Bordisms & Probes of Quantum Gravity

Memoria de Tesis Doctoral realizada por

Matilda Delgado

presentada ante el Departamento de Física Teórica de la Universidad Autónoma de Madrid para optar al Título de Doctor en Física Teórica

Tesis Doctoral dirigida por **Dr. Miguel Montero Muñoz**, Investigador del Instituto de Física Teórica UAM-CSIC

y **Dr. Ángel M. Uranga Urteaga**, Profesor de Investigación del Instituto de Física Teórica UAM-CSIC

Departamento de Física Teórica Universidad Autónoma de Madrid

Instituto de Física Teórica UAM-CSIC





Febrero de 2024

The following papers, some of them unrelated to the content of this document, were published by the candidate while this thesis was being developed:

- Global anomalies & bordism of non-supersymmetric strings, Ivano Basile, Arun Debray, M. Delgado, Miguel Montero JHEP 02 (2024) 092 arXiv:2310.06895 - Inspire
- Emergence of Species Scale Black Hole Horizons, José Calderón-Infante, M. Delgado, Angel M. Uranga JHEP 01 (2024) 003 arXiv:2310.04488 - Inspire
- Black holes as probes of moduli space geometry, M. Delgado, Miguel Montero, Cumrun Vafa JHEP 04 (2023) 045 arXiv:2212.08676 - Inspire
- 4. Elastic scattering of 3He+4He with SONIK, S.N. Paneru , C.R. Brune, D. Connolly et al. Phys.Rev.C 109 (2024) 1, 015802 arXiv:2211.14641 - Inspire
- Dynamical Cobordism and the beginning of time: supercritical strings and tachyon condensation,
 Roberta Angius, M. Delgado, Angel M. Uranga JHEP 08 (2022) 285 arXiv:2207.13108 - Inspire
- At the end of the world: Local Dynamical Cobordism, Roberta Angius, José Calderón-Infante, M. Delgado, Jesús Huertas, Angel M. Uranga JHEP 06 (2022) 142 arXiv:2203.11240 - Inspire
- Dynamical Cobordism and Swampland Distance Conjectures, Ginevra Buratti, José Calderón-Infante, M. Delgado, Angel M. Uranga JHEP 10 (2021) 037 arXiv:2107.09098 - Inspire
- Dynamical Tadpoles, Stringy Cobordism, and the SM from Spontaneous Compactification,
 Ginevra Buratti, M. Delgado, Angel M. Uranga
 JHEP 06 (2021) 170 arXiv:2104.02091 - Inspire

This thesis, written as a compendium of articles, investigates some broad consequences of allowing for the dynamical change of spacetime topology in string theory, in the context of the Swampland Program.

In the first four articles presented in this thesis, we investigate dynamical realizations of the cobordisms to nothing predicted by the Cobordism Conjecture. This comes down to considering solutions to the equations of motion where the compact space pinches off at finite distance in spacetime. We studied these setups in a wide variety of examples in string theory, which allowed us to characterize the backreaction of the singular extended objects (cobordism defects) that are located at the point of pinch-off. We found that they always explore infinite distances in moduli space, pointing to a link with the Distance Conjecture. This is not surprising from the perspective that these cobordisms represent dualities between any two effective field theories of quantum gravity in a general sense. It also means that the UV nature of these objects goes well beyond an EFT description. Furthermore, we showed that all of these solutions feature a generic behaviour, locally, near the singularity. These dynamical cobordisms are generally intricately linked to the presence of a scalar (tadpole) potential, which shows the importance of understanding the Distance Conjecture beyond proper moduli spaces. Finally, we discussed dynamical cobordisms where the solution runs along a time- or light-like dimension, and described known vacuum destroying bubbles in supercritical string theories as bordisms to nothing.

In the fifth article we discussed the identification and cancellation of global gauge/diffeomorphism anomalies in effective field theories of quantum gravity, where topology change should be allowed. We described how this can be done through the computation of an appropriate bordism group. We then implemented this for the three ten dimensional nonsupersymmetric and non tachyonic string theories, showing that they are free of anomalies. Anomaly inflow arguments also allowed us to shine light on the worldvolume theories of fivebranes in these theories.

Finally, we considered whether the various extended objects in string theory (like Dbranes, or the cobordism defects) could be used to probe UV physics, since they naturally warp spacetime and source a potential for the scalars. In the last two articles we discuss how to probe UV physics using black holes. We first showed that large BPS black holes in $4d \mathcal{N} = 2$ compactifications of type II string theories can allow us to probe the geometry of the underlying compact space solely by measuring its charges and its size. If one wants to probe the physics at the species scale where full-fledged quantum gravity comes into play, one needs to consider instead the smallest possible black hole that one can define in a theory. We considered small, singular black holes and showed that they developed a stretched species scale sized horizon when curvature corrections are taken into account near its core. This had the virtue of unifying the various definitions of the species scale. It also begs the question of whether or not full-fledged quantum gravity is *always* hidden behind a horizon.

After briefly motivating quantum gravity and string theory, we introduce bordism groups and outline the various aspects of the Swampland Program that will be needed in this thesis. After the articles, we finish with an overview of the results published in the papers. Esta tesis, escrita como un compendio de artículos, investiga algunas consecuencias amplias de permitir el cambio dinámico de la topología del espacio-tiempo en la teoría de cuerdas, en el contexto del Programa del Swampland.

En los primeros cuatro artículos presentados en esta tesis, investigamos realizaciones dinámicas de los cobordismos hacia la nada predichos por la Conjetura de Cobordismo. Esto implica considerar soluciones a las ecuaciones de movimiento donde el espacio compacto se pellizca a una distancia finita en el espacio-tiempo. Estudiamos estos escenarios en una variedad de ejemplos en la teoría de cuerdas, lo que nos permite caracterizar los objetos extendidos singulares (defectos de cobordismo) que se encuentran en el punto de pellizco. Encontramos que siempre exploran distancias infinitas en el espacio de módulos, apuntando a un vínculo con la Conjetura de Distancia. Esto no es sorprendente desde la perspectiva de que estos cobordismos representan dualidades entre cualquier par de EFTs de gravitación cuántica en un sentido general. También significa que la naturaleza UV de estos objetos va mucho más allá de la EFT. Además, mostramos que todas estas soluciones presentan un comportamiento genérico, localmente, cerca de la singularidad. Estos cobordismos dinámicos están generalmente vinculados a la presencia de un potencial escalar, lo que muestra la importancia de comprender la conjetura de distancia más allá de los espacios de módulos. Finalmente, discutimos cobordismos dinámicos donde la solución se desarrolla a lo largo de una dimensión temporal o tipo luz, y describimos burbujas que destruyen el vacío en teorías de cuerdas supercríticas como bordismos hacia la nada.

En el quinto artículo discutimos la identificación y cancelación de anomalías globales de gauge/difeomorfismo en teorías efectivas de campos de gravitación cuántica, donde se debería permitir el cambio de topología. Describimos cómo esto puede hacerse mediante el cálculo de un grupo de cobordismo apropiado. Luego lo implementamos para las tres teorías de cuerdas no supersimétricas y no taquiónicas de diez dimensiones, mostrando que están libres de anomalías. Los argumentos de "anomaly inflow" también nos permitieron arrojar luz sobre las teorías de worldvolume de las cincobranas en estas teorías.

Finalmente, consideramos si los diversos objetos extendidos en la teoría de cuerdas (como las D-branas o los defectos de cobordismo) podrían usarse para sondear la física UV, ya que naturalmente deforman el espacio-tiempo y generan un potencial para los escalares. En los dos últimos artículos discutimos cómo sondear la física UV usando agujeros negros. Primero mostramos que los agujeros negros grandes BPS en compactificaciones de 4d $\mathcal{N} = 2$ de teorías de cuerdas de tipo II nos permiten sondear la geometría del espacio compacto subyacente solo midiendo sus cargas y su tamaño. Si uno quiere sondear la física a la escala de especies donde la gravedad cuántica completa entra en juego, es necesario considerar en su lugar el agujero negro más pequeño posible que se pueda definir en una teoría. Consideramos agujeros negros pequeños y singulares y mostramos que desarrollan un horizonte del tamaño de la escala de especies cuando se tienen en cuenta las correcciones de curvatura cerca de su centro. Esto tenía la virtud de unificar las diversas definiciones de la escala de especies. También plantea la pregunta de si la gravedad cuántica completa está *siempre* oculta detrás de un horizonte.

Después de motivar brevemente la gravedad cuántica y la teoría de cuerdas, introducimos grupos de cobordismo y delineamos los diversos aspectos del Programa del Swampland que serán necesarios en esta tesis. Después de los artículos, terminamos con una descripción general de los resultados publicados en los artículos.

Acknowledgments

First and foremost, I am grateful to my advisors Ángel and Miguel not only for teaching me everything that I know about quantum gravity, but also for their continued friendliness, kindness and support throughout these four years. Your passion for physics is inspiring; hopefully, some of your creativity will have rubbed off on me.

Secondly, I would like to thank Roberta Angius, Ivano Basile, Ginevra Buratti, Arun Debray, Jesús Huertas and Cumrun Vafa for all that they have taught me during these friendly and productive collaborations. I would also like to acknowledge the hospitality of Harvard University during my stay there. A special thanks to José, for the numerous arguments, discussions and laughs.

A great many thanks to the IFT gang, the old and the new, for making the IFT such a fun and wholesome place to work at. Thanks to the 3Farteen for the many laughs: Gonzalo (Cosmo) for being a human calculator, Lorenzoni for being as bold as you are bald, and Alex for wearing pants. Thanks to all of the stringy/holo people for the serious and less serious discussions. Thanks to the eternal disfrutones Alberto, Alessandra, Edu, Carmen, Christian, David (Asturiano), Fer, Gonzalo (Stringy), Luca, Manu, Marina, Mattia, Michelangelo, Nacho, Pablo, Pau, Sara for the many manzanitas and cervecitas. Thanks as well as to Fran, Sergio, Walter, Xurxo for making so much noise across the hall. To Naredo; rest in pieces, see you never.

Thanks to the original gang, Alvaro, David, Dani, Raquel, Wax, for the lasting friendships.

Thank you to Anais, Amandine and Carlos for their love and support.

Merci à Pippa, pour être là et pour avoir été à mes côtés depuis le premier jour.

To my family, without whom none of this would have been possible; thank you, merci.

Contents

1	Motivation & Context					
	1.1	Aspec	ts of Quantum Gravity		• •	. 2
	1.2				• •	. 3
	1.3	1.3 The Swampland Program				. 7
		1.3.1	Bordisms and the Swampland		•	. 7
		1.3.2	The Distance Conjecture		•	. 14
		1.3.3	Probes of Quantum Gravity			. 15
	1.4	Outlin	e of the thesis		•	. 16
	Bibli	iograph	y			. 18
2	Dynamical Cobordism					23
3	Global Anomalies					153
4	Probes of Quantum Gravity					227
5	Global summary of results and discussion					285
6	Resumen global de los resultados y discusión					289

Motivation & Context

Quantum gravity (QG) stands at the edge of modern theoretical physics, aiming to reconcile the two main paradigms of modern physics: quantum mechanics and general relativity. While both theories are extremely successful in their respective domains, they conflict when applied together at very small length scales, such as those encountered near the singularities of black holes or during the earliest moments of the universe.

On the one hand, general relativity (GR) describes gravity as the curvature of spacetime caused by the presence of mass and energy. It provides a comprehensive framework for understanding the dynamics of massive objects, such as planets, stars, and galaxies. It has been extensively tested and confirmed through numerous astrophysical observations. On the other hand, quantum mechanics (QM) revolutionized our understanding of the microscopic nature of matter, describing the behavior of particles and fundamental forces with remarkable precision. Most of the time, these two drastically different theories do not clash since they operate at different scales: in most places in the universe, gravity is weak and can be treated classically with GR, providing a background on which to define the QM that describes the other forces. There are however isolated regions of space and time where the two theories clash. Schematically, these a regions where gravity is as strong as the other fundamental forces and the classical description of gravity by GR breaks down. For instance, in the early universe, during periods of extremely high energy and density, the classical description of spacetime by GR breaks down. Similarly, classical descriptions of gravity break down near a black hole's singularity. Understanding the behavior of matter and energy near black holes and at the beginning of the universe thus requires a quantum description of gravity.

The quest for a theory of quantum gravity has therefore become one of the central goals of contemporary theoretical physics. Such a theory would not only shed light on the earliest moments of cosmic evolution and on black hole singularities but more importantly provide a deeper understanding of the fundamental nature of space, time, and gravity. String theory, loop quantum gravity, and other approaches offer promising avenues for exploring these enigmatic questions.

Understanding the unification of GR and QM into a full-fledged quantum gravity theory remains a huge challenge to this day. The first obstacle is due to the fact that there are many conceptual differences between the two frameworks that make such a unification very subtle. We will now go over some of these aspects, in an effort to identify some of the characteristics that a quantum gravity theory should have.

1.1 Aspects of Quantum Gravity

Classical theories of physics are formulated with respect to a fixed background spacetime. For instance, in Newtonian mechanics, time is absolute, and space is considered an unchanging background along which particles move. In special relativity, while the laws of physics are invariant under Lorentz transformations, there is still a fixed Minkowski spacetime background. GR, however, revolutionized our understanding of gravity by introducing the idea that spacetime curvature is *caused* by the distribution of matter and energy. In GR, the gravitational field is encoded in the curvature of spacetime, and particles move along geodesics in this curved spacetime. The key feature of general relativity is its **background independence**. The theory does not presume any fixed spacetime background against which gravitational phenomena occur. Instead, the geometry of spacetime is determined dynamically by the distribution of matter and energy according to Einstein's equations.

In GR, spacetime is treated as a smooth, continuous manifold. In this framework, spacetime can be divided into infinitely small intervals, allowing for precise measurements and predictions of physical phenomena. However, QM introduces the idea of quantization, where physical quantities such as energy, momentum, and angular momentum are divided into discrete quanta. In the context of quantum gravity, the notion of quantized spacetime suggests that spacetime itself might be composed of fundamental building blocks or quanta at the Planck scale, which is the scale at which quantum gravitational effects become important. At that scale, quantized spacetime could give rise to quantum fluctuations in the fabric of spacetime itself. These fluctuations would manifest as **uncertainty in the geometry and topology of spacetime**. The idea that a quantum gravity theory should account for the possibility of changes in the topology of spacetime will play a fundamental role in this thesis.

If spacetime is quantized into discrete little blocks in full-fledged quantum gravity, then why is it that we observe the smooth background spacetime that we know and love? This apparent tension is resolved through the concept of the **emergence** of spacetime. Background independence asserts that the geometry of spacetime should not be imposed from the outside but should emerge from the fundamental principles of the theory itself. In this view, spacetime and gravitational dynamics emerge at low energies as collective phenomena from the underlying quantum degrees of freedom, similar to how fluid dynamics emerges from the collective behavior of molecules. This emergent nature of spacetime and dynamics has been studied in various contexts and there have been some attempts to make it into a quantitative statement in known theories of quantum gravity. We will come back to this in later sections.

The existence of a fundamental length scale associated with the quantization of spacetime nevertheless sets a limit on the resolution of spacetime measurements. Below this scale, the notion of continuous spacetime would lose its meaning. What would it mean to probe quantum gravitational effects in a physical (thought) experiment? In other words, if we could "see" a singularity in spacetime where quantum gravitational effects came into play, what would it look like? Another question is whether it is even possible to have such a naked singularity? Probing length scales smaller than the Planck length would require energies so high that they would generically collapse into a black hole due to the strong gravitational forces involved. This is a heuristic argument, but it goes in the favour of hiding quantum gravitational effects behind a horizon. Characterizing such black holes and

naked singularities and studying whether or not they can be used as **probes of quantum** gravitational effects will be a large part of this thesis.

Finally, one last aspect of quantum gravity that ties in with all that is mentioned above is **the holographic principle**, which suggests that spacetime itself may be an emergent phenomenon arising from the entanglement structure of a quantum field theory living on its boundary [1,2]. This principle was initially motivated by the study of black holes. Black holes possess an entropy proportional to their horizon area, suggesting a deep connection between the information content of a black hole and its surface area. This led to the proposal that the entropy of a black hole could be thought of as encoded on the horizon surface rather than within its volume. The holographic principle found its most profound realization in the AdS/CFT correspondence [3]. This correspondence states that the dynamics of a gravitational theory in an Anti de-Sitter spacetime can be fully described by the behavior of a quantum field theory on a fixed and flat spacetime, effectively encoding five-dimensional gravity in terms of a four-dimensional quantum field theory.

We have just described some features that a theory of quantum gravity should have. We saw that the notion of a background spacetime on which one can define quantum field theories and interactions should arise as a low energy description, at scales well bellow the Planck mass. This leads to a physical theory that one can compare to the laws of our universe, and perhaps even attempt to make predictions. In the next section, we will detail how this can be done in the context of a specific theory of quantum gravity: string theory. While loop quantum gravity offers intriguing insights and has made significant progress in addressing certain issues related to the quantization of gravity, it is hard to connect to real world physics since there are many subtleties in identifying a low-energy limit in which one can recover GR [4]. For this reason, our focus being on the low energy theories that stem from quantum gravity, we will base our discussion in string theory for the rest of this thesis.

1.2 Aspects of String Theory and the String Landscape

This section is meant to give a glimpse into the structure of string theory and outline how it exhibits the special features of quantum gravity theories described in the previous section. We will then describe how we can connect string theory to low-energy, four-dimensional physics. For a more in-depth introduction to string theory, see [5,6].

String theory is a theory of quantum gravity with renowned success. In its essence, it posits that the fundamental building blocks of the universe are not point-like particles but rather tiny, one-dimensional entities known as strings. String theory thus only has one fundamental scale, the string length l_s . These strings are assumed to propagate in a d-dimensional spacetime. In perturbative string theory, we assume that spacetime is flat Minkowski space and we quantize the oscillatory modes of the string and find its spectrum. All of the modes of the string describe particle states in spacetime. The massless modes lead (at the very least) to a d-dimensional graviton, a 2-form gauge field and scalar field ϕ , the *dilaton*, that parametrizes the coupling of the string $g_s = e^{\phi}$. All of the heavy modes of the string have masses at the string scale, they therefore decouple from physics at energy scales much lower than $m_s = l_s^{-1}$. We therefore obtain a d-dimensional low-energy effective field theory (EFT) that describes these three fields propagating on a flat spacetime.



Figure 1.1: A diagram describing our understanding of string theory according to the value of the string coupling g_s and the curvature of spacetime relative to the string scale. At weak coupling and low curvatures, we have 10D EFTs of string theory, describing various particles moving on the background described by general relativity. At strong curvatures, the worldsheet theory becomes strongly coupled but string perturbation theory is still well defined (the path integral over the moduli space of Riemann surfaces still makes sense). At strong coupling, string perturbation theory completely breaks down and we have to resort to other methods.

We can incorporate weakly curved spacetime into this discussion as a perturbation around the flat background. The big surprise that is responsible for some of the fame of string theory is that, in order for the worldsheet theory to remain consistent in curved spacetime, the metric has to satisfy Einstein's equations. In this way, GR arises naturally in the low-energy limit of string theory. Therefore, at low energies $E \ll l_s^{-1}$ and as long as spacetime is weakly curved, string theory is described by a low-energy effective theory that describes the dynamics of the 2-form gauge field, the dilaton and all other massless modes of the string, propagating on a background spacetime dictated by GR. In this limit, string theories thus reduce to theories that look like our world, though in a higher number of dimensions and possibly with a lot more supersymmetry.

For the simplicity of this discussion, we will mostly consider the eight known tendimensional non-tachyonic string theories, five of which lead to supersymmetric spacetimes [7-12]. There are other consistent string theories but they always have a tachyon in their spectrum, which signals an instability of the ten-dimensional vacuum. All eight of these theories, shown in black in Figure 1.2, reduce at low energies to ten-dimensional EFTs with matter and gauge fields minimally coupled to gravity.



Figure 1.2: A diagram representing the eight ten-dimensional non-tachyonic string theories in black and some arrows demonstrating how they are related to each other, to M-theory (in purple) and to tachyonic type 0 string theories (in gray).

Emergent Spacetime At low energies $E \ll l_s^{-1}$, a massless oscillatory mode of the string gives rise to a graviton in spacetime, which satisfies the equations of motion of GR in weakly curved spacetime. When one goes up in energies and considers string theory at the string scale, this effective description completely breaks down and one needs to take into account all of the massive oscillatory modes of the string. That is, you can not only consider the massless excitations which lead to familiar particle states in spacetime, you need to consider the string itself as the primary object of interest. In this sense, spacetime and GR only emerge at low energies. Importantly, at the string scale, you consider strings themselves moving in a flat "background", but in no shape or form is this theory describing GR. The better description would be in terms of some gas of strings, perhaps in the spirit of [13], or in terms of string field theory [14]. Even in the limit $g_s \to 0$, one would have to consider a non-interacting theory of strings moving around in flat space. When talking about the emergence of a *spacetime*, we always refer to the one that is governed by general relativity and that only exists in the low-energy limit of string theory.

Background Independence in String Theory Above, we discussed how we quantized the oscillations of the string on a flat background and showed at low-energies that this could be considered to be a flat (or weakly-curved) spacetime. This was for practical reasons, as quantizing the string on curved backgrounds is a very daunting task. From the perspective of the worldsheet, it is not obvious that string theory is background independent: because we picked a background to quantize it, background independence of the quantum theory is not manifest. However, in the low-energy EFTs, there are subtle ways in which the background independence of the full theory can be seen. For instance, there are dualities between string theories that map a string theory on a background X to another one (possibly the same one) on a different background Y, as is schematically depicted in Figure 1.2. The duality states that the spectrum of the two theories are exactly the same. This hints to the fact that the choice of background may be somewhat irrelevant. Furthermore, the most studied realization of the holographic principle is the AdS/CFT correspondence which equates the partition function of a string theory on an AdS spacetime with the partition function of a conformal field theory in flat space, in one dimension less. This means that all of the gravity in AdS can be recast in terms of CFT correlators in flat space. While the original formulation of the AdS/CFT correspondence involves AdS spacetime, there have been extensions and generalizations to other spacetime backgrounds. Furthermore, in holography the geometry of the extra (holographic) dimension is not meant to be fixed, and one technically has to sum over all geometries (even topologically different ones) that are compatible with the asymptotic boundary conditions. This versatility suggests that the essential features of the correspondence are not tied to any specific background geometry but rather reflect the holographic principle in a profound way. Finally, in non-perturbative formulations of string theory, such as string field theory, background independence could play a more prominent role (see e.g. [15]).

The String Landscape In order to make contact with our four-dimensional world, we need to compactify some of the extra dimensions. This comes down to wrapping some of them on a compact manifold of size l_{KK} and making $l_{KK}^{-1} \gg E$. This will have the effect of decomposing all of the ten-dimensional fields into their Fourrier modes: the massless modes survive the compactification and appear in the lower-dimensional theory whilst the heavy modes, often called Kaluza-Klein replicas, decouple from the lower-dimensional physics due to their mass $m_{KK} = l_{KK}^{-1} \gg E$. During the process of compactification, the parts of the ten-dimensional fields that propagate in the internal dimensions lead to new dynamical scalars in the lower-dimensional theory, called *moduli*. They parametrize the geometry of the compact manifold.

Through this process, from the ten-dimensional effective field theories that arise as the low-energy limit of string theories, we can recover four-dimensional effective field theories with gravity, gauge and matter fields, and an array of scalar fields that are either the dilaton or the moduli associated to the compact space. Do these theories look like our world? can we use them to make predictions from string theory?

The fact is that there are many (perhaps infinitely) different ways of performing this compactification, depending on which compact manifold we choose and what extra ingredients we add in along the way. The bigger problem is that only a few of these choices are simple enough to make the full compactification tractable mathematically and they all yield very different four dimensional low energy theories. These theories all describe general relativity, but have different particle contents, vacuum energy density and different symmetries. They are said to populate the Landscape of string theory. Indeed, although compactifying string theory to four dimensions is well understood conceptually, it is very hard to perform in mathematical detail except for some isolated cases that preserve a lot of (super)symmetries. A lot of progress has been made in recent years to find detailed, top-down examples of compactifications that lead to realistic universes, but it is fair to say that very few, if any at all, are under full computational control.

1.3 The Swampland Program

Faced with this computational difficulty, one can ask whether or not it is possible to obtain any low energy theory from string theory. In recent years, this has been shown not to be true. There are some theories that simply cannot be obtained from string theory at low energies. Identifying the characteristics that separate these theories from the ones that can has been the whole premise of the Swampland Program [16]. The Swampland Program aims to distinguish between effective theories that arise from a consistent theory of quantum gravity - those in the "Landscape" - and those that do not, that are said to be in the "Swampland". In other words, the Swampland Program provides criteria that effective field theories must satisfy to be considered as low-energy limits of a consistent theory of quantum gravity, such as string theory. EFTs that are in the Swampland are therefore thought to be inconsistent with fundamental principles of a UV-complete theory of quantum gravity. The Swampland Program has generated significant interest and research activity within the string theory community and beyond, as it provides valuable insights into the structure of quantum gravity and the landscape of effective field theories that truly stem from it.

The Swampland Conjectures, which form the basis of the program, are the criteria that segregate the EFTs in the landscape from those in the swampland. They are obtained in general by noticing patterns in theories that are known to stem from top-down constructions in string theory. Sometimes we can trace back these patterns to a fundamental aspect of quantum gravity theories such as background independence. There are also occasions where they can be motivated by black hole physics. A subset of these conjectures are on such good theoretical standing that they can be proven in (near-)generality in string theory. We will now outline some of the conjectures and Swampland-related themes that will be relevant for this thesis. Along the way we will point out how they resonate with some of the fundamental features of quantum gravity theories discussed above. For a more complete review of the Swampland, see e.g. [17, 18].

1.3.1 Bordisms and the Swampland

In this section, we first introduce bordism groups for the layman, in the hopes of providing a bit of clarity on what kind of information these groups contain. Then, we will use them in the context of the cobordism conjecture [19] and later in the context of identifying and classifying global anomalies in theories with dynamical gravity.

1.3.1.1 Bordism Groups for dummies

In this chapter we provide a brief introduction to bordism groups for readers that have little (or no) knowledge of algebraic topology. For a mathematical introduction on bordism groups see [20]. For more in-depth discussions with a special focus on physical applications, see [19, 21, 22].

Bordism groups are elaborate mathematical objects used in algebraic topology to classify closed manifolds up to a certain equivalence relation called bordism. Put simply, a bordism groups collect equivalence classes of closed manifolds of a given dimension, considering two manifolds X_d into \tilde{X}_d to be equivalent if there exists a higher-dimensional manifold W_{p+1} whose boundary is the disjoint union of X_d into \tilde{X}_d . This is depicted schematically in Figure 1.3a. This higher dimensional manifold parametrizes the deformation of X_d into \tilde{X}_d .



Figure 1.3: a) X_d and \tilde{X}_d are bordant to eachother, they are in the same equivalence class. b) X_d is in the same equivalence class as the neutral element. X_d is said to be bordant to nothing.

Schematically, the bordism group Ω_d is the set of equivalence classes of d-dimensional manifolds, where two manifolds are considered equivalent if they are bordant to each other:

$$\Omega_d \cong \{ d\text{-dimensional manifolds } X_d \} / \text{bordism}$$
(1.1)

Group Structure Bordism groups Ω_d have an abelian group structure, with addition under the disjoint union of manifolds. There is therefore a neutral element, the equivalence class of "nothing". This means that some manifolds X_d can be bordant to nothing, which is to say that they are boundaries. In that case, there is a higher dimensional manifold W_{p+1} that parametrizes how X_d can be shrunk into a point, as depicted in Figure 1.3b. The inverse is given by an appropriate orientation-reversed version of X_d such that their disjoint union is bordant to nothing, as depicted in Figure 1.4b.



Figure 1.4: a) The disjoint union of X_d and Y_d is bordant to X_d . b) The inverse of X_d is defined as the manifold that is such that the disjoint union of X_d and its inverse is bordant to nothing.

Structure Things get a bit more complicated when you want to consider bordisms between manifolds that satisfy extra conditions. So far, we have not specified anything about X_d except for the fact that it is a manifold, this is the most basic form of bordism, known as unoriented bordism. We can also consider the bordism of oriented manifolds in which case we call it oriented bordism. More generally, we can consider the bordisms of manifolds with a specific *structure*. When there is additional structure, the notion of cobordism must be formulated more precisely: the structure has to extend to the W_{d+1} dimensional manifolds that defines the bordism.

Defining a structure intuitively comes down to finding a way to translate a bundle of a topological group G into a vector bundle on the entire tangent space of X_d . For instance, for G = SO, a structure is equivalent to a choice of orientation of X_d . For G = Spin, it is a choice of spin structure. Let us denote a generic structure by a ξ -structure. When defined formally, a ξ -structure on W_{d+1} is known to induce a structure on its boundary. In that way, we can define bordisms between manifolds X_d and \tilde{X}_d with ξ -structure through a manifold W_{d+1} with a ξ -structure.

It can help to think of adding structure as trivializing a new class of topological invariants. For instance, for a d-manifold to be orientable, its first Stiefel-Whitney class [23] must vanish, and on top of that, for it to be a spin manifold, its second Stiefel-Whitney class must vanish as well. Once the vanishing of these classes is imposed, the manifold under consideration can be shown to admit a spin structure. The point is that there are various spin structures for each dimension d, and this is exactly what the spin bordism groups quantify.

Generators and Invariants If the bordism group under consideration Ω_d^{ξ} vanishes, then all d-manifolds with ξ -structure can be deformed into one another and into nothing. However, when they are not trivial, they generically take the form:

$$\Omega_d^{\xi} \cong \underbrace{\mathbb{Z} \times \cdots \times \mathbb{Z}}_{r \text{ times}} \times \underbrace{\mathbb{Z}_{p_1} \times \mathbb{Z}_{p_2} \cdots}_{p \text{ times}}$$
(1.2)

for p^i non-negative integers. This decomposition is true in all cases of interest in this thesis but there is no theorem proving that is it true in general. The first part correspond to the free part of the group, whilst the second part corresponds to torsional classes. Manifolds that are part of a torsional class indexed by \mathbb{Z}_{p^i} are such that only the disjoint union of p^i copies of them is bordant to nothing. Equation (1.2) means that any d-manifold with ξ -structure is in an equivalence class of bordism that can be indexed by r integers and pfractions of integers given by $1/p^i$.

To see where these integers (and fractions of integers) are coming from, we need to consider bordism invariants. These are topological invariants that, when evaluated on a manifold, yield (one of) the aforementioned indices. With a complete set of r + p bordism invariants, one can identify precisely in what bordism equivalence class a manifold belongs to. In particular, these invariants help in identifying the generators of the bordism group: the set of d-manifolds that are representative of the bordism class under consideration. For instance, Pontryagin numbers and Stiefel-Whitney numbers [23] are oriented cobordism invariants; they determine an oriented manifold's oriented cobordism class completely.

Determining the bordism invariant is in general easier said than done, they are not always simple integrals of characteristic classes of some tangent or gauge bundle. For instance, for spin bordism they are more complicated quantities known as eta-invariants of Dirac operators. In such cases, finding the relevant generator is not always easy.

Bordisms and Cohomology To a bordism theory Ω^{ξ} , we can associate a generalized cohomology theory, with homology groups $\Omega_d^{\xi}(W)$ for any space W. $\Omega_d^{\xi}(W)$ describes the set of equivalence classes of d-manifolds X_d equipped with a map into W. If W is a point then we fall back on the bordism group described above. Generalized cohomology

theories share all of the axioms of a cohomology theory except for the dimension axiom: the bordism groups of a point are not trivial for d > 0.

In physical setups, we often have to describe the bordisms of d-manifolds with ξ structure and with a non trivial G-bundle, which one can imagine to be some sort of gauge bundle. In this case, the bordism groups one has to consider are $\Omega_d^{\xi}(BG)$ where BG is the classifying space of G which classifies the various physically distinct G-bundles. Importantly, even if we know all of the $\Omega_d^{\xi} = \Omega_d^{\xi}(pt)$ and the homology of the space W, this does not mean that we can compute $\Omega_d^{\xi}(W)$. The Atiyah-Hirzebruch spectral sequence however gives a starting point for these calculations by reducing the computation of $\Omega_d^{\xi}(W)$ to that of various homology groups on W (for more information on this, see e.g. [24]).

Final Remarks

- For many types of ξ -structure (but not all of them) $\Omega_*^{\xi} = \bigoplus_d \Omega_d^{\xi}$ is a \mathbb{Z} -graded ring. This is the case for oriented bordism, spin bordism, string bordism etc. but not for pin[±] bordism. That means that on top of abelian group structure described above, they allow for multiplication under the cartesian product of two manifolds. The ring is \mathbb{Z} -graded by the dimensions of the manifolds.
- Given the discussion above, one might wonder what is the difference between considering $\Omega_d^{\xi}(BG)$ and considering $\Omega_x^{\xi+BG}(pt)$ where the G-bundle is added to the structure. The difference is that adding structure can in general involve intertwining the G-bundle with the tangent bundle of X_d . Intuitively, the difference is the same as the difference between considering a manifold with spin structure and considering a spinor bundle on a manifold. One is a characteristic of the tangent bundle everywhere, and the other is just a localized bundle.

In the next sections we will see how bordism groups can be used in the context of QG and the Swampland Program.

1.3.1.2 The Cobordism Conjecture

The cobordism conjecture [19] applies to string theories compactified on a p-dimensional manifolds X_p . The conjecture states that any two backgrounds must equivalent in a precise sense described by the mathematical construction of bordism groups. In this way, it speaks to the fundamental nature of quantum gravity: that its observables should be *background independent*. It quantifies the fact that there should always exist a physical process in quantum gravity that can deform any background into any other.

Saying that one should be able to deform any and all p-dimensional backgrounds into one another in quantum gravity comes down to saying that all p-dimensional backgrounds are part of the same equivalence class in bordism. If all p-dimensional manifolds are in the same bordism class, then they have to be in the class of "nothing", i.e. the trivial class. The cobordism conjecture can therefore be stated as:

The Cobordism Conjecture

Consider a D-dimensional QG theory compactified on a d-dimensional compact manifold, all bordism classes must be trivial:

$$\forall d \text{ such that } d \le D, \ \Omega_d^{QG} = 0.$$
(1.3)

From the perspective of the lower-dimensional effective field theories obtained from the compactification, this means that any two of these theories can be connected by a domain wall as in Figure 1.5a and that any such theory should admit a boundary, as in Figure 1.5b.



Figure 1.5: Bordisms from the perspective of the lower dimensional EFTs one gets after compactifying on X_d and \tilde{X}_d . A bordism between X_d and \tilde{X}_d looks like a domain wall (a) and a bordim to nothing looks like a boundary (b).

The difficulty here is in defining what are the topology-changing processes allowed in full-fledged quantum gravity. Of course, we do not know the whole set of such processes. In mathematics, bordism are usually considered between manifolds with specific structures. For instance, spin bordism groups Ω_d^{spin} describe the bordisms between d-dimensional manifolds equipped with a spin structure. Ideally, we would know what kind of structure is appropriate to describe quantum gravity backgrounds but for that we would need a full-fledged, non-perturbative description of string theory. Instead, we consider much simpler bordism groups that are only meant to approximate those of full-fledged quantum gravity. For instance, we know that the low-energy EFTs we get from string theory generically contain fermions. This means that they should only be compactified on manifolds that admit a spin structure. We can therefore consider spin bordism as a first approximation of Ω_d^{QG} . As we go along and further constrain the type of structure that our theories should admit, we can refine our description in terms of more elaborate bordism groups that account for this added structure.

If one of these approximate bordism groups are non-trivial, it only signals the fact that the bordism groups we consider are too poor an approximation of Ω_d^{QG} and that new structure needs to be added to the theory in order to trivialize these classes. In this way, the cobordism is predicting the existence of new structure in the theory that should trivialize all the classes:

$$\Omega_d^{approx.} \neq 0 \Longrightarrow \Omega_d^{approx+new \ structure} = 0.$$
(1.4)

More precisely, the presence of a non-trivial bordism class would signal the presence of an uncancelled global topological charge in the compact space, given by the bordism invariants, that prevent you from shrinking it to a point:

$$Q = \int_{X_d} [\text{topological OP}]_d \tag{1.5}$$

where we have represented the bordism invariant as an integral of some sort of characteristic class over the compact space, but it can be something more exotic, as discussed in the previous section. The presence of this topological operator supported on X_d signals the presence of a (D - d - 1)-form global symmetry (see e.g. [25] for a review on higher-form symmetries). This global charge has the effect of putting a "label" on the background X_d , which should not be allowed in a background independent theory. The statement that $\Omega_d^{QG} = 0$ thus means that there must be new structure in quantum gravity that gets rid of this global symmetry.¹ How can we get rid of a global symmetry? There are two options:

1. Breaking the symmetry: this amounts to introducing a defect that is charged under this symmetry and can thus absorb the uncancelled topological charge in the compactification. The objects that are charged under the (D - d - 1)-form global symmetry generated by this charge are (D - d - 1)-dimensional defects.

A simple example of this was discussed in [19]. It is the case of the one-dimensional spin bordism group

$$\Omega_1^{spin} = \mathbb{Z}_2, \qquad (1.6)$$

which is generated by the circle with periodic spin structure. A fermion with antiperiodic boundary conditions on this circle creates a configuration that you cannot deform to a point. In the case of the ten-dimensional heterotic string theories, their compactification on the circle with periodic string structure should be perfectly consistent, so this symmetry has to be broken by an eight-dimensional defect, which has yet to be identified.

2. Gauging the symmetry: this amounts to introducing a dynamical (D - d)-form that couples to (the Hodge dual of) the (D - d)-form current associated to the symmetry. Doing so in a consistent way eventually leads to new consistency conditions on the theory.

A simple example of this was discussed in [19]. It is the case of the four-dimensional spin bordism group

$$\Omega_4^{spin} = \mathbb{Z}\,,\tag{1.7}$$

which is generated by the K3 surface. Gauging this symmetry means to couple it to a 6-form gauge field. Consider heterotic string theories turning off all gauge bundles

¹As a side note, this shows how the cobordism conjecture can thus be seen as a direct implication of a different, more general Swampland conjecture: the no global symmetries conjecture [26]. In a few words, this conjecture states that any global symmetry should be gauged or broken in quantum gravity. When applied to topological symmetries such as that generated by (1.5), it reduces to the cobordism conjecture.

and forgetting about the branes. Consistency of the theory requires this 6-form to be the magnetic NSNS field, B_6 , and leads to the observation that K3 carries five units of NS5 brane charge. This is the known fact that compactifying heterotic string theories on K3 is inconsistent on its own. Therefore, one can either choose to ignore K3 completely, or to refine the description by considering heterotic string theories with gauge bundles and/or five-branes that can cancel this charge. The first option comes down to considering twisted-string bordism $\Omega_4^{twisted-string} = 0$ and the second comes down to breaking the symmetry by introducing NS5 branes in the compactification to cancel the charge. The second option seems the most natural, given that we know heterotic string theories have gauge fields and five-branes in their spectrum.

The new (D - d - 1)-dimensional defects predicted by the cobordism conjecture can in general be highly non-supersymmetric and unstable and will most likely back-react on the geometry in a singular way. Isolating the properties of these objects from the EFT's perspective is thus a very daunting task. For examples of direct applications of the cobordism conjecture (at the topological level), see for instance [22, 27–36]).

In particular, even in cases where the bordism classes are trivial and we need not introduce a new defect, describing how the domain walls between EFTs (or equivalently, the boundaries, since you can glue two boundaries to make a domain wall) predicted by the cobordism conjecture should appear in spacetime is difficult. This can be seen from the simple fact that you can have two completely different string theories with different degrees of freedom on each side. The domain wall has to be such that it converts all of the degrees of freedom of one theory into that of the other. In this way, the domain wall is implementing a duality in a general sense. This means that its physics are highly nonperturbative and therefore out of reach of the low-energy physics of the EFT. From this, one can already see that such objects will generally be singular from the perspective of the EFT.

The work in Chapter 2 of this thesis takes a step in the direction of describing these new defects and how they backreact on spacetime. We now turn to another corner of the Swampland Program where bordism groups play and important role.

1.3.1.3 Bordisms and Global Anomalies

Bordism groups also play an important role in the identification of global gauge and diffeomorphism anomalies in EFTs of quantum gravity. Anomaly cancellation is important in any consistent theory, but there are specific types of anomalies that arise when topologychanging processes should be allowed.

Consider an anomaly in a theory on a spacetime X_d , that is, consider a gauge transformation or a diffeomorphism under which the path integral is not invariant. When this anomalous transformation can be made arbitrarily small (close to the identity), we call the anomaly "local". Local anomalies are the most problematic type of gauge/gravitational anomaly since the path integral is not invariant under an infinitesimal transformation. They are the sort of anomaly that can be computed using triangle Feynman diagrams (in four dimensions). These anomalies are reviewed in [37], using the modern formalism of the anomaly polynomial in two dimensions more. When local anomalies are shown to vanish, we are left with global anomalies. These are anomalies associated to gauge transformations and diffeomorphisms that cannot be deformed to the identity. A famous example of a global gauge anomaly is Witten's SU(2) anomaly [38]. The modern way of computing global anomalies is through what is known as an anomaly theory, which lives on a (d+1)-dimensional manifold whose boundary is the spacetime manifold X_d . This theory is not meant to be describing a physical theory in one dimension more, instead, it is thought of as a means to parametrize the anomalous global gauge transformation. Indeed, the anomaly theory is engineered to give exactly the opposite anomaly of that of the path integral, such that when coupled together, the total anomaly cancels (for a pedagogical review, see [21]).

It turns out that is it can be easier to compute the anomaly using the anomaly theory than it is to compute it in the partition function. The reason, is that when local anomalies cancel, global anomalies become bordism invariants of the (d+1)-dimensional spaces. This means that detecting a potential global anomaly can be done "simply" by computing the relevant bordism groups in d+1 dimensions. Of course, this can be easier said than done, depending on the structure we have to impose on the bordism groups.

The anomaly theory on the higher-dimensional manifold Y_{d+1} parametrizes the space of gauge and metric configurations on X_d . In particular, the space Y_{d+1} can have holes and accounts for topology change of X_d . Evaluating the path integral on a topology changing transition is impossible without a non-pertubative description of QG since these processes are usually singular from the perspective of the EFT. The bordism groups thus shed light on the consistency conditions that arise from global anomaly cancellation in theories with dynamical gravity. This can be viewed as a bottom-up approach to the Swampland Program.

Chapter 3 will describe how this works in more detail. In particular, we will show how bordism groups were used to assess whether or not global anomalies cancel in the three ten-dimensional non-supersymmetric and non-tachyonic string theories in Figure 1.2. This ends our introduction to bordism groups in a Swampland context, we now turn to introducing another seemingly unrelated Swampland conjecture.

1.3.2 The Distance Conjecture

Before stating the distance conjecture, let us recall that there are two kinds of scalar fields in EFTs that come from string theory.

Firstly there is the dilaton, that exists already in ten dimensions and therefore also in any EFT obtained from compactifying the ten-dimensional ones. It is linked to the string coupling as $g_s = e^{\phi}$. When the dilaton goes to $\pm \infty$, our string theory becomes infinitely weakly or strongly coupled. These are limits that are usually linked to string dualities. In these limits, the string theory we started with is leaving its regime of validity and the degrees of freedom of the dual string become the appropriate degrees of freedom to describe the process at low energies. When $g_s \to 0$, they are the modes of the string theory we started with, and when $gs \to \infty$ they are usually the modes of a dual string. For example, one can see in Figure 1.2 that type I at strong coupling is dual to SO(32) string theory at infinitely weak coupling. In both cases, there is a tower of heavy string modes that becomes light in both of these limits, that of the string at strong coupling, and that of the dual string at weak coupling. Then, there are the moduli that arise only after compactification, that parametrize the geometry of the compact manifold. For instance, if you compactify one of the ten dimensional theories on a circle or radius R, you will obtain a dynamical scalar in the ninedimensional theory that is linked to R as follows: $\sigma \sim \log R$. This scalar can also go to $\pm \infty$ in the nine-dimensional spacetime. We see that if $\sigma \to +\infty$, $R \to \infty$ and so we observe a decompactification of the circle. In this limit, the nine-dimensional EFT breaks down completely as all of the KK modes become light. If $\sigma \to -\infty$, then $R \to 0$ and the circle is pinching off. In these limits, stringy physics are at play, since the better description is that of the T-dual string theory compactified on a radius $R^{-1} \to \infty$. Therefore the pinching off in the original frame is a decompactification in the T-dual frame, and the tower of states becoming light is that of the KK modes in the T-dual frame. These modes correspond to winding states in the original frame. T-duality is where the stringy physics comes in to play: the string being an extended objects can not only propagate in internal dimensions but also wrap around them.

The moral of the story is that no matter what scalars you consider in your EFTs of string theory, there is always an infinite tower of states becoming light when you go to infinite distance in scalar field space,² that signal the break down of the EFT. This is exactly the statement of the distance conjecture. It is motivated by a wide variety of examples in string theory, where in these limits, there is always an infinite tower of massive states getting exponentially light. The conjecture is stated as follows:

The Distance Conjecture

In any effective field theory of quantum gravity, there is a tower of states becoming light as one goes to infinite distance in field space $\varphi \to \infty$ with characteristic mass m [39]:

$$\frac{m}{M_p} \sim e^{-\lambda\varphi}, \ \lambda \in O(1).$$
 (1.8)

The reason for the exponential dependence on distance in field space is motivated by the two types of towers that we observe in string theory: KK towers and string towers, which both have masses that depend exponentially on the corresponding moduli. There is in fact a conjecture that those are the only two options in string theory [40]. The SDC has been discussed in many examples in string theory, see for instance [39–61].

From the discussion above, it is clear that infinite distance limits in field space are where the UV physics happen. They are always taking us to regions of the parameter space of our theory where the EFT is at the limit of its range of validity. When they are not simple decompactification limits, these infinite distance limits generically take us to a frame where the dual theory is a better low-energy description than the one we started with.

1.3.3 Probes of Quantum Gravity

It is of obvious interest to quantum gravity phenomenologists to be able to probe quantum gravitational effects in a physical experiment. One option would be to exploit the extended objects that naturally exist in string theory. These can be black holes, (black-) branes,

 $^{^{2}}$ Importantly, note that in the context of the distance conjecture, we always consider canonically normalized kinetic terms for the scalar fields, in the Einstein frame.

they can be singular, or have a horizon. A simple example of this are D-branes in type II theories. In full-fledged string theory they correspond to the end points of open strings and are very well understood. In the low-energy EFTs of string theory, they arise as non-perturbative objects that back-react on gravity in a very singular way. All branes except the D3 are the location where both the curvature and the dilaton blow up, reflecting the fact that D-branes cannot be fully described in the EFT.

Interestingly, these D-branes as well as other examples of UV (cobordism) defects (see Chapter 2) are generally located at infinite distance in moduli space, showcasing the fact that they are better described in a dual frame. This also explains how the domain walls between theories predicted by the cobordism conjecture back-react so singularly on spacetime, they are intrinsically linked to dualities, which in turn comes from the background independence of String Theory.

Being that they are singular at their core, these objects are by definition not controlled probes of UV physics. It is tempting to take a black hole instead or a black brane, whose horizon is large enough such that curvatures are small and corrections can be ignored. These large black holes can in fact constitute controlled probes, and they will be used in Chapter 4 this thesis to uncover topological properties about the underlying compactification.

However, if one wants to probe physics at such high energies that the true effects of full-fledged string theory come into play, one ought to use more singular probes. In fact one might even wonder whether it makes sense to probe such an energy scale, since spacetime and dynamics completely break down there. Let us first ask what this scale actually is. The naive answer is to assume it is the Planck scale, since that is when loop corrections to the graviton propagator become relevant. In reality, there is a subtlety to this argument: in a theory with light particle spaces, this scale is actually [62,63]:

$$\Lambda_{sp} = \frac{M_{pl}}{N^{\frac{1}{d-2}}} \tag{1.9}$$

This energy scale has been dubbed "Species scale" in the literature. The point is, that in the presence of a large number of light species, the scale at which full-fledged quantum gravity comes into effect can be a lot lower than the Planck scale. With the distance conjecture in mind, we see that the species scale gets lowered at infinite distance in moduli space. This means that the singular probes discussed above (D-branes and other cobordism defects) are regions of spacetime where the species scale is greatly lowered. Identifying the tower and how fast the species scale becomes light in these limits has been studied in [61,64–66]. The species scale has also been studied in the interior of moduli space in [67–69].

One may wonder then whether or not we could use these probes to ride our way up all the way to the Species scale. In Chapter 4 we will discuss this and see that curvature corrections near the singular core might always hide the full-fledged quantum gravity effects behind a species-scale sized horizon.

1.4 Outline of the thesis

After this general introduction, the second part of the thesis contains the collection of seven articles in the published versions. Four of them should be read one after the other, as they pertain to the same subject: dynamical realizations of cobordisms in spacetime.

They are presented in Chapter 2. Chapter 3 contains one stand-alone article, on the topic of bordisms and global gauge and gravitational anomalies in non-supersymmetric string theories. The last two articles are presented together in Chapter 4 as they both deal with probing UV effects with extended objects. We leave the conclusions for the last part of the thesis, in Chapter 6 where the main results of these works are summarized.

Bibliography

- G. 't Hooft, Dimensional reduction in quantum gravity, Conf. Proc. C 930308 (1993) 284 [gr-qc/9310026].
- [2] L. Susskind, The World as a hologram, J. Math. Phys. 36 (1995) 6377 [hep-th/9409089].
- [3] J. M. Maldacena, The Large N limit of superconformal field theories and supergravity, Adv. Theor. Math. Phys. 2 (1998) 231 [hep-th/9711200].
- T. Thiemann, Modern Canonical Quantum General Relativity, Cambridge Monographs on Mathematical Physics. Cambridge University Press, 2007, 10.1017/CBO9780511755682.
- [5] D. Tong, String Theory, 0908.0333.
- J. Polchinski, String theory. Vol. 1: An introduction to the bosonic string, Cambridge Monographs on Mathematical Physics. Cambridge University Press, 12, 2007, 10.1017/CBO9780511816079.
- J. Polchinski, String theory. Vol. 2: Superstring theory and beyond, Cambridge Monographs on Mathematical Physics. Cambridge University Press, 12, 2007, 10.1017/CBO9780511618123.
- [8] L. Alvarez-Gaume, P. H. Ginsparg, G. W. Moore and C. Vafa, An O(16) x O(16) Heterotic String, Phys. Lett. B 171 (1986) 155.
- [9] L. J. Dixon and J. A. Harvey, String Theories in Ten-Dimensions Without Space-Time Supersymmetry, Nucl. Phys. B 274 (1986) 93.
- [10] A. Sagnotti, Some properties of open string theories, in International Workshop on Supersymmetry and Unification of Fundamental Interactions (SUSY 95), pp. 473-484, 9, 1995, hep-th/9509080.
- [11] A. Sagnotti, Surprises in open string perturbation theory, Nucl. Phys. B Proc. Suppl. 56 (1997) 332 [hep-th/9702093].
- S. Sugimoto, Anomaly cancellations in type I D-9 anti-D-9 system and the USp(32) string theory, Prog. Theor. Phys. 102 (1999) 685 [hep-th/9905159].
- [13] R. H. Brandenberger and C. Vafa, Superstrings in the Early Universe, Nucl. Phys. B 316 (1989) 391.
- [14] E. Witten, On background independent open string field theory, Phys. Rev. D 46 (1992) 5467 [hep-th/9208027].
- [15] A. Sen and B. Zwiebach, Background independent algebraic structures in closed string field theory, Commun. Math. Phys. 177 (1996) 305 [hep-th/9408053].
- [16] C. Vafa, The String landscape and the swampland, hep-th/0509212.
- [17] E. Palti, The Swampland: Introduction and Review, Fortsch. Phys. 67 (2019) 1900037 [1903.06239].

- [18] M. van Beest, J. Calderón-Infante, D. Mirfendereski and I. Valenzuela, Lectures on the Swampland Program in String Compactifications, 2102.01111.
- [19] J. McNamara and C. Vafa, Cobordism Classes and the Swampland, 1909.10355.
- [20] R. E. Stong, Notes on Cobordism Theory. Princeton University Press, Princeton, 1968, doi:10.1515/9781400879977.
- [21] I. n. García-Etxebarria and M. Montero, Dai-Freed anomalies in particle physics, JHEP 08 (2019) 003 [1808.00009].
- [22] A. Debray, M. Dierigl, J. J. Heckman and M. Montero, The Chronicles of IIBordia: Dualities, Bordisms, and the Swampland, 2302.00007.
- [23] J. Milnor and J. D. Stasheff, *Characteristic Classes (AM-76)*, Volume 76. Princeton: Princeton University Press, 1974.
- [24] J. McCleary, A Userâs Guide to Spectral Sequences, Cambridge Studies in Advanced Mathematics. Cambridge University Press, 2 ed., 2000.
- [25] L. Bhardwaj, L. E. Bottini, L. Fraser-Taliente, L. Gladden, D. S. W. Gould,
 A. Platschorre et al., *Lectures on generalized symmetries*, *Phys. Rept.* **1051** (2024) 1
 [2307.07547].
- [26] T. Banks and L. J. Dixon, Constraints on String Vacua with Space-Time Supersymmetry, Nucl. Phys. B307 (1988) 93.
- [27] M. Dierigl, J. J. Heckman, M. Montero and E. Torres, IIB string theory explored: Reflection 7-branes, Phys. Rev. D 107 (2023) 086015 [2212.05077].
- [28] J. Kaidi, K. Ohmori, Y. Tachikawa and K. Yonekura, Nonsupersymmetric Heterotic Branes, Phys. Rev. Lett. 131 (2023) 121601 [2303.17623].
- [29] D. Andriot, N. Carqueville and N. Cribiori, Looking for structure in the cobordism conjecture, SciPost Phys. 13 (2022) 071 [2204.00021].
- [30] I. n. García Etxebarria, M. Montero, K. Sousa and I. Valenzuela, Nothing is certain in string compactifications, JHEP 12 (2020) 032 [2005.06494].
- [31] H. Ooguri and T. Takayanagi, Cobordism Conjecture in AdS, 2006.13953.
- [32] M. Montero and C. Vafa, Cobordism Conjecture, Anomalies, and the String Lamppost Principle, JHEP 01 (2021) 063 [2008.11729].
- [33] Y. Hamada and C. Vafa, 8d supergravity, reconstruction of internal geometry and the Swampland, JHEP 06 (2021) 178 [2104.05724].
- [34] R. Blumenhagen and N. Cribiori, Open-Closed Correspondence of K-theory and Cobordism, 2112.07678.
- [35] M. Dierigl and J. J. Heckman, Swampland cobordism conjecture and non-Abelian duality groups, Phys. Rev. D 103 (2021) 066006 [2012.00013].
- [36] M. Dierigl, J. J. Heckman, M. Montero and E. Torres, R7-branes as charge conjugation operators, Phys. Rev. D 109 (2024) 046004 [2305.05689].

- [37] L. Alvarez-Gaume and P. H. Ginsparg, The Structure of Gauge and Gravitational Anomalies, Annals Phys. 161 (1985) 423.
- [38] E. Witten, An SU(2) Anomaly, Phys. Lett. B 117 (1982) 324.
- [39] H. Ooguri and C. Vafa, On the Geometry of the String Landscape and the Swampland, Nucl. Phys. B766 (2007) 21 [hep-th/0605264].
- [40] S.-J. Lee, W. Lerche and T. Weigand, Emergent strings from infinite distance limits, JHEP 02 (2022) 190 [1910.01135].
- [41] R. Blumenhagen, I. Valenzuela and F. Wolf, The Swampland Conjecture and F-term Axion Monodromy Inflation, JHEP 07 (2017) 145 [1703.05776].
- [42] T. W. Grimm, E. Palti and I. Valenzuela, Infinite Distances in Field Space and Massless Towers of States, JHEP 08 (2018) 143 [1802.08264].
- [43] S.-J. Lee, W. Lerche and T. Weigand, Tensionless Strings and the Weak Gravity Conjecture, JHEP 10 (2018) 164 [1808.05958].
- [44] S.-J. Lee, W. Lerche and T. Weigand, A Stringy Test of the Scalar Weak Gravity Conjecture, Nucl. Phys. B 938 (2019) 321 [1810.05169].
- [45] T. W. Grimm, C. Li and E. Palti, Infinite Distance Networks in Field Space and Charge Orbits, JHEP 03 (2019) 016 [1811.02571].
- [46] G. Buratti, J. Calderón and A. M. Uranga, Transplanckian axion monodromy!?, JHEP 05 (2019) 176 [1812.05016].
- [47] P. Corvilain, T. W. Grimm and I. Valenzuela, The Swampland Distance Conjecture for Kähler moduli, JHEP 08 (2019) 075 [1812.07548].
- [48] S.-J. Lee, W. Lerche and T. Weigand, Modular Fluxes, Elliptic Genera, and Weak Gravity Conjectures in Four Dimensions, JHEP 08 (2019) 104 [1901.08065].
- [49] A. Joshi and A. Klemm, Swampland Distance Conjecture for One-Parameter Calabi-Yau Threefolds, JHEP 08 (2019) 086 [1903.00596].
- [50] F. Marchesano and M. Wiesner, Instantons and infinite distances, JHEP 08 (2019) 088 [1904.04848].
- [51] A. Font, A. Herráez and L. E. Ibáñez, The Swampland Distance Conjecture and Towers of Tensionless Branes, JHEP 08 (2019) 044 [1904.05379].
- [52] S.-J. Lee, W. Lerche and T. Weigand, Emergent Strings, Duality and Weak Coupling Limits for Two-Form Fields, 1904.06344.
- [53] F. Baume, F. Marchesano and M. Wiesner, Instanton Corrections and Emergent Strings, JHEP 04 (2020) 174 [1912.02218].
- [54] S. Cecotti, Special Geometry and the Swampland, JHEP 09 (2020) 147 [2004.06929].
- [55] N. Gendler and I. Valenzuela, Merging the weak gravity and distance conjectures using BPS extremal black holes, JHEP 01 (2021) 176 [2004.10768].

- [56] S.-J. Lee, W. Lerche, G. Lockhart and T. Weigand, Quasi-Jacobi forms, elliptic genera and strings in four dimensions, JHEP 01 (2021) 162 [2005.10837].
- [57] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, Swampland Conjectures for Strings and Membranes, JHEP 02 (2021) 006 [2006.15154].
- [58] D. Klaewer, S.-J. Lee, T. Weigand and M. Wiesner, Quantum corrections in 4d N = 1 infinite distance limits and the weak gravity conjecture, JHEP 03 (2021) 252 [2011.00024].
- [59] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, The EFT stringy viewpoint on large distances, JHEP 09 (2021) 197 [2104.05726].
- [60] E. Palti, Stability of BPS states and weak coupling limits, JHEP 08 (2021) 091 [2107.01539].
- [61] M. Etheredge, B. Heidenreich, S. Kaya, Y. Qiu and T. Rudelius, Sharpening the Distance Conjecture in diverse dimensions, JHEP 12 (2022) 114 [2206.04063].
- [62] G. Dvali, Black Holes and Large N Species Solution to the Hierarchy Problem, Fortsch. Phys. 58 (2010) 528 [0706.2050].
- [63] G. Dvali and M. Redi, Black Hole Bound on the Number of Species and Quantum Gravity at LHC, Phys. Rev. D 77 (2008) 045027 [0710.4344].
- [64] D. van de Heisteeg, C. Vafa, M. Wiesner and D. H. Wu, Bounds on Field Range for Slowly Varying Positive Potentials, 2305.07701.
- [65] M. Etheredge, B. Heidenreich, J. McNamara, T. Rudelius, I. Ruiz and I. Valenzuela, Running Decompactification, Sliding Towers, and the Distance Conjecture, 2306.16440.
- [66] J. Calderón-Infante, A. Castellano, A. Herráez and L. E. Ibáñez, Entropy Bounds and the Species Scale Distance Conjecture, 2306.16450.
- [67] D. van de Heisteeg, C. Vafa, M. Wiesner and D. H. Wu, Moduli-dependent Species Scale, 2212.06841.
- [68] N. Cribiori, D. Lust and G. Staudt, Black hole entropy and moduli-dependent species scale, 2212.10286.
- [69] D. van de Heisteeg, C. Vafa and M. Wiesner, Bounds on Species Scale and the Distance Conjecture, 2303.13580.

2 Dynamical Cobordism

This chapter contains the articles:

- Dynamical Tadpoles, Stringy Cobordism, and the SM from Spontaneous Compactification,
 Ginevra Buratti, M. Delgado, Angel M. Uranga JHEP 06 (2021) 170 arXiv:2104.02091 - Inspire
- Dynamical Cobordism and Swampland Distance Conjectures, Ginevra Buratti, José Calderón-Infante, M. Delgado, Angel M. Uranga JHEP 10 (2021) 037 arXiv:2107.09098 - Inspire
- At the end of the world: Local Dynamical Cobordism, Roberta Angius, José Calderón-Infante, M. Delgado, Jesús Huertas, Angel M. Uranga JHEP 06 (2022) 142 arXiv:2203.11240 - Inspire
- Dynamical Cobordism and the beginning of time: supercritical strings and tachyon condensation,
 Roberta Angius, M. Delgado, Angel M. Uranga JHEP 08 (2022) 285 arXiv:2207.13108 Inspire

Published for SISSA by 2 Springer

RECEIVED: April 15, 2021 ACCEPTED: June 1, 2021 PUBLISHED: June 29, 2021

Dynamical tadpoles, stringy cobordism, and the SM from spontaneous compactification

Ginevra Buratti, Matilda Delgado and Angel M. Uranga

Instituto de Física Teórica IFT-UAM/CSIC, C/ Nicolás Cabrera 13-15, Campus de Cantoblanco, 28049 Madrid, Spain E-mail: ginevra.buratti@uam.es, matilda.delgado@uam.es, angel.uranga@csic.es

ABSTRACT: We consider string theory vacua with tadpoles for dynamical fields and uncover universal features of the resulting spacetime-dependent solutions. We argue that the solutions can extend only a finite distance Δ away in the spacetime dimensions over which the fields vary, scaling as $\Delta^n \sim \mathcal{T}$ with the strength of the tadpole \mathcal{T} . We show that naive singularities arising at this distance scale are physically replaced by ends of spacetime, related to the cobordism defects of the swampland cobordism conjecture and involving stringy ingredients like orientifold planes and branes, or exotic variants thereof. We illustrate these phenomena in large classes of examples, including $AdS_5 \times T^{1,1}$ with 3-form fluxes, 10d massive IIA, M-theory on K3, the 10d non-supersymmetric USp(32) strings, and type IIB compactifications with 3-form fluxes and/or magnetized D-branes. We also describe a 6d string model whose tadpole triggers spontaneous compactification to a semirealistic 3-family MSSM-like particle physics model.

KEYWORDS: Flux compactifications, Superstring Vacua, Supersymmetry Breaking

ARXIV EPRINT: 2104.02091



Contents

1	Introduction and conclusions				
2	The	e fluxed conifold: KS solution as spontaneous cobordism	4		
	2.1	The 5d tadpole and its solution	5		
	2.2	Singularity at finite distance	6		
	2.3	Dynamical cobordism and the KS solution	7		
	2.4	More general throats	8		
3	Type IIB fluxes and spontaneous compactification				
	3.1	The 5d tadpole and its solution	9		
	3.2	The singularities	10		
	3.3	Cobordism and spontaneous compactification	11		
4	Solving dynamical tadpoles via magnetization				
	4.1 Solving dynamical tadpoles of magnetized branes				
	4.2	Backreaction and spontaneous compactification	13		
5	Solving tadpoles in 10d strings				
	5.1	Massive IIA theory	15		
		5.1.1 Solution in 9d and type I' as cobordism	15		
		5.1.2 A non-supersymmetric Freund-Rubin solution	16		
	5.2 An aside on M-theory on K3				
	5.3	Solving tadpoles in the non-supersymmetric 10d $USp(32)$ theory	18		
		5.3.1 The Dudas-Mourad solution and cobordism	18		
		5.3.2 Solving the tadpole via magnetization	20		
6	The	e SM from spontaneous compactification	21		
\mathbf{A}	Dyi	namical tadpoles and swampland constraints	23		

1 Introduction and conclusions

Supersymmetry breaking string vacua (including 10d non-supersymmetric strings) are generically affected by tadpole sources for dynamical fields, unstabilizing the vacuum [1, 2]. We refer to them as *dynamical tadpoles* to distinguish them from *topological tadpoles*, such as RR tadpoles, which lead to topological consistency conditions on the configuration (note however that dynamical tadpoles were recently argued in [3] to relate to violation of swampland constraints of quantum gravity theories). Simple realizations of dynamical tadpoles arose in early models of supersymmetry breaking using antibranes in type II (orientifold) compactifications [4–7], or in 10d non-supersymmetric string theories [8].

Dynamical tadpoles indicate the fact that equations of motion are not obeyed in the proposed configuration, which should be modified to a spacetime-dependent solution (more precisely, solution in which some fields do not preserve the maximal symmetry in the corresponding spacetime dimension, but we stick to the former nomenclature), e.g. rolling down the slope of the potential. This approach has been pursued in the literature (see e.g. [9-13]), although the resulting configurations often contain metric singularities or strong coupling regimes, which make their physical interpretation difficult.

In this work we present large classes of spacetime¹ dependent field configurations sourced by dynamical tadpoles, which admit a simple and tractable smoothing out of such singularities. Remarkably, these examples reveal a set of notable physical principles and universal scaling behaviours. We argue that the presence of a dynamical tadpole implies the appearance of ends of spacetime (or walls of nothing) at a finite spacetime distance, which is (inversely) related to the strength of the tadpole. These ends of spacetime moreover correspond to cobordism defects (or end of the world branes) of the theory implied by the swampland cobordism conjecture [14, 15]. In most setups the cobordism defects end up closing off the space into a compact geometry (possibly decorated with branes, fluxes or other ingredients), thus triggering spontaneous compactification.

We can sum up the main features described above, and illustrated by our examples, in two lessons:

Finite distance. In the presence of a dynamical tadpole controlled by an order parameter \mathcal{T} , the spacetime-dependent solution of the equations of motion cannot be extended to spacetime distances beyond a critical value Δ scaling inversely proportional to \mathcal{T} , with a scaling relation

$$\Delta^{-n} \sim \mathcal{T}.\tag{1.1}$$

In our examples, n = 1 or n = 2 for setups with an underlying AdS-like or Minkowski vacuum, respectively.

Dynamical cobordism. The physical mechanism cutting off spacetime dimensions at scales bounded by the Δ above, is a cobordism defect of the initial theory (including the dynamical tadpole source).

To be precise, when there are multiple spacetime directions to be closed off, the actual defect is the cobordism defect corresponding to circle or toroidal compactifications of the initial theory, with suitable monodromies on non-trivial cycles. This is analogous to the mechanism by which F-theory on half a \mathbf{P}_1 provides the cobordism defect for type IIB on \mathbf{S}^1 with $\mathrm{SL}(2, \mathbf{Z})$ monodromy [14] (see also [16]).

As explained, we present large classes of models illustrating these ideas, including (susy and non-susy) 10d string theories and type II compactifications with D-branes, orientifold planes, fluxes, etc. For simplicity, we present models based on toroidal examples (and orbifolds and orientifolds thereof), although many of the key ideas easily extend to more

 $^{^{1}}$ Actually, we restrict to configurations of fields varying over spatial dimensions (rather than time); yet we abuse language and often refer to them as spacetime-dependent.

general setups. This strongly suggests that they can apply to general string theory vacua. Very remarkably, the tractability of the models allows to devise spontaneous compactification whose endpoint corresponds to some of the (supersymmetric extensions of the) SM-like D-brane constructions in the literature. As will be clear, our examples can often be regarded as novel reinterpretations of models in the literature.

Although our examples are often related to supersymmetric models, supersymmetry is not a crucial ingredient in our discussion. Dynamical tadpoles correspond to sitting on the slope of potentials, which, even in theories admitting supersymmetric vacua, correspond to non-supersymmetric points in field space. On the other hand, supersymmetry of the final spacetime-dependent configuration is a useful trick to guarantee that dynamical tadpoles have been solved, but it is possible to build solutions with no supersymmetry but equally solving tadpoles.

Our results shed new light on several features observed in specific examples of classical solutions to dynamical tadpoles, and provide a deeper understanding of the appearance of singularities, and the stringy mechanism smoothing them out and capping off dimensions to yield dynamical compactification. In particular, we emphasize that our discussion unifies several known phenomena and sheds new light on the strong coupling singularities of type I' in [17] and in heterotic M-theory [18] (and its lower bound on the 4d Newton's constant). There are several directions which we leave for future work, for instance:

- As is clear from our explicit examples, many constructions of this kind can be obtained via a reinterpretation of known compactifications. This strongly suggests that our lessons have a general validity in string theory. It would be interesting to explore the discussion of tadpoles, cobordism and spontaneous compactifications in general setups beyond tori.
- A general consequence of (1.1) is a non-decoupling of scales between the geometric scales controlling the order parameter of the dynamical tadpole and the geometric size of the spontaneously compactified dimensions. This is reminiscent of the swampland AdS distance conjecture [19]. It would be interesting to explore the generation of hierarchies between the two scales, possibly based on discrete \mathbf{Z}_k gauge symmetries as in [20].
- Our picture can be regarded as belonging to the rich field of swampland constraints on quantum gravity [21] (see [22–24] for reviews). It would be interesting to study the interplay with other swampland constraints. In particular, the relation between the strength of the dynamical tadpole and the size of the spacetime dimensions is tantalizingly reminiscent of the first condition on |∇V|/V of the de Sitter conjecture [25–27], with *T* = |∇V| and if we interpret V as the inverse Hubble volume and hence a measure of size or length scale in the spacetime dimensions. It would be interesting to explore cosmological setups and a possible role of horizons as alternative mechanisms to cut off spacetime. Also, the inequality admittedly works in different directions in the two setups, thus suggesting they are not equivalent, but complementary relations.

- It would be interesting to apply our ideas to the study of other setups in which spacetime is effectively cut off, such as the capping off of the throat in near horizon NS5-branes due to strong coupling effects, or the truncation in [28] of throats of the euclidean wormholes in pure Einstein+axion theories [29].
- Finally, we have not discussed time-dependent backgrounds.² These are obviously highly interesting, but their proper understanding is likely to require new ingredients, such as *end (or beginning) of time* defects (possibly as generalization of the spacelike S-branes [30, 31]).

Until we come back to these questions in future work, the present paper is organized as follows. In section 2 we reinterpret the Klebanov-Strassler (KS) warped throat supported by 3-form fluxes as a template illustrating our two tadpole lessons. Section 2.1 explains that the introduction of RR 3-form flux in type IIB theory on $AdS_5 \times T^{1,1}$ produces a tadpole. The varying field configuration is the Klebanov-Tseytlin solution, which leads to a metric singularity at a finite distance scaling as (1.1), as we show in section 2.2. In section 2.3 we relate the KS smoothing of this singularity with cobordism defects. In section 2.4 we extend the discussion to other warped throats. In section 3 we present a similar discussion in toroidal compactifications with fluxes. Section 3.1 introduces a T_5 compactification with RR 3-form flux, whose tadpole backreacts producing singularities at finite distance as we show in section 3.2. In section 3.3 we argue they are smoothed out by capping off dimensions and triggering spontaneous compactification. In section 4 we build examples in the context of magnetized D-branes. In section 4.1 we describe the tadpole backreaction and its singularities, which are removed by spontaneous compactification in section 4.2. In section 5 we turn to the dilaton tadpole of several 10d strings. In section 5.1 we consider massive type IIA theory, where the running dilaton solutions produce dynamical cobordisms by introduction of O8-planes as cobordism defects of the IIA theory, eventually closely related to type I' compactifications. In section 5.2 we discuss a similar picture for M-theory on K3 with G_4 flux, and a Horava-Witten wall as its cobordism defect. In section 5.3 we consider the 10d non-supersymmetric USp(32) theory, in two different approaches. In section 5.3.1 we build on the classical solution in [9] and discuss its singularities in the light of the cobordism conjecture. In section 5.3.2 we describe an explicit (and remarkably, supersymmetry preserving) configuration solving its tadpole via magnetization and spontaneous compactification on \mathbf{T}^{6} . In section 6 we discuss an interesting application, describing a 6d model with tadpoles, which upon spontaneous compactification reproduces a semi-realistic MSSM-like brane model. Finally, appendix A discusses the violation of swampland constraints of type IIB on $AdS_5 \times T^{1,1}$ when its tadpole is not duly backreacted, in a new example of the mechanism in [3].

2 The fluxed conifold: KS solution as spontaneous cobordism

In this section we consider the question of dynamical tadpoles and their consequences in a particular setup, based on the gravity dual of the field theory of D3-branes at a coni-

²For classical solutions of tadpoles involving time dependence, see e.g. [11].

fold singularity. The discussion is a reinterpretation, in terms useful for our purposes, of the construction of the Klebanov-Tseytlin (KT) solution [32] and its deformed avatar, the Klebanov-Strassler (KS) solution [33]. This reinterpretation however provides an illuminating template to discuss dynamical tadpoles in other setups in later sections.

We consider type IIB on $\operatorname{AdS}_5 \times T^{1,1}$, where $T^{1,1}$ is topologically $\mathbf{S}^2 \times \mathbf{S}^3$ [34]. This is the near horizon geometry of D3-branes at the conifold singularity [34] (see also [35–37]), which has been widely exploited in the context of holographic dualities. The vacuum is characterized by the IIB string coupling $e^{\phi} = g_s$ and the RR 5-form flux N. The model has no scale separation, since the $T^{1,1}$ and AdS_5 have a common scale R, given by

$$R^4 = 4\pi g_s N \alpha'^2 \,. \tag{2.1}$$

In any event, we will find useful to discuss the model, and its modifications, in terms of the (KT) 5d effective theory introduced in [32]. This is an effective theory not in the Wilsonian sense but in the sense of encoding the degrees of freedom surviving a consistent truncation. In particular, it includes the dilaton ϕ (we take vanishing RR axion for simplicity), the NSNS axion $\Phi = \int_{\mathbf{S}^2} B_2$ and the $T^{1,1}$ breathing mode q (actually, stabilized by a potential arising from the curvature and the 5-form flux), which in the Einstein frame enters the metric as

$$ds_{10}^2 = R^2 \left(e^{-5q} \, ds_5^2 + e^{3q} ds_{T^{1,1}}^2 \right) \,. \tag{2.2}$$

This approach proved useful in [38] in the discussion of the swampland distance conjecture [39] in configurations with spacetime-dependent field configurations (see [19] for a related subsequent development, and [40, 41]).

2.1 The 5d tadpole and its solution

Let us introduce M units of RR 3-form flux in the \mathbf{S}^3 , namely

$$F_3 = M \,\omega_3\,,\tag{2.3}$$

where ω_3 is defined in eq. (27) in [33]. We do not need its explicit expression, it suffices to say that it describes a constant field strength density over the \mathbf{S}^3 . The introduction of this flux sources a backreaction on the dilaton and the metric, namely a dynamical tadpole for ϕ and q. In addition, as noticed in [38], it leads to an axion monodromy potential for Φ [42–45]. The situation is captured by the KT effective action (with small notation changes) for the 5d scalars ϕ , Φ and q, collectively denoted by φ^a

$$S_5 = -\frac{2}{\kappa_5^2} \int d^5 x \,\sqrt{-g_5} \left[\frac{1}{4} R_5 - \frac{1}{2} G_{ab}(\varphi) \partial \varphi^a \partial \varphi^b - V(\varphi) \right],\tag{2.4}$$

with the kinetic terms and potential given by

$$G_{ab}(\varphi)\partial\varphi^a\partial\varphi^b = 15(\partial q)^2 + \frac{1}{4}(\partial\phi)^2 + \frac{1}{4}e^{-\phi-6q}(\partial\Phi)^2, \qquad (2.5)$$

$$V(\varphi) = -5e^{-8q} + \frac{1}{8}M^2 e^{\phi - 14q} + \frac{1}{8}(N + M\Phi)^2 e^{-20q}.$$
 (2.6)

Clearly $g_s M^2$ is an order parameter of the corresponding dynamical tadpole. In the following we focus on the case³ of N being a multiple of M.

Ignoring the backreaction of the dynamical tadpole (i.e. considering constant profiles for the scalars over the 5d spacetime) is clearly incompatible with the equations of motion. Furthermore, as argued in [3], it can lead to violations of swampland constraints. In particular, since the introduction of F_3 breaks supersymmetry, if the resulting configuration was assumed to define a stable vacuum, it would violate the non-susy AdS conjecture [46]; also, as we discuss in appendix A, it potentially violates the Weak Gravity Conjecture [47].

Hence, we are forced to consider spacetime-dependent scalar profiles to solve the equations of motion. Actually, this problem was tackled in [33], with the scalars running with r, as we now review in the interpretation in [38]. There is a non-trivial profile for the axion Φ , given by

$$\Phi = 3g_s M \log(r/r_0). \tag{2.7}$$

This implies the cancellation of the dilaton tadpole, which can be kept constant $e^{\phi} = g_s$, as follows from its equation of motion from (2.5), (2.6)

$$\nabla\phi \sim -e^{-6q-\phi} (\partial\Phi)^2 + e^{-14q+\phi} M^2.$$
(2.8)

2.2 Singularity at finite distance

The varying Φ corresponds to the introduction of an NSNS 3-form flux in the configuration

$$H_3 = -g_s *_{6d} F_3, \qquad (2.9)$$

where the 6d refers to $T^{1,1}$ and the AdS₅ radial coordinate r, and the Hodge duality is with the AdS₅ × $T^{1,1}$ metric. This is precisely such that the complexified flux combination $G_3 = F_3 - \tau H_3$ satisfies the imaginary self duality (ISD) constraint making it compatible with 4d Poincaré invariance in the remaining 4d coordinates (and in fact, it also preserves supersymmetry). The backreaction on the metric thus has the structure in [48, 49]. The metric (2.2) takes the form

$$ds_{10}^2 = Z^{-\frac{1}{2}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z^{\frac{1}{2}} \left(dr^2 + r^2 ds_{T^{1,1}}^2 \right), \qquad (2.10)$$

where Z obeys a Laplace equation in AdS_5 , sourced by the fluxes, and reads

$$Z(r) = \frac{1}{4r^4} (g_s M)^2 \log(r/r_0).$$
(2.11)

The warp factor also enters in the RR 5-form flux, which decreases with r as

$$N(r) = \int_{\mathbf{S}^5} F_5 = g_s M^2 \log(r/r_0) \,. \tag{2.12}$$

This matches nicely with the monodromy for the axion Φ as it runs with r [38]. These features (as well as some other upcoming ones) were nicely explained as the gravity dual of a Seiberg duality cascade in [33].

³This implies that the configuration is uncharged under a discrete \mathbf{Z}_M symmetry, measured by N mod M, and associated to the redundancy generated by transformation $\phi \to \phi + 1$, $N \to N - M$, see footnote 5.

This 5d running solution in [32] solves the dynamical tadpole, but is not complete, as it develops a metric singularity at $r = r_0$. This is a physical singularity at finite distance in spacetime, whose parametric dependence on the parameters of the initial model is as follows

$$\Delta(r) = \int_{r_0}^r Z(r)^{\frac{1}{4}} dr \sim \int_{r_0}^r (g_s M)^{\frac{1}{2}} \left[\log(r/r_0) \right]^{\frac{1}{4}} \frac{dr}{r} \sim (g_s M)^{\frac{1}{2}} \left[\log(r/r_0) \right]^{\frac{5}{4}} = (g_s N)^{\frac{1}{4}} \frac{N}{g_s M^2} \sim R \frac{N}{g_s M^2}.$$
(2.13)

In the last equalities we used (2.12), (2.1). Hence, starting with an $AdS_5 \times T^{1,1}$ theory with N units of RR 5-form flux, the introduction of M units of RR 3-form flux leads to a breakdown of the corresponding spacetime-dependent solution at a distance scaling as $\Delta \sim M^{-2}$. Recalling that the dynamical tadpole is controlled by an order parameter $\mathcal{T} = g_s M^2$, this precisely matches the scaling relation (1.1) of the Finite Distance Lesson.

2.3 Dynamical cobordism and the KS solution

As is well known, the singularity in the KT solution is smoothed out in the KS solution [33]. This is given by a warped version of the deformed conifold metric, instead of the conical conifold singularity, with warp factor again sourced by an ISD combination of RR 3-form flux on \mathbf{S}^3 and NSNS 3-form flux on \mathbf{S}^2 times the radial coordinate. At large r the KS solution asymptotes to the KT solution, but near $r \sim r_0$, the solutions differ and the KT singularity is replaced by the finite size \mathbf{S}^3 of the deformed conifold.

Hence, the Finite Distance Lesson still applies even when the singularity is removed, and the impossibility to extend the coordinate r to arbitrary distances is implemented by a smooth physical end of spacetime. The purpose of this section is to highlight a novel insight on the KS solution, as a non-trivial realization of the swampland cobordism conjecture [14, 15]. The latter establishes that any consistent quantum gravity theory must be trivial in (a suitably defined version of) cobordism. Namely in an initial theory given by an n-dimensional internal compactification space (possibly decorated with additional ingredients, like branes or fluxes), there must exist configurations describing an (n+1)-dimensional (possibly decorated) geometry whose boundary is the initial one. The latter describes an end of the world defect (which we will refer to as the 'cobordism defect') for the spacetime of the initial theory. Since the arguments about the swampland cobordism conjecture are topological, there is no claim about the unprotected properties of the cobordism defect, although in concrete examples it can preserve supersymmetry; for instance, in maximal dimensions, the Horava-Witten boundary is the cobordism defect for 11d M-theory, and similarly the O8-plane is the cobordism defect of type IIA theory.⁴

In our setup, the initial theory is $AdS_5 \times T^{1,1}$ with N units of RR 5-form flux and M units of RR 3-form flux on S^3 . From the above discussion, it is clear that the KS solution

 $^{^{4}}$ Other 10d theories are conjectured to admit cobordism branes, but they cannot be supersymmetric and their nature is expected to be fairly exotic, and remains largely unknown. We will come back to this point in section 5.3.1.
is just the cobordism defect of this theory.⁵ The remarkable feature is that the end of spacetime is triggered dynamically by the requirement of solving the equations of motion after the introduction of the RR 3-form flux, hence it is fair to dub it dynamical cobordism. Hence, this is a very explicit illustration of the Dynamical Cobordism Lesson.

This powerful statement will be realized in many subsequent examples in later sections, and will underlie the phenomenon of spontaneous compactification, when the cobordisms close off the spacetime directions bounding them into a compact variety.

2.4 More general throats

A natural question is the extension of the above discussion to other $AdS_5 \times \mathbf{X}_5$ vacua with 3-form fluxes. This question is closely related to the search for general classes of gravity duals to Seiberg duality cascades and their infrared deformations, for which there is a concrete answer if \mathbf{X}_5 is the real base of a non-compact toric CY threefold singularity \mathbf{Y}_6 , which are very tractable using dimer diagrams [50, 51] (see [52] for a review).

From our perspective, the result in [53] is that the X_5 compactification with 3-form flux F_3 admits a KS-like end of the world (cobordism defect⁶) if Y_6 admits a complex deformation which replaces its conical singularity by a finite-size 3-cycle corresponding to the homology dual of the class $[F_3]$. In cobordism conjecture terms, in these configurations the corresponding global symmetry is broken, and spacetime may close off without further ado (as the axion monodromy due to the 3-form fluxes allows to eat up the RR 5-form flux before reaching the end of the world). Such complex deformations are easily discussed in terms of the web diagram for the toric threefold, as the splitting of the web diagram into consistent sub-diagrams [53]. Simple examples include the deformation of the complex cone over dP₂ to a smooth geometry, or the deformation of the complex cone over dP₃ to a conifold, or to a smooth geometry.

There are however singularities (or 3-form flux assignments), for which the complex deformations are simply not available. One may then wonder about how our Dynamical Cobordism lesson applies. The answer was provided in particular examples in [54–56]: the infrared end of the throat contains an explicit system of fractional D-branes, which in the language of the cobordism conjecture kill the corresponding cobordism classes, and allow the spacetime to end. As noticed in these references, the system breaks supersymmetry, and in [55] it was moreover noticed (as later revisited in [57]) to be unstable and lead to a runaway behaviour for the field blowing up the singularity. Hence, this corresponds to an additional dynamical tadpole, requiring additional spacetime dependence, to be solved. Simple examples include the complex cone over dP₁, and the generic $Y^{p,q}$ theories. We will not enter the discussion of possible mechanisms to stabilize these models, since following [58] they are likely to require asymptotic modifications of the warped throat ansatz (i.e. at all positions in the radial direction, including the initial one).

⁵Recalling footnote 3, the case of N multiple of M implies the vanishing of a \mathbf{Z}_M charge, and allows the cobordism defect to be purely geometrical; otherwise the cobordism defect ending spacetime must include explicit D3-branes, which are the defect killing the corresponding cobordism class [14].

⁶We note in passing that the regions between different throats in the multi-throat configurations [53] can be regarded as domain walls interpolating between two different, but bordant, type IIB vacua.

3 Type IIB fluxes and spontaneous compactification

In this section we construct an explicit 5d type IIB model with a tunable dynamical tadpole, and describe the spacetime-dependent solution solving its equations of motion, which is in fact supersymmetry preserving. The configuration displays dynamical cobordism resulting in spontaneous compactification to 4d. The resulting model is a simple toroidal compactification with ISD NSNS and RR 3-form fluxes [48, 49], in particular it appeared in [59, 60]. With this perspective in hindsight, one can regard this section as a reinterpretation of the latter flux compactification. Our emphasis is however in showing the interplay of the dynamical tadpoles in the 5d theory and the consequences in the spacetime configuration solving them.

3.1 The 5d tadpole and its solution

Consider type IIB on \mathbf{T}^5 , which for simplicity we consider split as $\mathbf{T}^2 \times \mathbf{T}^2 \times \mathbf{S}^1$. We label the coordinates of the \mathbf{T}^2 's as (x^1, y^1) and (x^2, y^2) , with periodicity 1, and introduce complex coordinates as $z^1 = x^1 + \tau_1 y^1$, $z^2 = x^2 + \tau_2 y^2$. We also use a periodic coordinate $x^3 \simeq x^3 + 1$ to parametrize the \mathbf{S}^1 . For simplicity, we do not consider moduli deviating from this rectangular structure,⁷ and also take the \mathbf{T}^5 to have an overall radius R,

$$ds^{2} = R^{2}[(dz^{1})^{2} + (dz^{2})^{2} + (dx^{3})^{2}].$$
(3.1)

The result so far is a standard 5d supersymmetric \mathbf{T}^5 compactification.

We introduce a non-trivial dynamical tadpole source by turning on an RR 3-form flux (using conventions in [49])

$$F_3 = (2\pi)^2 \alpha' N \, dx^1 \, dx^2 \, dx^3 \,. \tag{3.2}$$

The introduction of this flux does not lead to RR topological tadpoles, but induces dynamical tadpoles for diverse fields. In the following we focus on the dynamics of the 5d light fields R, τ_1 , τ_2 , the dilaton ϕ and the NSNS axion Φ defined by

$$B_2 = \Phi \, dy^1 \, dy^2 \,. \tag{3.3}$$

The discussion of the dynamical tadpole is similar to the $T^{1,1}$ example in section 2, so we sketch the result. There is a dilaton tadpole, arising from the dimensional reduction of the 10d kinetic term for the 3-form flux,

$$\nabla^2 \phi = \frac{1}{12} e^{\phi} (F_3)^2 . \tag{3.4}$$

Since $(F_3)^2$ is a constant source density, which does not integrate to zero over \mathbf{T}^5 , there is no solution for this Laplace equation if we assume the solution to be independent of the 5d spacetime coordinates. One possibility would be to allow for 5d spacetime dependence of ϕ (at least in one extra coordinate, as in [9]). Here we consider a different possibility, which

⁷As usual, they can be removed in orbifold models, although we will not focus on this possibility.

is to let the NSNS axion Φ acquire a dependence on one of the 5d coordinates, which we denote by y, as follows

$$\Phi = -(2\pi)^2 \alpha' \frac{N}{t_3} y \quad \Rightarrow \quad H_3 = -(2\pi)^2 \alpha' \frac{N}{t_3} \, dy^1 \, dy^2 \, dy \,. \tag{3.5}$$

We have thus turned on NSNS 3-form field strength in the directions y^1 , y^2 in \mathbf{T}^5 and the 5d spacetime coordinate y. Here the sign has been introduced for later convenience, and t_3 is a positive real parameter allowing to tune the field strength density, whose meaning will become clear later on.

Including this new source, the dilaton equation of motion becomes

$$\nabla^2 \phi = \frac{1}{12} \left[e^{\phi} (F_3)^2 - e^{-\phi} (H_3)^2 \right].$$
(3.6)

Hence, the spacetime-dependent profile (3.5) can cancel the right hand side and solve the dilaton tadpole when

$$e^{2\phi} (F_3)^2 = (H_3)^2 . (3.7)$$

We can thus keep the dilaton constant $e^{\phi} = g_s$. Taking for simplicity purely imaginary $\tau_1 = it_1$ and $\tau_2 = it_2$, the condition (3.7) is simply

$$g_s t_1 t_2 t_3 = 1. (3.8)$$

In addition to the dilaton, the 3-form fluxes backreact on the metric and other fields, which we discuss next.

3.2 The singularities

We now discuss the backreaction on the metric and other fields. For convenience, we use the complex coordinates z^1 , z^2 and $z^3 = x^3 + iy$. In terms of these, we can write the combination

$$G_3 = F_3 - \tau H_3 = \frac{(2\pi)^2}{4} \alpha' N(d\overline{z}_1 \, dz_2 \, dz_3 + dz_1 d\overline{z}_2 dz_3 + dz_1 d\overline{z}_2 d\overline{z}_3 + d\overline{z}_1 d\overline{z}_2 d\overline{z}_3) \,. \tag{3.9}$$

Regarding $\mathbf{T}^5 \times \mathbf{R}_y^1$ as a (non-compact) CY, this is a combination of (2,1) and (0,3) components, which is thus ISD. There is a backreaction on the metric and RR 4-form of the familiar black 3-brane kind. In particular, the metric includes a warp factor Z

$$ds_{10}^2 = Z^{-\frac{1}{2}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z^{\frac{1}{2}} R^2 [dz^1 d\overline{z}^1 + dz^2 d\overline{z}^2 + dz^3 d\overline{z}^3], \qquad (3.10)$$

where x^{μ} runs through the four Poincaré invariant spacetime coordinates. The warp factor is determined by the Laplace equation

$$-\tilde{\nabla}^2 Z = \frac{g_s}{12} G_3 \cdot \overline{G}_3 = \frac{g_s}{6} (F_3)^2, \qquad (3.11)$$

with the tilde indicating the Laplacian is computed with respect to the unwarped, flat metric, and in the last equation we used (3.7).

Note that, since y parametrizes a non-compact dimension, there is no tadpole problem in solving (3.11) i.e. we need not add background charge. One may then be tempted to conclude that this provides a 5d spacetime-dependent configuration solving the 5d tadpole. However, the solution is valid locally in y, but cannot be extended to arbitrary distances in this direction. Since the local flux density in \mathbf{T}^5 is constant, we can take Z to depend only on⁸ y, hence leading to a solution

$$-\frac{d^2 Z}{dy^2} = \frac{g_s}{6} (F_3)^2 \quad \Rightarrow \quad Z = 1 - \frac{g_s}{12} (F_3)^2 y^2, \qquad (3.12)$$

where we have set an integration constant to 1. The solution hits metric singularities at

$$y^{-2} = \frac{1}{12} g_s(F_3)^2 , \qquad (3.13)$$

showing there is a maximal extent in the direction y. Let us introduce the quantity $\mathcal{T} = \frac{1}{12}g_s(F_3)^2$, which controls the parametric dependence of the tadpole. Then, the distance between the singularities is

$$\Delta = \int_{-\mathcal{T}^{-1/2}}^{\mathcal{T}^{-1/2}} Z^{\frac{1}{4}} \, dy = \frac{2}{\sqrt{\mathcal{T}}} \int_{0}^{1} (1-t^2)^{\frac{1}{4}} \, dt \,, \tag{3.14}$$

with $t = \sqrt{\mathcal{T}}y$. We thus recover the scaling (1.1) with n = 2,

$$\Delta^{-2} \sim \mathcal{T} \,. \tag{3.15}$$

Hence the appearance of the singularities as a consequence of the dynamical tadpole is as explained in the introduction.

3.3 Cobordism and spontaneous compactification

The appearance of singularities is a familiar phenomenon. In this section we argue that they must be smoothed out, somewhat analogously to the KS solution in section 2. The fact that it is possible follows from the swampland cobordism conjecture [14, 15], namely there must exist an appropriate cobordism defect closing off the extra dimension into nothing. Since there are two singularities, the formerly non-compact dimension becomes compact, in an explicit realization of spontaneous compactification.⁹

In the following, we directly describe the resulting geometry, which turns out to be a familiar \mathbf{T}^6 (orientifold) compactification with ISD 3-form fluxes. Consider type IIB theory on $\mathbf{T}^2 \times \mathbf{T}^2 \times \mathbf{T}^2$, with

$$F_3 = (2\pi)^2 \alpha' N \, dx^1 \, dx^2 \, dx^3 \,, \quad H_3 = (2\pi)^2 \alpha' N \, dy^1 \, dy^2 \, dy^3 \,. \tag{3.16}$$

We use $z^i = x^i + it_i y^i$, hence the above defined t_3 is the complex structure modulus for the \mathbf{T}^2 involving the newly compact dimension. For moduli satisfying (3.8) the \mathbf{T}^6 flux combination G_3 is given by (3.9), which is ISD and indeed compatible with 4d Poincaré

⁸In fact, this is the leading behaviour at long distances, compared with the \mathbf{T}^5 size scale R.

⁹Spontaneous compactification has been discussed in the context of dynamical tadpoles in [9].

invariance as usual. Notice that in this case, it is possible to achieve a large size for the new compact dimension $t_3 \gg 1$ by simply e.g. taking small g_s . This corresponds to the regime of small 5d tadpole, with the relation

$$t_3^{-2} \sim g_s^2 \sim \mathcal{T}^2 \,, \tag{3.17}$$

in agreement with the maximal distance relation in the previous section.

Consistency, in the form of C_4 RR tadpole cancellation, requires the introduction of O3-planes at fixed points of the involution $\mathcal{R} : z^i \to -z^i$ (together with mobile D3-branes). From the perspective of the 5d theory, the additional dimension is compactified on an interval, with two end of the world defects given by the O3-planes, which constitute the cobordism defects of the configuration (possibly decorated with explicit D3-branes if needed).

4 Solving dynamical tadpoles via magnetization

In this section we consider a further setup displaying dynamical tadpoles, based on compactifications with magnetized D-branes [61–65]. In toroidal setups, these have been (either directly or via their T-dual intersecting brane world picture) widely used to realize semi-realistic particle physics models in string theory. In more general setups, magnetized 7-branes are a key ingredient in the F-theory realization of particle physics models [66–68].

4.1 Solving dynamical tadpoles of magnetized branes

We consider a simple illustrative example. Consider type IIB theory compactified on $\mathbf{T}^2 \times \mathbf{T}^2$ (labelled 1 and 2, respectively) and mod out by $\Omega \mathcal{R}_1(-1)^{F_L}$, where $\mathcal{R}_1 : z_1 \to -z_1$. This introduces 4 O7₁-planes spanning $(\mathbf{T}^2)_2$ and localized at the fixed points on $(\mathbf{T}^2)_1$. We also have 32 D7-branes (as counted in the covering space), split as 16 D7-branes (taken at generic points) and their 16 orientifold images. This model is related by T-duality on $(\mathbf{T}^2)_1$ to a type I toroidal compactification, but we proceed with the D7-brane picture.

We introduce M units of worldvolume magnetic flux along $(\mathbf{T}^2)_2$ for the U(1) of a D7-brane¹⁰

$$\frac{1}{2\pi\alpha'}\int_{\mathbf{T}^2}F_2 = M\,.\tag{4.1}$$

The orientifold requires we introduce -M units of flux on the image D7-brane.¹¹ This also ensures that there is no net induced **Z**-valued D5-brane charge in the model, and hence no associated RR tadpole, in agreement with the fact that the RR 6-form is projected out. In addition, there is a **Z**₂ K-theory charge [70] which is cancelled as long as $M \in 2\mathbf{Z}$.

The introduction of the worldvolume flux leads to breaking of supersymmetry. As is familiar in the discussion of supersymmetries preserved by different branes [71], we introduce the angle

$$\theta_2 = \arctan(2\pi\alpha' F) = \arctan(M\chi), \qquad (4.2)$$

where F is the field strength and χ is the inverse of the \mathbf{T}^2 area, in string units.

¹⁰If N the D7-branes are coincident, it is also possible to use the overall $U(1) \subset U(N)$. We will stick to the single D7-brane for the moment, but such generalization will arise in later examples.

¹¹For simplicity we consider vanishing discrete NSNS 2-form flux [69], although such generalization will arise in later examples.

This non-supersymmetric configuration introduces dynamical tadpoles. For small θ_2 , the extra tension can be described in effective field theory as an FI term controlled by θ [72–74]. In fact, in [75] a similar parametrization was proposed for arbitrary angles. By using the DBI action, the extra tension has the structure

$$V \sim \frac{1}{g_s} \left(\sqrt{1 + (\tan \theta_2)^2} - 1 \right).$$
 (4.3)

This leads to a tadpole for the dilaton and the $(\mathbf{T}^2)_2$ Kähler modulus.

We now consider solving the tadpole by allowing for some spacetime-dependent background. Concretely, we allow for a non-trivial magnetic field -F on two of the non-compact space coordinates, parametrized by the (for the moment, non-compact) coordinate z_3 . In fact this leads to a configuration preserving supersymmetry since, defining the angle θ_3 in analogy with (4.2), we satisfy the SU(2) rotation relation $\theta_3 + \theta_2 = 0$ [71]. In other words, the field strength flux has the structure

$$F_2 = F(dz_2 d\overline{z}_2 - dz_3 d\overline{z}_3), \qquad (4.4)$$

which is (1, 1) and primitive (i.e. $J \wedge F_2 = 0$), which are the supersymmetry conditions for a D-brane worldvolume flux.

Hence, it is straightforward to find spacetime-dependent solutions to the tadpole of the higher-dimensional theory, at the price of breaking part of the symmetry of the lowerdimensional spacetime. In the following we show that, as in earlier examples, this eventually also leads to spontaneous compactification.

4.2 Backreaction and spontaneous compactification

The spacetime field strength we have just introduced couples to gravity and other fields, so we need to discuss its backreaction.

In fact, this is a particular instance of earlier discussions, by considering the F-theory lift of the D7-brane construction. This can be done very explicitly by taking the configuration near the SO(8)⁴ weak coupling regime [76]. The configuration without magnetic flux M = 0 simply lifts to F-theory on K3×T² × R², where the (T²)₁ (modulo the Z₂ orientifold action) is the P₁ base of K3, and the T² and R² explicit factors correspond to the directions z_2 and z_3 , respectively. As is familiar, the 24 degenerate fibers of the K3 elliptic fibration form 4 pairs, reproducing the 4 orientifold planes, and 16 D7-branes in the orientifold quotient. Actually, the discussion below may be carried out for F-theory on K3 at generic points in moduli space, even not close to the weak coupling point.

The introduction of magnetization for one 7-brane corresponds to the introduction of a G_4 flux along the local harmonic (1, 1)-form supported at an I_1 degeneration (or enhanced versions thereof, for coincident objects), of the form

$$G_4 = \omega_2 \wedge F(dz_2 d\overline{z}_2 - dz_3 d\overline{z}_3).$$

$$(4.5)$$

This flux is self-dual, and in fact (2, 2) and primitive, which is the supersymmetry preserving condition for 4-form fluxes in M/F-theory [48, 77]. The backreacted metric is described

by a warp factor satisfying a Laplace equation sourced by the fluxes, similar to (3.11). Considering the regime in which the warp factor is taken independent of the internal space and depends only on the coordinates in the \mathbf{R}^2 parametrized by z_3 , the constant flux density leads to singularities at a maximal length scale Δ

$$\Delta^{-2} \sim F^2 \,. \tag{4.6}$$

This is another instance of the universal relation (1.1) with $\mathcal{T} \sim F^2$, hence n = 2.

This is in complete analogy with earlier examples. Hence, we are led to propose that the smoothing out of these singularities is provided by the compactification of the corresponding coordinates, e.g. on a \mathbf{T}^2 , with the addition of the necessary cobordism defects, namely orientifold planes and D-branes.¹²

To provide an explicit solution, we introduce the standard notation (see e.g. [63, 64]) of (n, m) for the wrapping numbers and the magnetic flux quanta on the $(\mathbf{T}^2)_i$'s for the directions i = 1, 2, 3. In this notation, the O7₁-planes and unmagnetized D7₁-branes are associated to $(0, 1) \times (1, 0) \times (1, 0)$, while the magnetized D7₁-branes¹³ correspond to $(0, 1) \times (1, M) \times (1, -M)$, and $(0, 1) \times (1, -M) \times (1, M)$ for the orientifold images. In other words, we require a flux quantization condition on $(\mathbf{T}^2)_3$ as in (4.1), up to a sign flip.

Since now the last complex dimension is compact, there is an extra RR tadpole cancellation condition, which requires the introduction of 16 O7₃-planes, wrapped on $(\mathbf{T}^2)_1 \times (\mathbf{T}^2)_2$ and localized at fixed points in $(\mathbf{T}^2)_1$, namely with wrapping numbers $(1,0) \times (1,0) \times (0,1)$. This introduces an extra orbifold action generated by $(z_1, z_2, z_3) \rightarrow (z_1, -z_2, -z_3)$, so the model can be regarded as a (T-dual of a) magnetized version of the D9/D5-brane $\mathbf{T}^4/\mathbf{Z}_2$ orientifolds in [78, 79]. Allowing for *n* additional mobile D7₃-branes (as counted in the covering space, and arranged in orbifold and orientifold invariant sets), the RR tadpole cancellation conditions is

$$2M^2 + n = 32. (4.7)$$

The supersymmetry condition is simply that the \mathbf{T}^2 parameters satisfy $\chi_2 = \chi_3$.

From the perspective of the original 6d configuration, the tadpole in the initial $\mathbf{T}^2 \times \mathbf{T}^2$ configuration has triggered a spontaneous compactification. Since the additional O-planes and D-branes required to cancel the new RR tadpoles are localized in z_3 , they can be interpreted as the addition of I-branes to cancel the cobordism charge of the original model.

It should be possible to generalize the above kind of construction to global K3-fibered CY threefolds with O7-planes. The local fibration in a small neighbourhood of a generic point of the base provides a local 6d model essentially identical to our previous one. On

¹²To be precise, the cobordism defects of an \mathbf{S}^1 compactification of the model. This is analogous to the mechanism by which F-theory on half a \mathbf{P}_1 provides the cobordism defect for type IIB on \mathbf{S}^1 with SL(2, **Z**) monodromy [14] (see also [16]). In fact, since magnetized branes often lead to chiral theories in the bulk, this extra circle compactification allows them to become non-chiral and admit an end of the world describable at weak coupling, see the discussion below (5.14) in section 5.3. We will nevertheless abuse language and refer as cobordism defect to the structures involved in the final spontaneous compactification under discussion.

¹³If the magnetization is in the U(1) \subset U(N) of a stack of N coincident branes, see footnote 10, the corresponding wrapping goes as (N, M).

the other hand, the global geometry defining how the two extra dimensions compactify would correspond to another possible spontaneous compactification, with the ingredients required for the cancellation of the new RR tadpoles.

However, a general drawback of this class of models is that the scales of the compact spaces in the directions 2 and 3 are of the same order.¹⁴ Thus, there is no separation of scales, and no reliable regime in which the dynamics becomes that of a 6d model. This is easily avoided in more involved models, as we will see in the examples in coming sections.

5 Solving tadpoles in 10d strings

In this section we consider dynamical tadpoles arising in several 10d string theories, and confirm the general picture. We illustrate this with various examples, with superymmetry (massive type IIA and M-theory on K3), and without it (non-supersymmetric 10d USp(32) theory).

5.1 Massive IIA theory

We consider 10d massive type IIA theory [80]. This can be regarded as the usual type IIA string theory in the presence of an additional RR 0-form field strength $F_0 \equiv m$. The string frame effective action for the relevant fields is

$$S_{10} = \frac{1}{2\kappa_{10}^2} \int d^{10}x \sqrt{-G} \left\{ e^{-2\phi} [R + 4(\partial\phi)^2] - \frac{1}{2} (F_0)^2 - \frac{1}{2} (F_4)^2 \right\} + S_{\text{top}}, \qquad (5.1)$$

where S_{top} includes the Chern-Simons terms. In the Einstein frame $G_E = e^{-\frac{\phi}{2}}G$, we have

$$S_{10,E} = \frac{1}{2\kappa^2} \int d^{10}x \sqrt{-G_E} \left\{ \left[R - \frac{1}{2} (\partial \phi)^2 \right] - \frac{1}{2} e^{\frac{5}{2}\phi} m^2 - \frac{1}{2} e^{\frac{1}{2}\phi} (F_4)^2 \right\}.$$
 (5.2)

Here we have used m to emphasize this quantity is constant. This theory is supersymmetric, but at a given value of ϕ , it has a tadpole controlled by

$$\mathcal{T} \sim e^{\frac{3}{2}\phi} m^2 \,. \tag{5.3}$$

This is in particular why the massive IIA theory does not admit 10d maximally symmetric solutions. In the following we discuss two different ways of solving it, leading to Minkowski or AdS-like configurations.

5.1.1 Solution in 9d and type I' as cobordism

To solve the tadpole (5.3) we can consider a well-known 1/2 BPS solution with the dilaton depending on one coordinate x^9 . Since the flux m can be regarded as generated by a set of m distant D8-branes, this is closely related to the solution in [81]. We describe it in

¹⁴In the toroidal example, if the magnetization along $(\mathbf{T}^2)_2$ is on the overall $U(1) \subset U(16)$ of 16 coincident D7-branes, the magnetic field along z_2 is $F \sim M/16$; this weakened tadpole implies an increase of the critical size of the spontaneously compactified dimensions by a factor of 4.

conventions closer to [17], for later use. In the Einstein frame, the metric and dilaton background have the structure

$$(G_E)_{MN} = Z(x^9)^{\frac{1}{12}} \eta_{MN}, \quad e^{\phi} = Z(x^9)^{-\frac{5}{6}}, \quad \text{with } Z(x^9) \sim B - mx^9,$$
 (5.4)

where B is some constant (in the picture of flux generated by distant D8-branes, it relates to the D8-brane tensions). The solution hits a singularity at $x^9 = B/m$. Starting at a general position x^9 , the distance to the singularity is

$$\Delta = \int_{x^9}^{\frac{B}{m}} Z(x^9)^{\frac{1}{24}} dx^9 \sim Z(x^9)^{\frac{25}{24}} m^{-1} \sim m^{-1} e^{-\frac{5}{4}\phi}, \qquad (5.5)$$

where in the last equality we have traded the position for the value the dilaton takes there. Recalling (5.3), this reproduces the Finite Distance scaling relation (1.1) with n = 2,

$$\Delta^{-2} \sim \mathcal{T} \,. \tag{5.6}$$

It is easy to propose the stringy mechanism capping off spacetime before or upon reaching this singularity, according to the Dynamical Cobordism lesson. This should be the cobordism defect of type IIA theory, which following [14] is an O8-plane, possibly with D8-branes.

In fact, this picture is implicitly already present in [17], which studies type I' theory, namely type IIA on an interval, namely IIA on S^1 modded out by $\Omega \mathcal{R}$ with $\mathcal{R} : x^9 \to -x^9$, which introduces two O8⁻-planes which constitute the interval boundaries. There are 32 D8-branes (in the covering space), distributed on the interval, which act as domain walls for the flux $F_0 = m$, which is piecewise constant in the interval. The metric and the dilaton profile are controlled by a piecewise linear function $Z(x^9)$. The location of the boundaries at points of strong coupling was crucial to prevent contradiction with the appearance of certain enhanced symmetries in the dual heterotic string (the role of strong coupling at the boundaries for the enhancements was also emphasized from a different perspective in [82, 83]). In our setup, we interpret the presence of (at least, one) O8-plane as the cobordism defect triggered by the presence of a dynamical tadpole in the bulk theory.

5.1.2 A non-supersymmetric Freund-Rubin solution

We now consider for illustration a different mechanism to cancel the dynamical tadpole, which in fact underlies the spontaneous compactification to (non-supersymmetric¹⁵) $AdS_4 \times$ \mathbf{S}^6 in [80]). The idea is that, rather than solving for the dilaton directly, one can introduce an additional flux F_4 along three space dimensions and time (or its dual F_6 on six space dimensions) to balance off the dilaton sourced by F_0 . This can be used to fix ϕ to a constant, and following [80] leads to a scaling

$$F_4 \sim m^2 \, d(\mathrm{vol})_4 \,, \tag{5.7}$$

¹⁵Thus, it should be unstable according to [46]. However, being at a maximum of a potential is sufficient to avoid dynamical tadpoles, so the solution suffices for our present purposes.

where $d(vol)_4$ is the volume form in the corresponding 4d. Using arguments familiar by now, the constant F_4 backreaction on the metric is encoded in a solution of the 4d Laplace equation with a constant source, leading to a solution quadratic in the coordinates (to avoid subtleties, we take solutions depending only on the space coordinates). This develops a singularity at a distance scaling as

$$\Delta^2 \sim |F_4|^{-2} \sim m^{-4} \sim \mathcal{T} \,, \tag{5.8}$$

where in comparison with (5.3) we have taken constant dilaton.

The singularities are avoided by an $\operatorname{AdS}_4 \times \mathbf{S}^6$ compactification, whose curvature radius is $R \sim m^2$, in agreement with the above scaling. From our perspective, the compactification should be regarded as a dynamical cobordism (where the cobordism is actually that of the 10d theory on an \mathbf{S}^5 (i.e. equator of \mathbf{S}^5)).

5.2 An aside on M-theory on K3

In this section we relate the above system to certain compactifications of M-theory and to the Horava-Witten end of the world branes as its cobordism defect. Although the results can be obtained by direct use of M-theory effective actions, we illustrate how they can be recovered by applying simple dualities to the above system.

Consider the above massive IIA theory with mass parameter m, and compactify on $\mathbf{T}^4/\mathbf{Z}_2$. This introduces O4-planes, and requires including 32 D4-branes in the configuration, either as localized sources, or dissolved as instantons on the D8-branes. Actually this can be considered as a simple model of K3 compactifications, where in the general K3 the O4-plane charge is replaced by the contribution to the RR C_5 tadpole arising from the CS couplings of D8-branes and O8-planes to tr \mathbb{R}^2 .

We now perform a T-duality in all the directions of the $\mathbf{T}^4/\mathbf{Z}_2$ (Fourier-Mukai transform in the case of general K3). We obtain a similar model of type I' on $\mathbf{T}^4/\mathbf{Z}_2$, but now with the tadpole being associated to the presence of m units of non-trivial flux of the RR 4-form field-strength over \mathbf{T}^4 (namely, K3). Also, the dilaton of the original picture becomes related to the overall Kähler modulus of K3. Finally, we lift the configuration to M-theory by growing an extra \mathbf{S}^1 and decompactifying it. We thus end up with a 7d compactification of M-theory on K3, with m units of G_4 flux,

$$\int_{K3} G_4 = m \,. \tag{5.9}$$

This leads to a dynamical tadpole, cancelled by the variation of the overall Kähler modulus (i.e. the K3 volume) along one the 7d space dimensions, which we denote by x^{11} . As in previous sections, this will trigger a singularity at a finite distance in x^{11} , related to the tadpole by $\Delta^{-2} \sim \mathcal{T}$. The singularity is avoided by the physical appearance of a cobordism defect, which for M-theory is a Horava-Witten (HW) boundary [84]. This indeed can support the degrees of freedom to kill the G_4 flux, as follows. From [85], the 11d G_4 is sourced by the boundary as

$$dG_4 = \delta(x^{11}) \left(\operatorname{tr} F^2 - \frac{1}{2} \operatorname{tr} R^2 \right), \qquad (5.10)$$

where $\delta(x^{11})$ is a bump 1-form for the HW brane, and F is the field-strength for the E_8 gauge fields in the boundary. Hence, the m units of G_4 in the K3 compactification can be absorbed by a HW boundary with an E_8 bundle with instanton number 12 + m (the 12 coming from half the Euler characteristic of K3 $\int_{K3} \operatorname{tr} R^2 = 24$).

The above discussion is closely related to the picture in [18], which discusses compactification of HW theory (namely, M-theory on an interval with two HW boundaries) on K3 and on a CY threefold. It includes a Kähler modulus varying over the interval according to a linear function¹⁶ and the appearance of a singularity at finite distance. In that case, the HW brane was located at the strong coupling point, based on heuristic arguments, and this led, in the CY₃ case, to a lower bound on the value of the 4d Newton's constant.

Our perspective remarkably explains that the location of the HW wall is not an arbitrary choice, but follows our physical principle of Dynamical Cobordism, and the bound on the Newton's constant is a consequence of that of Finite Distance!

5.3 Solving tadpoles in the non-supersymmetric 10d USp(32) theory

The previous examples were based on an underlying supersymmetric vacuum, on top of which the dynamical tadpole is generated via the introduction of fluxes or other ingredients. In this section we consider the opposite situation, in which the initial theory is strongly non-supersymmetric and displays a dynamical tadpole from the start. In particular we consider the non-supersymmetric 10d USp(32) theory constructed in [4], in two different ways: first, we use our new insights to revisit the spacetime-dependent solution proposed in [9] (see also [10] for other proposals); then we present a far more tractable solution involving magnetization, which in fact provides a supersymmetric compactification of this non-supersymmetric 10d string theory.

5.3.1 The Dudas-Mourad solution and cobordism

The non-supersymmetric 10d USp(32) theory in [4] is obtained as an Ω orientifold of type IIB theory. The closed string sector is as in type I theory, except that the O9⁻-plane is replace by an O9⁺-plane. Cancellation of RR tadpoles requires the introduction of open strings, which must be associated to 32 D9-branes. The closed string sector is a 10d $\mathcal{N} = 1$ supergravity multiplet; the orientifold action on the D9-branes breaks supersymmetry, resulting in an open string sector with USp(32) gauge bosons and gauginos in the two-index antisymmetric representation. All anomalies cancel, a remarkable feat from the field theory viewpoint, which is just a consequence of RR tadpole cancellation from the string viewpoint.

Although the RR tadpoles cancel, the NSNS tadpoles do not, implying that there is no maximally symmetric 10d solution to the equations of motion. In particular there is a dynamical dilaton tadpole of order the string scale, as follows from the terms in the 10d (Einstein frame) action

$$S_E = \frac{1}{2\kappa^2} \int d^{10}x \sqrt{-G} \left[R - \frac{1}{2} (\partial\phi)^2 \right] - T_9^E \int d^{10}x \sqrt{-G} \, 64 \, e^{\frac{3\phi}{2}} \,, \tag{5.11}$$

¹⁶In the presence of explicit M5-branes, it is a piecewise linear function. It is straightforward to include them in our cobordism description if wished, with explicit branes considered as part of the cobordism defect.

where T_9^E is the (anti)D9-brane tension. The tadpole scales as $\mathcal{T} \sim T_9^E g_s^{3/2}$, with the dilaton dependence arising from the fact that the supersymmetry breaking arises from the Moebius strip worldsheet topology, with $\chi = 3/2$.

Ref. [9] proposed solutions of this dynamical tadpole with 9d Poincaré invariance, and the dilaton varying over one spacetime dimension (see also [86, 87] for more recent, related work). In the following we revisit the solution with dependence on one spatial coordinate y, from the vantage point of our Lessons.

The 10d solution is, in the Einstein frame,

$$\phi = \frac{3}{4} \alpha_E y^2 + \frac{2}{3} \log |\sqrt{\alpha_E} y| + \phi_0 ,$$

$$ds_E^2 = |\sqrt{\alpha_E} y|^{\frac{1}{9}} e^{-\frac{\alpha_E y^2}{8}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + |\sqrt{\alpha_E} y|^{-1} e^{-\frac{3\phi_0}{2}} e^{-\frac{9\alpha_E y^2}{8}} dy^2 , \qquad (5.12)$$

where $\alpha_E = 64k^2T_9$. There are two singularities, at y = 0 and $y \to \infty$, which despite appearances are separated by a finite distance

$$\Delta \sim \int_0^\infty |\sqrt{\alpha_E}y|^{-\frac{1}{2}} e^{-\frac{3\phi_0}{4}} e^{-\frac{9\alpha_E y^2}{16}} \, dy \sim e^{-\frac{3\phi_0}{4}} \alpha_E^{-\frac{1}{2}} \,. \tag{5.13}$$

The fact that the solution has finite extent in the spatial dimension on which the fields vary is in agreement with the Finite Distance Lesson, and in fact satisfying its quantitative bound (1.1)

$$\Delta^{-2} \sim \mathcal{T} \,. \tag{5.14}$$

We can now consider how the Dynamical Cobordism Lesson applies in the present context. Following it, we expect the finite extent in the spatial dimensions to be physically implemented via the cobordism defect corresponding to the 10d USp(32) theory. In general the cobordism defect of bulk chiral 10d theories are expected to be non-supersymmetric, and in fact rather exotic, as their worldvolume dynamics must gap a (non-anomalous) set of chiral degrees of freedom. In fact, on general grounds they can be expected to involve strong coupling.¹⁷ An end of the world defect imposes boundary conditions on bulk supergravity fields, which at weak coupling should be at most linear in the fields, to be compatible with the superposition principle. A typical example are boundary conditions that pair up bulk fermions of opposite chiralities. However, the anomaly cancellation in the 10d USp(32) theory involves fields of different spins, which cannot be gapped by this simple mechanism, and should require strong coupling dynamics (a similar phenomenon in a different context occurs in [88]).

This strong coupling fits nicely with the singularity at $y \to \infty$, but the singularity at y = 0 lies at weak coupling. The simplest way out of this is to propose that the singularity at y = 0 is actually smoothed out by perturbative string theory (namely, α' corrections, just like orbifold singularities are not singular in string theory), and does not turn into an end of the world defect. Hence the solution (5.12) extends to y < 0, and, since the background is even in y, develops a singularity at $y \to -\infty$. This is still at finite distance Δ scaling as (5.14), and lies at strong coupling, thus allowing for the possibility that the singularity is turned into the cobordism defect of the 10d USp(32) theory.

¹⁷We are indebted to Miguel Montero for this argument, and for general discussions on this section.

It would be interesting to explore this improved understanding of this solution to the dilaton tadpole. Leaving this for future work, we turn to a more tractable solution in the next section.

5.3.2 Solving the tadpole via magnetization

We now discuss a more tractable alternative to solve the dynamical tadpole via magnetization, following section 4.

Stabilizing the tadpole via magnetization is, ultimately, equivalent to finding a compactification (on a product of \mathbf{T}^2 's) which is free of tadpoles, for instance by demanding it to be supersymmetric. Hence we need to construct a supersymmetric compactification of the non-supersymmetric 10d USp(32) theory [4].

As explained above, the 10d model is constructed with an O9⁺-plane and 32 $\overline{\text{D9}}$ -branes. Hence, we need to introduce worldvolume magnetic fields in different 2-planes, in such a way that the corresponding angles add up to 0 mod 2π . It is easy to convince oneself that this requires magnetization in at least three complex planes, ultimately triggering a $\mathbf{T}^2 \times \mathbf{T}^2 \times \mathbf{T}^2 \times \mathbf{T}^2$ compactification. In order to preserve supersymmetry, we need the magnetization to induce D5-brane charges, rather than $\overline{\text{D5}}$ -brane charge, hence we need the presence of three independent kinds of negatively charged $O5_i^-$ -planes, where i = 1, 2, 3 denotes the \mathbf{T}^2 wrapped by the corresponding O5-plane. We are thus considering an orientifold of $\mathbf{T}^6/(\mathbf{Z}_2 \times \mathbf{Z}_2)$ with an O9⁺-plane, and 8 $O5_i^-$ -planes.¹⁸

The wrapping numbers for the O-planes, and for one simple solution of all constraints for the D9-branes (and their explicitly included orientifold image D9-branes), are

Object	N_{α}	(n^1_α,m^1_α)	$(n_{\alpha}^2,m_{\alpha}^2)$	(n_{lpha}^3,m_{lpha}^3)
$O9^+$	32	(1, 0)	(1, 0)	(1, 0)
$O5_{1}^{-}$	-32	(1, 0)	(0,1)	(0, -1)
$O5_{2}^{-}$	-32	(0, 1)	(1, 0)	(0, -1)
$O5_{3}^{-}$	-32	(0, 1)	(0, -1)	(1, 0)
D9	16	(-1, 1)	(-1, 1)	(-1, 1)
D9'	16	(-1, -1)	(-1, -1)	(-1, -1)

It obeys the RR tadpole conditions for the **Z**-valued D9- and D5-brane charges, and the discrete \mathbf{Z}_2 RR tadpole conditions for D3- and D7_i-brane charges [70].

The supersymmetry condition determined by the O-plane wrappings is

$$\sum_{i} \arctan(-\chi_i) \equiv \theta_1 + \theta_2 + \theta_3 = 0 \mod 2\pi.$$
(5.15)

The model is in fact T-dual (in all \mathbf{T}^6 directions) to that in section 5 of [59].

It is easy to see that the above condition forces at least one of the \mathbf{T}^2 to have $\mathcal{O}(1)$ area in α' units. From our perspective, this a mere reflection of the fact that the 10d dynamical

¹⁸For such combinations of orientifold plane signs, see the analysis in [89], in particular its table 6. We will not need its detailed construction for our purposes.

tadpole to be canceled is of order the string scale, hence it agrees with the scaling $\Delta^{-2} \sim \mathcal{T}$. Happily, the use of an α' exact configuration, which is moreover supersymmetric, makes our solution reliable. This is an improvement over other approaches e.g. as in section 5.3.1.

Although we have discussed the compactification on (an orientifold of) \mathbf{T}^6 directly, we would like to point out that it is easy to describe it as a sequence of \mathbf{T}^2 spontaneous compactifications, each eating up a fraction of the initial 10d tadpole until it is ultimately cancelled upon reaching \mathbf{T}^6 . However, this picture does not really correspond to a physical situation, given the absence of decoupling of scales. This is true even in setups which seemingly allow for one \mathbf{T}^2 of parametrically large area. Indeed, consider for instance the regime $\chi_3 \sim 2\lambda$ and $\chi_1, \chi_2 \sim \lambda^{-1}$, for $0 < \lambda \ll 1$, which corresponds to $\theta_1, \theta_2 \sim \frac{\pi}{2} + \lambda$, $\theta_3 \sim \pi - 2\lambda$. This corresponds to a compactification on substringy size $(\mathbf{T}^2)_1 \times (\mathbf{T}^2)_2$ and a parametrically large $(\mathbf{T}^2)_3$. However, the fact that the $(\mathbf{T}^2)_1, (\mathbf{T}^2)_2$ can be T-dualized into large area geometries shows that there is not true decoupling of scales: in the original picture, the small sizes imply that there are towers of light winding modes, whose scale is comparable with the KK modes of $(\mathbf{T}^2)_3$. Hence, the lack of decoupling is still present, as expected from our general considerations in the introduction.

6 The SM from spontaneous compactification

In this section we explore an interesting application of the above mechanism, and provide an explicit example of a 6d theory with brane-antibrane pairs, and a dynamical tadpole triggering spontaneous compactification to a 4d (MS)SM-like particle physics model. Interestingly, the complete chiral matter and electroweak sector, including the Higgs multiplets, are generated as degrees of freedom on cobordism branes. Only the gluons are present in some form in the original 6d models.

Consider the type IIB orientifold of $\mathbf{T}^4/\mathbf{Z}_2$ with orientifold action Ω constructed in [78, 79], possibly with magnetization. To describe it, we introduce the notation in [69, 90] of wrapping numbers $(n^i_{\alpha}, m^i_{\alpha})$, where n^i_{α} and m^i_{α} provide the wrapping number and magnetic flux quantum of the D-brane α on the *i*th \mathbf{T}^2 , respectively. We consider the following stacks of D-branes (and their orientifold images, not displayed explicitly)

N_{lpha}	(n^1_α,m^1_α)	$(n_{\alpha}^2, m_{\alpha}^2)$
$N_{a+d} = 6 + 2$	(1,3)	(1, -3)
$N_{h_1} = 4$	(1, -3)	(1, -4)
$N_{h_2} = 4$	(1, -4)	(1, -3)
40	(0, 1)	(0, -1)

The O9- and O5-planes correspond to the wrapping numbers $(1,0) \times (1,0)$ and $(0,1) \times (0,-1)$ respectively. The stacks *a* and *d* are taken different and separated by Wilson lines, but they can be discussed jointly for the time being. They correspond to 8 D9-branes with worldvolume magnetic fluxes 72 units of D5-brane charge. The stacks h_1 and h_2 correspond to 8 additional D9-branes, with 96 with units of induced $\overline{\text{D5}}$ -branes charge.

The addition of 40 explicit D5-branes leads to RR tadpole cancellation (once orientifold images are included). In terms of the wrapping numbers, we have

$$\sum_{\alpha} N_{\alpha} n_{\alpha}^2 n_{\alpha}^3 = 16 , \quad \sum_{\alpha} N_{\alpha} m_{\alpha}^2 m_{\alpha}^3 = -16 .$$
 (6.1)

The model is far from supersymmetric due to the presence of D5- $\overline{\text{D5}}$ pairs, and in fact has a decay channel to supersymmetric model by their annihilation. On the other hand, even at the top of the tachyon potential, the theory is not at a critical point of its potential due to dynamical tadpole for the closed string moduli, namely the area moduli of the \mathbf{T}^{2} 's. In other words, the excess tension depends on these, as they enter the angles determining the deviation from the supersymmetry condition

$$\arctan\left(\frac{m_{\alpha}^{1}}{n_{\alpha}^{1}}\chi_{1}\right) + \arctan\left(\frac{m_{\alpha}^{2}}{n_{\alpha}^{2}}\chi_{2}\right) = 0.$$
(6.2)

For instance, we can make the stacks a, d supersymmetric, by choosing

$$\chi_1 = \chi_2 \,, \tag{6.3}$$

but the D-branes h_1 and h_2 break supersymmetry. Hence, there is a dynamical tadpole associated to the excess tension of these latter objects.

The dynamical tadpole can be solved by introducing magnetization along two of the 6d spacetime dimensions. The backreaction of this extra flux forces these two dimensions to be compactified on a \mathbf{T}^2 , with the addition of cobordism I-branes [15], which in general includes orientifold planes and D-branes, as in the examples above. We take these extra branes to be arranged in two new stacks b and c. Overall, we end up with an orientifold of $\mathbf{T}^6/(\mathbf{Z}_2 \times \mathbf{Z}_2)$, with D-brane stacks and topological numbers given by

N_{lpha}	(n^1_α,m^1_α)	$(n_{\alpha}^2,m_{\alpha}^2)$	$(n_{\alpha}^3,m_{\alpha}^3)$
$N_{a+d} = 6 + 2$	(1,3)	(1, -3)	(1, 0)
$N_b = 2$	(0,1)	(1, 0)	(0, 1)
$N_c = 2$	(-1, 0)	(0, -1)	(0, 1)
$N_{h_1} = 2$	(1, -3)	(1, -4)	(2, -1)
$N_{h_2} = 2$	(1, -4)	(1, -3)	(2, -1)
40	(0, 1)	(0, -1)	(0, 1)

The model satisfies the RR tadpole conditions

$$\sum_{\alpha} N_{\alpha} n_{\alpha}^{1} n_{\alpha}^{2} n_{\alpha}^{3} = 16, \qquad \sum_{\alpha} N_{\alpha} n_{\alpha}^{1} m_{\alpha}^{2} m_{\alpha}^{3} = 16,$$
$$\sum_{\alpha} N_{\alpha} m_{\alpha}^{1} n_{\alpha}^{2} m_{\alpha}^{3} = 16, \qquad \sum_{\alpha} N_{\alpha} m_{\alpha}^{1} m_{\alpha}^{2} n_{\alpha}^{3} = -16.$$
(6.4)

This corresponds to O9-planes along $(1,0) \times (1,0) \times (1,0)$, and O5-planes along $(0,1) \times (0,-1) \times (1,0)$, as already present in the 6d theory, and cobordism O5-planes along $(0,1) \times (1,0) \times (0,1)$ and $(1,0) \times (0,1) \times (0,1)$.

The model still contains only 3 stacks of D-branes with non-trivial angles, so that they are just enough to fix the 2 parameters χ_i of the \mathbf{T}^2 's. The O-planes fix the supersymmetry condition signs to

$$\arctan\left(\frac{m_{\alpha}^{1}}{n_{\alpha}^{1}}\chi_{1}\right) + \arctan\left(\frac{m_{\alpha}^{2}}{n_{\alpha}^{2}}\chi_{2}\right) - \arctan\left(\frac{m_{\alpha}^{3}}{n_{\alpha}^{3}}\chi_{3}\right) = 0.$$
(6.5)

Using the branes above, we get

$$\chi_1 = \chi_2, \quad \chi_3 = \frac{14\chi_1}{1 - 12\chi_1^2}.$$
 (6.6)

The regime of large $(\mathbf{T}^2)_3$ corresponds to small χ_3 , which is also attained for small χ_1 . Note that in this context the last condition $\chi_1 \sim \chi_3$ encodes the relation between the 6d tadpole and the inverse area of the spontaneously compactified \mathbf{T}^2 .

The model is, up to exchange of directions in the \mathbf{T}^6 and overall sign flips, precisely one of the examples of 4d MSSM-like constructions in [59, 60]. The gauge group is $U(3)_a \times$ $USp(2)_b \times U(1)_c \times U(1)_d$, where we break the naive $USp(2)_c$ by Wilson lines or shifting off the O-plane for the corresponding D5-branes. Taking into account the massive U(1)'s due to *BF* couplings, this reproduces the SM gauge group. In addition, open strings between the different brane stacks reproduce a 3-family (MS)SM chiral matter content, and the MSSM Higgs doublet pair. Hence, we have described the spontaneous compactification of a 6d model to a semi-realistic MSSM-like 4d theory.

A fun fact worth emphasizing is that most of the SM spectrum is absent in the original 6d model, and arises only after the spontaneous compactification. In particular, all the MSSM matter and Higgs chiral multiplets, as well as the electroweak gauge sector, arise from open string sectors involving the b and c branes, which arises as cobordism branes. It is remarkable that cobordism entails that spontaneous compactification implies not just the removal of spacetime dimensions, but also the dynamical appearance of novel degrees of freedom. It is tantalizing to speculate on the potential implications of these realizations in cosmological or other dynamical setups.

Acknowledgments

We are pleased to thank Inaki García-Etxebarria, Luis Ibánez, Fernando Marchesano, Miguel Montero and Irene Valenzuela for useful discussions. This work is supported by the Spanish Research Agency (Agencia Española de Investigación) through the grants IFT Centro de Excelencia Severo Ochoa SEV-2016-0597, the grant GC2018-095976-B-C21 from MCIU/AEI/FEDER, UE.

A Dynamical tadpoles and swampland constraints

In this appendix we use the model in section 2 to illustrate the result in [3] that, in theories with a dynamical tadpole which is not duly backreacted on the field configuration, the mistreatment can show up as violations of swampland constraints.

We consider type IIB theory on $AdS_5 \times T^{1,1}$ and introduce M units of RR 3-form flux. In the coordinates in [33, 34], it reads

$$F_3 = \frac{1}{2}M[\sin\theta_1(\cos\theta_2d\theta_1d\phi_1d\phi_2 + d\theta_1d\phi_1d\psi) + \sin\theta_2(\cos\theta_1d\theta_2d\phi_1d\phi_2 - d\theta_2d\phi_2d\psi)].$$

It has constant coefficients in terms of $f\ddot{u}nf$ -bein 1-forms g^i in [33] $F_3 = \frac{1}{2}Mg^5(g^1g^2 + g^3g^3)$, hence its kinetic term $|F_3|^3$ is constant over the $T^{1,1}$ geometry. This acts as a constant background source for e.g. the Laplace equation for the dilaton, which has no solution over the compact $T^{1,1}$ geometry. This inconsistency of the equations of motion, assuming no backreaction on the underlying geometry, signals the dynamical tadpole in the configuration. In the following we will argue that it moreover can lead to violation of the Weak Gravity Conjecture [47].

For concreteness we focus on the simplest set of states, corresponding to 5d BPS particle states in the original theory (M = 0), with the BPS bound corresponding to the WGC bound, for the gauge interaction associated to the KK U(1) dual to the U(1)_R symmetry of the dual CFT. For small R-charge $n \ll N$, these particle states are dual to chiral primary single-trace mesonic operators of the SU(N)² theory, e.g. tr $(A_1B_1 \ldots A_1B_1)$; in the AdS side, they correspond to KK gravitons with momentum n on the \mathbf{S}^1 . For very large R-charge, the KK gravitons polarize due to Myers' effect [91] into giant gravitons [92], and their dual operators are determinant or sub-determinant operators [93]. Note that on $T^{1,1}$ we have D3-branes wrapped on homologically trivial 3-cycles (but sustained as BPS states by their motion on \mathbf{S}^1), hence they are different from (di)baryonic operators, which correspond to D3-branes wrapped on the non-trivial \mathbf{S}^3 [94].

Our strategy is to consider these states in the presence of F_3 , but still keeping the geometry as $\operatorname{AdS}_5 \times T^{1,1}$ (i.e. with no backreaction of the dynamical tadpole), and show that the interaction of F_3 makes these states non-BPS, hence violating the WGC bound. This analysis will be quite feasible in the giant graviton regime $1 \ll n \sim N$, by using the wrapped D3-brane worldvolume action. Admittedly, proving a full violation of the WGC would require showing the violation of the BPS condition for all values of n; we nevertheless consider the large n result as a compelling indication that the WGC is indeed violated in this configuration, thus making its inconsistency manifest.

Supersymmetric 3-cycles for D3-branes are easily obtained from holomorphic 4-cycles in the underlying CY threefold [95, 96] (see also [97]). Describing the conifold as $z^1z^2 - z^3z^4 = 0$, any holomorphic function of these coordinates f(x, y, z, w) = 0 defines a holomorphic 4-cycle corresponding to a giant graviton D3-branes, i.e. wrapped on a trivial¹⁹ 3-cycle in $T^{1,1}$. We focus on a simple class of D3-branes studied in detail in [98]. They are defined by the 4-cycle $z^1 = \sqrt{1 - \alpha^2}$, with $\alpha \in [0, 1]$ being a real constant, encoding the size of the 3-cycle (with $\alpha = 0, 1$ corresponding to the pointlike KK graviton and the maximal giant graviton, respectively). We will follow the analysis in [98] with the inclusion of the effect of F_3 on the D3-brane probe.

¹⁹Di-baryonic D3-branes are on the other hand associated to non-Cartier divisors in the conifold, i.e. 4-cycles which can be defined in terms of the a_i , b_i homogeneous coordinates of the linear sigma model, but cannot be expressed as a single equation $f(z^i) = 0$.

It is convenient to change to new coordinates $\{\chi_1, \chi_2, \chi_3, \alpha, \nu\}$

$$\begin{cases} \chi_1 = \frac{1}{3}(\psi - \phi_1 - \phi_2) \\ \chi_2 = \frac{1}{3}(\psi + 3\phi_1 - \phi_2) \\ \chi_3 = \frac{1}{3}(\psi - \phi_1 + 3\phi_2) \end{cases} \begin{cases} \sqrt{1 - \alpha^2} = \sin\frac{\theta_1}{2}\sin\frac{\theta_2}{2} \\ \nu = \frac{2u}{\alpha^2 + u^2} \text{ with } u = \cos\frac{\theta_1}{2}\cos\frac{\theta_2}{2} \end{cases}$$
(A.1)

These are adapted to the D3-brane embedding, which simply reads

$$\sigma^0 = t, \ \sigma^1 = \nu \ (\text{doubly-covered}), \ \sigma^2 = \chi_2, \ \sigma^3 = \chi_3 \,.$$

The double covering is very manifest for the maximal giant graviton, $\alpha = 1$, $z^1 = 0$. It corresponds to the defining equation $z^3 z^4 = 0$, which splits in two components, corresponding to two (oppositely oriented) copies of the non-trivial²⁰ \mathbf{S}^3 . The double covering remains even for non-maximal giants, even though they correspond to irreducible 4-cycles.

The RR 3-form field strength in these coordinates is

$$F_{3} = M([a_{12} d\chi_{1} \wedge d\chi_{2} + a_{13} d\chi_{1} \wedge d\chi_{3} + a_{23} d\chi_{2} \wedge d\chi_{3}] \wedge d\alpha + [v_{12} d\chi_{1} \wedge d\chi_{2} + v_{13} d\chi_{1} \wedge d\chi_{3} + v_{23} d\chi_{2} \wedge d\chi_{3}] \wedge d\nu),$$
(A.2)

with

$$\begin{cases} a_{12} = \frac{9}{4}\alpha(1 \pm \frac{\sqrt{1-\nu^2}}{1-c}) \\ a_{13} = \frac{9}{4}\alpha(-1 \pm \frac{\sqrt{1-\nu^2}}{1-c}) \\ a_{23} = \frac{9}{4}\alpha(\frac{-c}{1-c}) \end{cases} \qquad \begin{cases} v_{12} = \mp \frac{9}{4} \frac{c(\nu^2 - c)}{\nu^3 \sqrt{1-\nu^2}(1-c)} \\ v_{13} = \mp \frac{9}{4} \frac{c(\nu^2 - c)}{\nu^3 \sqrt{1-\nu^2}(1-c)} \\ v_{23} = -\frac{9}{4} \frac{c^2}{\nu^3(1-c)} \end{cases}$$

where we have introduced $c = 1 - \sqrt{1 - \alpha^2 \nu^2}$. We can fix a gauge and find the RR 2-form

$$C_2 = M(c_{12}d\chi_1 \wedge d\chi_2 + c_{13}d\chi_1 \wedge d\chi_3 + c_{23}d\chi_2 \wedge d\chi_3),$$
 (A.3)

with

$$\begin{cases} c_{12} = -\frac{9}{8}(-\alpha^2 \mp \frac{2\sqrt{1-\nu^2}c}{\nu^2}) \\ c_{13} = -\frac{9}{8}(\alpha^2 \mp \frac{2\sqrt{1-\nu^2}c}{\nu^2}) \\ c_{23} = \frac{9}{8}\frac{(\alpha^2\nu^2 - 2c)}{\nu^2} \end{cases}$$

Its pullback on the D3-brane worldvolume is

$$P[C_2] = M\dot{\chi}_1(c_{12}dt \wedge d\chi_2 + c_{13}dt \wedge d\chi_3) + Mc_{23}d\chi_2 \wedge d\chi_3.$$
(A.4)

We can now compute the effect of this background on the D3-brane by using its worldvolume action. This is easy in the S-dual frame, in which the RR 2-form couples to the D3-brane just like the NSNS 2-form in the original DBI+CS D3-brane action.²¹ After integrating over χ_2 , χ_3 , this reads

$$S = S_{\rm BDI} + S_{\rm CS} = \frac{64\pi^2}{9} \int dt L , \qquad (A.5)$$

with $L = \int_0^1 d\nu \ 2 \left(-T_3 \sqrt{-\det\left(P[G]_{\mu\nu} + P[C_2]_{\mu\nu}\right)} + \mu_3 R^4 c_4 \dot{\chi}_1 \right) ,$

²⁰In terms of the linear sigma model coordinates we have $z^1 = a_1b_1$, $z^2 = a_2b_2$, $z^3 = a_1b_2$, $z^4 = a_2b_1$, and the two components correspond to $a_1 = 0$ and $b_1 = 0$, which are non-Cartier divisors.

²¹Related to this, one can check that the above background is neither pure gauge on the D3, nor cannot be removed by a change in the worldvolume gauge field strength flux.

where the factor of 2 of the double-covering of ν has been added, and the last term arises from the CS coupling to the RR 4-form as in [98].

We are interested in focusing on the angular momentum of the state $P_{\chi_1} = \frac{\partial L}{\partial \dot{\chi}_1}$ conjugate to the angular coordinate χ_1 . This reads

$$P_{\chi_1} = \frac{3}{2} \int_0^1 d\nu \left(\frac{\sqrt{3\pi} A g_\nu N T_3(\dot{\chi}_1)}{\sqrt{N g_\nu (B - A(\dot{\chi}_1)^2)}} + 9\pi c_4 \mu_3 N \right), \tag{A.6}$$

with

$$A = \frac{81}{64\nu^4} \{ -4 M^2 [(2(1-\alpha^2\nu^2)c - \alpha^2\nu^2)c] - 3\pi N(\alpha^2 - 1)\nu^2 c^2 \} \equiv A_{M^2}M^2 + A_NN,$$

$$B = \frac{81}{64\nu^4} \{ 4M^2 [(4-\alpha^2\nu^2)c^2 - 2\alpha^2\nu^2 c] + \pi N [2\alpha^4\nu^4 + (\alpha^2\nu^2 - 2c)(-3\alpha^2\nu^2 - 3\nu^2 + 8)] \}$$

$$\equiv B_{M^2}M^2 + B_NN.$$
(A.7)

In the last equalities we have highlighted the parametric dependence on N and M.

Despite the fact that we have not managed to find a closed form for the result, since $M \ll N$ we can find an expansion for the integrand in the form

$$n = p_0(\alpha, \nu, \dot{\chi}_1) N + p_2(\alpha, \nu, \dot{\chi}_1) M^2 + \mathcal{O}(M^4), \qquad (A.8)$$

where the coefficient functions are computable, but we will not need their explicit expressions.

The coefficient p_0 is the survivor for the M = 0 case, and leads to an integer momentum. On the other hand, the subleading correction p_2 produces a momentum which is not integer. This already signals a problem, since (as the geometry is considered undeformed even after introducing F_3) the gauge coupling of the KK U(1) is as in the M = 0 case, hence charges under it should be integer in the same units. Hence one can directly claim that the assumption of ignoring the dynamical tadpole backreaction lead to violation of charge quantization, in contradiction with common lore for consistency with quantum gravity [99].

The above discussion however seems to contradict the fact that any quantum excitation on a periodic \mathbf{S}^1 direction must have quantized momentum to have a well-defined wavefunction. In fact, an alternative interpretation of the above mismatch is that the D3brane probe computation assumes a well-defined worldvolume embedding, in particular well-defined (hence classical) trajectories for the 5d particle. It is only for BPS states in supersymmetric vacua that such a computation is guaranteed to end up producing quantized momenta. The fact that our holomorphic embedding ansatz fails to do so is just a reflection that the actual integer-quantized states are *not* described by holomorphic equations. Since the latter condition is the one ensuring the match between the particle mass and charge, it is clear that non-holomorphic embeddings will produce larger masses for the same charge, hence violating the BPS/WGC bound.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- W. Fischler and L. Susskind, Dilaton Tadpoles, String Condensates and Scale Invariance, Phys. Lett. B 171 (1986) 383 [INSPIRE].
- [2] W. Fischler and L. Susskind, Dilaton Tadpoles, String Condensates and Scale Invariance. 2., Phys. Lett. B 173 (1986) 262 [INSPIRE].
- [3] A. Mininno and A.M. Uranga, Dynamical tadpoles and Weak Gravity Constraints, JHEP 05 (2021) 177 [arXiv:2011.00051] [INSPIRE].
- [4] S. Sugimoto, Anomaly cancellations in type-I D9-D9 system and the USp(32) string theory, Prog. Theor. Phys. 102 (1999) 685 [hep-th/9905159] [INSPIRE].
- [5] I. Antoniadis, E. Dudas and A. Sagnotti, Brane supersymmetry breaking, Phys. Lett. B 464 (1999) 38 [hep-th/9908023] [INSPIRE].
- [6] G. Aldazabal and A.M. Uranga, Tachyon free nonsupersymmetric type IIB orientifolds via brane-antibrane systems, JHEP 10 (1999) 024 [hep-th/9908072] [INSPIRE].
- [7] A.M. Uranga, Comments on nonsupersymmetric orientifolds at strong coupling, JHEP 02 (2000) 041 [hep-th/9912145] [INSPIRE].
- [8] J. Polchinski, *String theory. Vol. 2: Superstring theory and beyond*, Cambridge Monographs on Mathematical Physics, Cambridge University Press (2007) [DOI] [INSPIRE].
- [9] E. Dudas and J. Mourad, Brane solutions in strings with broken supersymmetry and dilaton tadpoles, Phys. Lett. B 486 (2000) 172 [hep-th/0004165] [INSPIRE].
- [10] R. Blumenhagen and A. Font, Dilaton tadpoles, warped geometries and large extra dimensions for nonsupersymmetric strings, Nucl. Phys. B 599 (2001) 241 [hep-th/0011269] [INSPIRE].
- [11] E. Dudas, J. Mourad and C. Timirgaziu, Time and space dependent backgrounds from nonsupersymmetric strings, Nucl. Phys. B 660 (2003) 3 [hep-th/0209176] [INSPIRE].
- [12] E. Dudas, G. Pradisi, M. Nicolosi and A. Sagnotti, On tadpoles and vacuum redefinitions in string theory, Nucl. Phys. B 708 (2005) 3 [hep-th/0410101] [INSPIRE].
- [13] J. Mourad and A. Sagnotti, AdS Vacua from Dilaton Tadpoles and Form Fluxes, Phys. Lett. B 768 (2017) 92 [arXiv:1612.08566] [INSPIRE].
- [14] J. McNamara and C. Vafa, Cobordism Classes and the Swampland, arXiv:1909.10355 [INSPIRE].
- [15] M. Montero and C. Vafa, Cobordism Conjecture, Anomalies, and the String Lamppost Principle, JHEP 01 (2021) 063 [arXiv:2008.11729] [INSPIRE].
- [16] M. Dierigl and J.J. Heckman, Swampland cobordism conjecture and non-Abelian duality groups, Phys. Rev. D 103 (2021) 066006 [arXiv:2012.00013] [INSPIRE].
- [17] J. Polchinski and E. Witten, Evidence for heterotic type-I string duality, Nucl. Phys. B 460 (1996) 525 [hep-th/9510169] [INSPIRE].
- [18] E. Witten, Strong coupling expansion of Calabi-Yau compactification, Nucl. Phys. B 471 (1996) 135 [hep-th/9602070] [INSPIRE].
- [19] D. Lüst, E. Palti and C. Vafa, AdS and the Swampland, Phys. Lett. B 797 (2019) 134867
 [arXiv:1906.05225] [INSPIRE].

- [20] G. Buratti, J. Calderon, A. Mininno and A.M. Uranga, Discrete Symmetries, Weak Coupling Conjecture and Scale Separation in AdS Vacua, JHEP 06 (2020) 083 [arXiv:2003.09740] [INSPIRE].
- [21] C. Vafa, The String landscape and the swampland, hep-th/0509212 [INSPIRE].
- [22] T.D. Brennan, F. Carta and C. Vafa, The String Landscape, the Swampland, and the Missing Corner, PoS TASI2017 (2017) 015 [arXiv:1711.00864] [INSPIRE].
- [23] E. Palti, The Swampland: Introduction and Review, Fortsch. Phys. 67 (2019) 1900037
 [arXiv:1903.06239] [INSPIRE].
- [24] M. van Beest, J. Calderón-Infante, D. Mirfendereski and I. Valenzuela, Lectures on the Swampland Program in String Compactifications, arXiv:2102.01111 [INSPIRE].
- [25] G. Obied, H. Ooguri, L. Spodyneiko and C. Vafa, de Sitter Space and the Swampland, arXiv:1806.08362 [INSPIRE].
- [26] S.K. Garg and C. Krishnan, Bounds on Slow Roll and the de Sitter Swampland, JHEP 11 (2019) 075 [arXiv:1807.05193] [INSPIRE].
- [27] H. Ooguri, E. Palti, G. Shiu and C. Vafa, Distance and de Sitter Conjectures on the Swampland, Phys. Lett. B 788 (2019) 180 [arXiv:1810.05506] [INSPIRE].
- [28] M. Montero, A.M. Uranga and I. Valenzuela, Transplanckian axions!?, JHEP 08 (2015) 032 [arXiv:1503.03886] [INSPIRE].
- [29] S.B. Giddings and A. Strominger, Axion Induced Topology Change in Quantum Gravity and String Theory, Nucl. Phys. B 306 (1988) 890 [INSPIRE].
- [30] M. Kruczenski, R.C. Myers and A.W. Peet, Supergravity S-branes, JHEP 05 (2002) 039 [hep-th/0204144] [INSPIRE].
- [31] G. Jones, A. Maloney and A. Strominger, Nonsingular solutions for S-branes, Phys. Rev. D 69 (2004) 126008 [hep-th/0403050] [INSPIRE].
- [32] I.R. Klebanov and A.A. Tseytlin, Gravity duals of supersymmetric $SU(N) \times SU(N + M)$ gauge theories, Nucl. Phys. B 578 (2000) 123 [hep-th/0002159] [INSPIRE].
- [33] I.R. Klebanov and M.J. Strassler, Supergravity and a confining gauge theory: Duality cascades and chi SB resolution of naked singularities, JHEP 08 (2000) 052
 [hep-th/0007191] [INSPIRE].
- [34] I.R. Klebanov and E. Witten, Superconformal field theory on three-branes at a Calabi-Yau singularity, Nucl. Phys. B 536 (1998) 199 [hep-th/9807080] [INSPIRE].
- [35] D.R. Morrison and M.R. Plesser, Nonspherical horizons. 1., Adv. Theor. Math. Phys. 3 (1999) 1 [hep-th/9810201] [INSPIRE].
- [36] A.M. Uranga, Brane configurations for branes at conifolds, JHEP 01 (1999) 022 [hep-th/9811004] [INSPIRE].
- [37] K. Dasgupta and S. Mukhi, Brane constructions, conifolds and M-theory, Nucl. Phys. B 551 (1999) 204 [hep-th/9811139] [INSPIRE].
- [38] G. Buratti, J. Calderón and A.M. Uranga, Transplanckian axion monodromy!?, JHEP 05 (2019) 176 [arXiv:1812.05016] [INSPIRE].
- [39] H. Ooguri and C. Vafa, On the Geometry of the String Landscape and the Swampland, Nucl. Phys. B 766 (2007) 21 [hep-th/0605264] [INSPIRE].

- [40] F. Baume and J. Calderón Infante, *Tackling the SDC in AdS with CFTs*, arXiv:2011.03583 [INSPIRE].
- [41] E. Perlmutter, L. Rastelli, C. Vafa and I. Valenzuela, A CFT Distance Conjecture, arXiv:2011.10040 [INSPIRE].
- [42] E. Silverstein and A. Westphal, Monodromy in the CMB: Gravity Waves and String Inflation, Phys. Rev. D 78 (2008) 106003 [arXiv:0803.3085] [INSPIRE].
- [43] N. Kaloper and L. Sorbo, A Natural Framework for Chaotic Inflation, Phys. Rev. Lett. 102 (2009) 121301 [arXiv:0811.1989] [INSPIRE].
- [44] F. Marchesano, G. Shiu and A.M. Uranga, F-term Axion Monodromy Inflation, JHEP 09 (2014) 184 [arXiv:1404.3040] [INSPIRE].
- [45] A. Hebecker, S.C. Kraus and L.T. Witkowski, D7-Brane Chaotic Inflation, Phys. Lett. B 737 (2014) 16 [arXiv:1404.3711] [INSPIRE].
- [46] H. Ooguri and C. Vafa, Non-supersymmetric AdS and the Swampland, Adv. Theor. Math. Phys. 21 (2017) 1787 [arXiv:1610.01533] [INSPIRE].
- [47] N. Arkani-Hamed, L. Motl, A. Nicolis and C. Vafa, The String landscape, black holes and gravity as the weakest force, JHEP 06 (2007) 060 [hep-th/0601001] [INSPIRE].
- [48] K. Dasgupta, G. Rajesh and S. Sethi, *M theory, orientifolds and G-flux, JHEP* 08 (1999) 023 [hep-th/9908088] [INSPIRE].
- [49] S.B. Giddings, S. Kachru and J. Polchinski, *Hierarchies from fluxes in string compactifications*, *Phys. Rev. D* 66 (2002) 106006 [hep-th/0105097] [INSPIRE].
- [50] A. Hanany and K.D. Kennaway, Dimer models and toric diagrams, hep-th/0503149 [INSPIRE].
- [51] S. Franco, A. Hanany, K.D. Kennaway, D. Vegh and B. Wecht, Brane dimers and quiver gauge theories, JHEP 01 (2006) 096 [hep-th/0504110] [INSPIRE].
- [52] K.D. Kennaway, Brane Tilings, Int. J. Mod. Phys. A 22 (2007) 2977 [arXiv:0706.1660]
 [INSPIRE].
- [53] S. Franco, A. Hanany and A.M. Uranga, Multi-flux warped throats and cascading gauge theories, JHEP 09 (2005) 028 [hep-th/0502113] [INSPIRE].
- [54] D. Berenstein, C.P. Herzog, P. Ouyang and S. Pinansky, Supersymmetry breaking from a Calabi-Yau singularity, JHEP 09 (2005) 084 [hep-th/0505029] [INSPIRE].
- [55] S. Franco, A. Hanany, F. Saad and A.M. Uranga, Fractional branes and dynamical supersymmetry breaking, JHEP 01 (2006) 011 [hep-th/0505040] [INSPIRE].
- [56] M. Bertolini, F. Bigazzi and A.L. Cotrone, Supersymmetry breaking at the end of a cascade of Seiberg dualities, Phys. Rev. D 72 (2005) 061902 [hep-th/0505055] [INSPIRE].
- [57] K.A. Intriligator and N. Seiberg, The Runaway quiver, JHEP 02 (2006) 031 [hep-th/0512347] [INSPIRE].
- [58] G. Buratti, E. García-Valdecasas and A.M. Uranga, Supersymmetry Breaking Warped Throats and the Weak Gravity Conjecture, JHEP 04 (2019) 111 [arXiv:1810.07673] [INSPIRE].
- [59] F. Marchesano and G. Shiu, Building MSSM flux vacua, JHEP 11 (2004) 041 [hep-th/0409132] [INSPIRE].

- [60] F. Marchesano and G. Shiu, MSSM vacua from flux compactifications, Phys. Rev. D 71 (2005) 011701 [hep-th/0408059] [INSPIRE].
- [61] C. Bachas, A Way to break supersymmetry, hep-th/9503030 [INSPIRE].
- [62] C. Angelantonj, I. Antoniadis, E. Dudas and A. Sagnotti, Type I strings on magnetized orbifolds and brane transmutation, Phys. Lett. B 489 (2000) 223 [hep-th/0007090] [INSPIRE].
- [63] R. Blumenhagen, L. Görlich, B. Körs and D. Lüst, Noncommutative compactifications of type-I strings on tori with magnetic background flux, JHEP 10 (2000) 006 [hep-th/0007024] [INSPIRE].
- [64] G. Aldazabal, S. Franco, L.E. Ibáñez, R. Rabadán and A.M. Uranga, D = 4 chiral string compactifications from intersecting branes, J. Math. Phys. 42 (2001) 3103 [hep-th/0011073] [INSPIRE].
- [65] G. Aldazabal, S. Franco, L.E. Ibáñez, R. Rabadán and A.M. Uranga, *Intersecting brane worlds*, *JHEP* 02 (2001) 047 [hep-ph/0011132] [INSPIRE].
- [66] R. Donagi and M. Wijnholt, Model Building with F-theory, Adv. Theor. Math. Phys. 15 (2011) 1237 [arXiv:0802.2969] [INSPIRE].
- [67] C. Beasley, J.J. Heckman and C. Vafa, *GUTs and Exceptional Branes in F-theory I*, *JHEP* **01** (2009) 058 [arXiv:0802.3391] [INSPIRE].
- [68] C. Beasley, J.J. Heckman and C. Vafa, GUTs and Exceptional Branes in F-theory II: Experimental Predictions, JHEP 01 (2009) 059 [arXiv:0806.0102] [INSPIRE].
- [69] R. Blumenhagen, B. Körs and D. Lüst, Type I strings with F flux and B flux, JHEP 02 (2001) 030 [hep-th/0012156] [INSPIRE].
- [70] A.M. Uranga, D-brane probes, RR tadpole cancellation and k-theory charge, Nucl. Phys. B 598 (2001) 225 [hep-th/0011048] [INSPIRE].
- [71] M. Berkooz, M.R. Douglas and R.G. Leigh, Branes intersecting at angles, Nucl. Phys. B 480 (1996) 265 [hep-th/9606139] [INSPIRE].
- S. Kachru and J. McGreevy, Supersymmetric three cycles and supersymmetry breaking, Phys. Rev. D 61 (2000) 026001 [hep-th/9908135] [INSPIRE].
- [73] M. Cvetič, G. Shiu and A.M. Uranga, Chiral four-dimensional N = 1 supersymmetric type 2A orientifolds from intersecting D6 branes, Nucl. Phys. B 615 (2001) 3 [hep-th/0107166]
 [INSPIRE].
- [74] M. Cvetič, G. Shiu and A.M. Uranga, Three family supersymmetric standard-like models from intersecting brane worlds, Phys. Rev. Lett. 87 (2001) 201801 [hep-th/0107143] [INSPIRE].
- [75] D. Cremades, L.E. Ibáñez and F. Marchesano, SUSY quivers, intersecting branes and the modest hierarchy problem, JHEP 07 (2002) 009 [hep-th/0201205] [INSPIRE].
- [76] A. Sen, F theory and orientifolds, Nucl. Phys. B 475 (1996) 562 [hep-th/9605150] [INSPIRE].
- [77] K. Becker and M. Becker, Supersymmetry breaking, M-theory and fluxes, JHEP 07 (2001)
 038 [hep-th/0107044] [INSPIRE].
- [78] G. Pradisi and A. Sagnotti, Open String Orbifolds, Phys. Lett. B 216 (1989) 59 [INSPIRE].
- [79] E.G. Gimon and J. Polchinski, Consistency conditions for orientifolds and d manifolds, Phys. Rev. D 54 (1996) 1667 [hep-th/9601038] [INSPIRE].

- [80] L.J. Romans, Massive N=2a Supergravity in Ten-Dimensions, Phys. Lett. B 169 (1986) 374 [INSPIRE].
- [81] E. Bergshoeff, M.B. Green, G. Papadopoulos and P.K. Townsend, *The IIA supereight-brane*, hep-th/9511079 [INSPIRE].
- [82] N. Seiberg, Five-dimensional SUSY field theories, nontrivial fixed points and string dynamics, Phys. Lett. B 388 (1996) 753 [hep-th/9608111] [INSPIRE].
- [83] S. Kachru and E. Silverstein, On gauge bosons in the matrix model approach to M-theory, Phys. Lett. B 396 (1997) 70 [hep-th/9612162] [INSPIRE].
- [84] P. Hořava and E. Witten, Heterotic and type-I string dynamics from eleven-dimensions, Nucl. Phys. B 460 (1996) 506 [hep-th/9510209] [INSPIRE].
- [85] P. Hořava and E. Witten, Eleven-dimensional supergravity on a manifold with boundary, Nucl. Phys. B 475 (1996) 94 [hep-th/9603142] [INSPIRE].
- [86] R. Antonelli and I. Basile, Brane annihilation in non-supersymmetric strings, JHEP 11 (2019) 021 [arXiv:1908.04352] [INSPIRE].
- [87] I. Basile and S. Lanza, de Sitter in non-supersymmetric string theories: no-go theorems and brane-worlds, JHEP 10 (2020) 108 [arXiv:2007.13757] [INSPIRE].
- [88] S.S. Razamat and D. Tong, Gapped Chiral Fermions, Phys. Rev. X 11 (2021) 011063 [arXiv:2009.05037] [INSPIRE].
- [89] M. Klein and R. Rabadán, $Z_N \times Z_M$ orientifolds with and without discrete torsion, JHEP 10 (2000) 049 [hep-th/0008173] [INSPIRE].
- [90] L.E. Ibáñez, F. Marchesano and R. Rabadán, Getting just the standard model at intersecting branes, JHEP 11 (2001) 002 [hep-th/0105155] [INSPIRE].
- [91] R.C. Myers, Dielectric branes, JHEP 12 (1999) 022 [hep-th/9910053] [INSPIRE].
- [92] J. McGreevy, L. Susskind and N. Toumbas, Invasion of the giant gravitons from Anti-de Sitter space, JHEP 06 (2000) 008 [hep-th/0003075] [INSPIRE].
- [93] V. Balasubramanian, M. Berkooz, A. Naqvi and M.J. Strassler, Giant gravitons in conformal field theory, JHEP 04 (2002) 034 [hep-th/0107119] [INSPIRE].
- [94] S.S. Gubser and I.R. Klebanov, Baryons and domain walls in an N = 1 superconformal gauge theory, Phys. Rev. D 58 (1998) 125025 [hep-th/9808075] [INSPIRE].
- [95] A. Mikhailov, Giant gravitons from holomorphic surfaces, JHEP 11 (2000) 027 [hep-th/0010206] [INSPIRE].
- [96] C.E. Beasley, BPS branes from baryons, JHEP 11 (2002) 015 [hep-th/0207125] [INSPIRE].
- [97] D. Forcella, I. Garcia-Etxebarria and A. Uranga, *E3-brane instantons and baryonic operators* for D3-branes on toric singularities, JHEP **03** (2009) 041 [arXiv:0806.2291] [INSPIRE].
- [98] A. Hamilton, J. Murugan and A. Prinsloo, Lessons from giant gravitons on $AdS_5 \times T^{1,1}$, JHEP 06 (2010) 017 [arXiv:1001.2306] [INSPIRE].
- [99] T. Banks and N. Seiberg, Symmetries and Strings in Field Theory and Gravity, Phys. Rev. D 83 (2011) 084019 [arXiv:1011.5120] [INSPIRE].

Published for SISSA by 🖄 Springer

RECEIVED: July 30, 2021 ACCEPTED: September 4, 2021 PUBLISHED: October 5, 2021

Dynamical Cobordism and Swampland Distance Conjectures

Ginevra Buratti, José Calderón-Infante, Matilda Delgado and Angel M. Uranga

Instituto de Física Teórica IFT-UAM/CSIC, C/ Nicolás Cabrera 13-15, Campus de Cantoblanco, 28049 Madrid, Spain E-mail: ginevra.buratti@uam.es, j.calderon.infante@csic.es, matilda.delgado@uam.es, angel.uranga@csic.es

ABSTRACT: We consider spacetime-dependent solutions to string theory models with tadpoles for dynamical fields, arising from non-trivial scalar potentials. The solutions have necessarily finite extent in spacetime, and are capped off by boundaries at a finite distance, in a dynamical realization of the Cobordism Conjecture. We show that as the configuration approaches these cobordism walls of nothing, the scalar fields run off to infinite distance in moduli space, allowing to explore the implications of the Swampland Distance Conjecture. We uncover new interesting scaling relations linking the moduli space distance and the SDC tower scale to spacetime geometric quantities, such as the distance to the wall and the scalar curvature. We show that walls at which scalars remain at finite distance in moduli space correspond to domain walls separating different (but cobordant) theories/vacua; this still applies even if the scalars reach finite distance singularities in moduli space, such as conifold points.

We illustrate our ideas with explicit examples in massive IIA theory, M-theory on CY threefolds, and 10d non-supersymmetric strings. In 4d $\mathcal{N} = 1$ theories, our framework reproduces a recent proposal to explore the SDC using 4d string-like solutions.

KEYWORDS: Flux compactifications, Superstring Vacua, Supersymmetry Breaking

ARXIV EPRINT: 2107.09098





Contents

1	I Introduction and conclusions		
2	Cobordism walls in massive IIA theory	3	
3	Cobordism walls in M-theory on CY3	5	
	3.1 M-theory on CY3 with G_4 flux	5	
	3.2 Traveling across finite distance singularities in moduli space	7	
4	${ m S}^1$ compactification of 4d ${\cal N}=1$ theories and EFT strings	8	
	4.1 Cobordism walls in 4d $\mathcal{N} = 1$ theories on a circle	9	
	4.2 Comparison with EFT strings	12	
5	4d $\mathcal{N} = 1$ theories with flux-induced superpotentials	13	
6	Walls in 10d non-supersymmetric strings	15	
7	Final remarks	16	
\mathbf{A}	Holographic examples	18	
	A.1 Domain wall to a new vacuum	18	
	A.2 Domain wall to a new running solution	19	
	A.3 Cobordism domain walls to disconnected solutions	22	

1 Introduction and conclusions

A remarkable proposal in the Swampland Program of quantum gravity constraints on effective field theories [1] (see [2–4] for reviews) is the Cobordism Conjecture [5], that is based on the expected absence of exact global symmetries in quantum gravity. In short, it states that any configuration in a consistent theory of quantum gravity should not carry any cobordism charge. In practice, it implies that any configuration in a consistent theory of quantum gravity should not carry of quantum gravity should admit, at the topological level, the introduction of a boundary ending spacetime into nothing,¹ in the sense of [6] (see [7, 8] for recent related discussions). Accordingly, we will refer to such boundaries as *walls of nothing*. Equivalently, it implies that any two consistent theories of quantum gravity must admit, at the topological level, an interpolating configuration connecting them, as a generalized domain wall separating the two theories. We will refer to such configurations as *interpolating domain walls*.

¹This boundary may be dressed by additional defects, such as D-branes or O-planes in string setups, to absorb the relevant charges.

The Cobordism Conjecture is topological in nature. However, it can lead to remarkable breakthroughs when supplemented by additional assumptions. For instance, the extra ingredient of supersymmetry of the theory (and possibly of its walls) has led to highly non-trivial constraints in lower dimensional theories, see e.g. [9, 10].

An important step forward in endowing cobordism walls with dynamics was taken in [11], in the study of theories with tadpoles for dynamical fields (dubbed dynamical tadpoles, as opposed to topological tadpoles, such as RR tadpoles, which lead to topological consistency conditions on the configuration²). These are ubiquitious in the presence of scalar potentials, and in particular in non-supersymmetric string models. In theories with dynamical tadpoles the solutions to the equations of motion vary over the non-compact spacetime dimensions. Based on the behaviour of large classes of string models, it was proposed in [11] that such spacetime-dependent running solutions must hit cobordism walls of nothing at a finite distance Δ in spacetime³ (as measured in the corresponding Einstein frame metric), scaling as $\Delta^{-n} \sim \mathcal{T}$ with the strength of the tadpole \mathcal{T} . These examples included holographic AdS₅ × $T^{1,1}$ compactifications with RR 3-form flux, type IIB 3-form flux compactifications, magnetized D-brane models, massive IIA theory, M-theory on K3 with G_4 flux, and the 10d non-supersymmetric USp(32) string theory. On the other hand, interpolating cobordism walls connecting different theories were not discussed. One of the motivations of this work is to fill this gap.

We argue that, when a running solution in theories with dynamical tadpoles hits a wall, the behaviour of the configuration across the wall, and in particular the sharp distinction between interpolating domain walls and walls of nothing, is determined by the behaviour of scalar fields as one reaches the wall, via a remarkable correspondence:

- When scalars remain at finite distance points in moduli space as one hits the wall, it corresponds to an interpolating domain wall, and the solution continues across it in spacetime (with jumps in quantities as determined by the wall properties);
- On the other hand, when the scalars run off to infinity in moduli space as one reaches the wall (recall, at a finite distance in spacetime), it corresponds to a wall of nothing, capping off spacetime beyond it.

We also argue that scalars reaching singular points at finite distance in moduli space upon hitting the wall still define interpolating domain walls, rather than walls of nothing; hence, walls of nothing are not a consequence of general singularities in moduli space, but actually to those at infinity in moduli space. This suggests that, in the context of dynamical solutions,⁴ the walls of nothing of the Cobordism Conjecture are closely related to the Swampland Distance Conjecture.⁵ We indeed find universal scaling relations between the

²Note however that dynamical tadpoles were recently argued in [12] to relate to violation of swampland constraints of quantum gravity theories.

³For related work on dynamical tadpoles in non-supersymmetric theories, see [13–20].

⁴Note that, in setups with no dynamical tadpole, one can still have e.g. cobordism walls of nothing without scalars running off to infinity: for instance, 11d M-theory, which does not even have scalars, admits walls of nothing defined by Horava-Witten boundaries; similar considerations may apply to potential theories with no moduli (or with all moduli stabilized at high enough scale).

⁵The status of the SDC in spacetime dependent running solutions was addressed in [21].

(finite) distance to the wall in spacetime and the scale of the SDC tower [22]. In addition, we uncover a universal scaling relation between the curvature scalar in running solutions and the SDC tower scale that is reminiscent of the Anti de Sitter Distance Conjecture (ADC) [23].

We illustrate these ideas in several large classes of string theory models, including massive IIA, and M-theory on CY threefolds. Moreover, we also argue that our framework encompasses the recent discussion of EFT string solutions in 4d $\mathcal{N} = 1$ theories in [24] (see also [25]), where saxion moduli were shown to attain infinity in moduli space at the core of strings magnetically charged under the corresponding axion moduli. We show that EFT string solutions are the cobordism walls of nothing of \mathbf{S}^1 compactifications of the 4d $\mathcal{N} = 1$ theory with certain axion fluxes on the \mathbf{S}^1 . Our scalings also relate to those between EFT string tensions and the SDC tower scale in [24].

The paper is organized as follows. In section 2 we present the main ideas in the explicit setup of running solutions in massive IIA theory, and their interplay with type I' solutions [26]. In section 3 we carry out a similar discussion for M-theory on CY threefolds with G_4 flux (in section 3.1) and their relation to strongly coupled heterotic strings [27]. In section 3.2 we use it to discuss domain walls across singularities at finite distance in moduli space, following [28]. In section 4 we discuss the \mathbf{S}^1 compactification of general 4d $\mathcal{N} = 1$ theories. In section 4.1 we introduce dynamical tadpoles from axion fluxes, whose running solutions hit walls of nothing at which saxions run off to infinity. In section 4.2 we relate the discussion to the EFT strings of [24]. In section 5 we discuss the moduli space distances in walls of nothing and interpolating walls in 4d $\mathcal{N} = 1$ theories with non-trivial superpotentials of the kind arising in flux compactifications. In section 6 we discuss our proposal in non-supersymmetric string theories, in particular the 10d USp(32) string. In section 7 we offer some final remarks and outlook. Appendix A provides some observations on cobordism walls in holographic throats.

2 Cobordism walls in massive IIA theory

Walls of nothing and infinite moduli space distance. In this section we consider different kinds of cobordism walls in massive IIA theory [29], extending the analysis in [11]. The Einstein frame 10d effective action for the relevant fields is

$$S_{10,E} = \frac{1}{2\kappa^2} \int d^{10}x \sqrt{-G_E} \left\{ \left[R - \frac{1}{2} (\partial \phi)^2 \right] - \frac{1}{2} e^{\frac{5}{2}\phi} F_0^2 - \frac{1}{2} e^{\frac{1}{2}\phi} (F_4)^2 \right\}, \qquad (2.1)$$

where the Romans mass parameter is denoted by F_0 to suggest it is a 0-form field strength flux. This theory is supersymmetric, but has a dilaton tadpole

$$\mathcal{T} \sim e^{\frac{5}{2}\phi} F_0^2 \,, \tag{2.2}$$

so the theory does not admit 10d maximally symmetric solutions. The solutions with maximal (super)symmetry are 1/2 BPS configurations with the dilaton depending on one coordinate x^9 , closely related to that in [30]. In conventions closer to [26], the Einstein frame metric and dilaton are

$$(G_E)_{MN} = Z(x^9)^{\frac{1}{12}} \eta_{MN}, \quad e^{\phi} = Z(x^9)^{-\frac{5}{6}}, \quad \text{with } Z(x^9) \sim -F_0 x^9, \qquad (2.3)$$

where we have set some integration constant to zero. The solution hits a singularity at $x^9 = 0$. The spacetime distance from a general position x^9 to the singularity is [11]

$$\Delta = \int_{x^9}^0 Z(x^9)^{\frac{1}{24}} dx^9 \sim Z(x^9)^{\frac{25}{24}} F_0^{-1} \sim F_0^{-1} e^{-\frac{5}{4}\phi} \sim \mathcal{T}^{-\frac{1}{2}}, \qquad (2.4)$$

in agreement with the scaling relation $\Delta^{-2} \sim \mathcal{T}$, that was dubbed Finite Distance lesson in [11]. Following the Dynamical Cobordism proposal therein, the singularity is resolved in string theory into a cobordism wall of nothing, defined by an O8-plane (possibly dressed with D8-branes to match the F_0 flux to be absorbed),⁶ ending the direction x^9 as a boundary.

We now notice that, since $Z \to 0$ implies $\phi \to \infty$ as $x^9 \to 0$, the dilaton runs off to infinity in moduli space as one hits the wall, as befits a wall of nothing from our discussion in the introduction. According to the SDC, there is an infinite tower of states becoming massless in this region, with a scale decaying exponentially with the moduli space distance D as

$$M_{\rm SDC} \sim e^{-\lambda D}$$
, (2.5)

with some positive $\mathcal{O}(1)$ coefficient λ .

It is interesting to find a direct relation between these quantities and the spacetime distance to the wall. The distance in moduli space is given by $\phi = \sqrt{2} D$, as can be seen from the kinetic term for ϕ in (2.1). From (2.4) we have

$$\Delta \sim e^{-\frac{5}{2\sqrt{2}}D}, \quad M_{\rm SDC} \sim \Delta^{\frac{2\sqrt{2}}{5}\lambda}.$$
(2.6)

Hence the SDC tower scale goes to zero with the distance to the wall with a power-like scaling.

It is a natural question to ask if this tower of states becomes light in the actual dynamical configuration (rather than in the adiabatic framework of the standard formulation of the SDC). In this particular setup, the SDC tower corresponds to D0-branes which end up triggering the decompactification of the M-theory eleventh dimension. In the dynamical solution, there are a finite number of extra massless states, responsible for the enhancement of the perturbative open string gauge group to the exceptional symmetries which are known to arise from the heterotic dual theory [26] (see also [31]). On the other hand, there is no signal of an infinite tower of states becoming massless simultaneously. The appearance of the SDC in the dynamical context has thus different implications as compared with the usual adiabatic formulation.

Let us now turn to another novel, and tantalizing, scaling. The scalar curvature for the running solution reads

$$|R| \sim (-x^9)^{-\frac{25}{12}} \sim e^{\frac{5}{\sqrt{2}}D}$$
 (2.7)

Using this, we can write the SDC tower scale in terms of the scalar curvature as

$$M_{\rm SDC} \sim e^{-\lambda D} \sim |R|^{-\frac{\sqrt{2}}{5}\lambda}$$
 (2.8)

⁶This imposes a swampland bound on the possible values of F_0 that are consistent in string theory.

This scaling is highly reminiscent of the Anti de Sitter Distance Conjecture (ADC) of [23],⁷ even though the setup under consideration is very different.⁸ Note however that, as in the ADC, it signals a failure of the decoupling of scales, and hence a breakdown of the effective field theory near the wall of nothing. This fits nicely with our observation that the wall can only be microscopically defined in the UV complete theory, and works as a boundary condition defect at the level of the effective theory.

Interpolating domain walls. There is a well known generalization of the above solutions, which involves the inclusion of D8-branes acting as interpolating domain walls across which F_0 jumps by one unit. The general solution of this kind is provided by (2.3) with a piecewise constant F_0 and a piecewise continuous function Z [26].

The D8-brane domain walls are thus (a very simple realization of) cobordism domain walls interpolating between different Romans IIA theories (differing just in their mass parameter). The point we would like to emphasize is that, since Z remains finite across them, the dilaton remains at finite distance in moduli space, as befits interpolating domain walls from our discussion in the introduction.

3 Cobordism walls in M-theory on CY3

In this section we recall results from the literature on the strong coupling limit of the heterotic string, also known as heterotic M-theory [27, 32–34] (see [35, 36] for review and additional references). They provide straightforward realizations of the different kinds of cobordism walls in M-theory compactifications on CY threefolds. The discussion generalizes that in [11], and allows to study the behaviour at singular points at finite distance in moduli space, in particular flops at conifold points.

3.1 M-theory on CY3 with G_4 flux

We consider M-theory on a CY threefold \mathbf{X} , with G_4 field strength fluxes on 4-cycles. For later convenience, we follow the presentation in [28]. We introduce dual basis of 2- and 4-cycles $C^i \in H_2(\mathbf{X})$ and $D_i \in H_4(\mathbf{X})$, and define

$$\int_{D_i} G_4 = a_i \,, \quad \int_{C^i} C_6 = \tilde{\lambda}^i \,. \tag{3.1}$$

We also denote by b_i the 5d vector multiplet of real Kähler moduli, with the usual Kähler metric and the 5d $\mathcal{N} = 1$ prepotential

$$G_{ij} = -\frac{1}{2} \frac{\partial^2}{\partial b_i \partial b_j} \ln \mathcal{K}, \qquad \mathcal{K} \equiv \frac{1}{3!} d_{ijk} b^i b^j b^k, \qquad (3.2)$$

with d_{ijk} being the triple intersection numbers of **X**. We have the familiar constraint $\mathcal{K} = 1$ removing the overall modulus V, which lies in a hypermultiplet.

⁷It is possible that the result is ultimately linked to the generalized distance conjectures in [23]; we leave this as an open question for future work.

⁸In contrast to the ADC, that considers the limit of vanishing curvature of a family of AdS vacua, in our setup the scalar curvature blows up as the singularity is approached. However, we do find a power-like scaling similar to the ADC one.

The 5d effective action for these fields is

$$S_{5} = -\frac{M_{p,11}^{9}}{2} L^{6} \bigg[\int_{M_{5}} \sqrt{-g_{5}} \left(R + G_{ij}(b) \partial_{M} b^{i} \partial^{M} b^{j} + \frac{1}{2V^{2}} \partial_{M} V \partial^{M} V + \lambda(\mathcal{K} - 1) \right) \\ + \frac{1}{4V^{2}} G^{ij}(b) a_{i} \wedge \star a_{j} + d\tilde{\lambda}^{i} \wedge a_{i} \bigg] - \sum_{n=0}^{N+1} \alpha_{i}^{(n)} \int_{M_{4}^{(n)}} \left(\tilde{\lambda}^{i} + \frac{b^{i}}{V} \sqrt{g_{4}} \right).$$
(3.3)

Here λ is a Lagrange multiplier, and L the reference length scale of the Calabi-Yau. With hindsight, we include 4d localized terms which correspond to different walls in the theory, with induced 4d metric g_4 .

The G_4 fluxes a_i induce dynamical tadpoles for the overall volume and the Kähler moduli b_i . There are 1/2 supersymmetric solutions running in one spacetime coordinate, denoted by y, with the structure

$$ds_{5}^{2} = e^{2A} ds_{4}^{2} + e^{8A} dy^{2},$$

$$V = e^{6A}, \quad b^{i} = e^{-A} f^{i},$$

$$e^{3A} = \left(\frac{1}{3!} d_{ijk} f^{i} f^{j} f^{k}\right),$$

$$(d\tilde{\lambda}^{i})_{\mu\nu\rho\sigma} = \epsilon_{\mu\nu\rho\sigma} e^{-10A} \left(-\partial_{11} b^{i} + 2b^{i} \partial_{11} A\right).$$

(3.4)

The whole solution is determined by a set of one-dimensional harmonic functions. They are given in terms of the local values of the G_4 fluxes,

$$d_{ijk}f^{j}f^{k} = H_{i}, \qquad H_{i} = a_{i}y + c_{i}.$$
 (3.5)

Here the c_i are integration constants set to have continuity of the H_i , and hence of the f_i , across the different interpolating domain walls in the system, which produce jumps as follows. Microscopically, the interpolating domain walls correspond to M5-branes wrapped on 2-cycles $[C] = \sum n_i C^i$, leading to jumps in the fluxes that in units of M5-brane charge are given by

$$\Delta a_i = n_i \,. \tag{3.6}$$

Hence, interpolating domain walls maintain the theory at finite distance in moduli space. This is not the case for cobordism walls of nothing, which arise when $e^A \to 0$, and hence $V \to 0$, which sits at infinity in moduli space. This regime was already discussed (in the simpler setup of K3 compactifications) in [11], where the cobordism domain was argued to be given by a Horava-Witten boundary (dressed with suitable gauge bundle degrees of freedom, as required to absorb the local remaining G_4 flux), in agreement with the strong coupling singularity discussed in [27]. The wall appears at a finite spacetime distance Δ following the scaling $\Delta^{-2} \sim \mathcal{T}$ in [11]. In what follows, we describe the scaling relations of the moduli space distance and the SDC tower at these walls of nothing.

Since they are characterized by the vanishing of the overall volume of \mathbf{X} , it is enough to follow the behaviour of V and the discussion simplifies. Restriction to this sector amounts to setting all $f_i \equiv f$ in (3.4), and all $H_i \equiv H$. Also, since the wall of nothing arises when $H \to 0$, we can take this location as y = 0 and write

$$e^{2A} \sim H(y) \sim \alpha y$$
. (3.7)

Using the metric in (3.4), the spacetime distance from a point y > 0 is

$$\Delta = \int_0^y (\alpha y)^2 dy = \frac{1}{3} \alpha^2 y^3.$$
(3.8)

We are also interested in the traversed distance in moduli space D. Using the kinetic term in (3.3), the relevant integral to compute is

$$D = -\int \frac{1}{\sqrt{2}V} \frac{dV}{dy} dy.$$
(3.9)

Using $V \sim H^3$, we get as leading behavior near the singularity

$$D \simeq -\frac{3}{\sqrt{2}}\log y = -\frac{1}{\sqrt{2}}\log\frac{3\Delta}{\alpha^2},\qquad(3.10)$$

where in the last equality we used (3.8). This implies

$$\Delta \sim e^{-\sqrt{2}D},\tag{3.11}$$

and leads to a power-like scaling of the SDC tower mass

$$M_{\rm SDC} \sim \Delta^{\frac{\gamma}{\sqrt{2}}}$$
 (3.12)

Computing the curvature scalar from (3.4), we get

$$|R| \sim e^{2\sqrt{2D}}$$
. (3.13)

So the SDC tower scale can be expressed, in an ADC-like manner, as

$$M_{\rm SDC} \sim |R|^{-\frac{\lambda}{2\sqrt{2}}}.$$
(3.14)

We thus recover a similar behaviour to the examples in section 2.

3.2 Traveling across finite distance singularities in moduli space

The setup of M-theory on a CY3 \mathbf{X} allows to address the question of whether walls of nothing could arise at finite distance in moduli space, if the scalars hit a singular point in moduli space. This is actually not the case, as can be explicitly shown by following the analysis in [28] for flop transitions.

Specifically, they considered the flop transition between two Calabi-Yau manifolds with $(h_{1,1}, h_{2,1}) = (3, 243)$, in the setup of a CY3 compactification of the Horava-Witten theory, namely with two boundaries restricting the coordinate y to an interval. In our more general setup, one may just focus on the dynamics in the bulk near the flop transition as one moves along y. Hence we are free to locate the flop transition point at y = 0.

In terms of the Kähler moduli $t^i = V^{\frac{1}{3}}b_i$ of **X**, and changing to a more convenient basis

$$t^{1} = U, \quad t^{2} = T - \frac{1}{2}U - W, \quad t^{3} = W - U,$$
 (3.15)

and similar (proper transforms under the flop) for $\mathbf{\tilde{X}}$, the Kähler cones of \mathbf{X} and $\mathbf{\tilde{X}}$ are defined by the regions

$$\mathbf{X}: \quad W > U > 0, \quad T > \frac{1}{2}U + W, \quad (3.16)$$

$$\tilde{\mathbf{X}}: \quad U > W > 0, \quad T > \frac{3}{2}U.$$
 (3.17)

This shows that the flop curve is C_3 , and the area is W - U, changing sign across the flop.

Near the flop point y = 0, the harmonic functions for the two CYs X and X have the form

$$\begin{aligned}
\mathbf{X} & \text{at } y \leq 0 & \mathbf{X} & \text{at } y \geq 0 \\
H_T &= -18y + k_T, & \tilde{H}_T &= 18y + k_T, \\
H_U &= -25y + k_0, & \tilde{H}_U &= 24y + k_0, \\
H_W &= 6y + k_0, & \tilde{H}_W &= -5y + k_0.
\end{aligned}$$
(3.18)

Hence

X at
$$y \le 0$$
 X at $y \ge 0$
 $H_{W-U} = 31y$, $\tilde{H}_{W-U} = -29y$. (3.19)

Even though the flop point is a singularity in moduli space, and despite the sign flip for W-U, the harmonic functions are continuous and the solution remains at finite distance in moduli space. This agrees with the picture that it corresponds to an interpolating domain wall. In fact, as discussed in [28], the discontinuity in their slopes (and the related change in the G_4 fluxes) makes the flop point highly analogous to the above described interpolating domain walls associated to M5-branes.

The above example illustrates a further important aspect. It provides an explicit domain wall interpolating between two different (yet cobordant) topologies. It would be extremely interesting to extend this kind of analysis to other topology changing transitions, such as conifold transitions⁹ [38]. This would allow for a further leap for the dynamical cobordism proposal, given that moduli spaces of all CY threefolds are expected to be connected by this kind of transitions [39].

We have thus established that physics at finite distance in moduli space gives rise to interpolating domain walls, rather than walls of nothing, even at singular points in moduli space. The implication is that the physics of walls of nothing is closely related to the behaviour near infinity in moduli space and hence to the SDC. In the following section we explore further instances of this correspondence in general 4d $\mathcal{N} = 1$ theories.

4 S¹ compactification of 4d $\mathcal{N} = 1$ theories and EFT strings

In this section we study a systematic way to explore infinity in moduli space in general 4d $\mathcal{N} = 1$ theories. This arises in a multitude of string theory constructions, ranging from

 $^{^{9}}$ For a proposal to realize conifold transitions dynamically in a time-dependent background, see [37].

heterotic CY compactifications to type II orientifolds on CY spaces [40]. Our key tool is an S^1 compactification to 3d with certain axion fluxes. We will show that the procedure secretly matches the construction of EFT strings in [24] (see also [25]). Actually, this correspondence was the original motivation for this paper.

4.1 Cobordism walls in 4d $\mathcal{N} = 1$ theories on a circle

We want to consider general 4d $\mathcal{N} = 1$ theories near infinity in moduli space. According to [41–43], the moduli space in this asymptotic regime is well approximated by a set of axion-saxion complex fields, with metric given by hyperbolic planes. We start discussing the single-field case, and sketch its multi-field generalization at the end of this section.

Consider a 4d $\mathcal{N} = 1$ theory with complex modulus S = s + ia, where a is an axion of unit periodicity and s its saxionic partner. We take a Kähler potential

$$K = -\frac{2}{n^2} \log(S + \bar{S}) \,. \tag{4.1}$$

The 4d effective action is

$$S = \frac{M_{P,4}^2}{2} \int d^4x \sqrt{-g_4} \left\{ R_4 - \frac{n^{-2}}{s^2} \left[(\partial s)^2 + (\partial a)^2 \right] \right\},$$

$$= \frac{M_{P,4}^2}{2} \int d^4x \sqrt{-g_4} \left\{ R_4 - (\partial \phi)^2 - e^{-2n\phi} (\partial a)^2 \right\},$$
 (4.2)

where in the last equation we have defined $\phi = \frac{1}{n} \log ns$.

We now perform an S^1 compactification to 3d with the following ansatz for the metric¹⁰ and the scalars

$$ds_4^2 = e^{-\sqrt{2}\sigma} ds_3^2 + e^{\sqrt{2}\sigma} R_0^2 d\theta^2 ,$$

$$\phi = \phi(x^{\mu}) , \qquad a = \frac{\theta}{2\pi} q + a(x^{\mu}) , \qquad (4.3)$$

where x^{μ} denote the 3d coordinates and $\theta \sim \theta + 2\pi$ is a periodic coordinate. Regarding the axion as a 0-form gauge field, the ansatz for *a* introduces *q* units of its field strength flux (we dub it axion flux) on the S^1 . We allowed for a general saxion profile to account for its backreaction, as we see next.

The dimensional reduction of the action (4.2) gives (see e.g. [44])

$$S_3 = \frac{M_{P,3}}{2} \int d^3x \sqrt{-g_3} \left\{ R_3 - G_{ab} \partial_\mu \varphi^a \partial^\mu \varphi^b - V(\varphi) \right\} , \qquad (4.4)$$

where

$$G_{ab}\partial_{\mu}\varphi^{a}\partial^{\mu}\varphi^{b} = (\partial\sigma)^{2} + (\partial\phi)^{2} + e^{-2n\phi}(\partial a)^{2}, \qquad (4.5a)$$

$$V(\varphi) = e^{-2\sqrt{2}\sigma - 2n\phi} \left(\frac{q}{2\pi R_0}\right)^2, \qquad (4.5b)$$

and $M_{P,3} = 2\pi R_0 M_{P,4}^2$ is the 3d Planck mass.

¹⁰We omit the KK U(1) because it will not be active in our discussion.

The last term in the 3d action corresponds to a dynamical tadpole for a linear combination of the saxion and the radion, induced by the axion flux. We thus look for running solutions of the 3d equations of motion. We focus on solutions with constant axion in 3d $a(x^{\mu}) = 0$, for which the equations of motion read

$$\frac{1}{\sqrt{-g_3}}\partial_\nu\left(\sqrt{-g_3}g^{\mu\nu}\partial_\mu\sigma\right) = -\sqrt{2}\,e^{-2\sqrt{2}\sigma-2n\phi}\left(\frac{q}{2\pi R_0}\right)^2\,,\tag{4.6a}$$

$$\frac{1}{\sqrt{-g_3}}\partial_{\nu}\left(\sqrt{-g_3}g^{\mu\nu}\partial_{\mu}\phi\right) = -n\,e^{-2\sqrt{2}\sigma-2n\phi}\left(\frac{q}{2\pi R_0}\right)^2\,.\tag{4.6b}$$

We consider solutions in which the fields run with one of the coordinates x^3 (which with hindsight we denote by $r \equiv x^3$). We focus on a particular 3d axion-saxion ansatz

$$s(r) = s_0 - \frac{q}{2\pi} \log \frac{r}{r_0}, \quad a(r) = a_0,$$
(4.7)

for which the radion can be solved as

$$\sqrt{2}\sigma = \frac{2}{n}(\phi - \phi_0) + 2\log\frac{r}{R_0} = \frac{2}{n^2}\log\left(1 - \frac{q}{2\pi s_0}\log\frac{r}{r_0}\right) + 2\log\frac{r}{R_0}.$$
 (4.8)

This, together with (4.7), provides the scalar profiles solving the dynamical tadpole. The motivation for this particular solution is that it preserves 1/2 supersymmetry, as we discuss in the next section in the context of its relation with the 4d string solutions in [24].

Note that as $r \to 0$, the radion blows up as $\sigma \to -\infty$, implying that the \mathbf{S}^1 shrinks to zero size, and the metric becomes singular. As one hits this singularity, the saxion goes to infinity, so we face a wall at which the scalars run off to infinity in moduli space. According to our arguments, it must correspond to a cobordism wall of nothing, capping off spacetime so that the r < 0 region is absent; hence the suggestive notation to regard this coordinate as a radial one, an interpretation which will become more clear in the following section. The finite distance Δ to the wall can be shown to obey the scaling $\Delta^{-2} \sim \mathcal{T}$ introduced in [11].

Note that the asymptotic regime near infinity in moduli space $s \gg 1$ corresponds to the regime

$$r \ll r_0 e^{\frac{2\pi}{q}(s_0 - 1)}.$$
(4.9)

Hence the exploration of the SDC's implications requires zooming into the region close to the wall of nothing.

Let us emphasize that the microscopic structure of the wall of nothing cannot be determined purely in terms of the effective field theory, and should be regarded as provided by its UV completion.¹¹ On the other hand, we can use effective field theory to obtain the scaling relations between different quantities, as in the string theory examples in the previous sections.

 $^{^{11}\}mathrm{In}$ particular, possible constraints on q could arise from global consistency of the backreaction.

The scaling relations. We can now study the scaling relations between spacetime and moduli space distances, and the SDC tower scale. From the spacetime profiles for σ and ϕ , it is easy to check that the contribution from the radion dominates in the $r \to 0$ limit. The resulting scaling between the moduli space distance D and r is

$$r \simeq e^{-D} \,, \tag{4.10}$$

showing again that $D \to \infty$ as $r \to 0$. On the other hand, the spacetime distance Δ in the same limit gives

$$d\Delta \simeq \frac{r}{R_0} \left(-\frac{q}{2\pi s_0} \log\left(\frac{r}{r_0}\right) \right)^{\frac{2}{n^2}} \simeq \frac{1}{R_0} \left(-\frac{q}{2\pi s_0} \right)^{\frac{2}{n^2}} D^{2/n^2} e^{-2D} dD \,. \tag{4.11}$$

Upon integration one gets an incomplete gamma function that, after keeping the leading order in $D \to \infty$, finally gives

$$\Delta \sim e^{-2D + \frac{2}{n^2} \log D} \,. \tag{4.12}$$

This is an exponential behaviour up to logarithmic corrections. It would be interesting to relate this to existing results on log corrections to Swampland conjectures (see [45]), but we skip them for now. The resulting relation allows to write the scalings of the SDC tower scale as

$$M_{\rm SDC} \sim e^{-\lambda D} \sim \Delta^{\frac{\lambda}{2}},$$
 (4.13)

that is again a power-like relation with $\mathcal{O}(1)$ exponents.

Let us turn to computing the scaling of the SDC scale with the scalar curvature R. The general expression for R is rather complicated, but simplifies in the leading order approximation at r = 0

$$\log|R| \simeq -4\log r \simeq 4D. \tag{4.14}$$

Hence, the SDC tower mass scales as

$$M_{\rm SDC} \sim e^{-\lambda D} \sim |R|^{-\frac{1}{4}\lambda} \,. \tag{4.15}$$

Amusingly, we again recover a power-like scaling highly reminiscent of the ADC.

Multi-field generalization. Let us end this section by mentioning that the above simple model admits a straightforward generalization to several axion-saxion moduli a^i , s^i . One simply introduces a vector of axion fluxes q^i and generalizes the above running solution to

$$a^{i} = a_{0}^{i} + \frac{\theta}{2\pi}q^{i}, \quad s^{i}(r) = s_{0}^{i} - \frac{q^{i}}{2\pi}\log\frac{r}{r_{0}}.$$
 (4.16)

The corresponding backreaction on σ is

$$\sqrt{2}\,\sigma = -K(r) + K_0 + 2\log\frac{r}{R_0}\,. \tag{4.17}$$

We leave this as an exercise for the reader, since the eventual result is more easily recovered by relating our system to the 4d string-like solutions in [24], to which we now turn.
4.2 Comparison with EFT strings

The ansatz (4.7) is motivated by the relation of our setup with the string-like solutions to $4d \mathcal{N} = 1$ theories discussed in [24], which we discuss next. This dictionary implies that those results can be regarded as encompassed by our general understanding of cobordism walls of nothing and the SDC.

In a 4d perspective, (4.7) corresponds to a holomorphic profile $z = re^{i\theta}$

$$S = S_0 + \frac{q}{2\pi} \log \frac{z}{z_0} \,. \tag{4.18}$$

The axion flux in (4.3) implies that there is a monodromy $a \to a+q$ around the origin z = 0. Hence, the configuration describes a BPS string with q units of axion charge. The solution for the metric can easily be matched with that in [24]. The 4d metric takes the form

$$ds_4^2 = -dt^2 + dx^2 + e^{2Z} dz d\bar{z}, \qquad (4.19)$$

with the warp factor

$$2Z = -K + K_0 = \frac{2}{n^2} \log \frac{s}{s_0} \,. \tag{4.20}$$

This matches the 3d metric (4.19) by writing

$$ds_3^2 = e^{\sqrt{2}\sigma} \left(-dt^2 + dx^2 \right) + e^{2Z + \sqrt{2}\sigma} dr^2 , \qquad (4.21)$$

and (4.8) ensures the matching of the S^1 radion with the 4d angular coordinate range.

$$\int_{0}^{2\pi} d\theta e^{\sigma/\sqrt{2}} R_0 = \int_{0}^{2\pi} d\theta e^Z r \,. \tag{4.22}$$

Hence, in 4d $\mathcal{N} = 1$ theories there is a clear dictionary between running solutions in \mathbf{S}^1 compactifications with axion fluxes and EFT string solutions. The compactification circle maps to the angle around the string; the axion fluxes map to string charges; the coordinate in which fields run (semi-infinite, due to the wall of nothing) maps to the radial coordinate away from the string; the saxion running due to the axion flux induced dynamical tadpole maps to the string backreaction on the saxion, i.e. the string RG flow; the scalars running off to infinity in moduli space as one hits the wall of nothing map to the scalars running off to infinity in moduli space as one reaches the string core. Note that the fact that the wall of nothing is not describable within the effective theory maps to the criterion for an EFT string, i.e. it is regarded as a UV-given defect providing boundary conditions for the effective field theory fields.

This dictionary allows to extend the interesting conclusions in [24] to our context. For instance, the relation between the string tension and its backreaction on the geometry provides a scaling with the spacetime distance. This is the counterpart of the scaling relations we found in our 3d dynamical cobordism discussion in the previous section.

On another line, the Distant Axionic String Conjecture in [24] proposes that every infinite field distance limit of a 4d $\mathcal{N} = 1$ effective theory consistent with quantum gravity can be realized as an RG flow UV endpoint of an EFT string. We can thus map it into the

proposal that every infinite field distance limit of a 4d $\mathcal{N} = 1$ effective theory consistent with quantum gravity can be realised as the running into a cobordism wall of nothing in some axion fluxed \mathbf{S}^1 compactification to 3d. It is thus natural to extend this idea to a general conjecture

Cobordism Distance Conjecture. Every infinite field distance limit of a effective theory consistent with quantum gravity can be realized as the running into a cobordism wall of nothing in (possibly a suitable compactification of) the theory.

The examples in the previous sections provide additional evidence for this general form of the conjecture, beyond the above 4d $\mathcal{N} = 1$ context.

5 4d $\mathcal{N} = 1$ theories with flux-induced superpotentials

In the previous section we discussed cobordism walls in compactifications of 4d $\mathcal{N} = 1$ theories on \mathbf{S}^1 with axion fluxes. Actually, it is also possible to study running solutions and walls in these theories without any compactification. This requires additional ingredients to introduce the dynamical tadpoles triggering the running. Happily, there is a ubiquitous mechanism, via the introduction of non-trivial superpotentials, such as those arising in flux compactifications. We discuss those vacua and their corresponding walls in this section. The discussion largely uses the solutions constructed in [46], whose notation we largely follow.

Let us consider a theory with a single axion-saxion complex modulus $\Phi = a + iv$. The 4d effective action, in Planck units, is

$$S = -\int d^4x \sqrt{-g} \left[\frac{1}{2}R + \frac{|\partial \Phi|^2}{4(\operatorname{Im} \Phi)^2} + V(\Phi, \overline{\Phi}) \right]$$
(5.1)

with the $\mathcal{N} = 1$ scalar potential

$$V(\Phi,\overline{\Phi}) = e^{K} \left(K^{\Phi\overline{\Phi}} |D_{\Phi}W|^{2} - 3|W|^{2} \right).$$
(5.2)

We focus on theories of the kind considered in [46], where the superpotential is induced from a set of fluxes m^{I}, e_{I} , with I = 0, 1, and is given by

$$W = e_I f^I(\Phi) - m^I \mathcal{G}_I(\Phi) \tag{5.3}$$

for f^I , \mathcal{G}_I some holomorphic functions whose detailed structure we do not need to specify.

In general, these fluxes induce a dynamical tadpole for Φ , unless it happens to sit at the minimum of the potential. The results in [46] allow to build 1/2 BPS running solutions depending on one space coordinate y with

$$ds^2 = e^{2Z(y)} dx_\mu dx^\mu + dy^2 \,. \tag{5.4}$$

For the profile of the scalar, the solution has constant axion a, but varying saxion. Defining the 'central charge' $\mathcal{Z} = e^{\mathcal{K}/2} W$ and \mathcal{Z}_* its value at the minimum of the potential (and similarly for other quantities), the profile for the scalar v is

$$v(y) = v_* \coth^2\left(\frac{1}{2}|\mathcal{Z}_*|y\right).$$
(5.5)

Note that in [46] this solution was built as 'the left hand side' of an interpolating domain wall solution (more about it later), but we consider it as the full solution in our setup. Note also that we have shifted the origin in y with respect to the choice in [46].

The backreaction of the scalar profile on the metric is described by

$$Z(y) = d + e^{-\frac{1}{2}\hat{\mathcal{K}}_0} \left[\log\left(-\sinh\left(\frac{1}{2}|\mathcal{Z}_*|y\right)\right) + \log\cosh\left(\frac{1}{2}|\mathcal{Z}_*|y\right) \right], \quad (5.6)$$

where d is just an integration constant and $\hat{\mathcal{K}}_0$ is an additive constant in the Kähler potential.

The solution exhibits a singularity at y = 0, which (since the metric along y is flat) is at finite distance in spacetime from other points. On the other hand it is easy to see that the scalar v runs off to infinity as we hit the wall, since

$$v(y) \to 4 v_* |\mathcal{Z}_*|^{-2} y^{-2} \text{ as } y \to 0.$$
 (5.7)

We can obtain the scaling of the moduli space distance with the spacetime distance. Using the kinetic term in (5.1),

$$D = -\int \frac{1}{\sqrt{2v}} \frac{dv}{dy} dy \simeq -\sqrt{2} \log y \simeq -\sqrt{2} \log \Delta.$$
 (5.8)

In the last two equalities we have used (5.7) and (5.4) respectively. We thus get a familiar power-like scaling for the SDC scale

$$M_{\rm SDC} \sim \Delta^{\sqrt{2}\lambda}$$
 (5.9)

We also recover the ADC-like scaling with the scalar curvature. At leading order in $y \to 0$ one finds

$$\log|R| \simeq -2\log y \simeq \sqrt{2D}, \qquad (5.10)$$

which gives

$$M_{\rm SDC} \sim |R|^{-\frac{1}{\sqrt{2}}\lambda}.$$
(5.11)

This all fits very nicely with our picture that the solution is describing a cobordism wall of nothing, and that the solution for y > 0 is unphysical and not realized. This provides an effective theory description of the cobordism defects for general 4d $\mathcal{N} = 1$ theories, in a dynamical framework. It would be interesting to find explicit microscopic realizations of this setup.

Let us conclude this section by mentioning that it is possible to patch together several solutions of the above kind and build cobordism domain walls interpolating between different flux vacua. In particular in [46] the solution provided 'the left hand side' of one such interpolating domain wall solution whose 'right hand side' was glued before reaching (in our choice of origin) y = 0, hence before encountering the wall of nothing. The particular solution on the right hand side was chosen to sit at the minimum of the corresponding potential, for which there is no tadpole and thus the functions D and v are simply set to constants, fixed to guarantee continuity. Consequently, the solutions remain at finite distance in moduli space, in agreement with our picture for interpolating domain walls. In some sense, the flux changing membrane is absorbing the tadpole, thus avoiding the appearance of the wall of nothing. We refer the reader to [46] (see also [25]) for a detailed discussion.

6 Walls in 10d non-supersymmetric strings

The above examples all correspond to supersymmetric solutions, and even the resulting running solutions preserve some supersymmetry. This is appropriate to establish our key results, but we would like to illustrate that they are not restricted to supersymmetric setups. In order to illustrate that these ideas can apply more generally, and can serve as useful tools for the study of non-supersymmetric theories, we present a quick discussion of the 10d non-supersymmetric USp(32) theory [47], building on the solution constructed in [13] and revised in [11].¹²

The 10d (Einstein frame) action reads

$$S_E = \frac{1}{2\kappa^2} \int d^{10}x \sqrt{-G} \left[R - \frac{1}{2} (\partial \phi)^2 \right] - T_9^E \int d^{10}x \sqrt{-G} \, 64 \, e^{\frac{3\phi}{2}} \,, \tag{6.1}$$

where T_9^E is the (anti)D9-brane tension. The theory has a dynamical dilaton tadpole $\mathcal{T} \sim T_9^E g_s^{3/2}$, and does not admit maximally symmetric solutions. The running solution in [13] preserves 9d Poincaré invariance, and reads

$$\phi = \frac{3}{4} \alpha_E y^2 + \frac{2}{3} \log |\sqrt{\alpha_E} y| + \phi_0 ,$$

$$ds_E^2 = |\sqrt{\alpha_E} y|^{\frac{1}{9}} e^{-\frac{\alpha_E y^2}{8}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + |\sqrt{\alpha_E} y|^{-1} e^{-\frac{3\phi_0}{2}} e^{-\frac{9\alpha_E y^2}{8}} dy^2 , \qquad (6.2)$$

where $\alpha_E = 64k^2T_9$. There are two singularities, at y = 0 and $y \to \infty$, which despite appearances are located at finite spacetime distance, satisfying the scaling $\Delta^{-2} \sim \mathcal{T}$ introduced in [11]. In this case, there is no known microscopic description for the underlying cobordism defect, but we can still consider the effective theory solution to study the theory as we hit the walls.

We consider the two singularities at $y = 0, \infty$, and look at the behaviour of the solution near them. The distance from a generic point y to the singularities is given by the integral [11]

$$\Delta \sim \int |\sqrt{\alpha_E}y|^{-\frac{1}{2}} e^{-\frac{3\phi_0}{4}} e^{-\frac{9\alpha_E y^2}{16}} \, dy \,, \tag{6.3}$$

on the intervals [y, 0] when $y \to 0$, and $[y, \infty]$ when $y \to \infty$. They give (lower and upper) incomplete gamma functions

$$\Delta_0 \sim \gamma \left(\frac{1}{4}, \frac{9\alpha_E y^2}{16}\right) \quad \text{and} \quad \Delta_\infty \sim \Gamma \left(\frac{1}{4}, \frac{9\alpha_E y^2}{16}\right).$$
 (6.4)

By expanding at leading order as $y \to 0$ and $y \to \infty$, one gets

$$\Delta_0 \sim y^{\frac{1}{2}}$$
 and $\Delta_\infty \sim y^{-\frac{3}{2}} e^{-\frac{9\alpha_E y^2}{16}}$. (6.5)

The moduli space distance is $\phi = \sqrt{2}D$. Its leading behavior is $D \simeq -\frac{\sqrt{2}}{3}\ln y$ as $y \to 0^+$ and $D \simeq \frac{3\alpha_E}{4\sqrt{2}}y^2$ as $y \to \infty$. This leads to the scaling relations

$$y \to 0^{+}: \quad \Delta_{0} \sim e^{-\frac{3}{2\sqrt{2}}D}, y \to \infty: \quad \Delta_{\infty} \sim D^{-\frac{3}{4}} e^{-\frac{3}{2\sqrt{2}}D} \sim e^{-\frac{3}{2\sqrt{2}}D - \frac{3}{4}\ln D}.$$
(6.6)

 $^{^{12}}$ For other references related to dynamical tadpoles in non-supersymmetric theories, see [14–20].

In both cases we have the moduli space distance running off to infinity as we approach the wall. This is in agreement with their interpretation as cobordism walls of nothing.¹³ Moreover, we recover the a familiar power-like scaling of the SDC mass scale with the same numerical factors in both cases

$$M_{\rm SDC} \sim e^{-\lambda D} \sim \Delta^{\frac{2\sqrt{2}}{3}\lambda}$$
 (6.7)

It is interesting to see that one can also recover a standard power-like scaling for both singularities if the moduli space distance D is compared with the spacetime curvature scalar R. The latter reads

$$|R| = \sqrt{\alpha_E} \, e^{\frac{3\phi_0}{2}} \left(\frac{2}{9}y^{-1} + \frac{7}{2}\alpha_E y + \frac{9}{8}\alpha_E^2 y^3\right) e^{\frac{9\alpha_E}{8}y^2} \,. \tag{6.8}$$

Let us start with the $y \to 0$ singularity. We can approximate the logarithm of the scalar curvature as

$$\log|R| \simeq -\log y \simeq \frac{3}{\sqrt{2}}D.$$
(6.9)

This allows to rewrite the SDC scaling in the form of the ADC-like scaling

$$M_{\rm SDC} \sim e^{-\lambda\Delta} \sim |R|^{-\frac{\sqrt{2}}{3}\lambda}$$
 (6.10)

Let us now turn to the $y \to \infty$ limit. In this case the logarithm of the scalar curvature is approximated to

$$\log|R| \simeq \frac{9\alpha_E}{8}y^2 \simeq \frac{3}{\sqrt{2}}D, \qquad (6.11)$$

thus recovering the same behavior as for the other singularity.

As announced, we find a nice power-like scaling, reminiscent as usual of the ADC relations. It is amusing that the precise coefficient arises in both the strong and weak coupling singularities, which may hint towards some universality or duality relation in this non-supersymmetric 10d model.

7 Final remarks

In this work we have considered running solutions solving the equations of motion of theories with tadpoles for dynamical fields. These configurations were shown to lead to cobordism walls of nothing at finite distance in spacetime [11], in a dynamical realization of the Cobordism Conjecture. We have also discussed interpolating domain walls across which we change to a different (but cobordant) theory/vacuum. We have shown that the key criterion distinguishing both kinds of walls is related to distance in field space: walls of nothing are characterized by the scalars attaining infinite distance in moduli space, while interpolating domain walls remain at finite distance in moduli space.

¹³The interpretation of the $y \to 0$ singularity as a wall of nothing was deemed unconventional, since it would arise at weak coupling. It is interesting that we get additional support for this interpretation from the moduli space distance behaviour.

Hence, cobordism walls of nothing provide excellent probes of the structure of the effective theory near infinite distance points, and in particular the Swampland Distance Conjectures. This viewpoint encompasses and generalizes that advocated for EFT strings in 4d $\mathcal{N} = 1$ theories in [24]. We have found interesting new general scaling relations linking, for running solutions, the moduli space distance and the SDC tower mass scale to geometric spacetime quantities, such as the distance to the wall or the scalar curvature. The latter takes a form tantalizingly reminiscent of the Anti de Sitter Distance Conjecture (ADC), suggesting it may relate to the generalized distance in [23].

We have illustrated the key ideas in several large classes of string models, most often in supersymmetric setups (yet with nontrivial scalar potentials to produce the dynamical tadpole triggering the running); however, we emphasize that we expect similar behaviours in non-supersymmetric theories, as we have shown explicitly for the 10d non-susy USp(32) theory.

There are several interesting open question that we leave for future work:

- We have mainly focused on space-dependent running solutions. It is clearly interesting to consider time-dependent solutions, extending existing results in the literature [13–20], and exploit them in applications, in particular with an eye on possible implications for inflationary models or quintessence.
- A particular class of time-dependent solutions are dynamical bubbles. In particular, a tantalizing observation is that in the original bubble of nothing in [6], the 4d radion modulus goes to zero size (which lies at infinite distance in moduli space of the S¹ compactification) as one hits the wall. Although the setup is seemingly unrelated, it would be interesting to understand universal features of bubbles of nothing along the lines considered in our work.
- The appearance of ADC-like scaling relations in our running solutions possibly signals an underlying improved understanding of infinite distance limits in dynamical (rather than adiabatic) configurations. For instance, as shown in [21], the r → ∞ limit in the Klebanov-Strassler solution [48] avoids the appearance of a tower of states becoming massless exponentially with the distance. This was related to having a non-geodesic trajectory in moduli space (see [49] for a general discussion about non-geodesics and the SDC). However, as dictated by the lack of separation of scales in this model, an ADC-like scaling is yet respected as the scalar curvature goes to zero in this limit. This could point to a more universal way of writing the SDC in dynamical configurations.
- In all the examples we find precise numbers relating the parameter in the SDC λ to the power in the ADC-like scaling. It would certainly be interesting to find a pattern in these values and possibly relate them to properties of the infinite distance limits along the lines of [41–43]. On a similar line of thought, it has been argued that in supersymmetric cases the ADC's scaling parameter should be 1/2 [23], assuming this applies to our setup, it would be interesting to extract the SDC's parameter λ from our supersymmetric examples with an ADC-like scaling. It would be remarkable that they match the existing proposals for the value of λ.

- The ADC-like scaling may also signal some potential interplay with the Gravitino Distance Conjecture [50, 51]. One expects to find a power relation between the mass of the gravitino and the scalar curvature of the solution, it would be certainly interesting to test this and to look for some pattern in the corresponding powers.
- The trajectory in moduli space in spacetime-dependent solutions has a strong presence in the study of black holes, in particular attractor equations and flows. The attempts to relate them to the SDC (see e.g. [52]) can have an interesting interplay with our general framework.
- We certainly expect interesting new applications of our results to the study of nonsupersymmetric strings, and to supersymmetry-breaking configurations in string theory.

We hope to report on these problems in the near future.

Acknowledgments

We are pleased to thank Ivano Basile, Inaki García-Etxebarria, Luis Ibánez, Fernando Marchesano, Miguel Montero, Pablo Soler and Irene Valenzuela for useful discussions. This work is supported by the Spanish Research Agency (Agencia Española de Investigación) through the grants IFT Centro de Excelencia Severo Ochoa SEV-2016-0597, the grant GC2018-095976-B-C21 from MCIU/AEI/FEDER, UE. The work by J.C. is supported by the FPU grant no. FPU17/04181 from Spanish Ministry of Education.

A Holographic examples

In [11] it was shown that Dynamical Cobordism underlies the structure of the gravity dual of the $SU(N) \times SU(N + M)$ conifold theory, namely fractional brane deformation of $AdS_5 \times T^{1,1}$. This in fact explains the appearance of a singularity at finite radial distance [53] and its smoothing out into a configuration capping of the warped throat [48], as a cobordism wall of nothing. In this appendix we provide some examples of other warped throat configurations which illustrate the appearance of other cobordism walls of nothing, and cobordism domain walls interpolating between theories corresponding to compactification on horizons of different topology. The discussion is strongly inspired by the constructions in [54] (see also [55]).

A.1 Domain wall to a new vacuum

As a first example we consider a configuration in which a running of the conifold theory hits a wall (given by the tip of a KS throat) interpolating to an $AdS_5 \times S^5$ vacuum. The latter is the maximally symmetric solution of a theory at the bottom of its potential, i.e. with no dynamical tadpole. We carry out the discussion in terms of the dual field theory, which translates easily into the just explained gravity picture. The dilaton is constant in the whole configuration, so we skip factors of g_s . Consider the conifold theory with $SU(N) \times SU(N + M)$ at some scale, i.e. at some position r there are N units of 5-form flux and M units of 3-form flux. The Klebanov-Tseytlin solution [53] gives a running for the effective flux

$$N(r) = N + M^2 \log(r), \qquad (A.1)$$

and we get a singu at a value r_0 defined by

$$N + M^2 \log r_0 = 0 \quad \Rightarrow \quad r_0 = e^{-N/M^2}.$$
 (A.2)

Naively, the singularity would seem to be smoothed out into a purely geometric background with a finite size S^3 . Indeed, this is the full story if N is multiple of M, namely N = KM: in the field theory, the $SU(KM) \times SU(KM + M)$ theory suffers a cascade of K Seiberg dualities in which K decreases by one unit in each step. Morally, the cascade ends when the effective K = 0 and then we just have a pure SU(M) SYM, which confines and develops a mass gap. This is the end of the RG flow, with no more running, hence the spacetime in capped off in the IR region of the dual throat.

However, as also noticed in [48], the story is slightly different if N = KM + P. After the K steps in the duality cascade, one is left over with an SU(P) gauge theory with three complex scalar degrees of freedom parametrizing a deformed mesonic moduli space corresponding to (the symmetrization of P copies of) the deformed conifold. This gauge theory flows to $\mathcal{N} = 4$ SU(P) SYM in the infrared, which is a conformal theory. In the parameter range $1 \ll P \ll M \ll N$, the whole configuration admits a weakly coupled supergravity dual given by a KS throat at which infrared region we have a finite size \mathbf{S}^3 , at which P D3-branes (which we take coincident) would be located; however, since P is large, they backreact and carve out a further $\mathrm{AdS}_5 \times \mathbf{S}^5$ with P units of RR 5-form flux, which continues the radial direction beyond the KS throat endpoint region. Hence, this region acts as an interpolating domain wall between two different (but cobordant) theories, namely the conifold throat (with a dynamical tadpole from the fractional brane charge), and the $\mathrm{AdS}_5 \times \mathbf{S}^5$ vacuum (with no tadpole, and preserving maximal symmetry). The picture is summarized in figure 1.

A.2 Domain wall to a new running solution

Running can lead to an interpolating domain wall, across which we find not a vacuum, but a different running solution (subsequently hitting a wall of nothing, other interpolating domain walls, or just some AdS vacuum). We now illustrate this idea with an example of a running solution A hitting a domain wall interpolating to a second running solution B, which subsequently hits a wall of nothing. The example is based on the multi-flux throat construction in [54] (whose dimer picture is given in [56]). It is easy to devise other generalizations displaying the different behaviours mentioned above.

Consider the system of D3-branes at the singularity given by the complex cone over dP_3 . The gauge theory is described by the quiver and dimer diagrams¹⁴ in figure 2.

 $^{^{14}}$ For references, see [55, 57–59].



Figure 1. Domain wall interpolating between the conifold theory with fractional branes, and an AdS vacuum. Figure a) shows a heuristic intermediate step of a KS solution with a number P of left-over probe D3-branes. If P is large, the appropriate description requires including the backreaction of the D3-branes, which lead to a further AdS throat, to the left of the picture in figure b). Hence the running of the dynamical tadpole in the right hand side ends in a domain wall separating it from an AdS vacuum.



Figure 2. The quiver and dimer diagrams describing the gauge theory on D3-branes at the tip of the complex cone over dP_3 .

We can add fractional branes, i.e. rank assignments compatible with cancellation of non-abelian anomalies. There are several choices, corresponding to different fluxes on the 3-cycles in the dual gravitational theory. Some of them correspond to 3-cycles which can be grown out of the singular origin to provide a complex deformation of the CY. These are described as the splitting of the web diagram into sub-webs in equilibrium, see [56]. In particular we focus on the complex deformation of complex cone over dP_3 to a conifold, see the web diagram in figure 3.

There are two kinds of fractional branes, associated to M and P. In the gravity dual, these correspond to RR 3-form fluxes on 3-cycles (obtained by an S^1 fibration over a 2-cycle on dP_3), and there are NSNS 3-form fluxes in the dual 3-cycles. These are non-compact, namely they span a 2-cycle (dual to the earlier 2-cycle in dP_3) and the radial direction. For more details about the quantitative formulas of this kind of solution, see section 5 of [60].



Figure 3. a) Web diagram of the complex cone over dP_3 splitting into three sub-webs. b) Rank assignment (fractional branes) that trigger those complex deformations.



Figure 4. a) Quiver of the dP_3 theory in the last step of the first cascade, which turns into the conifold upon strong dynamics of the nodes 1 and 4. b) Same story in the dimer picture.

If we focus in the regime¹⁵ $P \ll M$, then the larger flux M implies a larger corresponding component of the H_3 flux, which means a faster running of the corresponding 5d NSNS axion. The axion associated to the flux P also runs, but more slowly. In the field theory, the duality cascade is controlled by M, so that N is reduced in multiples of M (at leading order in P/M). When N is exhausted we are left with a rank assignment as given in figure 4a. The result of the strong dynamics triggered by M can be worked out in field theory as in [54] or using dimers as in [56]. All the info about this last description is in figure 4b.

The result is a conifold theory with M regular branes and P fractional branes. This is the standard KS story (with just different labels for the branes): M decreases in sets of Puntil it is exhausted, then the running stops due to strong dynamics. In the gravity dual, we

 $^{^{15}}$ Note that in [54] the regime is the opposite, but both kinds of fractional branes are similar, so the result is the same up to relabeling.



Figure 5. Domain wall interpolating between the theory on dP_3 with (M + P) fractional branes, and a conifold theory with M regular branes and P fractional ones. The running of one of the dynamical tadpoles in the dP_3 theory stops at the wall but the other continues running until it reaches the \mathbf{S}^3 at the bottom of the KS throat.

have a KS throat sticking out and spacetime ends on the usual \mathbf{S}^3 (alternatively, if M is not a multiple of P, there is a number P of leftover D3-branes, which, if large, can trigger a further AdS throat as in section A.1. A sketch of the gravity dual picture is shown in figure 5.

Note that this kind of domain wall interpolates into two topologically different compactifications. As we cross it, the compactification space changes, and the spectrum of light fields changes (at the massless level, one of the axions ceases to exist). In this sense, it is a cobordism domain wall connecting two different quantum gravity theories [5].

A.3 Cobordism domain walls to disconnected solutions

The construction of singularities admitting complicated patterns of complex deformations (or resolutions) can be carried out systematically for toric singularities, using the techniques in [55]. This can be used to build sequences of domain walls realizing a plethora of possibilities. For our last class of examples, we consider cobordism domain walls to disconnected theories.

This has already been realized in the geometry used in [61] to build a bifd throat, i.e. two throats at the bottom of a throat, see figure 6. These had been proposed in [62] as possible hosts of axion monodromy inflation models (see [63-67] for additional references).

Actually, a far simpler way of getting a running solution with a domain wall to a disconnected set of e.g. vacua is to consider the KS setup in section A.1, with the P leftover D3-branes split into two stacks P_1 and P_2 of D3-branes at separated locations on the \mathbf{S}^3 (with $P_1, P_2 \gg 1$). This corresponds to turning on a vev v for a Higgsing $\mathrm{SU}(P) \to \mathrm{SU}(P_1) \times \mathrm{SU}(P_2)$ (with $P_1 + P_2 = P$) with a scale for v much smaller than the scale of confinement Λ of the original $\mathrm{SU}(KM + P) \times \mathrm{SU}(KM + M + P)$ theory. In the gravity dual, we have a running solution in the holographic direction, towards low energies; upon reaching Λ , we have the \mathbf{S}^3 domain wall, out of which we have one $\mathrm{AdS}_5 \times S^5$ -like



Figure 6. Picture of a bifid throat. It represents a domain wall implementing a cobordism between one theory and a disconnected set of two quantum gravity theories.



Figure 7. Picture of a bifid throat with two AdS tongues. It represents a domain wall implementing a cobordism between one theory and a disconnected set of two AdS theories.

vacuum (with flux P), until we hit the scale v, and the single throat splits into two AdS₅×S⁵ throats (with fluxes P_1, P_2). If $v \simeq \Lambda$, the splitting of throats happens in the same regime as the domain wall ending the run of the initial solution. This is depicted in figure 7.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- [1] C. Vafa, The String landscape and the swampland, hep-th/0509212 [INSPIRE].
- [2] T.D. Brennan, F. Carta and C. Vafa, The String Landscape, the Swampland, and the Missing Corner, PoS TASI2017 (2017) 015 [arXiv:1711.00864] [INSPIRE].
- [3] E. Palti, The Swampland: Introduction and Review, Fortsch. Phys. 67 (2019) 1900037 [arXiv:1903.06239] [INSPIRE].

- [4] M. van Beest, J. Calderón-Infante, D. Mirfendereski and I. Valenzuela, Lectures on the Swampland Program in String Compactifications, arXiv:2102.01111 [INSPIRE].
- [5] J. McNamara and C. Vafa, Cobordism Classes and the Swampland, arXiv: 1909.10355 [INSPIRE].
- [6] E. Witten, Instability of the Kaluza-Klein Vacuum, Nucl. Phys. B 195 (1982) 481 [INSPIRE].
- [7] H. Ooguri and L. Spodyneiko, New Kaluza-Klein instantons and the decay of AdS vacua, Phys. Rev. D 96 (2017) 026016 [arXiv:1703.03105] [INSPIRE].
- [8] I. García Etxebarria, M. Montero, K. Sousa and I. Valenzuela, Nothing is certain in string compactifications, JHEP 12 (2020) 032 [arXiv:2005.06494] [INSPIRE].
- [9] M. Montero and C. Vafa, Cobordism Conjecture, Anomalies, and the String Lampost Principle, JHEP 01 (2021) 063 [arXiv:2008.11729] [INSPIRE].
- [10] H.-C. Tarazi and C. Vafa, On The Finiteness of 6d Supergravity Landscape, arXiv:2106.10839 [INSPIRE].
- [11] G. Buratti, M. Delgado and A.M. Uranga, *Dynamical tadpoles, stringy cobordism, and the* SM from spontaneous compactification, JHEP 06 (2021) 170 [arXiv:2104.02091] [INSPIRE].
- [12] A. Mininno and A.M. Uranga, Dynamical tadpoles and Weak Gravity Constraints, JHEP 05 (2021) 177 [arXiv:2011.00051] [INSPIRE].
- [13] E. Dudas and J. Mourad, Brane solutions in strings with broken supersymmetry and dilaton tadpoles, Phys. Lett. B 486 (2000) 172 [hep-th/0004165] [INSPIRE].
- [14] R. Blumenhagen and A. Font, Dilaton tadpoles, warped geometries and large extra dimensions for nonsupersymmetric strings, Nucl. Phys. B 599 (2001) 241 [hep-th/0011269] [INSPIRE].
- [15] E. Dudas, J. Mourad and C. Timirgaziu, Time and space dependent backgrounds from nonsupersymmetric strings, Nucl. Phys. B 660 (2003) 3 [hep-th/0209176] [INSPIRE].
- [16] E. Dudas, G. Pradisi, M. Nicolosi and A. Sagnotti, On tadpoles and vacuum redefinitions in string theory, Nucl. Phys. B 708 (2005) 3 [hep-th/0410101] [INSPIRE].
- [17] J. Mourad and A. Sagnotti, AdS Vacua from Dilaton Tadpoles and Form Fluxes, Phys. Lett. B 768 (2017) 92 [arXiv:1612.08566] [INSPIRE].
- [18] I. Basile, J. Mourad and A. Sagnotti, On Classical Stability with Broken Supersymmetry, JHEP 01 (2019) 174 [arXiv:1811.11448] [INSPIRE].
- [19] R. Antonelli and I. Basile, Brane annihilation in non-supersymmetric strings, JHEP 11 (2019) 021 [arXiv:1908.04352] [INSPIRE].
- [20] I. Basile, On String Vacua without Supersymmetry: brane dynamics, bubbles and holography, Ph.D. Thesis, Scuola normale superiore di Pisa, Pisa, Scuola Normale Superiore (2020)
 [arXiv:2010.00628] [INSPIRE].
- [21] G. Buratti, J. Calderón and A.M. Uranga, Transplanckian axion monodromy!?, JHEP 05 (2019) 176 [arXiv:1812.05016] [INSPIRE].
- [22] H. Ooguri and C. Vafa, On the Geometry of the String Landscape and the Swampland, Nucl. Phys. B 766 (2007) 21 [hep-th/0605264] [INSPIRE].
- [23] D. Lüst, E. Palti and C. Vafa, AdS and the Swampland, Phys. Lett. B 797 (2019) 134867
 [arXiv:1906.05225] [INSPIRE].

- [24] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, The EFT stringy viewpoint on large distances, arXiv:2104.05726 [INSPIRE].
- [25] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, Swampland Conjectures for Strings and Membranes, JHEP 02 (2021) 006 [arXiv:2006.15154] [INSPIRE].
- [26] J. Polchinski and E. Witten, Evidence for heterotic type-I string duality, Nucl. Phys. B 460 (1996) 525 [hep-th/9510169] [INSPIRE].
- [27] E. Witten, Strong coupling expansion of Calabi-Yau compactification, Nucl. Phys. B 471 (1996) 135 [hep-th/9602070] [INSPIRE].
- [28] B.R. Greene, K. Schalm and G. Shiu, Dynamical topology change in M-theory, J. Math. Phys. 42 (2001) 3171 [hep-th/0010207] [INSPIRE].
- [29] L.J. Romans, Massive N = 2a Supergravity in Ten-Dimensions, Phys. Lett. B 169 (1986) 374 [INSPIRE].
- [30] E. Bergshoeff, M.B. Green, G. Papadopoulos and P.K. Townsend, The IIA supereight-brane, hep-th/9511079 [INSPIRE].
- [31] N. Seiberg, Five-dimensional SUSY field theories, nontrivial fixed points and string dynamics, Phys. Lett. B 388 (1996) 753 [hep-th/9608111] [INSPIRE].
- [32] A. Lukas, B.A. Ovrut, K.S. Stelle and D. Waldram, The Universe as a domain wall, Phys. Rev. D 59 (1999) 086001 [hep-th/9803235] [INSPIRE].
- [33] A. Lukas, B.A. Ovrut, K.S. Stelle and D. Waldram, *Heterotic M-theory in five-dimensions*, Nucl. Phys. B 552 (1999) 246 [hep-th/9806051] [INSPIRE].
- [34] A. Lukas, B.A. Ovrut and D. Waldram, Heterotic M-theory vacua with five-branes, Fortsch. Phys. 48 (2000) 167 [hep-th/9903144] [INSPIRE].
- [35] B.A. Ovrut, N = 1 supersymmetric vacua in heterotic M-theory, hep-th/9905115 [INSPIRE].
- [36] B.A. Ovrut, Lectures on heterotic M-theory, in Theoretical Advanced Study Institute in Elementary Particle Physics (TASI 2001): Strings, Branes and EXTRA Dimensions, pp. 359–406 (2002) [hep-th/0201032] [INSPIRE].
- [37] T. Mohaupt and F. Saueressig, Dynamical conifold transitions and moduli trapping in M-theory cosmology, JCAP 01 (2005) 006 [hep-th/0410273] [INSPIRE].
- [38] B.R. Greene, D.R. Morrison and A. Strominger, Black hole condensation and the unification of string vacua, Nucl. Phys. B 451 (1995) 109 [hep-th/9504145] [INSPIRE].
- [39] T.-m. Chiang, B.R. Greene, M. Gross and Y. Kanter, Black hole condensation and the web of Calabi-Yau manifolds, Nucl. Phys. B Proc. Suppl. 46 (1996) 82 [hep-th/9511204] [INSPIRE].
- [40] L.E. Ibáñez and A.M. Uranga, String theory and particle physics: An introduction to string phenomenology, Cambridge University Press (2012) [INSPIRE].
- [41] T.W. Grimm, E. Palti and I. Valenzuela, Infinite Distances in Field Space and Massless Towers of States, JHEP 08 (2018) 143 [arXiv:1802.08264] [INSPIRE].
- [42] P. Corvilain, T.W. Grimm and I. Valenzuela, The Swampland Distance Conjecture for Kähler moduli, JHEP 08 (2019) 075 [arXiv:1812.07548] [INSPIRE].
- [43] T.W. Grimm, C. Li and I. Valenzuela, Asymptotic Flux Compactifications and the Swampland, JHEP 06 (2020) 009 [Erratum ibid. 01 (2021) 007] [arXiv:1910.09549]
 [INSPIRE].

- [44] C. Pope, Lectures on Kaluza-Klein, http://people.tamu.edu/~c-pope/ihplec.pdf.
- [45] R. Blumenhagen, M. Brinkmann and A. Makridou, Quantum Log-Corrections to Swampland Conjectures, JHEP 02 (2020) 064 [arXiv:1910.10185] [INSPIRE].
- [46] I. Bandos, F. Farakos, S. Lanza, L. Martucci and D. Sorokin, *Three-forms, dualities and membranes in four-dimensional supergravity*, *JHEP* 07 (2018) 028 [arXiv:1803.01405]
 [INSPIRE].
- [47] S. Sugimoto, Anomaly cancellations in type-I D9-D9 system and the USp(32) string theory, Prog. Theor. Phys. 102 (1999) 685 [hep-th/9905159] [INSPIRE].
- [48] I.R. Klebanov and M.J. Strassler, Supergravity and a confining gauge theory: Duality cascades and χSB-resolution of naked singularities, JHEP 08 (2000) 052 [hep-th/0007191] [INSPIRE].
- [49] J. Calderón-Infante, A.M. Uranga and I. Valenzuela, The Convex Hull Swampland Distance Conjecture and Bounds on Non-geodesics, JHEP 03 (2021) 299 [arXiv:2012.00034]
 [INSPIRE].
- [50] N. Cribiori, D. Lüst and M. Scalisi, The gravitino and the swampland, JHEP 06 (2021) 071 [arXiv:2104.08288] [INSPIRE].
- [51] A. Castellano, A. Font, A. Herraez and L.E. Ibáñez, A Gravitino Distance Conjecture, arXiv:2104.10181 [INSPIRE].
- [52] Q. Bonnefoy, L. Ciambelli, D. Lüst and S. Lüst, Infinite Black Hole Entropies at Infinite Distances and Tower of States, Nucl. Phys. B 958 (2020) 115112 [arXiv:1912.07453]
 [INSPIRE].
- [53] I.R. Klebanov and A.A. Tseytlin, Gravity duals of supersymmetric $SU(N) \times SU(N + M)$ gauge theories, Nucl. Phys. B 578 (2000) 123 [hep-th/0002159] [INSPIRE].
- [54] S. Franco, A. Hanany and A.M. Uranga, Multi-flux warped throats and cascading gauge theories, JHEP 09 (2005) 028 [hep-th/0502113] [INSPIRE].
- [55] I. Garcia-Etxebarria, F. Saad and A.M. Uranga, Quiver gauge theories at resolved and deformed singularities using dimers, JHEP 06 (2006) 055 [hep-th/0603108] [INSPIRE].
- [56] S. Franco, A. Hanany, F. Saad and A.M. Uranga, Fractional branes and dynamical supersymmetry breaking, JHEP 01 (2006) 011 [hep-th/0505040] [INSPIRE].
- [57] A. Hanany and K.D. Kennaway, Dimer models and toric diagrams, hep-th/0503149 [INSPIRE].
- [58] S. Franco, A. Hanany, K.D. Kennaway, D. Vegh and B. Wecht, Brane dimers and quiver gauge theories, JHEP 01 (2006) 096 [hep-th/0504110] [INSPIRE].
- [59] K.D. Kennaway, Brane Tilings, Int. J. Mod. Phys. A 22 (2007) 2977 [arXiv:0706.1660] [INSPIRE].
- [60] S. Franco, Y.-H. He, C. Herzog and J. Walcher, Chaotic duality in string theory, Phys. Rev. D 70 (2004) 046006 [hep-th/0402120] [INSPIRE].
- [61] A. Retolaza, A.M. Uranga and A. Westphal, Bifid Throats for Axion Monodromy Inflation, JHEP 07 (2015) 099 [arXiv:1504.02103] [INSPIRE].
- [62] L. McAllister, E. Silverstein and A. Westphal, Gravity Waves and Linear Inflation from Axion Monodromy, Phys. Rev. D 82 (2010) 046003 [arXiv:0808.0706] [INSPIRE].

- [63] E. Silverstein and A. Westphal, Monodromy in the CMB: Gravity Waves and String Inflation, Phys. Rev. D 78 (2008) 106003 [arXiv:0803.3085] [INSPIRE].
- [64] N. Kaloper and L. Sorbo, A Natural Framework for Chaotic Inflation, Phys. Rev. Lett. 102 (2009) 121301 [arXiv:0811.1989] [INSPIRE].
- [65] F. Marchesano, G. Shiu and A.M. Uranga, F-term Axion Monodromy Inflation, JHEP 09 (2014) 184 [arXiv:1404.3040] [INSPIRE].
- [66] A. Hebecker, S.C. Kraus and L.T. Witkowski, D7-Brane Chaotic Inflation, Phys. Lett. B 737 (2014) 16 [arXiv:1404.3711] [INSPIRE].
- [67] L. McAllister, E. Silverstein, A. Westphal and T. Wrase, *The Powers of Monodromy*, *JHEP* 09 (2014) 123 [arXiv:1405.3652] [INSPIRE].

Published for SISSA by 2 Springer

RECEIVED: April 4, 2022 REVISED: May 18, 2022 ACCEPTED: May 23, 2022 PUBLISHED: June 24, 2022

At the end of the world: Local Dynamical Cobordism

Roberta Angius, José Calderón-Infante, Matilda Delgado, Jesús Huertas and Angel M. Uranga

Instituto de Física Teórica IFT-UAM/CSIC, C/ Nicolás Cabrera 13-15, Campus de Cantoblanco, 28049 Madrid, Spain E-mail: roberta.angius@csic.es, j.calderon.infante@csic.es, matilda.delgado@uam.es, j.huertas@csic.es, angel.uranga@csic.es

ABSTRACT: The Cobordism Conjecture states that any Quantum Gravity configuration admits, at topological level, a boundary ending spacetime. We study the dynamical realization of cobordism, as spacetime dependent solutions of Einstein gravity coupled to scalars containing such end-of-the-world 'branes'. The latter appear in effective theory as a singularity at finite spacetime distance at which scalars go off to infinite field space distance. We provide a local description near the end-of-the-world branes, in which the solutions simplify dramatically and are characterized in terms of a critical exponent, which controls the asymptotic profiles of fields and the universal scaling relations among the spacetime distance to the singularity, the field space distance, and the spacetime curvature. The analysis does not rely on supersymmetry. We study many explicit examples of such Local Dynamical Cobordisms in string theory, including 10d massive IIA, the 10d nonsupersymmetric USp(32) theory, Bubbles of Nothing, 4d $\mathcal{N} = 1$ cosmic string solutions, the Klebanov-Strassler throat, Dp-brane solutions, brane configurations related to the D1/D5 systems, and small black holes. Our framework encompasses diverse recent setups in which scalars diverge at the core of defects, by regarding them as suitable end-of-the-world branes. We explore the interplay of Local Dynamical Cobordisms with the Distance Conjecture and other swampland constraints.

KEYWORDS: D-Branes, Flux Compactifications, Superstring Vacua

ARXIV EPRINT: 2203.11240



Contents

1	Introduction	1
2	Local dynamical cobordisms	3
	2.1 General ansatz	3
	2.2 Local description of end of the world branes	5
3	Some 10d examples	8
	3.1 The 10d massive type IIA theory	8
	3.2 The 10d non-supersymmetric $USp(32)$ string	9
	3.2.1 r o 0	10
	3.2.2 $r \to \infty$	10
4	Branes as cobordism defects	11
	4.1 Compactification to a running solution	13
	4.2 D-branes as Dynamical Cobordisms	14
	4.3 Revisiting the EFT strings	17
	4.4 The Klebanov-Strassler throat	19
5	Small black holes as Dynamical Cobordisms	21
	5.1 The D2/D6 system on \mathbf{T}^4	21
	5.2 Small black holes from the D2/D6 system on $\mathbf{T}^4 \times \mathbf{T}^2$	24
	5.3 General small black holes	25
6	Swampland constraints and surprises from the UV	27
	6.1 Swampland distance conjecture and other constraints	27
	6.2 Large N surprises from the UV	29
7	Conclusions	30
\mathbf{A}	Local Dynamical Cobordisms with curved $(d-1)$ -dimensional slice	s 32
	A.1 General analysis for curved slices	32
	A.2 Witten's bubble of nothing	35
в	Subleading corrections to the local description	36

1 Introduction

The Cobordism Conjecture [1] states that in any consistent theory of quantum gravity all cobordism classes are trivial. In simple terms, it must admit at the topological level a configuration ending spacetime.¹ Such end-of-the-world configuration may correspond to a boundary (such as the 10d Horava-Witten boundary of 11d M-theory [2, 3]), a bubble of nothing in which some compactification space shrinks to zero size [4] (see [5–8] for some recent works), or more exotic possibilities, and may possibly be dressed with charged objects, such as branes, orientifold planes or generalizations (dubbed I-folds in [9]). The cobordism conjecture, already at this topological level, has produced interesting results, see [6, 9–13] for some references.²

An exploration of the Cobordism Conjecture beyond the topological level was undertaken in [21, 22] via the study of spacetime varying solutions to the equations of motion in theories with dynamical tadpoles, namely, potentials which do not have a minimum and thus do not admit maximally symmetric solutions (see [23–26] for early work and [27–30] for related recent developments, and [31, 32] for a complementary approach to cobordism solutions). In the solutions in [21, 22], which we refer to as Dynamical Cobordisms, the fields run along a spatial coordinate until the solution hits a singularity at finite distance in spacetime, which (once resolved in the full UV theory) ends spacetime.

These solutions exhibit sharp features in the region near the singularity. For instance, the scalars go off to infinite distance in moduli (or field) space at the spacetime singularity. Moreover, in the effective field theory description, the field space distance D, the spacetime curvature R and the spacetime distance Δ to the singularity are related by interesting scaling laws, namely (in Planck units)

$$\Delta \sim e^{-\frac{1}{2}\delta D}, \qquad |R| \sim e^{\delta D} \tag{1.1}$$

for suitable positive coefficient δ .

The singularities in these solutions are resolved in the full UV description, in terms of the corresponding cobordism configuration. In string theory examples, the latter often admits a tractable microscopic description involving geometries closing-off spacetime, possibly dressed with defects, as explained above. In this spirit, they were dubbed 'cobordism defects' or 'walls of nothing' in [21, 22]. In this work we will mainly focus on the effective field theory description, where they remain as singular sources, which we refer to as End-of-The-World (ETW) branes.³

The universal form of the scaling relations (1.1) was found by inspecting several explicit examples, but it suggests that a simple universal local description near the ETW branes should be possible in the effective theory. In this paper we provide this local description by studying Dynamical Cobordisms near walls at which the scalars run off to infinite field space

¹Equivalently, any two quantum gravity theories admit, at the topological level, a domain wall connecting them. For this paper we will emphasize the formulation as in the main text.

 $^{^{2}}$ Spacetimes with boundaries have also been considered in the holographic setup, see [14–20] for some recent approaches.

³This follows the nomenclature in some of the references in footnote 2.

distance. In this local description, the solutions simplify dramatically and are characterized in terms of a critical exponent δ , which controls the asymptotic profiles of fields and the scaling relations (1.1) in a very direct way. The analysis does not rely on supersymmetry and can be applied to non-supersymmetric setups.

This provides a powerful universal framework to describe ETW branes within effective field theory, as we illustrate in many different examples. We exploit it to describe Dynamical Cobordisms in several 10d string theories, including non-supersymmetric cases. We also use it to characterize warped throats [33, 34] as Dynamical Cobordisms. We moreover show that the familiar 10d D*p*-brane supergravity solutions can be regarded as Dynamical Cobordisms of sphere compactifications with flux, and are described by our local analysis with the D-branes playing the role of ETW branes. Finally, we argue that 4d small black hole solutions (see [35, 36] for some reviews), including those of the recent work [37], can be similarly regarded as Dynamical Cobordisms of \mathbf{S}^2 compactifications with flux, with the small black hole core playing the role of ETW brane.

Our models provide setups in which scalars explore large field space distances in a dynamical setup (as pioneered in [38], see also [39]), in contrast with the alternative adiabatic approach. Hence our description of Local Dynamical Cobordisms is the natural arena for the dynamical realization of swampland proposals⁴ dealing with infinity in scalar moduli/field space.

The paper is organized as follows. In section 2 we present the general formalism for the local description of Dynamical Cobordisms. In section 2.1 we present the general equations of motion, and in section 2.2 we apply them to describe the local dynamics near ETW branes, and derive the universal scaling relations. In section 3 we apply the local description to several 10d examples, including massive IIA theory in section 3.1 and the non-supersymmetric USp(32) theory of [40] in section 3.2. In section 4 we interpret D-brane supergravity solutions as Dynamical Cobordisms (section 4.1) and express them as ETW branes in the local description (section 4.2). Similar ideas are applied in section 4.3 to the EFT string in 4d $\mathcal{N} = 1$ theories in [41], and in section 4.4 to the Klebanov-Strassler warped throat [33, 34]. In section 5 we discuss small black holes as Dynamical Cobordisms. In section 5.1 we warm up by expressing the supergravity solution of D2/D6-branes on \mathbf{T}^4 as a Dynamical Cobordism, and in section 5.2 we relate it to small black holes via a further \mathbf{T}^2 compactification. In section 5.3 we consider more general small black holes, such as those in [37], and derive scaling relations despite the absence of a proper Einstein frame in 2d. In section 6 we discuss the interplay of Swampland constraints with the results of our local description for the behaviour of several quantities near infinity in field space. In section 6.1 we consider the Distance Conjecture, the de Sitter conjecture and the Transplanckian Censorship Conjecture. In section 6.2 we discuss potentially large backreaction effects when the UV description of the ETW branes involve a large number of degrees of freedom, suggesting mechanisms to generate non-trivial minima near infinity in field space. In section 7 we offer some final thoughts. In appendix A we generalize the ansatz in the main text to allow for non-zero constant curvature in the ETW brane

⁴See [38] for a related viewpoint.

worldvolume directions (section A.1), and apply it to describe Witten's bubble of nothing as a 4d Dynamical Cobordism and provide its local description (section A.2). In appendix B we discuss subleading corrections to the local description, specially relevant in cases where the leading contributions vanish.

2 Local dynamical cobordisms

In this section we formulate our local effective description near End of The World (ETW) branes, in terms of gravity coupled to a scalar field. We would like to emphasize that we consider a general scalar potential, but remarkably derive non-trivial results for its asymptotic behaviour near infinity in field space. The key input is just that the dynamics should allow for the scalar to go off to infinity in field space in a finite spacetime distance.

Interestingly, the scalar potential generically behaves as an exponential near infinity in moduli/field space, suggesting a first-principles derivation of the 'empirical' evidence for such exponential potentials, coming from string theory and other swampland considerations (see [42–44] for reviews). In particular, exponential potentials and constraints on them have been discussed in [31, 32], for the restricted case of bubbles of nothing (i.e. UV completed to a purely geometrical higher dimensional configuration, à la [4]). In contrast, our analysis holds for fully general ETW branes (and hence, allows for more general potentials, including cases without this asymptotic growth).

We focus on the case of a single scalar; however, our discussion also applies to setups with several scalars, by simply combining them into one effective scalar encapsulating the dynamics of the solutions (as illustrated in several of our examples in later sections).

2.1 General ansatz

Consider d-dimensional Einstein gravity coupled to a real scalar⁵ field with a potential,

$$S = \int d^{d}x \sqrt{-g} \left(\frac{1}{2}R - \frac{1}{2} \left(\partial \phi \right)^{2} - V(\phi) \right) \,, \tag{2.1}$$

where we are taking $M_{Pl} = 1$ units. We focus on d > 2, and deal with the d = 2 case in some explicit examples in section 5.

ETW branes define boundaries of the d-dimensional theory, hence they are described as real codimension 1 solutions. We take the ansatz

$$ds^{2} = e^{-2\sigma(y)} ds^{2}_{d-1} + dy^{2},$$

$$\phi = \phi(y),$$
(2.2)

where y parametrizes the coordinate transverse to the ETW brane.

We consider flat metric in the (d-1)-dimensional slices. The corresponding analysis for general non-zero constant curvature, carried out in the same spirit and leading to essentially similar results, is presented in appendix A.

⁵Even though our analysis holds for general potential, we often refer to the scalar as modulus, and its field space as moduli space.

The equations of motion are

$$\phi'' - (d-1)\sigma'\phi' - \partial_{\phi}V = 0, \qquad (2.3)$$

$$\frac{1}{2}(d-1)(d-2)\sigma'^2 + V - \frac{1}{2}\phi'^2 = 0, \qquad (2.4)$$

$$(d-2)\sigma'' - \phi'^2 = 0, \qquad (2.5)$$

where prime denotes derivative with respect to y. The first one is the equation of motion for the scalar; for the Einstein equations, they split into transverse and longitudinal components to the ETW brane, giving two independent equations, subsequently combined into the last two equations.

The analysis of these equations is more amenable in terms of a new quantity, the tunneling potential introduced in [45, 46] (see also [47-52])

$$V_t(\phi) \equiv V(\phi) - \frac{1}{2}{\phi'}^2$$
. (2.6)

Using it to eliminate the scalar from the eoms we get

$$(d-1)\sqrt{2(V-V_t)\sigma' - \partial_{\phi}V_t} = 0, \qquad (2.7)$$

$$\frac{1}{2}(d-1)(d-2)\sigma'^2 + V_t = 0, \qquad (2.8)$$

$$(d-2)\sigma'' - 2(V - V_t) = 0.$$
(2.9)

Finally, combining the first two equations to eliminate σ we get

$$\frac{1}{4}(d-2)\left(\partial_{\phi}V_{t}\right)^{2} + (d-1)\left(V - V_{t}\right)V_{t} = 0.$$
(2.10)

This is a d-dimensional generalization of a condition found in [50] in the context of domain walls.

Now, given a potential $V(\phi)$, one can use this equation to solve for the tunneling potential $V_t(\phi)$, and then use (2.6) and (2.8) to solve for $\phi(y)$ and $\sigma(y)$ respectively. In addition, one should check that (2.9) is also satisfied.

Before moving on, let us comment on the implications that these equations have for the signs of the relevant quantities. From equation (2.8) we learn that $V_t \leq 0$. In addition, from (2.6) we get that $V - V_t \geq 0$. Notice that these two facts are consistent with equation (2.10). Finally, combining the last inequality with (2.9) we learn that $\sigma'' \geq 0$. When solving our system of equations we will systematically pick signs so that these inequalities are satisfied.

A nice way of parametrizing the freedom of choosing the potential is by writing

$$V(\phi) = a(\phi)V_t(\phi), \qquad (2.11)$$

where we have to impose that $a(\phi) \leq 1$ for the reason explained above. Plugging this into (2.10) one can easily get to the solution

$$\log\left(\frac{V_t}{V_t^0}\right) = \pm 2\sqrt{\frac{d-1}{d-2}} \int_{\phi_0}^{\phi} \sqrt{1-a\left(\tilde{\phi}\right)} d\tilde{\phi}, \qquad (2.12)$$

where we are taking $V_t^0 \equiv V_t(\phi_0)$ as boundary condition.

2.2 Local description of end of the world branes

As explained in the introduction, we are interested in solutions for which the scalar attains infinity in field space i.e. $\phi \to \pm \infty$ at a point at finite distance in spacetime, defining an ETW brane. Without loss of generality we take this boundary to be y = 0, and the infinity in field space as $\phi \to \infty$.

From (2.12), it is clear that the asymptotic behavior as $y \to 0$, $\phi \to \infty$ is controlled by the asymptotic profile of $a(\phi)$. We know from the previous section that $a(\phi) \leq 1$ and we restrict our analysis to the cases where $a(\phi)$ has a well-defined and constant limit a < 1as $\phi \to \infty$ (we briefly remark on the behavior $a \to 1$ below (2.15)). Indeed, although one can cook up potentials realizing other possibilities, we have not encountered them in any of the string theory examples in later sections. We therefore ignore other possibilities in what follows, leaving for future work the question about the consistency of such behaviors from the viewpoint of UV completions. Note that the constraint a < 1 includes a = 0, which corresponds to solutions with potential negligible with respect to the kinetic energy for the scalar (at least asymptotically).

Taking constant a, (2.12) gives

$$V_t(\phi) \simeq -c \, e^{\delta \, \phi} \,, \tag{2.13}$$

where c > 0 is related to the boundary condition used before. As explained in appendix B, we also allow c to hide some ϕ -dependence, corresponding to subleading corrections. The leading behaviour is an exponential controlled by the critical exponent δ , given by

$$\delta = 2\sqrt{\frac{d-1}{d-2}(1-a)} \,. \tag{2.14}$$

Here we choose the plus sign for δ . As we will see later this will imply that ETW brane explores $\phi \to \infty$ as explained above.

The critical exponent δ controls the structure of the local solution, in particular the asymptotic profile of fields as $y \to 0$, and the scaling relations among different physical local quantities.

Recall that the freedom of choosing a potential is parametrized by a. It is then interesting to ask how the potential itself looks like when approaching the end of the world. Plugging (2.13) into (2.11) we find

$$V(\phi) \simeq -a c e^{\delta \phi} \,. \tag{2.15}$$

Note that we get an exponential dependence, for any value of a < 1. As a side-note, for a = 1, the potential V may take different forms e.g. power-like, growing strictly slower than exponentials.

Also notice that, since c > 0, the sign of the potential is completely determined by that of a. Moreover, using the relation between a and the critical exponent δ in (2.14), we can put bounds on the latter depending on the sign of the potential. Namely, for V > 0 we must have a < 0, which implies $\delta > 2\sqrt{\frac{d-1}{d-2}}$, while if V < 0 then 0 < a < 1, yielding $\delta < 2\sqrt{\frac{d-1}{d-2}}$. We thus neither have negative potentials whose exponential behaviour is arbitrarily strong, nor positive potentials whose exponential behaviour is arbitrarily mild. The explanation is that such exponentials would lead to $\phi'^2 \gg V$ as we approach the ETW brane, and therefore they correspond to the a = 0 case of our analysis.

It is interesting that we have derived fairly generically an exponential shape of the potential near infinity in moduli space, from the requirement that the theory contains ETW branes, namely configurations reaching infinity in moduli space at finite spacetime distance. In section 6.1 we will study its interplay with a variety of swampland constraints on scalar potentials. We note however that theories with milder growth of the potential (most prominently, theories with vanishing potential and exact moduli spaces) are still included in the analysis, and correspond to $a(\phi) \rightarrow 0$. The corresponding statement that $V \rightarrow 0$ in this case actually means that the theory can have any potential as long as it grows slower than ϕ'^2 .

From (2.6) we can obtain the asymptotic profile of ϕ as $y \to 0$

$$\phi(y) \simeq -\frac{2}{\delta} \log\left(\frac{\delta^2}{4} \sqrt{2c \frac{d-2}{d-1}} y\right) .$$
(2.16)

Here we are ignoring an additive integration constant, irrelevant in the $\phi \to \infty$ limit. We have also fixed another integration constant by demanding that the function blows up for $y \to 0$. The leading term as $y \to 0$ is

$$\phi(y) \simeq -\frac{2}{\delta} \log y \,. \tag{2.17}$$

Hence the scalar goes off to infinity as we approach the end of the world. This motivates the appearance of a lowered cutoff as we approach the wall, above which a more complete microscopic description simply resolves the singularity; this resonates with the swampland distance conjecture, as we discuss in section 6.1.

Plugging (2.16) into (2.8) we can also solve for $\sigma(y)$. The final result is

$$\sigma(y) \simeq \pm \frac{4}{(d-2)\delta^2} \log y \,. \tag{2.18}$$

Here we ignore an integration constant which can be reabsorbed by a change of coordinates. Note that, to comply with (2.9), we only need to pick the minus sign.

Furthermore, the d-dimensional scalar curvature is given by

$$R = (d-1)\left(2\sigma'' - d\sigma'^{2}\right) \sim \frac{1}{y^{2}}.$$
(2.19)

We thus recover that the curvature blows up as we approach the end of the world, leading to a naked singularity in the effective field theory description.

Notice that we have ignored a prefactor that, interestingly, vanishes for the special case $\delta^2 = \frac{2d}{d-2}$. For that value one should consider the next-to-leading order term in the $y \to 0$ expansion. In what follows we ignore this case and keep the generic one.

Since the scalar ϕ is normalized canonically, the field space distance D as $y \to 0$ is (2.17). Also, the distance in spacetime to the singularity is given by y. Hence from (2.17)

and (2.19) we obtain the universal relations

$$\Delta \sim e^{-\frac{\delta}{2}D}, \qquad |R| \sim e^{\delta D}. \tag{2.20}$$

The solutions provides a simple universal description of dynamical cobordism in terms of the effective field theory. The microscopic description of the cobordism defect is available only in the UV complete theory, and is thus model-dependent (but known in many cases, see our explicit examples in later sections). From our present perspective, the only microscopic information we need is the very existence of such defects, guaranteed by the swampland cobordism conjecture [1].⁶ It is thus remarkable that, the simple requirement that scalars go to infinity at finite spacetime distance leads to a complete local description of the EFT behaviour near a dynamical cobordism. Moreover, it constrains the structure of the theory, in particular it naturally yields an exponential behavior of the scalar potential near infinity in field space.

The above local description can be used to prove a general relation, introduced in [21], between the dynamical tadpole (defined as the derivative of the potential $\mathcal{T} = \partial_{\phi} V(\phi)$) at a given point and the spacetime distance Δ to the ETW brane, which in our examples is given by

$$\Delta \sim (\mathcal{T})^{-\frac{1}{2}} . \tag{2.21}$$

Indeed, using (2.15) and (2.17), we obtain \mathcal{T} evaluated at a point y^* :

$$\mathcal{T}|_{y=y^*} = \partial_{\phi} V|_{y=y^*} = -a \ c \ \delta \ e^{\delta\phi}|_{y=y^*} = -a \ c \ \delta \ (y^*)^{-2} \,, \tag{2.22}$$

 Δ is constructed as the distance from a point y^* to the singularity at y = 0, we therefore have $\Delta = y^*$. We hence have a general relation⁷

$$\Delta = \left(\frac{-\mathcal{T}}{a\ c\ \delta}\right)^{-\frac{1}{2}} \sim (\mathcal{T})^{-\frac{1}{2}} \ . \tag{2.23}$$

This relation places a bound on the spacetime extent of a solution whose running is induced by a dynamical tadpole, as emphasized in [21, 22], due the dynamical appearance of an end of spacetime. We would nevertheless mention that there exist solutions with spacetime boundaries even in situations with no dynamical tadpole. The simplest example is Horava-Witten theory, which corresponds to M-theory on an interval with two boundaries. Even in our present context of scalars running off to infinity at finite spacetime distance, it is possible to find ETW branes in cases with vanishing potential V = 0 (or asymptotically negligible potentials, a = 0).

 $^{^{6}}$ To be more precise, there are theories in which the cobordism higher-form symmetry is gauged, rather than broken by the existence of the defects. In such cases, the gauging imposes the constraint that the total charge cancels in the configuration; our analysis applies to those cases as well, with the ETW brane corresponding to a mere ending of spacetime with no explicit charged defects, similar to a bubble of nothing, see appendix A.2.

⁷For the particular case of the warped throat in 4.4 this corrects the statement in [21].

3 Some 10d examples

In this section we consider examples of 10d theories with Dynamical Cobordism solutions in [21, 22], and use the above local description to easily derive their structure. The results nicely match the asymptotic behavior of the complete solutions in the literature.

3.1 The 10d massive type IIA theory

We consider the 10d massive type IIA theory. The effective action in the Einstein frame for the relevant fields is

$$S_{10,E} = \frac{1}{2\kappa^2} \int d^{10}x \sqrt{-g} \left\{ R - (\partial\phi)^2 - \frac{1}{2}e^{\frac{5}{2}\sqrt{2}\phi}F_0^2 - \frac{1}{2}e^{\frac{\sqrt{2}}{2}\phi}|F_4|^2 \right\},$$
(3.1)

where F_0 denotes the Romans mass parameter. The $\sqrt{2}$ factors in the exponents ensure that the normalization of the scalar agrees with our conventions.

This theory has a potential

$$V = \frac{1}{2} e^{\frac{5}{\sqrt{2}}\phi} F_0^2, \qquad (3.2)$$

hence it does not admit 10d maximally symmetric solutions. On the other hand there are 9d Poincaré invariant (and in fact 1/2 supersymmetric) running solutions of the equations of motion in which the dilaton (and other fields) depend on a space coordinate, e.g. x^9 . The metric and dilaton profile read

$$ds_{10}^{2} = Z \left(x^{9}\right)^{1/12} \eta_{\mu\nu} dx^{\mu} dx^{\nu} ,$$

$$e^{\sqrt{2}\phi} = Z \left(x^{9}\right)^{-5/6} ,$$
(3.3)

where the coordinate function is $Z(x^9) = -F_0 x^9$. This solution hits a singularity at $x^9 = 0$, which was proposed to correspond to an end of the world brane in [21, 22]. In the microscopic theory, it corresponds to an O8-plane (possibly with D8-branes), as in one of the boundaries of the interval of type I' theory [53].

In the following we show how the local structure of the Dynamical Cobordism can be obtained from the analysis in the previous section.

The only input of the local analysis is the potential (3.2). Matching it with the local analysis expression (2.15), we obtain the following values for δ and, using (2.14) for a:

$$\delta = \frac{5}{\sqrt{2}}, \qquad a = -\frac{16}{9}.$$
(3.4)

Plugging this into (2.17) we obtain the dilaton profile

$$\phi \simeq -\frac{2\sqrt{2}}{5}\log y. \tag{3.5}$$

We can now obtain the profile for σ (2.18)

$$\sigma \simeq -\frac{1}{25}\log y \,, \tag{3.6}$$

which determines the metric via (2.2). As usual, the local description predicts the scalings

$$\Delta \sim e^{-\frac{5}{2\sqrt{2}}D}, \quad |R| \sim e^{\frac{5}{\sqrt{2}}D}.$$
 (3.7)

These results from the local analysis are in agreement with the scaling relations obtained in the paper [21] from the complete solution. In fact, this can be done very easily from (3.3), by a change of coordinates

$$y = \int_0^{x^9} \left(-F_0 \tilde{x}^9 \right)^{1/24} d\tilde{x}^9, \tag{3.8}$$

in terms of which the solution acquires the form of (2.2)

$$ds_{10}^{2} = \left[\frac{25}{24} \left(-F_{0}\right)y\right]^{2/25} ds_{9}^{2} + dy^{2},$$
$$e^{\sqrt{2}\phi} = \left[\frac{25}{24} \left(-F_{0}\right)y\right]^{-4/5}.$$

This indeed corresponds to profiles for σ (via (2.2)) and ϕ in agreement with (3.6) and (3.5) respectively.

3.2 The 10d non-supersymmetric USp(32) string

Let us consider a second example in the same spirit, but in the absence of supersymmetry. We consider the 10d non-supersymmetric USp(32) theory, built in [40] as a type IIB orientifold with a positively charged O9-plane and 32 anti-D9-branes. The 10d Einstein frame action for the relevant fields is

$$S_E = \frac{1}{2\kappa^2} \int d^{10}x \sqrt{-G} \left\{ R - (\partial\phi)^2 \right\} - T_9^E \int d^{10}x \sqrt{-G} \, 64 \, e^{\frac{3}{\sqrt{2}}\phi} \,. \tag{3.9}$$

We have introduced factors of $\sqrt{2}$ relative to the conventions in [40], to normalize the scalar as in previous sections.

This theory has a dilaton tadpole, due to the uncanceled NSNS tadpoles, and hence does not admit maximally symmetric 10d solution. On the other hand, there are 9d Poincaré invariant running solutions of its equations of motion [23], given by

$$ds_{E}^{2} = |\sqrt{\alpha_{E}}r|^{\frac{1}{9}} e^{-\frac{\alpha_{E}r^{2}}{8}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + |\sqrt{\alpha_{E}}r|^{-1} e^{-\frac{3\phi_{0}}{\sqrt{2}}} e^{-\frac{9\alpha_{E}r^{2}}{8}} dr^{2},$$

$$\phi = \frac{3}{4\sqrt{2}} \alpha_{E}r^{2} + \frac{\sqrt{2}}{3} \log|\sqrt{\alpha_{E}}r| + \phi_{0},$$
 (3.10)

where $\alpha_E = 64\kappa^2 T_9^E$, and ϕ_0 is a reference value for the dilaton. The coordinate r was denoted by y in [23] but here, we preserve y for the coordinate of the local analysis near end of the world branes.

The solution hits two singularities, at $r \to 0$ and at $r \to +\infty$, which are at finite spacetime distance, yet the scalar attains infinity in fields space ($\phi \to -\infty$ at $r \to 0$, and $\phi \to \infty$ at $r \to \infty$, respectively). As discussed in [21, 22], it thus describes a Dynamical Cobordism with two end of the world branes. The existence of two boundaries, and hence a finite size spacetime coordinate, arises in this example, but is not a general feature of running solutions, as we have seen in previous sections. It would be interesting to understand a general criterion discriminating between the two possibilities, but we leave this question for future work. In any event, even in setups with two boundaries, our local analysis applies to each of them individually, as we discuss next. Indeed, let us now exploit the local analysis to display the scalings near these walls, with the scalar potential in (3.9) as sole input.

$3.2.1 \quad r \rightarrow 0$

From equation (3.10), we see that $r \to 0$ corresponds to the limit $\phi \to -\infty$. The potential in (3.9) vanishes in that limit. As a consequence, we have an ETW brane in which the potential becomes negligible, i.e., the critical exponents for the local model are

$$\delta = \frac{3}{\sqrt{2}}, \qquad a = 0. \tag{3.11}$$

The local analysis then leads to the dilaton and radion profiles

$$\phi \simeq \frac{2\sqrt{2}}{3}\log y$$
, $\sigma \simeq -\frac{1}{9}\log y$. (3.12)

Note that we have chosen the sign of $\phi \to -\infty$ as $y \to 0$.

These results allow to obtain the universal scalings for the curvature and spacetime distance with the field space distance (2.20), namely

$$\Delta \sim e^{-\frac{3}{2\sqrt{2}}D}, \qquad |R| \sim e^{\frac{3}{\sqrt{2}}D}.$$
 (3.13)

It is easy to check that the above profiles and scaling reproduce the behaviour of the complete solution (3.10). This can be shown by the following coordinate change to bring it into the ansatz (2.2):

$$y = \int \sqrt{|\sqrt{\alpha_E}r|^{-1}e^{-\frac{3\phi_0}{\sqrt{2}}}e^{-\frac{9\alpha_E r^2}{8}}} dr \sim \left[\Gamma\left(\frac{1}{4}, \frac{9\alpha_E}{16}r^2\right) - \Gamma\left(\frac{1}{4}, 0\right)\right] \sim \sqrt{r}.$$
 (3.14)

In the last step we have taken the leading behaviour as $r \to 0$. By also taking the leading behaviour in (3.10), plugging in y, and reading off σ as it appears in (2.2) we finally recover the profiles predicted by the local analysis in (3.12).

3.2.2 $r o \infty$

This should be described by a local model where $\phi \to +\infty$ at $y \to 0$, i.e. the origin of a new local coordinate (which corresponds to $r \to \infty$). In this case the potential in (3.9) is blowing up, hence via (2.15) and (2.14), we get $\delta = 3/\sqrt{2}$, a = 0, just as in (3.11). The result a = 0 may seem puzzling, since from (2.15) this would seem to imply $V \to 0$. However, one should recall that in the local description a = 0 simply means that $V \ll \phi'^2$. Indeed, it may happen that c blows up as $\phi \to \infty$ in such a way that it compensates having $a \to 0$ in this same limit. We will explicitly check this later on.

The dilaton and radion profiles read

$$\phi \simeq -\frac{2\sqrt{2}}{3}\log y, \qquad \sigma \simeq -\frac{1}{9}\log y.$$
 (3.15)

The dilaton sign differs from (3.12) in order to have $\phi \to +\infty$ as $y \to 0$. We also recover the scalings for Δ and R with D, which are again given by (3.13).

Let us now show that the above local model indeed reproduces the $r \to \infty$ regime of (3.10). The required change of variables is now

$$y = \int_{r}^{\infty} |\sqrt{\alpha_E}\tilde{r}|^{-1/2} e^{-\frac{3}{4}\phi_0} e^{-\frac{9\alpha_E r^2}{16}} d\tilde{r} \sim \Gamma\left(\frac{1}{4}, \frac{9\alpha_E}{16}r^2\right) \sim r^{-\frac{3}{2}} e^{-\frac{9\alpha_E}{16}r^2}.$$
 (3.16)

The integration limits are chosen so that the finite distance singularity at $r \to \infty$ is located at the origin for the new coordinate. In the last step we have taken the leading behaviour of the Gamma function as $r \to \infty$.

Taking the logarithm of this expression and keeping the leading behaviour we get

$$\log y \simeq -\frac{9\alpha_E}{16}r^2. \tag{3.17}$$

Finally, by also taking the leading behaviour in (3.10), reading off σ as it appears in (2.2) and plugging in our previous expression for y, we recover the profiles anticipated by the local analysis in (3.15).

Let us now come back to the issue of having a = 0 while not having vanishing potential. First, let us check that indeed $\phi'^2/V \to \infty$ as we approach the ETW brane. We can compute it, with no approximations, as

$$\frac{\phi'^2}{V} \sim \left(\frac{3\alpha_E}{2\sqrt{2}}r + \frac{\sqrt{2}}{3}\frac{1}{r}\right)^2,$$
(3.18)

where we are ignoring irrelevant numerical prefactors. Importantly, for this computation one has to remember that ϕ' is the derivative with respect to y, not with respect to r. As advanced, we find that this blows up to infinity in both $r \to 0$ and $r \to \infty$ limits.

Moreover, using this result one can compute the tunneling potential as $\phi \to \infty$ as

$$V_t \simeq \frac{\phi'^2}{2} \sim r^2 V \sim \phi \, e^{\frac{3}{\sqrt{2}}\phi} \sim e^{\frac{3}{\sqrt{2}}\phi + \log \phi} \,, \tag{3.19}$$

where we have plugged in the value of V from (3.9) and $r^2 \sim \phi$ from the $r \to \infty$ limit of $\phi(r)$ in (3.10). As advertised, we find a case in which the coefficient c in (2.13) blows up as we approach the wall of nothing. This is consistent with our local analysis because, as we see in the last equality, c does not blow up faster than the exponential, i.e., it gives subleading corrections to log V_t (see appendix B for more details).

4 Branes as cobordism defects

The local analysis of section 2 provides a general framework to describe effective ETW branes, encapsulating Dynamical Cobordisms of the underlying theory. An interesting observation is that, in compactified theories with fluxes, the cobordism requires the introduction of charged objects. Namely, those required to break the corresponding cobordism charge to avoid a global symmetry, which should be absent in Quantum Gravity. A typical example



Figure 1. D*p*-branes as cobordism defect in theories with (8-p) compact dimensions from the higher and lower dimensional perspective. Our local (p+2)-dimensional description of the \mathbf{S}^{8-p} -truncation corresponds to the local structure of a Dynamical Cobordism of a more general compactification on \mathbf{X}_{8-p} .

is the introduction of NS5- and D-branes in bubbles of nothing in compactifications with NSNS and RR fluxes (see [6] for a recent discussion on bubbles of nothing).

Therefore it is interesting to explore the description of such objects in the local picture of section 2. As a simple illustrative setup, in this section we describe the geometry around a stack of D*p*-branes in the language of the local analysis of section 2. In local terms, it corresponds to regarding the D*p*-brane supergravity solution as a compactification of the 10d theory on \mathbf{S}^{8-p} with flux, yielding a d = (p+2)-dimensional running solution along one of the coordinates (morally the radial coordinate), which has finite extent and end on an effective ETW brane. The microscopic description of the latter is actually given by the D*p*-brane in the UV.

The above idea generalizes the description in [22] of the EFT strings solutions in [41] as cobordism defects of \mathbf{S}^1 compactifications of the underlying 4d $\mathcal{N} = 1$ theory with axion flux along the \mathbf{S}^1 .

We note that the compactification of the 10d theory on the \mathbf{S}^{8-p} around a D*p*-brane actually corresponds to a truncation onto the SO(9-*p*)-invariant sector. Sphere truncations have long been studied in the literature, in particular in the holographic context, see [54] for a discussion for D*p*-brane solutions. However, in our context we should regard the sphere truncation as a fair local description of Dynamical Cobordisms in actual compactifications, including those with scale separation, allowing for a more physical notion of lower-dimensional effective theory. Our local analysis should be regarded as part of the latter. This is depicted in figure 1, and is illustrated quantitatively in a similar example for Witten's bubble of nothing in appendix A.2. Finally, although we phrase our discussion in terms of D*p*-branes, notice that other string theory branes admit similar analysis; in fact, the NS5-brane is essentially the same as the D5-brane, since we are working in the Einstein frame, in which S-duality acts manifestly.

4.1 Compactification to a running solution

Let us begin with a precise description of the general procedure of compactifying a codimension (n+1) brane-like solution in d+n dimensions down to a running solution (codimension 1) in d dimensions. In next sections, we will apply this reasoning to the D*p*-branes as cobordism defects of \mathbf{S}^{8-p} compactifications.

Take the general metric of a codimension n object in d + n-dimensions:

$$ds^{2} = e^{-2\mu(r)} ds^{2}_{d-1} + e^{2\nu(r)} \left(dr^{2} + r^{2} d\Omega^{2}_{n} \right) .$$
(4.1)

The directions in ds_{d-1}^2 span the worldvolume of the object, while we have split the transverse directions into radial and angular ones.

We want to perform an \mathbf{S}^n truncation to look at this solution from the *d*-dimensional perspective. We thus take the compactification ansatz

$$ds^{2} = e^{-2\alpha\omega(r)}ds_{d}^{2} + e^{2\beta\omega(r)}r_{0}^{2}ds_{n}^{2}, \qquad (4.2)$$

where r_0 is a reference scale. By requiring that the *d*-dimensional action is in the Einstein frame and has canonically normalized kinetic term for the radion ω we get the following constraints for α and β :

$$\gamma \equiv \frac{\alpha}{\beta} = \frac{n}{d-2} \qquad \beta^2 = \frac{d-2}{n(d+n-2)}.$$
(4.3)

The first one implements the Einstein frame requirement, while in the second one we already apply both conditions. Note that for d = 2 we recover the familiar statement that there is no Einstein gravity in 2 dimensions. We will deal with reductions to 2d in section 5, and consider d > 2 in what follows.

By matching the compactification ansatz (4.2) with the metric in (4.1) we obtain the profile for the radion

$$e^{2\beta\omega(r)} = e^{2\nu(r)} \left(\frac{r}{r_0}\right)^2, \qquad (4.4)$$

as well as the lower-dimensional metric

$$ds_d^2 = e^{2\alpha\omega(r)} \left(e^{-2\mu(r)} ds_{d-1}^2 + e^{2\nu(r)} dr^2 \right) \,. \tag{4.5}$$

In order to put solutions in the general form (2.2) used for the local description in section 2, we introduce a new coordinate

$$y = \int e^{\alpha \omega(r)} e^{\nu(r)} dr , \qquad (4.6)$$

in terms of which we can borrow the results (2.15)-(2.20) from the local description.

From the viewpoint of the *d*-dimensional theory, there is a non-trivial potential arising from the curvature of the \mathbf{S}^n , and possibly other sources (such as fluxes, etc). Generically the total potential does not have a minimum, hence the running solutions can be regarded as induced by a dynamical tadpole. Applying the results in [21, 22], the *d*-dimensional solution must describe a Dynamical Cobordism ending on an ETW brane, to which we can apply the local analysis in section 2.

Note that however there are cases with a non-trivial minimum. A prominent example is the \mathbf{S}^5 compactification with a large number N of RR 5-form field string flux units (see [55] for a discussion in similar terms). The minimum corresponds to a setup with no tadpole, and admits a maximally symmetric solution, namely the celebrated $\operatorname{AdS}_5 \times \mathbf{S}^5$. Because of this, we will not consider the D3-branes in our discussion, and focus on genuinely running solutions.

4.2 D-branes as Dynamical Cobordisms

In this section we regard the 10d D*p*-brane solutions as \mathbf{S}^{8-p} compactifications and reexpress them in terms of the local description of ETW branes of the (p+2)-dimensional theory of section 2. Note that, in contrast with section 3, we do not intend to *derive* the local solutions from a (p+2)-dimensional scalar potential; rather we take the familiar 10d solutions and express their near brane asymptotics as local (p+2)-dimensional ETW brane solutions.

Consider the D*p*-brane solution in the 10d Einstein frame, with $0 \le p \le 8$. The 10d metric and dilaton profile take the form

$$ds_{10}^2 = Z\left(r\right)^{\frac{p-7}{8}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z\left(r\right)^{\frac{p+1}{8}} \left(dr^2 + r^2 d\Omega_{8-p}^2\right), \qquad (4.7)$$

$$\Phi = \frac{(3-p)}{4\sqrt{2}} \log \left(Z(r) \right) \,, \tag{4.8}$$

where the warp factor is given by the harmonic functions

$$Z(r) = 1 + \left(\frac{\rho}{r}\right)^{7-p} \quad \text{for } 0 \le p \le 6,$$
(4.9)

$$Z(r) = 1 - \frac{N}{2\pi} \log\left(\frac{r}{\rho}\right) \qquad \text{for } p = 7, \qquad (4.10)$$

$$Z(r) = 1 - \frac{|r|}{\rho}$$
 for $p = 8$. (4.11)

Here $\rho > 0$ is a length scale. For the cases $p \neq 7$ it depends on the number of D*p*-branes, N, while for p = 7 this dependence does not enter in ρ but has been made explicit in the solution.

As we have explained, these formulas should be regarded as the local description near the D-branes in possibly more general compactifications, namely the above Z(r) should be though of as local expansions around the D-brane location of the warp factor in more general compactification spaces, cf. figure 1.

We immediately see that for $p \neq 3$, the dilaton reaches infinite values near the point r = 0, the core of the D*p*-brane. As explained above, we do not consider the case p = 3,

since it relates to AdS minimum of the theory. The solution does not run towards an ETW brane but towards a minimum in the potential. Similarly, for p = 8, the dilaton reaches finite values at r = 0. This fits with the identification of D8-branes as interpolating walls instead of walls of nothing in [22]. In the following we restrict to $p \neq 3$ and $1 \le p \le 7$, the lower bound to avoid reduction to 2d (postponed until section 5), and the upper bound to have non-trivial sphere compactification.

The D*p*-brane is a solution of the following generic type II theory with a dilaton and RR field:

$$S_{10} \sim \frac{1}{2} \int d^{10}x \sqrt{-g_{10}} \left\{ R_{10} - (\partial \Phi)^2 - \frac{1}{2n!} e^{a\Phi} |F_n|^2 \right\} \,. \tag{4.12}$$

where n = 8 - p. This 10d theory does not have a scalar potential. However, once compactified on \mathbf{S}^{8-p} with N units of F_{8-p} flux, the curvature of the sphere as well as the flux itself will generate dynamical tadpoles for the ensuing radion and (p + 2)dimensional dilaton. Indeed, let us perform this compactification explicitly and show that we find ourselves in an end-of-the-world scenario, reproducing the associated scaling relations of [22].

Taking a compactification ansatz of the form (4.2) we obtain the d = (p+2)-dimensional Einstein frame metric:

$$ds_d^2 = \left(\frac{r^2}{r_0^2} Z(r)^{\frac{p+1}{8}}\right)^{\frac{8-p}{p}} \left\{ Z(r)^{\frac{p-7}{8}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z(r)^{\frac{p+1}{8}} dr^2 \right\},$$
(4.13)

where the Greek indices correspond to directions along the world volume of the *p*-brane. The (p+2)-dimensional dilaton inherits the same profile as the original one and one obtains the radion's profile through matching:

$$e^{2\beta\omega(r)} = \frac{r^2}{r_0^2} Z(r)^{\frac{p+1}{8}}.$$
(4.14)

The radion is canonically normalized if $\beta^2 = \frac{p}{8(8-p)}$.

The solution has a spacetime singularity at r = 0, at which both the dilaton and radion blow up. We can now compute the relevant scaling quantities, namely the spacetime distance Δ_d to the singularity, the curvature scalar $|R_d|$ near the singularity, and the distance Dtraversed in field space. For the former two we obtain:

$$\Delta_d \sim \begin{cases} r^{\frac{(p-3)^2}{2p}} & \text{for } p \in [1,6] \text{ and } p \neq 3, \\ r^{8/7} & \text{for } p = 7. \end{cases}$$
(4.15)

$$|R_d| \sim \begin{cases} r^{-\frac{(p-3)^2}{p}} & \text{for } p \in [1,6] \text{ and } p \neq 3, \\ r^{-16/7} & \text{for } p = 7. \end{cases}$$
(4.16)

For the field space distance near the singularity, we obtain the following by plugging in the profiles of the radion (4.14) and dilaton (4.7):

$$D(r) = \int (d\omega^2 + d\Phi^2)^{1/2} dr \simeq \begin{cases} -\frac{|3-p|}{2} \sqrt{\frac{9-p}{p}} \log r & \text{for } p \in [1,6] \text{ and } p \neq 3, \\ -\frac{4}{\sqrt{14}} \log r & \text{for } p = 7. \end{cases}$$
(4.17)

The solution thus describes Dynamical Cobordisms with the following scaling relations:

$$\Delta_d \sim e^{\frac{|p-3|}{p}\sqrt{\frac{p}{9-p}}D}, \qquad |R_d| \sim e^{\frac{2|p-3|}{p}\sqrt{\frac{p}{9-p}}D} \qquad \text{for } p \in [1,6] \text{ and } p \neq 3,$$
(4.18)

and

$$\Delta_9 \sim e^{-\frac{2\sqrt{14}}{7}D}, \qquad |R_9| \sim e^{\frac{4\sqrt{14}}{7}D} \qquad \text{for } p = 7.$$
 (4.19)

This shows that Dp-brane are cobordism defects, which reduced on the surrounding \mathbf{S}^{8-p} can be described as ETW branes. In the following we describe their structure in terms of the local description of section 2.2. This will allow us a much simpler computation of the above scaling relations.

The objective is to put the d-dimensional metric in domain-wall form (2.2). In the notation of section 4.1, one obtains:

$$\sigma(r) = -\alpha\omega(r) + \mu(r) = -\frac{8-p}{p}\log\left(Z(r)^{\frac{p+1}{16}}\left(\frac{r}{r_0}\right)\right) - \frac{p-7}{16}\log\left(Z(r)\right) \,. \tag{4.20}$$

The new coordinate y is obtained as

$$y = \int^{r} e^{\alpha \omega(r)} e^{\nu(r)} dr = \int^{r} Z(r)^{\frac{p+1}{2p}} \left(\frac{r}{r_0}\right)^{\frac{8-p}{p}} dr.$$
(4.21)

For a general D*p*-brane with $p \neq 3, 7$, in the limit $r \rightarrow 0$ we have

$$y = \int^{r} \left(\frac{\rho}{r}\right)^{(7-p)\frac{p+1}{2p}} \left(\frac{r}{r_{0}}\right)^{\frac{8-p}{p}} dr \sim r^{\frac{(p-3)^{2}}{2p}}.$$
(4.22)

Using equation (4.20), this yields

$$\sigma(r) \simeq \sigma\left(y^{\frac{2p}{(p-3)^2}}\right) \simeq -\frac{(9-p)}{(p-3)^2}\log y.$$

$$(4.23)$$

We may compare this to the profile for σ put forward by the local description described in section 2.2:

$$\sigma(y) \simeq -\frac{4}{p \,\delta^2} \log y \,. \tag{4.24}$$

We can thus extract the value of δ and, for completeness, that of a:

$$\delta^2 = \frac{4(p-3)^2}{p(9-p)}, \qquad 1-a = -\frac{(p-3)^2}{(p-9)(p+1)}. \tag{4.25}$$

Thus, we have, from equation (2.17):

$$D(y) \simeq -\frac{2}{\delta} \log y \simeq -\frac{|p-3|\sqrt{9-p}}{2\sqrt{p}} \log r.$$

$$(4.26)$$

We have thus recovered exactly the profile (4.17), without having to use the explicit scalar profile. From (2.20), we also recover the scaling relations (4.18), namely:

$$\Delta_d = y \sim e^{-\frac{|p-3|}{p}\sqrt{\frac{p}{9-p}}D}, \qquad |R_d| \sim e^{2\frac{|p-3|}{p}\sqrt{\frac{p}{9-p}}D}.$$
(4.27)

Hence, in this case we have used the local description to recover the field-space distance and scaling relations near the singularity without knowing the full details of the *d*-dimensional theory. In fact, we can use the local description to derive the asymptotic behaviour of interesting *d*-dimensional quantities. For instance, the scalar potential scales near the singularity as (2.15):

$$V(D) = -c \left(1 - \frac{(p-3)^2}{(9-p)(p+1)}\right) e^{\frac{2(p-3)}{\sqrt{p(9-p)}}D}.$$
(4.28)

This is a very interesting bottom-up approach. In the actual d-dimensional action, the potential would depend on the radion and dilaton with contributions from the curvature of the sphere and the flux traversing it. However, the local description encapsulates only the dependence on the effective scalar dominating the field distance D near the ETW brane, erasing any other irrelevant UV information. From the previous equation we find that the potential is negative as we approach the ETW brane (recall c > 0). With the extra input that the curvature and the flux contributions to the potential are negative and positive respectively, the local description is then telling us that it is the curvature term the one that dominates in this limit.

For the D7-brane, the coordinate y is given by

$$y = \int^r e^{\alpha\omega(r)} e^{\nu(r)} dr = \int^r \left(-\frac{N}{2\pi} \log\left(\frac{r}{\rho}\right)\right)^{\frac{4}{7}} \left(\frac{r}{r_0}\right)^{\frac{1}{7}} dr \sim r^{\frac{8}{7}}, \qquad (4.29)$$

where we have neglected the logarithmic contribution compared to the polynomial one. Similarly, we have:

$$\sigma(r) \simeq \sigma\left(y^{\frac{7}{8}}\right) \simeq -\alpha\omega\left(y^{\frac{7}{8}}\right) \simeq -\frac{1}{8}\log y.$$
(4.30)

Hence, comparing this to equation (2.18), we find:

$$\delta^2 = \frac{32}{7}, \qquad a = 0.$$
(4.31)

This means that the asymptotic potential vanishes, in the sense of $\phi'^2 \gg V$. Plugging this value of δ^2 into equation (2.17) and (2.20), we recover the same field space distance and scaling relations as in the computations of the previous section:

$$D(y) \simeq -\sqrt{\frac{7}{8}} \log y \simeq -\frac{4}{\sqrt{14}} \log r$$
, (4.32)

$$\Delta_9 = y \sim e^{-\sqrt{\frac{8}{7}}D}, \qquad |R_9| \sim e^{2\sqrt{\frac{8}{7}}D}.$$
(4.33)

4.3 Revisiting the EFT strings

In [41, 56] it was proposed that in 4d $\mathcal{N} = 1$ theories the limits in which saxionic scalars go to infinity in moduli space can be studied as radial flows in 4d supersymmetric EFT string solutions magnetically charged under the corresponding axionic partners. In [22] the result was recovered by considering running solution of the compactification of the theory to 3d with axion fluxes along the \mathbf{S}^1 : the solutions implement a Dynamical Cobordism ending spacetime along the running direction, and the EFT string arises as the cobordism defect required to get rid of the axion flux. In this section we revisit the analysis in [22] from the local description, with the EFT string becoming an ETW brane. As expected, the analysis is fairly similar to the 10d D7-brane example in the previous section; indeed, upon compactification of the 10d theory on a CY3, the wrapped D7-branes turns into the simplest avatar of the EFT strings in [41, 56].

In the 4d EFT string solution [41, 56], the profile for the scalars is given by

$$s(r) = s_0 - \frac{q}{2\pi} \log \frac{r}{r_0}, \qquad (4.34)$$

$$a(\theta) = a_0 + \frac{\theta}{2\pi}q. \qquad (4.35)$$

In our 3d interpretation, equation (4.35) describes the axionic flux over the S^1 , and equation (4.34) solves the dynamical tadpole for the saxion.

The 4d metric takes the form

$$ds_4^2 = -dt^2 + dx^2 + e^{2D} dz d\bar{z}, \qquad (4.36)$$

with $z = re^{i\theta}$. The warp factor is given by

$$2D = -K + K_0 = \frac{2}{n^2} \log \frac{s}{s_0}, \qquad (4.37)$$

where the Kähler potential is $K = -\frac{2}{n^2} \log s$. This *D* should not be confused with the field space distance, and we trust the reader to distinguish them by the context.

Matching the 4d metric (4.36) to the setup in section 4.1 with n = 1, we obtain the 3d coordinate y:

$$y = \int^{r} e^{\alpha \omega(r)} e^{\nu(r)} dr = \int^{r} \left(1 - \frac{q}{2\pi s_0} \log \frac{r}{r_0} \right)^{\frac{2}{n^2}} \frac{r}{r_0} dr \sim r^2, \qquad (4.38)$$

where we have once more neglected the logarithm compared to the polynomial contribution. Then, we can put the 3d metric in the domain-wall form (2.2), in the $r \to 0$ limit, with:

$$\sigma\left(y^{\frac{1}{2}}\right) = -\gamma\beta\omega\left(y^{\frac{1}{2}}\right) \simeq -\log\left(\left(1 - \frac{q}{2\pi s_0}\log\frac{y^{\frac{1}{2}}}{r_0}\right)^{\frac{1}{n^2}}\frac{y^{\frac{1}{2}}}{r_0}\right) \simeq -\frac{1}{2}\log y.$$
(4.39)

Comparing this to (2.18), we obtain

$$\delta^2 = 8, \quad a = 0. \tag{4.40}$$

We can use these parameters to recover the profiles and scaling of the local solution. For instance, we obtain that $\phi'^2 \gg V$, as in the D7-brane case. We also obtain the field-space profile and scaling relations⁸ from (2.17) and (2.20):

$$D(y) \simeq -\sqrt{\frac{1}{2}\log y}, \qquad (4.41)$$

$$\Delta = y \sim e^{-\sqrt{2}D}, \qquad |R| = e^{2\sqrt{2}D}.$$
(4.42)

We thus find that the full solution can be described in terms of the local description, with the EFT string described in terms of an ETW brane.

⁸This result corrects a factor of $\sqrt{2}$ arising from $|D| \simeq |\sigma_p| \simeq -\sqrt{2} \log(r)$, which was missing in [22].
4.4 The Klebanov-Strassler throat

In the previous examples we have shown that D-branes can play the role of ETW branes in running solutions of compactifications with fluxes. We would like to mention, however, an alternative mechanisms in which Dynamical Cobordisms can get rid of fluxes in the compactification, namely when the running involves axion monodromy.⁹ This is most clearly illustrated in the celebrated Klebanov-Strassler (KS) solution [34], related to the compactification of type IIB theory on the 5d Sasaki-Einstein space $T^{1,1}$ with N units of RR 5-form flux and M units of RR 3-form flux on an $\mathbf{S}^3 \subset T^{1,1}$.

As shown in [21], the KS solution can be regarded as a Dynamical Cobordism, in which the tip of the throat ends spacetime at finite spacetime distance in the radial direction, smoothing out (or UV completing) the singularity of the related Klebanov-Tseytlin (KT) solution [33]. In this section we show that the structure of the KT solution is indeed that of an ETW brane from the viewpoint of the 5d effective theory.

Consider the KT solution [33], whose 10d Einstein frame metric reads:

$$ds_{10}^2 = h^{-1/2}(r)\eta_{\mu\nu}dx^{\mu}dx^{\nu} + h^{1/2}(r)\left(dr^2 + r^2ds_{T^{1,1}}^2\right), \qquad (4.43)$$

with

$$h(r) = b_0 + \frac{M^2 \log \left(r/r_* \right)}{4r^4} \,. \tag{4.44}$$

The singularity is at r_s such that $h(r_s) = 0$, signalling the location of the ETW brane. One can show that $\partial_r h \neq 0$ at $r = r_s$, hence we may expand this harmonic function near this point as

$$h(r) \sim r - r_s \equiv \tilde{r} \,. \tag{4.45}$$

We now take the compactification ansatz

$$ds_{10}^2 = L^2 \left(e^{-5q} ds_5^2 + e^{3q} ds_{T^{1,1}}^2 \right)$$
(4.46)

with L an overall scale. Matching with (4.43) we get the profile for the breathing mode

$$q(r) = \frac{1}{6} \log\left(\left(\frac{r}{L}\right)^4 h(r)\right) \simeq \frac{1}{6} \log \tilde{r} , \qquad (4.47)$$

where in the last equality we have taken the near ETW limit. We also get the 5d Einstein frame metric:

$$L^{2}ds_{5}^{2} = \left(\left(\frac{r}{L}\right)^{2}h^{\frac{1}{2}}\right)^{\frac{3}{3}} \left(h^{-\frac{1}{2}}\eta_{\mu\nu}dx^{\mu}dx^{\nu} + h^{\frac{1}{2}}dr^{2}\right).$$
(4.48)

From it we can derive the relation between \tilde{r} and the radial coordinate y in the local analysis, which is

$$\tilde{r} \sim y^{\frac{3}{5}} \,. \tag{4.49}$$

Reading off the warp factor

$$e^{-2\sigma} = \left(\left(\frac{r}{L}\right)^2 h^{\frac{1}{2}} \right)^{\frac{5}{3}} h^{-\frac{1}{2}} \sim \tilde{r}^{\frac{1}{3}} \sim y^{\frac{1}{5}}, \qquad (4.50)$$

⁹For axion monodromy in inflation, see [57-64].

we finally find

$$\sigma(y) \simeq -\frac{1}{10} \log y \,. \tag{4.51}$$

Hence, the 5d KT solution near the singularity fits with the form of an ETW brane in our local description with

$$\delta = \frac{2\sqrt{30}}{3}, \qquad a = -\frac{3}{2}. \tag{4.52}$$

We can also check that the solution for the scalars also fits in the local model description. The NSNS axion is given by

$$T(r) = \tilde{T} + M \log r \simeq T_s + \frac{M}{r_s} \tilde{r}, \qquad (4.53)$$

again in the near ETW brane limit. Here $T_s = T(r_s)$, which we can keep arbitrary. The field space metric from the 5d action in [33] is given by

$$dD^{2} = 30(\partial q)^{2} + \frac{1}{2}g_{s}^{-1}e^{-6q}(\partial T)^{2}.$$
(4.54)

Using the profiles for q and T in the $\tilde{r} \to 0$ limit, we have

$$(\partial q)^2 \simeq \frac{1}{36\tilde{r}^2}, \qquad e^{-6q} (\partial T)^2 \simeq \left(\frac{M}{r_s}\right)^2 \frac{1}{\tilde{r}}.$$

$$(4.55)$$

For $\tilde{r} \to 0$, the breathing modes dominates the field space distance in field-space. Following [65], it is then an asymptotically geodesic trajectory. This is in contrast with the $r \to \infty$ limit, for which the field-space trajectory was shown to be highly non-geodesic in [38]. Hence we have

$$dD^2 \simeq 30(\partial q)^2 \simeq \frac{5}{6}\tilde{r}^{-2}$$
. (4.56)

Upon integration and using (4.49) we obtain

$$D(y) \simeq -\frac{\sqrt{30}}{10} \log y$$
. (4.57)

This again takes the form found in our local analysis, for the above coefficients (4.52).

Finally, we also check that the 5d scalar potential from [33] scales as predicted by the local model. The complete potential is

$$V(\phi) = -5e^{-8q} + \frac{1}{8}g_s M^2 e^{-14q} + \frac{1}{8}(N + MT)^2 e^{-20q}.$$
(4.58)

Plugging in $T = T_s$ and $D \simeq -\sqrt{30} q$ as dictated by (4.56), we get

$$V(D) = -5e^{\frac{4\sqrt{30}}{15}D} + \frac{1}{8}g_s M^2 e^{\frac{7\sqrt{30}}{15}D} + \frac{1}{8}(N + MT_s)^2 e^{\frac{2\sqrt{30}}{3}D}.$$
 (4.59)

For $N + MT_s \neq 0$, we find that the last term dominates as $D \to \infty$. As predicted by our local analysis, it has an exponential behaviour with D with the coefficient δ given in (4.52). Moreover, as predicted by finding a < 0, the coefficient in front of this exponential is positive.

We hope these examples suffice to convince the reader that the local description provides a simple and efficient framework to discuss the structure of Dynamical Cobordisms near the ETW brane.

5 Small black holes as Dynamical Cobordisms

The analysis of the previous section for single-charge D-brane solutions can be similarly carried out for systems of multiple charges, namely combining D-branes of different dimensionalities. Such systems have been extensively employed in the construction and microscopic understanding of black holes, both with finite horizon, starting with [66], or with vanishing classical horizon area (small black holes) (see [35, 36] for some reviews). In this section we describe brane configurations, closely related to the celebrated D1/D5 system, leading to small black holes, and describe them as cobordism defects of suitable sphere compactifications of the underlying theory. The resulting dimensionally truncated theory corresponds to a 2d theory of gravity and an effective scalar (2d dilaton gravity), for which we find scaling relations analogous to the higher dimensional cases. This description relates the Dynamical Cobordisms to the realization of the Swampland Distance Conjecture in small black holes¹⁰ in [37].

5.1 The D2/D6 system on T^4

We consider a configuration of D6- and D2-branes in the following (1/4 susy preserving) configuration

$$D6: 0 \ 1 \ 2 \times \times \times 6 \ 7 \ 8 \ 9 \tag{5.1}$$

$$D2: 0 \ 1 \ 2 \times \times \times \times \times \times \tag{5.2}$$

where the numbers correspond to directions spanned by the brane worldvolumes and \times 's mark transverse directions. We consider all branes to coincide in the mutually transverse directions 345. We moreover smear the D2-branes in the direction 6789. Eventually these directions will be taken to be compact, so the smeared description is valid for small compactification size.

In the 10d Einstein frame the metric and dilaton profile are given by harmonic superposition (see [70] for background)

$$ds^{2} = Z_{6}(r)^{-\frac{1}{8}} Z_{2}(r)^{-\frac{5}{8}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z_{6}(r)^{\frac{7}{8}} Z_{2}(r)^{\frac{3}{8}} (dr^{2} + r^{2} d\Omega_{2}^{2}) + Z_{6}(r)^{-\frac{1}{8}} Z_{2}(r)^{\frac{3}{8}} dx^{m} dx^{m},$$

$$\Phi(r) = \frac{1}{2\sqrt{2}} \log \left(Z_{6}(r)^{-\frac{3}{2}} Z_{2}(r)^{\frac{1}{2}} \right),$$
(5.3)

where r is the radial coordinate in 345, $d\Omega_2^2$ is the volume of a unit \mathbf{S}^2 in this \mathbf{R}^3 , and m = 6, 7, 8, 9. The harmonic functions are

$$Z_6(r) = 1 + \frac{\rho_6}{r}, \qquad Z_2(r) = 1 + \frac{\rho_2}{r}.$$
 (5.4)

As announced, we now consider compactifying the directions 6789 on a \mathbf{T}^4 (similar results hold for K3 compactification, as usual), with the compactification ansatz

$$ds^{2} = e^{-\frac{t}{\sqrt{2}}} ds_{6}^{2} + e^{\frac{t}{\sqrt{2}}} ds_{T^{4}}^{2} \,.$$
(5.5)

¹⁰For other approaches to Swampland constraints using (large and small) black holes, see e.g. [67–69].

Matching this ansatz to (5.3), we obtain the canonically normalized radion

$$t(r) = \sqrt{2} \log \left(Z_6(r)^{-\frac{1}{8}} Z_2(r)^{\frac{3}{8}} \right) .$$
 (5.6)

The 6d Einstein frame metric reduces to:

$$ds_{6}^{2} = e^{\frac{t}{\sqrt{2}}} \left(Z_{6} \left(r \right)^{-\frac{1}{8}} Z_{2} \left(r \right)^{-\frac{5}{8}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z_{6} \left(r \right)^{\frac{7}{8}} Z_{2} \left(r \right)^{\frac{3}{8}} \left(dr^{2} + r^{2} d\Omega_{2}^{2} \right) \right)$$

$$= Z \left(r \right)^{-\frac{1}{4}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z \left(r \right)^{\frac{3}{4}} \left(dr^{2} + r^{2} d\Omega_{2}^{2} \right)$$
(5.7)

where $Z(r) = Z_6(r)Z_2(r)$.

One can see that the dilaton and radion are both blowing up upon reaching the point r = 0, which is at finite spacetime distance, hence the configuration can be dubbed a 6d small black 2-brane.

As in section 4, we can describe the configuration as a Dynamical Cobordism of the 6d theory compactified on an S^2 with suitable 2-form fluxes (for the RR 2-form field strength and the T^4 reduction of the RR 6-form field strength). To implement this, we take the general ansatz:

$$ds_6^2 = e^{-2\alpha\sigma} ds_4^2 + r_0^2 e^{2\beta\sigma} d\Omega_2^2 \,. \tag{5.8}$$

In the resulting 4d theory, there are non-trivial potential terms for the new radion σ arising from the curvature of \mathbf{S}^2 and the 2-form fluxes. Imposing the Einstein frame in 4d comes down to setting $\gamma = \frac{\alpha}{\beta} = 1$. One can then choose β such that the radion $\sigma(r)$ has a canonically normalized kinetic term and one obtains $\beta = \frac{1}{2}$. From matching this compactification ansatz to equation (5.7), we obtain the canonically normalized radion σ ,

$$\sigma(r) = \log\left(\frac{r^2}{r_0^2} Z(r)^{\frac{3}{4}}\right),$$
 (5.9)

and the following 4d Einstein frame metric

$$ds_4^2 = e^{\sigma} \left(Z(r)^{-\frac{1}{4}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z(r)^{\frac{3}{4}} dr^2 \right) = \left(\frac{r}{r_0}\right)^2 \left(Z(r)^{\frac{1}{2}} \eta_{\mu\nu} dx^{\mu} dx^{\nu} + Z(r)^{\frac{3}{2}} dr^2 \right) \,.$$

This solution is a 4d Dynamical Cobordism, with the D2/D6-brane system playing the role of cobordism defect. The solution has the structure of an ETW brane; there are 3 running scalars going off to infinite distance at the singularity at r = 0, which is straightforward to show lies at finite spacetime distance. Indeed, near r = 0, we have

$$\Delta = \int_0^r \left(\frac{r}{r_0}\right) (Z_6(r)Z_2(r))^{\frac{3}{4}} dr \sim \sqrt{r} \,. \tag{5.10}$$

Furthermore, near the singularity, the distance in field space goes like:

$$dD^{2} = d\Phi^{2} + d\sigma^{2} + dt^{2} \simeq \frac{1}{2} \frac{dr^{2}}{r^{2}} \to D \simeq -\frac{1}{\sqrt{2}} \log(r)$$
(5.11)

Near the singularity, the Ricci scalar in 4d behaves as:

$$|R| \sim r^{-1}$$
 (5.12)

These lead to the familiar scaling relations near r = 0:

$$|R|^{-\frac{1}{2}} \sim \Delta \sim e^{-\frac{1}{\sqrt{2}}|D|}.$$
(5.13)

Since the above full solution has the structure of a Dynamical Cobordism, it should be possible to express it in the framework of our local description, with the D2/D6-brane system playing the role of the ETW brane. Let us define the new coordinate:

$$y = \int^{r} \left(\frac{r}{r_0}\right) (Z_6(r)Z_2(r))^{\frac{3}{4}} dr \sim \sqrt{r}, \qquad (5.14)$$

where we have considered the leading behaviour near r = 0.

Using equation (4.4), we have:

$$\sigma\left(y^2\right) = -\frac{1}{2}\log\left(\frac{y^4}{r_0^2}Z\left(y^2\right)^{\frac{1}{2}}\right) \simeq -\log y\,. \tag{5.15}$$

Matching this to the profile in (2.17), we see that $\delta^2 = 2$ and $a = \frac{2}{3}$. Then we automatically fall back on the previous field-space distance and scaling relations using equations (2.17) and (2.20):

$$D(y) \simeq -\sqrt{2}\log y \,, \tag{5.16}$$

$$\Delta = y \sim e^{-\frac{1}{\sqrt{2}}D} \sim |R|^{-\frac{1}{2}}.$$
(5.17)

This gives yet another nice check of the usefulness of the local analysis.

Beyond the Einstein frame. One last remark that will be relevant in the next sections is that scaling relations similar to those of (5.13) can be found, independent of the frame chosen during the compactification. Indeed, if one insists on keeping γ (and thus, also β) general and tracking it throughout the computations, one obtains the new coordinate near r = 0:

$$\Delta = y \sim r^{\frac{1}{4}(\gamma+1)} \quad \text{and} \quad |R| \sim r^{-\frac{1}{2}(\gamma+1)}.$$
(5.18)

Note that, if $\gamma < -1$, then these scalings behave opposite to those we have seen for ETW branes. This illustrates that the scalings mentioned rely on using the Einstein frame metric to describe the ETW brane.

In setups where one needs (or finds convenient) to use general frames, the condition for an ETW brane is that the picture of a scalar going off to infinity at finite spacetime distance can be attained by a suitable change of frame. In this respect, we note that there is an extra subtlety in dealing with the field space distance in general frames. Indeed, not being in the Einstein frame implies that the radion is multiplying the Einstein-Hilbert term in the action:

$$S_4 \supset \frac{1}{2} \int d^4x \sqrt{-g_4} e^{2\beta\sigma(1-2\gamma)} \left\{ e^{2\beta\gamma\sigma} \left(R_4 - (\partial t)^2 - (\partial \Phi)^2 - \beta^2 \left(6\gamma^2 - 8\gamma + 6 \right) (\partial \sigma)^2 \right) \right\}.$$
(5.19)

It thus makes sense to define the field space distance measured in units set by this coefficient of the Ricci scalar in the action. This field space distance near the singularity in this general frame reads:

$$dD^{2} = d\Phi^{2} + \beta^{2} \left(6\gamma^{2} - 8\gamma + 6 \right) d\sigma^{2} + dt^{2}$$

$$D \simeq -\frac{\sqrt{6\gamma^{2} - 8\gamma + 10}}{4} \log r \,.$$
(5.20)

Hence, we can derive the following universal scaling relations in a general frame:

$$\Delta \sim e^{-\frac{\gamma+1}{\sqrt{6\gamma^2 - 8\gamma + 10}}|D|} \sim |R|^{-\frac{1}{2}}.$$
(5.21)

Note that these reduce to those of (5.13) when setting $\gamma = 1$, as required by the Einstein frame. As a side note, one cannot recover this result in the local description detailed in section 2.2 as it was constructed in the Einstein frame. We leave such a more general formulation of the local construction for future work.

5.2 Small black holes from the D2/D6 system on $T^4 \times T^2$

Let us now consider turning our D6/D2-brane systems into a (small) black hole, by a further compactification on \mathbf{T}^2 .

We take the ansatz

$$ds_6^2 = e^{-q} ds_4^2 + e^q ds_{T^2}^2 \,. \tag{5.22}$$

By matching this ansatz to the 6d metric obtained previously (5.7), we get the 4d Einstein frame metric:

$$e^{q(r)} = Z(r)^{-\frac{2}{8}},$$

$$ds_4^2 = (g_4)_{ij} dx^i dx^j = e^{q(r)} \left(-Z(r)^{-\frac{2}{8}} dt^2 + Z(r)^{\frac{6}{8}} \left(dr^2 + r^2 d\Omega_2^2 \right) \right)$$

$$= -Z(r)^{-\frac{1}{2}} dt^2 + Z(r)^{\frac{1}{2}} \left(dr^2 + r^2 d\Omega_2^2 \right).$$
(5.23)

This solution describes a small black hole (in fact, equivalent to the celebrated D1/D5-brane one, by T-duality in one of the \mathbf{T}^2 directions), of the kind considered in [37].

To motivate the relation with the more general discussion in the next section, let us make the following heuristic argument. Although our solution has three scalar fields, the radial evolution can be reduced to one effective scalar as follows. Near r = 0, all three scalars have the same profile, so we may combine them in one effective scalar D whose effective action near r = 0 is of the form

$$S_4 \sim \frac{1}{2} \int d^4x \sqrt{g_4} \left\{ R_4 - (\partial D)^2 - \frac{1}{4} e^{\frac{1}{\sqrt{2}}D} |F_2|^2 \right\}$$
(5.24)

where we have restricted to the U(1) linear combination under which the D2/D6 system is charged.

With this proviso, we can frame this particular example with the more general class of small Black Holes considered in [37], to be discussed next.

5.3 General small black holes

In the context of the swampland program, [37] proposed the use of 4d small black hole solutions to provide further evidence for a number of a number of Swampland conjectures. A particularly important property is that the 4d solutions contain scalars going off to infinite field space distance at the black hole core. In the spirit of previous sections, in this section we show that these 4d solutions can be turned into 2d Dynamical Cobordisms upon reducing on the S^2 , with the small black hole playing the role of the ETW brane. In fact we will check that the 2d running solution satisfies the familiar scaling relations (for a general frame, since there is no Einstein frame in 2d).

Let us briefly review the key features of such solutions. We consider 4d Einstein-Maxwell coupled to a scalar controlling the gauge coupling. We take the action

$$S_{4d} \sim \frac{1}{2} \int d^4x \sqrt{-g_4} \left(R_4 - (\partial \phi)^2 - e^{2a\phi} |F_2|^2 \right) \,. \tag{5.25}$$

We focus on exponential dependence, since it provided the most explicit class considered in [37]. It also fits with the special role of exponential functions in local descriptions of ETW branes.

Without loss of generality, we take a > 0 so that $\phi \to \infty$ corresponds to weak coupling for the U(1) gauge field. Note that this a should not be confused with the parameter in (2.11), and we trust the reader to distinguish them by the context.

In this theory, electrically charged extremal black holes take the form

$$ds_4^2 = -f(r)dt^2 + f(r)^{-1}dr^2 + r^2 R(r)^2 d\Omega_2^2, \qquad (5.26)$$

where

$$R(r) = \left(1 - \frac{r_h}{r}\right)^{\frac{a^2}{1+a^2}}, \qquad f(r) = \left(1 - \frac{r_h}{r}\right)^{\frac{2}{1+a^2}}.$$
(5.27)

In addition, the profile for the scalar is given by

$$\phi(r) = \phi_0 - \frac{\sqrt{2}a}{1+a^2} \log\left(1 - \frac{r_h}{r}\right) \,. \tag{5.28}$$

The scalar goes off to infinity at the horizon $r = r_h$, which is however not smooth, since the \mathbf{S}^2 shrinks to zero size, leading to a small black hole.

In the string theory context, small black holes can be easily built by using D-branes. In fact, we now recast the above solution in a form closer to the solution (5.23), which described our system of D2- and D6-branes on $\mathbf{T}^4 \times \mathbf{T}^2$. This was already anticipated when we obtained (5.24), which has the structure of (5.25) (for $a = \frac{1}{2\sqrt{2}}$).

Carrying out the coordinate change $r \to r + r_h$, the metric (5.26) becomes

$$ds_4^2 = -\left(1 + \frac{r_h}{r}\right)^{-\frac{2}{1+a^2}} dt^2 + \left(1 + \frac{r_h}{r}\right)^{\frac{2}{1+a^2}} \left(dr^2 + r^2 d\Omega_2^2\right) \,. \tag{5.29}$$

Similarly, the scalar reads

$$\phi(r) = \phi_0 + \frac{\sqrt{2}a}{1+a^2} \log\left(1 + \frac{r_h}{r}\right) \,. \tag{5.30}$$

This has the structure of (5.23) with $Z(r) = (1 + r_h/r)^{\frac{4}{1+a^2}}$. Note that the core of the small black hole now lies at r = 0.

We now perform the reduction on S^2 to express these solutions as 2d running solutions describing a local Dynamical Cobordism, with the small black hole playing the role of the ETW brane. We will also recover the corresponding (general frame) scaling relations.

Since there is no Einstein frame in 2d, we perform the S^2 reduction with the following general ansatz:

$$ds_4^2 = e^{-2\alpha\omega} ds_2^2 + e^{2\beta\omega} r_0^2 d\Omega_2 \,. \tag{5.31}$$

The 2d action obtained from the compactification contains the terms

$$S_{2d} \supset \frac{1}{2} \int d^2x \sqrt{-g_2} e^{2\beta\omega} \left(R_2 - (\partial\phi)^2 - 6\beta^2 \left(\partial\omega\right)^2 \right) \,. \tag{5.32}$$

These expressions already show the impossibility to define an Einstein frame: it would require $\beta = 0$, and this would kill the radion's kinetic term. We therefore keep β general, so we deal with a dilaton-gravity theory. By matching the ansatz (5.31) with the 4d metric (5.29) we get the profile for the radion

$$\omega(r) = \frac{1}{\beta} \log\left(\frac{r}{r_0} \left(1 + \frac{r_h}{r}\right)^{\frac{1}{1+a^2}}\right), \qquad (5.33)$$

and the 2d metric

$$ds_2^2 = \left(\frac{r}{r_0}\right)^{2\gamma} \left(-\left(1+\frac{r_h}{r}\right)^{-\frac{2(1-\gamma)}{1+a^2}} dt^2 + \left(1+\frac{r_h}{r}\right)^{\frac{2(1+\gamma)}{1+a^2}} dr^2\right),$$
(5.34)

where $\gamma = \frac{\alpha}{\beta}$.

Computing the 2d Ricci scalar and taking the leading order in $r \to 0$ we get

$$|R| \sim r^{-2\frac{(\gamma+1)a^2}{1+a^2}},\tag{5.35}$$

where we are ignoring a constant prefactor.¹¹

Similarly, the spacetime distance from a given r to the singularity, at leading order in $r \to 0$, scales as

$$\Delta \sim r^{\frac{(\gamma+1)a^2}{(1+a^2)}}.$$
 (5.36)

We note that, as expected, the scaling is the familiar ETW one if $\gamma > -1$. As explained above, the fact that 2d gravity is topological means that the criterion for an ETW brane in a solution should be that the usual relations hold in *some* suitable frame.

Let us now recover the usual scalings with the field distance. Recalling the latter is measured in units set by the coefficient of the Ricci scalar in the action, we can it read off from (5.32) as:

$$dD^2 = d\phi^2 + 6\beta^2 d\omega^2 \,. \tag{5.37}$$

¹¹This prefactor vanishes for either $a^2 = 1$ or $a^2 = -2\gamma$. We will skip these cases without further discussion.

Plugging the profiles (5.28) and (5.33) at leading order as $r \to 0$ and integrating the line element we recover

$$D(r) \simeq -\frac{a\sqrt{2+6a^2}}{1+a^2}\log r$$
. (5.38)

Finally, together with the previous results for the distance to the end of the world and the curvature, we obtain the scalings

$$\Delta \sim e^{-\frac{\delta}{2}D}, \qquad |R| \sim e^{\delta D}, \qquad (5.39)$$

with

$$\delta = \frac{2(\gamma+1)a}{\sqrt{2+6a^2}} \,. \tag{5.40}$$

Hence, we recover the general frame scaling relations introduced in section 5.1. This shows that small black hole solutions can be regarded as just another instance of Dynamical Cobordism, and that they admit local scaling relations identifying the small black hole core with ETW branes in 2d.

6 Swampland constraints and surprises from the UV

In this section we discuss interesting interplays of the scalar running off to infinity in field space in Local Dynamical Cobordisms and the Swampland constraints.

6.1 Swampland distance conjecture and other constraints

Many studies of Swampland constraints are related to infinity in scalar moduli/field space (see [42–44] for reviews). Since Dynamical Cobordisms explore infinite field space distances, in this section we discuss the interplay with different Swampland constraints, especially the Distance Conjecture [71] (see [38, 39, 65, 72–83] and the reviews above for other approaches).

Let us focus on the simplest expression of the Distance Conjecture, which states that, when the scalars are taken to infinite field space distance D (in an adiabatic approach, namely, by changing the spacetime independent vevs), there is a tower of states becoming exponentially light, and thus the cutoff of the effective theory is lowered as

$$\Lambda \sim e^{-\alpha D} \,, \tag{6.1}$$

with some positive order 1 coefficient α .

This scaling can be combined in an interesting way with our scalings near ETW branes. For instance, using (2.20), we have

$$\Lambda \sim \Delta^{\frac{2\alpha}{\delta}} \,. \tag{6.2}$$

This matches with our intuition that the full description of the ETW brane requires UV completing the effective theory. It is important to note that the appearance of an infinite tower in the adiabatic version of the Distance Conjecture does not necessarily imply the appearance of a tower in the present Dynamical Cobordism context. On the other hand, the lowered cutoff certainly signals that there could be situations where the naive ETW brane

picture as described in effective theory may be corrected. We will see explicit examples in section 6.2.

Using also (2.20), we get that the cutoff scale relates to the spacetime curvature as

$$R| \sim \Lambda^{-\frac{\partial}{\alpha}} \,, \tag{6.3}$$

(where we have taken the generic case $\delta \neq (2d/(d-2))^{1/2}$ for concreteness). This relation, already noted in [22] is reminiscent of (although admittedly different in spirit from) that in [39] for AdS vacua.

From this perspective, the correlation between the appearance of the naked singularity and the running of the scalar going off to infinity suggests that the lowered cutoff of the swampland distance conjecture is responsible for regulating the singularity, which would be resolved in a more complete microscopic UV description. This remark is in the spirit of [41] (see also [56]) and [37], where the singular behaviour of certain defects (EFT strings or small black holes, respectively) is related to scalars going off to infinite distance.

From our perspective, the relation follows from the Dynamical Cobordism Distance Conjecture in [22]. In our present terms: every infinite field distance limit of an effective theory consistent with quantum gravity can be realized as a solution running into a cobordism ETW brane (possibly in a suitable compactification of the theory).

In particular, in sections 4 and 5 we provided a description of general defects as ETW branes of Dynamical Cobordisms. This general framework encompasses the defects in [37, 41] as particular examples.

An interesting spin-off of our local analysis is that it constrains the asymptotic form of the potential. Namely, whenever it is not vanishing (actually, negligible as compared with the scalar kinetic energy) it has an exponential form with a critical exponent δ , cf. (2.15). It is thus interesting to compare this asymptotic form of the potential with Swampland constraints expected to hold near infinity in scalar field space.

Let us consider the de Sitter conjecture in the version of [84] (see [76, 85] for the refined one), namely $|\nabla V|/V > O(1)$. From (2.15) we have

$$\frac{V'}{V} = \delta \,. \tag{6.4}$$

Since in general the critical exponent $\delta \sim \mathcal{O}(1)$, the potential satisfies the de Sitter conjecture. This fits nicely with the idea that the latter is expected to hold near infinity in moduli/field space.

Moreover, let us compare with the Transplanckian Censorship Conjecture [86]

$$|\nabla V| \ge \frac{2}{\sqrt{(d-1)(d-2)}} V.$$
 (6.5)

When V < 0, the constraint is trivial; on the other hand, when V > 0, in our setup we must have a < 0, and the expression (2.14) for δ guarantees that the above inequality is satisfied. A caveat for the above statements is that both the de Sitter and the Transplanckian Censorship conjectures involve the gradient ∇V , whereas our local description provides the potential only along one direction, the effective scalar dominating the running near the ETW brane. Hence, the comments above would hold under the assumption that the effective scalar in the local description follows a gradient flow. It would be interesting to assess this point in explicit models, and we leave this as an open question for future work.

6.2 Large N surprises from the UV

In the previous section we have discussed that the Distance Conjecture implies a lowered cutoff as one approaches the ETW brane. Indeed, as mentioned at several points, the microscopic description of the ETW branes lies in the underlying UV completion. In most of our examples, the corresponding cobordism defect is known, so that the end of the world picture can be confirmed in the full theory. However, it is conceivable that in some specific cases there exist UV effects hidden at the core of the ETW brane potentially modifying this picture. In this section we present two examples, where such corrections exist and lead to large backreactions, ultimately turning the candidate ETW brane into a domain wall interpolating to a new region beyond the apparent singularity. A further interesting observation is that both examples are related to large N physics and holography.

Large number of M2-branes. Consider as our first example a stack of N D2-branes in flat 10d spacetime (or at a smooth point in any other compactification). Locally around the D2-brane location the S^6 truncation yields a 4d theory with an ETW brane, at which a scalar (a combination of the radion and the dilaton) goes to infinity in field space. One may follow the theory in this limit and, as noted in [54], realize that the strong coupling is solved by lifting to M-theory, and turning the D2-branes into M2-branes. For small N, the UV completion of the effective ETW brane is thus merely a stack of M2-branes removing the flux and allowing spacetime to end, as befits a Dynamical Cobordism.

On the other hand, for N large we have a different behavior: the large number of M2-branes backreact on the geometry and generate an infinite $AdS_4 \times S^7$ throat. The effective theory ETW brane has a UV description with so many degrees of freedom that it actually generates a gravity dual beyond the wall.

From the perspective of the running scalars, the $AdS_4 \times S^7$ represents a minimum of the $(S^7 \text{ radion})$ potential. Hence the full D2/M2 solution describes the running of the theory from the slope of the potential down to a stable minimum, at which the theory relaxes to a maximally symmetric solution, instead of hitting an end of the world. The location of the minimum in field space is hidden near infinity in the original D2-brane effective description. Hence, the large N allows for the appearance of a minimum at strong coupling, which is nevertheless tractable.¹²

Moreover, the full D2/M2 solution describes a dynamical cobordism from the M-theory perspective. Far away from the stack of branes we can use the description in terms of D2-branes. As described above the 4d theory would be obtained by compactifying Type IIA on an \mathbf{S}^6 . This would be further lifted to M-theory on $\mathbf{S}^6 \times \mathbf{S}^1$. On the other hand, we have just argued that close to the stack of branes the 4d theory is given by M-theory on \mathbf{S}^7 . We then see that this solution describes a dynamical cobordism between to different

¹²This is reminiscent of the argument [87] that the scale separation (and hence the tractability) of the AdS minima in [88, 89] is controlled by a large number of flux units.

compactifications. Notice that this is not a cobordism to nothing, described by ETW brane solutions.

Warped KS throat with large number of D3-branes. Our second example is based on the warped throat considered in section 4.4. Recall we have type IIB theory compactified on $T^{1,1}$ with N units of RR 5-form flux and M units of RR 3-form flux on the S^3 , and we focus on the choice of parameters N = KM + P. At the level of the 4d effective theory, we recover a KT solution with a singularity at a finite spacetime distance, at which a scalar (a combination of the $T^{1,1}$ radion and the dilaton, but dominated by the former) goes off to infinite field space distance.

The UV smoothing of this singularity is slightly trickier than the N = KM case of section 4.4. It involves the smoothing of the singular conifold geometry into a deformed conifold, with a finite size S^3 , but there remain P D3-branes at the tip of the throat. This can be shown using the holographic dual field theory, as follows. There is a Seiberg duality cascade from the initial $SU(N) \times SU(N + M)$ theory in which N effectively decreases in multiples of M; hence, in the last step of the cascade we have an $SU(P) \times SU(M + P)$ gauge theory, whose strong coupling dynamics leads to an remnant $\mathcal{N} = 4$ SU(P) theory, as befits the above mentioned P probe D3-branes.

Hence, for small P the ETW brane of the 5d theory is microscopically described by the smooth Klebanov-Strassler throat dressed with P explicit D3-branes, required to absorb the remnant 5-form flux and allow spacetime to end.

On the other hand, for P large we have a different behavior: the large number of D3-branes backreact on the geometry and generate an infinite $AdS_5 \times S^5$ throat. The effective theory ETW brane has a UV description with so many degrees of freedom that it actually generates a gravity dual beyond the wall. The interpretation of this strong correction in terms of the running scalars is similar to the one mentioned above, as the apperance of an AdS minimum hidden near the infinite field space distance limit of the effective description.

We have seen two examples in which a naive ETW brane in the effective description has a UV description encoding large backreactions on the geometry recreating a geometry beyond the wall. Alternatively, the corrections generate minima in the scalar potential in the region near field space infinity of the effective description. It would be interesting to explore in more detail these and other possible classes of examples exhibiting this phenomenon. We hope to report on this in the future.

7 Conclusions

In this paper we have studied Dynamical Cobordism solutions in which theories of gravity coupled to scalars develop an end of spacetime. The latter is encoded in the effective theory as the appearance of a singularity at finite spacetime distance, at which some scalars run off to infinite field space distance. We have provided a local description of the configurations in the near ETW brane regime, and shown that the solutions are largely simplified, and fall in universality classes characterized by a critical exponent δ , which controls the profiles of

Example	d	δ	a
Massive IIA	10	$\frac{5}{\sqrt{2}}$	$-\frac{16}{5}$
Non-susy $USp(32)$ string	10	$\frac{3}{\sqrt{2}}$	0
D7 branes	9	$\frac{4\sqrt{14}}{7}$	0
D6 branes	8	$\sqrt{2}$	$\frac{4}{7}$
D5 branes	7	$\frac{2}{\sqrt{5}}$	$\frac{5}{6}$
D4 branes	6	$\frac{1}{\sqrt{5}}$	$\frac{24}{25}$
Klebanov- $Strassler$	5	$\frac{2\sqrt{30}}{3}$	$-\frac{3}{2}$
Bubble of Nothing	4	$\sqrt{6}$	0
D2 branes	4	$\frac{\sqrt{14}}{7}$	$\frac{20}{21}$
$D2/D6 \ on \ T^4 \times S^2$	4	$\sqrt{2}$	$\frac{2}{3}$
D1 branes	3	$\sqrt{2}$	$\frac{3}{4}$
EFT string	3	$2\sqrt{2}$	0

Table 1. Table of examples in this paper, with the corresponding parameters for the local descriptionnear the ETW brane.

the different fields and the scaling relations among the field space distance D, spacetime distance Δ and scalar curvature R.

We have studied several explicit models of ETW branes and characterized them in the local description, computing their critical exponent. The different examples and their key parameters are displayed in table 1. This list is intended to illustrate typical values of these parameters. It would be interesting to explore more examples and to explore possible connections among ETW branes described by the same parameters.

We have moreover shown that small black holes can also be regarded as Dynamical Cobordisms, and satisfy similar scaling laws. It would be interesting to explore from the cobordism perspective the recent applications of small black holes to the derivation of swampland constraints.

There are several interesting open directions for the future:

- We have focused on solutions with spatial dependence. It would certainly be interesting to explore time-dependent backgrounds, and their possible application to cosmology.
- In our local analysis we have focused on certain particular choices. For instance, we have not considered solutions where $|V| \gg |V_t|$, and we have moreover taken solutions controlled by a constant parameter a < 1. More general possibilities are in principle allowed from a mere effective field theory perspective, but they are not realized in any of the string theory examples we have explored. It is thus an interesting question if there are UV complete models realizing them, or on the contrary, they are excluded by some further arguments of consistency with Quantum Gravity.
- Finally, it would be interesting to get a better understanding of the possible appearance of non-trivial corrections in the large field region near the ETW branes, in particular

those leading to large backreactions signalling the existence of new minima of the scalar potential. This could lead to further insights into the stabilization of moduli in asymptotic regions of moduli/field space. The two examples mentioned in our work signal an interesting interplay with large N limits and holography, which may provide an extra leverage on these configurations.

We hope our work motivates interesting results in this and other directions.

Acknowledgments

We are pleased to thank I. Basile, J. R. Espinosa, L. Ibánez, F. Marchesano, M. Montero and I.Valenzuela for useful discussions. J.C. and M.D. also wish to acknowledge the hospitality of the Department of Physics of Harvard University during the early and late stages of this work, respectively. This work is supported through the grants CEX2020-001007-S and PGC2018-095976-B-C21, funded by MCIN/AEI/10.13039/501100011033 and by ERDF A way of making Europe. The work by R.A. is supported by the grant BESST-VACUA of CSIC. The work by M.D. is supported by the FPI gran no. FPI SEV-2016-0597-19-3 from Spanish National Research Agency from the Ministry of Science and Innovation. The work by J.C. and J. H. is supported by the FPU grants no. FPU17/04181 and FPU20/01495 from the Spanish Ministry of Education.

A Local Dynamical Cobordisms with curved (d-1)-dimensional slices

A.1 General analysis for curved slices

We can generalize the discussion in section 2 to the case in which the ETW brane has constant internal curvature R_d . Namely we take the foliation ansatz (2.2) with ds_{d-1}^2 describing a constant curvature (d-1)-dimensional metric. The equations of motion read

$$(d-1)\sqrt{2(V-V_t)}\,\sigma' - \partial_{\phi}V_t = 0\,, \qquad (A.1)$$

$$\frac{1}{2}(d-1)(d-2)\sigma'^2 + V_t - \frac{1}{2}e^{2\sigma}R_d = 0, \qquad (A.2)$$

$$(d-2)\sigma'' - 2(V-V_t) - \frac{1}{d-1}e^{2\sigma}R_d = 0, \qquad (A.3)$$

where we have again introduced the tunneling potential defined in (2.6).

For $R_d \neq 0$, it is still possible to eliminate σ by combining the first two equations (and their derivatives):

$$(d \partial_{\phi} V_t - (d-1)\partial_{\phi} V) \partial_{\phi} V_t = 2(d-1)(V_t - V) \left[\partial_{\phi}^2 V_t + \frac{2}{d-2}((d-1)V - (d-2)V_t) \right].$$
(A.4)

Importantly, in this derivation we need to assume $R_d \neq 0$, so that we do not expect to necessarily recover the results in section 2.

Restricting to the case $V = aV_t$, with a constant, we find that the solution to this equation is

$$V_t(\phi) = -c \left(\cosh\left(\frac{(a(d-1)+2-d)\phi}{\sqrt{(1-a)(d-2)(d-1)}}\right) \right)^{2-\frac{2}{a(d-1)+2-d}},$$
(A.5)

where we have ignored an integration constant that is irrelevant for the $\phi \to \infty$ limit.

Notice that, for a > 1, the coefficient in front of ϕ becomes imaginary and then what we have is a cosine, rather than a hyperbolic cosine. As we are not interested in this behaviour we from now on require a < 1. From computing ϕ'^2 from this solution and requiring that it must be positive, we then learn that we must have c > 0.

In addition, as we are interested in ETW branes, we want to require that ϕ'^2 blows up as $\phi \to \infty$. This is equivalent to having $|V_t| \to \infty$ in this same limit, which in turn implies that the power in (A.5) must be positive. This gives us that the only ETW brane solutions are for $a < \frac{d-2}{d-1}$. For this range of a, we can approximate the hyperbolic cosine by an exponential (as we are interested in the limit $\phi \to \infty$) and we have

$$V_t(\phi) \simeq -c \left(\exp\left(\frac{(a(d-1)+2-d)\phi}{\sqrt{(1-a)(d-2)(d-1)}}\right) \right)^{2-\frac{2}{a(d-1)+2-d}} = -c e^{\delta \phi}.$$
 (A.6)

The coefficient δ is

$$\delta = 2\sqrt{\frac{d-1}{d-2}(1-a)} \,. \tag{A.7}$$

So for $a < \frac{d-2}{d-1}$ the case of a ETW brane with internal curvature coincides with the case studied in the paper. Interestingly, this case turns out to be more restrictive than the $R_d = 0$ one, for which any a < 1 described an ETW brane.

This solution was also assuming that $a \neq \frac{d-2}{d-1}$. Plugging that particular value in (A.4), we find that the equation of motion simplifies to

$$(\partial_{\phi} V_t)^2 = V_t \cdot \partial_{\phi}^2 V_t \,. \tag{A.8}$$

This equation has the solution

$$V_t = -c \, e^{\delta \, \phi} \,, \tag{A.9}$$

with c and δ arbitrary constants. In order to describe an ETW brane we require $\delta > 0$. Interestingly, for this special value of a with $R_d \neq 0$, we find that we recover the exponential behaviour, but with the freedom of choosing the critical exponent δ .

In both cases we find the same exponential behaviour for V_t . Therefore, just as in section 2.2, we find that the potential takes the form

$$V(\phi) \simeq -a c e^{\delta \phi} \,. \tag{A.10}$$

However, here we uncover that, for a given potential of this form, the setup with $R_d \neq 0$ allows for two possible values of a, namely the $a < \frac{d-2}{d-1}$ given in (A.7)), or the value $a = \frac{d-2}{d-1}$, with δ and a independent. For this reason, from now on we keep a and δ as different variables when solving the rest of the equations, and at the end we comment on the two possibilities.

Using (2.6) we can obtain the profile for ϕ

$$\phi(y) \simeq -\frac{2}{\delta} \log\left(\frac{\delta}{2}\sqrt{2(1-a)c}\,y\right)$$
 (A.11)

Notice that this is the equivalent to (2.16), but with a and δ kept independent. The leading behaviour is then given by

$$\phi(y) \simeq -\frac{2}{\delta} \log y \,, \tag{A.12}$$

and thus the field only depends on the critical exponent.

We can now use (A.1) to get the profile for the warp factor σ :

$$\sigma \simeq -\frac{1}{(d-1)(1-a)}\log y, \qquad (A.13)$$

where we have set an integration constant to zero without loss of generality. We recover the equivalent to (2.18), albeit with a and δ kept independent. We see that the warp factor doesn't depend on δ , but specifically on the prefactor a of the potential.

Finally, we have to check that the solution is compatible with (A.3). From it we obtain the condition

$$\frac{4}{\delta^2} - \frac{d-2}{(d-1)(1-a)} + \frac{R_d}{d-1} y^{2-\frac{2}{(d-1)(1-a)}} = 0.$$
 (A.14)

Let us now apply it for the two possible values for a:

- For $a < \frac{d-2}{d-1}$, the power of y in the last term is positive, so that it is subleading in the $y \to 0$ limit. Moreover, recall that in this case δ relates to a via (A.7), which is the precise the value for which the first two terms cancel each other. In conclusion, for $a < \frac{d-2}{d-1}$ having $R_d \neq 0$ becomes irrelevant as we approach the ETW and we basically recover the same results as in the $R_d = 0$ case.
- For $a = \frac{d-2}{d-1}$, the exponent of y vanishes, and hence the R_d term is relevant. In this case, consistency of the equations requires

$$\delta = 2\left(d - 2 - \frac{R_d}{d - 1}\right)^{-\frac{1}{2}}.$$
(A.15)

Therefore, for this case δ is also fixed, but in terms of R_d . Notice that this quantity must satisfy $R_d < (d-2)(d-1)$. Provided this condition, we find that δ can take any positive value.

This case corresponds to a metric $ds^2 = dy^2 + y^2 ds_{d-1}^2$, hence it describes a conical singularity. The singularity is absent in the case $R_d = (d-1)(d-2)$, namely the curvature of ds_{d-1}^2 is that of \mathbf{S}^{d-1} , and the geometry is locally smooth, and we have $\delta = 0$ and no exponential growth of the potential. Also, in order to have an ETW brane, the (d-1)-dimensional curvature must be lower than that of \mathbf{S}^{d-1} .

In conclusion, given a potential with an exponential behaviour as $\phi \to \infty$, in the $R_d \neq 0$ case there exist two different kind of solutions. In the first one the value of R_d is irrelevant and we recover the same behaviour as in the $R_d = 0$ case (but with a more constrained critical exponent, $\delta > \frac{2}{\sqrt{d-2}}$). In the second, the curvature R_d is relevant and it must be fixed by the critical exponent by (A.15).

A.2 Witten's bubble of nothing

To illustrate the above general formulation for curved (d-1)-dimensional slices, we consider the example of the celebrated Witten's bubble of nothing [4] (see [5–8] for other recent realization of bubbles of nothing). We show it admits a description in an effective 4d theory of gravity coupled to a scalar with zero potential, as a 4d Dynamical Cobordism, and characterize its local description and critical exponent δ .

Related discussion of a 4d effective description of the configuration have appeared in [90] (recently revisited in the context of bubbles in de Sitter space in [31, 32]).

Since we have restricted our discussion to dependence on spatial coordinates, we actually consider the euclidean 5d Schwarzschild black hole solution, before the Wick rotation to the expanding bubble solution. The 5d metric reads

$$ds^{2} = \left(1 - \frac{R^{2}}{r^{2}}\right)^{-1} dr^{2} + r^{2} d\Omega_{3}^{2} + \left(1 - \frac{R^{2}}{r^{2}}\right) d\phi^{2}.$$
 (A.16)

Here ϕ parametrizes an \mathbf{S}^1 fibered over the radial coordinate r, times and \mathbf{S}^3 ; the radial coordinate is constrained to the range $r \geq R$, and the \mathbf{S}^1 shrinks to zero size at the euclidean horizon r = R (in a smooth way for the periodicity $\phi \sim \phi + 2\pi R$).

We would like to perform a reduction to 4d along the S^1 . This is a sphere reduction analogous to those in section 4.1. Hence, we match this metric with (4.2), for n = 1, d = 4, and, using (4.3), $\alpha = -\sqrt{1/6}$ and $\beta = -\sqrt{2/3}$. We obtain that the radion ω in (4.2) is:

$$\omega = -\sqrt{\frac{3}{8}} \log\left(1 - \frac{R^2}{r^2}\right). \tag{A.17}$$

The 4d metric is given in (4.5) and reads

$$ds_4^2 = \left(1 - \frac{R^2}{r^2}\right)^{-\frac{1}{2}} dr^2 + \left(1 - \frac{R^2}{r^2}\right)^{\frac{1}{2}} r^2 d\Omega_3^2.$$
(A.18)

We would now like to zoom into the location of the ETW brane, the euclidean horizon r = R. So we introduce the coordinate $\tilde{r} = 1 - \frac{R^2}{r^2}$. Near $r \to R$ the metric scales as

$$ds_4^2 \sim \tilde{r}^{-\frac{1}{2}} d\tilde{r}^2 + \tilde{r}^{\frac{1}{2}} d\Omega_3^2.$$
 (A.19)

Now, we make the change (4.6):

$$y = \int \frac{d\tilde{r}}{\tilde{r}^{1/4}} \simeq \tilde{r}^{3/4}$$
 (A.20)

Replacing $\tilde{r} \simeq y^{\frac{4}{3}}$ in (A.19) we get the 4d metric as a foliation of \mathbf{S}^3 slices:

$$ds_4^2 \sim dy^2 + y^{\frac{2}{3}} d\Omega_3^2 \,. \tag{A.21}$$

This corresponds to a metric of the kind (2.2) for curved 3d slices, namely of the kind studied in appendix A.1. Using (A.13) we can see that a = 0, and from (A.7) $\delta = \sqrt{6}$. Interestingly, this corresponds to the case in which the curvature of the slices is irrelevant, and the solution is similar to the $R_d = 0$ case.

We could have also obtained the same result from the profile for the radion,

$$\omega = -\sqrt{\frac{3}{8}}\log\tilde{r} \simeq -\sqrt{\frac{2}{3}}\log y.$$
(A.22)

By using (A.12), $\omega \simeq -\frac{2}{\delta} \log y$, we read that $\delta = \sqrt{6}$, hence a = 0.

Hence Witten's bubble of nothing is described by a 4d Dynamical Cobordism running solution with the scalar reaching off to infinite distance in fields space at a rate controlled by the critical exponent $\delta = \sqrt{6}$. This provides a simple local description in terms of an ETW brane. From this perspective, the 5d solution provides the UV completion of the ETW brane, which in this case is purely a geometrical closing-off of the geometry.

We would like to emphasize that this example provides an explicit realization of the picture discussed in section 4, in particular figure 1 (albeit, with no brane dressing at the tip). Namely, the complete solution involves a genuine compactification on a finite size \mathbf{S}^1 , yet it is described by a local EWT brane model identical to that obtained as an \mathbf{S}^1 reduction on a flat \mathbf{R}^2 (which, given the vanishing potential, straightforwardly leads to a = 0, hence $\delta = \sqrt{6}$). This supports the picture in section 4 that the sphere reductions in the flat space transverse to the D-branes suffices to provide the local description even in the (physically more interesting case) in which the transverse space is globally given by a more involved geometry, implementing the actual compactification to the lower-dimensional theory.

B Subleading corrections to the local description

In section 2.2 we took constant a as a proxy for the leading behaviour of $a(\phi)$ as $\phi \to \infty$. Here we consider the role of possible subleading corrections. We notice that these corrections do not necessarily go to zero as $\phi \to \infty$ in (2.12). For example, let us take

$$\sqrt{1-a(\phi)} = \sqrt{1-a} + \frac{b}{\phi}.$$
(B.1)

It is clear that $a(\phi)$ asymptotes to a as $\phi \to \infty$, but after doing the integral in (2.12) the correction to the leading behaviour given by the second term behaves as $\log \phi$. Indeed, ignoring constant prefactors we get

$$V_t \sim \phi^2 \sqrt{\frac{d-1}{d-2}} {}^b e^{\delta \phi} , \qquad (B.2)$$

with δ defined in (2.14). Comparing with (2.13) we see that we can describe this example with our leading order analysis if we allow for $c \sim \phi^{2\sqrt{\frac{d-1}{d-2}}b}$. Notice that the example in section 3.2 precisely realise this behaviour (see equation (3.19)).

As a general lesson, we can include these kind of corrections that do not vanish in the $\phi \to \infty$ limit by promoting c from just a constant to a ϕ -dependent quantity that may hide subleading corrections. In this way, it may happen that $c \to \infty$ as $\phi \to \infty$ as long as it blows-up slower than an exponential (otherwise it would not represent a subleading behaviour).

This remark is specially interesting in the $a(\phi) \to 0$ case. From (2.15) we would conclude that $V \to 0$ if c is a finite constant. However, if allowing $c \to \infty$ because of possible subleading terms, it can happen that a times c remains finite in the $\phi \to \infty$ limit. In this way, we describe a solution in which $\phi'^2 \gg V$ (i.e., $a(\phi) \to 0$) without requiring that V vanishes asymptotically.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- J. McNamara and C. Vafa, Cobordism Classes and the Swampland, arXiv: 1909.10355 [INSPIRE].
- P. Hořava and E. Witten, Heterotic and type-I string dynamics from eleven-dimensions, Nucl. Phys. B 460 (1996) 506 [hep-th/9510209] [INSPIRE].
- P. Hořava and E. Witten, Eleven-dimensional supergravity on a manifold with boundary, Nucl. Phys. B 475 (1996) 94 [hep-th/9603142] [INSPIRE].
- [4] E. Witten, Instability of the Kaluza-Klein Vacuum, Nucl. Phys. B 195 (1982) 481 [INSPIRE].
- [5] H. Ooguri and L. Spodyneiko, New Kaluza-Klein instantons and the decay of AdS vacua, Phys. Rev. D 96 (2017) 026016 [arXiv:1703.03105] [INSPIRE].
- [6] I. García Etxebarria, M. Montero, K. Sousa and I. Valenzuela, Nothing is certain in string compactifications, JHEP 12 (2020) 032 [arXiv:2005.06494] [INSPIRE].
- [7] G. Dibitetto, N. Petri and M. Schillo, Nothing really matters, JHEP 08 (2020) 040
 [arXiv:2002.01764] [INSPIRE].
- [8] P. Bomans, D. Cassani, G. Dibitetto and N. Petri, *Bubble instability of m IIA on* $AdS_4 \times S^6$, arXiv:2110.08276 [INSPIRE].
- M. Montero and C. Vafa, Cobordism Conjecture, Anomalies, and the String Lamppost Principle, JHEP 01 (2021) 063 [arXiv:2008.11729] [INSPIRE].
- [10] H. Ooguri and T. Takayanagi, Cobordism Conjecture in AdS, arXiv: 2006.13953 [INSPIRE].
- [11] M. Dierigl and J.J. Heckman, Swampland cobordism conjecture and non-Abelian duality groups, Phys. Rev. D 103 (2021) 066006 [arXiv:2012.00013] [INSPIRE].
- [12] Y. Hamada and C. Vafa, 8d supergravity, reconstruction of internal geometry and the Swampland, JHEP 06 (2021) 178 [arXiv:2104.05724] [INSPIRE].

- [13] R. Blumenhagen and N. Cribiori, Open-Closed Correspondence of k-theory and Cobordism, arXiv:2112.07678 [INSPIRE].
- [14] T. Takayanagi, Holographic Dual of BCFT, Phys. Rev. Lett. 107 (2011) 101602
 [arXiv:1105.5165] [INSPIRE].
- [15] O. Aharony, L. Berdichevsky, M. Berkooz and I. Shamir, Near-horizon solutions for D3-branes ending on 5-branes, Phys. Rev. D 84 (2011) 126003 [arXiv:1106.1870] [INSPIRE].
- [16] B. Assel, C. Bachas, J. Estes and J. Gomis, Holographic Duals of D = 3 N = 4 Superconformal Field Theories, JHEP 08 (2011) 087 [arXiv:1106.4253] [INSPIRE].
- [17] C. Bachas and I. Lavdas, Quantum Gates to other Universes, Fortsch. Phys. 66 (2018) 1700096 [arXiv:1711.11372] [INSPIRE].
- [18] C. Bachas and I. Lavdas, Massive Anti-de Sitter Gravity from String Theory, JHEP 11 (2018) 003 [arXiv:1807.00591] [INSPIRE].
- [19] M.V. Raamsdonk and C. Waddell, Holographic and localization calculations of boundary F for $\mathcal{N} = 4$ SUSY Yang-Mills theory, JHEP **02** (2021) 222 [arXiv:2010.14520] [INSPIRE].
- [20] M. Van Raamsdonk and C. Waddell, Finding $AdS^5 \times S^5$ in 2 + 1 dimensional SCFT physics, JHEP 11 (2021) 145 [arXiv:2109.04479] [INSPIRE].
- [21] G. Buratti, M. Delgado and A.M. Uranga, Dynamical tadpoles, stringy cobordism, and the SM from spontaneous compactification, JHEP 06 (2021) 170 [arXiv:2104.02091] [INSPIRE].
- [22] G. Buratti, J. Calderón-Infante, M. Delgado and A.M. Uranga, Dynamical Cobordism and Swampland Distance Conjectures, JHEP 10 (2021) 037 [arXiv:2107.09098] [INSPIRE].
- [23] E. Dudas and J. Mourad, Brane solutions in strings with broken supersymmetry and dilaton tadpoles, Phys. Lett. B 486 (2000) 172 [hep-th/0004165] [INSPIRE].
- [24] R. Blumenhagen and A. Font, Dilaton tadpoles, warped geometries and large extra dimensions for nonsupersymmetric strings, Nucl. Phys. B 599 (2001) 241 [hep-th/0011269] [INSPIRE].
- [25] E. Dudas, J. Mourad and C. Timirgaziu, Time and space dependent backgrounds from nonsupersymmetric strings, Nucl. Phys. B 660 (2003) 3 [hep-th/0209176] [INSPIRE].
- [26] E. Dudas, G. Pradisi, M. Nicolosi and A. Sagnotti, On tadpoles and vacuum redefinitions in string theory, Nucl. Phys. B 708 (2005) 3 [hep-th/0410101] [INSPIRE].
- [27] I. Basile, J. Mourad and A. Sagnotti, On Classical Stability with Broken Supersymmetry, JHEP 01 (2019) 174 [arXiv:1811.11448] [INSPIRE].
- [28] R. Antonelli and I. Basile, Brane annihilation in non-supersymmetric strings, JHEP 11 (2019)
 021 [arXiv:1908.04352] [INSPIRE].
- [29] I. Basile, On String Vacua without Supersymmetry: brane dynamics, bubbles and holography, Ph.D. Thesis, Scuola Normale Superiore, Pisa Italy (2020) [arXiv:2010.00628] [INSPIRE].
- [30] I. Basile, Supersymmetry breaking, brane dynamics and Swampland conjectures, JHEP 10 (2021) 080 [arXiv:2106.04574] [INSPIRE].
- [31] P. Draper, I.G. Garcia and B. Lillard, Bubble of nothing decays of unstable theories, Phys. Rev. D 104 (2021) L121701 [arXiv:2105.08068] [INSPIRE].
- [32] P. Draper, I. Garcia Garcia and B. Lillard, de Sitter decays to infinity, JHEP 12 (2021) 154 [arXiv:2105.10507] [INSPIRE].

- [33] I.R. Klebanov and A.A. Tseytlin, Gravity duals of supersymmetric SU(N) × SU(N + M) gauge theories, Nucl. Phys. B 578 (2000) 123 [hep-th/0002159] [INSPIRE].
- [34] I.R. Klebanov and M.J. Strassler, Supergravity and a confining gauge theory: Duality cascades and chi SB resolution of naked singularities, JHEP 08 (2000) 052 [hep-th/0007191] [INSPIRE].
- [35] G. Mandal, A Review of the D1/D5 system and five-dimensional black hole from supergravity and brane viewpoint, hep-th/0002184 [INSPIRE].
- [36] A. Dabholkar and S. Nampuri, Quantum black holes, Lect. Notes Phys. 851 (2012) 165 [arXiv:1208.4814] [INSPIRE].
- [37] Y. Hamada, M. Montero, C. Vafa and I. Valenzuela, Finiteness and the swampland, J. Phys. A 55 (2022) 224005 [arXiv:2111.00015] [INSPIRE].
- [38] G. Buratti, J. Calderón and A.M. Uranga, Transplanckian axion monodromy!?, JHEP 05 (2019) 176 [arXiv:1812.05016] [INSPIRE].
- [39] D. Lüst, E. Palti and C. Vafa, AdS and the Swampland, Phys. Lett. B 797 (2019) 134867 [arXiv:1906.05225] [INSPIRE].
- [40] S. Sugimoto, Anomaly cancellations in type I D9 D9 system and the USp(32) string theory, Prog. Theor. Phys. 102 (1999) 685 [hep-th/9905159] [INSPIRE].
- [41] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, The EFT stringy viewpoint on large distances, JHEP 09 (2021) 197 [arXiv:2104.05726] [INSPIRE].
- [42] T.D. Brennan, F. Carta and C. Vafa, The String Landscape, the Swampland, and the Missing Corner, PoS TASI2017 (2017) 015 [arXiv:1711.00864] [INSPIRE].
- [43] E. Palti, The Swampland: Introduction and Review, Fortsch. Phys. 67 (2019) 1900037
 [arXiv:1903.06239] [INSPIRE].
- [44] M. van Beest, J. Calderón-Infante, D. Mirfendereski and I. Valenzuela, Lectures on the Swampland Program in String Compactifications, arXiv:2102.01111 [INSPIRE].
- [45] J.R. Espinosa, A Fresh Look at the Calculation of Tunneling Actions, JCAP 07 (2018) 036 [arXiv:1805.03680] [INSPIRE].
- [46] J.R. Espinosa, Fresh look at the calculation of tunneling actions including gravitational effects, Phys. Rev. D 100 (2019) 104007 [arXiv:1808.00420] [INSPIRE].
- [47] J.R. Espinosa and T. Konstandin, A Fresh Look at the Calculation of Tunneling Actions in Multi-Field Potentials, JCAP 01 (2019) 051 [arXiv:1811.09185] [INSPIRE].
- [48] J.R. Espinosa, Tunneling without Bounce, Phys. Rev. D 100 (2019) 105002
 [arXiv:1908.01730] [INSPIRE].
- [49] J.R. Espinosa, Vacuum Decay in the Standard Model: Analytical Results with Running and Gravity, JCAP 06 (2020) 052 [arXiv:2003.06219] [INSPIRE].
- [50] J.R. Espinosa, The Stabilizing Effect of Gravity Made Simple, JCAP 07 (2020) 061 [arXiv:2005.09548] [INSPIRE].
- [51] J.R. Espinosa and J. Huertas, Pseudo-bounces vs. new instantons, JCAP 12 (2021) 029 [arXiv:2106.04541] [INSPIRE].
- [52] J.R. Espinosa, J.F. Fortin and J. Huertas, Exactly solvable vacuum decays with gravity, Phys. Rev. D 104 (2021) 065007 [arXiv:2106.15505] [INSPIRE].

- [53] J. Polchinski and E. Witten, Evidence for heterotic type-I string duality, Nucl. Phys. B 460 (1996) 525 [hep-th/9510169] [INSPIRE].
- [54] N. Itzhaki, J.M. Maldacena, J. Sonnenschein and S. Yankielowicz, Supergravity and the large N limit of theories with sixteen supercharges, Phys. Rev. D 58 (1998) 046004
 [hep-th/9802042] [INSPIRE].
- [55] E. Silverstein, TASI/PiTP / ISS lectures on moduli and microphysics, in Theoretical Advanced Study Institute in Elementary Particle Physics (TASI 2003): Recent Trends in String Theory, Boulder, U.S.A. 2003, pp. 381–415 [DOI] [hep-th/0405068] [INSPIRE].
- [56] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, Swampland Conjectures for Strings and Membranes, JHEP 02 (2021) 006 [arXiv:2006.15154] [INSPIRE].
- [57] E. Silverstein and A. Westphal, Monodromy in the CMB: Gravity Waves and String Inflation, Phys. Rev. D 78 (2008) 106003 [arXiv:0803.3085] [INSPIRE].
- [58] L.E. Ibáñez and I. Valenzuela, The inflaton as an MSSM Higgs and open string modulus monodromy inflation, Phys. Lett. B 736 (2014) 226 [arXiv:1404.5235] [INSPIRE].
- [59] L. McAllister, E. Silverstein and A. Westphal, Gravity Waves and Linear Inflation from Axion Monodromy, Phys. Rev. D 82 (2010) 046003 [arXiv:0808.0706] [INSPIRE].
- [60] N. Kaloper and L. Sorbo, A Natural Framework for Chaotic Inflation, Phys. Rev. Lett. 102 (2009) 121301 [arXiv:0811.1989] [INSPIRE].
- [61] R. Flauger, L. McAllister, E. Pajer, A. Westphal and G. Xu, Oscillations in the CMB from Axion Monodromy Inflation, JCAP 06 (2010) 009 [arXiv:0907.2916] [INSPIRE].
- [62] F. Marchesano, G. Shiu and A.M. Uranga, F-term Axion Monodromy Inflation, JHEP 09 (2014) 184 [arXiv:1404.3040] [INSPIRE].
- [63] A. Hebecker, S.C. Kraus and L.T. Witkowski, D7-Brane Chaotic Inflation, Phys. Lett. B 737 (2014) 16 [arXiv:1404.3711] [INSPIRE].
- [64] S. Franco, D. Galloni, A. Retolaza and A. Uranga, On axion monodromy inflation in warped throats, JHEP 02 (2015) 086 [arXiv:1405.7044] [INSPIRE].
- [65] J. Calderón-Infante, A.M. Uranga and I. Valenzuela, The Convex Hull Swampland Distance Conjecture and Bounds on Non-geodesics, JHEP 03 (2021) 299 [arXiv:2012.00034] [INSPIRE].
- [66] A. Strominger and C. Vafa, Microscopic origin of the Bekenstein-Hawking entropy, Phys. Lett. B 379 (1996) 99 [hep-th/9601029] [INSPIRE].
- [67] Q. Bonnefoy, E. Dudas and S. Lüst, Weak gravity (and other conjectures) with broken supersymmetry, PoS CORFU2019 (2020) 117 [arXiv:2003.14126] [INSPIRE].
- [68] M. Lüben, D. Lüst and A.R. Metidieri, *The Black Hole Entropy Distance Conjecture and Black Hole Evaporation*, *Fortsch. Phys.* **69** (2021) 2000130 [arXiv:2011.12331] [INSPIRE].
- [69] N. Cribiori, M. Dierigl, A. Gnecchi, D. Lüst and M. Scalisi, Large and Small Non-extremal Black Holes, Thermodynamic Dualities, and the Swampland, arXiv:2202.04657 [INSPIRE].
- [70] T. Ortín, *Gravity and Strings*, 2nd edition, Cambridge Monographs on Mathematical Physics, Cambridge University Press, Cambridge, U.K. (2015) [DOI] [INSPIRE].
- [71] H. Ooguri and C. Vafa, On the Geometry of the String Landscape and the Swampland, Nucl. Phys. B 766 (2007) 21 [hep-th/0605264] [INSPIRE].

- [72] A. Nicolis, On Super-Planckian Fields at Sub-Planckian Energies, JHEP 07 (2008) 023 [arXiv:0802.3923] [INSPIRE].
- [73] F. Baume and E. Palti, Backreacted Axion Field Ranges in String Theory, JHEP 08 (2016)
 043 [arXiv:1602.06517] [INSPIRE].
- [74] I. Valenzuela, Backreaction Issues in Axion Monodromy and Minkowski 4-forms, JHEP 06 (2017) 098 [arXiv:1611.00394] [INSPIRE].
- [75] B. Heidenreich, M. Reece and T. Rudelius, The Weak Gravity Conjecture and Emergence from an Ultraviolet Cutoff, Eur. Phys. J. C 78 (2018) 337 [arXiv:1712.01868] [INSPIRE].
- [76] H. Ooguri, E. Palti, G. Shiu and C. Vafa, Distance and de Sitter Conjectures on the Swampland, Phys. Lett. B 788 (2019) 180 [arXiv:1810.05506] [INSPIRE].
- [77] B. Heidenreich, M. Reece and T. Rudelius, Emergence of Weak Coupling at Large Distance in Quantum Gravity, Phys. Rev. Lett. 121 (2018) 051601 [arXiv:1802.08698] [INSPIRE].
- [78] T.W. Grimm, E. Palti and I. Valenzuela, Infinite Distances in Field Space and Massless Towers of States, JHEP 08 (2018) 143 [arXiv:1802.08264] [INSPIRE].
- [79] P. Corvilain, T.W. Grimm and I. Valenzuela, The Swampland Distance Conjecture for Kähler moduli, JHEP 08 (2019) 075 [arXiv:1812.07548] [INSPIRE].
- [80] T.W. Grimm, C. Li and I. Valenzuela, Asymptotic Flux Compactifications and the Swampland, JHEP 06 (2020) 009 [Erratum ibid. 01 (2021) 007] [arXiv:1910.09549] [INSPIRE].
- [81] N. Gendler and I. Valenzuela, Merging the weak gravity and distance conjectures using BPS extremal black holes, JHEP **01** (2021) 176 [arXiv:2004.10768] [INSPIRE].
- [82] F. Baume and J. Calderón Infante, Tackling the SDC in AdS with CFTs, JHEP 08 (2021) 057 [arXiv:2011.03583] [INSPIRE].
- [83] E. Perlmutter, L. Rastelli, C. Vafa and I. Valenzuela, A CFT distance conjecture, JHEP 10 (2021) 070 [arXiv:2011.10040] [INSPIRE].
- [84] G. Obied, H. Ooguri, L. Spodyneiko and C. Vafa, de Sitter Space and the Swampland, arXiv:1806.08362 [INSPIRE].
- [85] S.K. Garg and C. Krishnan, Bounds on Slow Roll and the de Sitter Swampland, JHEP 11 (2019) 075 [arXiv:1807.05193] [INSPIRE].
- [86] A. Bedroya and C. Vafa, Trans-Planckian Censorship and the Swampland, JHEP 09 (2020) 123 [arXiv:1909.11063] [INSPIRE].
- [87] G. Buratti, J. Calderon, A. Mininno and A.M. Uranga, Discrete Symmetries, Weak Coupling Conjecture and Scale Separation in AdS Vacua, JHEP 06 (2020) 083 [arXiv:2003.09740] [INSPIRE].
- [88] O. DeWolfe, A. Giryavets, S. Kachru and W. Taylor, Type IIA moduli stabilization, JHEP 07 (2005) 066 [hep-th/0505160] [INSPIRE].
- [89] P.G. Cámara, A. Font and L.E. Ibáñez, Fluxes, moduli fixing and MSSM-like vacua in a simple IIA orientifold, JHEP 09 (2005) 013 [hep-th/0506066] [INSPIRE].
- [90] M. Dine, P.J. Fox and E. Gorbatov, Catastrophic decays of compactified space-times, JHEP 09 (2004) 037 [hep-th/0405190] [INSPIRE].

Published for SISSA by 🖉 Springer

RECEIVED: August 2, 2022 ACCEPTED: August 8, 2022 PUBLISHED: August 29, 2022

Dynamical Cobordism and the beginning of time: supercritical strings and tachyon condensation

Roberta Angius, Matilda Delgado and Angel M. Uranga

Instituto de Física Teórica IFT-UAM/CSIC, C/ Nicolás Cabrera 13-15, Campus de Cantoblanco, 28049 Madrid, Spain E-mail: roberta.angius@csic.es, matilda.delgado@uam.es, angel.uranga@csic.es

ABSTRACT: We describe timelike linear dilaton backgrounds of supercritical string theories as time-dependent Dynamical Cobordisms in string theory, with their spacelike singularity as a boundary defining the beginning of time. We propose and provide compelling evidence that its microscopic interpretation corresponds to a region of (a strong coupling version of) closed tachyon condensation. We argue that this beginning of time is closely related to (and shares the same scaling behaviour as) the bubbles of nothing obtained in a weakly coupled background with lightlike tachyon condensation. As an intermediate result, we also provide the description of the latter as lightlike Dynamical Cobordism.

KEYWORDS: String and Brane Phenomenology, Tachyon Condensation, Bosonic Strings

ARXIV EPRINT: 2207.13108





Contents

1	Introduction	1
2	Overview of Dynamical Cobordisms	3
3	Supercritical strings as time-dependent Dynamical Cobordism	4
	3.1 Linear timelike dilaton as Dynamical Cobordism	5
	3.2 The ETW brane at the beginning of time	5
	3.3 The timelike tachyon case	7
4	Lightlike tachyon condensation	8
	4.1 Lightlike tachyon in the worldsheet description	8
	4.2 Spacetime description and lightlike ETW brane	10
	4.2.1 Effective action	10
	4.2.2 The local scalings	11
	4.2.3 General f_1	13
5	The strong coupling region and the origin of time	14
6	Conclusions	17
\mathbf{A}	Dimension quenching as an interpolating domain wall	18
в	The partition function with a lightlike tachyon background	19
\mathbf{C}	The critical exponent for general f_1	21

1 Introduction

One of the outstanding questions in string theory is the understanding of time-dependent backgrounds and in particular the resolution of cosmological (i.e. spacelike) singularities (see [1, 2] for reviews). On general grounds, and in analogy with timelike singularities, one may expect that stringy effects smooth out the singularity, thus providing a microscopic description of the beginning of time.

This is a natural proposal from the perspective of the Swampland Cobordism Conjecture [3], which states that any consistent theory of quantum gravity should admit configurations ending spacetime, namely boundaries or general cobordism defects leading to walls of nothing.¹ This also resonates with (a Lorentzian version of) the no-boundary proposal for the Hartle-Hawking wavefunction of the universe [9].

¹A prototype is the infinite volume limit of bubbles of nothing in higher-dimensional compactifications in which the internal space shrinks to zero size [4] (see [5-8] for some recent work).

From this perspective, such cosmological solutions would correspond to dynamical time-dependent configurations with a beginning of time given by a cobordism defect extending in the spatial directions. This appealing picture is however hampered by the general lack of understanding of the microscopic structure of spacelike singularities.

The cobordism conjecture has been exploited at the topological level with interesting results, see e.g. [6, 10-15]. On the other hand, there is substantial progress in understanding the implications of the cobordism conjecture at the dynamical level. The configurations dubbed Dynamical Cobordisms in [16-18] (see also [19])² describe spacetime dependent solutions in which the fields run until they hit a real-codimension 1 singularity at finite distance in spacetime, at which certain scalars run off to infinite distance in field space. In several examples of such spatially varying solutions, the timelike singularities had a known string theory UV description, which displayed an end of spacetime. Remarkably, [18] showed that in the effective theory description these singularities (dubbed end-of-the-world (ETW) branes) follow universal scaling laws, and are characterized by a single critical exponent.

In this paper we take the natural next step of starting the study of time-dependent Dynamical Cobordism with spacelike singularities and of shedding some light on their resolution. The particular arena to explore these ideas are timelike linear dilaton backgrounds in supercritical bosonic string theory.

Supercritical string theories provide consistent versions of string theory in a general number D of spacetime dimensions, provided a suitable timelike linear dilaton background is turned on [30–33]. They provide an excellent testing ground for general features of string theory (see [34] for a recent example). In particular, and as will be relevant to our discussion, they constitute a setup in which closed string tachyon physics has been subject to quantitative analysis (see e.g. [35–41]).³ For our purposes, the main property of these theories is that the timelike linear dilaton background makes them one of the simplest time-dependent setups in string theory.

We express these backgrounds as time-dependent Dynamical Cobordisms, exhibiting their beginning of time singularity and characterizing it as an ETW brane, with a precise critical exponent. We moreover propose, providing non-trivial support for it, that the stringy resolution of the singularity involves a region of (the strong coupling version of) bulk tachyon condensation. This is a realization of the mechanism in [45] in a different setup which, as promised, provides a stringy analogue of the Hartle-Hawking proposal.

Our approach is based on the realization that the beginning of time singularity, and the walls of nothing described via lightlike tachyon condensation in [35] (see also [36–38, 46–48] for related results) admit an ETW brane description in the effective theory with exactly the same critical exponent. Moreover, we show that these configurations, which seemingly contain two intersecting ETW walls, actually contain a single recombined one with two

²For the related topic of solutions in theories with dynamical tadpoles, see [20-23] for early work and [24-29] for related recent developments.

 $^{{}^{3}}$ See [42–44] and references therein for discussion of the fate of localized closed tachyons and related instabilities.

different asymptotic regions, a lightlike one corresponding to tachyon condensation at weak coupling and a spacelike one at strong coupling corresponding to the beginning of time.

A potential caveat to our analysis is the use of effective theories to describe tachyon condensation phenomena, which involve stringy scales and are not fully understood for closed tachyons (see [49] and references therein for further discussion). We however encounter that the main feature of the ETW wall, the critical exponent, is surprisingly robust under corrections of the effective action. This suggest that the main results may survive beyond the validity of the tools used to extract it in the present work. The same considerations apply to the study of the beginning of time singularity, which lies at strong coupling.

The paper is organized as follows. In section 2 we recall the Dynamical Cobordisms of [16-18], and the structure of the ETW branes in terms of their critical exponent. In section 3 we discuss the timelike linear dilaton background as a time-dependent solution: in section 3.1 we express it as a Dynamical Cobordism with a beginning of time; in section 3.2 we describe the singularity at the beginning of time as an ETW brane; and in section 3.3 we explore its UV description in terms of a timelike tachyon condensate. In section 4 we discuss walls of nothing arising in lightlike tachyon condensation and show that they correspond to Dynamical Cobordisms with a lightlike ETW brane: in section 4.1 we recall the worldsheet description, and in section 4.2 we provide their spacetime description and characterize their ETW brane and critical exponent. In section 5 we combine results and formulate our proposal that the UV description of the beginning of time in the linear dilaton background is (a strong coupling version of) closed tachyon condensation. In section 6 we offer some final thoughts. In appendix A we mention that the dimension quenching mechanism in [36, 37] can be described as a dynamical cobordism describing an interpolating wall [17] between theories of different dimension. Some calculational details have been postponed to appendices B, C.

2 Overview of Dynamical Cobordisms

In a series of papers [16–18] the analysis of dynamical spacetime-dependent solutions realizing cobordisms to nothing was initiated (see also [19]). Such solutions, from the perspective of the lower-dimensional effective field theory, present universal features that allow them to be described in a general framework as follows.

Consider the lower-dimensional EFT to be *D*-dimensional⁴ Einstein gravity coupled to a scalar with arbitrary potential (in $M_{Pl} = 1$ units):

$$S = \int d^{D}x \sqrt{-g} \left(\frac{1}{2}R - \frac{1}{2} (\partial \phi)^{2} - V(\phi) \right) .$$
 (2.1)

We consider solutions in which the metric and scalar vary along one coordinate, denoted by y. The ansatz for the metric is

$$ds^{2} = e^{-2\sigma(y)} ds^{2}_{D-1} + dy^{2}.$$
(2.2)

⁴We use D for the spacetime dimension and \mathcal{D} for the field space distance.

Here we follow earlier references and considered space-dependent running solutions. The sign flips necessary for time-dependent ones will be taken into account in the concrete examples of later sections.

In the following we take a flat metric for the D-1 dimensional slices. All solutions that describe a cobordism to nothing present a spacetime singularity at finite spacetime distance Δ where the scalars explore an infinite distance \mathcal{D} in field space, this is the location of the ETW wall. Indeed, the solution does not extend beyond this point which, without loss of generality, we choose to be y = 0.

One of the highlights of the analysis in [18] is that the solutions near ETW branes behave in a simple way. We quote some of the main expressions encapsulating this

$$\phi(y) \simeq -\frac{2}{\delta} \log y \quad , \quad \sigma(y) \simeq -\frac{4}{(D-2)\delta^2} \log y \quad , \quad |R| \simeq \frac{1}{y^2} \,. \tag{2.3}$$

with δ a scaling coefficient which characterizes the local solution near the ETW brane, and |R| is the spacetime scalar curvature. Although [18] focused on space-dependent running solutions, it is straightforward to extend the discussion to time-dependent ones, and recover the same scaling laws.

From the above profiles, all solutions describing ETW walls present universal scaling relations between Δ , \mathcal{D} and the spacetime scalar curvature |R|, as follows

$$\Delta \sim e^{-\frac{\delta}{2}\mathcal{D}} \quad , \quad |R| \sim e^{\delta \mathcal{D}} \,. \tag{2.4}$$

We also get that the scalar potential behaves as^5

$$V(\phi) \simeq -a c e^{\delta \phi}, \qquad (2.5)$$

for a constant a < 1 related to δ by

$$\delta = 2\sqrt{\frac{D-1}{D-2}(1-a)} \,. \tag{2.6}$$

3 Supercritical strings as time-dependent Dynamical Cobordism

In this section we discuss the maximally symmetric configuration of supercritial strings, and interpret the necessary linear dilaton background as a running solution which satisfies the properties of a time-dependent Dynamical Cobordism. The local behaviour is hence that of an ETW brane.

In this work we focus on the supercritical bosonic theory. We expect similar ideas to apply to other supercritical theories, including supercritical type 0 or heterotic superstrings [36, 37].

⁵Note that if y is timelike, the overall sign of the potential changes. The quantity c is a positive constant related to the boundary condition used when solving the equations of motion. Subleading corrections to the potential can be included by promoting c to a function $c = c(\phi)$ with slower growth than an exponential.

3.1 Linear timelike dilaton as Dynamical Cobordism

Consider bosonic string theory in *D*-dimensional Minkowski space (in the string frame). In order to satisfy the central charge constraint for the theory, there is a linear dilaton background

$$\Phi = v_M X^M \,, \tag{3.1}$$

with

$$v \cdot v = -\frac{D - 26}{6\alpha'},\tag{3.2}$$

with contractions defined with respect to the flat Minkowski metric.

Hence, supercritical strings require a timelike dilaton gradient, whereas subcritical strings require an spacelike one. The critical D = 26 theory does not require a dilaton profile for consistency, but does admit a lightlike dilaton background. We thus expect our discussion to extend this background of the critical theory as well.

These linear dilaton theories define conformal theories exactly in α' , which implies that they satisfy the equations of motion of the spacetime (string frame) action

$$S_{\text{str.}} = \frac{1}{2} \int d^D x \sqrt{-G_{(s)}} e^{-2\Phi} \left[-\frac{2(D-26)}{3\alpha'} + R_{(s)} + 4(\partial\Phi)^2 \right].$$
(3.3)

We explicitly denote string frame quantities with an s subindex, while quantities with no subindex are implicitly defined in the Einstein frame.

In the following we focus on supercritical strings and timelike dilaton background

$$\Phi = -qX^0, \qquad (3.4)$$

where $q \equiv v^0$. Here we have absorbed a possible additive constant by shifting time, so that the dilaton vanishes at $X^0 = 0$. From the two solutions of (3.2) we choose the one leading to weak coupling $g_s = e^{\Phi}$ in the future $X^0 \to \infty$, namely

$$q = \sqrt{\frac{D - 26}{6\alpha'}} \ . \tag{3.5}$$

In the next section, we reinterpret this linear dilaton background as a running solution with an ETW wall at the origin of time.

3.2 The ETW brane at the beginning of time

The spacetime physics of the singularity was considered from a cosmological perspective in [35]. Here we instead study it from the perspective of the ETW branes of dynamical cobordism in section 2.

To discuss the spacetime physics, we focus on the Einstein frame, so the metric reads

$$ds^{2} = \exp\left(\frac{4qX^{0}}{D-2}\right) \eta_{MN} dx^{M} dx^{N} .$$
(3.6)

We see that at $X^0 \to -\infty$ the warp factor goes to zero, and we hit a singularity. We can introduce a time coordinate y giving the invariant interval to the singularity as

$$y = \frac{D-2}{2q} \exp\left(\frac{2qX^0}{D-2}\right), \qquad (3.7)$$

in terms of which (3.6) is recast as the time-dependent version of (2.2):

$$ds^{2} = -dy^{2} + \frac{4q^{2}y^{2}}{(D-2)^{2}} dx^{m} dx^{m}, \qquad (3.8)$$

for $m = 1, \ldots, D - 1$. We thus obtain

$$\sigma(y) = -\log y \,. \tag{3.9}$$

Comparing this to (2.3) gives:

$$\delta = \frac{2}{\sqrt{D-2}} \,. \tag{3.10}$$

Expressing the dilaton in terms of a scalar ϕ with canonical kinetic term in the Einstein frame as

$$\phi = \frac{2}{\sqrt{D-2}} \Phi \sim -\sqrt{D-2} \log y \,. \tag{3.11}$$

This is precisely the scaling relation for the scalar (2.3) for the value of δ in (3.10).

We also get the expected scaling of the potential. The Einstein frame action gives

$$S = \frac{1}{2} \int d^D x \sqrt{-G} \left[R - \frac{4}{D-2} \left(\partial \Phi \right)^2 - \frac{2(D-26)}{3\alpha'} \exp\left(\frac{4\Phi}{D-2}\right) \right], \quad (3.12)$$

which, comparing with (2.5) and using the normalized dilaton (3.11) yields the precise value of δ in (3.10).

To conclude, we recover the scaling relations (2.4), which state that the configuration hits an ETW singularity at finite time in the past at which the scalar runs off to infinite distance in field space. According to the cobordism interpretation of such singularities in [17, 18], it defines a beginning of time, a boundary in the time direction, for this solution.

The microscopic description of the ETW brane requires some understanding of spacelike defects in string theory, which remains mostly *terra incognita*.⁶ In our particular example, this is even more so since it lies at strong coupling.⁷ Despite these difficulties, we find compelling evidence that the microscopic description of our spacelike ETW brane is the strong coupling avatar of tachyon condensation. We propose a direct approach to this proposal in the next section, and a further indirect, but quantitatively more reliable, route to support this picture in section 4.

⁶See e.g. [50] for attemps invoking S-branes.

⁷For the critical type IIA with a lightlike dilaton background, a microscopic description for the analogous singularity was proposed in [51], based on M(atrix) theory [52]. Such a description does not seem feasible in our case.

3.3 The timelike tachyon case

The resolution of spacelike singularities in a tractable worldsheet approach was addressed in [45] in a setup with a shrinking 1-cycle, in terms of the condensation of a closed string tachyon in the winding sector (see [53, 54] for proposed higher-genus generalizations). In short, the regime near the singularity was proposed to be coated by a longer duration region in which the tachyon condenses with an exponential profile. The latter describes an effective Liouville wall in the time direction, beyond which no string excitation can propagate. This was argued to be a stringy definition of the *nothing* in the Hartle-Hawking description of the wavefunction of the universe [9]. In this picture, spacetime emerges smoothly as the tachyon turns off. In our terms, it describes a cobordism to nothing in the time direction.

In this section we explore a similar interpretation for the spacelike singularity encountered in our timelike linear dilaton setup. The idea is to consider an exponential profile for the closed string tachyon of supercritical bosonic theory. The tachyon couples to the worldsheet as a 2d potential. The condition for this deformation to be marginal, to linear order in conformal perturbation theory, or equivalently, the linearized spacetime equation of motion for the tachyon is

$$\partial^2 T(X) - 2v^M \partial_M T(X) + \frac{4}{\alpha'} T(X) = 0.$$
(3.13)

We will discuss corrections to this later on.

For a general tachyon exponential profile

$$T(X^M) = \mu \exp(\beta_M X^M) . \qquad (3.14)$$

we obtain a condition on β :

$$\beta \cdot \beta - 2v \cdot \beta + \frac{4}{\alpha'} = 0.$$
(3.15)

We now focus on a timelike tachyon profile

$$T = \mu \exp(-\beta^0 X^0),$$
 (3.16)

with the condition

$$-(\beta^0)^2 + 2q\beta^0 + \frac{4}{\alpha'} = 0.$$
(3.17)

There are two solutions to this quadratic equation. A possibility is to choose $\beta^0 < 0$, so that the tachyon grows for late times $X^0 \to \infty$. This is a good strategy to study the process of tachyon condensation in a weakly coupled regime, see e.g. [38]. In fact, it is closely related to our approach (albeit for lightlike tachyons) in section 4.

Here, instead, we are interested in having tachyon condensation at the beginning of time, to provide a resolution of the spacelike ETW brane at y = 0, hence we need the tachyon to grow in the past $X^0 \to -\infty$, we thus require $\beta^0 > 0$. Using (3.5) we have

$$\beta^{0} = \frac{\sqrt{D - 26} + \sqrt{D - 2}}{\sqrt{6\alpha'}}.$$
(3.18)

We may now compare the relative growth of the string coupling $g_s = e^{\Phi}$ and of the tachyon as $X^0 \to -\infty$ to assess if the tachyon condensation could be studied using world-sheet techniques. We have

$$T/g_s = \mu \exp\left(-\sqrt{\frac{D-2}{6\alpha'}}X^0\right) \,. \tag{3.19}$$

This shows that the tachyon grows parametrically faster than the string coupling as $X^0 \rightarrow -\infty$. This leads to the expectation that the worldsheet analysis provides a reliable description of the physics at early times. In analogy with [45], and based on the extensive analysis in [35–37], the presence of the worldsheet potential creates a Liouville wall expelling all string excitations, providing a microscopic definition of an ETW brane in time.

The drawback of this approach is that it relied on trusting the linearized deformation approximation, which is expected to experience strong higher order corrections.⁸ Therefore the scenario can be at most regarded as a qualitative description. In the next section we turn to a different approach, involving α' exact solutions.

4 Lightlike tachyon condensation

We are thus led to consider solutions under better control. In this section we consider an α' -exact solution of the supercritical linear dilaton theory with tachyon profile along a lightlike direction. As established in [35, 36] for the bosonic theory, at late times this leads to a wall of nothing moving at the speed of light, analogous to the asymptotic behaviour of a bubble of nothing. After recalling the argument, we carry out a new spacetime analysis that shows that at late times the background corresponds to a lightlike ETW brane, and show that its critical exponent is exactly the same as for the beginning of time ETW brane of the previous section. This tantalizing relation is a strong support for our interpretion of the beginning of time is (a strongly coupled version of) a closed string tachyon condensation phase, discussed in section 5.

4.1 Lightlike tachyon in the worldsheet description

Consider introducing an exponential tachyon background (3.14) along a lightlike direction $X^+ = (X^0 + X^1)/\sqrt{2}$

$$T = \mu \exp(\beta X^+). \tag{4.1}$$

The linearized tachyon marginality condition (3.13) is satisfied for

$$\beta = \frac{2\sqrt{2}}{q\alpha'}.\tag{4.2}$$

At late times $X^0 \to \infty$, the string coupling is small and one may perform a reliable worldsheet analysis. As shown in [35] and contrary to the timelike tachyon case, the

⁸For the $\beta^0 < 0$ solution, corrections are interestingly expected to be suppressed in a large *D* approximation, as exploited in [55]. Although large *D* could be interesting in our $\beta^0 > 0$ case to increase the hierarchy in (3.19), it does not lead to a similar suppression of corrections.

deformation by the operator (4.1) is exact, as higher order corrections in the perturbation vanish, since the lightlike nature of the insertions prevent the existence of non-trivial Wick contractions. Furthermore, in light-cone coordinates the propagator of the $X^{+/-}$ fields is oriented from X^+ to X^- and we know that all interaction vertices introduced by the tachyon potential only depend on X^+ . These two facts combined show that there are no possible Feynman diagrams beyond tree-level, which implies the solution is exact in α' . One can thus conclude that the linearized tree-level solution (4.1) is exactly conformally invariant.

The tachyon couples as a worldsheet potential, which grows infinitely at $X^+ \to \infty$. This 2d potential prevents any string modes from entering the corresponding region, which thus becomes a region of nothing. The physical interpretation of this is that the tachyon configuration describes a wall of nothing propagating at the speed of light, which effectively ends spacetime at an effective value of X^+ .

The finite range in X^+ can be estimated by e.g. cutting off X^+ when the T = 1. This gives

$$\Delta X^{+} = -\beta^{-1} \log \mu / \mu^{*}, \qquad (4.3)$$

where μ^* defines a reference position from which we measure the range to the wall. A more precise derivation follows from the *gedanken* experiment of solving the motion of classical strings incoming into the tachyon wall [35]. The initial speed reduces to zero at a turning point, after which the string is pushed back by the tachyon wall and its speed asymptotes to that of light. The turning point position in X^+ in the formulas in [35] gives back the result (4.3).

Given the importance of the notion of finiteness on the location of the tachyon wall, we provide an alternative derivation, carried out by adapting the techniques in [45]. We briefly sketch the results here and give more computational details in appendix B. Decomposing the field $X^+(\tau, \sigma)$ into its zero and nonzero modes $X^+(\tau, \sigma) = X_0^+ + \hat{X}^+(\tau, \sigma)$ and performing a Wick rotation, the Euclidean partition function reads:

$$Z(\mu) = \int dX_0^+ \int \mathcal{D}\hat{X}^+ \mathcal{D}X^- \mathcal{D}X^i \mathcal{D}g\mathcal{D}(\text{ghosts})e^{-S_E^{\text{deformed}}}, \qquad (4.4)$$

with the Euclidean action:

$$S_E^{\text{deformed}} = \frac{1}{2\pi\alpha'} \int d^2 \sigma_E \sqrt{g} \left[\partial_{\sigma^0} \hat{X}^+ \partial_{\sigma^0} X^- + \partial_{\sigma^1} X^- \partial_{\sigma^1} \hat{X}^+ + \partial_{\alpha} X^i \partial_{\alpha} X^i \right] + \frac{1}{2\pi} \int d^2 \sigma_E \sqrt{g} R_2 \Phi(X) + \frac{i\mu_E}{2\pi} \int d^2 \sigma_E \sqrt{g} e^{\beta X^+} .$$

$$(4.5)$$

From this we see that when the tachyon condenses, at large X^+ , the path integral becomes suppressed. This results in a truncation of contributions to the integral coming from string oscillations with $X^+ \mapsto \infty$. This is the same mechanism as that of a Liouville wall in Liouville theory: no physical degrees of freedom exist in this region. In fact, one can show that the partition function in (4.4) can be directly related to that of the free theory (with no tachyon deformation) as follows. After integrating out the zero-mode X_0^+ , one can show that:

$$\frac{\partial Z}{\partial \mu_E} = -\frac{1}{\beta \mu_E} \int \mathcal{D}\hat{X}^+ \mathcal{D}X^- \mathcal{D}X^i \mathcal{D}g\mathcal{D}(\text{ghosts})e^{-S_E^{\text{free}}}, \qquad (4.6)$$

where S_E^{free} is the euclidean action of the worldsheet theory without the tachyon potential. Integrating with respect to μ and fixing a cutoff for X^+ such that $\mu_* = e^{\beta X_*^+}$, we obtain:

$$Z_1 = -\frac{\log(\mu_E/\mu_*)}{\beta}\widehat{Z}, \qquad (4.7)$$

where \hat{Z} is the partition function for the 2d theory without the tachyon insertion. Hence the partition function Z in the presence of the tachyon background related to that of the theory without the tachyon \hat{Z} via the factor $\frac{\log(\mu_E/\mu_*)}{\beta}$, which thus provides an effective "size" of the direction X^+ , which matches that of (4.3).

The interpretation of the exponential tachyon as a wall of nothing receives further support from the dimension quenching mechanism in [36]. In appendix A we review it from the perspective of dynamical cobordisms in the spacetime perspective, to be discussed next.

4.2 Spacetime description and lightlike ETW brane

In this section we study the spacetime description of the wall of nothing corresponding to the lightlike tachyon, and show that it satisfies the properties of (the lightlike version of) and ETW brane. This nicely confirms the worldsheet arguments of the previous section.

4.2.1 Effective action

In order to describe the spacetime dynamics of the lightlike tachyon configuration, we need an effective spacetime action for the relevant fields, in particular for the tachyon. This is already a subtle point, since tachyon condensation processes may in principle backreact on the whole tower of stringy states, hence the validity of the truncation to an effective theory with a finite set of fields is to some extent questionable.

In any event, this approach has been successful enough in open string tachyon effective actions, and we may venture into its use for the closed case, hoping that fortune favors the brave.

The construction of the most general 2-derivative effective action for the metric, dilaton and tachyon in supercritical string theory has been discussed in [35] and [38], whose discussion we follow. In the string frame it has the structure

$$S = \frac{1}{2} \int d^D x \sqrt{-G_{(s)}} e^{-2\Phi} \left[f_1 R_{(s)} + 4f_2 (\nabla \Phi)^2 - f_3 (\nabla T)^2 - 2f_4 - f_5 \nabla T \cdot \nabla \Phi \right] .$$
(4.8)

where the $f_i(T)$ are general functions of the tachyon. By demanding that the equations of motion are compatible with the linear dilaton background with an exponential tachyon profile, one can show that the $f_i(T)$ can be expressed in terms of $f_1(T)$:

$$f_{2}(T) = f_{1}(T), \qquad f_{3}(T) = -f_{1}''(T) - \frac{f_{1}'(T)}{T}, \qquad f_{5}(T) = 4f_{1}'(T),$$

$$f_{4} = \frac{1}{2} \left[4f_{1}(T) \left(\frac{D-26}{6\alpha'} \right) + Tf_{1}'(T) \left(\beta \cdot \beta + \frac{8}{\alpha'} \right) - T^{2}f_{1}''(T) \beta \cdot \beta \right].$$
(4.9)

For general exponential tachyon profiles, the tachyon background is only a solution at linearized order, hence we expect the above relations to receive corrections. For lightlike tachyons, however, the solution is exact in worldsheet perturbation theory, hence the above relations hold, and the corrections at most modify the behaviour of f_1 at large T. Note also that for lightlike tachyon profiles $\beta \cdot \beta = 0$ and the tachyon potential $V_{(s)} = f_4$ becomes β -independent.

Going to the Einstein frame, we redefining the metric to absorb the f_1 prefactor as well as the usual dilaton factor,

$$(G_{(s)})_{MN} = e^{\frac{4}{D-2}\Phi} f_1^{-\frac{2}{D-2}} G_{MN}, \qquad (4.10)$$

the spacetime action is

$$S = \frac{1}{2} \int d^{D}x \sqrt{-G} \left[R - \frac{4}{D-2} \left(\partial_{M} \Phi \partial^{M} \Phi - \frac{f_{1}'}{f_{1}} \partial_{M} \Phi \partial^{M} T \right) - \left[\frac{D-1}{D-2} \frac{f_{1}'^{2}}{f_{1}^{2}} - \frac{f_{1}''}{f_{1}} - \frac{f_{1}'}{f_{1}T} \right] \partial_{M} T \partial^{M} T - \frac{2}{3\alpha'} e^{\frac{4\Phi}{D-2}} f_{1}^{\frac{-D}{D-2}} \left((D-26)f_{1} + 12Tf_{1}' \right) \right].$$

$$(4.11)$$

The complete expression for f_1 is actually not known, beyond its expansion around T = 0

$$f_1 = 1 - T^2 + \dots \tag{4.12}$$

Nevertheless, [38] proposed a set of regularity conditions on the effective action, which to some extent constrain f_1 further, and several explicit solutions were proposed, concretely $f_1 = \exp(-T^2)$ and $f_1 = 1/\cosh(\sqrt{2}T)$. Interestingly, in the large T regime (which is near the wall of nothing, our main focus), both can be parametrized as

$$f_1 \sim A \exp\left(-b T^k\right). \tag{4.13}$$

Actually, an outcome of [38] is that the behaviour of the system is not particularly sensitive to the precise form f_1 . In the following we focus on the dependence (4.13), but later show that the same results hold, even at quantitative level, for very general forms of f_1 .

4.2.2 The local scalings

We propose that the lightlike tachyon background, in the weak coupling regime, corresponds to a dynamical cobordism in X^+ , and that the tachyon wall corresponds to an ETW brane, namely a singularity in effective theory at finite spacetime distance, and at which some scalar runs of to infinite field theory distance. In the following we show that the scalings derived from the Einstein frame spacetime solution are indeed of the ETW kind.

Note that, because the dynamical cobordism takes place via dependence on the lightlike coordinate X^+ , in order to discuss spacetime *distance*, we choose slices of constant X^0 , and measure spatial distance along X^1 , along which the dilaton remains constant.

Again, recall that we focus on the dependence (4.13), but similar conclusions hold for a very general class of profiles of f_1 . The running scalar along X^1 is only the tachyon, hence the distance in field space as we approach the wall is given by:

$$\mathcal{D} = \int^{T_{\rm ETW}} \left(\sqrt{\frac{f_1''}{f_1} + \frac{f_1'}{f_1} \left(\frac{1}{T} - \frac{D-1}{D-2} \frac{f_1'}{f_1} \right)} \right) dT \sim \frac{b}{\sqrt{D-2}} T_{\rm ETW}^k, \qquad (4.14)$$

which diverges (for k > 1) since the tachyon goes to infinity at the ETW brane at $X^1 \to \infty$.

Let us now check that the wall is indeed at finite spacetime distance in the Einstein frame. The length along X^1 is

$$\Delta = \int^{\text{ETW}} f_1^{\frac{1}{D-2}} dx^1 = A^{\frac{1}{D-2}} \int^{\text{ETW}} \exp\left(-\frac{bT^k}{D-2}\right) dx^1$$

$$= \frac{\sqrt{2}}{bk} A^{\frac{1}{D-2}} \int^{\text{ETW}} \exp\left(-\frac{\mathcal{D}}{\sqrt{D-2}}\right) \frac{d\mathcal{D}}{\mathcal{D}} = \frac{\sqrt{2}}{bk} A^{\frac{1}{D-2}} \operatorname{Ei}\left(-\frac{\mathcal{D}}{\sqrt{D-2}}\right), \qquad (4.15)$$

where this last function is the exponential integral, and is clearly convergent, showing that the tachyon background behaves as dynamical cobordism ending at an ETW brane, where the (tachyon) scalar runs off to infinite field space distance at a finite spacetime distance.

We can check the scaling relations of ETW branes of section 2. We can expand (4.15) for $\mathcal{D} \to \infty$ as

$$\operatorname{Ei}\left(-\frac{\mathcal{D}}{a}\right) \sim e^{-\mathcal{D}/a}\left(\frac{a}{\mathcal{D}}+\ldots\right),$$
(4.16)

and get

$$\Delta \sim \exp\left[-\frac{\mathcal{D}}{\sqrt{D-2}} - \log\left(\frac{\mathcal{D}}{\sqrt{D-2}}\right)\right].$$
(4.17)

Comparing this with (2.4) gives a value

$$\delta = \frac{2}{\sqrt{D-2}} \,. \tag{4.18}$$

Namely, we recover an exponential relation. It is interesting to point out that the log correction is reminiscent of that encountered in [17, 18] for the EFT strings in [56]. Also note that, restricting to the leading exponential scaling, the critical exponent is independent of k. This is a particular case of the claimed robustness of the results under changes of f_1 , and will be explored in general in section 4.2.3.

We can also compute the scaling of the Ricci scalar, which, upon direct computation gives

$$|R| \sim \exp\left[\frac{2\mathcal{D}}{\sqrt{D-2}} + \log\left(\frac{\mathcal{D}^2}{D-2}\right)\right].$$
 (4.19)

Again, the leading terms gives the scaling corresponding to an ETW brane, with δ given by (4.18), again remarkably independent of k.
The potential, computed in the limit $T \mapsto \infty$ with (4.11), also agrees nicely with the general formula provided by the local analysis in (2.5):

$$V(T) = -\frac{8}{\alpha'} A^{-\frac{2}{D-2}} k b T^k e^{\frac{4\phi}{D-2}} e^{\delta \mathcal{D}} = -a c(T) e^{\delta \mathcal{D}}, \qquad (4.20)$$

with $a \in [0, 1]$ and the subleading polynomial correction can be absorbed by the function c(T).⁹

Even more remarkably, the value of δ (4.18) for the lightlike tachyon agrees with the critical exponent (3.10) of the ETW brane at the beginning of time of the linear dilaton solution. This shows that both kinds of ETW branes are very similar, and is strongly suggestive that they may admit similar microscopic descriptions. Hence, we claim that the singularity at the beginning of time is a dynamical cobordism to nothing triggered by (the strong coupling version of) the condensation of the closed string tachyon. We look deeper into this argument in section 5 but before then, we show that this surprising matching of the critical exponents holds for general profiles of f_1 .

4.2.3 General f_1

Let us now show that the above structure, and in particular the same value for the critical exponent δ , holds for general f_1 under very mild conditions. In particular we demand that f_1 decays at large T faster then 1/T. This is a very reasonable requirement, in particular notice that this ensures the convergence of the integral for the spacetime distance Δ to the ETW brane. Hence it implements the intuition that the wall of nothing propagating at the spped of light hits in finite time any point at finite spacetime distance.

Consider now the integral for the field space distance

$$\mathcal{D} = \int \left[\frac{f_1''}{f_1} + \frac{f_1'}{f_1} \left(\frac{1}{T} - \frac{D-1}{D-2} \frac{f_1'}{f_1} \right) \right]^{\frac{1}{2}} dT.$$
(4.21)

We start massaging the integrand of \mathcal{D} , by noticing that

$$\frac{f_1''}{f_1} + \frac{f_1'}{f_1} \left(\frac{1}{T} - \frac{D-1}{D-2}\frac{f_1'}{f_1}\right) = \left(\frac{f_1'}{f_1}\right)' + \frac{f_1'}{Tf_1} + \frac{1}{D-2}\left(\frac{f_1'}{f_1}\right)^2, \qquad (4.22)$$

it is easy to show that for f_1 decaying faster than 1/T, the dominant term is the last one. In fact one can see by considering different profiles (e.g. power-law, exponential, exponential of an exponential, etc) that, the faster the decay, the more the last terms dominates. Then

$$\mathcal{D} \sim \int \frac{1}{\sqrt{D-2}} \frac{f_1'}{f_1} dT.$$
 (4.23)

We may write this as

$$d\mathcal{D} = \frac{1}{\sqrt{D-2}} \frac{f_1'}{f_1} dT = \frac{1}{\sqrt{D-2}} d\log f_1 = \sqrt{D-2} \ d\log f_1^{\frac{1}{D-2}} . \tag{4.24}$$

⁹More details about such subleading corrections can be found in the appendix B of [18].

Namely

$$f_1^{\frac{1}{D-2}} = \exp(-\mathcal{D}/\sqrt{D-2}),$$
 (4.25)

where we have chosen the appropriate sign for the distance to be positive (recall that f_1 is a function that decreases to zero). This allow to express the spacetime distance as

$$\Delta \sim \int f_1^{1/(D-2)} dx^1 \sim \int \exp(-\mathcal{D}/\sqrt{D-2}) \frac{dT}{T} \,. \tag{4.26}$$

This has a similar structure to the intermediate expression in (4.15). Similar to the exponential integral there, the above integral behaves just like the exponential in the integrand, leading to the scaling

$$\Delta = \exp(-\mathcal{D}/\sqrt{D-2}), \qquad (4.27)$$

which reproduces the value of δ in (4.18). Indeed, one can check that the additional terms in the integrand lead to subleading corrections, of the kind in (4.17) (for a proof of this statement under mild assumptions, we refer the reader to appendix C).

5 The strong coupling region and the origin of time

In this section we argue that the microscopic description of the ETW brane at the beginning of time is a region of (the strong coupling version of) closed string tachyon condensation.

The ETW brane recombination. Let us now consider the full lightlike tachyon configuration, including the strongly coupled region, and consider the interplay of the two ETW branes we have encountered.

In the string frame variables there are two asymptotic regions, controlled by seemingly different physics. The first is the region $X^0 \to -\infty$, with X^1 finite (hence $X^+ \to -\infty$), which corresponds to a linear timelike dilaton configuration, with negligible tachyon background. The second is the region $X^+ \to \infty$ at finite X^1 (hence $X^0 \to \infty$), which corresponds to a lightlike tachyon configuration at weak string coupling. Both regions are disjoint, as they only coincide at infinity in $X^0 \to -\infty$, $X^+ \to \infty$ (hence we need $X^1 \to \infty$).

In the Einstein frame, these asymptotic regimes turn into singularities at finite distance in spacetime, triggered by the running off of suitable scalars (the tachyon or the dilaton) to infinite distance in field space. Following the dynamical cobordism interpretation advocated in [16–18], these are ETW branes chopping off the region of spacetime beyond them.

An important observation is that the effective theory in which one describes ETW branes is not valid at arbitrarily short distances to the singularity. The singularity is expected to be smoothed out by new UV physics which implies the existence of a cutoff in the applicability of the effective theory. This translates into cutting of a strip of spacetime around the singularities, hence providing a notion of 'strechted' ETW brane in effective theory. This can be obtained in different ways, for instance by imposing a maximal bound on the scalar curvature. Instead, we use a criterion directly inspired by the swampland distance conjecture [57], as follows.

The distance conjecture states that when an effective theory reaches to a large distance \mathcal{D} in field space, its effective cutoff scales as

$$\Lambda \sim e^{-\alpha \mathcal{D}} \,, \tag{5.1}$$

for some order 1 coefficient α . The actual distance conjecture moreover claims that there is an infinite tower of states becoming light with Λ , but this formulation corresponds to an adiabatic motion in moduli space, and such towers may actually not arise in dynamical situations with spacetime dependence of the scalars [58].¹⁰ Hence we stick to the milder statement that a cutoff is developed, whose origin in our context would stem from the UV completion of the ETW brane.

In our configuration we hence consider the slice of spacetime at which the field space distance (in the combined tachyon-dilaton system) reaches a large but finite value. From the Einstein frame action (4.11), the relevant kinetic terms read

$$\frac{4}{D-2}\partial\Phi\cdot\partial\Phi - \frac{4}{D-2}\frac{f_1'}{f_1}\partial\Phi\cdot\partial T + \left(-\frac{f_1''}{f_1} - \frac{f_1'}{Tf_1} + \frac{D-1}{D-2}\frac{f_1'^2}{f_1^2}\right)\partial T\cdot\partial T.$$
 (5.2)

We are interested in the behaviour near the intersection of the two singularities. Since this lies at large T, we can simplify the last term using the argument in section 4.2.3. Using $(f'_1/f_1)\partial T = \partial \log f_1$, the kinetic term may be written

$$\frac{1}{D-2}(2\partial\Phi - \partial\log f_1)^2, \qquad (5.3)$$

so that the slices of constant distance are defined by

$$\mathcal{D} \sim \frac{1}{\sqrt{D-2}} (2\Phi - \log f_1) = \text{const}.$$
(5.4)

Note that interestingly, the swampland distance cutoff (5.1) is

$$\Lambda \sim e^{-\alpha \mathcal{D}} \sim \exp\left(-\frac{\alpha \delta}{2} \left(2\Phi - \log f_1\right)\right) = \left(e^{-2\Phi} f_1\right)^{\alpha \delta/2},\tag{5.5}$$

where δ is given by (4.18). The factor inside brackets is the prefactor of the Einstein term in the string frame action. The fact that it relates to the cutoff scale shows that one gets the same spacetime slices if one uses a bound in the scalar curvature to limit the applicability of the effective theory, rather than in the field space distance. Indeed, this is expected from the scaling (2.3) of R with \mathcal{D} near ETW branes.

The curve in the (x^0, x^1) -plane defined by (5.4) asymptotes to constant X^0 on one side and to constant X^+ on the other. For illustration we may consider f_1 as in (4.13) and get

$$2qx^0 - b\mu^k e^{\beta kx^+} = \text{const.}$$
(5.6)

¹⁰A simple example is the Taub-NUT geometry, which can be regarded as a spacetime-dependent solution of S^1 compactifications, in which the circle shrinks to zero size at a point in the base. Hence, it attains infinite distance in the naive circle compactification moduli space, but no tower of light particles or other disasters arise.



Figure 1. Depiction of a spacetime slice at which the solutions attain a large fixed field space distance in the tachyon-dilaton space. The asymptotes near $x^0 \sim 0$ and $x^+ \sim 0$, should be regarded as corresponding to the physical location of the ETW branes, which are hence recombined in the middle region, and which for a boundary of spacetime, with nothing beyond them. The spray line signal the location of the singularities.

This leads to slices of the form

$$x^{1} = \frac{\sqrt{2}}{k\beta} \log(x^{0} + cst) - x^{0} + \frac{\sqrt{2}}{k\beta} \log\frac{2q}{b\mu^{k}}, \qquad (5.7)$$

for some constant cst related to the cutoff. In figure 1 we depict the structure of such curves and of the resulting spacetime picture.

From this it is clear that what seemed to be an intersection between the two ETW walls is in fact a smooth region that interpolates between the two. This hence strongly motivates that this solution describes a one and only recombined ETW wall. Another interesting indication for this is the fact that the dilaton-tachyon mixing in the effective action (scaling like f'_1/f_1 times powers of the string couplings) gets large for large tachyon and strong coupling, namely near the naive intersection of the singularities. Hence, the two walls, which asymptotically correspond to the dilaton or the tachyon running off to infinite distance in their field space, become of a very similar nature in that intersection region.

As a side note, one may wonder if our analysis extends to the case of critical D = 26bosonic string theory. Indeed, this theory is tachyonic and, as mentioned in section 3.1, it admits a lightlike dilaton background. It is then clear from the equation of motion for the tachyon (3.13) that, if we take the dilaton along the direction X^+ , the only option is to take the tachyon along X^- . This means that the ETW branes corresponding to large tachyon and strong coupling are of the intersecting kind, exactly like those described in this section. It would be interesting to explore these lightlike linear dilaton backgrounds in other critical theories, and make contact with existing proposals concerning the resulting singularities [36, 51].

6 Conclusions

In this paper we have studied timelike linear dilaton backgrounds of supercritical string theories as time-dependent solutions in string theory, and addressed the question of the resulting spacelike singularity, from the perspective of the cobordism conjecture. We have quantitatively characterized the solution as a Dynamical Cobordism in which the dilaton rolls until it hits infinite field space distance at a singularity at finite time in the past. We have shown that the singularity in effective theory follows the scaling behaviour of ETW branes. In order to clarify its microscopic description, we have considered lightlike tachyon condensation backgrounds, whose microscopic description had been argued to correspond to a stringy version of a bubble of nothing. Using an effective theory approach, we have characterized them as ETW branes and have encountered precisely the same scaling exponent as the beginning of time singularity. Together with the fact that both ETW branes join smoothly from the effective theory perspective, this has motivated our proposal that the spacelike singularity should correspond in string theory to a region of (a strong coupling version of) closed tachyon condensation, giving rise to a cobordism boundary defining the beginning of time.

There are several open questions and new directions:

- We have used an effective theory for the tachyon in terms of the undetermined function f_1 . Even though our results are robust under changes of the precise form of this function, it would be interesting to determine it, or at least its asymptotic behavior for large T.
- Conversely, it would be interesting to understand if the criterion that the theory allows for a resolution of spacelike singularities can be used as a constraint on effective theories. For instance, there exist choices of f_1 which lead to lightlike tachyon ETW branes with scalings different from the beginning of time one. Such theories may not be compatible with a microscopic description of the latter, as the ETW brane may not be compatible for recombination. It is thus tantalizing to claim that this can be used as a criterion to exclude such choices of f_1 . It would also be interesting to understand this possibility, possibly invoking other swampland constraints or physical considerations.
- Although we have focused on the bosonic theory, there is a rich set of phenomena arising in lightlike tachyon backgrounds in other string theories. We expect our ideas to lead to interesting new insights into this web of transitions.
- Finally, an interesting corner in this circle of ideas is that of lightlike dilaton backgrounds in critical string theories. They are toy models of cosmological singularities, which in certain supersymmetric cases admit interesting proposals for their microscopic description [51]. It would be exciting to use cobordism ideas to make progress on the understanding of such backgrounds.

We hope to report on these and other questions in the near future.

Acknowledgments

We are pleased to thank Luis Ibánez, Fernando Marchesano, Miguel Montero, Lorenz Schlechter and Irene Valenzuela for useful discussions, and Ginevra Buratti, José Calderón, Jesús Huertas for collaboration on related topics. M.D. also wishes to acknowledge the hospitality of the Department of Physics of Harvard University during the development of this work. This work is supported through the grants CEX2020-001007-S and PGC2018-095976-B-C21, funded by MCIN/AEI/10.13039/501100011033 and by ERDF A way of making Europe. The work by R.A. is supported by the grant BESST-VACUA of CSIC. The work by M.D. is supported by the FPI gran no. FPI SEV-2016-0597-19-3 from Spanish National Research Agency from the Ministry of Science and Innovation.

A Dimension quenching as an interpolating domain wall

In [17] it was argued that, when the scalars remain at finite distance in field space as one hits the wall, the corresponding configuration described an interpolating wall between different QG theories. This scenario is built in contrast with the end-of-the-world walls where the fields reach infinite distance in field-space at the wall. Instead of the solution ending abruptly at the location of a singularity, these interpolating solutions continue across the wall into another theory. On each side of the interpolating wall, the field spaces may have different structures but the location of the wall itself is at finite distance in both of them. As a result, the interpolating wall must have all the right properties for communicating between the two theories, whatever they may be. It is clear from this that the microscopic nature of these walls can be hard to describe; it may in general be non-supersymmetric and may involve strong-coupling physics. The existence of such objects is one of the predictions of the Cobordism Conjecture [3].

The examples of interpolating walls in [17] were simple enough to be described by standard supersymmetric objects. For example, D8 branes in massive type IIA string theory were identified as such walls interpolating between "different" massive IIA theories with different units of 0-form flux. Here, we re-interpret the dimension-quenching bubbles of [36, 37] as interpolating walls between bosonic string theories of different dimensions.

Throughout this paper we have dealt with tachyons with exponential profiles along one light-like direction. These solutions were shown to lead to bubbles of nothing, which fit in the wall-of-nothing description. There are, however, slightly more complicated solutions to the tachyon equation of motion (3.13). Following [36], we can consider a profile with oscillatory dependence on another coordinate, denoted by X^2 :

$$T(X) = \mu_0^2 \exp(\beta X^+) - \mu_k^2 \cos(kX^2) \exp(\beta_k X^+).$$
 (A.1)

This is a solution to the equation of motion with a timelike linear dilaton $\Phi = -qX^0$ background if:

$$q\beta_k = \sqrt{2} \left(\frac{2}{\alpha'} - \frac{1}{2}k^2\right). \tag{A.2}$$

Since the tachyon couples to the worldsheet as a potential, the theory has a vacuum at $X^2 = 0$. One can show that expanding around this vacuum in the limit where the wavelength of oscillations k^{-1} is much larger than the string length l_s yields:

$$T(X^+, X^2) = \frac{\mu^2}{2\alpha'} \exp(\beta X^+) : (X^2)^2 :,$$
(A.3)

where $\mu^2 = \alpha' k^2 \mu_k^2$, and dots denote normal ordering. We refer the reader to [36] for additional aspects of the detailed derivation.

The physical interpretation of this is clear. Before the tachyon condenses, at $X^+ \rightarrow -\infty$, the string propagates in D-1 spatial dimensions. As the string reaches a regime where $T \sim 1$ (namely $X^+ \sim \beta^{-1} \log \mu$), the potential confines the string to the region where it is vanishing, at $X^2 = 0$. Strings that oscillate along the X^2 dimension will be expelled from the region of large tachyon condensate. This bubble thus interpolates between a region of D-1 spatial dimensions to one where the string can effectively propagate in D-2 dimensions. These types of bubbles were dubbed dimension-quenching or dimension-changing bubbles.

Turning now to the dynamical cobordism perspective, one can see in (A.3) that the tachyon field remains at a finite value (hence at a finite distance in field space) at the location of this bubble at $X^2 = 0$. This fits perfectly with the description of an interpolating wall as described in [17]. We thus interpret these dimension-quenching bubbles as examples of dynamical cobordism interpolating walls between bosonic theories of different dimensions.

As a side note, one can construct similar bubbles that kill more than one dimension by granting oscillatory dependence of the tachyon on extra dimensions. Furthermore, this dimension-quenching mechanism also extends to superstring theories and can be used to draw connections between supercritical Type 0 theories and their 10-dimensional critical counterparts [36].

B The partition function with a lightlike tachyon background

The computation of the partition function in the presence of the lightlike tachyon background is obtained evaluating the path integral without vertex operator insertions:

$$Z(\mu) = \int \mathcal{D}X^{+} \mathcal{D}X^{-} \mathcal{D}X^{i} \mathcal{D}g \mathcal{D}(\text{ghosts}) e^{iS^{\text{deformed}}}, \qquad (B.1)$$

where we have emphasized the fact that the integration along the lightlike directions does not affect the spacelike directions (i = 2, ..., D - 1). Being at weak coupling, we only consider the one-loop contribution and so we have to evaluate the 2d action on a genus one worldsheet:

$$S^{\text{deformed}} = -\frac{1}{2\pi\alpha'} \int_{\mathcal{M}_1} d^2 \sigma \sqrt{g} g^{\alpha\beta} \left[-\partial_\alpha X^+ \partial_\beta X^- - \partial_\alpha X^- \partial_\beta X^+ + \partial_\alpha X^i \partial_\beta X_i \right] + \frac{1}{2\pi} \int_{\mathcal{M}_1} d^2 \sigma \sqrt{g} R_2 \phi(X) - \frac{1}{2\pi} \mu \int_{\mathcal{M}_1} d^2 \sigma \sqrt{g} e^{\beta X^+} .$$
(B.2)

In analogy with the procedure in [45], we decompose the field $X^+(\tau, \sigma)$ into its zero and nonzero modes:

$$X^{+}(\tau,\sigma) = X_{0}^{+} + \hat{X}^{+}(\tau,\sigma), \qquad (B.3)$$

we get a standard integration for the zero mode:

$$\mathcal{D}X^+ = dX_0^+ \mathcal{D}\hat{X}^+. \tag{B.4}$$

Choosing the following convention to perform a Wick rotation:

$$\tau \mapsto i\tau_E \qquad X^i \mapsto iX^i_E \qquad \mu \mapsto -i\mu_E \,, \tag{B.5}$$

equation (B.1) becomes:

$$Z(\mu) = \int dX_0^+ \int \mathcal{D}\hat{X}^+ \mathcal{D}X^- \mathcal{D}X^i \mathcal{D}g\mathcal{D}(\text{ghosts})e^{-S_E^{\text{deformed}}}.$$
 (B.6)

where the tachyonic potential gives an oscillating contribution to the integral in the condensate region. Such a behavior produces a truncation of the contributions to the integral coming from configurations with $X^+ \mapsto \infty$. The Euclidean 2d action is:

$$S_E^{\text{deformed}} = \frac{1}{2\pi\alpha'} \int d^2 \sigma_E \sqrt{g} \left[\partial_{\sigma^0} \hat{X}^+ \partial_{\sigma^0} X^- + \partial_{\sigma^1} X^- \partial_{\sigma^1} \hat{X}^+ + \partial_{\alpha} X^i \partial_{\alpha} X^i \right] + \frac{1}{2\pi} \int d^2 \sigma_E \sqrt{g} R_2 \phi(X) + \frac{i\mu_E}{2\pi} \int d^2 \sigma_E \sqrt{g} e^{\beta X^+} .$$
(B.7)

Using the variable change:

$$y = e^{\beta X_0^+} \longrightarrow dX_0^+ = \frac{dy}{\beta y},$$
 (B.8)

and making the dependence of the integrand on X_0^+ explicit, we obtain:

$$Z(\mu_E) = \int \mathcal{D}\widehat{X}^+ \mathcal{D}X^- \mathcal{D}X^i \mathcal{D}g\mathcal{D}(\text{ghosts}) \int_0^\infty \frac{dy}{\beta y} e^{-S_E^{\text{kinetic}} - S_E^{\text{dilaton}} - \frac{i\mu_E}{2\pi} \int d^2 \sigma_E \sqrt{g} y e^{\beta \widehat{X}^+}},$$
(B.9)

where $S_E^{\text{kinetic}} + S_E^{\text{dilaton}} = S_E^{\text{free}}$ are respectively the kinetic and the dilaton contributions in the Euclidean action (B.7).

Now, let us consider the following quantity:

$$\frac{\partial Z}{\partial \mu_E} = \int \mathcal{D}\widehat{X}^+ \mathcal{D}(\text{others}) \int_0^\infty \frac{dy}{\beta} e^{-S_E^{\text{free}} + \frac{-i\mu_E}{2\pi} \int d^2 \sigma_E \sqrt{g} y e^{\beta \widehat{X}^+}} \left(\frac{-i}{2\pi} \int d^2 \sigma_E \sqrt{g} e^{\beta \widehat{X}^+}\right). \tag{B.10}$$

Let us finally perform the integration on the zero mode. We obtain:

$$\frac{\partial Z}{\partial \mu_E} = -\frac{1}{\beta \mu_E} \int \mathcal{D}\hat{X}^+ \mathcal{D}X^- \mathcal{D}X^i \mathcal{D}g \mathcal{D}(\text{ghosts}) e^{-S_E^{\text{free}}}, \qquad (B.11)$$

where S_E^{free} is the euclidean action of the world-sheet theory without the tachyon potential. Integrating with respect to μ and fixing a cutoff for X^+ such that $\mu_* = e^{\beta X_*^+}$, we obtain:

$$Z_1 = -\frac{\log(\mu_E/\mu_*)}{\beta}\widehat{Z}, \qquad (B.12)$$

where \hat{Z} is the partition function for the free 2d theory, namely without the tachyon insertion and without integrating the zero modes of X^+ . Note that the tachyon's contribution to the partition function is entirely encoded in the zero modes.

We can interpret this factor as a "size" of the direction X^+ . Indeed, because of the potential barrier created by the condensation of the tachyon, no physical degrees of freedom penetrate inside the bubble wall, beyond $X^+ \sim 1$. As mentioned previously, the path integral is suppressed in this region. The direction X^+ thus has an effectively finite "size" that agrees with the estimate in (4.3).

C The critical exponent for general f_1

In this appendix we provide more details regarding how we obtain the scaling relation (4.27) for a general f_1 decaying faster than T^{-1} . The starting point is (4.26), which we rewrite as follows:

$$\Delta \sim \int \exp\left(-\frac{\mathcal{D}}{\sqrt{D-2}} - \log(T\mathcal{D}')\right) d\mathcal{D}, \qquad (C.1)$$

where the prime stands for derivation with respect to T. Proving that the first term in the exponential is the dominant one in the limit $T \to \infty$ comes down to comparing the two terms:

$$-\frac{\mathcal{D}}{\sqrt{D-2}} \sim \log f_1 \quad \text{and} \quad -\log T\mathcal{D}' \sim \log\left(\frac{f_1}{|f_1'|T}\right).$$
 (C.2)

Notice that in the special case where f_1 is power-like $f_1 = T^{-k}$, with k > 0, the second term in the exponential is constant so one automatically obtains the scaling relation (4.27). For other choices of f_1 , we wish to check that in the limit $T \to \infty$,

$$\frac{\left|\log\left(\frac{f_1}{|f_1'|T}\right)\right|}{\left|\log f_1\right|} \to 0.$$
(C.3)

We consider a positive and monotonically decreasing function f_1 , we require:

$$\left|\log\left(\frac{f_1}{|f_1'|T}\right)\right| \ll \left|\log f_1\right| \text{ as } T \to \infty.$$
 (C.4)

We know for a fact that $\log f_1$ is negative when $T \to \infty$. As we will show shortly, $\log(\frac{f_1}{|f_1'|T})$ is negative, then one can easily show that (C.4) implies:

$$|f_1'| \ll T^{-1}$$
, (C.5)

which is true for any function f_1 under consideration since $f_1 \ll T^{-1}$.

The question of whether (C.3) is verified is thus recast into the question of the sign of $\log(\frac{f_1}{|f'_i|T})$. In order to determine if this term is negative, we can write it as follows,

$$\log\left(\frac{f_1}{|f_1'|T}\right) = \log\left(\frac{f_1}{T}\right) - \log(|f_1'|).$$
(C.6)

One clearly sees that the first term is negative whilst the second is positive. We would therefore like to show that the first one dominates:

$$\left|\log\left(\frac{f_1}{T}\right)\right| \gg \left|\log(|f_1'|) \rightarrow \frac{f_1}{T} \ll |f_1'|$$
(C.7)

This is condition is verified for all valid choices of f_1 that are not power-like. Indeed, we can use the trivial fact that $f_1 \ll T$ as $T \to \infty$ to show that $\frac{f_1}{|f_1'|} \ll T$. In the case where f_1 is a power-like function the inequality is not strict as $\frac{f_1}{T} \sim |f_1'|$, but we still recover the right scaling relations as mentioned previously.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited. SCOAP³ supports the goals of the International Year of Basic Sciences for Sustainable Development.

References

- F. Quevedo, Lectures on string/brane cosmology, Class. Quant. Grav. 19 (2002) 5721 [hep-th/0210292] [INSPIRE].
- B. Craps, Big Bang Models in String Theory, Class. Quant. Grav. 23 (2006) S849 [hep-th/0605199] [INSPIRE].
- [3] J. McNamara and C. Vafa, Cobordism Classes and the Swampland, arXiv:1909.10355 [INSPIRE].
- [4] E. Witten, Instability of the Kaluza-Klein Vacuum, Nucl. Phys. B 195 (1982) 481 [INSPIRE].
- [5] H. Ooguri and L. Spodyneiko, New Kaluza-Klein instantons and the decay of AdS vacua, Phys. Rev. D 96 (2017) 026016 [arXiv:1703.03105] [INSPIRE].
- [6] I. García Etxebarria, M. Montero, K. Sousa and I. Valenzuela, Nothing is certain in string compactifications, JHEP 12 (2020) 032 [arXiv:2005.06494] [INSPIRE].
- [7] G. Dibitetto, N. Petri and M. Schillo, Nothing really matters, JHEP 08 (2020) 040 [arXiv:2002.01764] [INSPIRE].
- [8] P. Bomans, D. Cassani, G. Dibitetto and N. Petri, *Bubble instability of m IIA on* $AdS_4 \times S^6$, *SciPost Phys.* **12** (2022) 099 [arXiv:2110.08276] [INSPIRE].
- [9] J.B. Hartle and S.W. Hawking, Wave Function of the Universe, Phys. Rev. D 28 (1983) 2960 [INSPIRE].
- [10] H. Ooguri and T. Takayanagi, Cobordism Conjecture in AdS, arXiv: 2006.13953 [INSPIRE].
- [11] M. Montero and C. Vafa, Cobordism Conjecture, Anomalies, and the String Lampost Principle, JHEP 01 (2021) 063 [arXiv:2008.11729] [INSPIRE].
- [12] M. Dierigl and J.J. Heckman, Swampland cobordism conjecture and non-Abelian duality groups, Phys. Rev. D 103 (2021) 066006 [arXiv:2012.00013] [INSPIRE].
- [13] Y. Hamada and C. Vafa, 8d supergravity, reconstruction of internal geometry and the Swampland, JHEP 06 (2021) 178 [arXiv:2104.05724] [INSPIRE].
- [14] R. Blumenhagen and N. Cribiori, Open-closed correspondence of k-theory and cobordism, JHEP 08 (2022) 037 [arXiv:2112.07678] [INSPIRE].

- [15] D. Andriot, N. Carqueville and N. Cribiori, Looking for structure in the cobordism conjecture, arXiv:2204.00021 [INSPIRE].
- [16] G. Buratti, M. Delgado and A.M. Uranga, Dynamical tadpoles, stringy cobordism, and the SM from spontaneous compactification, JHEP 06 (2021) 170 [arXiv:2104.02091] [INSPIRE].
- [17] G. Buratti, J. Calderón-Infante, M. Delgado and A.M. Uranga, Dynamical Cobordism and Swampland Distance Conjectures, JHEP 10 (2021) 037 [arXiv:2107.09098] [INSPIRE].
- [18] R. Angius, J. Calderón-Infante, M. Delgado, J. Huertas and A.M. Uranga, At the end of the world: Local Dynamical Cobordism, JHEP 06 (2022) 142 [arXiv:2203.11240] [INSPIRE].
- [19] R. Blumenhagen, N. Cribiori, C. Kneissl and A. Makridou, Dynamical cobordism of a domain wall and its companion defect 7-brane, JHEP 08 (2022) 204 [arXiv:2205.09782] [INSPIRE].
- [20] E. Dudas and J. Mourad, Brane solutions in strings with broken supersymmetry and dilaton tadpoles, Phys. Lett. B 486 (2000) 172 [hep-th/0004165] [INSPIRE].
- [21] R. Blumenhagen and A. Font, Dilaton tadpoles, warped geometries and large extra dimensions for nonsupersymmetric strings, Nucl. Phys. B 599 (2001) 241 [hep-th/0011269] [INSPIRE].
- [22] E. Dudas, J. Mourad and C. Timirgaziu, Time and space dependent backgrounds from nonsupersymmetric strings, Nucl. Phys. B 660 (2003) 3 [hep-th/0209176] [INSPIRE].
- [23] E. Dudas, G. Pradisi, M. Nicolosi and A. Sagnotti, On tadpoles and vacuum redefinitions in string theory, Nucl. Phys. B 708 (2005) 3 [hep-th/0410101] [INSPIRE].
- [24] I. Basile, J. Mourad and A. Sagnotti, On Classical Stability with Broken Supersymmetry, JHEP 01 (2019) 174 [arXiv:1811.11448] [INSPIRE].
- [25] R. Antonelli and I. Basile, Brane annihilation in non-supersymmetric strings, JHEP 11 (2019) 021 [arXiv:1908.04352] [INSPIRE].
- [26] A. Mininno and A.M. Uranga, Dynamical tadpoles and Weak Gravity Constraints, JHEP 05 (2021) 177 [arXiv:2011.00051] [INSPIRE].
- [27] I. Basile, On String Vacua without Supersymmetry: brane dynamics, bubbles and holography, Ph.D. thesis, Pisa, Scuola Normale Superiore, 2020. arXiv:2010.00628 [INSPIRE].
- [28] I. Basile, Supersymmetry breaking, brane dynamics and Swampland conjectures, JHEP 10 (2021) 080 [arXiv:2106.04574] [INSPIRE].
- [29] J. Mourad and A. Sagnotti, A 4D IIB Flux Vacuum and Supersymmetry Breaking. I. Fermionic Spectrum, arXiv:2206.03340 [INSPIRE].
- [30] R.C. Myers, New Dimensions for Old Strings, Phys. Lett. B 199 (1987) 371 [INSPIRE].
- [31] I. Antoniadis, C. Bachas, J.R. Ellis and D.V. Nanopoulos, Cosmological String Theories and Discrete Inflation, Phys. Lett. B 211 (1988) 393 [INSPIRE].
- [32] I. Antoniadis, C. Bachas, J.R. Ellis and D.V. Nanopoulos, An Expanding Universe in String Theory, Nucl. Phys. B 328 (1989) 117 [INSPIRE].
- [33] I. Antoniadis, C. Bachas, J.R. Ellis and D.V. Nanopoulos, Comments on cosmological string solutions, Phys. Lett. B 257 (1991) 278 [INSPIRE].
- [34] Y. Chen and J. Maldacena, String scale black holes at large D, JHEP 01 (2022) 095 [arXiv:2106.02169] [INSPIRE].
- [35] S. Hellerman and I. Swanson, Cosmological solutions of supercritical string theory, Phys. Rev. D 77 (2008) 126011 [hep-th/0611317] [INSPIRE].

- [36] S. Hellerman and I. Swanson, Dimension-changing exact solutions of string theory, JHEP 09 (2007) 096 [hep-th/0612051] [INSPIRE].
- [37] S. Hellerman and I. Swanson, Charting the landscape of supercritical string theory, Phys. Rev. Lett. 99 (2007) 171601 [arXiv:0705.0980] [INSPIRE].
- [38] I. Swanson, Cosmology of the closed string tachyon, Phys. Rev. D 78 (2008) 066020
 [arXiv:0804.2262] [INSPIRE].
- [39] M. Berasaluce-González, M. Montero, A. Retolaza and A.M. Uranga, Discrete gauge symmetries from (closed string) tachyon condensation, JHEP 11 (2013) 144
 [arXiv:1305.6788] [INSPIRE].
- [40] I. García-Etxebarria, M. Montero and A. Uranga, Heterotic NS5-branes from closed string tachyon condensation, Phys. Rev. D 90 (2014) 126002 [arXiv:1405.0009] [INSPIRE].
- [41] I. García-Etxebarria, M. Montero and A.M. Uranga, Closed tachyon solitons in type-II string theory, Fortsch. Phys. 63 (2015) 571 [arXiv:1505.05510] [INSPIRE].
- [42] A. Adams, J. Polchinski and E. Silverstein, Don't panic! Closed string tachyons in ALE space-times, JHEP 10 (2001) 029 [hep-th/0108075] [INSPIRE].
- [43] A.M. Uranga, Localized instabilities at conifolds, hep-th/0204079 [INSPIRE].
- [44] S.P. De Alwis and A.T. Flournoy, Closed string tachyons and semiclassical instabilities, Phys. Rev. D 66 (2002) 026005 [hep-th/0201185] [INSPIRE].
- [45] J. McGreevy and E. Silverstein, The Tachyon at the end of the universe, JHEP 08 (2005) 090 [hep-th/0506130] [INSPIRE].
- [46] S. Hellerman and I. Swanson, Cosmological unification of string theories, JHEP 07 (2008)
 022 [hep-th/0612116] [INSPIRE].
- [47] S. Hellerman and I. Swanson, Supercritical N = 2 string theory, arXiv:0709.2166 [INSPIRE].
- [48] S. Hellerman and I. Swanson, A Stable vacuum of the tachyonic E₈ string, arXiv:0710.1628 [INSPIRE].
- [49] H. Yang and B. Zwiebach, A Closed string tachyon vacuum?, JHEP 09 (2005) 054 [hep-th/0506077] [INSPIRE].
- [50] M. Gutperle and A. Strominger, Space-like branes, JHEP 04 (2002) 018 [hep-th/0202210] [INSPIRE].
- [51] B. Craps, S. Sethi and E.P. Verlinde, A Matrix big bang, JHEP 10 (2005) 005 [hep-th/0506180] [INSPIRE].
- [52] T. Banks, W. Fischler, S.H. Shenker and L. Susskind, M theory as a matrix model: A Conjecture, Phys. Rev. D 55 (1997) 5112 [hep-th/9610043] [INSPIRE].
- [53] J. McGreevy, E. Silverstein and D. Starr, New dimensions for wound strings: The Modular transformation of geometry to topology, Phys. Rev. D 75 (2007) 044025 [hep-th/0612121]
 [INSPIRE].
- [54] D.R. Green, A. Lawrence, J. McGreevy, D.R. Morrison and E. Silverstein, *Dimensional duality*, *Phys. Rev. D* 76 (2007) 066004 [arXiv:0705.0550] [INSPIRE].
- [55] S. Hellerman and X. Liu, Dynamical dimension change in supercritical string theory, hep-th/0409071 [INSPIRE].

- [56] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, The EFT stringy viewpoint on large distances, JHEP 09 (2021) 197 [arXiv:2104.05726] [INSPIRE].
- [57] H. Ooguri and C. Vafa, On the Geometry of the String Landscape and the Swampland, Nucl. Phys. B 766 (2007) 21 [hep-th/0605264] [INSPIRE].
- [58] G. Buratti, J. Calderón and A.M. Uranga, Transplanckian axion monodromy!?, JHEP 05 (2019) 176 [arXiv:1812.05016] [INSPIRE].



This chapter contains the article:

• Global anomalies & bordism of non-supersymmetric strings, Ivano Basile, Arun Debray, M. Delgado, Miguel Montero JHEP 02 (2024) 092 arXiv:2310.06895 - Inspire

Published for SISSA by 2 Springer

RECEIVED: November 7, 2023 ACCEPTED: January 22, 2024 PUBLISHED: February 14, 2024

Global anomalies & bordism of non-supersymmetric strings

Ivano Basile^(D), Arun Debray^(D), Matilda Delgado^(D) and Miguel Montero^(D)

^aArnold Sommerfeld Center for Theoretical Physics, Ludwig-Maximilians-Universität (LMU), Theresienstraße 37, 80333 München, Germany

^bPurdue University,

West Lafayette, IN 47907, U.S.A.

^cInstituto de Física Teórica IFT-UAM/CSIC,

C/ Nicolás Cabrera 13-15, Campus de Cantoblanco, 28049 Madrid, Spain

E-mail: ivano.basile@lmu.de, adebray@purdue.edu, matilda.delgado@uam.es, miguel.montero@csic.es

ABSTRACT: The three tachyon-free non-supersymmetric string theories in ten dimensions provide a handle on quantum gravity away from the supersymmetric lamppost. However, they have not been shown to be fully consistent; although local anomalies cancel due to versions of the Green-Schwarz mechanism, there could be global anomalies, not cancelled by the Green-Schwarz mechanism, that could become fatal pathologies. We compute the twisted string bordism groups that control these anomalies via the Adams spectral sequence, showing that they vanish completely in two out of three cases (Sugimoto and SO(16)²) and showing a partial vanishing also in the third (Sagnotti 0'B model). We also compute lower-dimensional bordism groups of the non-supersymmetric string theories, which are of interest to the classification of branes in these theories via the Cobordism Conjecture. We propose a worldvolume content based on anomaly inflow for the SO(16)² NS5-brane, and discuss subtleties related to the torsion part of the Bianchi identity. As a byproduct of our techniques and analysis, we also reprove that the outer \mathbb{Z}_2 automorphism swapping the two E_8 factors in the supersymmetric heterotic string is also non-anomalous.

KEYWORDS: String and Brane Phenomenology, Superstrings and Heterotic Strings

ARXIV EPRINT: 2310.06895



Contents

1	Introduction 2 Local anomalies and the Green-Schwarz mechanism			1
2				4
3	Global anomalies and bordism groups			10
	3.1	3.1 Bianchi identities and twisted string structures		12
	3.2	Evidence for torsional Bianchi identities		17
		3.2.1	Type I and HO heterotic theories	18
		3.2.2	$E_8 \times E_8$ theory	19
		3.2.3	Non-supersymmetric theories	20
	3.3 Vanis		hing bordism classes	21
		3.3.1	$\operatorname{Sp}(16)$	21
		3.3.2	U(32)	27
		3.3.3	$\operatorname{Spin}(16) \times \operatorname{Spin}(16)$	34
	3.4	4 Physical intuition from fivebrane anomaly inflow		37
		3.4.1	Sp(16) and $U(32)$	39
		3.4.2	$SO(16) \times SO(16)$	40
		3.4.3	Anomaly inflow on puffed-up fivebrane instantons	44
4	4 Anomalies and bordism for the swap \mathbb{Z}_2 action			46
	4.1 Overview and the bordism computation			46
	4.2 Cancelling the anomaly		56	
5	Conclusions			59

1 Introduction

It is widely known that there are five supersymmetric string theories in ten dimensions [1]. It is slightly less known that there are several other non-supersymmetric string theories in ten dimensions, many of which have tachyons; and there are just three known models in ten dimensions which are both non-supersymmetric in spacetime and tachyon free: the $SO(16) \times SO(16)$ string [2, 3], the Sugimoto model [4], and the Sagnotti 0'B model [5, 6]. These non-supersymmetric models and their compactifications have been the subject of a renovated interest in the recent literature (see e.g. [7–35]), presumably because they constitute a promising arena to study quantum gravity away from the supersymmetric lamppost.

In spite of this recent surge of work, we still know remarkably little about the three tachyonfree non-supersymmetric models in ten dimensions, particularly when compared with their supersymmetric counterparts. In particular, the spectrum of all three non-supersymmetric models is chiral in ten dimensions, and so it is potentially anomalous. Local anomalies have long been known to cancel via non-supersymmetric versions of the Green-Schwarz mechanism [2–6]. However, to our knowledge, except for an inconclusive analysis in [36] these models have not been shown to be free of global anomalies, as was done early on in [37-39] for the supersymmetric chiral theories.¹ A global anomaly could lead to an inconsistency of the model, and having to discard it, or to new topological couplings that cancel it [40].

The purpose of this paper is twofold: on one hand, we compute the potential global anomaly of the three tachyon-free non-supersymmetric string theories in quantum gravity (albeit only a subclass of anomalies for the Sagnotti 0'B model), showing that it vanishes. We do this using cobordism theory, and computing the relevant twisted string bordism groups. This is a standard technique for proving anomaly cancellation results; see [40–47] for recent anomaly cancellation theorems in string and supergravity theories using this technique.

The second important result of our paper is precisely the calculation of these twisted string bordism groups (which have not appeared in the literature before), which are summarized in table 1 in the conclusions. The physics use of these bordism groups is that they can be used to predict new, singular configurations (branes) of the corresponding non-supersymmetric string theories, by means of the Cobordism Conjecture [48] (see [49–52] for similar recent work in type II and supersymmetric heterotic string theories). Furthermore, the calculation of the bordism groups themselves by means of the Adams spectral sequence is interesting in its own right, and we expect that similar techniques can be used to compute string bordism groups of e.g. six-dimensional compactifications, and more generally, to study anomalies of any theory with a 2-group symmetry or a Green-Schwarz mechanism.

Along the way, we will encounter and comment on issues such as whether the heterotic Bianchi identity can be taken to take values on the free part of cohomology or the torsion piece must be included, or the connections between anomaly cancellation in eleven-dimensional backgrounds and anomaly inflow on non-supersymmetric NS5 branes on these theories. We also include a quick introduction to the Green-Schwarz mechanism in the modern formalism of anomaly theory, providing for the first time a candidate for the worldvolume degrees of freedom for the NS5 brane in the SO(16) × SO(16) string. We also study global anomalies in the \mathbb{Z}_2 outer automorphism swapping the two factors of the SO(16) × SO(16) string, showing that anomalies vanish.

The upshot of our paper is:

- The bordism group controlling anomalies of the Sugimoto string, $\Omega_{11}^{\text{String-Sp}(16)}$, vanishes (theorem 3.48), and therefore the theory is anomaly-free.²
- The bordism group $\Omega_{11}^{\text{String-SU}(32)\langle c_3 \rangle}$, controlling the anomaly of the Sagnotti 0'B model, is isomorphic to 0 or \mathbb{Z}_2 (theorem 3.63). We do not know whether the anomaly vanishes, although it does in all specific backgrounds we looked into.
- For the $SO(16) \times SO(16)$ heterotic string, (where the identity component of the global form of the gauge group is actually $Spin(16) \times Spin(16)$, since the massless spectrum

¹Even in this case, only gravitational and global anomalies in the identity component of the gauge group have been considered.

²Modulo potential subtleties regarding the global structure of the gauge group, that we comment on in the Conclusions.

contains both spinors and vectors³), the bordism group $\Omega_{11}^{\text{String-Spin}(16)^2}$ controlling the anomaly vanishes (theorem 3.78), and therefore this theory is anomaly-free.

• There is also a \mathbb{Z}_2 gauge symmetry swapping the two factors of Spin(16), whose anomaly we also studied. The bordism group controlling the anomaly has order 64 — but nevertheless (theorem 4.30), the anomaly vanishes.

As a consequence of our calculations, we also can cancel an anomaly in a supersymmetric string theory.

• When one takes into account the \mathbb{Z}_2 symmetry of the $E_8 \times E_8$ heterotic string swapping the two copies of E_8 , the anomaly vanishes (corollary 4.34).

The cancellation of this anomaly is not a new result: it is a special case of the more general work of [42]. Our argument rests on different physical assumptions and is a different mathematical result; for example, we do not assume the Stolz-Teichner conjecture. Thus we answer a question of [46], who showed the bordism group controlling this anomaly has order 64 but did not address the anomaly, and asked for a bordism-theoretic argument that the anomaly vanishes.

One key application of this \mathbb{Z}_2 symmetry of the $E_8 \times E_8$ heterotic string is constructing the CHL string [54], a nine-dimensional string theory obtained by compactifying the $E_8 \times E_8$ heterotic string on a circle, where the monodromy around the circle is the \mathbb{Z}_2 symmetry we discussed above. An anomaly in the \mathbb{Z}_2 symmetry would have implied an inconsistency in the CHL string. We find that the anomaly vanishes, in agreement with the results in [42], which showed this from a worldsheet perspective; by contrast, we approach the question from a pure spacetime perspective.

One can make an analogous construction for the $SO(16) \times SO(16)$ heterotic string, compactifying it on a circle whose monodromy exchanges the two bundles. The result is a nine-dimensional non-supersymmetric string theory whose gauge group is (perhaps a quotient of) Spin(16). Studying this theory would be an interesting extension of similar constructions in the $E_8 \times E_8$ case [55] and in the SO(16) × SO(16) case without the monodromy [35]. Analogously to the CHL string, an anomaly in the \mathbb{Z}_2 symmetry of the SO(16) × SO(16) heterotic string would lead to an inconsistency of this new theory, and our anomaly cancellation result implies a consistency check for this theory on backgrounds where the gauge group is Spin(16). It would be interesting to study this theory on more general backgrounds.

We have also identified a plethora of non-trivial bordism classes on these theories. It is a natural direction to explore the nature and physics of the bordism defects associated to these branes [49–51], a task we will not pursue in this paper. Furthermore, representatives of the bordism classes we encountered provide natural examples of interesting compactification manifolds for these non-supersymmetric strings to various dimensions; studying these, finding out whether moduli are stabilized (including SUSY-breaking stringy corrections to the potential) etc. is another important open direction to study.

³See [53] for an analysis detailing some possibilities for a global quotient in the $SO(16)^2$ gauge group. In this paper, we assume the simply-connected global form $Spin(16)^2$ (so " $SO(16)^2$ " is an abuse of notation); this has the advantage that all anomalies we find also exist for any other possibility (although with a nontrivial global quotient, there could be more anomalies than the ones that we study here).

This paper is organized as follows: in section 2 we provide a lightning review of modern methods to study anomalies and how these cancel via the Green-Schwarz mechanism, as well as a detailed description of how this happens for each of the three tachyon-free, nonsupersymmetric string theories. We believe this is the first time these important results are collected together in a single reference, and with a unified notation. Section 3 contains our main result — the calculation of bordism groups for these theories using the Adams spectral sequence — together with a discussion of the natural cohomology theory for the Bianchi identity to take values in. We also study in detail the relationship of higher-dimensional anomaly cancellation to the worldvolume theory of magnetic NS branes. Section 4 extends the anomaly calculation to the SO(16) × SO(16) string including the (gauged) automorphism swapping the two \mathbb{Z}_2 factors. This multiplies the number of interesting bordism classes, but anomalies still cancel. Finally, section 5 presents a table with our results, conclusions, and potential further directions, including a few comments on how these anomalies might be studied from a worldsheet point of view, in the line of [42, 45].

2 Local anomalies and the Green-Schwarz mechanism

The word "anomaly" describes the breaking of a classical symmetry by quantum effects. In a Lagrangian theory, anomalies correspond to a lack of invariance of the path integral under a symmetry transformation. They can arise for both global and gauge symmetries in field theories. Anomalies in global symmetries only point to the fact that the symmetry cannot be gauged; they can lead to anomaly matching conditions that heavily constrain the RG-flow and strong coupling dynamics of the theory [56]. In contrast, anomalies in gauge theories point to true inconsistencies: a gauge symmetry is by definition a redundancy of the theory and as such can never be broken. In this paper, we will only consider anomalies in gauge symmetries.

The anomalies under consideration arise in field theories when they are coupled to gauge fields and dynamical gravity. They then correspond to a lack of invariance of the path integral under a gauge transformation/diffeomorphism (for a review, see [57]). When this transformation can be continuously connected to the identity, we speak of local anomalies. Their cancellation heavily constrains a theory; for example the gauge group of $\mathcal{N} = 1$ supergravities in ten dimensions is constrained by anomaly cancellation to be either one of four gauge groups: (a quotient of) Spin(32), $E_8 \times E_8$, U(1)⁴⁹⁶ or U(1)²⁴⁸ × E_8 . The last two can be ruled out as low-energy EFTs of a consistent theory of quantum gravity by demanding the consistency of the worldvolume theory of brane probes in [58] (see also [11] for developments in the context of orientifold models) and using more general arguments in [59].

When the anomalous symmetry transformation cannot be continuously connected to the identity, then we speak of global anomalies (not to be confused with anomalies in global symmetries!). Global anomalies and their cancellation will be at the heart of this paper. It only makes sense to study them once local anomalies cancel. We therefore review local anomaly cancellation in the remainder of this section, before discussing global anomalies in the next ones.

The most direct way to study local anomalies is to compute certain one-loop Feynman diagrams involving external gauge bosons and/or gravitons, and chiral fermions in the internal legs; for ten-dimensional theories, the relevant diagram has 6 external legs [60].

There is, however, a much more concise way of studying such anomalies, through what is called an anomaly polynomial [57]. This is a certain formal polynomial in the gauge-invariant quantities $\operatorname{tr} F^m$ and $\operatorname{tr} R^m$, which are certain contractions of Riemann and gauge field strength tensors that do not involve the metric. If the theory we wish to study lives in d spacetime dimensions, the anomaly polynomial is of degree (d+2). Although the anomaly polynomial is often discussed in the physics literature directly in terms of $\operatorname{tr} F^m$ and $\operatorname{tr} R^m$, we find it more natural and convenient to write it down in terms of (the free part of) Chern and Pontryagin characteristic classes, more common in the mathematical literature. These can be written as linear combinations of $\operatorname{tr} F^m$ and $\operatorname{tr} R^m$ via Chern-Weil theory as follows. The *i*-th Chern class $c_{\mathbf{r},i}$ is associated to a complex vector bundle, in some representation \mathbf{r} of the gauge group. Via Chern-Weil theory, they are represented in cohomology by the following characteristic polynomial of the field strength (here, t is just a dummy variable):⁴

$$\sum_{i} c_{\mathbf{r},i} t^{i} = \det\left(\frac{iF}{2\pi}t + 1\right), \qquad (2.1)$$

or, expanding the determinant,

$$\sum_{i} c_{\mathbf{r},i} t^{i} = 1 + \frac{i \operatorname{tr}_{\mathbf{r}}(F)}{2\pi} t + \frac{\operatorname{tr}_{\mathbf{r}}(F^{2}) - \operatorname{tr}_{\mathbf{r}}(F)^{2}}{8\pi^{2}} t^{2} + \cdots$$
(2.2)

The traces are over the gauge indices and as such the *i*-th Chern class $c_{\mathbf{r},i}$ is a 2*i*-form. The Pontryagin classes are characteristic classes associated to a real vector bundle, which we will always take to be the spacetime tangent bundle. One way to define them is in terms of the Chern classes of the complexification of the vector bundle. The total Pontryagin class is the sum of the Pontryagin classes and its first few terms are as follows:

$$p = 1 + p_1 + p_2 + \dots \tag{2.3}$$

$$p = 1 - \frac{\operatorname{tr}(R^2)}{8\pi^2} + \frac{\operatorname{tr}(R^2)^2 - 2\operatorname{tr}(R^4)}{128\pi^4} \cdots$$
(2.4)

Similarly to the case of the Chern classes, the Pontryagin class p_i is a 4i-form. The reason we prefer these characteristic classes over the trace notation trF^m and trR^m is that, as will be clear later, the anomaly polynomial is a sum of Atiyah-Singer indices, and these indices are written in terms of these classes in the mathematical literature.

The precise relationship between local anomalies and the anomaly polynomial is as follows. The anomalous variation of the quantum effective action $\delta_{\Lambda}\Gamma$ can be related to the (d+2)-dimensional anomaly polynomial through what is called the Wess-Zumino descent procedure, which we briefly outline here. Since the characteristic classes are (locally) exact, the anomaly polynomial itself is also (locally) exact. We can therefore locally write it as

$$P_{d+2} = dI_{d+1} \tag{2.5}$$

⁴Representing characteristic classes with differential forms misses any torsional components of integral cohomology, which is the more natural domain of characteristic classes. This subtlety will play an important role when discussing certain global anomalies in later parts of this paper, but it is immaterial in the present discussion.

where I_{d+1} is called the (Lagrangian density of the) anomaly theory and locally satisfies the descent property $\delta_{\Lambda}I_{d+1} = dI_d$. It is related to an anomalous variation of the effective action $\delta_{\Lambda}\Gamma$, which is local, by extending the spacetime manifold X_d into a (d+1)-dimensional manifold Y_{d+1} whose boundary is X_d . One finds

$$\delta_{\Lambda}\Gamma = \int_{X_d = \partial Y_{d+1}} I_d = \int_{Y_{d+1}} dI_d = \delta_{\Lambda} \left[\int_{Y_{d+1}} I_{d+1} \right] \equiv \delta_{\Lambda} \left[\alpha(Y_{d+1}) \right], \tag{2.6}$$

where the anomaly I_d is only defined up to a closed form and $\alpha(Y_{d+1})$ is called the anomaly theory. The anomaly corresponding to a given gauge transformation is then computed as the integral of I_d over the spacetime manifold X_d . It follows that local anomalies vanish if and only if the anomaly polynomial vanishes.

The anomaly polynomial can be written as a sum of contributions from all of the fields in the theory. Each contribution is given by an index density in (d + 2)-dimensions, whose integral over a compact manifold (with suitable structure, e.g. spin for fermions) gives the index of the corresponding Dirac operator via the Atiyah-Singer index theorem. We now list some of these contributions that will be relevant in what follows. The index density associated to a left-handed Weyl fermion in the representation \mathbf{r} of a gauge group with field strength F is:

$$\mathcal{I}_{1/2} = \left[\hat{A}(R)\operatorname{tr}_{\mathbf{r}} e^{iF/2\pi}\right]_{d+2},\tag{2.7}$$

where the term $\operatorname{tr}_{\mathbf{r}} e^{iF/2\pi}$ is sometimes referred to as the Chern character. The notation $[\cdots]_{d+2}$ means that one should select the (d+2)-form part of the enclosed expression. $\hat{A}(R)$ is called the *A*-roof polynomial, and it can be expanded as:

$$\hat{A}(R) = 1 - \frac{p_1}{24} + \frac{(7p_1^2 - 4p_2)}{5760} + \frac{-31p_1^3 + 44p_1p_2 - 16p_3}{967680} \cdots$$
(2.8)

Note that expression (2.7) can be easily applied to a fermion that is a singlet under the gauge group, in which case the Chern character reduces to 1. The contribution to the anomaly polynomial corresponding to a left-handed fermion singlet is therefore simply:

$$\mathcal{I}_{\text{Dirac}} = \left[\hat{A}(R)\right]_{d+2}.$$
(2.9)

The index density associated to a left-handed Weyl gravitino in (d+2) dimensions is:

$$\mathcal{I}_{3/2} = \left[\hat{A}(R) \left(\operatorname{tr} e^{iR/2\pi} - 1 \right) \operatorname{tr}_{\mathbf{r}} e^{iF/2\pi} \right]_{d+2}.$$
 (2.10)

Finally, a self-dual tensor gives a contribution:

$$\mathcal{I}_{\rm SD} = \left[-\frac{1}{8} L(R) \right]_{d+2},\tag{2.11}$$

where L(R) is called the *L*-polynomial or Hirzebruch genus, and it can be expanded as follows:

$$L(R) = 1 + \frac{p_1}{3} + \frac{-p_1^2 + 7p_2}{45} + \frac{2p_1^3 - 13p_1p_2 + 62p_3}{945} + \dots$$
(2.12)

Armed with these index densities, we can now review anomalies in ten-dimensional $\mathcal{N} = 1$ supergravities. With this supersymmetry, the only multiplets are the gravity and vector

multiplets [60]. There are pure gravitational anomalies coming from the chiral gravitini in the gravity multiplet as well as Majorana-Weyl spinors in both gravity and vector multiplets. There can also be gauge anomalies involving the fermions in the vector multiplet, which transform in the adjoint representation of the gauge group (they are gaugini). The anomaly polynomial of these theories is computed using (2.7)-(2.11) summing over all chiral fields (dilatino, gravitino, and gaugini), and reduces to the following expressions for the Spin(32)/Z₂ and the $E_8 \times E_8$ theories respectively:

$$P_{12}^{\text{Spin}(32)/\mathbb{Z}_2} = -\frac{p_1 + c_{32,2}}{2} \times \frac{1}{192} \left(16 \, c_{32,2}^2 - 32 \, c_{32,4} + 4 \, c_{32,2} \, p_1 + 3 \, p_1^2 - 4 \, p_2 \right) \,, \quad (2.13)$$

$$P_{12}^{E_8 \times E_8} = -\frac{p_1 + c_{16,2}^{(1)} + c_{16,2}^{(2)}}{2} \times \frac{1}{192} \left(8 \left(c_{16,2}^{(1)} \right)^2 + 8 \left(c_{16,2}^{(2)} \right)^2 + 4 \left(c_{16,2}^{(1)} + c_{16,2}^{(2)} \right) p_1 - 8 c_{16,2}^{(1)} c_{16,2}^{(2)} + 3 p_1^2 - 4 p_2 \right).$$

$$(2.14)$$

Here, the Chern classes are expressed in the vector representations of Spin(32) and the SO(16) subgroups of each E_8 respectively, and the (1) and (2) indices differentiate between each of the two E_8 gauge groups. These expressions agree with the ones in [60] once (2.1)–(2.4) are used to put Chern and Pontryagin classes back into traces. The anomaly polynomial does not vanish, and at first sight this would mean that the theories are inconsistent. However, both anomaly polynomials factorize in the following schematic form:

$$P_{12} = X_4 X_8 \,, \tag{2.15}$$

where the X_4 is a four-form and X_8 is an eight-form. This factorization property is the key to cancelling the anomaly, through what is known as the Green-Schwarz mechanism [60, 61]. There is another field in these theories that can contribute to the aforementioned diagrams: the massless Kalb-Ramond B-field. Consistency of the 10-dimensional supergravity requires that the B-field not be invariant under gauge and gravitational interactions, and in fact it must satisfy an identity of the form

$$dH_3 = X_4 \,, \tag{2.16}$$

where $H_3 \equiv dB_2 - \omega_{\rm CS}$ is the gauge-invariant curvature of the B-field built from the Chern-Simons forms of gauge and spin connections that appears in the (super)gravity couplings. $X_4 = d\omega_{\rm CS}$ is a linear combination of characteristic classes of gauge and gravitational bundles. There is more to the Bianchi identity at the global level, a subject which we will discuss in section 3.1.

In this case, the anomaly corresponding to a factorized anomaly polynomial (2.15) can be cancelled by introducing the following term in the action:

$$-\int B_2 \wedge X_8 \,. \tag{2.17}$$

This term in the action is actually generated by string perturbation theory, as shown in [62] in the heterotic case, and it contributes a term $-dB_2 \wedge X_8$ to the Lagrangian density of the anomaly theory. Using the Bianchi identity and (2.5), we see that this adds a term $-X_4X_8$ to



Figure 1. A diagram showing how the three tachyon-free non-supersymmetric string theories relate to the supersymmetric ones and M-theory, via various worldsheet orbifolds. For instance, the SO(16)² is obtained from the E_8^2 heterotic by orbifolding spacetime and gauge group fermion number, $F = F_L + F_R$. We also list some tachyonic examples via red dot-dashed arrows, although we are not exhaustive.

the anomaly polynomial, exactly canceling the term in (2.15) so that the total local anomaly vanishes. This is the basic principle of the Green-Schwarz mechanism where the anomaly is cancelled by introducing an extra term in the action.

The main focus of this paper is however the three known⁵ ten-dimensional non-tachyonic, non-supersymmetric string theories, which also feature a B-field and a Green-Schwarz mechanism to cancel local anomalies. The diagram in figure 1 sums up how these non-supersymmetric theories are related to eachother and to the supersymmetric ones via gaugings of various worldsheet symmetries. We will now briefly describe the matter content of these theories as well as their anomalies.

• The Sugimoto model

The Sugimoto string [4] can be thought of as the non-supersymmetric sibling of the supersymmetric type I Spin(32)/ \mathbb{Z}_2 string. The main departure for us is that the gauge algebra is $\mathfrak{sp}(16)$ instead, and thus the Chern classes are taken in the fundamental representation of this group.⁶ This distinction arises from the different kind of orientifold projection of type IIB, which introduces anti-D9 branes and an O9 plane with positive Ramond-Ramond charge *and* tension. The sign change in the reflection coefficients for unoriented strings scattering off the O9 is such that the Chan-Paton degeneracies reconstruct representations of the symplectic group Sp(16).

As in the type I case, the closed-string sector arranges into an $\mathcal{N} = 1$ supergravity multiplet, while the chiral fermions from the open-string sector arrange into the anti-

⁵It was recently proven that there are no other examples in the heterotic context [63].

⁶Note that we use $\mathfrak{sp}(16)$ and $\mathrm{Sp}(16)$ instead of the notation USp(32) that is often employed in the literature.

symmetric rank-two representation of the gauge group, leading to the same anomaly polynomial formally. This representation is however reducible and contains a singlet; this is nothing but the Goldstino that accompanies the breaking of supersymmetry [64–67]. The low-energy interactions comply with the expected Volkov-Akulov structure of nonlinear supersymmetry [68–70], although there is no tunable parameter that recovers a linear realization. All in all, since the anomaly polynomial is formally identical to the type I case, it factorizes as follows:

$$P_{12}^{\text{Sugimoto}} = -\frac{p_1 + c_{32,2}}{2} \times \frac{1}{192} \left(16 c_{32,2}^2 - 32 c_{32,4} + 4 c_{32,2} p_1 + 3 p_1^2 - 4 p_2 \right).$$
(2.18)

• The Sagnotti model

The type 0'B string [5, 6] of Sagnotti is built from an orientifold projection of the tachyonic type 0B string, where the unique tachyon-free choice involves an O9 plane with zero tension. The resulting gauge group is U(32), Ramond-Ramond *p*-form potentials with p = 0, 2, 4 (the latter having a self-dual curvature) survive the projection and they get anomalous Bianchi identities for the gauge-invariant curvatures,

$$dH_1 = X_2,$$

 $dH_3 = X_4,$ (2.19)
 $dH_5 = X_6$

where $X_2 = c_{32,1}$, X_4 is formally identical to the one in Sugimoto and type I strings, and X_6 is a polynomial in p_1 , $c_{32,1}$, $c_{32,2}$ and $c_{32,3}$. The Bianchi identity for X_2 tells us that the low-energy gauge group reduces to SU(32), since $c_{32,1}$ is set to zero. Physically, the anomalous Bianchi identity for the RR axion induces the kinetic term $|dC_0 + A|^2$, with A the gauge field of the diagonal $\mathfrak{u}(1)$. This is just the Stückelberg mass term for A. All these RR fields with anomalous Bianchi identities play a crucial role in the cancellation of local anomalies via a more complicated Green-Schwarz mechanism involving a decomposition of the anomaly polynomial [6] of the form

$$P_{12}^{0'B} = X_2 X_{10} + X_4 X_8 + X_6 X_6.$$
(2.20)

As we will see, setting X_6 to zero implies that $c_{32,3}$ is also trivial, and so we shall impose this condition when studying global anomalies of this theory.

• The heterotic model

The case of the SO(16) × SO(16) string [2, 3] is slightly different. Along with its two supersymmetric counterparts, it is the unique ten-dimensional heterotic model that is devoid of tachyons. It is built from a projection of either of the two heterotic models, most directly the $E_8 \times E_8$ one under the projector built from a combination of spacetime fermion number and an E_8 lattice symmetry. As a result, it does not have any chiral fields that are uncharged under the gauge symmetry, and in particular it does not have a gravitino. Its anomaly polynomial was derived in [2, 3] and factorizes as:

$$P_{12}^{\mathrm{SO}(16)^2} = -\frac{p_1 + c_{\mathbf{16},2}^{(1)} + c_{\mathbf{16},2}^{(2)}}{2} \times \frac{1}{24} \left((c_{\mathbf{16},2}^{(1)})^2 + (c_{\mathbf{16},2}^{(2)})^2 + c_{\mathbf{16},2}^{(1)} c_{\mathbf{16},2}^{(2)} - 4 c_{\mathbf{16},4}^{(1)} - 4 c_{\mathbf{16},4}^{(2)} \right),$$
(2.21)

where the (1) and (2) indices differentiate between each of the two SO(16) gauge groups. The Green-Schwarz mechanism is carried by the Kalb-Ramond field, which survives the projection as befits a heterotic model [71].

All in all, local anomalies vanish for all three non-supersymmetric string theories, by the Green-Schwarz mechanism (or a more complicated version of it). This was already known in the literature, but leaves open the possibility for the presence of global anomalies. Global anomalies are those that arise in gauge/diffeomorphism transformations that cannot be continuously connected to the identity. These anomalies are not detected by the anomaly polynomial. In the following section we detail how one can study these anomalies and we evaluate them for the case of the three non-supersymmetric tachyon-free string theories.

3 Global anomalies and bordism groups

In the previous section, we have summarized prior results in the literature regarding anomaly cancellation of ten-dimensional non-supersymmetric string theories via the Green-Schwarz mechanism. Importantly, the Green-Schwarz mechanism only guarantees cancellation of *local* anomalies — it guarantees that the (super)gravity path integral is gauge invariant as long as we only consider gauge transformations infinitesimally close to the identity. More generally, one also need discuss *global* anomalies, namely anomalies in gauge transformations that cannot be continuously deformed to the identity. The archetypal example of such a global anomaly is Witten's SU(2) anomaly [72]. If one includes topology-changing transitions, one has even more general anomalies (dubbed Dai-Freed anomalies in [73]), involving a combination of gauge transformations and spacetime topology change. In this paper, we will take the point of view that such anomalies should cancel in a consistent quantum theory of gravity, where spacetime topology is supposed to fluctuate.

The framework of anomaly theories introduced briefly in the previous section (2.5) can also be used to study global anomalies of Lagrangian theories such as the ones we are interested in. Given a *d*-dimensional quantum field theory, an anomaly on a manifold X_d (possibly decorated with gauge field, spin structure, etc.) means that the partition function $Z(X_d)$ is not invariant under gauge transformations (or diffeomorphisms, for the case of a gravitational anomaly). In a modern understanding (see [73–77] for detailed reviews, and also [78, 79] for a discussion in the context of the 6d Green-Schwarz mechanism), the anomaly can be captured by an invertible (d + 1)-dimensional field theory α with the property that, when evaluated on a manifold with boundary Y_{d+1} with $\partial Y_{d+1} = X_d$, the product

$$Z(X_d) \cdot e^{-2\pi i \alpha(Y_{d+1})} \tag{3.1}$$

is invariant under gauge transformations. The *d*-dimensional QFT arises as a boundary mode of the (d + 1)-dimensional invertible field theory α , and the anomaly is re-encoded

in the fact that (3.1) is not the partition function of a d-dimensional quantum field theory — its value depends in general on Y_{d+1} and the particular way on which the fields on X_d are extended to Y_{d+1} .

In general, it may be very difficult to determine α . However, in weakly coupled Lagrangian theories, we have a prescription to associate an anomaly theory to each of the chiral degrees of freedom involved. For instance, the anomaly theory for a Weyl fermion in *d*-dimensions (d even) is given by the so-called eta invariant of a (d + 1)-dimensional Dirac operator with the same quantum numbers as the fermion we started with [75, 80],

$$e^{2\pi i \alpha_{\text{fermion}}(Y_{d+1})} = e^{2\pi i \eta_{d+1}(Y_{d+1})}.$$
(3.2)

If one has several fermions, the total anomaly theory is simply the product of these (so that the η invariants add up). There are other topological couplings that can also contribute to the anomaly theory, as we will see below.

Two different open manifolds Y_{d+1} and Y'_{d+1} , both having X_d as a boundary, will yield values for the partition function (3.1) differing by a factor

$$e^{2\pi i\alpha(Y_{d+1})}e^{-2\pi i\alpha(Y'_{d+1})} = e^{2\pi i\alpha(Y_{d+1}\cup\overline{Y'}_{d+1})}.$$
(3.3)

The manifold $Y_{d+1} \cup \overline{Y'}_{d+1}$ is just a general closed (d+1)-dimensional manifold. In an anomaly free-theory, the partition function in (3.1) should not depend on the choice of extension; therefore, in this picture, anomaly cancellation is simply the statement that the anomaly theory $\alpha(\tilde{Y}_{d+1})$ be trivial when evaluated on a closed manifold \tilde{Y}_{d+1} .

The particular case in which \tilde{Y}_{d+1} itself is a boundary, $\tilde{Y}_{d+1} = \partial Z_{d+2}$, corresponds to local anomalies, which allows us to connect the discussion to the preceding section. The η invariants introduced above, that give the anomalies for chiral fermions, can in this case be evaluated by means of the APS index theorem [81],

$$\eta(\tilde{Y}_{d+1}) = \text{Index} - \int_{Z_{d+2}} P_{d+2}, \qquad (3.4)$$

where P_{d+2} is the anomaly polynomial of the previous subsection, and the "Index" is an integer. We thus recover the usual, perturbative, anomaly cancellation condition in terms of the anomaly theory. In theories where anomalies are cancelled via the Green-Schwarz mechanism, another ingredient is necessary. The ten-dimensional action has an extra Green-Schwarz term (2.17), which is the boundary mode of an 11d invertible field theory

$$\alpha_{\rm GS}(Y_{11}) = \int_{Y_{11}} H \wedge X_8. \tag{3.5}$$

The total anomaly theory is therefore the sum of the fermion anomaly and $\alpha_{\rm GS}(Y_{11})$. On a manifold which is itself a boundary, $\tilde{Y}_{11} = \partial Z_{12}$,

$$\int_{\tilde{Y}_{11}} H \wedge X_8 = \int_{Z_{12}} dH \wedge X_8 = \int_{Z_{12}} X_4 \wedge X_8, \tag{3.6}$$

where in the last equality we used the constraint that we are restricting to twisted string manifolds satisfying the (anomalous) Bianchi identity $dH = X_4$. Taking this last contribution

into account, we see that the local anomaly coming from the GS term can cancel that of the fermions, provided that the anomaly polynomial factorizes as discussed in section 2.

In the rest of this paper, we will assume that local anomalies cancel, and ask what is the value of the total anomaly theory,

$$e^{2\pi i \alpha_{\rm tot}} = e^{2\pi i \alpha_{\rm fermions}} e^{2\pi i \alpha_{\rm GS}} \tag{3.7}$$

when evaluated on 11-dimensional closed manifolds which are *not* boundaries. This task seems daunting at first, since, depending on the collection of background fields, there can be infinitely many such manifolds. Fortunately, one can prove⁷ [75] that the partition function of the anomaly theory $\alpha_{tot} \pmod{1}$ is a *bordism invariant*,⁸

$$e^{2\pi i \alpha_{\rm tot}(Y_{11}^{(1)})} = e^{2\pi i \alpha_{\rm tot}(Y_{11}^{(2)})} \quad \text{if} \quad Y_{11}^{(1)} \cup \overline{Y_{11}^{(2)}} = \partial Z_{d+2}.$$
(3.8)

This reduces the problem significantly: since $\alpha_{tot} \pmod{1}$ is a bordism invariant, one need only evaluate it on a single representative per bordism class. Furthermore, these classes form an abelian group, the bordism group (of manifolds suitably decorated with a twisted string structure and gauge bundle). Bordism groups have appeared prominently in the field theory and quantum gravity literature, and there are many techniques available for their computation (see [51] for a detailed introduction). Thus, to compute these anomalies, we will just compute the relevant bordism groups and evaluate the anomaly theory on generators. Notice that if it happens that the relevant bordism group Ω_{11} is 0, there are no global anomalies to check! That this happens was in fact shown by Witten in [37, 38] for the $E_8 \times E_8$ string when one does not take into account the \mathbb{Z}_2 symmetry switching the two E_8 gauge fields.⁹ See [39] for an analysis of type I string.

More recently, [42, 45] used the Stolz-Teichner conjecture to analyze global anomalies in supersymmetric, heterotic string theory even in stringy backgrounds, lacking a geometric description. In this paper we content ourselves with the target space treatment described above, which may miss anomalies of non-geometric backgrounds. In the following, we present the calculation and results for the three ten-dimensional non-supersymmetric string theories described in section 2. But before that, we will describe and justify more carefully the precise structure that will be assumed in our bordism calculations.

3.1 Bianchi identities and twisted string structures

As described in the previous subsection, the computation of global anomalies can be organized in terms of a bordism calculation and an anomaly theory, which is just a homomorphism from the bordism group to U(1). The precise bordism group to be used (i.e. the particular

⁷The proof is a straightforward application of the APS index theorem (3.4), see [75].

⁸There is also a more theoretical and more general proof that the partition function of the anomaly theory is a bordism invariant, due to Freed-Hopkins-Teleman [82] and Freed-Hopkins [83]; they show that up to a deformation, which is irrelevant for anomaly calculations, the partition function of *any* reflection-positive invertible field theory is a bordism invariant.

⁹If one does want to take this \mathbb{Z}_2 symmetry into account, for example to study the CHL string, the relevant Ω_{11} is nonzero [46, Theorem 2.62], and it was not known whether the global anomaly cancels. We will show that it does cancel in this paper, in section 4.

structure that our manifolds are required to have) depends on the theory we are interested in. For instance, all heterotic string theories under consideration include fermions, so we will consider only manifolds (and bordism between them) carrying a spin structure; the anomaly theory is related to the η invariant for a certain Dirac operator on this manifold. This means that the second Stiefel-Whitney class of the allowed manifolds where the anomaly theory is to be evaluated will vanish,

$$w_2 = 0.$$
 (3.9)

In heterotic string theories, also the Bianchi identity (2.16) needs to be taken into account. Equation (3.6) illustrates that cancellation of perturbative anomalies requires us to assume $dH = X_4$ even off-shell.¹⁰ Therefore, we will restrict our bordism groups to consist of 11-dimensional manifolds in which (2.16) is satisfied. In particular, we will set

$$\int_{M_4} X_4 = 0 \tag{3.10}$$

for any closed 4-manifold M_4 . The precise expression of X_4 in terms of characteristic classes depends on the particular theory under study. The particular case

$$X_4 = \frac{p_1}{2} \tag{3.11}$$

has been studied in the mathematical literature, and receives the name of a *string* structure. The X_4 's that appear in heterotic string theories are always of the form

$$X_4 = a \frac{p_1}{2} (\text{Tangent bundle}) + b c_2 (\text{Gauge bundle}), \quad a, b \in \mathbb{Z},$$
(3.12)

and we will refer to the data of a solution to this equation for chosen a and b as a *twisted* string structure. This notion appeared in the mathematical literature in [84, Definition 8.4].

The bordism groups related to the three non-supersymmetric string theories we are going to consider are

$$\Omega_{11}^{\text{String}-Sp(16)}, \quad \Omega_{11}^{\text{String}-\text{SU}(32)\langle c_3 \rangle}, \quad \Omega_{11}^{\text{String}-Spin(16)^2}, \tag{3.13}$$

for the Sugimoto, Sagnotti, and $SO(16) \times SO(16)$ heterotic theories, respectively; these are the bordism groups of twisted string manifolds where the particular choice of twisted string structure is spelled out by the Green-Schwarz mechanisms for these theories as we discussed in section 2.

Remark 3.14. Before presenting the results for the bordism groups, we must discuss an important subtlety, which affects the bordism calculation. Up to this point in this paper, we have been cavalier when writing down characteristic classes such as " p_1 " or " c_2 ", and defined these characteristic classes as closed differential forms (e.g. in (2.4)) by way of Chern-Weil theory. However, these differential forms have quantized periods, as is the case for data coming out of any quantum theory, and a proper treatment of the Green-Schwarz mechanism should take this into account. There are two ways to do this.

 $^{^{10}}$ If we insist on keeping the *B*-field as a background; see the discussion at the end of this section.

- 1. The simplest approach is to lift to \mathbb{Z} -valued cohomology: the quantized periods are a reminder that the de Rham classes of the Chern-Weil forms of p_1 , c_2 , etc., lift canonically to classes in $H^4(BG;\mathbb{Z})$ for various Lie groups G, and on many manifolds M, these integer-cohomology lifts of these characteristic classes can be torsion! Thus it is natural to wonder whether the B-field should be an element of $H^3(-;\mathbb{Z})$ and the Bianchi identity (3.12) should take place in $H^4(-;\mathbb{Z})$. The definitions of string structure and twisted string structure in mathematics assume this lift has taken place.
- 2. Alternatively, one could lift to differential cohomology $\check{H}^4(-;\mathbb{Z})$, which amounts to observing that it is not just the Z-cohomology lift which is natural, but also the data of the Chern-Weil form; differential cohomology is a toolbox for encoding both of these pieces of data. Indeed, for any compact Lie group G and class $c \in H^*(BG;\mathbb{Z})$, there is a canonical differential refinement $\check{c} \in \check{H}^*(B_{\nabla}G;\mathbb{Z})$ [85, 86], where $B_{\nabla}G$ is the classifying stack of G-connections.¹¹ Thus we could instead ask: should we begin with a B-field in $\check{H}^3(-;\mathbb{Z})$ and ask for the Bianchi identity to take place in $\check{H}^4(-;\mathbb{Z})$? This combines the two other formalisms we considered, differential forms and integral cohomology.

The answer in the mathematics literature is often the second option, beginning with Freed [39, section 3] and continuing in, for example, [46, 87–104]. In particular, [94, 95] interpret the data entering into the Green-Schwarz mechanism as specifying a connection for a Lie 2-group built as an extension of the gauge group by BU(1), providing an appealing physical interpretation of the lift to differential cohomology.

We are interested in classifying anomalies, and while there is an interesting differential refinement of the story of the bordism classification of anomalies due to Yamashita-Yonekura [105–107], the deformation classification of anomalies ultimately can proceed without differential-cohomological information, because it boils down to studying bordism groups. Because of this, we will work with characteristic classes in integral cohomology, noting here that the correct setup of the Green-Schwarz mechanism taking torsion and Chern-Weil forms into account uses differential cohomology, and that for our computations it makes no difference.

Note that cancellation of perturbative anomalies around (3.6) only requires the free part of $X_4 \in H^4(-;\mathbb{Z})$ to be trivial in a compact manifold, and poses no obvious restriction on torsion. Reference [38] studies a particular example suggesting that this should be the case, but does not attempt to make a general argument. To ascertain whether the torsion piece of X_4 must also be trivialized or not, consider the physical origin of the Bianchi identity, which is itself a two-dimensional version of the Green-Schwarz mechanism described above (see e.g. [58, 108]). Consider a worldsheet wrapped on a 2-manifold Σ_2 of the ambient ten-dimensional spacetime manifold M_{10} . The configuration should be invariant under spacetime diffeomorphisms, and gauge transformations, which are manifested as global symmetries of the worldsheet. However, in heterotic or type I theories, the worldvolume degrees of freedom are chiral, and anomalous under these transformations. The anomaly theory, which we denote $\alpha^{\text{worldsheet}}$, is encoded by

¹¹If you do not want to think about stacks, this statement is essentially equivalent to the notion that for a principal *G*-bundle $P \to M$ with connection Θ , the differential characteristic class $\check{c}(P,\Theta) \in \check{H}^*(M;\mathbb{Z})$ is natural in (P,Θ) .

a three-dimensional η invariant. Applying the APS index theorem (3.4), we obtain

$$\exp(2\pi i\alpha^{\text{worldsheet}}) = \exp\left(2\pi i\int X_4\right),\tag{3.15}$$

where X_4 is a certain differential form built out of characteristic classes, and which is precisely the X_4 appearing above (indeed, (3.15) is usually taken to give the definition of X_4). As things stand, any configuration with an insertion of a fundamental string worldsheet on Σ_2 has a gravitational anomaly; however, the worldsheet also has an electric coupling to the B-field,

$$\exp\left(2\pi i \int_{\Sigma_2} B_2\right),\tag{3.16}$$

whose anomaly theory is simply

$$\alpha_B = \int H. \tag{3.17}$$

Now, the total worldsheet anomaly is

$$\exp\left(2\pi i\alpha_{\text{total}}^{\text{worldsheet}}\right) = \exp\left(2\pi i\alpha^{\text{worldsheet}}\right)\exp\left(2\pi i\alpha_B\right).$$
(3.18)

The physical consistency condition is that the total anomaly is trivial

$$\exp\left(2\pi i\alpha_{\text{total}}^{\text{worldsheet}}\right) = 1, \quad \text{for all } M_3. \tag{3.19}$$

When M_3 is a boundary, anomaly cancellation is achieved, as above, by setting $dH = X_4$, precisely the Bianchi identity described above. However, this is not all there is to (3.19). Assuming that anomalies vanish when M_3 is a boundary, $\exp(2\pi i \alpha_{\text{total}}^{\text{worldsheet}}(M_3))$ is actually only dependent on the integer homology class of M_3 . In fact, since it is a map that assigns a phase to each 3-cycle in the ambient 10-dimensional manifold M_{10} , it can be regarded as an element of $H^3(M_{10}; U(1))$, with U(1) coefficients. Using the long exact sequence in cohomology associated to the short exact sequence of groups $\mathbb{Z} \to \mathbb{R} \to U(1)$, we obtain that [76]

$$H^{3}(M_{10};\mathbb{R}) \to H^{3}(M_{10};\mathrm{U}(1)) \to H^{4}(M_{10};\mathbb{Z}) \to H^{4}(M_{10};\mathbb{R}),$$
 (3.20)

where the third map is taking the free part of the integer cohomology class. In general, $\exp(2\pi i \alpha_{\text{total}}^{\text{worldsheet}})$ will have pieces both in the image of the first map and in its cokernel. An example where the anomaly theory has a non-trivial piece in the image of the first map of (3.20) can be obtained by compactifying heterotic string theory on a Bieberbach 3- manifold, a fixed-point free quotient of the torus T^3 .¹² Since T^3 is Riemann-flat, a quick analysis would suggest that the Bianchi identity is satisfied automatically with no gauge bundle or B-field turned on. However, trying to implement this manifold directly in the worldsheet results in a theory which is not level-matched. The problem is that $\exp(2\pi i \alpha_{\text{total}}^{\text{worldsheet}})$ with no gauge bundle turned on is nontrivial for most Bieberbach manifolds, and so the anomaly theory is a nontrivial class in $H^3(M_{10}; \mathbb{R})$. Cancelling this anomaly forces either a B-field (discrete torsion) to be turned on, or a non-trivial flat gauge bundle to be present.

 $^{^{12}\}mathrm{We}$ thank Cumrun Vafa for pointing out this example to one of us.

The rest of the anomaly theory is in the image of the second map in (3.20), and can therefore be represented by a certain torsion integer cohomology class in $H^4(M_{10};\mathbb{Z})$, whose free part vanishes. We will now show that this is in fact the torsion part of $X_4 - dH$. Consider a torsion 3-cycle M_3 of order k, i.e. such that kM_3 is the boundary of some 4-manifold N_4 . Let us see how to compute the anomaly in this case. First, the anomaly theory $\alpha_{\text{total}}^{\text{worldsheet}}$ is a linear combination of η invariants, which in this particular case can be re-expressed as linear combinations of gravitational and gauge Chern-Simons numbers as discussed above. The Chern-Simons invariant is additive on disconnected sums, and so, we have

$$\alpha_{\text{total}}^{\text{worldsheet}}(M_3) = \frac{1}{k} \alpha_{\text{total}}^{\text{worldsheet}}(kM_3).$$
(3.21)

Next, we can use the fact that $kM_3 = \partial N_4$, to write (after exponentiation)¹³

$$\exp\left(2\pi i\alpha_{\text{total}}^{\text{worldsheet}}(M_3)\right) = \exp\left(\frac{2\pi i}{k}\left(\text{Index}_{N_4} + \int_{N_4} (X_4 - dH)\right)\right).$$
(3.22)

This expression is not obviously independent of the choice of N_4 , but when $(X_4 - dH)$ is pure torsion, it actually is. The reason is that the quantity $\int_{N_4} (X_4 - dH)$ may be rewritten as a linking pairing in homology [109]. If we Poincaré dualize $(X_4 - dH)_{\text{tor}}$ to a torsion 6-cycle M_6 , the linking pairing between M_6 and M_3 is constructed by choosing a boundary N_4 for kM_3 and computing $\int_{N_4} (X_4 - dH)$ modulo k. Importantly, the result does not depend on the choice of N_4 (see [109] for a review and proof of these facts).

In short, the full analog of the Bianchi identity is (3.19). Unpacking this condition, we recover that:

• There is the condition on any 3-manifold M_3 that

$$\int_{M_3} H = \int_{M_3} CS_3^{X_4}, \tag{3.23}$$

where $CS_3^{X_4}$ is a (local) Chern-Simons form obeying $dCS_3^{X_4} = X_4$. This will force discrete B-fields to be turned on in certain situations, such as on Bieberbach manifolds (these were referred to as "worldsheet discrete theta angles" in [79]).

• As a consequence of the previous point, when M_3 is a boundary, we get that the Bianchi identity $dH = X_4$ must hold over the integers.

The general analysis we just carried out is somewhat abstract; in the next subsection, we will verify its correctness by explicitly checking, in a variety of backgrounds, that anomalies in ten dimensions only cancel if the torsional part of the Bianchi identity holds.

Finally, we comment on another possible way in which the anomaly calculation could have been set, avoiding the calculation of string bordism groups altogether, as in [79]. Anomalies are always studied with respect to a choice of background fields. The approach we have followed here takes the metric g, the gauge field A, and the 2-form field B as background fields, and imposes the Bianchi identity as a restriction on the allowed backgrounds. However,

¹³The anomaly theory is a linear combination of real-valued eta invariants, thus division by k is well-defined and there is no phase ambiguity.

in a quantum theory of gravity, there are no global symmetries, and therefore, there are no background fields either. This is manifested in the fact that all three of q, A, B are actually dynamical fields that we are supposed to path-integrate over. Treating these as backgrounds is justified if there is some sort of weak coupling limit in which the fields become frozen. This is automatically the case at low energies in any ten-dimensional string theory, since the couplings of all of q, A, B are dimensionful and become irrelevant in the deep IR. It is not the case e.g. in six dimensions, where antisymmetric tensor fields are often strongly coupled and cannot be treated perturbatively. In such cases, the only approach available is to explicitly perform the path integral over the tensor fields, compute their effective action, and verify that the resulting path integral indeed cancels against the contributions of other chiral fields. There is no meaningful analog of the notion of having a string structure, since no weak coupling notion is available. The anomaly theory (as a function of the metric and background gauge fields) can then studied on general spin manifolds (and not just string manifolds), and anomalies cancel in a standard way, because the B-field (which is integrated over) couples to background 4 and 8-forms X_4 and X_8 , and has a mixed anomaly captured by the anomaly polynomial $\int X_4 X_8$, just what is needed to cancel the anomaly of the fermions. From a perturbative string worldsheet point of view, we feel it is more natural to keep Bas a background field; furthermore, the techniques we use in this paper can be extended to compute lower-dimensional string bordism groups, which control solitonic objects in these non-supersymmetric theories via the Cobordism Conjecture [48].

3.2 Evidence for torsional Bianchi identities

In the previous subsection, we gave an argument that the Bianchi identity holds at the level of torsion, too. The argument relies heavily on string perturbation theory, and one may worry e.g. that it does not capture strongly coupled situations. In this section, we provide independent evidence, which does not rely on the worldsheet at all, that the Bianchi identity holds at the level of integer cohomology. We do so by computing Dai-Freed anomalies of supersymmetric and non-supersymmetric string theories on simple eleven-dimensional lens spaces. Lens spaces are quotients of spheres by \mathbb{Z}_p groups; they are the simplest examples of manifolds whose cohomology is purely torsional (except in bottom and top degrees, as usual). In particular, their first Pontryagin classes are torsion; the upshot of the calculation in this section is that spacetime anomalies on lens spaces seem to vanish if and only if the Bianchi identities are satisfied at the level of integral cohomology, including torsional classes.

Now we turn to the details of evaluating anomalies on lens spaces; we refer the reader to [40, 51] for more on lens spaces and the corresponding expressions for eta invariants. (Eleven-dimensional) lens spaces are defined to be quotients of the form

$$L_p^{11} = S^{11} / \mathbb{Z}_p, \tag{3.24}$$

where the \mathbb{Z}_p action acts as scalar multiplication by $e^{2\pi i/p}$ on the six complex coordinates \mathbb{C}^6 and where we embed the covering S^{11} as the unit sphere. An important property of these lens spaces is that the Green-Schwarz term,

$$H \wedge X_8, \tag{3.25}$$

will automatically vanish on lens spaces, since $H^3(S^{11}/\mathbb{Z}_n;\mathbb{Z}) = 0$. As a result, the calculation of the full anomalies of string theories on lens spaces reduces to determining the anomalies of the chiral fields. We will now evaluate the anomaly theory of the Type I and the Heterotic string theories (supersymmetric and non-supersymmetric) on certain eleven-dimensional lens spaces.

3.2.1 Type I and HO heterotic theories

As the Green-Schwarz (GS) contribution to the anomaly theory vanishes on lens spaces, the remaining fermion anomaly theory of the type I and HO heterotic theory is given by

$$\alpha(L_p^{11}) = \eta_0^{\rm RS}(L_p^{11}) - 3\,\eta_0^{\rm D}(L_p^{11}) + \eta_{\rm adj}^{\rm D}(L_p^{11})\,.$$
(3.26)

The Rarita-Schwinger eta invariant η^{RS} arises from the anomaly theory of a ten-dimensional gravitino according to $\alpha_{\text{gravitino}} = \eta^{\text{RS}} - 2\eta^{\text{D}}$ [40]. In order to evaluate this anomaly theory on L_p^{11} , one can derive the branching rules for the adjoint representation of Spin(32) in terms of the charge-q irreducible \mathbb{Z}_p representations \mathcal{L}^q . This branching depends on how \mathbb{Z}_p is included in the gauge group. We choose a family of inclusions of the form

$$\mathbb{Z}_p \hookrightarrow \mathrm{U}(1) \stackrel{k}{\hookrightarrow} \mathrm{SU}(N) \hookrightarrow \mathrm{Spin}(2N), \qquad (3.27)$$

according to which the (complexified) vector representation of Spin(2N) splits as

$$\mathbf{2N} \longrightarrow \mathbf{N} \oplus \mathbf{N}^* \,. \tag{3.28}$$

The parameter k denotes an inclusion that places the U(1) fundamental representation \mathcal{L} in k diagonal blocks, in pairs $L \equiv \mathcal{L} \oplus \mathcal{L}^{-1}$, and the rest in the trivial representation \mathcal{L}^{0} . Then, the vector representation of Spin(2N) further splits into

$$\mathbf{2N} \longrightarrow \mathbf{N} \oplus \mathbf{N}^* \longrightarrow [kL \oplus (N-2k)\mathcal{L}^0] \oplus [kL \oplus (N-2k)\mathcal{L}^0].$$
(3.29)

In order to find the branching rules for other representations, it is convenient to use Chern characters. Letting $x \equiv c_1(\mathcal{L})$, the Chern character of **N** (and **N**^{*}) decomposes into

$$ch(\mathbf{N}) \longrightarrow k \left(e^x + e^{-x} \right) + \left(N - 2k \right).$$
(3.30)

Then we can build the characters for adjoint, symmetric and antisymmetric SU(N) representations, from which we can reconstruct the characters for Spin(2N) representations of interest, such as the adjoint (antisymmetric) and spinorial. The resulting branching rules involve the representations $L^q \equiv \mathcal{L}^q \oplus \mathcal{L}^{-q}$. In particular, the adjoint of Spin(2N) branches according to

$$\mathbf{adj} \longrightarrow k(2k-1)L^2 \oplus 4k(N-2k)L \oplus [N(2N-1) - 2k(2k-1) - 8k(N-2k)]\mathcal{L}^0, \quad (3.31)$$

which gives the corresponding eta invariant. Using the expressions

$$\begin{split} \eta^{\rm D}_q(L^{11}_p) &= \frac{2p^6 + 21p^4 + 168p^2 - 191 - 42p^4q^2 + 210p^2q^4 - 630p^2q^2}{60480p} \\ &\quad + \frac{-252pq^5 + 1260pq^3 - 1008pq + 84q^6 - 630q^4 + 1008q^2}{60480p} \,, \end{split} \tag{3.32}$$

$$\eta^{\rm RS}_0(L^{11}_p) &= \frac{22p^6 - 273p^4 - 3192p^2 + 3443}{60480p} \,, \end{split}$$

the anomaly simplifies to

$$\alpha_{\text{Spin}(32)}^{(k)}(L_p^{11}) = \frac{(p^2 - 1)(p^4 + (11 - 5k)p^2 + 10(k - 3)^2)}{60p}.$$
(3.33)

In order to compare the cases in which $\alpha = 0 \mod 1$ to the Bianchi identity, let us recall that the total Pontryagin class of L_p^{2k-1} is $p(L_p^{2k-1}) = (1+y)^k$ with y a generator of $H^4(L_p^{2k-1};\mathbb{Z}) \cong \mathbb{Z}_p$. Thus $p_1(L_p^{11}) = 6y$, and one can show that the canonical choice of $\frac{p_1}{2}$ afforded by the spin structure is $\frac{p_1}{2} = 3y$. On the other hand, according to the branching rule (3.29), the total Chern class of the associated vector bundle is $c = (1-y)^{2k}$, and thus $c_2 = -2ky$. Therefore,

$$\frac{p_1 + c_2}{2} = (3 - k) y \tag{3.34}$$

vanishes if and only if $k = 3 \mod p$. Plugging in k = 3 + mp with m integer, the anomaly does vanish (mod 1), and it does not vanish otherwise.

3.2.2 $E_8 \times E_8$ theory

The calculation for the $E_8 \times E_8$ theory is almost the same as in the preceding case. The anomaly theory has the same form of (3.26), the only difference being the branching of the adjoint representation $\mathbf{adj} = (\mathbf{248}, \mathbf{1}) \oplus (\mathbf{1}, \mathbf{248})$. We employ the same construction as before, embedding \mathbb{Z}_p into the Spin(16) subgroup of E_8 . The general construction is thus specified by a pair (k_1, k_2) pertaining to the two E_8 factors. One then has to compute the branching for the **120** and the spinorial **128** of Spin(16) which compose the adjoint representation of E_8 . The former has been presented in the preceding section, now with N = 8, while the latter can be constructed computing Chern characters of antisymmetric representations of SU(8) whose direct sum gives the branching of the spinorial representation:

$$\mathbf{128}_{+} \oplus \mathbf{128}_{-} \longrightarrow \bigoplus_{m=0}^{8} \begin{pmatrix} \mathbf{8} \\ \mathbf{m} \end{pmatrix}.$$
(3.35)

The Chern character for the various antisymmetric representations can be found by expanding the graded Chern character for the exterior algebra $\Lambda(V) = \bigoplus_n \Lambda^n(V)$

$$\operatorname{ch}(\Lambda(V)) \equiv \sum_{n} t^{n} \operatorname{ch}(\Lambda^{n}(V)), \qquad (3.36)$$

which can be computed exploiting the property $\Lambda(U \oplus V) \simeq \Lambda(U) \otimes \Lambda(V)$ and that, for line bundles \mathcal{L} ,

$$\operatorname{ch}(\Lambda(\mathcal{L})) = 1 + t \, e^{c_1(\mathcal{L})} \,. \tag{3.37}$$

Thus, (3.30) gives

$$ch(\Lambda(\mathbf{N})) \longrightarrow (1+t e^x)^k (1+t e^{-x})^k (1+t)^{N-2k}.$$
 (3.38)

For instance for N = 8 and k = 1, summing the even or odd rank characters leads to

$$ch(128) \longrightarrow 64 + 32 (e^x + e^{-x}),$$
 (3.39)

which means that the spinorial representations branch according to $128 \rightarrow 32L \oplus 64\mathcal{L}^0$. Analogously, $120 \rightarrow L^2 \oplus 24L \oplus 70\mathcal{L}^0$, so that all in all

$$\mathbf{248} \quad \longrightarrow \quad L^2 \oplus 56L \oplus 134\mathcal{L}^0 \,. \tag{3.40}$$

The anomaly for this particular choice $(k_1, k_2) = (1, 0)$ then simplifies to

$$\alpha_{E_8 \times E_8}^{(1,0)}(L_p^{11}) = \frac{p^6 + 5p^4 + 34p^2 - 40}{60p} \tag{3.41}$$

which vanishes (mod 1) for p = 2. Let us now look at the Bianchi identity. The Chern class of the adjoint $E_8 \times E_8$ associated bundle is

$$c = (1 - 4y) (1 - y)^{56}, \qquad (3.42)$$

with y a generator of $H^4(L_p^{11};\mathbb{Z})$, and thus $c_2 = -60y$. For $E_8 \times E_8$ we have to divide $\frac{c_2}{2}$ by 30 in the Bianchi identity, thus getting

$$\frac{p_1}{2} + \frac{c_2}{60} = 2y. \tag{3.43}$$

This class only vanishes if p = 2, which is the same value for which the anomaly vanishes! Similarly, for $(k_1, k_2) = (1, 1)$ the Bianchi class is y, which never vanishes (except for the trivial case p = 1), and accordingly the anomaly never vanishes either.

One can carry on with more complicated embeddings computing the spinorial branching of **128**: for $(k_1, k_2) = (2, 1)$ the Bianchi class vanishes, and indeed the anomaly turns out to always vanish mod 1. At first glance, the case $(k_1, k_2) = (2, 0)$ appears to present an exception, since the Bianchi class is $y \neq 0$ but the anomaly vanishes for p = 5. However, in order to find the relationship between torsional Bianchi identities and anomalies, for given torsion the anomaly should vanish for all allowed backgrounds, and the (1, 1) embedding has the same Bianchi class but nonvanishing anomaly for p = 5.

The general expression for any (k_1, k_2) for the $E_8 \times E_8$ theory is more involved due to how the spinorial representations branch, but the procedure to compute the anomaly is systematic.

3.2.3 Non-supersymmetric theories

Let us now address the non-supersymmetric cases. An immediate consequence of the above result for the supersymmetric heterotic theories is that the anomaly on lens spaces satisfying the torsional Bianchi identity also vanishes for the non-SUSY heterotic theory, since its chiral matter content is in the virtual difference of the corresponding representations [71]. This fact will turn out to be useful when discussing fivebrane anomaly inflow in section 3.4.2.

For the Sagnotti model, the anomaly theory can be written as [5, 6]

$$\alpha_{0'B}(L_p^{11}) = \alpha_{\text{self-dual}}(L_p^{11}) - \eta_{\text{antisym}}^{D}(L_p^{11}) = -\alpha_0^{\text{RS}}(L_p^{11}) + 3\eta_0^{D}(L_p^{11}) - \eta_{\text{antisym}}^{D}(L_p^{11})$$
(3.44)

since it contains a four-form RR field with self-dual curvature, similarly to type IIB. Following the same procedure as before, now with the simpler inclusion $\mathbb{Z}_p \hookrightarrow U(1) \hookrightarrow SU(32)$, one can evaluate the fermionic anomalies; for the self-dual field, in the second line of eq. (3.44) we have used anomaly cancellation in type IIB supergravity to recast its anomaly theory in terms of fermionic contributions, along the lines of [40]. Thus we obtain

$$\alpha_{0'B}^{(k)}(L_p^{11}) = -\frac{(p^2 - 1)(5k^2 - 5k(p^2 + 12) + 2(p^4 + 11p^2 + 90))}{120p}.$$
(3.45)

The Chern class of the associated fundamental bundle is now $c = (1 - y)^k$, so that $c_2 = -ky$ and the Bianchi class

$$\frac{p_1 + c_2}{2} = \left(3 - \frac{k}{2}\right)y \tag{3.46}$$

vanishes for $k = 6 \mod p$. Notice that $c_3 = 0$ as well for these bundles, since the total Chern class only contains powers of $y \in H^4(L_p^{11}, \mathbb{Z})$. Substituting k = 6 + mp for integer m, the anomaly vanishes as expected, but not otherwise.

The calculation for the Sugimoto model is essentially identical: the anomaly theory is simply $\alpha_{\text{Sugimoto}} = -\alpha_{0'B}$, since the antisymmetric fermion has now positive chirality and the gravitino and dilatino contribute the opposite of the self-dual tensor. The inclusion we employ is $\mathbb{Z}_p \hookrightarrow U(1) \hookrightarrow \text{Sp}(1) \simeq \text{SU}(2) \stackrel{k}{\hookrightarrow} \text{Sp}(16)$, under which the **32** representation branches according to

$$32 \quad \longrightarrow \quad kL \oplus (32 - 2k)\mathcal{L}^0 \,. \tag{3.47}$$

Since the resulting Bianchi class is also the same, one obtains the same result: the anomalies cancel on lens backgrounds which satisfy the Bianchi identity at the torsional level.

3.3 Vanishing bordism classes

We now turn to the main results of this paper — the calculation of string bordism groups with twisted string structures corresponding to the non-supersymmetric strings, by means of homotopy theory. These sections cover in some detail the mathematical aspects of the calculation; a table summarizing the results can be found in the Conclusions.

$3.3.1 \quad Sp(16)$

At this point we make our first bordism computation: that every closed, spin 11-manifold M with a principal Sp(16)-bundle P satisfying the Green-Schwarz identity $\frac{1}{2}p_1(M) + c_2(P) = 0$ is the boundary of a compact spin 12-manifold on which the Sp(16)-bundle and Green-Schwarz data extend. This implies that the anomalies we study in this paper vanish for the Sugimoto string.

To make these computations, we use the Adams and Atiyah-Hirzebruch spectral sequences. By now these are standard tools in the mathematical physics literature, so we point the reader to [51, 110] for background and many example computations written for mathematical physicists. The computations in this paper are a little more elaborate: twisted string bordism rather than twisted spin bordism. There are fewer such calculations in the literature, but we found the references [46, 111–114] helpful.

On to business. The data of a G-gauge field and a B-field satisfying a Bianchi identity is expressed mathematically as a principal bundle for a Lie 2-group extension of G by
BU(1). Such extensions are classified by $H^4(BG;\mathbb{Z})$ [115, Corollary 97]. Let String(n)-Sp(16) denote the Lie 2-group which is the extension of Spin(n) × Sp(16) by BU(1) classified by $\frac{1}{2}p_1 + c_2 \in H^4(B(\text{Spin}(n) \times \text{Sp}(16));\mathbb{Z})$, and String-Sp(16) be the colimit as $n \to \infty$ as usual. A string-Sp(16)-structure on a manifold M is data of a spin structure, a Sp(16)-bundle P, and a trivialization of the Green-Schwarz term $\frac{1}{2}p_1(M) + c_2(P)$: exactly what we need for the Sugimoto string.

Though we are primarily interested in showing $\Omega_{11}^{\text{String-Sp}(16)} = 0$, the lower-dimensional bordism groups are barely more work.

Theorem 3.48. The low-dimensional String-Sp(16) bordism groups are:

$$\begin{split} \Omega_0^{\mathrm{String-Sp}(16)} &\cong \mathbb{Z} & \Omega_6^{\mathrm{String-Sp}(16)} \cong \mathbb{Z}_2 \\ \Omega_1^{\mathrm{String-Sp}(16)} &\cong \mathbb{Z}_2 & \Omega_7^{\mathrm{String-Sp}(16)} \cong \mathbb{Z}_4 \\ \Omega_2^{\mathrm{String-Sp}(16)} &\cong \mathbb{Z}_2 & \Omega_8^{\mathrm{String-Sp}(16)} \cong \mathbb{Z}^{\oplus 3} \oplus \mathbb{Z}_2 \\ \Omega_3^{\mathrm{String-Sp}(16)} &\cong 0 & \Omega_9^{\mathrm{String-Sp}(16)} \cong (\mathbb{Z}_2)^{\oplus 3} \\ \Omega_4^{\mathrm{String-Sp}(16)} &\cong \mathbb{Z} & \Omega_{10}^{\mathrm{String-Sp}(16)} \cong (\mathbb{Z}_2)^{\oplus 3} \\ \Omega_5^{\mathrm{String-Sp}(16)} &\cong \mathbb{Z}_2 & \Omega_{11}^{\mathrm{String-Sp}(16)} \cong 0. \end{split}$$

Proof. Let $V \to BSp(16)$ be the vector bundle associated to the defining representation; it is rank 64 as a real vector bundle. Then by an argument analogous to [51, section 10.4], there is an isomorphism $\Omega^{\text{String-Sp}(16)}_* \cong \Omega^{\text{String}}_*((BSp(16))^{V-64})$, where $(BSp(16))^{V-64}$ is the *Thom spectrum* of the virtual vector bundle $V - \mathbb{R}^{64} \to BSp(16)$. The Thom spectrum X^V of $V \to X$ is a homotopy-theoretic object whose homotopy groups can be expressed as certain kinds of bordism groups by the Pontrjagin-Thom theorem; the upshot is that string bordism groups of X^V are isomorphic to (X, V)-twisted string bordism groups of a point. See [51, section 10.4] for more information and references.

If tmf denotes the spectrum of connective topological modular forms, then it follows that the map $\Omega^{\text{String}}_*(X) \to tmf_*(X)$ is an isomorphism in degrees 15 and below whenever X is a space or connective spectrum [112, Theorem 2.1] (the latter condition includes all Thom spectra we study in this paper). Therefore for the rest of the proof we focus on $tmf_*((BSp(16))^{V-64})$. These are finitely generated abelian groups, so we may work one prime at a time (see [51, section 10.2]).

As input, we will need the following calculation of Borel.

Proposition 3.49 (Borel [116, section 29]). $H^*(BSp(16); \mathbb{Z}) \cong \mathbb{Z}[c_2, c_4, \ldots, c_{32}]$, where c_i is the pullback of the *i*th Chern class under the map $BSp(16) \to BU(32)$.

For large primes p (i.e. $p \ge 5$), we want to show that $tmf_*((BSp(16))^{V-64})$ lacks p-torsion in degrees 11 and below. This follows because when $p \ge 5$, the homotopy groups of the p-localization $tmf_{(p)}$ are free and concentrated in even degrees [117, section 13.1], and the $\mathbb{Z}_{(p)}$ cohomology of BSp(16) (hence also of $(BSp(16))^{V-64}$, by the Thom isomorphism) is always free and concentrated in even degrees as a consequence of proposition 3.49 and the universal coefficient theorem, so the Atiyah-Hirzebruch spectral sequence computing plocalized $tmf_*((BSp(16))^{V-64})$ collapses with only free summands on the E_{∞} -page, preventing p-torsion in $tmf_*((BSp(16))^{V-64})$ in the range we care about. For p = 3, the 3-localized Atiyah-Hirzebruch spectral sequence does not immediately collapse, so we use the Adams spectral sequence (and we will see that this Adams spectral sequence does immediately collapse). The Adams spectral sequence takes the form

$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{A}}^{s,t}(H^*(X;\mathbb{Z}_p),\mathbb{Z}_p) \Longrightarrow \pi_{t-s}^s(X)_p^{\wedge}.$$
(3.50)

Let us explain this notation. We pick a prime p; then \mathcal{A} is the p-primary Steenrod algebra, the \mathbb{Z}_p -algebra of all natural transformations $H^*(-;\mathbb{Z}_p) \to H^{*+t}(-;\mathbb{Z}_p)$ that commute with the suspension isomorphism. The mod p cohomology of any space or spectrum X is thus naturally a \mathbb{Z} -graded \mathcal{A} -module, so we may apply $\operatorname{Ext}_{\mathcal{A}}$, the derived functor of $\operatorname{Hom}_{\mathcal{A}}$. This gives us two gradings: the original \mathbb{Z} -grading on cohomology is the one labeled t, and the grading arising from the derived functors is the one labeled s. On the right-hand side of (3.50), π^s_* denotes stable homotopy groups, and $(-)_p^{\wedge}$ denotes p-completion. We will not need to worry in too much detail about p-completion: we will only ever p-complete finitely generated abelian groups \mathcal{A} , for which the p-completion carries the same information as the free summands and the p-power torsion summands of \mathcal{A} . Thus we will typically be implicit about p-completion in particular, \mathbb{Z}_p always denotes the cyclic group of order p, not the p-adic integers.

We are interested in *tmf*-homology (or really string bordism), rather than stable homotopy, which means replacing X with $tmf \wedge X$ in (3.50); then the Adams spectral sequence converges to $tmf_{t-s}(X)_p^{\wedge}$.

By work of Henriques and Hill (see [111, 117]), building on work of Behrens [118] and unpublished work of Hopkins-Mahowald, there is a change-of-rings theorem for the 3-primary Adams spectral sequence for tmf simplifying (3.50) to

$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{A}^{tmf}}^{s,t}(H^*(X;\mathbb{Z}_3),\mathbb{Z}_3) \Longrightarrow tmf_*(X)_3^{\wedge}.$$
(3.51)

Here \mathcal{A}^{tmf} is the graded \mathbb{Z}_3 -algebra

$$\mathcal{A}^{tmf} = \mathbb{Z}_3 \langle \beta, \mathcal{P}^1 \rangle / (\beta^2, (\mathcal{P}^1)^3, \beta(\mathcal{P}^1)^2 \beta - (\beta \mathcal{P}^1)^2 - (\mathcal{P}^1 \beta)^2), \qquad (3.52)$$

with $|\beta| = 1$ and $|\mathcal{P}^1| = 4$. For the Adams E_2 -page, \mathcal{A}^{tmf} acts on $H^*(X; \mathbb{Z}_3)$ by sending β to the Bockstein for $0 \to \mathbb{Z}_3 \to \mathbb{Z}_9 \to \mathbb{Z}_3 \to 0$ and \mathcal{P}^1 to the first mod 3 Steenrod power. See [111–114] for more information and some example computations with this variant of the Adams spectral sequence.

As input, we need to know how β and \mathcal{P}^1 act on $H^*((BSp(16))^{V-32};\mathbb{Z}_3)$. This is determined in [114, Corollary 2.37] from the input data of the action of the images of β and \mathcal{P}^1 on the mod 3 Steenrod algebra on $H^*(BSp(16);\mathbb{Z}_3)$. As the cohomology of BSp(16) is concentrated in even degrees, β must act trivially, and thus likewise for the Thom spectrum $(BSp(16))^{V-64}$. Shay [119] computes the action of \mathcal{P}^1 on mod 3 Chern classes;¹⁴ the formula implies that in $H^*(BSp(16);\mathbb{Z}_3)$, $\mathcal{P}^1(c_2) = c_4 + c_2^2$ and $\mathcal{P}^1(c_4) = c_4c_2$. For the Thom class, $\mathcal{P}^1(U) = Uc_2$ [114, Theorem 2.28]. Using the Cartan formula, we can compute the \mathcal{A}^{tmf} -module structure on $H^*((BSp(16))^{V-64};\mathbb{Z}_3)$.

Definition 3.53. If M is a \mathbb{Z} -graded module over a \mathbb{Z} -graded algebra A, we will let $\Sigma^k M$ denote the same underlying A-module with the grading shifted up by k, i.e. if $x \in M$ is homogeneous of degree m, then $x \in \Sigma^k M$ has degree m + k. We will write Σ for Σ^1 .

¹⁴We also found Sugawara's explicit calculations of this formula in [120, section 5] helpful.

The notation Σ^k is inspired by the suspension of a topological space, which has the effect of increasing the degrees of elements in cohomology by 1.

Definition 3.54. Let N_3 denote the nontrivial \mathcal{A}^{tmf} -module extension of $C\nu$ by $\Sigma^8\mathbb{Z}_3$, where $C\nu$ is the \mathcal{A}^{tmf} -module defined in [114, section 3.2].

Then, there is an \mathcal{A}^{tmf} -module isomorphism

$$H^*((BSp(16))^{V-64}; \mathbb{Z}_3) \cong \mathbb{N}_3 \oplus \Sigma^8 \mathbb{N}_3 \oplus \mathbb{P},$$
 (3.55)

where P is concentrated in degrees 12 and above (so we can ignore it). We draw the decomposition (3.55) in figure 3, left.

We need to compute $\operatorname{Ext}_{\mathcal{A}^{tmf}}(N_3, \mathbb{Z}_3)$. To do so, we use the fact that the short exact sequence of \mathcal{A}^{tmf} -modules (which we draw in figure 2, top)

$$0 \longrightarrow \Sigma^8 \mathbb{Z}_3 \longrightarrow N_3 \longrightarrow C\nu \longrightarrow 0$$
(3.56)

induces a long exact sequence on Ext groups; traditionally one draws the Ext of the first and third terms of a short exact sequence in the same Adams chart, so that the boundary maps have the same degree as a d_1 differential. See Beaudry-Campbell [110, section 4.6, section 5] for more information and some examples for modules over a different algebra $\mathcal{A}(1)$, and [114, figures 2, 3, and 5] for some \mathcal{A}^{tmf} -module examples.

We will draw the long exact sequence in Ext corresponding to (3.56) in figure 2. To do so, we need $\operatorname{Ext}_{\mathcal{A}^{tmf}}(\mathbb{Z}_3)$, which is due to Henriques-Hill [111, 117], and $\operatorname{Ext}_{\mathcal{A}^{tmf}}(C\nu)$, which is computed in topological degree 14 and below in [114, figure 2]. Our notation for names of Ext classes follows [114, section 3]; $\operatorname{Ext}_{\mathcal{A}^{tmf}}(\Sigma^8\mathbb{Z}_3)$ is a free $\operatorname{Ext}_{\mathcal{A}^{tmf}}(\mathbb{Z}_3)$ -module on a single generator, so call that generator z.¹⁵ Most boundary maps are nonzero for "degree reasons," meaning that their domain or codomain is the zero group. For $t - s \leq 14$, there are two exceptions: $\partial(z)$ could be $\pm \alpha y$ or 0, and $\partial(\alpha z)$ could be $\pm \beta x$ or 0. Since the boundary maps commute with the $\operatorname{Ext}_{\mathcal{A}^{tmf}}(\mathbb{Z}_3)$ -action and $\alpha(\alpha y) = \beta x$,¹⁶ these two boundary maps are either both zero or both nonzero. To see that they are both nonzero, we use that $\operatorname{Ext}_{\mathcal{A}^{tmf}}^{0,8}(N_3) \cong \operatorname{Hom}_{\mathcal{A}^{tmf}}(N_3, \Sigma^8\mathbb{Z}_3) = 0$, so $z \in \operatorname{Ext}_{\mathcal{A}^{tmf}}^{0,8}(\Sigma^8\mathbb{Z}_3)$ is not the image of an Ext class for N_3 , so $\partial(z) \neq 0$.

With this Ext in hand, we can draw the E_2 -page of the Adams spectral sequence in figure 3, right. The spectral sequence collapses at E_2 in the range we study for degree reasons. The straight lines denote actions by $h_0 \in \operatorname{Ext}_{\mathcal{A}^{tmf}}^{1,1}(\mathbb{Z}_3,\mathbb{Z}_3)$, which lift to multiplication by 3, so we see there is no 3-torsion in degrees 11 and below.

Lastly, for p = 2, we use the Adams spectral sequence again; the outline of the proof is quite similar to the p = 3 case, but the details are different. Specifically, we will once again

¹⁵There are two classes which generate $\operatorname{Ext}_{\mathcal{A}^{tmf}}(\Sigma^{8}\mathbb{Z}_{3})$ as an $\operatorname{Ext}_{\mathcal{A}^{tmf}}(\mathbb{Z}_{3})$ -module, and one is -1 times the other. For the purposes of this paper, it does not matter which one we call z and which one we call -z.

¹⁶The equation $\alpha(\alpha y) = \beta x$ is stated in [114, Remark 3.21], but not proven there. One way to prove it is to compare with the equivalent α -action $\alpha y \mapsto \beta x$ in $\operatorname{Ext}_{\mathcal{A}^{tmf}}(N_1)$ in the long exact sequence in (*ibid.*, figure 5): because $\partial(\alpha y) = \pm \beta w$ and $\alpha \beta w \neq 0$, and because $\alpha(\partial(-)) = \partial(\alpha \cdot -)$, $\alpha(\alpha y) \neq 0$, hence must be $\pm \beta x$, and we can choose the generator x so that we obtain βx and not $-\beta x$. The calculation of $\partial(\alpha x)$ in (*ibid.*, Lemma 3.24) does not use any information about $\alpha(\alpha y)$.



Figure 2. Top: the short exact sequence (3.56) of \mathcal{A}^{tmf} -modules. Lower left: the induced long exact sequence in Ext. Lower right: Ext_{\mathcal{A}^{tmf}}(N₃) as computed by the long exact sequence.



Figure 3. Left: the \mathcal{A}^{tmf} -module structure on $H^*((BSp(16))^{V-64}; \mathbb{Z}_3)$ in low degrees; the pictured submodule contains all elements in degrees 11 and below. Right: the E_2 -page of the Adams spectral sequence computing $tmf_*((BSp(16))^{V-64})^{\wedge}_3$.

use the Adams spectral sequence and a standard change-of-rings theorem to simplify the calculation of the E_2 -page, but the algebra of cohomology operations is different.

Let $\mathcal{A}(2)$ be the subalgebra of the mod 2 Steenrod algebra generated by Sq¹, Sq², and Sq⁴. There is an isomorphism $H^*(tmf; \mathbb{Z}_2) \cong \mathcal{A} \otimes_{\mathcal{A}(2)} \mathbb{Z}_2$ [121, 122], which by a standard argument simplifies the E_2 -page of the 2-primary Adams spectral sequence to

$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{A}(2)}^{s,t}(H^*(X;\mathbb{Z}_2),\mathbb{Z}_2) \Longrightarrow tmf_*(X)_2^{\wedge}.$$
(3.57)

The next thing to do is to determine how $\mathcal{A}(2)$ acts on $H^*((BSp(16))^{V-64}; \mathbb{Z}_2)$. Since this cohomology ring vanishes in degrees not divisible by 4, Sq¹ and Sq² act trivially. For Sq⁴, [110, section 3.3] says Sq⁴(U) = $Uw_4(V) = c_2$, and the Wu formula computes the Steenrod squares in $H^*(BSp(16); \mathbb{Z}_2)$, using that the mod 2 reductions of the generators in proposition 3.49 are Stiefel-Whitney classes. This allows us to completely describe the $\mathcal{A}(2)$ action on $H^*((BSp(16))^{V-64}; \mathbb{Z}_2)$ in the degrees we need: Sq⁴(U) = Uc_2 , Sq⁴(Uc_2^2) = Uc_2^3 ,



Figure 4. Left: the $\mathcal{A}(2)$ -module structure on $H^*((BSp(16))^{V-64}; \mathbb{Z}_2)$ in low degrees; the pictured submodule contains all elements in degrees 11 and below. Right: the E_2 -page of the Adams spectral sequence computing $tmf_*((BSp(16))^{V-64})_2^{\wedge}$.

 $Sq^4(Uc_4) = Uc_6$, and all other actions by Sq^1 , Sq^2 , or Sq^4 starting in degree 11 or below vanish. Thus, if M_4 denotes the $\mathcal{A}(2)$ -module consisting of two \mathbb{Z}_2 summands in degrees 0 and 4 connected by a Sq^4 , there is an isomorphism

$$H^*((BSp(16))^{V-64}; \mathbb{Z}_2) \cong M_4 \oplus \Sigma^8 M_4 \oplus \Sigma^8 M_4 \oplus P,$$
 (3.58)

where P contains no elements in degrees 11 or below, and hence will be irrelevant to our calculations. We draw (3.58) in figure 4, left.

Bruner-Rognes [113, section 4.4] compute $\operatorname{Ext}_{\mathcal{A}(2)}(M_4)$; using their result, we give the E_2 -page of the Adams spectral sequence computing $tmf_*((BSp(16))^{V-64})^{\wedge}_2$ in figure 4, right.

Looking at the E_2 -page, most differentials are ruled out by degree considerations or the fact that they must commute with the action of h_0 or h_1 . The only options left are d_2 and d_3 out of $E_r^{0,8}$ and $d_2 \colon E_2^{2,12} \to E_2^{4,13}$.

Lemma 3.59. All classes in $E_2^{0,8} \cong (\mathbb{Z}_2)^{\oplus 2}$ survive to the E_{∞} -page.

Proof. Classes $x \in E_2^{0,\bullet}$ of an Adams spectral sequence for *G*-bordism correspond naturally to (a subset of) \mathbb{Z}_2 -valued characteristic classes c_x for manifolds with *G*-structure, and x survives to the E_{∞} -page if and only if there is a closed manifold *M* with *G*-structure such that $\int_M c_x = 1$; see [41, section 8.4].

For the Adams spectral sequence for string-Sp(16) bordism at p = 2, the two classes corresponding to a basis of $E_2^{0,8}$ are the mod 2 reductions of c_2^2 and c_4 . To finish this lemma, we will find closed string-Sp(16) 8-manifolds on which these classes do not vanish.

• The quaternionic projective plane \mathbb{HP}^2 has a tautological principal Sp(1)-bundle $P := S^{11} \to \mathbb{HP}^2$; let $P^{\vee} \to \mathbb{HP}^2$ be the same space with the quaternion-conjugate Sp(1)action, and let $Q \to \mathbb{HP}^2$ be the principal Sp(16)-bundle induced from P^{\vee} by the inclusion $i: \operatorname{Sp}(1) \to \operatorname{Sp}(16)$. Using the fact that i pulls c_2 back to c_2 and Borel and Hirzebruch's calculation of the characteristic classes of \mathbb{HP}^2 [123, section 15.5, section 15.6] (see also [41, section 5.2] for a good review), the reader can verify that (\mathbb{HP}^2, Q) has a unique string-Sp(16) structure, meaning in particular that $c_2(Q) = -\frac{1}{2}p_1(\mathbb{HP}^2)$, and that $\int_{\mathbb{HP}^2} c_2(Q)^2 = 1$.

• For c_4 , take S^8 with principal Sp(16)-bundle $P \to S^8$ classified by either generator of

$$[S^8, BSp(16)] = \pi_8(BSp(16)) \xrightarrow{\cong} \pi_8(BSp) = \pi_0(BSp) = \mathbb{Z},$$
(3.60)

using Bott periodicity. Since $H^4(S^8; \mathbb{Z}) = 0$, $c_2(P)$ and $\frac{1}{2}p_1(S^8)$ vanish and therefore (S^8, P) is string-Sp(16); and $\int_{S^8} c_4(P) = 1$ essentially by definition. \Box

The differential out of $E_2^{2,12}$ does not vanish — to see this, consider the map

$$f: \mathbb{HP}^1 \longrightarrow \mathbb{HP}^\infty \simeq BSp(1) \longrightarrow BSp(16)$$
 (3.61a)

and the induced map on Thom spectra

$$f_* \colon tmf_*((\mathbb{HP}^1)^{f^*V-64}) \longrightarrow tmf_*((BSp(16))^{V-64}).$$
(3.61b)

The map f_* induces a pullback map on mod 2 cohomology and on Adams spectral sequences; the map on mod 2 cohomology is the quotient by all elements of degree greater than 4, so the effect on Adams spectral sequences is to kill all summands in Ext except for the red summands. As $H^*((\mathbb{HP}^1)^{f^*V-64};\mathbb{Z}_2)$ consists of two \mathbb{Z}_2 summands in degrees 0 and 4, joined by a Sq⁴, the Adams spectral calculating its 2-completed *tmf*-homology is worked out by Bruner-Rognes [113, Theorem 8.1], who show that $d_2: E_2^{2,12} \to E_2^{4,13}$ is an isomorphism. Thus this differential persists to the Sp(16) Adams spectral sequence.

3.3.2 U(32)

Now we discuss the Sagnotti string, whose gauge group is U(32). The Green-Schwarz mechanism for this theory involves three classes in degrees 2, 4, and 6 canceling c_1 , c_2 , and c_3 of the gauge bundle, respectively.

We may impose the degree-2, 4, and 6 conditions on BU(32) in any order. Starting with c_1 , we obtain BSU(32); then, let $BSU(32)\langle c_3 \rangle$ denote the fiber of the map

$$c_3 \colon BSU(32) \longrightarrow K(\mathbb{Z}, 6). \tag{3.62}$$

A map $X \to BSU(32)\langle c_3 \rangle$ is equivalent data to a rank-32 complex vector bundle $V \to X$ with SU-structure and a trivialization of $c_3(V)$. There is a tautological such vector bundle $V_t \to BSU(32)\langle c_3 \rangle$, which is the pullback of the tautological bundle over BSU(32).

Finally, the degree-4 condition for a U(32)-bundle V over a manifold M asks for a trivialization of $\frac{1}{2}p_1(M) + c_2(V)$. Thus, we ask for a $(BSU(32)\langle c_3\rangle, V_t)$ -twisted string structure on M, i.e. a map $f: M \to BSU(32)\langle c_3\rangle$ and a string structure on $TM \oplus f^*V_t$; the Whitney sum formula for $\frac{1}{2}p_1$ unwinds this into the usual Green-Schwarz condition. We will be a little casual with the notation and call a $(BSU(32)\langle c_3\rangle, V_t)$ -structure a String-SU(32) $\langle c_3\rangle$ -structure, even though we do not construct a Lie 2-group String-SU(32) $\langle c_3\rangle$ realizing this twisted string structure (and indeed, there is no guarantee one exists).

Theorem 3.63. $\Omega_{11}^{\text{String-SU}(32)\langle c_3 \rangle}$ is isomorphic to either 0 or \mathbb{Z}_2 .

The ambiguity is in a differential we were not able to resolve. Unfortunately, this means we were not able to use bordism-theoretic methods alone to calculate the anomaly of the Sagnotti string. Our proof also yields partial information on lower-dimensional bordism groups; there is ambiguity due to Adams spectral sequence differentials, some of which we suspect are nonzero.

Proof. Before we start our analysis, we need to understand $H^*(BSU(3)\langle c_3\rangle; A)$ for various coefficient rings A. As $BSU(32)\langle c_3\rangle$ is the fiber of $c_3: BSU(32) \to K(\mathbb{Z}, 6)$, the fiber of $BSU(32)\langle c_3\rangle \to BSU(32)$ is $\Omega K(\mathbb{Z}, 6) \simeq K(\mathbb{Z}, 5)$; moreover, this fibration pulls back from the universal fibration with fiber $K(\mathbb{Z}, 5)$, namely the loop-space-path-space fibration for $K(\mathbb{Z}, 6)$:



We will compute $H^*(BSU(32)\langle c_3\rangle; A)$ for various A using the Serre spectral sequence, along with some information gained from the map of Serre spectral sequences induced by (3.64).

As for the Sugimoto string, we work one prime at a time.

Lemma 3.65. For $p \ge 5$, there is no p-torsion in $H^*(BSU(32)\langle c_3 \rangle; \mathbb{Z})$ in degrees 12 and below, and all free summands are concentrated in even degrees.

Proof. It suffices to work with cohomology valued in the ring $\mathbb{Z}[1/6]$ of rational numbers whose denominators in lowest terms are of the form $2^m 3^n$, as tensoring with $\mathbb{Z}[1/6]$ preserves all *p*-power torsion for $p \geq 5$.

Cartan [124] and Serre [125] computed $H^*(K(\mathbb{Z}, n); \mathbb{Z}[1/6])$; their formulas imply that when $p \geq 5$, $H^k(K(\mathbb{Z}, 5); \mathbb{Z}[1/6])$ is torsion-free for $k \leq 12$, and vanishes apart from $H^5(K(\mathbb{Z}, 5); \mathbb{Z}[1/6]) \cong \mathbb{Z}[1/6].$

Now consider the Serre spectral sequence for the fibration on the left in (3.64) using cohomology with $\mathbb{Z}[1/6]$ coefficients. The map of fibrations (3.64) induces a map of Serre spectral sequences, and this map is an isomorphism on $E_2^{0,\bullet}$. Since this map commutes with differentials, this means the fate of all classes in $E_2^{0,\bullet}$ is determined by their preimages in the spectral sequence for $K(\mathbb{Z}, 5) \to * \to K(\mathbb{Z}, 6)$. For example, we know thanks to Serre [126, section 10] that in that spectral sequence, E transgresses to the mod 2 reduction of the tautological class F of $K(\mathbb{Z}, 6)$. Therefore in the spectral sequence for $BSU(32)\langle c_2\rangle$, Etransgresses to the pullback of F, which is c_3 . The Leibniz rule then tells us $d_6(xE) = xc_3$ for $x \in H^*(BSU(32); \mathbb{Z}[1/6])$; since this cohomology ring is polynomial, $xc_3 \neq 0$ as long as $x \neq 0$, so these differentials never vanish. Therefore the nonzero part of the E_{∞} -page, at least in total degree 12 and below, is a quotient of $E_2^{*,0} = H^*(BSU(32); \mathbb{Z}[1/6])$. Since $H^*(BSU(32); \mathbb{Z}[1/6])$ is free and concentrated in even degrees. This implies the E_{∞} -page is also free and concentrated in even degrees in total degree 12 and below, which implies the lemma statement. **Corollary 3.66.** For $p \ge 5$, $\Omega_k^{\text{String-SU}(32)\langle c_3 \rangle}$ lacks p-torsion for $k \le 11$.

Proof. We want to compute $\Omega^{\text{String}}_{*}((BSU(32)\langle c_3\rangle)^{V_t-64})$, and as noted above, we may replace $\Omega^{\text{String}}_{*}$ with tmf for the degrees k in the corollary statement. Because of lemma 3.65 and the fact that $tmf_{(p)}$ has homotopy groups concentrated in even degrees and lacks p-torsion, the Atiyah-Hirzebruch spectral sequence computing $tmf_{*}((BSU(32)\langle c_3\rangle)^{V_t-64})_p^{\wedge}$ collapses with no p-torsion in the range 11 and below.

As usual, p = 2 and p = 3 are harder.

Lemma 3.67.

- 2. $H^*(BSU(32)\langle c_3 \rangle; \mathbb{Z}_2) \cong \mathbb{Z}_2[c_2, G, c_4, H, J, K, c_6, L, \dots]/(\dots)$, with $|c_i| = 2i$, |G| = 7, |H| = 8, |J| = 10, |K| = 11, and |L| = 12; all missing generators and relations are in degrees 13 and above. In addition, we have the following Steenrod squares:
 - Sq¹ vanishes on the named generators except Sq¹(G) = $H + \lambda_1 c_4$ and Sq¹(K) = $L + \lambda_2 c_2 c_4 + \lambda_3 c_6$ for some $\lambda_1, \lambda_2, \lambda_3 \in \mathbb{Z}_2$.
 - Sq² vanishes on the named generators except for Sq²(H) = J and possibly on c₆, K, and L.
 - $\operatorname{Sq}^4(c_2) = c_2^2$, $\operatorname{Sq}^4(c_4) = c_2c_4 + c_6$, $\operatorname{Sq}^4(G) = K$, and $\operatorname{Sq}^4(H) = L$.
- 3. $H^*(BSU(32)\langle c_3 \rangle; \mathbb{Z}_3) \cong \mathbb{Z}_3[c_2, c_4, J, c_6, \dots]/(\dots)$ with $|c_i| = 2i$ and |J| = 10, and with all missing generators and relations in degrees 13 and above; c_i denotes the pullback of the mod 3 reduction of the *i*th Chern class along $BSU(32)\langle c_3 \rangle \to BSU(32)$, and $\mathcal{P}^1(c_2) = c_2^2 + c_4$ and $\mathcal{P}^1(c_4) = c_2c_4$.

Proof. This is a standard argument with the Serre spectral sequence for the fibration on the left in (3.64), so we sketch the details.

For the mod 2 cohomology, we need as input $H^*(BSU(32); \mathbb{Z}_2) \cong \mathbb{Z}_2[c_2, c_3, \ldots, c_{32}]$ with $|c_i| = 2i$ [116, section 29]: these are the mod 2 reductions of the Chern classes. We also need $H^*(K(\mathbb{Z}, 5); \mathbb{Z}_2)$, which was computed by Serre [126, section 10]. This is a polynomial ring on infinitely many generators; the six in degrees below 13 are $E \in H^5$, the mod 2 reduction of the tautological class; $G := \operatorname{Sq}^2(E)$; $H := \operatorname{Sq}^1(H)$; $I := \operatorname{Sq}^4(E)$; $K := \operatorname{Sq}^4(G)$; and $L := \operatorname{Sq}^5(G)$.

In the Serre spectral sequence, the class E transgresses to c_3 , and the proof is the same as in the proof of lemma 3.65. Similarly, we divine the fate of the other classes on the line p = 0:

- In the Serre spectral sequence for the rightmost fibration in (3.64), G transgresses via d_8 to $\operatorname{Sq}^2(F)$ by the Kudo transgression theorem [127], so in the leftmost fibration, $d_r(G) = 0$ for $r \leq 7$, and $d_8(G) = \operatorname{Sq}^2(c_3) = 0$. Thus G is a permanent cycle.
- In a similar way, H transgressing to $\operatorname{Sq}^3(F)$ via d_9 pulls back to imply $d_9(H) = \operatorname{Sq}^3(c_3) = 0$, so H is also a permanent cycle. Likewise, E^2 is a permanent cycle, because in the fibration over $K(\mathbb{Z}, 6)$, it supports the transgressing $d_{11}(E^2) = \operatorname{Sq}^5(F)$, and $\operatorname{Sq}^5(c_3) = 0$, and similarly K and L are permanent cycles.
- $I = \operatorname{Sq}^4(E)$ transgresses to $\operatorname{Sq}^4(c_3) = c_5 + c_2c_3$.

The Leibniz rule then cleans up the rest of the spectral sequence in degrees 12 and below. This gives us the ring structure. For the Steenrod squares, we use the information of the \mathcal{A} -action on $H^*(BSU(32);\mathbb{Z}_2)$ coming from the Wu formula, together with the \mathcal{A} -actions we gave when describing $H^*(K(\mathbb{Z},5);\mathbb{Z}_2)$ above. There is ambiguity in the Steenrod squares in $H^*(BSU(32)\langle c_3\rangle;\mathbb{Z}_2)$ coming from the loss of information passing to the associated graded on the E_{∞} -page, which is the source of λ_1 , λ_2 , and λ_3 in the theorem statement. However, by pulling back to the analogous fibration over BSU(2), where the fiber bundle admits a section (as c_3 of an SU(2)-bundle is canonically trivial), so we can use the Künneth formula to compute Steenrod squares. Pulling back to SU(2) loses all information about c_i for i > 2, so this leaves ambiguity in c_4 and c_6 as described in the theorem statement, but resolves the ambiguity involving c_2 . Some ambiguity can be erased by redefining generators, which is how we disambiguate Sq⁴(H) = L, but this still leaves the choices listed in the theorem statement.

For \mathbb{Z}_3 cohomology, we begin with $H^*(K(\mathbb{Z},5);\mathbb{Z}_3) \cong \mathbb{Z}_3[E, \mathcal{P}^1(E), \beta \mathcal{P}^1(E), \mathcal{P}^2(E), \ldots]$, with the remaining generators in degrees 15 and above, where $E \in H^5(K(\mathbb{Z},5);\mathbb{Z}_3)$ is the mod 3 reduction of the tautological class. Just like for \mathbb{Z}_2 cohomology, E transgresses via d_5 to c_3 ; then the Kudo transgression theorem [127] tells us

- $\mathcal{P}^{1}(E)$ transgresses via d_{10} to $\mathcal{P}^{1}(c_{3}) = c_{2}c_{3} c_{5} = -c_{5}$ by the E_{10} -page (as $d_{5}(c_{2}E) = c_{2}c_{3}$), and
- $\beta \mathcal{P}^1(E)$ is a permanent cycle (it transgresses to $\beta(-c_5) = 0$, which we know for degree reasons).

We obtain $\mathcal{P}^1(c_i)$ from Sugawara's calculations [120, section 5] of Shay's formula [119]. With the fate of these classes known, the Leibniz rule cleans up the rest of the spectral sequence in total degrees 12 and below to obtain the theorem statement.

Now, just as in the proof of theorem 3.48, we run the Adams spectral sequences at p = 3and p = 2. The twist by V_t twists the action of \mathcal{P}^1 at p = 3, and the action of Sq^4 at p = 2, in an analogous way. Reusing names of \mathcal{A}^{tmf} -modules from section 3.3.1, we conclude that there is an \mathcal{A}^{tmf} -module isomorphism

$$H^*((BSU(32)\langle c_3\rangle)^{V_t-64};\mathbb{Z}_3) \cong \mathbb{N}_3 \oplus \Sigma^8 N_3 \oplus \Sigma^{10} N_3 \oplus P, \tag{3.68}$$

where P is concentrated in degrees 12 and above, so will not affect us. We draw (3.68) in figure 5, left. We calculated $\operatorname{Ext}_{\mathcal{A}^{tmf}}(N_3)$ in figure 2; using this, we discover that, like for the Sugimoto string, in degrees 11 and below, the E_2 -page consists only of h_0 -towers in degrees 0, 4, and 8, so there can be no 3-torsion. See figure 5, right, for a picture of this Adams spectral sequence.

Last, p = 2. The ambiguity in the Steenrod actions is not severe enough to get in the way of the existence of an isomorphism of $\mathcal{A}(2)$ -modules

$$H^*((BSU(32)\langle c_3\rangle)^{V_t-64};\mathbb{Z}_2) \cong M_4 \oplus \Sigma^7 N_1 \oplus \Sigma^8 M_4 \oplus \Sigma^8 M_4 \oplus \Sigma^{11} N_2 \oplus P \tag{3.69}$$

where P is concentrated in degrees 12 and above, so will be irrelevant for us, and:



Figure 5. Left: the \mathcal{A}^{tmf} -module structure on $H^*((BSU(32)\langle c_3\rangle)^{V_t-64};\mathbb{Z}_3)$ in low degrees; the pictured submodule contains all elements in degrees 11 and below. Right: the E_2 -page of the Adams spectral sequence computing $tmf_*((BSU(32)\langle c_3\rangle)^{V_t-64})^{\wedge}_3$. This figure is part of the proof of theorem 3.63.

- N_1 is isomorphic to $\mathcal{A}(2) \otimes_{\mathcal{A}(1)} Q$ in degrees 6 and below (i.e. the quotients of these two $\mathcal{A}(2)$ -modules by their submodules of elements in degrees 7 and above are isomorphic); and
- N_2 is isomorphic to $\mathcal{A}(2) \otimes_{\mathcal{A}(1)} Q$ in degrees 3 and below.

Here Q is the "question mark," the $\mathcal{A}(1)$ -module which has a \mathbb{Z}_2 -vector space basis $\{x_0, x_1, x_3\}$ with $|x_i| = i$, and with $\operatorname{Sq}^1(x_0) = x_1$, $\operatorname{Sq}^2(x_1) = x_3$ (all other $\mathcal{A}(1)$ -actions are trivial for degree reasons). The module $\Sigma^{11}N_2$ is generated by Uc_2G . We draw the decomposition (3.69) in figure 6, left.

Bruner-Rognes [113, section 4.44] calculate $\operatorname{Ext}_{\mathcal{A}(2)}(M_4)$. For N_1 and N_2 , we use that an isomorphism of $\mathcal{A}(2)$ -modules in degrees k and below implies the existence of an isomorphism of Ext groups in topological degrees k - 1 and below, so it is good enough to know $\operatorname{Ext}_{\mathcal{A}(2)}(\mathcal{A}(2) \otimes_{\mathcal{A}(1)} Q)$; then the change-of-rings theorem (see, e.g., [110, section 4.5]) implies

$$\operatorname{Ext}_{\mathcal{A}(2)}(\mathcal{A}(2) \otimes_{\mathcal{A}(1)} Q) \cong \operatorname{Ext}_{\mathcal{A}(1)}(Q), \qquad (3.70)$$

and Adams-Priddy [128, table 3.11] compute $\operatorname{Ext}_{\mathcal{A}(1)}(Q)$. Putting all this together, we can draw the E_2 -page in figure 6, right. In the range relevant to us, the E_2 -page is generated as an $\operatorname{Ext}_{\mathcal{A}(2)}(\mathbb{Z}_2)$ -module by the following ten summands.

- 1. Coming from $\text{Ext}(M_4)$: $a_1 \in \text{Ext}^{0,0}, a_2 \in \text{Ext}^{3,7}, a_3 \in \text{Ext}^{1,6}, a_4 \in \text{Ext}^{2,9}$, and $a_5 \in \text{Ext}^{2,12}$.
- 2. Coming from $\operatorname{Ext}(\Sigma^7 N_1)$: $b_1 \in \operatorname{Ext}^{7,0}$ and $b_2 \in \operatorname{Ext}^{2,12}$.
- 3. Coming from $\operatorname{Ext}(\Sigma^8 M_4)$: $c \in \operatorname{Ext}^{0,8}$.
- 4. Coming from $\operatorname{Ext}(\Sigma^8 M_4)$: $d \in \operatorname{Ext}^{0,8}$.
- 5. Coming from $\operatorname{Ext}(\Sigma^{11}N_2)$: $e \in \operatorname{Ext}^{0,11}$.

These are subject to various relations: notably, if x is any one of these generators, $h_2 x = 0$.



Figure 6. Left: the $\mathcal{A}(2)$ -module structure on the quotient of $H^*((BSU(32)\langle c_3\rangle)^{V_t-64}; \mathbb{Z}_2)$ by its submodule of elements in degrees greater than 12; the pictured submodule contains all elements in degrees 11 and below. Right: the E_2 -page of the Adams spectral sequence computing $tmf_*((BSU(32)\langle c_3\rangle)^{V_t-64})_2^{\wedge}$. This figure is part of the proof of theorem 3.63. We were unable to determine the value of $d_2(e)$: it is either 0 or h_1^2d .

Once we take into account the fact that differentials commute with h_0 and h_1 , we still need to determine $d_2(b_1)$, $d_r(c)$, $d_r(d)$, $d_2(a_5)$, $d_2(b_2)$, and $d_r(e)$.

Lemma 3.71. For all r, $d_r(c) = 0$ and $d_r(d) = 0$; $d_2(a_5) = h_1^2 a_4$.

The values of these differentials are the same as for the corresponding classes in the Adams spectral sequence for the Sugimoto string, and the proofs are the same as we gave for them in section 3.3.1.

Ultimately we need to address $d_2: E_2^{0,11} \to E_2^{2,13}$. If this differential vanishes, there is also potential for $d_6: E_6^{0,11} \to E_6^{6,16}$ to be nonzero. The fate of these two differentials determines whether $\Omega_{11}^{\text{String-SU}(32)\langle c_3 \rangle}$ is nonzero, so it is unfortunate that the techniques we applied were unable to resolve them.

We are able to obtain some partial information, though.

Lemma 3.72. If $d_2(e) = 0$, so that $d_6(e)$ is defined, then $d_6(e) = 0$.

Proof. $d_6(e) \in E_6^{6,16}$. No other nonzero differentials have source or target $E_r^{6,16}$, so $E_6^{6,16} \cong E_2^{6,16} \cong (\mathbb{Z}_2)^{\oplus 2}$, spanned by the classes $w_1h_1^2a_1$ and $h_0^4b_2$. Here $w_1 \in \operatorname{Ext}_{\mathcal{A}(2)}^{4,12}(\mathbb{Z}_2)$ is the class whose image in $\operatorname{Ext}_{\mathcal{A}(1)}(\mathbb{Z}_2)$ is the Bott periodicity class. Thus there are $\lambda_1, \lambda_2 \in \mathbb{Z}_2$ such that

$$d_6(e) = \lambda_1 w_1 h_1^2 a_1 + \lambda_2 h_0^4 b_2. \tag{3.73}$$

Because d_6 commutes with h_0 and $h_0 e = 0$, $\lambda_2 = 0$.

To show $\lambda_1 = 0$, consider the map of Adams spectral sequences induced by the map from the *tmf*-homology to the *ko*-homology of $(BSU(32)\langle c_3\rangle)^{V_t-32}$. The map on E_2 -pages is the map

$$\operatorname{Ext}_{\mathcal{A}(2)}(H^*((BSU(32)\langle c_3\rangle)^{V_t-32};\mathbb{Z}_2),\mathbb{Z}_2) \longrightarrow \operatorname{Ext}_{\mathcal{A}(1)}(H^*((BSU(32)\langle c_3\rangle)^{V_t-32};\mathbb{Z}_2),\mathbb{Z}_2)$$

$$(3.74)$$

induced by the inclusion $\mathcal{A}(1) \to \mathcal{A}(2)$ of algebras. It is possible to compute the righthand Ext groups using the decomposition (3.69) and the techniques in [110]; one learns that e and $w_1h_1^2a_1$ both remain nonzero after (3.74), so it suffices to compute $d_6(e)$ in the ko-homology Adams spectral sequence. There, though, the submodule M_{a_1} of the Ext groups generated by a_1 splits off: because $V_t - 32$ is spin, there is a Thom isomorphism $ko_*((BSU(32)\langle c_3\rangle)^{V_t-32}) \cong ko_*(BSU(32)\langle c_3\rangle)$, so $ko_*(\text{pt})$ splits off; as this splitting lifts to the level of spectra, it also splits M_{a_1} off of the Adams spectral sequence, so all differentials into M_{a_1} from any other summand vanish. Thus $d_6(e)$ cannot be $h_1^2w_1a_1$ in the ko-homology Adams spectral sequence, so the same is true in the tmf-homology Adams spectral sequence. \Box

Likewise, since $E_2^{2,12} \cong (\mathbb{Z}_2)^{\oplus 4}$, spanned by the classes a_5 , b_2 , $h_1^2 c$, and $h_1^2 d$, then there are $\lambda_1, \ldots, \lambda_4 \in \mathbb{Z}_2$ such that

$$d_2(e) = \lambda_1 a_5 + \lambda_2 b_2 + \lambda_3 h_1^2 c + \lambda_4 h_1^2 d.$$
(3.75)

Lemma 3.76. In (3.75), $\lambda_1 = 0$, $\lambda_2 = 0$, and $\lambda_3 = 0$.

Proof. Because $h_0b_2 \neq 0$ but $h_0h_1^2 = 0$, if $\lambda_2 \neq 0$, then $h_0d_2(e) \neq 0$. However, since d_2 commutes with h_0 -multiplication, and $h_0e = 0$, λ_2 must vanish and $d_2(e) \in \text{span}(a_5, h_1^2c, h_1^2d)$.

By lemma 3.71, $d_2(c) = 0 = d_2(d) = 0$, so $d_2(h_1^2 c) = d_2(h_1^2 d) = 0$, and $d_2(a_5) = h_1^2 a_4$. Therefore d_2 : span $(a_5, h_1^2 c, h_1^2 d) \to E_2^{4,13}$ is nonzero on a class $\mu_1 a_5 + \mu_2 h_1^2 c + \mu_3 h_1^2 d$ if and only if $\mu_1 \neq 0$. Thus $\lambda_1 = 0$: otherwise $d_2(d_2(e)) \neq 0$, and it is always true that $d_2 \circ d_2 = 0$.

For λ_3 , consider the map $r: BSU(3)\langle c_3 \rangle \to BSU(32)\langle c_3 \rangle$ and the map r_* it induces of Adams spectral sequences. The pullback r^* on cohomology kills c_4 but leaves c_2 and G alone; therefore on Ext groups, $e \in \text{Im}(r_*)$ (because e is the filtration 0 class corresponding to c_2G), $h_1^2 c \in \text{Im}(r_*)$ (because c is the filtration 0 element corresponding to c_2^2), and $h_1^2 d \notin \text{Im}(r_*)$ (because d corresponds to c_4). The map r_* commutes with differentials, so $d_2(e) \in \text{Im}(r_*)$, which is only consistent if $\lambda_3 = 0$.

Determining whether $\lambda_4 = 0$ appears to be difficult. This would be a good problem to address because if $\lambda_4 \neq 0$, so that $d_2(e) \neq 0$, then the bordism group controlling the anomaly of the Sagnotti string would vanish, and the anomaly would cancel, at least on the class of backgrounds we studied.

Because the class e potentially causing a nonzero bordism group is in Adams filtration 0, the corresponding bordism invariant is the integral of a modulo 2 characteristic class, explicitly

$$\int c_2 G. \tag{3.77}$$

The class $G \in H^7(BSU(32)\langle c_3 \rangle; \mathbb{Z}_2)$ is a little mysterious, so we go into some more detail; it is an example of a *secondary characteristic class* in the sense of Peterson-Stein [129].

Recall that by a trivialization of a cohomology class $z \in H^k(X; A)$, where A is an abelian group, we mean a null-homotopy of a map $f_z \colon X \to K(A, k)$ whose homotopy class represents z. There is a space of such trivializations, and a standard result in obstruction theory implies that its set of path components is a torsor over $H^{k-1}(X; A)$. In other words, given two trivializations of f_z , their difference is well-defined as an element of $H^{k-1}(X; A)$.

The Wu formula implies $\operatorname{Sq}^2(c_3) = 0$ in $H^8(BSU(32); \mathbb{Z}_2)$, and in fact provides a canonical trivialization for $\operatorname{Sq}^2(c_3)$. Pulling back to $BSU(32)\langle c_3 \rangle$ trivializes c_3 , and therefore provides a second trivialization of $\operatorname{Sq}^2(c_3)$. The difference between these two trivializations is the class $G \in H^7(BSU(32)\langle c_3 \rangle; \mathbb{Z}_2)$.

As a final comment, if we knew of a manifold with a non-trivial integral of c_2G , it would by definition be a generator of the bordism group. We could evaluate the anomaly theory on it in order to determine whether or not the anomaly vanishes. Regrettably, we do not know of such a manifold.

3.3.3 $\operatorname{Spin}(16) \times \operatorname{Spin}(16)$

Next we discuss the symmetry type of the non-supersymmetric heterotic string with gauge Lie algebra $\mathfrak{so}(16) \oplus \mathfrak{so}(16)$. Although usually called SO(16)², there are fields transforming in spinor representations in the massless spectrum of the theory which means that we should instead consider Spin(16)². There is a further subtlety: according to [53], the gauge group G is the quotient of Spin(16) × Spin(16) by the diagonal \mathbb{Z}_2 subgroup $\langle (k, k) \rangle$, where $k \in \text{Spin}(16)$ is either central element not equal to ± 1 .¹⁷

As the computation of $H^*(BG)$ is complicated, we will make a simplifying assumption: only working with the double cover $\text{Spin}(16) \times \text{Spin}(16)$, as we mentioned above. Thus our anomaly cancellation results are only partial information: if we found an anomaly for $\text{Spin}(16)^2$, it would imply the existence of an anomaly for the actual gauge group G. However, we found that anomalies cancel for $\text{Spin}(16)^2$, which is only partial information: there could be an anomaly of the theory which vanishes when restricted to gauge fields induced from a $\text{Spin}(16)^2$ gauge field. It would be interesting to address the more general question of the anomaly for G.¹⁸

Let String-Spin(16)² be the Lie 2-group which is the string cover of Spin × Spin(16) × Spin(16) corresponding to the degree-4 cohomology class $\frac{1}{2}p_1^{(1)} - \frac{1}{2}p_1^{(2)} - \frac{1}{2}p_1^{(3)}$, where $c^{(i)}$ refers to the cohomology class c coming from the i^{th} factor of BSpin or BSpin(n).¹⁹ Quotienting String-Spin $(16)^2$ by the Spin $(16)^2$ factor produces a map to Spin; composing with Spin $\rightarrow O$ we obtain a tangential structure as usual.

¹⁷Strictly speaking, the analysis of [53] does not take into account the full string spectrum. Therefore, a priori the correct gauge group G may differ from this particular quotient of $\text{Spin}(16)^2$.

¹⁸Like for any double cover, for any odd prime p, the quotient $BSpin(16) \times BSpin(16) \rightarrow BG$ is a p-primary equivalence, so the lack of p-primary torsion we establish for $Spin(16) \times Spin(16)$ remains valid for G.

¹⁹Elsewhere in the paper we have referred to $\frac{1}{2}p_1^L$ and $\frac{1}{2}p_1^R$ as Chern classes, and indeed they are Chern classes of the representations that play a role in the Green-Schwarz mechanism for this string theory. However, the bordism computation we perform in this section only depends on the characteristic class, not the representation (this is the thesis of [114]), so to emphasize this independence, we use the more intrinsic name $\frac{1}{2}p_1$, as this class is one-half of the first Pontrjagin class of the vector representation of Spin(n).

Theorem 3.78. In degrees 11 and below, the String-Spin $(16)^2$ bordism groups are:

$\Omega_0^{\text{String-Spin}(16)^2} \cong \mathbb{Z}$	$\Omega_6^{\text{String-Spin}(16)^2} \cong 0$
$\Omega_1^{\text{String-Spin}(16)^2} \cong \mathbb{Z}_2$	$\Omega_7^{\text{String-Spin}(16)^2} \cong 0$
$\Omega_2^{\text{String-Spin}(16)^2} \cong \mathbb{Z}_2$	$\Omega_8^{\text{String-Spin}(16)^2} \cong \mathbb{Z}^2 \oplus \mathbb{Z}^3 \oplus \mathbb{Z}$
$\Omega_3^{\text{String-Spin}(16)^2} \cong 0$	$\Omega_9^{\text{String-Spin}(16)^2} \cong (\mathbb{Z}_2)^{\oplus 2} \oplus (\mathbb{Z}_2)^{\oplus 2} \oplus \mathbb{Z}_2$
$\Omega_4^{\operatorname{String-Spin}(16)^2} \cong \mathbb{Z} \oplus \mathbb{Z}$	$\Omega_{10}^{\text{String-Spin}(16)^2} \cong (\mathbb{Z}_2)^{\oplus 3} \oplus (\mathbb{Z}_2)^{\oplus 3} \oplus \mathbb{Z}_2$
$\Omega_5^{\text{String-Spin}(16)^2} \cong 0$	$\Omega_{11}^{\text{String-Spin}(16)^2} \cong 0.$

The colors in the theorem statement will be explained below; they correspond to different summands in (an approximation to) $MT(\text{String-Spin}(16)^2)$.

Proof. The inclusion $i: \text{Spin}(16) \to \text{Spin}$ induces a map $Bi: B\text{Spin}(16) \to B\text{Spin}$ which is 15-connected, because it is an isomorphism on cohomology in degrees 15 and below. This map sends $\frac{1}{2}p_1$ to $\frac{1}{2}p_1$, so is compatible with the construction of String-Spin $(16)^2$ — that is, if String-Spin² is defined in the same way as String-Spin $(16)^2$ but using Spin instead of Spin(16), then *i* induces a map of tangential structures

 $i_2: B(\text{String-Spin}(16)^2) \to B(\text{String-Spin}^2),$ (3.79)

as well as the analogous map on bordism groups. Because *i* is 15-connected, i_2 is also 15-connected, so the induced map of Thom spectra is also 15-connected (e.g. check on cohomology, where it follows from 15-connectivity of i_2 via the Thom isomorphism). Therefore for $k \leq 15$, the map $\Omega_k^{\text{String-Spin}(16)^2} \to \Omega_k^{\text{String-Spin}^2}$ induced by *i* is an isomorphism. Therefore for the rest of this proof, we can work only with String-Spin² bordism without affecting the results.

Concretely, a string-Spin² structure on a vector bundle $E \to X$ is data of a spin structure on E and two virtual spin vector bundles $V^L, V^R \to X$ and a trivialization of $\frac{1}{2}p_1(E) - \frac{1}{2}p_1(V^L) - \frac{1}{2}p_1(V^R)$. Since $\frac{1}{2}p_1$ is additive in direct sums [46, Lemma 1.6], this is equivalent to a trivialization of $\frac{1}{2}p_1(E - V^L - V^R)$, meaning that a string-Spin² structure is equivalent to the data of V^L and V^R and a string structure on $W := E - V^L - V^R$.

The data (E, V^L, V^R) and (E, W, V^R) are equivalent, as $V^L = E - W - V^R$, and the spin structure on V^L can be recovered from the spin structures on E, W, and V^R by the two-out-of-three property (the string structure on W includes data of a spin structure). Therefore the data of a string-Spin² structure on $E \to X$ is equivalent to the following data:

- a spin structure on E,
- a virtual string vector bundle $W \to X$, and
- a virtual spin vector bundle $V^R \to X$.

Taking bordism groups, we learn

$$\Omega^{\text{String-Spin}^2}_* \xrightarrow{\cong} \Omega^{\text{Spin}}_* (B\text{Spin} \times B\text{String}).$$
(3.80)

For any spaces A and B, the stable splitting $\Sigma^{\infty}_{+}(A) \simeq \Sigma^{\infty}A \vee \mathbb{S}$ and its analogue for B together imply a stable splitting

$$\Sigma^{\infty}_{+}(A \times B) \simeq \mathbb{S} \vee \Sigma^{\infty} A \vee \Sigma^{\infty} B \vee \Sigma^{\infty}(A \wedge B), \qquad (3.81a)$$

implying that for any generalized homology theory h,

$$h_*(A \times B) \cong h_*(\mathrm{pt}) \oplus \widetilde{h}_*(A) \oplus \widetilde{h}_*(B) \oplus \widetilde{h}_*(A \wedge B).$$
(3.81b)

Here $\tilde{h}(X)$ denotes "reduced *h*-homology" of a space X, meaning the quotient $h(X)/i_*(h(\text{pt}))$ induced by a choice of basepoint $i: \text{pt} \to X$. Thus for example $\tilde{\Omega}_*^{\text{Spin}}(X)$ denotes reduced spin bordism, etc. Apply (3.81b) for $h = \Omega_*^{\text{Spin}}$, A = BSpin, and B = BString:

$$\Omega^{\text{String-Spin}^2}_* \cong \Omega^{\text{Spin}}_*(B\text{Spin} \times B\text{String}) \cong \Omega^{\text{Spin}}_* \oplus \widetilde{\Omega}^{\text{Spin}}_*(B\text{Spin}) \oplus \widetilde{\Omega}^{\text{Spin}}_*(B\text{String}) \oplus \widetilde{\Omega}^{\text{Spin}}_*(B\text{Spin} \wedge B\text{String}).$$
(3.81c)

The colors in (3.81c) indicating the pieces of this direct-sum decomposition correspond to the colors in the theorem statement displaying which pieces of the bordism groups come from which summands in (3.81c).

The final step is to determine the four summands in (3.81c).

- Ω_*^{Spin} was calculated by Milnor [130, section 3] and Anderson-Brown-Peterson [131].
- $\widetilde{\Omega}^{\text{Spin}}_{*}(B\text{Spin})$ was calculated by Francis [132, section 2.2].
- $\widetilde{\Omega}_*^{\text{Spin}}(B\text{String})$ was computed by Davis [133] at p = 2. At odd primes, these groups are easy to calculate in the range we need: because BString is 7-connected, $\widetilde{\Omega}_k^{\text{Spin}}(B\text{String})$ vanishes for k < 8; for $8 \le k \le 11$, use the Atiyah-Hirzebruch spectral sequence. Work of Stong [134] and Giambalvo [135] implies that in degrees 11 and below, $\widetilde{H}^*(B\text{String};\mathbb{Z})$ consists of a single summand isomorphic to \mathbb{Z} in degree 8, and the remaining groups vanish. This suffices to collapse the Atiyah-Hirzebruch spectral sequence into the blue groups in the theorem statement.
- For $A = \mathbb{Z}$ or \mathbb{Z}_2 , $\widetilde{H}^k(B\mathrm{Spin}; A)$ vanishes for k < 4, and $H^k(B\mathrm{String}; A)$ vanishes for k < 8, so by the Künneth formula, $\widetilde{H}^k(B\mathrm{Spin} \wedge B\mathrm{String}; A)$ vanishes for k < 12. Therefore the Atiyah-Hirzebruch spectral sequence for $\widetilde{\Omega}^{\mathrm{Spin}}_*(B\mathrm{Spin} \wedge B\mathrm{String})$ vanishes in degrees 11 and below.

Remark 3.82 (Analogy with $E_8 \times E_8$). The two-step simplification of String-Spin(16)² (first replace Spin(16) with Spin, then recast as spin bordism of a space) is directly analogous to Witten's [136, section 4] simplification of the symmetry type of the $E_8 \times E_8$ heterotic string: first, there is a 15-connected map $BE_8 \to K(\mathbb{Z}, 4)$, so in dimensions relevant to string theory we may replace the former with the latter; then Witten recast the data of the two maps to $K(\mathbb{Z}, 4)$ and the twisted string structure given by the Green-Schwarz procedure as a spin structure and a single map to $K(\mathbb{Z}, 4)$. Remark 3.83 (Analogy with Spin(32) and detecting a non-supersymmetric 0-brane). The same two-step procedure also works for the $Spin(32)/\mathbb{Z}_2$ heterotic string when one restricts to Spin(32)-bundles, showing that the relevant twisted string bordism groups coincide with $\Omega_*^{\text{Spin}}(B\text{String})$, which vanishes in dimension 11. As with Spin(16) × Spin(16), this is only partial information towards a complete anomaly cancellation result.

However, the partial information provided by these bordism groups is already useful: combined with the Cobordism Conjecture [48], it detects Kaidi-Ohmori-Tachikawa-Yonekura's non-supersymmetric 0-brane [50]. To see this, consider $\Omega_8^{\text{Spin}}(B\text{String}) \cong \mathbb{Z}^3$: two of the \mathbb{Z} summands come from $\Omega_*^{\text{Spin}}(\text{pt})$, and as such are generated by \mathbb{HP}^2 and the Bott manifold; the third \mathbb{Z} summand is represented by S^8 with the map to BString given by the generator of $[S^8, B\text{String}] = \pi_8(B\text{String}) \cong \mathbb{Z}$. Tracing through the simplification from twisted string bordism of BSpin(32) to the spin bordism of BString, we see that this S^8 has the Spin(32)bundle arising from the generator of $\pi_8(\text{Spin}(32)) \cong \mathbb{Z}$, which is detected by p_2 .

The Cobordism Conjecture predicts that associated to this bordism class (or rather its image in the corresponding bordism group for $Spin(32)/\mathbb{Z}_2$), there is a 0-brane in Spin(32)/ \mathbb{Z}_2 heterotic string theory whose link is S^8 with this Spin(32)-bundle and twisted string structure. This is precisely the 0-brane discovered by Kaidi-Ohmori-Tachikawa-Yonekura [50]. Those authors also discuss a 6-brane in the $Spin(32)/\mathbb{Z}_2$ heterotic string, but its description uses $\pi_1(Spin(32)/\mathbb{Z}_2) \cong \mathbb{Z}_2$, so it is invisible to the Spin(32) computation we made here.

3.4 Physical intuition from fivebrane anomaly inflow

As we have just seen, the relevant bordism groups vanish, and therefore there are no Dai-Freed anomalies (except possibly for the Sagnotti string). It is instructive to study the vanishing of anomalies more explicitly in particular examples, to better understand the physics at play. Let us recall from section 3 the structure of the anomaly theory for ten dimensional theories that feature a Green-Schwarz mechanism:

$$\alpha_{\rm GS}(Y_{11}) = \int_{Y_{11}} H \wedge X_8. \tag{3.84}$$

The boundary mode of this eleven dimensional field theory gives exactly the contribution of the Green-Schwarz term to the classical action:

$$S_{\rm GS} = \int_{Y_{10}} B_2 \wedge X_8.$$
 (3.85)

In this section, we consider simple backgrounds of the factorized form $Y_{11} = S^3 \times M_8$ for the anomaly theory (3.84). We will also take one unit of three-form H flux threading the sphere, so that the Green-Schwarz term gives a nontrivial contribution, and M_8 a spin manifold equipped with a gauge bundle E such that $\frac{p_1(M_8)+c_2(E)}{2}$ is trivial in integer cohomology. Unlike more general backgrounds, these factorized ones allows for an intuitive understanding of how anomalies are cancelled, via inflow.

On these backgrounds, the eta invariant contribution to the anomaly theory (coming from the fermions) vanishes on account of the factorization property

$$\eta(A \times B) = \eta(A) \operatorname{index}(B), \tag{3.86}$$

where A is odd-dimensional. The eta invariant of fermions on S_H^3 vanishes modulo 1, as it is the same as the eta invariant on a three-sphere, which is the boundary of \mathbb{R}^4 . As a result, the anomaly theory simplifies to the Green-Schwarz term

$$\alpha(S_H^3 \times M_8) = \int_{M_8} X_8 \,. \tag{3.87}$$

If we can now show that this quantity is always an integer, Dai-Freed anomalies will vanish on all such factorized backgrounds. This result does not hinge on the precise bordism groups computed in the preceding section.

In order to prove that eq. (3.87) is always an integer, we can connect it with the anomaly inflow mechanism on a fivebrane. Specifically, S_H^3 is a non-trivial bordism class, and one possible boundary for it in string theory is a fivebrane. The fivebrane is a codimension four object, and it is characterized precisely by the fact that the angular S^3 in the transverse space is threaded by one unit of H-flux. These fivebranes are precisely D5-branes in the orientifold models and NS5-branes in the heterotic model. We will now show that X_8 coincides with the anomaly polynomial of such a fivebrane, up to terms which vanish when the Bianchi identity holds. In the presence of fivebranes coupling to B_6 , the dual of B_2 , the classical gauge variation of the effective action is compensated by the quantum anomaly of the chiral worldvolume degrees of freedom. For this inflow mechanism to work, the anomaly polynomial of a single fivebrane has to be $I_8 = X_8$ (up to terms that vanish on a twisted String manifold). To see this, notice that the bulk action receives additional worldvolume contributions of the form

$$S = S_{\text{bulk}} + S_{\text{GS}} + S_{\text{wv}} + \mu \int_{W} B_6 ,$$
 (3.88)

where W denotes the worldvolume of the fivebrane(s) and B_6 the dual of B_2 . The bulk action S_{bulk} , which describes the ten-dimensional effective (super)gravity theory, is accompanied by the Green-Schwarz term S_{GS} of eq. (2.17) to cancel bulk anomalies. The brane is instead described by the worldvolume DBI action S_{wv} accompanied by the magnetic coupling to B_6 , which is the relevant coupling in the following argument. The equation of motion for B_6 and the corresponding dual Bianchi identity are

$$d \star dB_6 = \mu \,\delta(W \hookrightarrow M_{10}) \,,$$

$$dH_3 = \mu \,\delta(W \hookrightarrow M_{10}) \,.$$
(3.89)

where $H_3 = dB_2$ and the δ is a distribution-valued four-form that describes the embedding of W in spacetime. Correspondingly, the Bianchi identity for the gauge invariant field strength $H \equiv \tilde{H}_3 = dB_2 - \omega_{\rm CS}$, which ordinarily reads $dH = X_4$, also receives a new localized contribution. Because of this Bianchi identity, there is a new classical contribution to the gauge variations. Using descent, $X_8 = dX_7^{(0)}$, $\delta X_7^{(0)} = dX_6^{(1)}$, one finds

$$\delta_{\text{new}} S_{\text{GS}} = -\int_{M_{10}} dB_2 \wedge \delta X_7^{(0)}$$

= $-\int_{M_{10}} dH_3 X_6^{(1)}$
= $-\mu \int_W X_6^{(1)}$, (3.90)

which cancels by inflow provided that the worldvolume theory of the D5-brane has an anomaly polynomial $I_8 = \mu X_8$ [137, 138]. With our choice of units, the elementary charge $\mu = n_5 \in \mathbb{Z}$ counts the number of fivebranes.

As described in section 2, the anomaly polynomial is a sum of indices, given by the APS index theorem for each one of the anomalous degrees of freedom propagating on the fivebrane. As such, we know that I_8 is an integer and we can conclude from the previous discussion that X_8 must also be an integer. This anomaly inflow argument thus allows one to show that the anomaly (3.87) always vanishes.

For the orientifold models, this mechanism can be implemented explicitly, since these theories have D5-branes whose worldvolume degrees of freedom are known. We describe this in detail in section 3.4.1.

On the heterotic side, although the $SO(16) \times SO(16)$ theory is known to have NS5 branes, their worldvolume degrees of freedom are not known and so we have to resort to other arguments to prove that the anomaly (3.87) vanishes. The proof can be found at the end of section 3.4.2. There is, however, a more physical way of understanding why anomalies cancel in the heterotic case: one can show that the anomaly polynomial of $SO(16) \times SO(16)$ can be directly related to that the supersymmetric heterotic theories. Therefore, one use this connection to show that anomalies cancel for $SO(16) \times SO(16)$ by showing that they cancel in the supersymmetric cases. This is done explicitly in section 3.4.2.

Finally, since we have proven that anomalies vanish in the heterotic case, we can reverse the anomaly inflow argument above to speculate about the worldvolume degrees of freedom of the NS5 brane. Indeed, we identify what kind of degrees of freedom give rise to the correct anomaly polynomial so as to have $X_8 = I_8$. We do so away from strong coupling effects, in the puffed-up instanton limit of the NS5 brane. This is detailed in section 3.4.3.

3.4.1 Sp(16) and U(32)

Let us begin with the orientifold models. This cancellation of anomalies by inflow was first constructed for the case of $\text{Spin}(32)/\mathbb{Z}_2$ in [137, 138]. The chiral fermions on the worldvolume of the D5-brane consist of one vector multiplet of $\text{Spin}(32)/\mathbb{Z}_2$ and two gauge singlets, such that,

$$(X_8)_{\text{Spin}(32)/\mathbb{Z}_2} - I_{\text{Spin}(32)/\mathbb{Z}_2} = -\frac{1}{24} p_1(X_4)_{\text{Spin}(32)/\mathbb{Z}_2}, \qquad (3.91)$$

where $(X_8)_{\text{Spin}(32)/\mathbb{Z}_2}$ and $(X_4)_{\text{Spin}(32)/\mathbb{Z}_2}$ can be read off from (2.13). This shows how one recovers $I_8 = X_8$ up to a term that vanishes on a twisted string manifold.

One may wonder where the extra term in (3.91) comes from, even if we know it to vanish on a twisted string manifold. This can be understood as follows; from the perspective of the ten-dimensional supergravity action, a D5-brane amounts to introducing a delta function localized on the brane. The D5-brane gives a localized contribution to the 10d action of the form $B_2 \wedge Y_4 \wedge \delta_4$ where Y_4 is some 4-form which in this case is reduces to $Y_4 = -\frac{1}{24}p_1$, expanding the A-roof genus in the Chern-Simons effective worldvolume action. Indeed, using the Bianchi identity for the H_3 flux, we see that this term contributes to the anomaly polynomial as:

$$\int_{Z_{12}} X_4 \wedge Y_4 \wedge \delta_4 = \int_{X_8} X_4 \wedge Y_4 \,. \tag{3.92}$$

Therefore, the appearance of the extra term in (3.91) can be traced down to not properly taking into account the delta-function source that corresponds to the localized D5-brane.

The same mechanism happens in the two non-supersymmetric orientifold models, as was found by [139] along the lines of [138, 140]. The worldvolume degrees of freedom on D5-branes can be extracted from one-loop open-string amplitudes [141], and the chiral fermions arrange in the virtual representation

$$\left(\frac{\mathbf{N}(\mathbf{N}+\mathbf{1})}{\mathbf{2}},\mathbf{1}\right) - \left(\frac{\mathbf{N}(\mathbf{N}-\mathbf{1})}{\mathbf{2}},\mathbf{1}\right) - (\mathbf{N},\mathbf{32}) \tag{3.93}$$

of $SO(N) \times Sp(16)$ (for the Sugimoto model²⁰) or $U(N) \times U(32)$ (for the Sagnotti model). In order to compare the anomaly polynomial I_8 (without worldvolume gauge field) with the bulk X_8 , one needs to decompose characteristic classes of the bulk tangent bundle in terms of the worldvolume tangent bundle TW and normal bundle N. In detail,

$$p_{1}(TM_{10}) = p_{1}(TW) + p_{1}(N),$$

$$p_{2}(TM_{10}) = p_{2}(TW) + p_{1}(TW) p_{1}(N) + p_{2}(N),$$

$$p_{1}(N) = c_{1}(N)^{2} - 2 c_{2}(N),$$

$$p_{2}(N) = c_{2}(N)^{2} = \chi(N)^{2}.$$
(3.94)

When the normal bundle of the worldvolume is trivial, one obtains

$$I_8 - X_8 \propto p_1(TM_{10}) X_4 \,, \tag{3.95}$$

and therefore the inflow mechanism implies that X_8 integrates to an integer on any spin 8-manifold with $X_4 = 0$. When the normal bundle is non-trivial, there are additional contributions to the above expression, proportional to the Euler class of N. However, the full brane action also contains another term [137, 138] proportional to B_2 rather than B_6 , which induces another classical variation to be canceled by inflow. As a result, the anomaly polynomial of the fivebrane worldvolume theory is not quite the above I_8 , but has an additional contribution that cancels the normal bundle terms [138]. In more detail, adding a coupling of the type $\int_W B_2 Y_4$ to the fivebrane worldvolume action contributed a new classical variation to the effective action, which arises by descent from $\Delta I_8 = -(X_4 + n_5 \chi(N))Y_4$. Therefore, the full anomaly polynomial of the fivebrane worldvolume ought to be $I_8 = n_5 X_8 - \Delta I_8$, again up to terms that vanish on twisted String backgrounds. This additional coupling can be shown to cancel the normal bundle terms in the anomaly [138] (see also [142] for a discussion in the context of M-theory).

3.4.2 $SO(16) \times SO(16)$

For the heterotic model, no such result is available, since the worldvolume degrees of freedom of NS5-branes are not understood without supersymmetry or dualities at one's disposal. However, one can nonetheless express X_8 as an index of six-dimensional chiral fields; since index are manifestly integers, this will be enough to establish that anomalies cancel. In order

²⁰In this case, a single brane corresponds to N = 2.

to do so, let us observe that the formal difference of representations of the chiral fermions of the non-supersymmetric heterotic model can be rewritten as^{21}

$$(128, 1) + (1, 128) - (16, 16)$$

= (128, 1) + (1, 128) + (120, 1) + (1, 120)
- (120, 1) - (1, 120) - (16, 16), (3.96)

The matter fields in the first line after the equal correspond precisely to the decomposition of the adjoint of $\mathfrak{e}_8 \oplus \mathfrak{e}_8$ into representations of the $\mathfrak{so}_{16} \oplus \mathfrak{so}_{16}$ subalgebra; they are the field content that would arise after giving a vev to an adjoint $\mathfrak{e}_8 \oplus \mathfrak{e}_8$ field. Similarly, the fields in the second line are (with reversed chirality) those fields that would arise after adjoint Higgsing from the $\mathfrak{so}(32)$ algebra to its $\mathfrak{so}_{16} \oplus \mathfrak{so}_{16}$ subalgebra. What we are seeing here is that, at a formal level (as far as the chiral spectrum is concerned), the SO(16)² is equivalent to one copy of the $E_8 \times E_8$ string stacked on top of a copy of the Spin(32)/ \mathbb{Z}_2 string, with opposite chirality, and Higgsed to a common subgroup with algebra $\mathfrak{so}_{16} \oplus \mathfrak{so}_{16}$. Therefore, we can write, at the level of anomaly polynomials, the equality

$$P_{12}^{E_8 \times E_8}|_{\mathrm{SO}(16)^2} - P_{12}^{\mathrm{Spin}(32)/\mathbb{Z}_2}|_{\mathrm{SO}(16)^2} = P_{12}^{\mathrm{SO}(16)^2}, \qquad (3.97)$$

where we have merely restricted to $SO(16)^2$ bundles inside of the two groups above. Since each of the supersymmetric string theories are anomaly-free by themselves, the formal linear combination will also be. This argument, which can be carried out at the level of eta invariants etc. and not just anomaly polynomials, is yet another proof of the fact that the $SO(16)^2$ theory is anomaly free,²² without relying explicitly on bordism calculations. Furthermore, in particular, this holds for the Green-Schwarz terms, which are

$$(X_8)_{\text{Spin}(32)/\mathbb{Z}_2}|_{\text{SO}(16)^2} - (X_8)_{E_8 \times E_8}|_{\text{SO}(16)^2} = \frac{1}{24} \left((c_{16,2}^{(1)})^2 + (c_{16,2}^{(2)})^2 + c_{16,2}^{(1)} c_{16,2}^{(2)} - 4 c_{16,4}^{(1)} - 4 c_{16,4}^{(2)} \right) = (X_8)_{\text{SO}(16)^2}.$$
(3.98)

It is unclear whether this connection between the non-supersymmetric $SO(16) \times SO(16)$ theory and the supersymmetric theories persists beyond a formal equality at the level of (super)gravity, or whether on the contrary it has a deeper meaning. Some previous work [143, 144] (see also [145]) identified connections between supersymmetric and non-supersymmetric strings via interpolating models, which are nine-dimensional compactifications recovering either supersymmetric or non-supersymmetric strings in different decompactification limits. In particular, an interpolating model was constructed between the $SO(16) \times SO(16)$ theory and the supersymmetric $Spin(32)/\mathbb{Z}_2$ theory, matching the worldsheet CFT descriptions and solitons in between the two.²³

The cancellation of anomalies by fivebrane inflow for the $SO(16) \times SO(16)$ theory thus follows from that of the two supersymmetric heterotic theories. The anomaly inflow in the

 $^{^{21}}$ To our knowledge, this was first explicitly stated in the literature in [71], though we learned from Luis Álvarez-Gaumé that the authors of [2] were also aware of this fact.

 $^{^{22}}$ Even with the right global quotient.

 $^{^{23}}$ Since the non-supersymmetric theories have NS-NS tadpoles and would-be moduli run in the absence of (large) stabilizing fluxes [146, 147] and/or spacetime warping [148, 149], there may be additional subtleties in understanding the dynamics of this duality.

case of Spin(32)/ \mathbb{Z}_2 was discussed above (3.91). The case of $E_8 \times E_8$ is slightly more involved and we will discuss it now. The anomaly inflow of the NS5-brane was famously discussed in [150] where the limit in which an instanton in $E_8 \times E_8$ becomes point-like was matched to the world volume theory of the NS5 brane at strong coupling. For our purposes, we can ignore strong coupling dynamics and focus on matching a 6-dimensional anomaly theory of chiral fermions to $(X_8)_{E_8 \times E_8}$ which can be read off from (2.14). This guarantees that $(X_8)_{E_8 \times E_8}$ is an integer and that local anomalies cancel in 10d. We now detail how this can be done.

One can show that $(X_8)_{E_8 \times E_8}$ can be decomposed as follows:

$$(X_8)_{E_8 \times E_8} = \frac{(c_{16,2}^{(1)} - c_{16,2}^{(2)})^2}{32} + \frac{1}{24}X_4^2 + \mathcal{I}_{SD} + 2\mathcal{I}_{Dirac}$$
(3.99)

where $\mathcal{I}_{SD} + 2\mathcal{I}_{Dirac}$ is the index of a self-dual form field and 2 fermion singlets in 8 dimensions. The index of a self-dual form field in 8 dimensions can be shown to be an integer over 8 [42]. Indeed, it can be written in terms of the signature of the 8-manifold as follows [103]:

$$\mathcal{I}_{\rm SD} = -\frac{\sigma}{8} \,. \tag{3.100}$$

On the other hand, the index of chiral fermions is always an integer. In order to simplify the first term in (3.99), we can rewrite the Chern classes in an embedded SU(2) subgroup of each E_8 , which are known to be integer-valued. Therefore, on a twisted string manifold, X_8 reduces to:

$$X_8 = \frac{(c_{2,2}^{(1)} - c_{2,2}^{(2)})^2}{8} - \frac{\sigma}{8} + n \quad \text{with} \ n \in \mathbb{Z}$$
(3.101)

where $\frac{1}{2}c_{16,2}^{(i)} \rightarrow c_{2,2}^{(i)}$ are the 2nd Chern classes in the fundamental of the SU(2) subgroup of the i-th E_8 . The Bianchi identity $(X_4)_{E_8 \times E_8} = 0$ gives us

$$c_{\mathbf{2},2}^{(2)} = -\frac{p_1}{2} - c_{\mathbf{2},2}^{(1)} \,. \tag{3.102}$$

Plugging this into (3.101), we see that the condition for anomalies to vanish comes down to showing that the following quantity is an integer:

$$\frac{(c_{\mathbf{2},2}^{(1)})^2}{2} + \frac{c_{\mathbf{2},2}^{(1)} p_1}{4} + \frac{p_1^2}{32} - \frac{\sigma}{8}.$$
(3.103)

As it happens, it was shown in [103] that the last two terms give (28 times) an integer. Indeed, one can show that:

$$28 \mathcal{I}_{\text{Dirac}} = \frac{p_1^2}{32} - \frac{\sigma}{8} \,. \tag{3.104}$$

Now, to show that the first two terms of (3.103) are an integer, one can note that on a twisted string manifold, $\frac{p_1}{2}$ is a characteristic vector of $H^4(X;\mathbb{R})$. This, in particular, means that:

$$\frac{p_1}{2}c_{2,2}^{(i)} \mod 2 = (c_{2,2}^{(i)})^2 \mod 2.$$
(3.105)

Therefore we have shown that on a twisted string manifold, X_8 is always an integer; and so there can never be an anomaly.

Given that the X_8 of SO(16) × SO(16) is a linear combination of those of $E_8 \times E_8$ and $Spin(32)/\mathbb{Z}_2$, we can infer that $(X_8)_{SO(16)^2}$ is an integer and so that all anomalies vanish for this non-supersymmetric theory. Nevertheless, for completeness, let us detail explicitly how $(X_8)_{SO(16)^2}$ can be proven to be an integer. One can write $(X_8)_{SO(16)^2}$ as follows:

$$(X_8)_{\rm SO(16)^2} = -\frac{1}{32} (c_{\mathbf{16},2}^{(1)} - c_{\mathbf{16},2}^{(2)})^2 - \mathcal{I}_{\rm SD} - 4 \mathcal{I}_{\rm Dirac}$$

$$-\frac{1}{48} (X_4)_{\rm SO(16)^2} (c_{\mathbf{16},2}^{(1)} + c_{\mathbf{16},2}^{(2)} + 3p_1) + \mathcal{I}_{\rm Dirac}^{\mathbf{16}_{(1)}} + \mathcal{I}_{\rm Dirac}^{\mathbf{16}_{(2)}}$$

$$(3.106)$$

where $\mathcal{I}_{\text{Dirac}}^{\mathbf{16}_{(i)}}$ is the contribution of a fermion that transforms in the **16** of SO(16)_i, which is known to be integer-valued. Therefore, on a twisted string manifold, the cancellation of anomalies comes down to showing that the following quantity is an integer:

$$-\frac{1}{32}(c_{\mathbf{16},2}^{(1)} - c_{\mathbf{16},2}^{(2)})^2 - \mathcal{I}_{SD} = -\frac{1}{32}(c_{\mathbf{16},2}^{(1)} - c_{\mathbf{16},2}^{(2)})^2 + \frac{\sigma}{8}$$
(3.107)

Given that one can put the 2nd Chern classes in the SU(2) subgroup of SO(16) as $c_{16,2}^{(i)} \rightarrow 2c_{2,2}^{(i)}$; the proof goes exactly as in the $E_8 \times E_8$ case.

One can sometimes read-off the chiral field content of a theory from the anomaly polynomial. For instance, for the $E_8 \times E_8$ case, the anomaly polynomial (3.99) suggests that the chiral field content of the NS5 brane is a self-dual form field and 2 fermion singlets. There are no chiral fields charged under the gauge group, since all the gauge-dependent parts of (3.99) are in the factorized piece. As it happens, this exactly the chiral field content of a 6d (1,0) tensor multiplet, which is precisely the worldvolume field content of the NS5 brane in $E_8 \times E_8$ string theory. This answer is essentially determined by anomalies together with supersymmetry. In the non-supersymmetric case of the SO(16)² string, reading off the chiral field content from (3.106) in the same way suggests that the chiral field content of the SO(16)² NS5 brane is:

- Four fermion singlets,
- A fermion transforming in the $(16, 1) \oplus (1, 16)$ of SO(16),
- A self-dual 2-form field.

There are some subtleties in assessing whether or not these are truly the chiral degrees of freedom propagating on this non-supersymmetric brane. First of all, there is no supersymmetry to constrain the worldvolume theory of the NS5 brane which can therefore carry any kind of chiral degrees of freedom. As we have seen from (3.104) and (3.100), indices can sometimes be exchanged for one another and yet give the same integer. This means that the X_8 does not completely fix the worldvolume content of the NS5 brane, and that any chiral field content with the same anomaly as the one proposed above remains a possibility. Another reason why we cannot be sure that (3.106) correctly describes the degrees of freedom propagating on the NS5 brane is that we cannot be sure that an NS5 brane (understood as a small instanton where the full spacetime gauge group symmetry gets restored) exists to begin with. Unlike in the supersymmetric case, in general we expect that the size modulus of the instanton, being



Figure 7. A sketch of a fivebrane puffing up into an instanton which can be described within the effective field theory.

non-supersymmetric, receives a potential due to quantum effects that may lead to the small instanton limit being obstructed. The study of the strong coupling effects near the small instanton limit is beyond the validity of effective field theory, and thus beyond the scope of this paper (although it may be amenable to a version of the constructions in [50]), but we point out that, if the limit does exist and the small instanton transition does survive, the transition point would be a natural place to look for a non-supersymmetric interacting CFT, a cousin of the E_8 SCFT. It would be interesting to explore this further. On the other hand, studying the anomaly inflow on the worldvolume of the puffed-up NS5 brane instanton is accessible within the effective field theory (see figure 7). We do this explicitly in the next section.

3.4.3 Anomaly inflow on puffed-up fivebrane instantons

The above result shows that there are no Dai-Freed anomalies on factorized backgrounds of the form $S_H^3 \times M_8$, since X_8 integrates to an integer. Anomaly inflow on fivebranes dictates that X_8 be the anomaly polynomial associated to the worldvolume theory on a single fivebrane, possibly up to terms that vanish when the Bianchi identity is satisfied. As explained above, without direct access to the relevant degrees of freedom on the fivebrane worldvolume, studying the anomaly inflow on the point-like NS5 brane is impossible. Luckily, one can still examine the anomaly inflow on the worldvolume of puffed-up fivebrane instantons, which can be described in the low-energy approximation. Puffing-up the fivebrane corresponds to delocalizing it along its transverse dimensions, making it look like a four dimensional gauge instanton.

Introducing an instanton Higgses one of the SO(16) factors, say the first SO(16)⁽¹⁾, according to SO(16) \rightarrow SU(2)^(a) × SU(2)^(b) × SO(12), so that the vector and spinor representations branch into

$$\mathbf{16} = (\mathbf{1^{(a)}}, \mathbf{1^{(b)}}, \mathbf{12}) + (\mathbf{2^{(a)}}, \mathbf{2^{(b)}}, \mathbf{1}), \qquad (3.108a)$$

$$128 = (2^{(a)}, 1^{(b)}, 32) + (1^{(a)}, 2^{(b)}, \bar{32}).$$
(3.108b)

If the instanton bundle only involves $SU(2)^{(a)}$, there an $SU(2)^{(b)} \times SO(12) \times SO(16)^{(2)}$ unbroken symmetry and the background has fermion zero modes (fzm) arising from the representations $(\mathbf{8_s}, \mathbf{16^{(1)}}, \mathbf{16^{(2)}})$ and $(\mathbf{8_c}, \mathbf{128^{(1)}}, \mathbf{1^{(2)}})$ of the spacetime isometries and the original gauge group. As a result, one has

$$\begin{cases} 1 \text{ fzm in the rep } (\mathbf{8_s}, \mathbf{2^{(b)}}, \mathbf{1}, \mathbf{16^{(2)}}), \\ 1 \text{ fzm in the rep } (\mathbf{8_c}, \mathbf{1^{(b)}}, \mathbf{32}, \mathbf{1^{(2)}}). \end{cases}$$
(3.109)

The two types of fermion zero modes have different chirality, and thus the corresponding worldvolume anomaly polynomial reads

$$P_8 = \frac{1}{2} [\hat{A}(R)(\operatorname{ch}(F)_{(\mathbf{2}^{(\mathbf{b})},\mathbf{1},\mathbf{16}^{(\mathbf{2})})} - \operatorname{ch}(F)_{(\mathbf{1}^{(\mathbf{b})},\mathbf{32},\mathbf{1}^{(\mathbf{2})})})]_8, \qquad (3.110)$$

which evaluates to

$$P_{8} = \frac{1}{24} \left[-2p_{1}c_{\mathbf{12},2} + p_{1}(c_{\mathbf{16},2}^{(2)} + 8c_{\mathbf{2},2}^{(b)}) - c_{\mathbf{12},2}^{2} - 4c_{\mathbf{12},4} + 2(c_{\mathbf{16},2}^{(2)} + 2c_{\mathbf{2},2}^{(b)})(c_{\mathbf{16},2}^{(2)} + 4c_{\mathbf{2},2}^{(b)}) - 4c_{\mathbf{16},4}^{(2)} \right].$$

$$(3.111)$$

The next step is to evaluate X_8 on this background. This amounts to decomposing characteristic classes according to the branching rules, and one finds

$$c_{\mathbf{16},2}^{(1)} \to c_{\mathbf{12},2} + 2c_{\mathbf{2},2}^{(a)} + 2c_{\mathbf{2},2}^{(b)}, c_{\mathbf{16},4}^{(1)} \to 2c_{\mathbf{12},2}c_{\mathbf{2},2}^{(a)} + 2c_{\mathbf{12},2}c_{\mathbf{2},2}^{(b)} + c_{\mathbf{12},4} - 2c_{\mathbf{2},2}^{(a)}c_{\mathbf{2},2}^{(b)} + (c_{\mathbf{2},2}^{(a)})^2 + (c_{\mathbf{2},2}^{(b)})^2.$$
(3.112)

Finally, $(c_{2,2}^{(a)})^2$ should be replaced by zero, since it is proportional to the square of the worldvolume current $\delta(W)$ of the fivebrane. All in all, when the dust settles one arrives at

$$X_8 - P_8 = \frac{1}{24} \left(2c_{12,2} - c_{16,2}^{(2)} - 8c_{2,2}^{(b)} \right) \left(c_{12,2} + 2c_{2,2}^{(b)} + c_{16,2}^{(2)} + p_1 \right) , \qquad (3.113)$$

where the second factor corresponds to X_4 for the unbroken piece of the gauge group. The inflow therefore works when X_8 and P_8 are equal on manifolds where the Bianchi identity holds. A similar argument works for more general choices of instanton bundles.

For the orientifold models, one expects the small limit of the "fat" fivebrane instantons to yield the worldvolume degrees of freedom of D5-branes. This is a nice crosscheck that we detail now. For the Sugimoto model (the calculation is identical in the Sagnotti model), the anomaly polynomial P_8 associated to the fermion zero modes of the instanton is

$$P_8 = [\hat{A}(R)\mathrm{ch}(F)_{\mathbf{30}}]_8 = \frac{1}{192} \left(8p_1c_{\mathbf{30},2} - 4\left(8c_{\mathbf{30},4} + p_2\right) + 16c_{\mathbf{30},2}^2 + 7p_1^2 \right), \qquad (3.114)$$

since under the branching $\text{Sp}(16) \rightarrow \text{SU}(2) \times \text{USp}(30)$ the adjoint representation, containing the gauginos, decomposes according to

$$f 495 = (f 2, f 30) + (f 1, f 434) + (f 1, f 1)$$

where the only charged contribution comes from the first term on the right-hand side. In the small limit, the SU(2) Chern classes vanish and the remaining ones are enhanced to Sp(16) classes, ending up with

$$P_8^{\text{small}} = \frac{1}{192} \left(8p_1 c_{32,2} - 4 \left(8c_{32,4} + p_2 \right) + 16c_{32,2}^2 + 7p_1^2 \right) = X_8 + \frac{1}{24} X_4 p_1.$$
(3.115)

Thus reproducing the anomaly polynomial of a D5-brane worldvolume up to terms that vanish on the allowed backgrounds.

4 Anomalies and bordism for the swap \mathbb{Z}_2 action

4.1 Overview and the bordism computation

The $E_8 \times E_8$ heterotic string theory has a \mathbb{Z}_2 symmetry given by swapping the two copies of E_8 , so it is possible to expand the gauge group of the theory to $(E_8 \times E_8) \rtimes \mathbb{Z}_2$. To our knowledge, this fact first appears in [53, section I] (see also [151, section 2.1.1]). The question of anomaly cancellation for this string theory is completely different in the absence versus in the presence of this extra \mathbb{Z}_2 : without it, the anomaly is known to vanish, as Witten [136, section 4] showed it is characterized by a bordism invariant $\Omega_{11}^{\text{Spin}}(BE_8) \to \mathbb{C}^{\times}$, and Stong [152] showed $\Omega_{11}^{\text{Spin}}(BE_8) \cong 0$. But with the \mathbb{Z}_2 swapping symmetry turned on, the relevant bordism group has order 64 [46, Theorem 2.62] courtesy of a harder computation; even though we cannot determine this group exactly, we will show that the anomaly vanishes, in accordance with the results in [107] obtained from a worldsheet perspective.

In this section, we discuss a closely analogous story for the Spin(16) × Spin(16) nonsupersymmetric heterotic string. The gauge group Spin(16) ×_{Z₂} Spin(16) (where the diagonal Z₂ we quotient by corresponds to either of the subgroups in each Spin(16) whose quotient is not SO(16)) admits a Z₂ automorphism switching the two Spin(16) factors, enlarging the gauge group of this theory to (Spin(16) ×_{Z₂} Spin(16)) × Z₂; see [53, section III].

In this paper, we chose to work with $\text{Spin}(16) \times \text{Spin}(16)$, which simplifies the bordism computations at the expense of applying to only some backgrounds. The \mathbb{Z}_2 symmetry enlarges the gauge group to $G_{16,16} := (\text{Spin}(16) \times \text{Spin}(16)) \rtimes \mathbb{Z}_2$. The Green-Schwarz mechanism is analogous: if $x \in H^*(B\text{Spin}(16); A)$ for some coefficient group A, let x^L and x^R denote the copies of x in $H^*(B(\text{Spin}(16) \times \text{Spin}(16)); A)$ coming from the first, resp. second copies of Spin(16) via the Künneth formula. Then the class $\frac{1}{2}p_1^L + \frac{1}{2}p_1^R$, which was the characteristic class of the Green-Schwarz mechanism in the absence of the \mathbb{Z}_2 symmetry, descends through the Serre spectral sequence for the fibration

to define a class in $H^*(BG_{16,16};\mathbb{Z})$, and the Green-Schwarz mechanism asks, on a spin manifold M with a principal $G_{16,16}$ -bundle $P \to M$, for a trivialization of

$$\frac{1}{2}p_1(M) - (\frac{1}{2}p_1^L + \frac{1}{2}p_1^R)(P).$$
(4.2)

Let $\mathbb{G}_{16,16}$ denote the Lie 2-group corresponding to this data, i.e. the string cover of Spin × $G_{16,16}$ corresponding to the class (4.2). Quotienting by $G_{16,16}$ defines a map to Spin and therefore a tangential structure in the usual way; a $\mathbb{G}_{16,16}$ -structure on a vector bundle $E \to M$ is a spin structure on E, a double cover $\pi \colon M' \to M$, a pair of rank-16 spin vector bundles V^L and V^R on M' identified under the deck transformation of M', and a trivialization of $\frac{1}{2}p_1(E) - (\frac{1}{2}p_1(V^L) - \frac{1}{2}p_1(V^R))$ (the class $\frac{1}{2}p_1(V^L) + \frac{1}{2}p_1(V^R)$ descends from M' to M). If the double cover $M' \to M$ is trivial, this is equivalent to a Spin-Spin(16)² structure as defined in section 3.3.3.

Theorem 4.3.

$$\begin{split} \Omega_{0}^{\mathbb{G}_{16,16}} &\cong \mathbb{Z} & \Omega_{6}^{\mathbb{G}_{16,16}} \cong \mathbb{Z}_{2} \\ \Omega_{1}^{\mathbb{G}_{16,16}} &\cong (\mathbb{Z}_{2})^{\oplus 2} & \Omega_{7}^{\mathbb{G}_{16,16}} \cong \mathbb{Z}_{16} \\ \Omega_{2}^{\mathbb{G}_{16,16}} &\cong (\mathbb{Z}_{2})^{\oplus 2} & \Omega_{8}^{\mathbb{G}_{16,16}} \cong \mathbb{Z}^{\oplus 3} \oplus (\mathbb{Z}_{2})^{\oplus i} \\ \Omega_{3}^{\mathbb{G}_{16,16}} &\cong \mathbb{Z}_{8} & \Omega_{9}^{\mathbb{G}_{16,16}} \cong (\mathbb{Z}_{2})^{\oplus j} \\ \Omega_{4}^{\mathbb{G}_{16,16}} &\cong \mathbb{Z} \oplus \mathbb{Z}_{2} & \Omega_{10}^{\mathbb{G}_{16,16}} \cong (\mathbb{Z}_{2})^{\oplus k} \\ \Omega_{5}^{\mathbb{G}_{16,16}} &\cong 0 & \Omega_{10}^{\mathbb{G}_{16,16}} \cong A, \end{split}$$

where either i = 1, j = 4, and k = 4, or i = 2, j = 6, and k = 5, and A is an abelian group of order 64 isomorphic to one of $\mathbb{Z}_8 \oplus \mathbb{Z}_8$, $\mathbb{Z}_{16} \oplus \mathbb{Z}_4$, $\mathbb{Z}_{32} \oplus \mathbb{Z}_2$, or \mathbb{Z}_{64} .

The fact that $\Omega_{11}^{\mathbb{G}_{16,16}} \neq 0$ implies that the $\text{Spin}(16) \times \text{Spin}(16)$ heterotic theory with its \mathbb{Z}_2 swapping symmetry could have an anomaly; we will nevertheless be able to cancel it later in this section.

Proof. The proof is nearly identical to the analogous calculation for the $E_8 \times E_8$ heterotic string, which is done in [46, section 2.2, section 2.3]; therefore we will be succinct and direct the reader there for the details.

Let $V \to BSpin(16) \times BSpin(16)$ be the direct sum of the tautological vector bundles on the two factors. The \mathbb{Z}_2 swapping action on $BSpin(16) \times BSpin(16)$ lifts to make Vinto a \mathbb{Z}_2 -equivariant vector bundle, so V descends to a vector bundle we will also call Vover $BG_{16,16}$. Since the action of \mathbb{Z}_2 is compatible with the spin structures on the two tautological bundles, $V \to BG_{16,16}$ is spin, so $w_1(V) = 0$ and $w_2(V) = 0$; and essentially by definition, $\frac{1}{2}p_1(V) = \frac{1}{2}p_1^L + \frac{1}{2}p_1^R$. Therefore just as for the other theories we studied, there is an isomorphism

$$\Omega_*^{\mathbb{G}_{16,16}} \stackrel{\cong}{\to} \Omega_*^{\text{String}}((BG_{16,16})^{V-32}).$$
(4.4)

This is the biggest difference between the computations for the $\text{Spin}(16) \times \text{Spin}(16)$ and $E_8 \times E_8$ theories: see [46, Lemma 2.2]. Much of the theory developed in [46, section 2] and in [114] and applied to the $E_8 \times E_8$ theory in *loc. cit.* can therefore be avoided for the $\text{Spin}(16) \times \text{Spin}(16)$ case; nevertheless, the calculation is pretty similar.

First we must establish the absence of p-torsion for primes p > 3. This is analogous to the other twisted string bordism computations in this paper, and we do not go into detail.

At p = 3, we follow [114, section 3.2]. First we need $H^*(BG_{16,16}; \mathbb{Z}_3)$; the Serre spectral sequence for \mathbb{Z}_3 cohomology and the fibration (4.1) collapses to an isomorphism

$$H^*(BG_{16,16};\mathbb{Z}_3) \xrightarrow{\cong} H^*(B(\operatorname{Spin}(16) \times \operatorname{Spin}(16));\mathbb{Z}_3)^{\mathbb{Z}_2}.$$
(4.5)

In the degrees relevant to us, $H^*(B\text{Spin}(16);\mathbb{Z}_3)$ is generated by the Pontrjagin classes p_1 and p_2 with no relations in degrees 11 and below, so we obtain the following additive basis for $H^*(BG_{16,16};\mathbb{Z}_3)$ in degrees 11 and below: 1, $p_1^L + p_1^R$, $(p_1^L)^2 + (p_1^R)^2$, $p_2^L = p_2^R$, and $p_1^L p_1^R$. Using this, we determine the \mathcal{A}^{tmf} -module structure on $H^*((BG_{16,6})^{V-32};\mathbb{Z}_3)$ using [114, Corollary 2.37]: if U denotes the Thom class, $\beta(U) = 0$ and $\mathcal{P}^1(U) = -U(p_1^L + p_1^R)$ (as



Figure 8. Left: the \mathcal{A}^{tmf} -module structure on $H^*((BG_{16,16})^{V-32};\mathbb{Z}_3)$ in low degrees; the pictured submodule contains all elements in degrees 11 and below. Right: the E_2 -page of the Adams spectral sequence computing $tmf_*((BG_{16,16})^{V-32})_3^{\wedge}$.

 $\frac{1}{2}x = -x$ in a \mathbb{Z}_3 -vector space). Using this and the Cartan formula, we find an \mathcal{A}^{tmf} -module isomorphism

$$H^*((BG_{16,16})^{V-32};\mathbb{Z}_3) \cong \mathbb{N}_3 \oplus \Sigma^8 N_3 \oplus \Sigma^8 N_3 \oplus P,$$
(4.6)

where N_3 is as in definition 3.54 and P is concentrated in degrees 12 and above, and will be irrelevant for us. We draw (4.6) in figure 8, left. Using the calculation of $\text{Ext}_{\mathcal{A}^{tmf}}(N_3)$ from figure 2, we can draw the E_2 -page of the Adams spectral sequence in figure 8, right; it collapses to show there is no 3-torsion in degrees 11 and below.

Finally p = 2. First, we need $H^*(BG_{16,16}; \mathbb{Z}_2)$; Evens' generalization [153] of a theorem of Nakaoka [154, Theorem 3.3] gives us the following additive basis for these cohomology groups in degrees 13 and below:

- classes of the form $c^L + c^R$, where c ranges over a basis of $H^*(BSpin(16); \mathbb{Z}_2)$ in degrees 13 and below;
- the classes $w_4^L w_4^R$, $w_6^L w_6^R$, $w_4^L w_k^R + w_k^L w_4^R$ for k = 6, 7, 8, and $(w_4^L)^2 w_4^R + w_4^L (w_4^2)^R$; and
- finally, we have classes of the form x^m , $w_4^L w_4^R x^m$, and $w_6^L w_6^R x^m$, where x is the generator of $H^1(B\mathbb{Z}_2;\mathbb{Z}_2)$, pulled back by the quotient $G_{16,16} \to \mathbb{Z}_2$ by the normal $\operatorname{Spin}(16) \times \operatorname{Spin}(16)$ subgroup.

Quillen's detection theorem [155, Proposition 3.1] computes the $\mathcal{A}(2)$ -action on these classes. Since V has vanishing w_1 and w_2 , but $w_4(V) = w_4^L + w_4^R$, $\operatorname{Sq}^1(U) = 0$, $\operatorname{Sq}^2(U) =$ 0, and $\operatorname{Sq}^4(U) = \operatorname{U}(w_4^L + w_4^R)$. Using this, we can obtain $\mathcal{A}(2)$ -module structure on $H^*((BG_{16,16})^{V-32};\mathbb{Z}_2)$ by direct computation with the Cartan formula similarly to [46, Proposition 2.41].

Proposition 4.7. Let \mathcal{M} be the quotient of $H^*((BG_{16,16})^{V-32}; \mathbb{Z}_2)$ by all elements in degrees 14 and above. Then \mathcal{M} is the direct sum of the following submodules.

- 1. M_1 , the summand containing U.
- 2. $M_2 := \widetilde{H}^*(\mathbb{RP}^\infty; \mathbb{Z}_2)$ (modulo elements in degrees 14 and above).



Figure 9. The $\mathcal{A}(2)$ -module structure on $H^*(B((\text{Spin}(16) \times \text{Spin}(16)) \rtimes \mathbb{Z}_2)^{V-32}; \mathbb{Z}_2)$ in low degrees. The figure includes all classes in degrees 13 and below. Here $\alpha := (w_4^L)^2 w_4^R + w_4^L (w_4^R)^2$.

- 3. M_3 , the summand containing $U((w_4^L)^2 + (w_4^R)^2)$.
- 4. M_4 , the summand containing $Uw_4^L w_4^R$.
- 5. M_5 , the summand containing $Uw_4^L w_4^R x$.
- 6. M_6 , the summand containing $U(w_4^L w_6^L + w_4^R w_6^R)$.
- 7. M_7 , the summand containing $U((w_4^L)^2 w_4^R + w_4^L (w_4^R)^2)$.
- 8. M_8 , the summand containing $U(w_4^L w_8^L + w_4^R w_8^R)$.
- 9. M_9 , the summand containing $U(w_4^L w_8^R + w_8^L w_4^R)$.

We draw this decomposition in figure 9.

The next step is to split off some of these summands in a manner similar to [46, Corollary 2.36]. Morally this is exactly the same simplification we used in theorem 3.78 and discussed further in remark 3.82, but the details are a little more complicated.

Definition 4.8. Let $\xi: B\mathbb{G}_{16,16'} \to BO$ be the tangential structure defined analogously to $B\mathbb{G}_{16,16}$, but with Spin in place of Spin(16).

Lemma 4.9. The map $\text{Spin}(16) \hookrightarrow \text{Spin}$ induces a map $\Omega_k^{\mathbb{G}_{16,16}} \to \Omega_k^{\mathbb{G}'_{16,16}}$ which is an isomorphism for $k \leq 14$.

This means that, for our string-theoretic applications, it does not matter whether we use $B\mathbb{G}_{16,16}$ or $B\mathbb{G}'_{16,16}$.

Proof. We want to show that the map $MT\mathbb{G}_{16,16} \to MT\mathbb{G}'_{16,16}$ of bordism spectra is an isomorphism on π_k for $k \leq 14$. By the Whitehead theorem we may equivalently use $H^k(-;\mathbb{Z})$, and by the Thom isomorphism, it suffices to show the map $B\mathbb{G}_{16,16} \to B\mathbb{G}'_{16,16}$ is an isomorphism on \mathbb{Z} -cohomology in degrees 14 and below. The cohomology rings of these spaces can be computed in two steps: first the Serre spectral sequence for the fibration $B(\operatorname{Spin}(16) \times \operatorname{Spin}(16)) \to BG_{16,16} \to B\mathbb{Z}_2$, then the Serre spectral sequence for the fibration $B^2\mathrm{U}(1) \to B\mathbb{G}_{16,16} \to BG_{16,16}$; and analogously for $B\mathbb{G}'_{16,16}$ with Spin in place of Spin(16). For each of these two steps, the map $\operatorname{Spin}(16) \to \operatorname{Spin}$ induces a map of Serre spectral sequences. and because $H^*(B\operatorname{Spin};\mathbb{Z}) \to H^*(B\operatorname{Spin}(16);\mathbb{Z})$ is an isomorphism in degrees 15 and below, we learn that at each of the two steps, the two spectral sequences are isomorphism on cohomology in degrees 14 and below. \Box

Proposition 4.10. There is a spectrum Q and a splitting

$$MT\mathbb{G}'_{16.16} \xrightarrow{\simeq} MTSpin \lor \mathcal{Q},$$
 (4.11)

such that the pullback map on cohomology corresponding to the projection $MT\mathbb{G}'_{16,16} \to \mathcal{Q}$ is a map

$$H^*(\mathcal{Q};\mathbb{Z}_2) \cong \mathcal{A} \otimes_{\mathcal{A}(2)} \mathcal{L} \longrightarrow H^*(MT\mathbb{G}'_{16,16};\mathbb{Z}_2) \cong \mathcal{A} \otimes_{\mathcal{A}(2)} H^*((BG_{16,16})^{V-32};\mathbb{Z}_2)$$
(4.12)

given by the inclusion of an $\mathcal{A}(2)$ -module $\mathcal{L} \hookrightarrow H^*((BG_{16,16})^{V-32}; \mathbb{Z}_2)$, followed by applying $\mathcal{A} \otimes_{\mathcal{A}(2)}$ -; the quotient of \mathcal{L} by all classes in degrees 14 and above is isomorphic to

$$M_2 \oplus M_3 \oplus M_4 \oplus M_5 \oplus M_7 \oplus M_8 \oplus M_9. \tag{4.13}$$

Proof. The idea is the same as [46, Corollary 2.36]: show that a spin structure induces a $\mathbb{G}'_{16,16}$ -structure, such that forgetting back down to BSpin recovers the original spin structure.

Any spin vector bundle $E \to M$ has a canonical $\mathbb{G}'_{16,16}$ -structure with a trivial double cover $M' := M \amalg M$, V^L equal to E on one copy of M inside M' and equal to 0 on the other copy of M, and V^R the image of V^L under the deck transformation, as $\frac{1}{2}p_1(V^L) + \frac{1}{2}p_1(V^R)$ (descended to M) is canonically identified with $\frac{1}{2}p_1(E)$. Composing with the forgetful map $B\mathbb{G}'_{16,16} \to B$ Spin gives a map BSpin $\to B$ Spin homotopy equivalent to the identity and therefore maps of spectra $MTSpin \to MT\mathbb{G}'_{16,16} \to MTSpin$, yielding the splitting as promised.

To see the statement on cohomology, one can look at the edge morphism in the Serre spectral sequence for $B^2U(1) \rightarrow B\mathbb{G}'_{16,16} \rightarrow BG'_{16,16}$.

As we already know spin bordism groups in the dimensions we need, we focus on computing $\pi_*(\mathcal{Q})_2^{\wedge}$. Because the cohomology of \mathcal{Q} is of the form $\mathcal{A} \otimes_{\mathcal{A}(2)} \mathcal{L}$, the change-of-rings theorem simplifies the Adams spectral sequence for \mathcal{Q} to the form

$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{A}(2)}^{s,t}(\mathcal{L}, \mathbb{Z}_2) \Longrightarrow \pi_{t-s}(\mathcal{Q})_2^{\wedge};$$
(4.14)

we will then add on the summands coming from Ω_*^{Spin} to obtain the groups in the theorem statement. The first thing we need is $\text{Ext}_{\mathcal{A}(2)}$ of M_2 , M_3 , M_4 , M_5 , M_7 , M_8 , and M_9 .

- 1. Davis-Mahowald [156, table 3.2] compute $Ext(M_2)$.
- 2. In degrees 14 and below, M_3 is isomorphic to $\Sigma^8 \mathcal{A}(2) \otimes_{\mathcal{A}(1)} \mathbb{Z}_2$ (meaning the quotients of these modules by their submodules of elements in degrees 15 and above are isomorphic). Therefore when $t s \leq 14$, there is an isomorphism

$$\operatorname{Ext}_{\mathcal{A}(2)}^{s,t}(M_3, \mathbb{Z}_2) \cong \operatorname{Ext}_{\mathcal{A}(2)}^{s,t}(\mathcal{A}(2) \otimes_{\mathcal{A}(1)} \mathbb{Z}_2, \mathbb{Z}_2),$$
(4.15a)

and the change-of-rings theorem (see, e.g., [110, section 4.5]) implies that in all degrees,

$$\operatorname{Ext}_{\mathcal{A}(2)}(\mathcal{A}(2) \otimes_{\mathcal{A}(1)} \mathbb{Z}_2, \mathbb{Z}_2) \cong \operatorname{Ext}_{\mathcal{A}(1)}(\mathbb{Z}_2, \mathbb{Z}_2).$$
(4.15b)

Liulevicius [157, Theorem 3] first calculated the algebra $\operatorname{Ext}_{\mathcal{A}(1)}(\mathbb{Z}_2,\mathbb{Z}_2)$.

- 3. As an $\text{Ext}(\mathbb{Z}_2)$ -module, $\text{Ext}(M_4) \cong \mathbb{Z}_2[h_0]$ with $h_0 \in \text{Ext}^{1,1}$ [46, (2.43)].
- 4. $\operatorname{Ext}(M_5)$ is computed in [46, figure 2].
- 5. Finally, for M_7 , M_8 , and M_9 , we only need to know their Ext groups in degrees 12 and below. For i = 7, 8, 9, there is a surjective map $M_i \to \Sigma^{12}\mathbb{Z}_2$ whose kernel is concentrated in degrees 14 and above, so (e.g. using the long exact sequence in Ext associated to a short exact sequence of $\mathcal{A}(2)$ -modules [110, section 4.6]) for $t - s \leq 12$, Ext of each of these modules is isomorphic to $\text{Ext}(\Sigma^{12}\mathbb{Z}_2)$, which was computed by May (unpublished) and Shimada-Iwai [158, section 8].

These assemble into a description of the E_2 -page of (4.14) (compare [46, Proposition 2.46]).

Proposition 4.16. The E_2 -page of the Adams spectral sequence for \mathcal{Q} in degrees $t - s \leq 12$ is as displayed in figure 10. In this range, the E_2 -page is generated as an $\operatorname{Ext}_{\mathcal{A}(2)}(\mathbb{Z}_2)$ -module by ten elements:

- $p_1 \in \text{Ext}^{0,1}, p_3 \in \text{Ext}^{0,3}, p_7 \in \text{Ext}^{0,7}, and b \in \text{Ext}^{2,10}, coming from \text{Ext}(M_2);$
- $a_1 \in \text{Ext}^{0,8}$ and $a_3 \in \text{Ext}^{3,15}$, coming from $\text{Ext}(M_3)$.
- $a_2 \in \operatorname{Ext}^{0,8}$, coming from $\operatorname{Ext}(M_4)$.
- $c \in \text{Ext}^{0,9}$ and $d \in \text{Ext}^{0,11}$, coming from $\text{Ext}(M_5)$.
- $e \in \operatorname{Ext}^{0,12}$, coming from $\operatorname{Ext}(M_7)$.
- $f \in \operatorname{Ext}^{0,12}$, coming from $\operatorname{Ext}(M_8)$.
- $g \in \operatorname{Ext}^{0,12}$, coming from $\operatorname{Ext}(M_9)$.

The next step is to evaluate the differentials. Unlike the other Adams spectral sequences we considered in this paper, there are several differentials to address, even after using that differentials commute with the action of h_0 , h_1 , and h_2 :

- d_2 on a_1 , a_2 , a_3 , c, d, e, f, and g,
- d_3 on a_1 , a_2 , and a_3 , and
- d_4 , d_5 , and d_6 on e, f, and g.

The argument is nearly the same as in [46, Lemmas 2.47, 2.50, and 2.56].



Figure 10. The E_2 -page of the Adams spectral sequence computing $tmf_*((BG_{16,16})^{V-32})^{\wedge}_2$. In lemma 4.17 we show that $d_2(a_2) = h_2^2 p_1$ and that many other differentials vanish. We do not know the values of $d_2(c)$ or $d_2(h_1c)$, which is why those differentials are denoted with dotted lines.

Lemma 4.17. $d_2(a_2) = h_2^2 p_1$, and all differentials vanish on a_1 , a_3 , e, f, and g.

Proof. If ξ' denotes the tangential structure identical to $\mathbb{G}_{16,16}$ except with $K(\mathbb{Z}, 4)$ in place of BSpin(16), then the class $\frac{1}{2}p_1$, interpreted as a map BSpin(16) $\rightarrow K(\mathbb{Z}, 4)$, induces a map of tangential structures from $\mathbb{G}_{16,16}$ -structure to ξ' -structure, hence also a map of Thom spectra, hence a map of Adams spectral sequences. The E_2 -page for $\Omega_*^{\xi'}$ is computed in [46, figure 3] in the range $t - s \leq 12$, and looks very similar to our E_2 -page in figure 10; using the comparison map between these two spectral sequences, we conclude the differentials in the lemma statement.

The comparison map would also tell us $d_2(c)$, except that the fate of this differential in ξ' -bordism is not known.

Lastly, we address the class $d \in E_2^{0,11}$. Since d has topological degree 11, its fate affects the size of $\Omega_{11}^{\mathbb{G}_{16,16}}$, hence the possible anomaly theories for the $\text{Spin}(16) \times \text{Spin}(16)$ theory.

Definition 4.18. Embedding each $S^k \hookrightarrow \mathbb{R}^{k+1}$ and using the notation $(\vec{x}, \vec{y}, \vec{z})$ for a vector in $\mathbb{R}^5 \times \mathbb{R}^5 \times \mathbb{R}^4$, let \mathbb{Z}_2 act on $S^4 \times S^4 \times S^3$ by the involution

$$(\vec{x}, \vec{y}, \vec{z}) \longmapsto (-\vec{y}, -\vec{x}, -\vec{z}). \tag{4.19}$$

This action is free on $S^4 \times S^4 \times S^3$; let Y_{11} denote the quotient.

 Y_{11} is an $(S^4 \times S^4)$ -bundle over \mathbb{RP}^3 .

Lemma 4.20. Y_{11} has a spin structure.

Proof. To prove this, we will stably split the tangent bundle of Y_{11} . This is a standard technique; for more examples from a similar perspective, see [41, section 5.2, section 5.5.2], [51, Examples 14.51 and 14.54; Lemma 14.56; Propositions 14.74, 14.83, and 14.101], and [46, Lemma 2.68].

Recall that, since the normal bundle to $S^k \hookrightarrow \mathbb{R}^{k+1}$ is trivialized by the unit outward normal vector field \vec{v} , there is an isomorphism $\phi: TS^k \oplus \mathbb{R} \cong \mathbb{R}^{k+1}$; since \vec{v} is O(k+1)-invariant, ϕ promotes to an isomorphism of O(k+1)-equivariant vector bundles, where O(k+1) acts trivially on the normal bundle and via the defining representation on \mathbb{R}^{k+1} .

Applying this thrice, we have an isomorphism of vector bundles

$$T(S^4 \times S^4 \times S^3) \oplus \underline{\mathbb{R}}^3 \xrightarrow{\cong} \underline{\mathbb{R}}^5 \oplus \underline{\mathbb{R}}^5 \oplus \underline{\mathbb{R}}^4.$$

$$(4.21)$$

The \mathbb{Z}_2 -action on $S^4 \times S^4 \times S^3$ we used to define in Y_{11} in definition 4.18 extends to a linear action on $\mathbb{R}^5 \times \mathbb{R}^5 \times \mathbb{R}^4$, upgrading (4.21) to an isomorphism of \mathbb{Z}_2 -equivariant vector bundles. In a little more detail:

- \mathbb{Z}_2 acts on $T(S^4 \times S^4 \times S^3)$ as the derivative of the involution (4.19).
- \mathbb{Z}_2 acts on $\mathbb{R}^5 \oplus \mathbb{R}^5 \oplus \mathbb{R}^4$ as the \mathbb{Z}_2 -representation described by the same formula (4.19).
- \mathbb{Z}_2 acts on the normal \mathbb{R}^3 by inverting and swapping the first two coordinates, and inverting the third: $(x, y, z) \mapsto (-y, -x, -z)$.²⁴

The isomorphism (4.21) of \mathbb{Z}_2 -equivariant vector bundles descends through the quotient by \mathbb{Z}_2 to an isomorphism of vector bundles on Y_{11} ; trivial bundles made equivariant by a \mathbb{Z}_2 -representation descend to vector bundles associated to that representation and the principal \mathbb{Z}_2 -bundle $\pi: S^4 \times S^4 \times S^3 \to Y_{11}$.

In particular, if $\sigma_{\pi} \to Y_{11}$ denotes the line bundle associated to π and the sign representation σ of \mathbb{Z}_2 on \mathbb{R} and \mathbb{R} denotes the trivial representation, then the \mathbb{Z}_2 -representation $(x, y) \mapsto (-y, -x)$ on \mathbb{R}^2 is isomorphic to $\sigma \oplus \mathbb{R}$. Using this, we obtain an isomorphism of vector bundles

$$TY_{11} \oplus \sigma_{\pi} \oplus \underline{\mathbb{R}}^2 \xrightarrow{\cong} \sigma_{\pi}^{\oplus 5} \oplus \underline{\mathbb{R}}^5 \oplus \sigma_{\pi}^{\oplus 4}.$$
(4.22a)

Therefore we have an isomorphism of *virtual* vector bundles

$$TY_{11} \xrightarrow{\cong}_{\text{virt.}} \sigma_{\pi}^{\oplus 8} + \underline{\mathbb{R}}^3.$$
 (4.22b)

For any vector bundle $V, V^{\oplus 4}$ is spin, as can be verified with the Whitney sum formula, and the existence of a spin structure is an invariant of the virtual equivalence class of a vector bundle, so we can conclude.

Proposition 4.23. Y_{11} admits a $\mathbb{G}_{16,16}$ -structure such that the bordism invariant

$$\int_{Y_{11}} w_4^L w_4^R x^3 = 1. \tag{4.24}$$

²⁴In particular, unlike most of the standard examples of the stable splitting technique, the normal bundle is *not* equivariantly trivial. This is because the image of the \mathbb{Z}_2 -representation in O(14) is not contained in the subgroup $O(5) \times O(5) \times O(4)$.

Proof. The following data describes a $\mathbb{G}_{16,16}$ -structure on Y_{11} : identify $S^4 = \mathbb{HP}^1$ and consider the tautological quaternionic line bundle $L \to \mathbb{HP}^1$ on the first S^4 factor, and $L^* := \operatorname{Hom}_{\mathbb{H}}(L, \underline{\mathbb{H}})$ on the second S^4 factor. These have associated $\operatorname{Sp}(1) = \operatorname{Spin}(3)$ bundles; inflate via $\operatorname{Spin}(3) \hookrightarrow \operatorname{Spin}(16)$ to obtain a $(\operatorname{Spin}(16) \times \operatorname{Spin}(16))$ -bundle on $S^4 \times S^4 \times S^3$. The two $\operatorname{Spin}(16)$ -bundles are switched when one applies the involution (4.19), so on the quotient Y_{11} , we obtain a principal $G_{16,16}$ -bundle $P \to Y_{11}$.

To verify the claim in the first sentence of our proof, we need to check that a spin structure on Y_{11} and the principal $G_{16,16}$ -bundle $P \to Y_{11}$ satisfy the Green-Schwarz condition $\frac{1}{2}p_1(TY_{11}) + \frac{1}{2}p_1(V^L) + \frac{1}{2}p_1(V^R) = 0$. In fact, the two parts of this expression vanish separately.

• In (4.22b), we learned that TY_{11} is virtually equivalent to $\sigma_{\pi}^{\oplus 8} \oplus \mathbb{R}^3$. This bundle turns out to admit a string structure, meaning $\frac{1}{2}p_1(TY_{11}) = 0$. It suffices to prove that $\sigma^{\oplus 8} \to B\mathbb{Z}_2$ admits a string structure, where σ is the tautological line bundle. To see this, recall that $\sigma^{\oplus 4}$ (like the sum of 4 copies of any vector bundle) is spin, so $\sigma^{\oplus 8} \cong \sigma^{\oplus 4} \oplus \sigma^{\oplus 4}$ factors $\sigma^{\oplus 8}$ as the direct sum of two spin vector bundles. Then use the Whitney sum formula for $\frac{1}{2}p_1$ of a direct sum of spin vector bundles [46, Lemma 1.6] to conclude that in $H^4(B\mathbb{Z}_2;\mathbb{Z})$,

$$\frac{1}{2}p_1(\sigma^{\oplus 8}) = 2 \cdot \frac{1}{2}p_1(\sigma^{\oplus 4}).$$
(4.25)

Maschke's theorem implies that for $k \ge 1$, multiplication by 2 kills all elements in $H^k(B\mathbb{Z}_2;\mathbb{Z})$, so $\frac{1}{2}p_1(\sigma^{\oplus 8}) = 0$.

• The bundles L and L^* over S^4 have inverse values of p_1 , hence also of $\frac{1}{2}p_1$ (since $H^4(S^4;\mathbb{Z})$ is torsion-free, the latter follows from the former). Therefore when we descend from $S^4 \times S^4 \times S^3$ to Y_{11} , the class $\frac{1}{2}p_1(V^L) + \frac{1}{2}p_1(V^R)$ is 0.

Finally, we need to verify $\int_{Y_{11}} w_4^L w_4^R x^3 = 1$. Since $H^{11}(Y_{11}; \mathbb{Z}_2) \cong \mathbb{Z}_2$, it suffices to show that the pullback of $w_4^L w_4^R x^3 \in H^{11}(BG_{16,16}; \mathbb{Z}_2)$ along the classifying map $f_P: Y \to BG_{16,16}$ for $P \to Y_{11}$ is nonzero. To do this, first factor f_P into the following diagram of three fibrations:



Here \mathcal{X} is the $S^4 \times S^4$ -bundle over $B\mathbb{Z}_2$ defined analogously to Y_{11} but using $S^{\infty} = E\mathbb{Z}_2$ instead of S^3 . The map $j: S^4 \times S^4 \to B\text{Spin}(16) \times B\text{Spin}(16)$ is the map classifying L and L^* .

The diagram (4.26) induces maps between the Serre spectral sequences of the three fibrations; using it, one can compute the pullback of $w_4^L w_4^R x^3$ to Y_{11} and see that it is nonzero, as promised.

Corollary 4.27. In the Adams spectral sequence in figure 10, d survives to the E_{∞} -page; in particular, $d_2(d) = 0$.

Proof. We reuse the strategy from lemma 3.59: since d is in filtration 0, it corresponds to some characteristic class $c \in H^{11}(BG_{16,16}; \mathbb{Z}_2)$, and d survives to the E_{∞} -page if and only if there is some closed 11-dimensional $\mathbb{G}_{16,16}$ -manifold M such that $\int_M c = 1$. By inspection of figure 9, $c = w_4^L w_4^R x^3$, so by proposition 4.23 we can take $M = Y_{11}$.

The last step in this calculation is to address extensions. The argument is nearly identical to [46, Lemma 2.59 and Proposition 2.60], though one now has the extra classes $h_0^k a_1$ for $k \ge 0$, $h_1 a_1$, and $h_1^2 a_1$ in degrees 8, 9, and 10 respectively which were not present in the $E_8 \times E_8$ spectral sequence. Fortunately, this new ambiguity is fully resolved by applying the " $2\eta = 0$ trick" to classes of the form $h_1 x$ in standard ways, for example as in [159, Corollary F.16(2)], [160, (5.47)], [51, Lemmas 14.29 and 14.33], and [46, Lemma 2.59], and one learns that there are no hidden extensions in degrees 10 and below. Unfortunately, just like in the $E_8 \times E_8$ case [46, Theorem 2.62], we have not ruled out the possibility of a hidden extension in $\Omega_{11}^{\mathbb{G}_{16,16}}$.

The generators described in [46, section 2.2.1, section 2.2.2] for Ω_*^{chet} pull back to generate most of the corresponding $\mathbb{G}_{16,16}$ bordism groups: the difference between a $\mathbb{G}_{16,16}$ -structure and a ξ^{het} -structure is that in the latter, Spin(16) is replaced by E_8 , so to support our claim that the generators there pull back to generators of $\mathbb{G}_{16,16}$ -bordism, we must argue that the $(E_8 \times E_8) \rtimes \mathbb{Z}_2$ -bundles are induced from $G_{16,16}$ -bundles. As usual we may replace BE_8 with $K(\mathbb{Z}, 4)$, so this amounts to checking that for the generating manifolds in [46, section 2.2.1, section 2.2.2], the degree-4 classes entering the Green-Schwarz mechanism can be written as $\frac{1}{2}p_1(V)$ for some rank-16 spin vector bundle V. By adding trivial summands, we may use lower-rank spin vector bundles.

By inspection of the list of generators in [46, section 2.2.1, section 2.2.2], it suffices to show this for \mathbb{HP}^2 : the rest of the list of generators there either have degree-4 classes equal to 0, have their ξ^{het} -structure induced from a spin structure (so that we may use the tangent bundle to define the $\mathbb{G}_{16,16}$ -structure as in the proof of proposition 4.10), or are products of manifolds otherwise accounted for. For \mathbb{HP}^2 , the degree-4 classes come from the tautological quaternionic line bundle, hence define a $\mathbb{G}_{16,16}$ -structure.

Thus the list of generators in [46, section 2.2.1, section 2.2.2] accounts for most of the generators of the $\mathbb{G}_{16,16}$ -bordism groups we have computed. A few manifolds are as yet unaccounted for.

- 1. There is an 8-dimensional $\mathbb{G}_{16,16}$ -manifold Y_8 generating a \mathbb{Z} and whose image in the Adams E_{∞} -page is a_1 . The \mathbb{Z}_2 summands lifting h_1a_1 and $h_1^2a_1$ are also unaccounted for, and can be generated by $Y_8 \times S_{nb}^1$, resp. $Y_8 \times S_{nb}^1 \times S_{nb}^1$.
- 2. Depending on the fate of $d_2(c)$, there may be a \mathbb{Z}_2 summand in $\Omega_9^{\mathbb{G}_{16,16}}$ whose generator lifts the class $c \in E_{\infty}^{9,0}$. In [46, section 2.2.1] no generator was provided and we also do not know what manifold this would be.
- 3. A generator lifting the class $d \in E_{\infty}^{0,11}$ was left as an open question in [46, section 2.2.1(11)]. Thanks to proposition 4.23, we can choose Y_{11} .

Proposition 4.28. Let $V \to S^8$ be the rank-16 spin vector bundle whose classifying map is either generator of $[S^8, BSpin(16)] = \pi_8(BSpin(16)) \cong \mathbb{Z}$ (by Bott periodicity), and let $P \to S^8$ be the $G_{16,16}$ -bundle induced by $V^L := V$, $V^R = 0$, and the trivial \mathbb{Z}_2 -bundle. Then (S^8, P) admits a $\mathbb{G}_{16,16}$ -structure, and for any such structure, its $\mathbb{G}_{16,16}$ -bordism class is linearly independent from the classes of \mathbb{HP}^2 , B, $\mathbb{RP}^7 \times S^1_{nb}$, and X_8 described in [46, section 2.2.1(8), section 2.2.2], and is not a multiple of any other class.

Thus (S^8, P) is the generator lifting $a_1 \in E_{\infty}^{0,8}$.

Proof. It suffices to find a bordism invariant $\psi: \Omega_8^{\mathbb{G}_{16,16}} \to \mathbb{Z}^m$ for some m such that the value of ψ on (S^8, P) is linearly independent from the values on the other generators, and also not a multiple of any other element of \mathbb{Z}^m . For X_8 and $\mathbb{RP}^7 \times S_{nb}^1$, this will be vacuously true, because the $\mathbb{G}_{16,16}$ -bordism classes of these manifolds are torsion, so we focus on the two \mathbb{HP}^2 s and the Bott manifold described in [46, section 2.2.1(8)].

Let m = 2 and ψ be given by the two Z-valued invariants $\int p_2(M)$ and $\int (p_2(V^L) + p_2(V^R))$. The latter is a priori an invariant of manifolds with a Spin(16) × Spin(16)-bundle, but it survives the Serre spectral sequence to define an invariant of $G_{16,16}$ -bundles and therefore of $\mathbb{G}_{16,16}$ -manifolds.

For the $\mathbb{G}_{16,16}$ -structure on the Bott manifold and both $\mathbb{G}_{16,16}$ -structures on \mathbb{HP}^2 specified in [46, section 2.2.1(8)], $\int p_2(M) \neq 0$. However,

$$\int_{S^8} p_2(S^8) = 0 \tag{4.29a}$$

$$\int_{S^8} (p_2(V^L) + p_2(V^R)) = \int_{S^8} p_2(V) = \pm 1, \qquad (4.29b)$$

the former because TS^8 is stably trivial and the latter somewhat tautologically from the definition of V. Therefore $\psi(S^8, P)$ is linearly independent from ψ evaluated on the other bordism classes we have considered. Finally, we know that the bordism class of (S^8, P) is not a multiple of some other class because (4.29b) is ± 1 , and if the class of (S^1, P) were a multiple, the values of all \mathbb{Z} -valued bordism invariants on it would be divisible by some natural number greater than 1.

Thus we have found generators for all classes except for c and h_1c , which may or may not be trivial, depending on the value of an Adams differential.

4.2 Cancelling the anomaly

Now that we have the generators of $\Omega_{11}^{\mathbb{G}_{16,16}}$ in hand, we proceed to calculate the partition function of the anomaly theory on these generators and show that it is trivial. We are able to do this without knowing the isomorphism type of $\Omega_{11}^{\mathbb{G}_{16,16}}$, similarly to Freed-Hopkins' approach in [41].

Theorem 4.30. Let α denote the anomaly field theory for the Spin(16)² heterotic string on $\mathbb{G}_{16,16}$ -manifolds. Then α is isomorphic to the trivial theory.

Proof. Recall that $\alpha \cong \alpha_f \otimes \alpha_{X_8}$, where α_f is the anomaly of the fermionic fields and α_{X_8} is the anomaly coming from the term $-\int B_2 \wedge X_8$ (2.17) that the Green-Schwarz mechanism adds to the action, as we discussed in section 2.

We will calculate α on a generating set for $\Omega_{11}^{\mathbb{G}_{16,16}}$. Based on [51] and the above discussion, the two generators are

$$B \times \mathbb{RP}^3$$
 and Y_{11} (4.31)

with $\mathbb{G}_{16,16}$ -structures described in the previous subsection, corresponding physically to turning on appropriate gauge bundles. Here *B* is a *Bott manifold*, i.e. a closed spin 8-manifold satisfying $\widehat{A}(B) = 1$, and indeed any choice of *B* that admits a string structure may be used in this computation. We will use the Bott manifold constructed by Freed-Hopkins in [41, section 5.3]; those authors show $\frac{1}{2}p_1(B) = 0$, so *B* is string, and that $p_2 = -1440b$, where *b* is a generator of $H^8(B;\mathbb{Z}) \cong \mathbb{Z}$. Any other Bott manifold is cobordant to this one, so we will not botter studying them all.

The first generator we evaluate α on is $B \times \mathbb{RP}^3$. As discussed in [46, section 2.2.1], this generator has $G_{16,16}$ -bundle induced from the principal $\mathbb{Z}/2$ -bundle $S^3 \times B \to \mathbb{RP}^3 \times B$ and any inclusion $\mathbb{Z}_2 \hookrightarrow G_{16,16}$ complementary to the normal Spin(16)² subgroup. From a physics point of view, this means the Spin(16) gauge bundles are trivial: the \mathbb{Z}_2 symmetry switches two copies of the trivial bundle. This implies $X_8 = 0$, so α_{X_8} is trivial. For α_f , we must calculate the η -invariants of the spinor bundles associated to the gauge bundles. We will first dimensionally reduce our theory on B, to obtain a 2d effective theory, and study the corresponding anomaly, which is the dimensional reduction of α_f , on \mathbb{RP}^3 . As the defining property of a Bott manifold is that the Dirac index is 1, and the gauge bundle is switched off in our example, the 2d spectrum is identical to the ten-dimensional one, so showing that the anomaly on \mathbb{RP}^3 is trivial will imply $\alpha_f (B \times \mathbb{RP}^3) = 1$.

We need to know the gauge bundle on \mathbb{RP}^3 ,²⁵ but because the gauge bundle is trivial on B, we can describe the $G_{16,16}$ -bundle on \mathbb{RP}^3 as induced from the \mathbb{Z}_2 -bundle $S^3 \to \mathbb{RP}^3$. Thus we should see how the $G_{16,16}$ -representations describing the fermions branch when we restrict to \mathbb{Z}_2 .

- The 10d fermions in the (128, 1) ⊕ (1, 128) give a total of 128 2d fermions transforming as singlets of the swap, and another 128 transforming in the sign representation.
- For the 10d fermions in the (16, 16), the swap is implemented via a matrix with sixteen blocks each having eigenvalues ±1, again giving 128 fermions in each of the trivial and sign representations of the swap Z₂. Since the 10d fermions have opposite chirality to those in the previous point, the resulting 2d fermions also come in opposite chirality.

With these matter assignments, we obtain a total of 128 \mathbb{Z}_2 charged fermions of each chirality, which collectively are anomaly-free (and therefore, gravitational anomalies cancel). Therefore, there is no anomaly under the swap on any background, such as \mathbb{RP}^3 : the η -invariants all cancel out. Thus $\alpha_f(B \times \mathbb{RP}^3)$ vanishes and the overall anomaly $\alpha_f \otimes \alpha_{X_8}$ vanishes on $B \times \mathbb{RP}^3$.

For Y_{11} , which is an $(S^4 \times S^4)$ -bundle over \mathbb{RP}^3 , we perform a twisted compactification on $S^4 \times S^4$ and study the anomaly of the resulting 2d theory on \mathbb{RP}^3 . Because Y_{11} is not

 $^{^{25}}$ In general keeping track of tangential structures on dimensional reductions can be complicated (see, e.g., [161, section 9], but because *B* has a string structure and the tangential structure of the theory is a twisted string structure, we do not need to worry about this detail.
a product, we must take a little more care with this procedure, but it is not so difficult to show that the assignment from a string 3-manifold N with principal $\mathbb{Z}/2$ -bundle $P \to N$ to the manifold

$$\kappa(N) := (S^4 \times S^4) \times_{\mathbb{Z}_2} N, \tag{4.32}$$

where the two copies of S^4 are given the same \mathbb{Z}_2 -action and Spin(16)-bundles as we used in the construction of Y_{11} , produces a $\mathbb{G}_{16,16}$ -manifold for all N and is compatible with bordism, allowing κ to define a functor of bordism categories and therefore a twisted compactification as promised.

The covering S^4 has $\text{Spin}(16)^2$ bundles characterized by a second Chern class

$$c_2^{\text{SO}(16),i} = (-1)^i (b_1 + b_2), \tag{4.33}$$

where b_1, b_2 are the volume forms of both S^4 factors. Now, rather than explicitly computing the dimensional reductions of α_f and α_{X_8} on \mathbb{RP}^3 , we take advantage of the fact that α is a deformation invariant, so we may deform our 2d theory into something where the value of the anomaly on \mathbb{RP}^3 is more obviously trivial.²⁶ Specifically, we can take a limit in moduli space where the instantons become singular and pointlike, turning into a non-supersymmetric version of the heterotic NS5-brane; as explained in section 3.4.2, the resulting theory becomes symmetric between the two Spin(16) factors, implying that, just like in the supersymmetric heterotic string theories, small instantons of both gauge factors are identified. After deforming in this way the gauge bundle on both Spin(16) factors, we are left with a single pointlike NS5 and a single anti-NS5 in each sphere, which annihilate, leading to a trivial and therefore anomaly-free configuration for the compactified theory, and implying that $\alpha(Y_{11}) = 1$.

As a bonus, we can answer a question of [46], giving a bordism-theoretic argument for the analogous anomaly cancellation question for the $E_8 \times E_8$ heterotic string. This anomaly cancellation result was first established by Tachikawa-Yamashita in [42] by a different argument.

Corollary 4.34. The anomaly field theory α for the $E_8 \times E_8$ heterotic string theory taking into account the \mathbb{Z}_2 swap symmetry is trivial.

Proof. The argument for Y_{11} also works in the supersymmetric $E_8 \times E_8$ theory, since the instantons may also be embedded in E_8 . In the supersymmetric case, the pointlike limit of the instanton is the ordinary, supersymmetric heterotic NS5-brane, as illustrated in [150], so the E_8^2 anomaly vanishes on Y_{11} .

For $B \times \mathbb{RP}^3$, dimensional reduction leads to 248 singlet and 248 fermions (from the E_8 adjoints) charged under the sign representation. Since the relevant anomaly is controlled by

$$\Omega_3^{\text{Spin}}(B\mathbb{Z}_2) = \mathbb{Z}_8,\tag{4.35}$$

 $^{^{26}}$ The SO(16) × SO(16) string is non-supersymmetric, and therefore the deformations we have just outlined in the previous paragraph may be obstructed dynamically; for instance, there may be a potential obstructing the small instanton limit. However, since we only wish to compute the anomaly, we may ignore such effects; the only ingredient we really need is the fact, proven in section 3.4.2, that in the small instanton limit the anomaly becomes symmetric between both Spin(16) factors.

and 248 is a multiple of 8, we conclude there is no swap anomaly either. Finally, we already know that gravitational anomalies cancel in $B \times \mathbb{RP}^3$, since if we forget about the swap this is just an ordinary string background.

In summary, we have shown that anomalies vanish under both generators of the swap bordism group, both for Spin(16)² and $E_8 \times E_8$. The supersymmetric case is covered by the worldsheet analysis in [42], which takes into account twists including the swap we just discussed. Thus, we recover a special case of the general anomaly cancellation result there. On the other hand, our approach covers the non-supersymmetric SO(16)² case (for the case of geometric target spaces only).

5 Conclusions

Our world is non-supersymmetric, and that fact alone means that non-supersymmetric corners of the string landscape warrant much more attention than they have received so far, both as a source of interesting backgrounds that might connect more directly to our universe, as well as a new trove of data to check and refine Swampland constraints. In this paper we have moved a bit in this direction by computing the bordism groups and anomalies associated to twisted string structures in the three known non-supersymmetric, tachyon free string models in ten dimensions. The results we obtained are summarized in table 1 for the Sugimoto and $\text{Spin}(16)^2$ groups; for the more complicated Sagnotti 0'B model, we were just able to show that there is a potential \mathbb{Z}_2 anomaly.

From the results of the table, it is clear that both $\text{Spin}(16)^2$ and Sugimoto models are free of global anomalies. One might have expected this from the fact that they have a consistent worldsheet description. However, there can be non-perturbative consistency conditions that are not automatically satisfied by the existence of a consistent worldsheet at one-loop, see for instance [162], where a K-theory tadpole is not detected by the closed string sector. It would be very interesting to determine in full generality whether existence of a consistent worldsheet is sufficient to guarantee consistence of the target spacetime. Although we have not settled the question of consistency in the Sagnotti string, we expect that it is also free of anomalies; for instance, upon circle compactification, it can be related to a "hybrid" type I' setup involving an $O8^+$ plane and an $\overline{O8^-}$, both of which are individually consistent [139].

Perhaps the more interesting result of our work is table 1 itself, listing the bordism groups of the $\text{Spin}(16)^2$ and Sugimoto theories. An obvious follow-up to this paper is to use the Cobordism Conjecture [48] together with the groups in table 1 to predict new, non-supersymmetric objects in the non-supersymmetric string theories, similarly to what has been done in type II in [51]. While it is natural to expect these new branes to be non-supersymmetric, it may be worthwhile to pursue this direction in more detail.

One subtlety that must be kept in mind, when considering our results, is that we did not necessarily use the correct global form of the gauge group in our calculations. With the exception of the $SO(16)^2 \rtimes \mathbb{Z}_2$, we focused on simply connected versions of all the groups, which immensely simplified the calculations. Since any bundle before taking a quotient is still an allowed bundle after taking the quotient, our results show that a very large class of

k	$\Omega_k^{\text{String}-\text{Spin}(16)^2}$	$\Omega_k^{\mathbb{G}_{16,16}}$	$\Omega_k^{\mathrm{String-Sp}(16)}$	$\Omega_k^{ m String-SU(32)\langle c_3 angle}$
0	Z	\mathbb{Z}	Z	\mathbb{Z}
1	\mathbb{Z}_2	\mathbb{Z}_2^2	\mathbb{Z}_2	\mathbb{Z}_2
2	\mathbb{Z}_2	\mathbb{Z}_2^2	\mathbb{Z}_2	\mathbb{Z}_2
3	0	\mathbb{Z}_8	0	0
4	\mathbb{Z}^2	$\mathbb{Z}\oplus\mathbb{Z}_2$	\mathbb{Z}	\mathbb{Z}
5	0	0	\mathbb{Z}_2	\mathbb{Z}_2
6	0	\mathbb{Z}_2	\mathbb{Z}_2	0 or \mathbb{Z}_2
7	0	\mathbb{Z}_{16}	\mathbb{Z}_4	$\mathbb{Z}_2 \text{ or } \mathbb{Z}_4 \oplus \mathbb{Z}_2$
8	\mathbb{Z}^6	$\mathbb{Z}^3\oplus\mathbb{Z}_2^i$	$\mathbb{Z}^3\oplus\mathbb{Z}_2$	$\mathbb{Z}^3\oplus\mathbb{Z}_2$ or $\mathbb{Z}^3\oplus\mathbb{Z}_2^2$
9	\mathbb{Z}_2^5	\mathbb{Z}_2^j	\mathbb{Z}_2^3	\mathbb{Z}_2^3
10	\mathbb{Z}_2^7	\mathbb{Z}_2^k	\mathbb{Z}_2^3	$\mathbb{Z}\oplus\mathbb{Z}_2^2$ or $\mathbb{Z}\oplus\mathbb{Z}_2^3$
11	0	A	0	0 or \mathbb{Z}_2

Table 1. Twisted string bordism groups computed in this paper for the $\text{Spin}(16)^2$ theory with and without including the swap (second and third columns), for the Sugimoto string (fourth column), and for the Sagnotti string (fifth column). In the second column, i, j, k are unknown integers, and A is an abelian group of order 64 (see section 4 for details). In the fifth column, there are ambiguities due to undetermined differentials in the Adams spectral sequence; see section 3.3.2 for details. In some cases, the bordism group vanishes in degree 11, which automatically implies the corresponding theory has no anomalies; we also show the anomaly can be trivialized for the \mathbb{Z}_2 outer automorphism of the $\text{Spin}(16)^2$ string, even though the bordism group is nonzero. The results in this table can be further used to classify bordism classes and predict new solitonic objects in these non-supersymmetric string theories following [48, 51].

allowed bundles in the Sugimoto and $SO(16)^2$ theories are anomaly free,²⁷ but particularly in the Sugimoto case there may be more bundles to check if the gauge group is actually $Sp(16)/\mathbb{Z}_2$. In the type I theory, we know the group is $Spin(32)/\mathbb{Z}_2$ and not just SO(32)due to the presence of K-theory solitons transforming in a spinorial representation [163]. In the Sugimoto theory, the relevant K-theory is symplectic, and there do not seem to be any such solitonic particles [4], suggesting that the group might actually be $Sp(16)/\mathbb{Z}_2$. It would be interesting to elucidate this point and figure out whether there really are any global anomalies beyond those studied here.

Another result of our paper is a series of arguments and checks that in any heterotic string theory, the Bianchi identity must hold at the level of integer coefficients. Furthermore, satisfying the Bianchi identity even at the level of integer coefficients is not enough to guarantee consistency of the string background; there is also a consistency condition (tadpole) that is detected by $H^3(M; \mathbb{R})$. The general consistency condition is of course that the anomaly of probe strings vanishes; more generally, it is natural to expect that all consistency conditions (tadpoles) of any quantum gravity background come from consistency of probe branes in said background.

Another limitation of our study is that, by following a (super)gravity approach, we must restrict to studying anomalies on *smooth* backgrounds. String theories, both with and without

 $^{^{27}}$ The equivalence discussed at the beginning of subsection 3.4.2 shows that anomalies vanish for SO(16)² even when the correct global form is taken into account.

spacetime supersymmetry, make sense on much larger classes of backgrounds that do not admit a geometric description, such as orbifolds, and which are only analyzed from a worldsheet perspective. These cases are not covered by our analysis. Using modular invariance one can show that the Green-Schwarz mechanism always cancels local anomalies in any consistent worldsheet background [71, 164], with or without spacetime supersymmetry. The question of whether global anomalies also cancel in these non-geometric backgrounds was addressed in [42, 45], where *all* global anomalies are shown to cancel for all gauge groups and dimensions in the ordinary supersymmetric heterotic string theories. This remarkable result rests on the validity of the Segal-Stolz-Teichner conjecture [165], which connects deformation classes of worldsheet theories (or, more generally, two-dimensional (0, 1) supersymmetric QFTs) to the spectrum of (connective) *topological modular forms* (TMFs) [117, 166] (see also [167]). The physical interpretation of this more refined generalized cohomology theory is related to "going up and down RG flows" [168], and it includes the familiar string bordism deformations of the target space manifold of a sigma model as well as more exotic, "non-geometric" deformations.

To construct an ordinary, spacetime-supersymmetric heterotic model, all that one needs is a (0,1) SQFT. Such a QFT always has a notion of a right-moving worldsheet fermion number F_R^w , which is gauged by the usual GSO projection to construct a modular-invariant partition function. The original Segal-Stolz-Teichner conjecture applies precisely to (0,1) SQFT's. If we wanted to make such an argument for a spacetime non-supersymmetric string theory (tachyonic or not), we face the obstacle that the GSO projection is different, and it involves *additional* worldsheet symmetries. For instance, the SO(16)² theory has a "diagonal" modular invariant partition function, which requires a notion of a left-moving worldsheet fermion number in addition to the (0,1) SQFT structure. Thus, valid SO(16)² worldsheet theories are equipped with an additional left-moving Z₂ symmetry, or equivalently, they are equipped with both a spin structure and a Z₂ symmetry. To repeat the argument of [42, 45], one must work with Z₂-equivariant TMF; it would be very interesting to do so.

When we started this project we were actually quite surprised that we could not find a comment on global anomalies of non-supersymmetric tachyon-free strings²⁸ anywhere in the literature. After all, these constructions are all 25+ years old, and they have a quite distinguished role in the string landscape. In a sense, they look more like our universe than the more familiar, supersymmetric theories! Maybe the reason for this neglect is simply lack of workforce; the last 25 years have brought so much progress on so many areas that the community just had to focus on the most novel or promising ones, and simply left many important questions unanswered. The physics of non-supersymmetric string theories was a victim to this rapid progress. Despite this, recent research in this direction has yielded e.g. metastable vacua [147, 169], novel end-of-the-world defects [170–176], and checks of Swampland constraints [12, 20, 177]. The results that we have presented in this paper are yet another step in this direction. We believe (and hope to have convinced at least some readers) that non-supersymmetric string theories constitute a very interesting arena where there seems to be an abundance of low-hanging fruit that is likely to yield novel lessons both in the Landscape and the Swampland.

²⁸Other than [36]; maybe we just missed it.

Acknowledgments

We are very thankful to Luis Álvarez-Gaumé, Carlo Angelantonj, Alberto Castellano, Markus Dierigl, Dan Freed, Christian Kneißl, Giorgio Leone, Ignacio Ruiz, Cumrun Vafa, Ángel Uranga, and Matthew Yu for many interesting discussions and insightful comments. We are particularly grateful to Yuji Tachikawa for explanations on equivariant topological modular forms. MM gratefully acknowledges the support and hospitality of KITP during the "Bootstrapping Quantum Gravity" Program, supported in part by the National Science Foundation under Grant No. NSF PHY-1748958, as well as the Aspen Center for Physics, which is supported by National Science Foundation grant PHY-2210452, in the context of the program "Traversing the Particle Physics Peaks". The travel of MM to ACP was supported by a grant of the Simons Foundation. During the early stages of this work, MM was supported by a grant from the Simons Foundation (602883,CV) and by the NSF grant PHY-2013858, by the Maria Zambrano Scholarship (UAM) CA3/RSUE/2021-00493, and currently by the Atraccion del Talento Fellowship 2022-T1/TIC-23956 from Comunidad de Madrid. The work of MD is supported by the FPI grant no. FPI SEV-2016-0597-19-3 from Spanish National Research Agency from the Ministry of Science and Innovation. MM and MD acknowledge the support of the grants CEX2020-001007-S and PID2021-123017NB-I00, funded by MCIN/AEI/10.13039/501100011033 and by ERDF A way of making Europe. Part of this work was completed while IB and AD participated in the "Geometry and the Swampland" conference at IFT in 2022; we thank the conference for support. MD also wishes to acknowledge the hospitality of the Arnold Sommerfeld Center for Theoretical Physics (ASC) at the Ludwig-Maximilians Universität throughout a research visit during which part of this work was completed.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- J. Polchinski, String theory. Vol. 1: An introduction to the bosonic string, Cambridge University Press (2007) [D0I:10.1017/CB09780511816079] [INSPIRE].
- [2] L. Álvarez-Gaumé, P.H. Ginsparg, G.W. Moore and C. Vafa, An O(16) × O(16) Heterotic String, Phys. Lett. B 171 (1986) 155 [INSPIRE].
- [3] L.J. Dixon and J.A. Harvey, String Theories in Ten-Dimensions Without Space-Time Supersymmetry, Nucl. Phys. B 274 (1986) 93 [INSPIRE].
- [4] S. Sugimoto, Anomaly cancellations in type I D-9-anti-D-9 system and the USp(32) string theory, Prog. Theor. Phys. 102 (1999) 685 [hep-th/9905159] [INSPIRE].
- [5] A. Sagnotti, Some properties of open string theories, in the proceedings of the International Workshop on Supersymmetry and Unification of Fundamental Interactions (SUSY 95), Palaiseau, France, May 15–19 (1995), p. 473–484 [hep-th/9509080] [INSPIRE].
- [6] A. Sagnotti, Surprises in open string perturbation theory, Nucl. Phys. B Proc. Suppl. 56 (1997) 332 [hep-th/9702093] [INSPIRE].

- [7] S. Abel, K.R. Dienes and E. Mavroudi, Towards a Phenomenology for the non-supersymmetric Heterotic String, PoS PLANCK2015 (2016) 001 [INSPIRE].
- [8] S. Abel, E. Dudas, D. Lewis and H. Partouche, *Stability and vacuum energy in open string models with broken supersymmetry*, *JHEP* **10** (2019) 226 [arXiv:1812.09714] [INSPIRE].
- [9] M. McGuigan, Dark Horse, Dark Matter: Revisiting the SO(16)x SO(16)' Nonsupersymmetric Model in the LHC and Dark Energy Era, arXiv:1907.01944 [INSPIRE].
- [10] H. Itoyama and S. Nakajima, Stability, enhanced gauge symmetry and suppressed cosmological constant in 9D heterotic interpolating models, Nucl. Phys. B 958 (2020) 115111
 [arXiv:2003.11217] [INSPIRE].
- [11] C. Angelantonj, Q. Bonnefoy, C. Condeescu and E. Dudas, String Defects, Supersymmetry and the Swampland, JHEP 11 (2020) 125 [arXiv:2007.12722] [INSPIRE].
- [12] I. Basile and S. Lanza, de Sitter in non-supersymmetric string theories: no-go theorems and brane-worlds, JHEP 10 (2020) 108 [arXiv:2007.13757] [INSPIRE].
- [13] I. Basile, On String Vacua without Supersymmetry: brane dynamics, bubbles and holography, Ph.D. thesis, Scuola Normale Superiore, I-56100 Pisa, Italy (2020) [arXiv:2010.00628]
 [INSPIRE].
- [14] J. Kaidi, Stable Vacua for Tachyonic Strings, Phys. Rev. D 103 (2021) 106026 [arXiv:2010.10521] [INSPIRE].
- [15] A.E. Faraggi, V.G. Matyas and B. Percival, Type 0 heterotic string orbifolds, Phys. Lett. B 814 (2021) 136080 [arXiv:2011.12630] [INSPIRE].
- [16] N. Cribiori, S. Parameswaran, F. Tonioni and T. Wrase, Misaligned Supersymmetry and Open Strings, JHEP 04 (2021) 099 [arXiv:2012.04677] [INSPIRE].
- [17] H. Itoyama and S. Nakajima, Marginal deformations of heterotic interpolating models and exponential suppression of the cosmological constant, Phys. Lett. B 816 (2021) 136195
 [arXiv:2101.10619] [INSPIRE].
- [18] E. Gonzalo, L.E. Ibáñez and I. Valenzuela, AdS swampland conjectures and light fermions, Phys. Lett. B 822 (2021) 136691 [arXiv:2104.06415] [INSPIRE].
- [19] R. Pérez-Martínez, S. Ramos-Sánchez and P.K.S. Vaudrevange, Landscape of promising nonsupersymmetric string models, Phys. Rev. D 104 (2021) 046026 [arXiv:2105.03460] [INSPIRE].
- [20] I. Basile, Supersymmetry breaking, brane dynamics and Swampland conjectures, JHEP 10 (2021) 080 [arXiv:2106.04574] [INSPIRE].
- [21] I. Basile, Supersymmetry breaking and stability in string vacua: Brane dynamics, bubbles and the swampland, Riv. Nuovo Cim. 44 (2021) 499 [arXiv:2107.02814] [INSPIRE].
- [22] A. Sagnotti and J. Mourad, String (In)Stability Issues with Broken Supersymmetry, LHEP
 2021 (2021) 219 [arXiv:2107.04064] [INSPIRE].
- [23] H. Itoyama, Y. Koga and S. Nakajima, Target space duality of non-supersymmetric string theory, Nucl. Phys. B 975 (2022) 115667 [arXiv:2110.09762] [INSPIRE].
- [24] N. Cribiori, S. Parameswaran, F. Tonioni and T. Wrase, Modular invariance, misalignment and finiteness in non-supersymmetric strings, JHEP 01 (2022) 127 [arXiv:2110.11973] [INSPIRE].
- [25] B. Percival, Classification of Non-Supersymmetric Heterotic String Vacua, Ph.D. thesis, University of Liverpool, Liverpool L69 3BX, U.K. (2022).

- [26] S. Raucci, On new vacua of non-supersymmetric strings, Phys. Lett. B 837 (2023) 137663 [arXiv:2209.06537] [INSPIRE].
- [27] I. Basile, S. Raucci and S. Thomée, Revisiting Dudas-Mourad Compactifications, Universe 8 (2022) 544 [arXiv:2209.10553] [INSPIRE].
- [28] Z.K. Baykara, D. Robbins and S. Sethi, Non-supersymmetric AdS from string theory, SciPost Phys. 15 (2023) 224 [arXiv:2212.02557] [INSPIRE].
- [29] Y. Koga, Interpolation and exponentially suppressed cosmological constant in non-supersymmetric heterotic strings with general Z2 twists, Nucl. Phys. B 990 (2023) 116160
 [arXiv:2212.14572] [INSPIRE].
- [30] E. Cervantes, O. Pérez-Figueroa, R. Pérez-Martínez and S. Ramos-Sánchez, Higgs-portal dark matter from nonsupersymmetric strings, Phys. Rev. D 107 (2023) 115007 [arXiv:2302.08520] [INSPIRE].
- [31] A.R.D. Avalos, A.E. Faraggi, V.G. Matyas and B. Percival, Fayet-Iliopoulos D-term in non-supersymmetric heterotic string orbifolds, Eur. Phys. J. C 83 (2023) 926 [arXiv:2302.10075] [INSPIRE].
- [32] S. Raucci, Fake supersymmetry with tadpole potentials, JHEP 07 (2023) 078 [arXiv:2304.12717] [INSPIRE].
- [33] J. Mourad and A. Sagnotti, Non-supersymmetric vacua and self-adjoint extensions, JHEP 08 (2023) 041 [arXiv:2305.09587] [INSPIRE].
- [34] A.R.D. Avalos, A.E. Faraggi, V.G. Matyas and B. Percival, *D-term uplifts in nonsupersymmetric heterotic string models*, *Phys. Rev. D* 108 (2023) 086007
 [arXiv:2306.16878] [INSPIRE].
- [35] B. Fraiman, M. Graña, H. Parra De Freitas and S. Sethi, Non-Supersymmetric Heterotic Strings on a Circle, arXiv:2307.13745 [INSPIRE].
- [36] Y.H. Kwon, The anomaly consideration of $O(16) \times O(16)$ heterotic string theory, Mod. Phys. Lett. A **2** (1987) 531 [INSPIRE].
- [37] E. Witten, Global gravitational anomalies, Commun. Math. Phys. 100 (1985) 197 [INSPIRE].
- [38] E. Witten, *Global anomalies in string theory*, in the proceedings of the *Symposium on Anomalies, Geometry, Topology*, Argonne, U.S.A., 28–30 March (1985) [INSPIRE].
- [39] D.S. Freed, Dirac charge quantization and generalized differential cohomology, hep-th/0011220 [INSPIRE].
- [40] A. Debray, M. Dierigl, J.J. Heckman and M. Montero, The anomaly that was not meant IIB, arXiv:2107.14227 [D0I:10.1002/prop.202100168] [INSPIRE].
- [41] D.S. Freed and M.J. Hopkins, Consistency of M-Theory on Non-Orientable Manifolds, Quart. J. Math. Oxford Ser. 72 (2021) 603 [arXiv:1908.09916] [INSPIRE].
- [42] Y. Tachikawa and M. Yamashita, Topological Modular Forms and the Absence of All Heterotic Global Anomalies, Commun. Math. Phys. 402 (2023) 1585 [Erratum ibid. 402 (2023) 2131]
 [arXiv:2108.13542] [INSPIRE].
- [43] Y. Lee and K. Yonekura, *Global anomalies in 8d supergravity*, *JHEP* 07 (2022) 125 [arXiv:2203.12631] [INSPIRE].
- [44] A. Debray and M. Yu, What bordism-theoretic anomaly cancellation can do for U, arXiv:2210.04911 [INSPIRE].

- [45] Y. Tachikawa, Topological modular forms and the absence of a heterotic global anomaly, PTEP 2022 (2022) 04A107 [arXiv:2103.12211] [INSPIRE].
- [46] A. Debray, Bordism for the 2-group symmetries of the heterotic and CHL strings, arXiv:2304.14764 [INSPIRE].
- [47] M. Dierigl, P.-K. Oehlmann and T. Schimannek, The discrete Green-Schwarz mechanism in 6D F-theory and elliptic genera of non-critical strings, JHEP 03 (2023) 090 [arXiv:2212.04503]
 [INSPIRE].
- [48] J. McNamara and C. Vafa, Cobordism Classes and the Swampland, arXiv:1909.10355 [INSPIRE].
- [49] M. Dierigl, J.J. Heckman, M. Montero and E. Torres, IIB string theory explored: Reflection 7-branes, Phys. Rev. D 107 (2023) 086015 [arXiv:2212.05077] [INSPIRE].
- [50] J. Kaidi, K. Ohmori, Y. Tachikawa and K. Yonekura, Nonsupersymmetric Heterotic Branes, Phys. Rev. Lett. 131 (2023) 121601 [arXiv:2303.17623] [INSPIRE].
- [51] A. Debray, M. Dierigl, J.J. Heckman and M. Montero, The Chronicles of IIBordia: Dualities, Bordisms, and the Swampland, arXiv:2302.00007 [INSPIRE].
- [52] D. Andriot, N. Carqueville and N. Cribiori, Looking for structure in the cobordism conjecture, SciPost Phys. 13 (2022) 071 [arXiv:2204.00021] [INSPIRE].
- [53] B. McInnes, The semispin groups in string theory, J. Math. Phys. 40 (1999) 4699 [hep-th/9906059] [INSPIRE].
- [54] S. Chaudhuri, G. Hockney and J.D. Lykken, Maximally supersymmetric string theories in D < 10, Phys. Rev. Lett. 75 (1995) 2264 [hep-th/9505054] [INSPIRE].
- [55] S. Nakajima, New non-supersymmetric heterotic string theory with reduced rank and exponential suppression of the cosmological constant, arXiv:2303.04489 [INSPIRE].
- [56] G. 't Hooft, Naturalness, chiral symmetry, and spontaneous chiral symmetry breaking, NATO Sci. Ser. B 59 (1980) 135 [INSPIRE].
- [57] L. Álvarez-Gaumé and P.H. Ginsparg, The Structure of Gauge and Gravitational Anomalies, Annals Phys. 161 (1985) 423 [Erratum ibid. 171 (1986) 233] [INSPIRE].
- [58] H.-C. Kim, G. Shiu and C. Vafa, Branes and the Swampland, Phys. Rev. D 100 (2019) 066006 [arXiv:1905.08261] [INSPIRE].
- [59] A. Adams, O. DeWolfe and W. Taylor, String universality in ten dimensions, Phys. Rev. Lett. 105 (2010) 071601 [arXiv:1006.1352] [INSPIRE].
- [60] M.B. Green, J.H. Schwarz and E. Witten, Superstring Theory Vol. 2: 25th Anniversary Edition, Cambridge University Press (2012) [D0I:10.1017/CB09781139248570] [INSPIRE].
- [61] M.B. Green and J.H. Schwarz, Anomaly Cancellation in Supersymmetric D = 10 Gauge Theory and Superstring Theory, Phys. Lett. B 149 (1984) 117 [INSPIRE].
- [62] W. Lerche, B.E.W. Nilsson and A.N. Schellekens, *Heterotic String Loop Calculation of the Anomaly Cancelling Term*, Nucl. Phys. B 289 (1987) 609 [INSPIRE].
- [63] P. Boyle Smith, Y.-H. Lin, Y. Tachikawa and Y. Zheng, Classification of chiral fermionic CFTs of central charge ≤ 16, arXiv:2303.16917 [INSPIRE].
- [64] I. Antoniadis, E. Dudas and A. Sagnotti, Brane supersymmetry breaking, Phys. Lett. B 464 (1999) 38 [hep-th/9908023] [INSPIRE].

- [65] C. Angelantonj, Comments on open string orbifolds with a nonvanishing B(ab), Nucl. Phys. B 566 (2000) 126 [hep-th/9908064] [INSPIRE].
- [66] G. Aldazabal and A.M. Uranga, Tachyon free nonsupersymmetric type IIB orientifolds via Brane-anti-brane systems, JHEP 10 (1999) 024 [hep-th/9908072] [INSPIRE].
- [67] C. Angelantonj et al., Type I vacua with brane supersymmetry breaking, Nucl. Phys. B 572 (2000) 36 [hep-th/9911081] [INSPIRE].
- [68] E. Dudas and J. Mourad, Consistent gravitino couplings in nonsupersymmetric strings, Phys. Lett. B 514 (2001) 173 [hep-th/0012071] [INSPIRE].
- [69] G. Pradisi and F. Riccioni, Geometric couplings and brane supersymmetry breaking, Nucl. Phys. B 615 (2001) 33 [hep-th/0107090] [INSPIRE].
- [70] N. Kitazawa, Brane SUSY Breaking and the Gravitino Mass, JHEP 04 (2018) 081
 [arXiv:1802.03088] [INSPIRE].
- [71] A.N. Schellekens and N.P. Warner, Anomalies, Characters and Strings, Nucl. Phys. B 287 (1987) 317 [INSPIRE].
- [72] E. Witten, An SU(2) Anomaly, Phys. Lett. B 117 (1982) 324 [INSPIRE].
- [73] I. García-Etxebarria and M. Montero, Dai-Freed anomalies in particle physics, JHEP 08 (2019)
 003 [arXiv:1808.00009] [INSPIRE].
- [74] E. Witten, Fermion Path Integrals And Topological Phases, Rev. Mod. Phys. 88 (2016) 035001
 [arXiv:1508.04715] [INSPIRE].
- [75] K. Yonekura, Dai-Freed theorem and topological phases of matter, JHEP 09 (2016) 022
 [arXiv:1607.01873] [INSPIRE].
- [76] J. Davighi, B. Gripaios and O. Randal-Williams, Differential cohomology and topological actions in physics, arXiv:2011.05768 [INSPIRE].
- [77] D.S. Freed, What is an anomaly?, arXiv:2307.08147 [INSPIRE].
- [78] J. Davighi, N. Lohitsiri and A. Debray, Toric 2-group anomalies via cobordism, JHEP 07 (2023)
 019 [arXiv:2302.12853] [INSPIRE].
- [79] Y. Tachikawa, Yamashita, and Mayuko, Anderson self-duality of topological modular forms, its differential-geometric manifestations, and vertex operator algebras, arXiv:2305.06196 [INSPIRE].
- [80] X.-Z. Dai and D.S. Freed, eta invariants and determinant lines, J. Math. Phys. 35 (1994) 5155 [hep-th/9405012] [INSPIRE].
- [81] M.F. Atiyah, V.K. Patodi and I.M. Singer, Spectral asymmetry and Riemannian Geometry 1, Math. Proc. Cambridge Phil. Soc. 77 (1975) 43 [INSPIRE].
- [82] D.S. Freed, M.J. Hopkins and C. Teleman, Consistent Orientation of Moduli Spaces, in The Many Facets of Geometry: A Tribute to Nigel Hitchin, O. Garcia-Prada et al. eds., Oxford University Press (2010), p. 395–420 [DOI:10.1093/acprof:oso/9780199534920.003.0019] [INSPIRE].
- [83] D.S. Freed and M.J. Hopkins, Reflection positivity and invertible topological phases, Geom. Topol. 25 (2021) 1165 [arXiv:1604.06527] [INSPIRE].
- [84] B.-L. Wang, Geometric cycles, index theory and twisted K-homology, J. Noncommut. Geom. 2 (2008) 497 [0710.1625].

- [85] J. Cheeger and J. Simons, Differential characters and geometric invariants, in Geometry and topology, Springer, Berlin (1985), pp. 50–80 [DOI:10.1007/BFb0075216].
- [86] U. Bunke and T. Nikolaus and M. Völkl, Differential cohomology theories as sheaves of spectra, J. Homotopy Relat. Struct. 11 (2016) 1.
- [87] D.S. Freed, K-theory in quantum field theory, in Current developments in mathematics, 2001, Cambridge, U.S.A., 16–17 November (2001) [International Press, Somerville, MA (2002), p. 41–87] [INSPIRE].
- [88] D.S. Freed, G.W. Moore and G. Segal, The Uncertainty of Fluxes, Commun. Math. Phys. 271 (2007) 247 [hep-th/0605198] [INSPIRE].
- [89] H. Sati, U. Schreiber and J. Stasheff, L_{∞} algebra connections and applications to String- and Chern-Simons n-transport, arXiv:0801.3480 [D0I:10.1007/978-3-7643-8736-5_17] [INSPIRE].
- [90] D.S. Freed and J. Lott, An index theorem in differential K-theory, Geom. Topol. 14 (2010) 903 [arXiv:0907.3508] [INSPIRE].
- [91] U. Bunke, String structures and trivialisations of a Pfaffian line bundle, Commun. Math. Phys. 307 (2011) 675.
- [92] J. Distler, D.S. Freed and G.W. Moore, Orientifold précis, in Mathematical foundations of quantum field theory and perturbative string theory, American Mathematical Society, Providence, RI (2011), pp. 159–172 [DOI:10.1090/pspum/083/2742428].
- [93] J. Distler, D.S. Freed and G.W. Moore, Spin structures and superstrings, in Surveys in differential geometry. Volume XV: Perspectives in mathematics and physics, International Press, Somerville, MA (2011), pp. 99–130 [DOI:10.4310/SDG.2010.v15.n1.a4].
- [94] D. Fiorenza, U. Schreiber and J. Stasheff, Čech cocycles for differential characteristic classes: an ∞-Lie theoretic construction, Adv. Theor. Math. Phys. 16 (2012) 149 [arXiv:1011.4735] [INSPIRE].
- [95] H. Sati, U. Schreiber and J. Stasheff, Differential twisted String and Fivebrane structures, Commun. Math. Phys. 315 (2012) 169 [arXiv:0910.4001] [INSPIRE].
- [96] D. Fiorenza, H. Sati and U. Schreiber, Extended higher cup-product Chern-Simons theories, J. Geom. Phys. 74 (2013) 130 [arXiv:1207.5449] [INSPIRE].
- [97] U. Schreiber, Differential cohomology in a cohesive infinity-topos, arXiv:1310.7930 [INSPIRE].
- [98] D. Fiorenza, H. Sati and U. Schreiber, Multiple M5-branes, String 2-connections, and 7d nonabelian Chern-Simons theory, Adv. Theor. Math. Phys. 18 (2014) 229 [arXiv:1201.5277]
 [INSPIRE].
- [99] D. Grady and H. Sati, Massey products in differential cohomology via stacks, J. Homotopy Relat. Struct. 13 (2018) 169.
- [100] S. Monnier and G.W. Moore, A Brief Summary Of Global Anomaly Cancellation In Six-Dimensional Supergravity, arXiv:1808.01335 [INSPIRE].
- [101] S. Monnier and G.W. Moore, Remarks on the Green-Schwarz Terms of Six-Dimensional Supergravity Theories, Commun. Math. Phys. 372 (2019) 963 [arXiv:1808.01334] [INSPIRE].
- [102] D. Fiorenza, H. Sati and U. Schreiber, The character map in (twisted differential) non-abelian cohomology, arXiv:2009.11909 [DOI:10.1142/13422] [INSPIRE].
- [103] C.-T. Hsieh, Y. Tachikawa and K. Yonekura, Anomaly Inflow and p-Form Gauge Theories, Commun. Math. Phys. 391 (2022) 495 [arXiv:2003.11550] [INSPIRE].

- [104] D. Grady and H. Sati, Differential cohomotopy versus differential cohomology for M-theory and differential lifts of Postnikov towers, J. Geom. Phys. 165 (2021) 104203 [arXiv:2001.07640]
 [INSPIRE].
- [105] M. Yamashita and K. Yonekura, Differential models for the Anderson dual to bordism theories and invertible QFT's, I, arXiv:2106.09270 [INSPIRE].
- [106] M. Yamashita, Differential models for the Anderson dual to bordism theories and invertible QFT's, II, arXiv:2110.14828 [INSPIRE].
- [107] M. Yamashita, Invertible QFTs and differential Anderson duals, arXiv:2304.08833 [INSPIRE].
- [108] I. García-Etxebarria, M. Montero and A. Uranga, Heterotic NS5-branes from closed string tachyon condensation, Phys. Rev. D 90 (2014) 126002 [arXiv:1405.0009] [INSPIRE].
- [109] J.S. Birman and J. Eisner, Seifert and Threlfall, A Textbook of Topology, Elsevier Science (1980) [ISBN: 9780080874050].
- [110] A. Beaudry and J.A. Campbell, A guide for computing stable homotopy groups, in Topology and quantum theory in interaction, Amererican Mathematical Society, Providence, RI (2018), pp. 89–136, [D0I:10.1090/conm/718/14476].
- [111] M.A. Hill, The 3-local tmf-homology of $B\Sigma_3$, Proc. Am. Math. Soc. **135** (2007) 4075.
- [112] M.A. Hill, The String bordism of BE_8 and $BE_8 \times BE_8$ through dimension 14, Illinois J. Math. 53 (2009) 183.
- [113] R.R. Bruner and J. Rognes, The Adams spectral sequence for topological modular forms, American Mathematical Society, Providence, RI (2021) [D0I:10.1090/surv/253].
- [114] A. Debray, Yu, and Matthew, Adams spectral sequences for non-vector-bundle Thom spectra, arXiv:2305.01678 [INSPIRE].
- [115] C.J. Schommer-Pries, Central extensions of smooth 2-groups and a finite-dimensional string 2-group, Geom. Topol. 15 (2011) 609 [arXiv:0911.2483] [INSPIRE].
- [116] A. Borel, Sur la cohomologie des espaces fibrés principaux et des espaces homogènes de groupes de Lie compacts, Annals Math. (2) 57 (1953) 115.
- [117] C.L. Douglas, J. Francis, A.G. Henriques and M.A. Hill, *Topological Modular Forms*, American Mathematical Society (2014) [DOI:10.1090/surv/201].
- [118] M. Behrens, A modular description of the K(2)-local sphere at the prime 3, Topology 45 (2006) 343 [math/0507184].
- [119] P.B. Shay, mod p Wu formulas for the Steenrod algebra and the Dyer-Lashof algebra, Proc. Am. Math. Soc. 63 (1977) 339.
- [120] T. Sugawara, Wu formulas for the mod 3 reduced power operations, Mem. Fac. Sci. Kyushu Univ. Ser. A 33 (1979) 297.
- [121] M.J. Hopkins and M. Mahowald, From elliptic curves to homotopy theory, in Topological modular forms, American Mathematical Society, Providence, RI (2014), pp. 261–285 [D0I:10.1090/surv/201/15].
- [122] A. Mathew, The homology of tmf, Homology Homotopy Appl. 18 (2016) 1 [arXiv:1305.6100].
- [123] A. Borel and F. Hirzebruch, Characteristic classes and homogeneous spaces. I, Amer. J. Math. 80 (1958) 458.
- [124] H. Cartan, Sur les groupes d'Eilenberg-Mac Lane. II, Proc. Nat. Acad. Sci. U.S.A. 40 (1954) 704.

- [125] J.-P. Serre, Sur les groupes d'Eilenberg-MacLane, C. R. Acad. Sci. Paris 234 (1952) 1243.
- [126] J.-P. Serre, Cohomologie modulo 2 des complexes d'Eilenberg-MacLane, Comment. Math. Helv. 27 (1953) 198.
- [127] T. Kudo, A transgression theorem, Mem. Fac. Sci. Kyūsyū Univ. A 9 (1956) 79.
- [128] J.F. Adams and S.B. Priddy, Uniqueness of BSO, Math. Proc. Cambridge Phil. Soc. 80 (1976) 475.
- [129] F.P. Peterson and N. Stein, Secondary characteristic classes, Annals Math. (2) 76 (1962) 510.
- [130] J.W. Milnor, Remarks concerning spin manifolds, in Differential and Combinatorial Topology (A Symposium in Honor of Marston Morse), Princeton Univiverityt Press, Princeton, N.J. (1965), p. 55–62.
- [131] D.W. Anderson, E.H. Brown, Jr. and F.P. Peterson, The structure of the Spin cobordism ring, Annals Math. (2) 86 (1967) 271.
- [132] J. Francis, Integrals on spin manifolds and the K-theory of K(Z, 4), https://sites.math.northwestern.edu/~jnkf/writ/bspin2011.pdf.
- [133] D.M. Davis, The splitting of $BO\langle 8 \rangle \wedge bo$ and $MO\langle 8 \rangle \wedge bo$, Trans. Am. Math. Soc. **276** (1983) 671.
- [134] R.E. Stong, Determination of $H^*(BO(k, \dots, \infty), Z_2)$ and $H^*(BU(k, \dots, \infty), Z_2)$, Trans. Am. Math. Soc. 107 (1963) 526.
- [135] V. Giambalvo, The mod p cohomology of BO(4k), Proc. Am. Math. Soc. 20 (1969) 593.
- [136] E. Witten, Topological tools in ten-dimensional physics, in Workshop on unified string theories, Santa Barbara, U.S.A., July 29 — August 16 (1985) [M.B. Green and D.J. Gross eds., World Scientific Publishing, Singapore (1986), p. 400–429].
- [137] J.A. Dixon, M.J. Duff and J.C. Plefka, Putting string / five-brane duality to the test, Phys. Rev. Lett. 69 (1992) 3009 [hep-th/9208055] [INSPIRE].
- [138] J. Mourad, Anomalies of the SO(32) five-brane and their cancellation, Nucl. Phys. B 512 (1998) 199 [hep-th/9709012] [INSPIRE].
- [139] E. Dudas and J. Mourad, D-branes in nontachyonic 0B orientifolds, Nucl. Phys. B 598 (2001) 189 [hep-th/0010179] [INSPIRE].
- [140] H. Imazato, S. Mizoguchi and M. Yata, Note on Anomaly Cancellation on SO(32) heterotic 5-brane, Mod. Phys. Lett. A 26 (2011) 1453 [arXiv:1010.1640] [INSPIRE].
- [141] E. Dudas, J. Mourad and A. Sagnotti, Charged and uncharged D-branes in various string theories, Nucl. Phys. B 620 (2002) 109 [hep-th/0107081] [INSPIRE].
- [142] E. Witten, Five-brane effective action in M theory, J. Geom. Phys. 22 (1997) 103
 [hep-th/9610234] [INSPIRE].
- [143] J.D. Blum and K.R. Dienes, Duality without supersymmetry: The case of the SO(16) \times SO(16) string, Phys. Lett. B 414 (1997) 260 [hep-th/9707148] [INSPIRE].
- [144] J.D. Blum and K.R. Dienes, Strong / weak coupling duality relations for nonsupersymmetric string theories, Nucl. Phys. B 516 (1998) 83 [hep-th/9707160] [INSPIRE].
- [145] A.E. Faraggi and M. Tsulaia, On the Low Energy Spectra of the Nonsupersymmetric Heterotic String Theories, Eur. Phys. J. C 54 (2008) 495 [arXiv:0706.1649] [INSPIRE].

- [146] J. Mourad and A. Sagnotti, AdS Vacua from Dilaton Tadpoles and Form Fluxes, Phys. Lett. B 768 (2017) 92 [arXiv:1612.08566] [INSPIRE].
- [147] I. Basile, J. Mourad and A. Sagnotti, On Classical Stability with Broken Supersymmetry, JHEP 01 (2019) 174 [arXiv:1811.11448] [INSPIRE].
- [148] E. Dudas and J. Mourad, Brane solutions in strings with broken supersymmetry and dilaton tadpoles, Phys. Lett. B 486 (2000) 172 [hep-th/0004165] [INSPIRE].
- [149] R. Blumenhagen and A. Font, Dilaton tadpoles, warped geometries and large extra dimensions for nonsupersymmetric strings, Nucl. Phys. B 599 (2001) 241 [hep-th/0011269] [INSPIRE].
- [150] P. Hořava and E. Witten, Eleven-dimensional supergravity on a manifold with boundary, Nucl. Phys. B 475 (1996) 94 [hep-th/9603142] [INSPIRE].
- [151] J. de Boer et al., Triples, fluxes, and strings, Adv. Theor. Math. Phys. 4 (2002) 995 [hep-th/0103170] [INSPIRE].
- [152] R.E. Stong, Appendix: calculation of $\Omega_{11}^{\text{Spin}}(K(Z, 4))$, in Workshop on unified string theories, Santa Barbara, U.S.A., July 29 — August 16 (1985) [M.B. Green and D.J. Gross eds., World Scientific Publishing, Singapore (1986), p. 430–437].
- [153] L. Evens, On the Chern classes of representations of finite groups, Trans. Am. Math. Soc. 115 (1965) 180.
- [154] M. Nakaoka, Homology of the infinite symmetric group, Annals Math. (2) 73 (1961) 229.
- [155] D. Quillen, The Adams conjecture, Topology 10 (1971) 67.
- [156] D.M. Davis and M. Mahowald, The immersion conjecture for RP^{8l+7} is false, Trans. Am. Math. Soc. 236 (1978) 361.
- [157] A.L. Liulevicius, The cohomology of a subalgebra of the Steenrod algebra, Trans. Am. Math. Soc. 104 (1962) 443.
- [158] N. Shimada and A. Iwai, On the cohomology of some Hopf algebras, Nagoya Math. J. 30 (1967) 103.
- [159] J. Kaidi, J. Parra-Martínez, Y. Tachikawa and A. Debray, Topological Superconductors on Superstring Worldsheets, SciPost Phys. 9 (2020) 10 [arXiv:1911.11780] [INSPIRE].
- [160] A. Debray, Invertible phases for mixed spatial symmetries and the fermionic crystalline equivalence principle, arXiv:2102.02941 [INSPIRE].
- [161] C. Schommer-Pries, Tori Detect Invertibility of Topological Field Theories, arXiv:1511.01772
 [INSPIRE].
- [162] A.M. Uranga, D-brane probes, RR tadpole cancellation and K-theory charge, Nucl. Phys. B 598 (2001) 225 [hep-th/0011048] [INSPIRE].
- [163] E. Witten, *D-branes and K-theory*, *JHEP* **12** (1998) 019 [hep-th/9810188] [INSPIRE].
- [164] A.N. Schellekens and N.P. Warner, Anomalies and Modular Invariance in String Theory, Phys. Lett. B 177 (1986) 317 [INSPIRE].
- [165] S. Stolz and P. Teichner, Supersymmetric field theories and generalized cohomology, arXiv:1108.0189 [INSPIRE].
- [166] M.J. Hopkins, Topological Modular Forms, the Witten Genus, and the Theorem of the Cube, in Proceedings of the International Congress of Mathematicians, Basel (1995), p. 554–565 [D0I:10.1007/978-3-0348-9078-6_49].

- [167] S. Gukov, D. Pei, P. Putrov and C. Vafa, 4-manifolds and topological modular forms, JHEP 05 (2021) 084 [arXiv:1811.07884] [INSPIRE].
- [168] D. Gaiotto, T. Johnson-Freyd and E. Witten, A Note On Some Minimally Supersymmetric Models In Two Dimensions, arXiv:1902.10249 [INSPIRE].
- [169] R. Antonelli and I. Basile, Brane annihilation in non-supersymmetric strings, JHEP 11 (2019)
 021 [arXiv:1908.04352] [INSPIRE].
- [170] G. Buratti, J. Calderón-Infante, M. Delgado and A.M. Uranga, Dynamical Cobordism and Swampland Distance Conjectures, JHEP 10 (2021) 037 [arXiv:2107.09098] [INSPIRE].
- [171] R. Blumenhagen, N. Cribiori, C. Kneissl and A. Makridou, Dynamical cobordism of a domain wall and its companion defect 7-brane, JHEP 08 (2022) 204 [arXiv:2205.09782] [INSPIRE].
- [172] R. Angius et al., At the end of the world: Local Dynamical Cobordism, JHEP 06 (2022) 142
 [arXiv:2203.11240] [INSPIRE].
- [173] R. Blumenhagen, C. Kneissl and C. Wang, Dynamical Cobordism Conjecture: solutions for end-of-the-world branes, JHEP 05 (2023) 123 [arXiv:2303.03423] [INSPIRE].
- [174] R. Angius, J. Huertas and A.M. Uranga, Small black hole explosions, JHEP 06 (2023) 070
 [arXiv:2303.15903] [INSPIRE].
- [175] J. Huertas and A.M. Uranga, Aspects of dynamical cobordism in AdS/CFT, JHEP 08 (2023)
 140 [arXiv:2306.07335] [INSPIRE].
- [176] J. Calderón-Infante, A. Castellano, A. Herráez and L.E. Ibáñez, Entropy bounds and the species scale distance conjecture, JHEP 01 (2024) 039 [arXiv:2306.16450] [INSPIRE].
- [177] I. Basile, Emergent Strings at an Infinite Distance with Broken Supersymmetry, Astronomy 2 (2023) 206 [arXiv:2201.08851] [INSPIRE].

4 Probes of Quantum Gravity

This chapter contains the articles:

- Black holes as probes of moduli space geometry, M. Delgado, Miguel Montero, Cumrun Vafa JHEP 04 (2023) 045 arXiv:2212.08676 - Inspire
- Emergence of Species Scale Black Hole Horizons, José Calderón-Infante, **M. Delgado**, Angel M. Uranga JHEP 01 (2024) 003 arXiv:2310.04488 - Inspire

Published for SISSA by 🖉 Springer

RECEIVED: January 12, 2023 ACCEPTED: April 1, 2023 PUBLISHED: April 11, 2023

Black holes as probes of moduli space geometry

Matilda Delgado,^a Miguel Montero,^b and Cumrun Vafa^b

ABSTRACT: We argue that supersymmetric BPS states can act as efficient finite energy probes of the moduli space geometry thanks to the attractor mechanism. We focus on 4d $\mathcal{N} = 2$ compactifications and capture aspects of the effective field theory near the attractor values in terms of physical quantities far away in moduli space. Furthermore, we illustrate how the standard distance in moduli space can be related asymptotically to the black hole mass. We also compute a measure of the resolution with which BPS black holes of a given mass can distinguish far away points in the moduli space. The black hole probes may lead to a deeper understanding of the Swampland constraints on the geometry of the moduli space.

KEYWORDS: Black Holes in String Theory, String and Brane Phenomenology

ARXIV EPRINT: 2212.08676



^aInstituto de Física Teórica IFT-UAM/CSIC,

C/ Nicolás Cabrera 13-15, Campus de Cantoblanco, 28049 Madrid, Spain

^bDepartment of Physics, Harvard University, Cambridge, MA 02138, U.S.A.

E-mail: matilda.delgado@uam.es, mmontero@g.harvard.edu, vafa@g.harvard.edu

Contents

1	Introduction	1
2	Review of 4d $\mathcal{N} = 2$ black hole solutions	3
3	Probing the prepotential with large black holes	5
4	Asymptotic black hole properties in moduli space	9
5	Resolution of the probing	11
6	Conclusions	12
\mathbf{A}	Asymptotic mass formulae for the mirror quintic	13

1 Introduction

One of the basic features of String Theory compactifications is the ubiquity of moduli fields — massless or light scalars that parametrize the internal compactification space and that control the masses and couplings of the theory. The geometry of moduli fields is relevant to the physical content of the theory and is captured by aspects of the Swampland Program including the Distance Conjecture [1]. Given their relevance, it seems an important question how would we probe the global geometry of moduli fields in terms of physical data at one point in moduli space.

More concretely, suppose one has a theory with an exact or approximate moduli space, and we live in a vacuum where the moduli ϕ take some value ϕ_0 . One can study directly the physics of the vacuum at $\phi = \phi_0$ by means of scattering experiments, and even learn about the local geometry of the moduli space by studying these couplings. However it is, in general, very difficult to design an experiment to probe the physics at some value $\phi = \phi_1$ which is very far away from ϕ_0 . It is precisely in faraway regions where interesting physics (such as decaying towers of states, emergence of perturbative string or decompactification limits, etc.) is supposed to take place. In most string theory literature, one is satisfied with studying the family of vacua parametrized by ϕ , as well as the ϕ dependence of various observables such as masses and couplings. Yet this approach is somewhat unphysical: changing the vev of ϕ everywhere at once costs infinite energy, and once one starts considering configurations where ϕ only changes in a region of finite size, other challenges can appear. One difficulty in designing setups that will probe large variations $\Delta \phi$ is that, when these are transplanckian, they will have a significant gravitational backreaction of their own, and whatever setup we consider is in danger of collapsing into a black hole. This was studied in [2], as well as in the more recent series of papers [3, 4], where it was pointed out that probing arbitrarily large $\Delta \phi$ is in principle possible in the effective field theory, but requires resources (masses, size of the laboratory...) exponentially large in $\Delta \phi$. Furthermore, the constructions described in [2] are not solving Einstein's equations, and so are regarded at best as interesting initial conditions, but they do not provide any concrete way to probe the faraway regions in moduli space.

The basic goal of this short note is to explain how the attractor mechanism allows one to overcome this challenge and do a form of "black hole spectroscopy", where properties of black holes at any one given point in moduli space can be used to probe the vacuum in faraway regions of the moduli space. Indeed, in a 4d $\mathcal{N}=2$ theory, thanks to the attractor mechanism (first constructed in [5] and further studied in [6-10]), the properties of the vector multiplet moduli space far away from any given vacuum can be studied reliably and in a robust way — simply engineer a black hole such that the near-horizon values of all scalars X_h^I differ significantly from those at infinity. The resulting geometry has the same asymptotics as the vacuum, but the near-horizon geometry constitutes a very long $AdS_2 \times S^2$ geometry, where the scalars are stabilized at the attractor value. The two regions are joined by an intermediate throat in which the fields X^{I} run. The size of the S^{2} (or equivalently, the curvature of the AdS_2) are furthermore controlled by the total value of the black hole charge, which may be safely rescaled to arbitrarily large values without affecting the attractor solution. What this means is that one can, while keeping the attractor value fixed, engineer an $AdS_2 \times S^2$ region where the size of the S^2 is arbitrarily large, and in which the physics looks locally like the vacuum solution on \mathbb{R}^4 with the attractor values of the moduli, thus achieving a concrete "laboratory" in which the asymptotic vacuum can be probed. To make this picture concrete, we will show explicitly how the triple intersection numbers of the infinite distance limit, as well as the first subleading correction, can be encoded in term of mass and degeneracy of charged BPS states that can be physically measured in the asymptotic region.

If one has access to arbitrary mass/charge states, it becomes possible to study points in moduli space which are arbitrarily far away and with any desired precision. It is more interesting to study how the distance in moduli space and the resolution of the probing depend on the resources — how well can we do if we have a maximum allowed mass, charge for the states. We study this question in a simple two-parameter family of black holes, finding agreement with the results in [2] that an exponential field range in moduli space are intimately related to the masses which trigger the flow. However, unlike in [2], we have a concrete setup where transplanckian field ranges are attained in the context of a supersymmetric solution, in a time-independent way, thanks to the attractor mechanism.

The rest of the note includes a review of 4d $\mathcal{N} = 2$ black holes and the attractor mechanism in section 2, followed by the main application of black hole spectroscopy in section 3, where we obtain the triple intersection numbers of an asymptotic limit in terms of degeneracy of states. Section 4 quantifies just how far can we go in probing the moduli space geometry for a given mass, and section 5 explains how we quantify the resolution of points on moduli space using physical data at far away points. Section 6 contains a few concluding remarks.

2 Review of 4d $\mathcal{N} = 2$ black hole solutions

We will start by reviewing some elements of Calabi-Yau three-fold X compactifications of Type IIB string theory and their associated black hole solutions, which will be the core of this paper. The reader interested in further details is encouraged to check [11, 12]; here we will only describe the essentials of what we need. At low energies, the effective field theory describing a Calabi-Yau compactification is a four-dimensional $\mathcal{N} = 2$ supergravity, coupled to $n_V = h^{1,2}$ abelian vector multiplets and $n_H = h^{1,1} + 1$ hypermultiplets, where $h^{i,j}$ are the Hodge numbers of the Calabi-Yau three-fold X. The vector multiplet moduli space is a special Kähler manifold, and its scalars parametrize the complex structure of X. The dynamics of the hypermultiplets decouples completely from that of the vectors in the black holes we will consider, due to the 4d $\mathcal{N} = 2$, so we will mostly ignore them in the following.

As explained e.g. in [13], the intersection pairing in the middle cohomology of the Calabi-Yau defines a symplectic (antisymmetric) inner product. Constructing the complex structure moduli space comes down to choosing a symplectic basis $\{A^I, B_I\}$ of 3-cycles in $H_3(X, \mathbb{Z})$ (and the corresponding basis of three forms $\{\alpha_I, \beta^J\}$ of $H^3(X, \mathbb{Z})$), which we take to be orthonormal in the following sense:

$$\langle \alpha_I, \beta^J \rangle = -\langle \beta^J, \alpha_I \rangle = \int_X \alpha_I \wedge \beta^J = \delta_I^J \int_{A^I} \alpha_J = -\int_{B_J} \beta^I = \delta_J^I ; \quad \int_{A^I} \beta^J = \int_{B_J} \alpha_I = 0 ,$$
 (2.1)

where $\{I, J\} \in \{0, ..., h^{2,1}\}$. Every Calabi-Yau manifold has a holomorphic (3,0)-form (see e.g. [11]) that can be decomposed as follows in terms of its A- and B-periods $\{X^I, F_J\}$:

$$X^{I} = \int_{A^{I}} \Omega_{3} \quad F_{J} = \int_{B_{J}} \Omega_{3} \quad \longleftrightarrow \quad \Omega_{3} = X^{I} \alpha_{I} - F_{J} \beta^{J} \,. \tag{2.2}$$

Performing a change $\Omega_3 \to e^f \Omega_3$ has no impact on the complex structure of X. In terms of the scalars X^I , this amounts to an overall re-scaling $X^I \to e^f X^I$ from which it is clear that only $h^{2,1}$ of these $h^{2,1} + 1$ scalars are independent. The Kähler potential is given by (see. eg [14])

$$\mathcal{K} = -\ln\left(i\int_X \Omega_3 \wedge \bar{\Omega}_3\right) = -\ln i\left(\bar{X}^I F_I - X^I \bar{F}_I\right).$$
(2.3)

One can now see that a rescaling of Ω_3 corresponds to a Kähler transformation in 4d $\mathcal{N} = 2$ language:

$$\Omega_3 \to e^f \Omega_3 \quad \mathcal{K} \to \mathcal{K} - f - \bar{f} \,.$$
 (2.4)

The fact that the complex structure of X is unchanged by a rescaling of Ω_3 translates to the 4d $\mathcal{N} = 2$ Lagrangian being invariant under Kähler transformations. One can therefore define a Kähler metric on the complex structure moduli space that is invariant under rescalings of Ω_3 by using \mathcal{K} as a Kähler potential. The metric obtained in this way on complex structure moduli space $g_{I\bar{J}} \sim \partial_I \partial_{\bar{J}} \mathcal{K}$ coincides with what can be read off of the kinetic term of the 4d Lagrangian for the complex structure moduli. One can choose a symplectic basis such that a single holomorphic function, the so-called prepotential F = F(X), encodes all the data of the topological theory. The B-periods can be reexpressed in terms of the prepotential as:

$$F_J(X) = \frac{\partial F(X)}{\partial X^J}.$$
(2.5)

One can construct black hole solutions in the 4d $\mathcal{N} = 2$ effective theory by wrapping D-branes on the various cycles of X (see eg. [9, 10, 15, 16]). These are generalizations of Reissner-Nordström black holes, charged under the $n_V = h^{1,2}$ abelian vector multiplets. These black holes have the remarkable property that they are *attractors* for the vector multiplet moduli. This means that these moduli, in general, run along the radial direction until they reach the black hole horizon where their value is entirely determined by the supersymmetric equations of motion, in what is known as the attractor mechanism [5]. The attractor equations that describe this flow relate the charges of the black hole to the values of the moduli at the horizon. Throughout this work, we will use the attractor mechanism as a tool to map black hole thermodynamic properties to the complex structure moduli space.

Let us now review the attractor mechanism of extremal 4d $\mathcal{N} = 2$ black holes in more detail. In type IIB language, one constructs such black holes by wrapping D3 branes on a general 3-cycle \mathcal{C} in X. Indeed, a black hole is identified by the decomposition of \mathcal{C} onto the basis $\{A^I, B_I\}$ or equivalently by its corresponding electric and magnetic charges $\{p^I, q_J\}$. Take Γ to be the 3-form that is Poincaré dual of \mathcal{C} , then the corresponding splitting of magnetic and electric charges $\{p^I, q_J\}$ is given by:

$$p^{I} = \int_{A^{I}} \Gamma \quad q_{J} = \int_{B_{J}} \Gamma.$$
(2.6)

Consider a BPS solution charged under the 3-form Γ . Then, the central charge of the black hole is given by:

$$Z = e^{\mathcal{K}/2} \int_X \Omega_3 \wedge \Gamma = e^{\mathcal{K}/2} (p^I F_I - q_I X^I) \,. \tag{2.7}$$

The attractor mechanism acts as a potential for the moduli and drives them to minimizing the central charge at the horizon of the black hole (note that the horizon values of the moduli will differ significantly, in general, from their values at spatial infinity) [8]. This minimization procedure leads to the attractor equations at the horizon, which relate the holomorphic periods to the charges of the black hole and can be written as follows:

$$p^{I} = \operatorname{Re}\left[C_{h}X_{h}^{I}\right] = C_{h}X_{h}^{I} + \bar{C}_{h}\bar{X}_{h}^{I},$$

$$q_{I} = \operatorname{Re}\left[C_{h}F_{hI}\right] = C_{h}F_{hI} + \bar{C}_{h}\bar{F}_{hI},$$
(2.8)

where the "h" subscripts emphasize that these quantities are evaluated at the horizon and where we have introduced $C \equiv -2i\bar{Z}e^{\mathcal{K}/2}$. The vector multiplet moduli, which can be expressed in terms of the X^{I} , have arbitrary values infinitely far from the black hole, they vary along the radial direction and are fixed by the attractor equations at the horizon. Solving these equations for the periods at the horizon allows one to obtain the entropy of the black hole (equivalently, its area), which is expressed in terms of the central charge as:

$$S = \pi |Z_h|^2 \,. \tag{2.9}$$

One can also compute the ADM mass of the black hole, which turns out to be

$$M_{ADM}^2 = |Z_{\infty}|^2, \tag{2.10}$$

where we have introduced the subscript to emphasize that the ADM mass is obtained by evaluating the central charge Z, viewed as a function of the charges p^I , q_I and the scalar values X^I given in (2.7), with the scalars X^I taken to have their asymptotic values, i.e. evaluated at infinite distance from the black hole. In the particular case where the asymptotic and near-horizon values of the scalars coincide, (2.10) and (2.8) agree: the attractor value of the mass is just given by the near-horizon dynamics. When they do not, the difference is due entirely to the running scalars outside of the horizon contributing to the mass. This follows from the attractor equations, which imply [8]

$$|Z_{\infty}|^{2} - |Z_{h}|^{2} = \int_{-\infty}^{0} d\tau \, e^{-U/2} \sqrt{g_{I\bar{J}} \frac{dt^{I}}{d\tau} \frac{dt^{\bar{J}}}{d\tau}}.$$
(2.11)

In this expression, $t^I \equiv X^I/X^0$ are the physical moduli, e^{2U} is the time-component of the black hole metric, and τ is a certain parametrization of the radial coordinate in which the horizon sits at $\tau = -\infty$ and spatial infinity is at $\tau = 0$. Thus, we see that the difference in mass above the attractor value is just the backreaction of the running moduli.

Finally, we also note that the attractor equations and in particular the charges are invariant under Kähler transformations, which act on the periods and C as follows:

$$\mathcal{K} \to \mathcal{K} - f - \bar{f}, \quad C \to e^{-f}C, \quad X^I \to e^f X^I, \quad F \to e^{2f}F.$$
 (2.12)

One can obtain the $h^{2,1}$ physical, invariant, moduli t^I by choosing special coordinates such as $X^I = t^I X^0$. Throughout the next sections we will be solving the attractor equations (2.8) by choosing a constant C_h . Solving the attractor equations with different values of C_h will generate a set of black hole solutions with the same attractor point in moduli space but different charges and masses.

In the next section we will exploit the attractor equations in an attempt to map topological data of the Calabi-Yau moduli space to thermodynamic properties of black holes.

3 Probing the prepotential with large black holes

Armed with the attractor mechanism described in the previous section, we will explain how it can be used to achieve a simple form of black hole moduli space spectroscopy, where we relate the properties of faraway points in moduli space to statistical, thermodynamic properties of large charge BPS states in a given vacuum. As described above, the attractor mechanism produces near-horizon $AdS_2 \times S^2$ geometries where the value of the moduli are controlled by the attractor mechanism and can in general be very different from the asymptotic values of the moduli. For concreteness and simplicity, we will be interested in black holes that take the vector multiplet moduli to near-infinite distance limits in their moduli space. In these regions, the prepotential is constrained to take the well-known form:

$$F(X) = -D_{IJK} \frac{X^{I} X^{J} X^{K}}{X^{0}}, \qquad (3.1)$$

where $C_{IJK} = 6D_{IJK}$ are integers which, in the Calabi-Yau context, receive the interpretation of the triple intersection numbers of the mirror Calabi-Yau. But it is expected that this structure follows from general quantum gravity principles, even when a Calabi-Yau description is not present (see [17]).

With this, we see from (2.5) that the attractor equations will relate the charges of the black hole directly to the X^{I} at the horizon and the parameters D_{IJK} (we take C_h to be a constant at the horizon). Naturally, these charges will be very large since the moduli are reaching near-infinite values. Turning things around, solving these equations for the moduli at the horizon would allow us to express the entropy of the black hole (2.9) in terms of the charges $\{p^{I}, q_{J}\}$ and D_{IJK} . This would show that if one could measure the entropy and charge of one of these large black holes experimentally, it would be possible to deduce the values of the D_{IJK} . One would therefore recover topological data of the underlying Calabi-Yau from measuring black hole observables at a very different point in moduli space. Furthermore, quantities like electric and magnetic charges, or the degeneracy of charged BPS states (i.e., entropy of the black holes), are actual observables, which one could measure experimentally.

We will just illustrate this method in the simplest example, and assume that we have a single vector multiplet $n_V = 1$. Then, there are just four periods, and from the prepotential (3.1) we have

$$F_1 = -3D_{111}\frac{(X^1)^2}{X^0}$$
 and $F_0 = D_{111}\frac{(X^1)^3}{(X^0)^2}$. (3.2)

One can set $C_h = 1$ at the horizon by a Kähler transformation, and then the attractor equations are given by:

$$p^{0} = \operatorname{Re} \begin{bmatrix} X^{0} \end{bmatrix} \qquad q_{0} = D_{111} \operatorname{Re} \begin{bmatrix} \frac{(X^{1})^{3}}{(X^{0})^{2}} \end{bmatrix}$$
$$p^{1} = \operatorname{Re} \begin{bmatrix} X^{1} \end{bmatrix} \qquad q_{1} = -3D_{111} \operatorname{Re} \begin{bmatrix} \frac{(X^{1})^{2}}{X^{0}} \end{bmatrix}. \qquad (3.3)$$

Solving these equations yields the central charge at the horizon in terms of the X^{I} fields,

$$|Z|^{2} = \frac{D_{111}|X^{1}\bar{X^{0}} - X^{0}\bar{X^{1}}|^{3}}{4|X^{0}|^{4}}.$$
(3.4)

Equivalently, one can solve (3.3) for the periods and express the entropy in terms of the charges and D_{111} . For simplicity, we will assume that one of the charges vanishes $(p_0 = 0)$, in which case we obtain the entropy as:

$$S = \pi |p^1| \frac{\sqrt{|q_1^2 - 12D_{111}p^1q_0|}}{\sqrt{3}} \,. \tag{3.5}$$

The argument of the square root is always positive if we pick charges such that the attractor equations have a solution. We emphasize that an expression such as (3.5) is anyway only expected to hold for very large charges, and in a one-parameter family of solutions such that

the attractor values of the scalar are approaching the infinite distance limit in which (3.1) is approximately valid. One example of such a family can be parametrized as follows: in terms of the physical modulus $t = X^1/X^0$, take the charges that scale, in the $y \sim |t| \to \infty$ limit, as

$$\mathcal{Q}_{\infty} = \begin{cases}
p^{0} = 0 \\
p^{1} = N \\
q_{0} = -y^{2}N \\
q_{1} = 0
\end{cases}$$
(3.6)

Here, N is an overall rescaling of the charges, that does not affect the attractor value, but which will be important in a number of applications in what follows. Importantly, we have chosen a family of black holes whose charges solve (3.3) but do not depend on D_{111} explicitly. We are trying to encode D_{111} in terms of observables such as charges and the degeneracy of BPS states, and therefore, choosing charges depending on D_{111} would amount to assuming the answer. For the family (3.6), the entropy formula simplifies to

$$S = 2\pi \sqrt{D_{111}} \, y N^2 \,. \tag{3.7}$$

By counting the number of BPS states with such charges in a 4d $\mathcal{N} = 2$ world, one could use this formula to obtain an experimental evaluation of the triple-intersection number D_{111} , and provides a direct link between moduli space properties and properties of the prepotential. This is an interesting result since one would not expect to be able to probe far away moduli data from measurements in the middle of moduli space, without any knowledge of the underlying compactification. Naturally, the formula (3.7) is to be taken as proof of concept that such a relation can be made. The exact expression will change if we consider a large black hole with charges that scale differently, or if one considers a different Calabi-Yau for the compactification.

A natural next step is to determine whether this framework can also be used to detect subleading corrections in the prepotential. To this effect, we will repeat the above analysis, now including the first correction to the prepotential as one moves slightly into the bulk of moduli space:

$$F(X) = -D_{IJK} \frac{X^{I} X^{J} X^{K}}{X^{0}} + d_{I} X^{I} X^{0}.$$
(3.8)

In the Calabi-Yau case, we can relate d_I to topological properties of the mirror Calabi-Yau via the formula

$$\int_{CY} c_2 \wedge \alpha_I = 24 \, d_I, \tag{3.9}$$

where c_2 and α_I are the second Chern class and the corresponding two-form of the mirror Calabi-Yau. Again, one can write the attractor equations in the simplified case where there is only one modulus, where they become

$$p^{0} = \operatorname{Re}\left[X^{0}\right]$$
 $q_{0} = D_{111} \operatorname{Re}\left[\frac{(X^{1})^{3}}{(X^{0})^{2}}\right] + d_{1} \operatorname{Re}\left[X^{1}\right]$ (3.10)

$$p^{1} = \operatorname{Re}\left[X^{1}\right]$$
 $q_{1} = -3D_{111}\operatorname{Re}\left[\frac{(X^{1})^{2}}{X^{0}}\right] + d_{1}\operatorname{Re}\left[X^{0}\right].$ (3.11)

One can solve these equations for the periods at the horizon and express the entropy in terms of the charges. For a black hole with a single vanishing charge $p^0 = 0$, one obtains:

$$S = \frac{\pi}{\sqrt{3}} |p^1| \sqrt{12d_1 D_{111}(p^1)^2 - 12D_{111}p^1 q_0 + (q^1)^2} \,. \tag{3.12}$$

It is easy to see that this reduces exactly to (3.5) when d_1 is set to zero. One can evaluate the entropy of the large black hole with charges that scale as (3.6) in the $y \sim |t| \to \infty$ limit and obtain:

$$S = 2\pi N^2 \sqrt{D_{111}(y^2 + d_1)} \,. \tag{3.13}$$

Expanding this near $y \to \infty$, one obtains

$$S = \pi N^2 \left[2\sqrt{D_{111}}y + d_1\sqrt{D_{111}}\frac{1}{y} + \mathcal{O}(y^{-3}) \right].$$
(3.14)

Having previously measured D_{111} using (3.7) with extremely large black holes, one could measure deviations of this expression for slightly smaller black holes and obtain d_1 from (3.13).

An important point in all of the above is to note that we have been using the leading behavior of the black hole entropy. This is valid for large N and one expects to receive corrections suppressed by powers of $1/N^2$ [18]. So for example if we want the D_{111} term to be measurable, we need to ensure that

$$\sqrt{D_{111}}y \gtrsim \mathcal{O}(1/N^2), \qquad (3.15)$$

which is automatically satisfied in our case since $y \gg 1$ and D_{111} is an integer. For the subleading term to be measurable we need to make sure

$$d_1 \sqrt{D_{111}} / y \gtrsim \mathcal{O}(1/N^2) \,, \tag{3.16}$$

which would be achievable if we pick $N \gtrsim \sqrt{y}$. In addition to polynomial corrections in prepotential, there are also exponential corrections:

$$\frac{F}{X_0^2} = -D_{IJK}t^I t^J t^K + d_I t^I + K_{\alpha_I} e^{-\alpha_I t^I} + \dots, \qquad (3.17)$$

where the coefficients K_{α_I} are quantized in the primitive directions [19]. In the one modulus example we have studied (where there is a single coefficient, $\alpha_1 = 1$), to get to the precision to be able to measure K_{α_I} , we need to measure the degeneracy of large charge states, where

$$\sqrt{D_{111}} N^2 \gtrsim \mathcal{O}(y^3 e^{2y}), \qquad (3.18)$$

which is consistent with the intuition that measurement of exponentially small corrections require exponentially large charged BPS states. As we will discuss in the next section yitself is exponential in distance in moduli space, so this is a double exponential in terms of distance.

The above procedure could be refined indefinitely, recovering more and more information about the Calabi-Yau by measuring the sizes of smaller and smaller black holes. At each order in y, one can solve the attractor equations with the corrected prepotential and obtain the entropy in terms of the charges, the previously determined parameters and the new undetermined ones. Measuring the size of an appropriate selection of black holes will allow one to obtain the value of the undetermined parameters. The examples above describe the case with a single vector multiplet; in general, when $n_V > 1$, one will need a multiparameter family of black holes at each step. For instance, in the first step, one would need a large black hole with charges analogous to (3.6) for each direction in moduli space in order to recover all of the triple intersection numbers D_{LIK} . Of course, this procedure becomes increasingly more complicated as we go further away from the controlled corners of the moduli space, though see [20], where a similar iterative procedure was used to find generic solutions to the attractor equations at all orders, also incorporating instanton contributions. At low orders, this method is equivalent to our own. Nevertheless, the fact that such a procedure can be carried out in principle suggests that there is no obstruction in recovering the geometry of moduli space at any point, using physical measurements of charged BPS states at other points. However, the BPS degeneracy of states is a function of the attractor point alone, and it is not helpful in relating the asymptotic and attractor values of the moduli in a meaningful way. In the next section we will address this question by considering the energetics of the BPS states and relating it to asymptotic distances in moduli space.

4 Asymptotic black hole properties in moduli space

As we saw in the previous section, one can directly relate the prepotential, and thus, the usual moduli space metric, to degeneracy of BPS states. This suggests that it might be possible to obtain the full geometry of moduli space from other physical measurements. We now show that the attractor flow can correctly capture the notion of asymptotic distance near the boundaries of moduli space; namely, we will show that asymptotically in moduli space, the entropy of large BPS states and also their masses can be directly related to the distance in moduli space to their attractor points.

To do this, we once more consider black holes whose attractor point is near the boundary of moduli space, as in the previous section. Using again the family of black holes with charges (3.6), the corresponding periods at the horizon are given by:

$$\operatorname{Re} \begin{bmatrix} X^{0} \end{bmatrix} = 0, \quad \operatorname{Im} \begin{bmatrix} X^{0} \end{bmatrix} = -ND_{111}^{1/2}y^{-1},$$

$$\operatorname{Re} \begin{bmatrix} X^{1} \end{bmatrix} = N, \quad \operatorname{Im} \begin{bmatrix} X^{1} \end{bmatrix} = 0.$$
(4.1)

From these expressions, it is straightforward to obtain the Kähler potential (2.3) in this limit:

$$\mathcal{K} = -\log\left(8N^2\sqrt{D_{111}}y\right) \,. \tag{4.2}$$

The metric in moduli space is

$$ds^{2} = 2\partial_{t}\partial_{\bar{t}}K|dt|^{2} = \frac{3}{2}\frac{|dt|^{2}}{\Im(t)^{2}} = \frac{3}{2}\frac{dy^{2}}{y^{2}}, \qquad (4.3)$$

so that the distance in moduli space is given by:

$$d \sim \sqrt{\frac{3}{2}} \log y \,. \tag{4.4}$$

From the formula for the entropy (3.7) obtained for this set of charges, one immediately obtains

$$S \sim N^2 e^{\sqrt{\frac{2}{3}}d} \,. \tag{4.5}$$

for this family of black hole solutions. A similar exponential relation holds asymptotically for the ADM mass: using the general expression (2.10), one gets that

$$M_{BPS} = e^{\mathcal{K}_{\infty}/2} |p_1 F_{\infty}^1 - q_0 X_{\infty}^0| = N e^{\mathcal{K}_{\infty}/2} |F_{\infty}^1 + y^2 X_{\infty}^0|, \qquad (4.6)$$

where the subscript ∞ in any quantity denotes its asymptotic values. For large y, the last term is leading, giving a dependence on the ADM mass that agrees parametrically with the entropy, and so

$$M_{BPS} \sim Ne^{\sqrt{\frac{8}{3}}d},\tag{4.7}$$

as well.

These expressions relate the mass of a large BPS state to the attractor point lying at a far away distance in moduli space. In practice, it means that if one wants to probe the moduli space at a large distance d, one needs to create a massive BPS state with energy which is exponentially large in distance. This is reminiscent of the work by Nicolis [2] where it was shown that, in a purely Newtonian setting, it was possible to construct setups with scalar sources that lead to arbitrarily large field ranges, with a size that grows exponentially on the field range. What we are finding is not only in agreement with this, but more broadly, with the findings of [3, 4], which studied transplanckian field displacements in a variety of setups (including 4d dilatonic black holes). As proposed in [4], there really seems to be a universal feature of quantum gravity that arbitrarily large field ranges can be probed at an exponential cost in physical resources. Other instances where one can see this include the extended objects of [21, 22] which probe infinite distances in moduli space; their tension goes exponentially with the distance. It would be very interesting to find the physical mechanism underlying this phenomenon.

A scaling similar to (4.5) was recovered in [23] in relation to the Black Hole Entropy Distance conjecture proposed in [24], a generalization of the ADC to black hole spacetimes. Both [23, 24] proposed identifying the logarithm of the black hole entropy (the horizon area) with a notion of distance, encoded in the change of the metric when the flux changes one unit. The connection to (4.7) is precisely that, when the quantized black hole charges change and the black hole area readjust, so do the vevs of the vector multiplet moduli, and the notion of distance using these or directly the metric as in [23, 24] agree. In any case, equation (4.5) shows that the black holes provide a thermodynamic interpretation for the distance in moduli space, if only asymptotically.

Appendix A discussed the precise realization of the general discussion in this paper in the context of a specific model, namely the mirror quintic M. The results agree with (4.5), as they should.

5 Resolution of the probing

As we have seen, BPS states can serve as effective probes of far away regions of moduli space via their attractor geometry. However, when solving the attractor equations (3.3), one needs to take into account Dirac quantization, which demands that the charges p^{I}, q_{I} are quantized. In the Calabi-Yau picture, the quantization simply maps to the fact that the D3 branes that form the black holes we study must wrap an integer homology class.

Dirac quantization implies that the picture of the moduli space provided by the black holes is not continuous; rather, it is naturally a mesh of points in the moduli space. These issues of quantization are however often ignored in the study of 4d $\mathcal{N} = 2$ black holes, simply because of the Kähler transformation (2.12). This transformation tells us that a homogeneous rescaling of the charges does not affect attractor values; as a result, one may simply scale the charges up, to very large values, achieving an arbitrarily dense mesh. While this is true, if one is constrained to finite resources (finite black hole mass, charge, or equivalently, entropy), the mesh allowed by Dirac quantization will be finite, leading to a finite "resolution" in the probing of moduli space. We will determine this resolution momentarily.

In more detail, consider the attractor equations near the boundary of moduli space (3.3). The general solution with non-vanishing charges $p^1 > 0, q^0 < 0$ is, asymptotically,

$$X^{1} = p_{1}, \quad X_{0} = \sqrt{-\frac{p_{1}^{3}}{q_{0}}D_{111}}, \quad t = \frac{X^{1}}{X^{0}} = \sqrt{-\frac{q_{0}}{p_{1}D_{111}}}.$$
 (5.1)

We see that the attractor value of the physical modulus is insensitive to an overall rescaling of the charges. As we make a small change in the charges, the value of t in (5.5) changes, and we probe a nearby point of moduli space. The smallest such change that can take place, compatible with Dirac quantization, is changing p_1 by one unit while keeping q_0 constant. Under such a change, we obtain that the infinitesimal change in moduli space distance, δd , is given by

$$\delta d = \sqrt{\frac{3}{2}} \frac{\delta t}{t} = \sqrt{\frac{3}{8}} \frac{1}{p_1},$$
(5.2)

where we have used the asymptotic form of the Kähler potential, $\mathcal{K} \sim -3 \log t$. Using the asymptotic relation between the moduli space distance and the change in t,

$$d \sim \sqrt{\frac{3}{2}} \log t \quad \Rightarrow \quad t \sim e^{\sqrt{\frac{2}{3}}d},$$
(5.3)

combined with (5.1), one can rewrite (5.2) as

$$\delta d = \sqrt{\frac{3}{8}} \frac{t^2 D_{111}}{|q_0|} \sim \sqrt{\frac{3}{8}} \frac{D_{111} e^{\sqrt{\frac{8}{3}d}}}{|q_0|},\tag{5.4}$$

in terms of the moduli space distance traversed by the black hole. Now, close to the infinite distance limit, $|t| \to \infty$, (5.1) tells us that p_1 is much smaller than q_0 , and so the ADM mass

$$M_{BPS} = e^{\mathcal{K}_{\infty}/2} \left| p_1 F_{\infty}^1 - q_0 X_{\infty}^0 \right|,$$
 (5.5)

can be approximated by the second term,

$$M_{BPS} \approx e^{\mathcal{K}_{\infty}/2} X_{\infty}^{0} |q_0|.$$
(5.6)

This last equation allows us to replace $|q_0|$ by the black hole mass M in (5.4), yielding an expression

$$\delta d = \sqrt{\frac{3}{8}} D_{111} e^{\mathcal{K}_{\infty}/2} X_{\infty}^{0} \frac{e^{\sqrt{\frac{8}{3}}d}}{M}, \qquad (5.7)$$

and finally, defining the resolution of the moduli space probing as the inverse spacing (in analogy with optics), we get

$$r \equiv \frac{1}{\delta d} \sim \frac{M_{BPS}}{e^{\sqrt{\frac{8}{3}d}}} = N,$$
(5.8)

where in the last equality we have used (4.7). This equation gives us a notion of how the resolution in moduli space scales with the size of a large black hole whose attractor point is at a distance d in moduli space. Taking d to be a constant, we see that the resolution increases with the amount of energy (black hole mass) at one's disposal. This makes sense, as higher masses and bigger black holes naturally mean higher charges and so the "mesh" of points in moduli space becomes smaller. Furthermore, keeping the mass of the black hole fixed, we see that the resolution will decrease exponentially as we try to explore farther points in moduli space. This fits naturally with previous discussions in [25] relating the Distance Conjecture to the Bekenstein bound and finiteness of quantum gravity amplitudes. The number of states that quantum gravity admits in a box should be finite, and bounded by the area of the box. This means that it should not be possible to construct distinguishable states which probe infinite swaths of moduli space with arbitrarily large resolution in a box of given size. The resolution of this puzzle is, precisely, that the resolution drops quickly and makes far away points indistinguishable without increasing the size of the box.

6 Conclusions

Although the physics of moduli spaces is arguably one of the most important aspects of the Swampland program and string compactifications, the question of how these moduli spaces could be probed in practice, if one was found, has received comparatively little attention. In this short note we have shown how, in 4d $\mathcal{N} = 2$ theories, asymptotic properties of moduli spaces are encoded in black hole solutions in a possibly very far away vacuum in moduli space, finding that quantitative features of the prepotential can be deduced from measurements of BPS degeneracies with appropriate charges. On top of this, the 4d $\mathcal{N} = 2$ BPS states are able to reach arbitrarily large regions in moduli space, at a finite energy cost.

We also studied the quality of the moduli space picture provided by the BPS black holes — how far in moduli space can we go, and with which resolution, given finite resources. While this question is in general a complicated optimization problem, we studied a couple of one-parameter families of black holes, finding that both the distance goes logarithmically with the mass of the BPS black hole and that the resolution of the probing (which is finite due to charge quantization) depend linearly on the size of the black hole. This is in general agreement with the findings in [3, 4], and shows that studying a transplanckian field range is possible in gravity, but takes an exponential amount of resources. It would be interesting to explore potential connections between this finding and more information-theoretic approaches to the Distance Conjecture recently put forth in [26, 27].

It is tantalizing that the distance conjecture is also an exponential relation between the mass of the tower and distance in moduli space. However, that involves the mass going exponentially down in distance, unlike the BPS mass that we need to probe such distances, which increases exponentially with distance. It would be interesting to see if there is a relation between these two facts.

One outstanding question is how to generalize our analysis to setups where supersymmetry is not protecting the answer, such as non-supersymmetric string theories or even questions involving hypermultiplets in 4d $\mathcal{N} = 2$ theories, for which the attractor mechanism does not provide protection. Although it is likely that qualitative different ingredients are needed, we suspect that the basic point — that black holes are appropriate probes of the moduli space — is likely to apply, too.

Finally, the perspective we have taken in this manuscript is reminiscent of the moduli space holography picture of [28, 29]. In that reference, it was proposed that the bulk of moduli space could be reconstructed from asymptotic data; in our setup, we have done the reverse, studying asymptotic regions from a bulk point in moduli space. And much as in the setup of [21, 22], we have a one-to-one mapping between moduli space and physical space, sourced by the gradients of the fields. As these gradients can in principle be studied via ordinary holography, our perspective may help bridge the gap between these two disparate notions of holography.

Acknowledgments

We thank Max Wiesner and Angel Uranga for helpful discussions and comments on the manuscript. The work of MD is supported by the FPI grant no. FPI SEV-2016-0597-19-3 from Spanish National Research Agency from the Ministry of Science and Innovation. MD also wishes to acknowledge the hospitality of the Department of Physics of Harvard University during the development of this work. The work of MM and CV is supported by a grant from the Simons Foundation (602883,CV) and by the NSF grant PHY-2013858. The authors acknowledge the support of the Spanish Agencia Estatal de Investigacion through the grant "IFT Centro de Excelencia Severo Ochoa" SEV-2016-0597.

A Asymptotic mass formulae for the mirror quintic

In this section we will particularize the general discussion in this paper to a specific model, namely the mirror quintic M. This is a Calabi-Yau threefold with hodge numbers $h^{1,1} =$ 101 and $h^{2,1} = 1$. One can define it by considering the following quotient in \mathbb{P}^4 :

$$M = \left(\Sigma_i z_i^5 - 5\psi \Pi_i z_i\right) / \mathbb{Z}_5^3.$$
(A.1)

The four periods of the mirror quintic were famously studied in [19]. In particular, they can be combined into a period vector with respect to an integer symplectic basis (A^i, B_j) of $H^3(M, \mathbb{Z})$, which in the large complex structure limit, $\psi \to \infty$, take the form [19]:

$$\Pi = \begin{pmatrix} F_0 \\ F_1 \\ X^0 \\ X^1 \end{pmatrix} \xrightarrow{\psi \to \infty} \Pi_{LCSL} = \left(\frac{2\pi i}{5}\right)^3 \begin{pmatrix} \frac{5}{6}t^3 + \frac{25}{12}t \\ -\frac{5}{2}t^2 - \frac{1}{2}t \\ 1 \\ t \end{pmatrix}, \text{ with } t = -\frac{5}{2\pi i}\log(5\psi). \quad (A.2)$$

The Kähler potential at the large complex structure point is thus given by

$$e^{-\mathcal{K}}|_{LCSL} = \frac{32\pi^3 \log^3(5|\psi|)}{75}$$
 (A.3)

Now, we will solve the attractor equations (2.8) in a slightly different way than in the main text; rather than the choice of charges in (3.6), we will use the choice of charges that exactly leads to the attractor values in (A.2), in a gauge where $C_h = N(\frac{2\pi i}{5})^{-3}$ and writing t = x + iy. One immediately obtains, in the limit $y \to \infty$:

$$Q_{\infty} = \begin{cases} p^{0} = N \\ p^{1} = Nx \\ q_{0} = -\frac{5}{2}Nxy^{2} \\ q_{1} = \frac{5}{2}Ny^{2} \end{cases}$$
(A.4)

From (A.3), one obtains the distance in moduli space as $d = \sqrt{\frac{3}{2} \log y}$. Finally, we obtain the entropy from (2.9) which, at leading order in y, is:

$$S \sim N^2 y^3 \sim N^2 e^{\sqrt{6}d} \,. \tag{A.5}$$

This agrees with (4.5), with a different exponent since we picked a different set of charges.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited. SCOAP³ supports the goals of the International Year of Basic Sciences for Sustainable Development.

References

- H. Ooguri and C. Vafa, On the geometry of the string landscape and the swampland, Nucl. Phys. B 766 (2007) 21 [hep-th/0605264] [INSPIRE].
- [2] A. Nicolis, On super-Planckian fields at sub-Planckian energies, JHEP 07 (2008) 023
 [arXiv:0802.3923] [INSPIRE].
- [3] M.J. Dolan, P. Draper, J. Kozaczuk and H. Patel, Transplanckian censorship and global cosmic strings, JHEP 04 (2017) 133 [arXiv:1701.05572] [INSPIRE].
- [4] P. Draper and S. Farkas, Transplanckian censorship and the local swampland distance conjecture, JHEP 01 (2020) 133 [arXiv:1910.04804] [INSPIRE].

- [5] S. Ferrara, R. Kallosh and A. Strominger, N = 2 extremal black holes, *Phys. Rev. D* **52** (1995) R5412–R5416 [hep-th/9508072] [INSPIRE].
- S. Ferrara and R. Kallosh, Supersymmetry and attractors, Phys. Rev. D 54 (1996) 1514 [hep-th/9602136] [INSPIRE].
- S. Ferrara and R. Kallosh, Universality of supersymmetric attractors, Phys. Rev. D 54 (1996) 1525 [hep-th/9603090] [INSPIRE].
- [8] S. Ferrara, G.W. Gibbons and R. Kallosh, Black holes and critical points in moduli space, Nucl. Phys. B 500 (1997) 75 [hep-th/9702103] [INSPIRE].
- [9] G.W. Moore, Arithmetic and attractors, hep-th/9807087 [YCTP-P17-98] [INSPIRE].
- [10] F. Denef, B.R. Greene and M. Raugas, Split attractor flows and the spectrum of BPS D-branes on the quintic, JHEP 05 (2001) 012 [hep-th/0101135] [INSPIRE].
- [11] B.R. Greene, String theory on Calabi-Yau manifolds, in the proceedings of Theoretical Advanced Study Institute in Elementary Particle Physics (TASI '96): fields, strings, and duality, (1996), p. 543 [hep-th/9702155] [INSPIRE].
- [12] L.E. Ibáñez and A.M. Uranga, String theory and particle physics: an introduction to string phenomenology, Cambridge University Press (2012) [INSPIRE].
- [13] P. Candelas and X. de la Ossa, Moduli space of Calabi-Yau manifolds, Nucl. Phys. B 355 (1991) 455 [INSPIRE].
- [14] J. Polchinski, String theory. Volume 2: superstring theory and beyond, Cambridge University Press (2007) [INSPIRE].
- [15] F. Denef, Attractors at weak gravity, Nucl. Phys. B 547 (1999) 201 [hep-th/9812049]
 [INSPIRE].
- [16] F. Denef, Supergravity flows and D-brane stability, JHEP 08 (2000) 050 [hep-th/0005049]
 [INSPIRE].
- [17] S. Cecotti, Special geometry and the swampland, JHEP 09 (2020) 147 [arXiv:2004.06929]
 [INSPIRE].
- [18] H. Ooguri, A. Strominger and C. Vafa, Black hole attractors and the topological string, Phys. Rev. D 70 (2004) 106007 [hep-th/0405146] [INSPIRE].
- [19] P. Candelas, X.C. De La Ossa, P.S. Green and L. Parkes, A pair of Calabi-Yau manifolds as an exactly soluble superconformal theory, Nucl. Phys. B 359 (1991) 21 [INSPIRE].
- [20] P. Candelas, P. Kuusela and J. McGovern, Attractors with large complex structure for one-parameter families of Calabi-Yau manifolds, JHEP 11 (2021) 032 [arXiv:2104.02718]
 [INSPIRE].
- [21] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, The EFT stringy viewpoint on large distances, JHEP 09 (2021) 197 [arXiv:2104.05726] [INSPIRE].
- [22] R. Angius et al., At the end of the world: local dynamical cobordism, JHEP 06 (2022) 142
 [arXiv:2203.11240] [INSPIRE].
- [23] Q. Bonnefoy, L. Ciambelli, D. Lüst and S. Lüst, Infinite black hole entropies at infinite distances and tower of states, Nucl. Phys. B 958 (2020) 115112 [arXiv:1912.07453]
 [INSPIRE].

- [24] D. Lüst, E. Palti and C. Vafa, AdS and the swampland, Phys. Lett. B 797 (2019) 134867
 [arXiv:1906.05225] [INSPIRE].
- [25] Y. Hamada, M. Montero, C. Vafa and I. Valenzuela, Finiteness and the swampland, J. Phys. A 55 (2022) 224005 [arXiv:2111.00015] [INSPIRE].
- [26] J. Stout, Infinite distance limits and information theory, arXiv:2106.11313 [INSPIRE].
- [27] J. Stout, Infinite distances and factorization, arXiv: 2208.08444 [INSPIRE].
- [28] T.W. Grimm, Moduli space holography and the finiteness of flux vacua, JHEP 10 (2021) 153 [arXiv:2010.15838] [INSPIRE].
- [29] T.W. Grimm, J. Monnee and D. van de Heisteeg, Bulk reconstruction in moduli space holography, JHEP 05 (2022) 010 [arXiv:2103.12746] [INSPIRE].

Published for SISSA by 🖄 Springer

RECEIVED: October 23, 2023 ACCEPTED: December 14, 2023 PUBLISHED: January 2, 2024

Emergence of species scale black hole horizons

José Calderón-Infante,^a Matilda Delgado^b and Angel M. Uranga^b

^aTheoretical Physics Department, CERN,

CH-1211 Geneva 23, Switzerland

^bInstituto de Física Teórica IFT-UAM/CSIC,

C/ Nicolás Cabrera 13-15, Campus de Cantoblanco, 28049 Madrid, Spain

E-mail: jose.calderon-infante@cern.ch, matilda.delgado@uam.es, angel.uranga@csic.es

ABSTRACT: The scale at which quantum gravity becomes manifest, the species scale Λ_s , has recently been argued to take values parametrically lower than the Planck scale. We use black holes of vanishing horizon area (small black holes) in effective field theories coupled to quantum gravity to shed light on how the three different physical manifestations of the species scale Λ_s relate to each other. (i) Near the small black hole core, a scalar field runs to infinite distance in moduli space, a regime in which the Swampland Distance Conjecture predicts a tower of exponentially light states, which lower Λ_s . (ii) We integrate out modes in the tower and generate via Emergence a set of higher derivative corrections, showing that Λ_s is the scale at which such terms become relevant. (iii) Finally, higher derivative terms modify the black hole solution and grant it a non-zero, species scale sized stretched horizon of radius Λ_s^{-1} , showcasing the species scale as the size of the smallest possible black hole describable in the effective theory.

We present explicit 4d examples of small black holes in 4d $\mathcal{N} = 2$ supergravity, and the 10d example of type IIA D0-branes. The emergence of the species scale horizon for D0-branes requires a non-trivial interplay of different 8-derivative terms in type IIA and M-theory, providing a highly non-trivial check of our unified description of the different phenomena associated to the species scale.

KEYWORDS: String and Brane Phenomenology, Black Holes in String Theory, D-Branes

ARXIV EPRINT: 2310.04488



Contents

Inti	roduction	1
Warmup example: 4d small black holes		4
2.1	4d $\mathcal{N} = 2$ supergravity, attractors and higher derivative corrections	5
	2.1.1 $\mathcal{N} = 2$ supergravity black holes	5
	2.1.2 Higher derivative corrections and quantum attractors	6
2.2	Small black holes, species scale horizons and the SDC tower	7
	2.2.1 An illustrative class of stretched small black holes	7
	2.2.2 The species scale horizon	8
	2.2.3 The SDC tower and higher derivative emergence	9
D0-	branes and the species scale horizon	11
3.1	Microscopic derivation of the entropy and of the species scale	12
3.2	Hot D0-branes	14
Spe	ccies scale horizon from emergent higher derivative terms	16
4.1	Entropy function analysis of general \mathbb{R}^4 terms	16
	4.1.1 The D0-brane solution at two-derivative level	16
	4.1.2 The D0-brane solution and higher derivative terms	18
	4.1.3 The species scale and cosmic censorship	20
4.2	The D0-brane SDC tower and higher derivative emergence	21
4.3	Supersymmetric 10d R^4 terms are not enough	23
4.4	Including F_2 : species scale D0-brane horizon from M-theory couplings	24
Cor	nclusions	27
Ten	sor structure of R^4 corrections	30
	Int: Wai 2.1 2.2 D0- 3.1 3.2 Spe 4.1 4.2 4.3 4.4 Cor Ten	Introduction Warmup example: 4d small black holes 2.1 4d $\mathcal{N} = 2$ supergravity, attractors and higher derivative corrections 2.1.1 $\mathcal{N} = 2$ supergravity black holes 2.1.2 Higher derivative corrections and quantum attractors 2.2 Small black holes, species scale horizons and the SDC tower 2.2.1 An illustrative class of stretched small black holes 2.2.2 The species scale horizon 2.2.3 The SDC tower and higher derivative emergence D0-branes and the species scale horizon 3.1 Microscopic derivation of the entropy and of the species scale 3.2 Hot D0-branes Species scale horizon from emergent higher derivative terms 4.1 Entropy function analysis of general R^4 terms 4.1.1 The D0-brane solution at two-derivative level 4.1.2 The D0-brane solution and higher derivative terms 4.1.3 The species scale and cosmic censorship 4.2 The D0-brane SDC tower and higher derivative emergence 4.3 Supersymmetric 10d R^4 terms are not enough 4.4 Including F_2 : species scale D0-brane horizon from M-theory couplings Conclusions Tensor structure of R^4 corrections

1 Introduction

Recent activity in the Swampland Program (see [1–4] for reviews) is providing strong support that the cutoff of an effective field theory (EFT) consistent with quantum gravity is not the Planck scale, but potentially a much lower one, the species scale Λ_s [5–9]. This implies that objects which can be reliably described in the EFT have a lower bound in their size of order Λ_s^{-1} . This is a very profound statement since it leads to quantum gravity effects at length scales which can potentially be parametrically larger than the Planck scale. This happens for instance near infinite distance points in moduli spaces, where the (Swampland) Distance Conjecture (SDC) [10] predicts the appearance of an infinite tower of light particles. Additionally, bounds on how fast these towers and the species scale become light have been proposed and tested in [11–14]. A natural setup in which to explore and sharpen these ideas is in black hole physics. In particular, charged extremal black holes with scalar dependent gauge kinetic functions have an attractor mechanism [15–18] that fixes the vevs of such scalars at the horizon. By appropriately choosing the charges of the black hole, one can tune the point of moduli space that is probed by the horizon. A particular choice of charges leads to small black holes, which classically have zero horizon area (i.e. a singularity), and for which scalars run off to infinite distance as one approaches the black hole core (see [19–21] for other uses of small black holes to explore swampland constraints and [20–29] for other spacetime-dependent configurations probing infinite distances in field space). One would then expect that quantum gravity effects smooth out the singularity by generating a stretched horizon of small but finite size. According to our discussion above, this size should be given by the species scale Λ_s^{-1} (hence parametrically larger than the Planck length). Indeed, it was argued in [19] that in order for small black hole solutions not to violate entropy bounds, they should acquire a finite size, of the order of the UV cut-off.

Black holes have been used at multiple occasions in the literature to probe the species scale [30, 31]. And conversely, the species scale itself is often defined as being the size of the smallest possible black hole (SPBH) that one can reliably describe within an EFT description [5, 32]. One of our objectives is to study how quantum gravitational effects at the species scale come to modify small black hole solutions and ultimately promote them to these non-singular, finite-sized SPBHs, in perfect agreement with the works of [19].

The appearance of stretched horizons for small black holes has long been explored in the context of string theory, e.g. for supersymmetric (and non-supersymmetric) charged black holes in 4d $\mathcal{N} = 4, 2$ supersymmetric compactifications, by the inclusion of higher derivative corrections, resulting in string scale stretched horizons.¹ Actually this corresponds in those examples to precisely the species scale in the infinite distance limit of the corresponding small black hole, in agreement with our discussion above (see [30, 31] for related works).

In the above discussion, the appearance of the species scale associated to the SDC tower seems accidental. We will however argue that there is a direct link, by showing that integrating out the SDC tower produces, in the spirit of Emergence [35–38] (see [39–43] for recent discussions), the higher curvature terms which ultimately lead to the stretched horizon, which is thus naturally set by the species scale. Hence, we provide an explicit microscopic mechanism for the emergence of species scale horizons for this class of small black holes.² Our result moreover matches the approach in [47], where the species scale has been argued to be the scale at which higher curvature terms become relevant.

Hence, our work allows us to bring together three definitions of the species scale: the original definition accounting for the number of species in the SDC tower, the scale defining the size of SPBHs, and the scale of higher curvature corrections. The central idea is that small black hole solutions get resolved as SPBHs at the species scale, by including the effect of higher derivative corrections emerging from integrating out the SDC tower. Given that our main tool is small black holes, our results hold in asymptotic regimes in moduli space;

¹For a complementary viewpoint, see [33, 34].

²It would be interesting to explore connections with other proposals to produce stretched horizons parametrically larger thank the Planck scale for small black holes, such as the fuzzball proposal (see [44-46] for reviews).

it would be interesting to extend our understanding to test recent proposal involving the species scale in the interior of moduli space [30, 47, 48].

We expect that the emergence of species scale stretched horizons should hold beyond 4d systems. In fact, the main point of this work is to explore this question for D0-brane solutions of 10d type IIA. These behave as 10d charged small black holes, of classically zero size, and at whose core the dilaton runs off to infinitely strong coupling. Swampland considerations imply that a species scale horizon should arise. However, despite the fact that systems of N D0-branes have been under scrutiny for decades from diverse perspectives, including supersymmetric quantum mechanics [49, 50], M(atrix) theory [51], holography [52], their combination [53], and MonteCarlo simulations (see [54] for a recent update), the only discussion about D0-brane horizons seems to have appeared in [55].

We will provide several arguments for the existence of this species scale horizon for D0brane solutions; our arguments include microstate counting, finite temperature considerations and the analysis of higher curvature terms using (a 10d version of) the entropy function. On the other hand, we compute the familiar higher curvature R^4 terms arising from emergence by integrating out the SDC tower of BPS particles (which are D0-branes themselves) and show that it does not suffice to generate the horizon. Hence, the appearance of the species scale horizon implied by Swampland considerations demands the inclusion of further higher derivative corrections. Indeed, we show that one indeed gets a finite size species scale horizon once we include 8-derivative terms involving curvatures and RR 2-form field strength, a computation which we carry out using a lift to 11d. This also nicely dovetails with the fact that the species scale is the 11d Planck scale.

We regard this as impressive evidence of the non-trivial power of the species scale proposal, which provides a rationale for the presence of these extra terms (which are otherwise largely ignored in the literature), and hence probes deep UV properties of M-theory.

One general aspect in our analysis is that the limited knowledge of higher derivative corrections implies that the computations can be carried out including only the leading terms. However, the scales probed, and in particular the species scale, are such that in principle the whole set of higher derivative terms contribute in comparable amounts (hence, the use of *leading* is misleading). The key point however is that the main effect of the correction is to turn the singular behaviour of the two-derivative approximation into a smoothed out behaviour controlled by the species scale, and it suffices to truncate the infinite set of higher derivative terms to a tractable finite subset which captures the essence of this change. namely the presence of the species scale horizon, with the expectation that the remaining terms modify only order 1 numerical factors, but not the parametric dependence producing the species scale. In fact, in specific setups, such as 4d small black holes, this has been substantiated using scaling and other arguments [56, 57]. We expect this lesson to apply to other setups as well, including the 10d type IIA D0-branes. For this case, we do not have a clear argument why a truncation of the higher derivative terms suffices to capture the existence and parametric dependence of the stretched horizon. It would be interesting to argue for this, for instance by looking for general scaling properties of the higher derivative terms in the type IIA effective action with the dilaton.

The paper is organized as follows. In section 2 we consider small black holes in 4d $\mathcal{N} = 2$. Section 2.1 reviews the attractor mechanism, including higher curvature corrections, and in
section 2.2 we study a class of 4d small black holes, the appearance of a species scale horizon from the higher curvature corrections, and the emergence of the latter from integrating out an SDC tower of KK gravitons of M-theory on \mathbf{S}^1 (times the CY₃). In section 3 we turn to the case of 10d type IIA D0-branes, regarded as small black holes. In section 3.1 we carry out a microscopic computation of the entropy of the system of N D0-branes and of the appearance of the species scale scaling as $N^{1/2}$. Section 3.2 presents a complementary discussion of the computation of the entropy from the perspective of the finite temperature deformation of the D0-brane supergravity solution.

In section 4 we study the appearance of the horizon from higher derivative terms in the action. In section 4.1 we perform an entropy function analysis of the spacetime solution, and show that general R^4 corrections can lead to the appearance of a species scale stretched horizon matching the microscopic results. In section 4.2 we consider the emergence of certain supersymmetric R^4 terms from integrating out the SDC tower in the limit explored by the solution (which is a D0-brane tower, or equivalently M-theory KK gravitons). In section 4.3 we apply the entropy function to these terms and show they still do not lead to the appearance of the horizon. Finally, in section 4.4 we show that including terms involving the RR 2-form field strength one recovers the species scale horizon for the system. We offer some final thoughts and future prospects in section 5.

2 Warmup example: 4d small black holes

In this section, we illustrate our picture of the diverse appearances of the species scale using small black holes in 4d $\mathcal{N} = 2$ EFTs obtained from compactifications of type IIA theory on a Calabi-Yau threefold. As is well known, there are large classes of BPS black hole solutions displaying the attractor mechanism [15]: vector multiple moduli vary along the radial direction, and, in the $AdS_2 \times S^2$ near-horizon geometry, attain values which are fixed in terms of the black hole charges and are independent of the asymptotic value of the moduli (while hypermultiplets remain fixed all along the flow). Small black holes correspond to solutions where the attractor mechanism drives the scalars to infinite distance in field space, leading to a horizon of formally zero size and to a singularity. As has been studied extensively in the literature (see [57, 58] for reviews), higher curvature corrections generically lead to a stretched horizon and a finite entropy, matching microscopic computations in many classes of examples. In section 2.1, we summarize these developments and describe the case of a specific class of small black hole solutions in full detail.

Our new angle is discussed in section 2.2, where we argue that the Swampland Distance Conjecture infinite tower of states becoming light in the infinite distance limit corresponding to small black holes is precisely responsible, via emergence, for the appearance of higher derivative terms producing the stretched horizon. We thus link the appearance of the species scale as derived from the SDC tower, as controlling higher derivative corrections, and as setting the size of the smallest possible black hole in the effective field theory. It is amusing to see that swampland ideas play such a central role in the almost three decade-long success story of microstate explanation of black hole entropy.³

³For other approaches to the Distance Conjecture and Black Holes, see [19-21, 28, 30, 59, 60]. For the interplay of higher curvature corrections and swampland constrains, mainly the Weak Gravity Conjecture [61], see e.g. [62-64].

2.1 4d $\mathcal{N} = 2$ supergravity, attractors and higher derivative corrections

2.1.1 $\mathcal{N} = 2$ supergravity black holes

Type II string theory on a Calabi-Yau threefold gives 4d $\mathcal{N} = 2$ supergravity theories. We will focus on the physics in vector multiplet moduli space, as hypermultiplets are inert. Including the graviphoton, there are $n_V + 1$ gauge bosons, labelled by $I = 0, \ldots, n_V$. The structure of the bosonic Lagrangian is

$$\mathcal{L} = R - 2g_{i\bar{j}}\partial_{\mu}z^{i}\partial^{\mu}\bar{z}^{\bar{j}} + \operatorname{Im}\mathcal{N}_{IJ}F^{I}_{\mu\nu}F^{J\,\mu\nu} + \operatorname{Re}\mathcal{N}_{IJ}F^{I}_{\mu\nu}\frac{\epsilon^{\mu\nu\rho\sigma}}{2\sqrt{-g}}F^{J}_{\rho\sigma}, \qquad (2.1)$$

where the different quantities are defined using special geometry (see [65] for reference). The scalars are parametrized by a set of projective coordinates X^{I} , and the Kähler potential is determined by the prepotential F(X), a holomorphic function of degree 2, as

$$\mathcal{K} = -\log i \left(\bar{X}^I F_I - X^I \bar{F}_I \right), \quad \text{with } F_I = \partial_I F.$$
(2.2)

The gauge kinetic functions \mathcal{N}_{IJ} are also determined by special geometry, but we will skip its detailed structure.

The microscopic description in type IIA theory is as follows. There are $h_{1,1}$ vector multiplets whose gauge bosons arise from the 10d RR 3-form integrated over 2-cycles ω_i , $i = 1, \ldots, h_{1,1}$. The Kähler moduli, complexified with the integrals of the NSNS 2-form over the 2-cycles, give rise to the vector moduli; morally, the affine coordinates Z^i . The corresponding $F_i \equiv \partial_i F$ are associated to the dual 4-cycles.

The structure of the prepotential in the large volume regime is

$$F_0(X) = -\frac{1}{6}C_{ijk}\frac{X^i X^j X^k}{X^0}, \qquad (2.3)$$

where C_{ijk} are the integer triple intersection numbers of the Calabi-Yau. Away from the large volume limit, the prepotential receives corrections from worldsheet instantons. Also, as discussed later, certain higher derivative corrections can also be encoded in a corrected prepotential.

The theory contains BPS black holes constructed out of D-branes wrapped on holomorphic cycles in the CY. There is a vector of electric and magnetic charges $\Gamma = (q_I, p^I)$. The central charge is

$$\mathcal{Z} = e^{\mathcal{K}/2} \left(q_I X^I - F_I p^I \right) \,. \tag{2.4}$$

Regular black holes have a near-horizon $AdS_2 \times S^2$ geometry. The values of the moduli at the horizon and the entropy are fixed by the attractor equations

$$\partial_i \mathcal{Z}(X^I, \bar{X}^I, q_I, p^I)|_h = 0, \qquad S = \pi |\mathcal{Z}(q_I, p^I)|_h, \qquad (2.5)$$

where the subindex h indicates the value at the horizon. The solution can be expressed as

$$p^{I} = \operatorname{Re}[C_{h}X_{h}^{I}] = C_{h}X_{h}^{I} + \bar{C}_{h}\bar{X}_{h}^{I},$$

$$q_{I} = \operatorname{Re}[C_{h}F_{h\,I}] = C_{h}F_{h\,I} + \bar{C}_{h}\bar{F}_{h\,I},$$
(2.6)

where we have introduced $C = -2i\bar{\mathcal{Z}}e^{\mathcal{K}/2}$.

In the type IIA setup, the charges q_0 and q_i correspond⁴ to D0-branes, and D2-branes on 2-cycles, while the charges p^0 and p^i correspond to D6-branes on the CY and D4-branes on 4-cycles.

A celebrated class (given its simple lift to M-theory [66]) is obtained by considering sets of D4-branes wrapped p_i times on the i^{th} 4-cycle and a number q_0 of D0-branes, for $|q_0| \gg p^i \gg 1$. Its attractor behaviour can be easily analyzed in the large volume limit (2.3), see [30] for a recent application. The attractor values for moduli, and the entropy, are (we introduce $q \equiv -q_0$)

$$X^{0} = -\frac{1}{2}\sqrt{\frac{\frac{1}{6}C_{ijk}p^{i}p^{j}p^{k}}{q}}, \qquad X^{i} = -\frac{i}{2}p^{i}, \qquad S = 2\pi\sqrt{\frac{1}{6}q C_{ijk}p^{i}p^{j}p^{k}}.$$
 (2.7)

The CY volume modulus at the horizon is thus

$$\mathcal{V}_h = \sqrt{\frac{q^3}{\frac{1}{6}C_{ijk}p^i p^j p^k}} \,. \tag{2.8}$$

Note that we need $q \gg p^i$ for a reliable large volume expansion. In section 2.2 we exploit this class of models to build small black holes and discuss its corrections.

2.1.2 Higher derivative corrections and quantum attractors

The discussion of higher derivative corrections to the attractor mechanism has been studied extensively in the literature. In particular, there is a class of $\mathcal{N} = 2 R^2 F^{2g-2}$ F-term corrections which are computed by the topological string, and whose effects on black holes can be included systematically [67–69], see also [58, 70] for reviews. Following these references, one uses the Weyl superfield $W_{\mu\nu} = F^+_{\mu\nu} - R^+_{\mu\nu\lambda\rho}\theta\sigma^{\lambda\rho}\theta + \ldots$, whose lowers component is the (self-dual piece of the) graviphoton, and defines $\Upsilon = W^2$. The F-term corrections can be encoded in a degree 2 generalized prepotential

$$F(X^{I},\Upsilon) = \sum_{g=0}^{\infty} F_{g}(X^{I})\Upsilon^{g}.$$
(2.9)

The lowest term F_0 corresponds to the usual prepotential, and higher F_g 's are the genus g topological string amplitude. In the presence of these corrections, the attractor equations and the entropy are modified so we have

$$p^{I} = i(X_{h}^{I} - \bar{X}_{h}^{I}), \qquad q_{I} = i(F_{I}(X, \Upsilon)_{h} - \overline{F}_{I}(\overline{X}, \overline{\Upsilon}_{h})),$$

$$\Upsilon|_{h} = -64, \qquad \qquad S = \pi(|\mathcal{Z}_{h}|^{2} + 4\mathrm{Im}(\Upsilon\partial_{\Upsilon}F)_{h}). \qquad (2.10)$$

The structure of corrections near the large volume limit is

$$F(X,\Upsilon) = -\frac{1}{6}C_{ijk}\frac{X^{i}X^{j}X^{k}}{X^{0}} + d_{i}\frac{X^{i}\Upsilon}{X^{0}}, \qquad (2.11)$$

⁴One should take into account induced D-brane charges, e.g. a single D4-brane wrapped on a K3 has -1 units of induced D0-brane charge due to Chern-Simons couplings to background curvature.

where the C_{ijk} are the triple-intersection numbers of the Calabi-Yau and the d_i are given in terms of the second Chern classes of the Calabi-Yau by

$$d_i = -\frac{1}{24} \frac{1}{64} \int_{\mathbf{X}_6} c_2(T\mathbf{X}_6) \wedge \omega_i$$
 (2.12)

with ω_i a basis of $H^{1,1}(\mathbf{X}_6)$. While these corrections are present for general black holes, for regular ones they are suppressed with respect to the two-derivative result. However, they are crucial for small black holes for which the classical piece vanishes, so that the extra pieces produce a non-zero entropy, namely a stretched horizon. In the following we consider a particular class of small black holes, and show that the leading correction F_1 , which contain 4-derivative R^2 corrections, produce a stretched horizon. In section 2.2 we will show that these corrections precisely follow from the SDC tower and that the horizon size is controlled by the species scale.

We would like to point out that the results can also be derived in the formalism of minimization of the entropy function in an $AdS_2 \times S^2$ ansatz near horizon geometry [71]. The added terms in the Lagrangian encoded in the corrected prepotential contribute to a modified entropy function. For small black holes, there is a run away but once the corrections are included, the entropy function can actually be minimized, signaling the presence of a horizon. This approach will be exploited in the 10d setup in section 4.1.

2.2 Small black holes, species scale horizons and the SDC tower

2.2.1 An illustrative class of stretched small black holes

We now consider a specific class of models, based on a 2-modulus version of the large volume expansion (2.3). We take the prepotential

$$F = -\frac{X^1 (X^2)^2}{X^0}, \qquad (2.13)$$

where we have chosen $C_{122} = 6$ for simplicity. This is a simple template that can model an internal space \mathbf{X}_6 given by $\mathrm{K3} \times \mathbf{T}^2$ (for which extensive studies of small black holes and stretched horizons have been carried out, see [57] for a review), or K3-fibered CY threefolds. The affine coordinates $Z^1 = X^1/X^0$ and $Z^2 = X^2/X^0$ correspond to the sizes of the \mathbf{T}^2 (or the base \mathbf{P}_1 of the K3 fibration) and of the K3 (fiber), respectively.

We can easily build small black holes with a choice of non-zero charges $p \equiv p^1$, $q \equiv -q_0$, and setting $q_i, p^0, p^2 = 0$. The central charge, in terms of $t_i = \text{Im}Z^i$, is

$$\mathcal{Z} = \frac{q + p(t_2)^2}{2\sqrt{2} (t_1)^{1/2} t_2} \,. \tag{2.14}$$

The attractor equations imply that t_2 is stabilized at the horizon at

$$t_2 = \sqrt{\frac{q}{p}},\tag{2.15}$$

which is still in the large volume regime if $q \gg p$, as we assume in the following. The attractor mechanism also leads to $t_1 \to \infty$, so this scalar runs off to infinite distance in moduli space. Consequently the overall volume in string units at the horizon also diverges:

$$e^{-\mathcal{K}_h} |X_h^0|^{-2} = 8\mathcal{V}_h = 8t_1 t_2^2 \to \infty,$$
 (2.16)

where the $|X_h^0|^{-2}$ factor accounts for the choice of projective coordinates on the special Kähler manifold such that \mathcal{V} is independent of the Kähler gauge. Using the formulas (2.5), the area of the horizon and therefore the entropy vanish:

$$S = \lim_{\substack{t_1 \to \infty \\ t_2 = \sqrt{\frac{q}{p}}}} \pi |\mathcal{Z}| \to 0.$$
(2.17)

We now show that the small black hole acquires a stretched horizon upon the inclusion of the correction of the kind (2.11). In particular, we focus on the correction d_1 , so the prepotential is

$$F = -\frac{X^1 (X^2)^2}{X^0} + d_1 \frac{X^1 \Upsilon}{X^0} , \qquad (2.18)$$

where the last two terms correspond to F_1 , the genus 1 topological string amplitude. This corresponds to the class of solutions considered in [30] with $p^i = 0$, $i \neq 1$ (see also [57] for a related class).

The moduli at the horizon and entropy are given by

$$Z^{1} = i \frac{\sqrt{pq}}{2\sqrt{d_{1}\Upsilon_{h}}}, \qquad Z^{2} = 0, \qquad S \sim 4\pi\sqrt{d_{1}\Upsilon_{h}pq}.$$
(2.19)

We see that the horizon no longer explores an infinite distance in moduli space and that the entropy is non-vanishing. Hence the solution develops a stretched horizon that cloaks up the singularity. The overall volume at the stretched horizon is modified by the curvature corrections (see [58] and references therein) and is given by:

$$e^{-\mathcal{K}_h} |X_h^0|^{-2} = 8\mathcal{V}_h \sim \frac{q(t_1)_h}{p} \sim \frac{q^{3/2}}{\sqrt{p \, d_1 \Upsilon_h}},$$
 (2.20)

which is finite, and still in the large volume regime if $q \gg p$.

Recall that the above result could have been obtained using the entropy function formalism, but we skip its explicit discussion. Rather we turn to the discussion of the appearance of the higher derivative correction from the SDC tower.

2.2.2 The species scale horizon

In a *d*-dimensional EFT weakly coupled to Einstein gravity, the species scale is given by:

$$\Lambda_s \sim \frac{M_d}{N_s^{\frac{1}{d-2}}},\tag{2.21}$$

where N_s is the number of species below the species scale and M_d is the *d*-dimensional Planck mass. In EFTs obtained from string theory (or compactifications thereof), the number of light species changes as one moves along the moduli space. Thus, the species scale is dependent on the region of moduli-space under consideration. In particular, in infinite distance limits in moduli space where infinite towers of states are becoming exponentially light [10], the species scale falls as the number of light species increases drastically. In order to determine the species scale, we must therefore study what infinite distance limit the (formerly) small black hole was probing.

Recall that the small black holes described in the previous section probes the infinite distance limit $t_1 \to \infty$, with t_2 fixed. Note that in this case, the overall volume modulus (2.16) $\mathcal{V} \sim t_1(t_2)^2$ also goes to infinity at the horizon, where \mathcal{V} is the volume of the CY \mathbf{X}_6 in string units. At fixed 4d Planck scale $M_p^2 = M_s^2 \mathcal{V}/g_s^2$, the large \mathcal{V} limit corresponds to a large g_s limit in string variables (a similar conclusion follows from the fact that the 4d dilaton lies in a hypermultiplet, so it is constant along the attractor flow, hence \mathcal{V}/g_s^2 remains fixed). Hence, the limit of our interest correspond to a large g_s regime. There is therefore a tower of D0-brane particles becoming light; this also follows from the fact that the central charge for kD0-branes is given by $Z_{D0} = k(t_1)^{-1/2}(t_2)^{-1}$, which goes to zero in the limit. Actually, there may be other towers of particles becoming light, but they correspond to branes wrapped on some internal cycles, so they are suppressed in the regime $q \gg p$. The D0-brane particles thus encode the leading correction. Incidentally, we also note that the infinite distance limit explored by the singular two-derivative solution is a standard decompactification limit to M-theory compactified on \mathbf{X}_6 , rather than an emergent string limit [72, 73]. Indeed, one can show that the volume of the Calabi-Yau in M-theory units is related to the 4d dilaton and is therefore constant, signalling that there is no decompactification to 11d M-theory.

The computation of the species scale in this infinite distance limit can be done in different ways. We choose to focus on the approach of [48], where it was argued that higher derivative corrections to the EFT could be used as a proxy for determining the species scale. In this framework, the species scale acts as the energy scale that suppresses higher derivative corrections to the EFT. This recipe was applied to the case of Calabi-Yau compactifications of type II string theories to find that the species scales is given by:

$$\Lambda_s = \frac{M_4}{\sqrt{F_1}}\,,\tag{2.22}$$

where once more, F_1 is the genus 1 topological free energy. Taking (2.19) and $F_1 \sim Z^1 d_1 \Upsilon$, we see that at the horizon of the stretched black hole, we have:

$$S \sim \left(\frac{\Lambda_s}{M_4}\right)^{-2} \,. \tag{2.23}$$

The radius of the stretched horizon is therefore given exactly by the species scale.

Evaluating the species scale can also be done by counting the number of D0 states that are light in the EFT, in the spirit of [39–43], using (2.21). In the next section we will bridge these two interpretations of the species scale by showing how the higher derivative corrections to the EFT Lagrangian originate from integrating out the tower of D0 states, in the spirit of Emergence.

2.2.3 The SDC tower and higher derivative emergence

The SDC implies that the description of infinite distance limits in moduli space require the inclusion of microscopic physics, in particular an infinite tower of states whose masses decrease exponentially with the field distance [10]. On the other hand, in the previous section we have argued that the singular behaviour of the solution is cloaked by a stretched horizon if suitable higher derivative corrections are included. In this section we show that the two UV ingredients are related, and in fact the distance conjecture *implies* the appearance of higher derivative corrections of the necessary kind to generate the stretched horizon. Hence, the two explanations are two sides of the same phenomenon and link the different physical interpretations of the species scale.

The key idea is that the R^2 terms can be seen as generated by an infinite tower of 4d BPS D0-brane states, namely KK gravitons running on an S^1 compactification of the 5d theory given by M-theory on X_6 . In fact, they correspond to the leading term in the Gopakumar-Vafa expansion [74, 75] in the large volume limit (where the contribution from D2-branes is subleading).

Indeed, M-theory on \mathbf{X}_6 contains 5d BPS states which correspond to 11d gravitons on the groundstate of the \mathbf{X}_6 compactification. These are given by the K3 cohomology classes, giving an overall multiplicity of 24, times a linear dependence on the Kähler modulus of Z^1 . These 5d BPS states can run with KK momentum in the \mathbf{S}^1 compactification to 4d, leading to the 4d tower of D0-brane states.⁵

Hence the problem is just the computation of 1-loop diagram, see [76] for computations in this spirit (see also [77–82] for related computations in other setups). For later convenience, we present the general n graviton scattering amplitude in d dimensions in the worldline formalism:

$$\mathcal{A}_{d,n} = \frac{1}{n! \, \pi^{d/2}} \sum_{k} \int d^d \mathbf{p} \int_0^\infty \frac{d\tau}{\tau} \, \tau^n \, e^{-\tau \left(\mathbf{p}^2 + \frac{k^2}{R^2}\right)} \,. \tag{2.24}$$

Here τ is the worldline parameter, k is the KK momentum, and R is the \mathbf{S}^1 radius.

We want to obtain the R^2 corrections to the effective action, which are encoded in two graviton scattering. For pedagogical reasons, it is easier to start from the 6d perspective of M-theory on K3×S¹ and postpone compactifying on the \mathbf{T}^2 . Thus, we particularize to d = 6, n = 2. The above expression corresponds to the contribution of a single K3 ground state, and diverges in two ways: because of the integral and because of the infinite sum. Since the sum over momentum modes is infinite, we can perform a Poisson resummation, so that both are nicely combined. We get

$$\frac{1}{\pi^{3/2}}\mathcal{A}_{6,2} = \pi^{3/2}\tilde{K}\int_0^\infty d\hat{\tau}\,\hat{\tau}^{1/2}\sum_l e^{-\pi\hat{\tau}R^2l^2} = C\tilde{K} + \frac{\zeta(3)}{\pi R^3}\tilde{K}\,,\tag{2.25}$$

where $\hat{\tau} = \tau^{-1}$. This Poisson summation trades the KK momentum k for the winding number l of the worldline along the \mathbf{S}^1 . The only divergence is now in the l = 0 piece, which has been isolated as the first terms in the second equality; C is an unknown coefficient regularizing the divergence. Including the K3 ground state multiplicity of 24 and the \mathbf{T}^2 modulus dependence upon compactification to 4d, we have

$$\frac{1}{\pi^{3/2}}\mathcal{A}_{4,2} = 24Z^1 \left[C\tilde{K} + \frac{\zeta(3)}{\pi R^3} \tilde{K} \right], \qquad (2.26)$$

where \tilde{K} is a 4d kinematical matrix. This is of the form introduced in the prepotential (2.18), hence it is precisely of the form required to produce a stretched horizon. We note that the same 4d result (2.26) could have been obtained via a 1-loop computation in the 10d theory

⁵The D0-brane tower is actually analogous to that we will consider in the 10d context in section 4.2, as they are simply related by compactification.

compactified directly on K3× \mathbf{T}^2 , with the Z^1 prefactor arising from the momentum integral over the \mathbf{T}^2 . Hence, the 4d coupling of interest arises from higher derivative emergence.

This nicely illustrates the links among the appearance of the species scale from the SDC tower, the higher-curvature corrections and the stretching of a small black hole into the SPBH.

We finish with a comment in hindsight. It is a well-known fact that the R^2 corrections encoded in the corrected prepotential can be seen to arise from the compactification of suitable 10d R^4 terms [83]. For concreteness, focusing on the particularly simple case of compactification on $\mathbf{X}_6 = \mathbf{K}_3 \times \mathbf{T}^2$, we start from the familiar 10d supersymmetric R^4 terms, and put two curvature insertions in K3 to get a 6d R^2 term, with a prefactor of $\chi(K3) = 24$; upon further integration over \mathbf{T}^2 (which provides no background curvature) we obtain the 4d R^2 term with the Z^1 prefactor. It is easy to generalize this argument to general CY₃ compactifications and obtain the corrections⁶ (2.11) with d_i essentially determined by the second Chern classes of \mathbf{X}_6

$$d_i \sim c_{2\,i} = \int_{\mathbf{X}_6} R \wedge R \wedge \omega_i \,, \tag{2.27}$$

where ω_i are the Poincaré dual 2-forms of the basis 2-cycles. In fact, the above is the simplest way to derive the 4d R^2 corrections [83] (see [30] for a recent application). The dimensional reduction just described underlies the fact that the 4d R^2 terms can be obtained from integrating out a tower of 5d KK gravitons is related to a similar derivation of 10d R^4 terms from 11d M-theory gravitons, as we describe in section 4.2.

3 D0-branes and the species scale horizon

We now turn to the discussion of the system of N D0-branes in 10d type IIA theory. We will argue for the existence of a stretched horizon controlled by the species scale, and explore its appearance from higher derivative couplings and their interplay with emergence upon integrating out the states in the corresponding SDC tower.

We start by describing D0-brane systems as small black holes in the theory. The relevant part of the string-frame type IIA action is

$$S = \frac{1}{2\kappa^2} \int d^{10}x \sqrt{-g} \left[e^{-2\phi} \left(R + 4\partial_\mu \phi \partial^\mu \phi \right) - \frac{1}{2} |F_2|^2 \right], \qquad (3.1)$$

where $2\kappa^2 = (2\pi)^7 \alpha'^4$. The D0-brane solution is

$$ds^{2} = f(r)^{-1/2} (-dt^{2}) + f(r)^{1/2} (dx_{\perp}^{2}),$$

$$e^{(\phi - \phi_{\infty})} = f(r)^{3/4},$$

$$f(r) = 1 + \frac{\rho^{7} g_{s}^{\infty} N}{r^{7}}, \quad \text{with } \rho^{7} = (4\pi)^{\frac{5}{2}} \alpha'^{\frac{7}{2}} \Gamma\left(\frac{7}{2}\right).$$
(3.2)

⁶Actually, there is one further constant correction to the prepotential, proportional to the Euler characteristic $\chi(\mathbf{X}_6)$, arising from integrating R^3 over the space \mathbf{X}_6 , and reabsorbing the resulting correction to the 4d Einstein term via a change of frame [83]. This will not be relevant for our discussion.

We will be more interested in the Einstein frame action

$$S_{\text{IIA}} = \frac{1}{2} \int d^{10}x \sqrt{-g} \left(R - \frac{1}{2} (\partial \phi)^2 - \frac{1}{2} e^{\frac{3}{2}\phi} |F_2|^2 \right) , \qquad (3.3)$$

where, from now on, we use 10d Planck units. The D0-brane solution reads

$$ds^{2} = f(r)^{-7/8} (-dt^{2}) + f(r)^{1/8} (dx_{\perp}^{2}),$$

$$e^{(\phi - \phi_{\infty})} = f(r)^{3/4},$$

$$f(r) = 1 + \frac{\rho^{7} g_{s}^{\infty} N}{r^{7}}, \quad \text{with } \rho^{7} = (4\pi)^{\frac{5}{2}} \alpha'^{\frac{7}{2}} \Gamma\left(\frac{7}{2}\right).$$
(3.4)

Although not usually described from this viewpoint, this solution provides an interesting analogy with small black holes in lower dimensional theories. It has an S^8 horizon of zero size at its core, where a scalar (the dilaton) goes off to infinite distance in moduli space, resulting in a singularity. The latter is not worrisome, because it simply reflects the presence of a source, whose microscopic description is beyond the two-derivative supergravity approximation.

On the other hand, swampland arguments would imply that, as we are driven to the core of the solution, namely near infinite distance limit in the dilaton moduli space, the effective field theory should include additional degrees of freedom, corresponding to an SDC tower of states. Inclusion of these in the effective field theory leads to a lowering of the cutoff scale and hence a minimal size for the system, which would naturally be associated to a species scale horizon. In the following sections we provide several arguments that such a horizon exists and is given by the species scale.

3.1 Microscopic derivation of the entropy and of the species scale

The most direct way to argue that the D0-brane system develops a finite size horizon is to show that it has a large number of microstates, to which one can associate an entropy. This provides a lower bound on the effective size of the system, by holographic bounds.⁷ In this section we perform the microscopic computation of the entropy of a system of D0-branes with total charge N (which is familiar from similar combinatorics problems in string theory), and also explain the microscopic derivation of the species scale (which is a novel result).

Recall that consistency of the description of 10d type IIA as M-theory on \mathbf{S}^1 requires that, for each value of k, there is exactly one threshold bound state in the k D0-brane system (corresponding to the 11d graviton with k units of KK momentum) [84]. Hence, for a total D0-brane charge N, the number of microstates is given by the number of partitions p(N). The asymptotic behaviour for large N is given by the celebrated Hardy-Ramanujan formula

$$p(N) \sim \frac{1}{4\sqrt{3}N} \exp\left(\pi\sqrt{\frac{2}{3}}\sqrt{N}\right)$$
 (3.5)

The entropy of this thermodynamic ensemble for large charge is thus

$$S \sim \sqrt{N}$$
, (3.6)

⁷Note that in general the Bekenstein-Hawking area law gets corrected when the action includes higher curvature terms; hence, this relation will be made more precise in section 4.1.

up to $\log N$ corrections. It is tantalizing to speculate the latter may be related to similar log corrections to the species scale discussed in [85].

The above expression implies that the corresponding horizon size in a gravitational description of the system is parametrically larger than the Planck scale, as we will explain in later sections. In fact, the expectation that it corresponds to the species scale can already be supported from the viewpoint of microscopic combinatorics, as we show next.

One may have expected that, in order for a charge N black hole to be describable in an effective field theory, the latter should include at least N species (i.e. different D0-brane bound states). However, this is not correct and highly overestimates the number of species. In fact, rephrasing the definition in [48], the number of species N_s should be the minimum necessary to explain the entropy of black hole entropy. In other words,⁸ if we consider a theory with the number of species given by k and want to build a charge N black hole, the number of ways to do so is given by $p_k(N)$, the number of partitions of N with each part not larger than k. This is equivalent to the number of partitions of N into at most k parts.⁹ A classic result in [88] gives the asymptotic behaviour

$$\lim_{N \to \infty} \frac{p_k(N)}{p(N)} = \exp\left(-\frac{2}{C}e^{-\frac{1}{2}Cx}\right)$$
(3.7)

for

$$k = C^{-1} N^{\frac{1}{2}} \log N + x N^{\frac{1}{2}}, \text{ with } C = \pi \sqrt{\frac{2}{3}}.$$
 (3.8)

Summing up, on the one hand the entropy of a black hole made out of N D0 branes in a theory with k species is given by $p_k(N)$ in (3.7). On the other, we know the entropy of such a black hole is given by $p(N) \sim \sqrt{N}$ by microstate counting. We now show that this scaling can be also recovered if and only if the number of species in the theory is at least of order \sqrt{N} . Indeed, equation (3.7) gives an order one number as $N \to \infty$ (so that both p(N) and $p_k(N)$ are of the same order) unless $x \to -\infty$. Imposing that x should be bounded from below, (3.8) automatically implies that $k \gtrsim \sqrt{N}$ asymptotically, where we are ignoring a multiplicative log N correction. Hence, the number of species $N_s = \sqrt{N}$ suffices to explain the black hole entropy scaling. Equivalently, most microscopic configurations can be built using blocks of bound states of at most $N_s = \sqrt{N}$ D0-branes.

In fact, we incidentally note that the distribution is dominated by states containing at least one bound state of *exactly* $N_s = \sqrt{N}$ D0-branes. Following [89], we introduce the number p(N, k) as the number of partitions of *exactly* k parts (equivalently, partitions with at least one part *exactly equal* to k). One can define a probability measure for the likelihood of a random microstate containing at least one bound state of k D0-branes:

$$f_{N,k} = \frac{p(N,k)}{p(N)}, \quad \sum_{k=1}^{N} f_{N,k} = 1.$$
 (3.9)

 $^{^{8}}$ Similar counting problems arise in many other setups, for instance characterizing 1/2 BPS states with potentially large R-charges in AdS/CFT, see e.g. [86, 87].

⁹The equivalence is clear by using the relation of partitions with Young diagrams. A partition of $N = l_1 + \ldots + l_k$ with $l_i \ge l_{i+1}$ with at most k parts can be depicted as a Young diagram with N boxes arranged in k rows of lengths l_i . By conjugating the diagram, i.e. exchanging the roles of rows and columns, it corresponds to a partition of N with arbitrary number of parts, but each part being at most k.

Defining the quantity

$$X(k) = \frac{k}{\sqrt{N}} - \frac{1}{C}\log n, \qquad (3.10)$$

the probability distribution takes the asymptotic form

$$f_{N,k} \sim e^{-\frac{C}{2}e^{-\frac{2}{C}X(k)}} - e^{-\frac{C}{2}e^{-\frac{2}{C}X(k-1)}}.$$
(3.11)

The probability was shown in [90] to be maximized at a value

$$k_0 \sim \frac{\sqrt{6}}{\pi} \sqrt{NL} + \frac{6}{\pi^2} (3(L+1)/2 - L^2/4) - \frac{1}{2}, \text{ with } L = \log(\sqrt{6N}/\pi).$$
 (3.12)

This shows that the species scale is exactly given by $N_s = \sqrt{N}$ (and for instance, not smaller), as we had anticipated.

In the previous description, we simply used the microscopic D0-brane combinatorics, and the abstract definition of species scale. In an actual spacetime description of the system, it will be manifest that the D0-brane stack is a small black hole exploring the infinite distance limit of strong coupling, i.e. decompactification to 11d Mtheory. This will allow for the interpretation of the species scales as the 11d Planck scale. We thus turn to the description of the stretched horizon and the species scale from the spacetime perspective.

3.2 Hot D0-branes

In the previous section we have argued that the D0-branes solution develops an species scalesized stretched horizon. Naively, a direct approach to confirm the existence of a stretched horizon would be to solve the equations of motion including all the higher derivative corrections. However, given the lack of knowledge about the precise form of all such corrections, this is actually not feasible. In this section, we propose an alternative way to use spacetime equations of motion to explore the finite entropy of the system, in a universal way insensitive to the details of such corrections. We can explore the resolution of the singularity by studying the system at a finite temperature, and determine its properties as a function of N in the limit of vanishing temperature. We will find that the entropy goes as $N^{\frac{1}{2}}$, in perfect agreement with the microstate counting of the previous section. This provides further evidence that the would-be stretched horizon of a stack of D0 branes should have an area that scales as $N^{\frac{1}{2}}$.

We take the near-extremal (finite temperature) solution for D0-branes (see e.g. [52]), with a horizon r_0 . This comes down to introducing factors of

$$f_0 = \left(1 - \frac{r_0^7}{r^7}\right) \tag{3.13}$$

in the metric (3.4), where r_0 is tied to the energy density ϵ of the branes above extremality:

$$r_0^7 \sim \alpha'^4 g_s^2 \ \epsilon \,. \tag{3.14}$$

In the limit where $\epsilon \to 0$, we recover the extremal (zero temperature) stack of D0 branes.

In the spirit of the gauge/gravity correspondence for general D-branes in [52], we will carry out the computation of the entropy in the near-horizon limit

$$r \to 0$$
, $\alpha' \to 0$, $U = r/\alpha' = \text{fixed}$, $g_{QM}^2 = (2\pi)^{-2} g_s {\alpha'}^{-3/2} = \text{fixed}$. (3.15)

Here g_{QM} is the coupling of the D0-brane worldvolume quantum mechanics. It is useful to define the dimensionless coupling at some energy scale U

$$g_{\rm eff}^2 \sim g_{QM}^2 N U^{-3} \sim \lambda U^{-3} ,$$
 (3.16)

where $\lambda = g_{QM}^2 N$ is the (dimensionful) 't Hooft coupling.

The solution in the limit becomes

$$\frac{ds^2}{\alpha'} = -\frac{U^{7/2}}{\sqrt{d_0\lambda}} f_0 dt^2 + \frac{\sqrt{d_0\lambda}}{U^{7/2}} \left(\frac{dU^2}{f_0} + U^2 d\Omega_8^2 \right) ,$$

$$e^{\phi} = \frac{(2\pi)^2}{d_0} \frac{1}{N} \left(\frac{\lambda d_0}{U^3} \right)^{7/4} \sim \frac{g_{\text{eff}}^{7/2}}{N} ,$$
(3.17)

where

$$d_0 = 240\pi^5, \qquad U_0^7 = a_0 g_{QM}^4 \epsilon, \qquad a_0 = \frac{4480\pi^7}{3}.$$
 (3.18)

The horizon U_0 can be expressed in terms of the temperature as:

$$T^{-1} = \frac{4}{7}\pi\sqrt{\lambda d_0} U_0^{-5/2} \,. \tag{3.19}$$

We now move on to the Einstein frame to compute the entropy. The Einstein frame spatial metric reads [91]:

$$ds_E^2 = CU^{-7/8} \left(\frac{dU^2}{f_0} + U^2 d\Omega_8^2 \right) \quad \text{with} \quad C = \frac{l_s^2 N^{1/2} d_0^{1/8}}{2\pi \lambda^{3/8}} \,. \tag{3.20}$$

It is convenient to introduce a new radial variable R, with dimension of length, defined as $U^{9/8} = (R^2/C)$. The metric becomes:

$$ds_E^2 = \frac{256}{81} \frac{dR^2}{1 - \left(\frac{R_H}{R}\right)^{112/9}} + R^2 d\Omega_8^2, \qquad (3.21)$$

where R_H has the parametric dependence

$$R_H \sim \left(\frac{T}{\lambda^{1/3}}\right)^{9/40} N^{1/4} l_s \,.$$
 (3.22)

So the entropy of the finite temperature black hole in terms of the dimensionless temperature $T/\lambda^{1/3}$ is given by

$$S \sim N^2 \times (T/\lambda^{1/3})^{9/5}$$
. (3.23)

Restoring the powers of N hidden in T and λ , we get the scaling behaviour

$$S \sim (U_0^{5/2} / g_{QM}^{5/3})^{9/5} N^{1/2} .$$
(3.24)

In the limit $U_0 \to 0$, one recovers the familiar zero entropy result. However, the point that the above computation shows is that entropy of the system, when cloaked with a fixed horizon size, scales as $N^{1/2}$, in agreement with our discussion in section 3.1.

4 Species scale horizon from emergent higher derivative terms

It is useful to have a more direct intuition, at the level of the geometry, about the appearance of the species scale horizon for D0-branes uncovered in the previous section. In analogy with the 4d small black holes in section 2.2, we expect this horizon to show up once higher derivative corrections are included in the effective theory, which become relevant precisely at the species scale. Moreover, such new terms are expected to arise from integrating out the towers of light particles associated to the infinite distance limit of the formerly small black hole, in the spirit of Emergence.

In this section we work out in detail this picture for the D0-brane system. We first show that generic 10d R^4 corrections lead to a finite size horizon. We then consider the specific R^4 couplings arising from integrating out a tower of D0-brane states (i.e. 11d KK gravitons), which constitute the SDC tower associated to the infinite distance limit probed by the small black hole solution (decompactification to 11d M-theory). Interestingly we find that these familiar R^4 terms do not suffice to generate the finite size horizon for the system (let alone a species scale one). Turning things around, this suggests that the appearance of the species scale horizon demanded by the swampland considerations requires a crucial role of new physics beyond the usually considered R^4 terms. In fact, we include a specific set of supersymmetric 8-derivative terms involving the RR field strength 2-form, tractable via a lift to 11d Mtheory, and show that they do produce the species scale horizon, thus confirming the swampland expectations.

4.1 Entropy function analysis of general R^4 terms

It is well established that the existence and properties of stretched horizons for 4d small black holes in theories with higher-curvature corrections can be studied using the entropy function formalism [71]. The idea is to use the ansatz for a putative near horizon AdS_2 solution to evaluate an entropy function, and to extremize it with respect to its parameters. In this section we apply a similar logic to the solution of D0-branes in 10d type IIA, and argue that the introduction of higher curvature corrections leads to a stretched horizon for the system.

Before entering the discussion, let us make a general comment. In this section we only include general terms involving only the curvature, but no other fields. This is for tractability, since curvature terms lead to closed expressions which can be analyzed in a model-independent way, and suffice to illustrate the fairly generic appearance of stretched horizons. The discussion of other corrections can be carried out for instance if a specific set of such correction is fixed, as we do in section 4.4 for a set of 8-derivative corrections including curvature and the RR 2-form field strength.

4.1.1 The D0-brane solution at two-derivative level

We start by recovering the key features of the two-derivative solution of the D0-brane system from the entropy function formalism [71], as warmup for coming sections. As explained above, we introduce an $AdS_2 \times S^8$ ansatz:

$$ds^{2} = v_{1} \left(-r^{2} dt^{2} + \frac{1}{r^{2}} dr^{2} \right) + v_{2} d\Omega_{8}^{2},$$

$$F_{rt} = e, \quad \tilde{F}_{\theta \phi_{1} \cdots \phi_{7}} = 0,$$

$$e^{\phi} = g_{s},$$
(4.1)

where v_1 and v_2 control the AdS and sphere length scales. For the 2-form field strength this introduces electric but not magnetic charge at the horizon. Finally, the last equation sets the value of the string coupling at the horizon.

The entropy function is given (modulo some irrelevant overall factor) by evaluating the action at the near horizon ansatz (4.1) and applying a Legendre transform with respect to e, namely

$$\mathcal{E}(N, g_s, v, \beta, e) = eN - \int_{S^8} d\Omega_8 \sqrt{-g} \mathcal{L}|_h .$$
(4.2)

To evaluate this quantity for the ansatz (4.1) we compute

$$\sqrt{-g} = v_1 v_2^4 \sqrt{g_{\Omega_8}} = \frac{v^5}{\beta} \sqrt{g_{\Omega_8}} \,, \tag{4.3}$$

$$R = -\frac{2}{v_1} + \frac{56}{v_2} = \frac{56 - 2\beta}{v}, \qquad (4.4)$$

$$|F_2|^2 = \frac{1}{2!} F_{\mu\nu} F^{\mu\nu} = -\frac{1}{v_1^2} e^2 = -\frac{\beta^2}{v^2} e^2.$$
(4.5)

With hindsight, we have introduced new variables $v = v_2$ and $\beta = v_2/v_1$, to describe the overall length scale of $AdS_2 \times S^8$ and the relative scale separation. The result is

$$\mathcal{E}(N, g_s, v, \beta, e) = eN - \frac{8\pi^4 v^3 \left(\beta^2 e^2 g_s^{3/2} - 4(\beta - 28)v\right)}{105\beta}.$$
(4.6)

The goal is to extremize this function with respect to g_s , v, β and e, expressing the result in terms of the D0-brane charge N. We start with the extremization with respect to e, which gives

$$\frac{\partial \mathcal{E}}{\partial e} = 0 \quad \to \quad e = \frac{105N}{16\pi^4 \beta v^3 q_s^{3/2}}. \tag{4.7}$$

Substituting this result, the entropy function reads

$$\mathcal{E}(N, g_s, v, \beta) = \frac{105N^2}{32\pi^4 \beta v^3 g_s^{3/2}} + \frac{32\pi^4 (\beta - 28)v^4}{105\beta}.$$
(4.8)

We now extremize with respect to v and β , and get

$$\frac{\partial \mathcal{E}}{\partial v} = \frac{\partial \mathcal{E}}{\partial \beta} = 0 \quad \to \quad v^7 = \frac{1575N^2}{4096\pi^8 g_s^{3/2}} \,, \ \beta = 49 \,. \tag{4.9}$$

Incidentally, the approximately order 1 value of β (taking volume factors into account) is natural from the absence of scale separation in AdS vacua [92]. The resulting entropy function now reads

$$\mathcal{E}(N, g_s) \sim \frac{N^{8/7}}{g_s^{6/7}}$$
 (4.10)

We see that there is no well-defined extremum for the parameter g_s , it lies at $g_s \to \infty$. This simply reflects the fact that the D0-brane solution drives the dilaton to infinity in moduli space at its core, as observed above from the 10d solution. Indeed, we can recover other features of the 10d solution from the entropy function, by studying the scaling of different quantities with g_s . For instance, from (4.9) and (4.4), we have that in the $g_s \to \infty$ limit

$$v_2 = v \sim \frac{1}{g_s^{3/14}} \to 0, \qquad R \sim g_s^{3/14} \to \infty.$$
 (4.11)

Namely, we recover that the horizon volume goes to zero, and the scalar curvature blows up with exactly the same scalings as would be obtained directly from the D0 solution at its core. Note that the g_s we consider in this section is to be compared to the dilaton near the core of the D0 stack in the solution (3.4). Indeed, g_s is the value of the string coupling at the horizon of the would-be black hole, whilst g_s^{∞} in (3.4) is the string coupling infinitely far away from the stack of branes.

We therefore recover that the process of extremization of the Wald entropy function of the near-horizon geometry of a black hole electrically charged under the RR 2-form in 10d describes the near-core behaviour of a stack of D0 branes. The fact that the Wald entropy is only extremized for a value of the string coupling at the core that is blowing up, with a horizon that tends to zero size is exactly the behaviour expected from small black holes, as in section 2. We conclude from this that the Wald entropy formalism constitutes a reliable description of D0-brane stacks as small black holes. In the next section, we will introduce higher derivative corrections to the 10d supergravity which will affect the Wald entropy in such a way that the same exact minimization procedure will lead to a finite sized black hole instead of a small one.

4.1.2 The D0-brane solution and higher derivative terms

In analogy with the 4d small black hole case, we may expect that the D0-brane solutions can develop a stretched horizon upon the inclusion of suitable higher derivative corrections. In 10d type IIA the first corrections allowed by supersymmetry arise at 8-derivative level. There is a vast literature devoted to the computation of these corrections at tree and one-loop level (see e.g. [76, 93–98]). However, for the present purposes it will be more efficient to assume the presence of general R^4 corrections (the example of supersymmetric R^4 corrections will be discussed explicitly in section 4.3). Also, as explained earlier, we focus on terms involving only curvatures, which allows for a tractable model-independent analysis. The inclusion of terms involving other fields is discussed in section 4.4 for a specific set of supersymmetric 8-derivative terms including the RR 2-form field strength.

We now describe the entropy function computation in the presence of general R^4 corrections. The corrections can be written as a linear combination of all possible contractions

between eight inverse metrics and four Riemann tensors with all the indices down. Inspecting the ansatz in (4.1), we see that each inverse metric gives either a $1/v_1$ or a $1/v_2$ contribution, while each Riemann gives either a v_1 or a v_2 . In terms of v and β , this means that any of these contributions will take the form $\beta^a v^{-4}$, with a ranging from -4 to 8. For instance, the a = -4one corresponds to the case in which all the Riemanns give v_1 and all the metrics give $1/v_2$, while the a = 8 one corresponds to all the Riemanns giving v_2 and all the metrics giving $1/v_1$.

Following this reasoning, we can write the R^4 term evaluated at the horizon as

$$\mathcal{L}_{R^4}|_h = -g_s^{1/2} \; \frac{p_{12}(\beta)}{\beta^4} \, \frac{1}{v^4} \,, \tag{4.12}$$

where p_{12} denotes a degree 12 polynomial, and the minus sign is there for future convenience. Note that we already make explicit the dependence on the string coupling, which is fixed to that of the R^4 terms arising at first loop in perturbative Type IIA string theory. Motivated by having $g_s \to \infty$ at the core of the D0 solution at the two-derivative level, we are neglecting the tree level piece. One could worry that even higher loop contributions would dominate as $g_s \to \infty$, but they actually vanish. This has been considered previously in the literature from the Type IIA perspective [76, 99, 100] and, as it will become clearer in section 4.4, it is required for a consistent uplift to M-theory in this limit.

Including the extra term in (4.12), the entropy function (4.2) reads

$$\mathcal{E}(N, g_s, v, \beta, e) = eN - \frac{8}{105}\pi^4 \beta e^2 g_s^{3/2} v^3 + \frac{16\pi^4 v \left(g_s^{1/2} p_{12}(\beta) + 2(\beta - 28)\beta^4 v^3\right)}{105\beta^5}.$$
 (4.13)

The extremization with respect to e is insensitive to the new R^4 term, so the result is again (4.7). Plugging this result, the entropy function reads

$$\mathcal{E}(N, g_s, v, \beta) = \frac{105N^2}{32\pi^4 \beta g_s^{3/2} v^3} + \frac{16\pi^4 \left(g_s^{1/2} p_{12}(\beta)v + 2(\beta - 28)\beta^4 v^4\right)}{105\beta^5}.$$
 (4.14)

In order to display the effect of the higher derivative term in the dilaton, let us now consider the extremization with respect to g_s . We obtain

$$\frac{\partial \mathcal{E}}{\partial g_s} = 0 \quad \to \quad g_s = \frac{105}{16\pi^4} \sqrt{\frac{3}{2}} \frac{\beta^2 N}{\sqrt{p_{12}(\beta)} v^2} \,. \tag{4.15}$$

We see that the new contributions generically stabilize the dilaton. We note that the solution only exist if $p_{12}(\beta_0) > 0$, where β_0 is the yet to be determined value for β extremizing the entropy function. The latter is determined by the extremization with respect to v, namely

$$\frac{\partial \mathcal{E}}{\partial v} = \frac{128\pi^4(\beta - 28)v^3}{105\beta} = 0 \quad \to \quad \beta = 28.$$
(4.16)

Note that we again obtain a value compatible with the absence of scale separation, suggesting that this property is fairly robust against inclusion of higher derivative corrections.

Finally, the extremization with respect to β gives

$$\frac{\partial \mathcal{E}}{\partial \beta}\Big|_{\beta=28} = 0 \quad \to \quad v^4 = \frac{4\,p_{12}(28) - 21\,p'_{12}(28)}{175616 \cdot 6^{1/4}\,\pi^2\,p_{12}(28)^{1/4}}\,\sqrt{\frac{5N}{7}}\,,\tag{4.17}$$

where $p'_{12}(\beta)$ is the derivative of the polynomial with respect to β . Hence, the existence of the solution requires

$$p_{12}(28) > 0, \quad 4 p_{12}(28) > 21 p'_{12}(28).$$
 (4.18)

We will see the explicit evaluation of such polynomials for a particular R^4 correction in section 4.3.

In conclusion, under fairly general circumstances, the presence of higher derivative corrections suggests the existence of a stretched horizon for 10d D0-brane solutions. From the above expressions, the scaling of its properties with N is

$$S \sim \sqrt{N}, \qquad v^4 \sim \sqrt{N}, \qquad \beta = 28, \qquad g_s \sim N^{3/4}, \qquad e \sim N^{-1/2}.$$
 (4.19)

The entropy of the system scales as \sqrt{N} , in agreement with the microstate counting in section 3.1. Notice that this, as well as all the scalings above, rely heavily in the string coupling dependence made explicit in (4.12) and corresponding to the one-loop contribution in perturbative Type IIA string theory. The string coupling at the horizon blows up as $N \to \infty$, which is consistent with neglecting the tree level contribution in this regime. In the next section we will shed some further light on these scalings, since they lead to the appearance of the species scale.

4.1.3 The species scale and cosmic censorship

The species scale. Here we shortly note that the above horizon scale indeed corresponds to the species scale of the SDC tower associated to the infinite distance limit probed by the solution. This infinite distance limit corresponds to a decompactification to 11d M-theory (consistently with the above mentioned fact that the horizon value of g_s gets strong in the large N regime). The extra dimension is reconstructed by the SDC tower of D0 branes, whose species scale corresponds to the 11d Planck scale. Restoring the 10d Planck scale, we have

$$\Lambda_s \sim M_{11} \sim g_s^{-1/12} M_{10} \sim N^{-1/16} M_{10} \,, \tag{4.20}$$

where we have used the scaling of g_s with N in equation (4.19). It is indeed easy to show that this coincides with the inverse radius of the stretched horizon. Using the scaling of v with N in (4.19), we have

$$r_h^{-1} \sim v^{-1/2} M_{10} \sim N^{-1/16} M_{10}$$
 (4.21)

Let us note that this agreement is easily understood by the fact that the stretched horizon arises from the competition between the classical and the R^4 terms in the effective action. The scale at which these two can compete indeed corresponds to one of the notions of the species scale [47]. The fact that it matches the species scale associated to the tower of D0s can be directly checked in the effective action and, as we explain in section 4.2, can be understood from the emergence of higher derivative terms coming from integrating out the tower of D0s. Here we see that these two notions of species scale also coincide with the one of the smallest BH describable within the EFT, in this case the stretched horizon of the D0 black hole. The D0-brane bound state size scale. We have emphasized the appearance of a finite size for the system from the spacetime viewpoint, via the appearance of a stretched horizon. However, as any quantum system, the D0-brane system has a characteristic size, in particular that associated to the size of the D0-brane bound states. This has been studied in particular in the context of M(atrix) theory. In this context, the outcome of [51, 101–103], is an estimate of the size of a bound state of N D0-branes scaling as

$$R \gtrsim N^{1/3} M_{11}^{-1} \,. \tag{4.22}$$

This grows with N in a way seemingly stronger¹⁰ than the stretched horizon size $r_h \sim M_{11}^{-1}$ discussed above. However, an important observation is that (4.22) is derived in the limit of $g_s \to 0$ with M_{11} fixed, with g_s the coupling in flat space. Hence, for a proper comparison, in the evaluation of the horizon size $r_h \sim N^{1/16} M_{10}^{-1} \sim g_s^{-1/12} N^{1/16} M_{11}^{-1}$, one should not use the horizon value $g_s \sim N^{3/4}$ in matching M_{10} , but rather consider g_s as a free asymptotic parameter. Thus, in the $g_s \to 0$ and M_{11} fixed limit, the horizon size is parametrically larger than the size of the bound state of gravitons, which are effectively cloaked behind the horizon.

A cosmic censorship interpretation. The above considerations motivate a cosmic censorship interpretation of our result, which provides a different angle to other uses of cosmic censorship in the analysis of swampland constraints (see e.g. [104, 105]). The classical 2-derivative solution displays a singularity. In the full theory, the pathological implications of such singularities must be avoided, either by some desingularization [106], addition of extra degrees of freedom as in orbifolds, or the appearance of a horizon cloaking it. We have shown that in our system the latter occurs via the use of higher derivative interactions, despite the fact that it does not happen in the low-energy two-derivative approximation. Interestingly, this happens in the realm of the effective field theory, on the verge of its validity. This is again tied up to the notion of the species scale as that at which the tower of higher derivative interactions becomes relevant.

4.2 The D0-brane SDC tower and higher derivative emergence

The infinite distance limit in moduli space explored by the core of the solution is the strong coupling limit of type IIA theory, which corresponds to the decompactification limit of 11d M-theory on \mathbf{S}^1 . The tower of states predicted by the swampland distance conjecture corresponds to D0-brane states,¹¹ namely the KK momentum states of 11d graviton multiplets. The computation is indeed simpler in this 11d picture, and is a higher-dimensional version of that encountered in section 2.2.3. In fact is a well-established story that this set of states can lead, at the 1-loop level, to the appearance of higher derivative corrections to the 10d effective action, which can be interpreted as tree- and one-loop level (in g_s) terms of type IIA theory [76], as we now discuss.

¹⁰Note that the stronger growth also holds even if we restrict to the dominant bound state of $k \sim N^{1/2}$ D0-branes, as in section 3.1, which would yield $R \sim N^{1/6}$.

¹¹We expect readers not to confuse the D0-brane states in the SDC tower with the D0-branes sourcing the solution in the previous sections.

The terms we are going to compute are the $t_8t_8R^4$ terms,¹² in principle of the kind included in the analysis in section 4.1.2 (see also section 4.3 for more details on their structure). They are protected by supersymmetry, so only the 1-loop diagram of BPS 11d graviton KK modes can contribute. These can be easily computed from a worldline formalism. Using the general expression (2.24), integrating over continuous momenta, and particularizing for d = 10, n = 4, we get

$$\mathcal{A}_{10,4} = \frac{\sqrt{\pi}}{2R} \tilde{K} \int_0^\infty \frac{d\tau}{\tau^2} \sum_k e^{-\pi\tau R^{-2}k^2} , \qquad (4.23)$$

where the prefactor \tilde{K} contains all the kinematics, and is the linearized approximation to the $t_8 t_8 R^4$ term in the effective action. As in the 4d case, the above expression diverges in two ways: because of the integral and because of the infinite sum. We can perform a Poisson resummation, so that both are nicely combined and we get

$$\frac{1}{\pi^{3/2}}\mathcal{A}_{10,4} = \tilde{K} \int_0^\infty d\hat{\tau} \, \hat{\tau}^{1/2} \sum_l e^{-\pi\hat{\tau}R^2l^2} = C\tilde{K} + \frac{\zeta(3)}{\pi R^3}\tilde{K} \,, \tag{4.24}$$

where $\hat{\tau} = \tau^{-1}$. This Poisson summation trades the KK momentum k for the winding number l of the worldline along the \mathbf{S}^1 . The only divergence is now in the l = 0 piece, which has been isolated as the first terms in the second equality; C is an unknown coefficient regularizing the divergence.¹³

As anticipated, the quantum corrections coming from the SDC tower lead to a higher derivative R^4 correction, in principle of the kind invoked to lead to a stretched horizon. There remains to perform a detailed evaluation of these terms and their effect in the entropy function argument, which we postpone to section 4.3.

An important remark is that the R^4 correction obtained in this way is ambiguous. The 4-graviton scattering is an on-shell quantity that is invariant under local field redefinitions $g_{ij} \rightarrow g_{ij} + \delta g_{ij}$, which is usually used to get rid of terms involving Ricci tensors (see e.g. [55]). However, since the AdS₂ × \mathbf{S}^8 ansatz has a non-vanishing Ricci tensor, doing this is not innocuous for our purposes. We will thus stick to the full $t_8 t_8 R^4$ term, including Ricci terms, as it is the natural kinematic function that appears in this computation à la Emergence. That this tensor structure gives the full off-shell R^4 correction to the effective action was also suggested in [95]. As evidence for this, it was found that the terms containing no more than one Ricci tensor, which were reliably computed from the vanishing of the worldsheet beta function, precisely match the ones appearing in the $t_8 t_8 R^4$ structure.

Higher derivative emergence. When translated to the Type IIA frame, the first and second terms in (4.24) precisely reproduce the $t_8t_8R^4$ tensor structure of the 1-loop and

¹²As we will explain in section 4.3, the actual 10d type IIA R^4 terms contain a further piece with an $\epsilon_{10}\epsilon_{10}$ structure. This however does not contribute to 4-graviton scattering amplitudes, or conversely is not captured by our 4-graviton 1-loop amplitude (it would require a 5-point computation). This is analogous to the similar statements for the odd-odd structure in the R^4 computation from scattering amplitudes in string theory, see e.g. [107].

¹³In [76], it was fixed by T-duality upon further S^1 compactification. In our present context, there is no way of fixing it without this kind of extra UV information. It would be interesting if performing the sum over KK modes up to the species scale one could reproduce its precise value, along the lines of [39–43].

tree-level string perturbation theory contributions, respectively. Therefore, this establishes that these can be interpreted as coming from integrating out the full tower of D0-particles, as well as the fast-movers in the massless 10d sector. Let us remark that this result is very much in the spirit of the Emergence Proposal, albeit for higher derivative terms rather than just the kinetic terms.

Note also that the above computation integrates out the full tower of states, i.e. not just up to the species scale as in [39–43]. In fact, interpreting the latter recipe as a regularization procedure to be applied when integrating out infinite number of states, one could hope to recover the right value for C; we leave this as an interesting open question. On the other hand, notice that integrating the full tower is in the spirit of the recent proposals in [42, 43]. It would be interesting to explore these connections further.

4.3 Supersymmetric 10d R^4 terms are not enough

In this section, we particularize the analysis in section 4.1.2 to the specific R^4 terms appearing in the Type IIA effective action (for a concrete choice fixing all the already mentioned ambiguities concerning these terms). By plugging the $AdS_2 \times S^8$ ansatz, we will compute the polynomial appearing in equation (4.12) and check the conditions in equation (4.18).

We will show that, perhaps surprisingly, these supersymmetric R^4 terms do not satisfy the constraints in section 4.1.2 required to generate a finite size horizon. Note that this does not invalidate the analysis of the entropy function, but rather shows that the appearance of the horizon from pure higher curvature terms demands the use of other possibly non-supersymmetric terms. Alternatively, one can maintain the problem in the realm of supersymmetric corrections, and include higher derivative couplings involving the RR 2-form field strength; these go beyond the analysis in section 4.1.2, and will be discussed in section 4.4.

Hence, we focus on the supersymmetric R^4 corrections in 10d type IIA theory. Following the notation in [98], at the eight derivatives level, the type IIA action gets supplemented by additional terms quartic in the Riemann tensor at both tree level and first loop in the string coupling g_s :

$$S_{\rm IIA} = S_{\rm IIA}^{class} + \frac{1}{3 \cdot 2^{11}} (S^{\rm tree} + S^{loop}), \qquad (4.25)$$

where S^{tree} and S^{loop} both arise at level α'^3 . In fact, it has been conjectured that these are the only two contributions to this type of term in string perturbation theory [76, 99, 100]. They can be written as follows in the Einstein frame and in 10d Planck units:

$$S^{\text{tree}} = \frac{\zeta(3)}{2} \int d^{10}x \sqrt{-g} \, e^{-\frac{3}{2}\phi} \left(t_8 t_8 R^4 + \frac{1}{8} \epsilon_{10} \epsilon_{10} R^4 \right) \,, \tag{4.26}$$

$$S^{loop} = \frac{\pi^2}{6} \int d^{10}x \sqrt{-g} \, e^{\frac{1}{2}\phi} \left(t_8 t_8 R^4 - \frac{1}{8} \epsilon_{10} \epsilon_{10} R^4 \right) \,, \tag{4.27}$$

where the form of the $t_8 t_8 R^4$ and $\epsilon_{10} \epsilon_{10} R^4$ structures can be found in appendix A. Let us only consider the 1-loop term, which is the leading one as $g_s \to \infty$.

Evaluating $t_8 t_8 R^4$ and $\frac{1}{8} \epsilon_{10} \epsilon_{10} R^4$ on the $AdS_2 \times S^8$ near-horizon geometry (4.1), one obtains:

$$t_8 t_8 R^4 = \frac{192 \left(3\beta^4 + 56\beta^2 + 2520\right)}{v^4} \,, \tag{4.28}$$

$$\frac{1}{8}\epsilon_{10}\epsilon_{10}R^4 = \frac{161280(4\beta - 1)}{v^4}.$$
(4.29)

Therefore, the 1-loop term can be put in the form (4.12) if

$$p_{12}(\beta) \sim -\beta^4 \left(3\beta^4 + 56\beta^2 - 3360\beta + 3360 \right) ,$$
 (4.30)

where we are ignoring an irrelevant positive-defined prefactor. Now we can check the conditions in (4.18) for the existence of an extremum of the entropy function. The second one is satisfied, while the first one is not. This implies that the familiar R^4 terms used so far do not suffice to generate the stretched horizon for the D0 solution.

This does not invalidate the main claim that the system should develop a species scale horizon. As explained earlier, we expect other higher derivative couplings to contribute non-trivially, and the horizon may very well develop when they are ultimately included. Indeed, we will show in the next section that a simple modification of the above computation promoted to M-theory allows to include couplings involving the RR 2-form F_2 , and lead to a non-trivial species scale stretched horizon for the D0-brane system.

4.4 Including F_2 : species scale D0-brane horizon from M-theory couplings

In the previous section we showed that 10d supersymmetric R^4 terms do not generate a finite size horizon. Hence, the requirement that there should arise species scale horizon, as demanded by swampland considerations, implies that one should look further. One possibility is to consider other possibly non-supersymmetric pure higher curvature terms. A more attractive avenue is to stay in the realm of supersymmetric terms but include higher derivative couplings involving the RR 2-form field strength F_2 . These go beyond the analysis in section 4.1.2, but, once a specific set of couplings is fixed, are amenable to a direct study again via the entropy function.

In this section we include such terms, completing the analysis at the supersymmetric eight derivatives level of the effective action. The most efficient way of doing this is by encoding these terms in the 11d M-theory effective action, since both the 10d metric and RR 1-form become part of the 11d metric. This is, by encoding the extremal near-horizon ansatz in the 11d metric and plugging it into the M-theory effective action, supplemented by R^4 terms, we can effectively recover the 10d Lagrangian evaluated at the horizon including both R^4 and terms involving F_2 . Let us remark that we will keep the 10d point of view at all times to focus in the presence of a stretched horizon for the D0 solution. This is, here we think about the 11d effective action as just an efficient way of encoding the eight derivatives terms involving F_2 and evaluating them at the extremal near-horizon ansatz.

Our analysis is close to that in [55], with the difference that we emphasize the 10d perspective in the computation. Even though we find analogous results, an important difference in the analysis is that we keep the full $t_8 t_8 R^4$ structure, while only purely Riemannian terms were included in [55]. As already discussed in section 4.2, the Wald entropy analysis

is sensitive to the terms including the Ricci tensor which, via Emergence, are naturally encoded in the $t_8 t_8 R^4$ structure. This is consistent with the results obtained in [95], where the terms in $t_8 t_8 R^4$ including one Ricci tensor were unambiguously shown to appear in the Type IIA effective action.

Following again the notation in [98], and setting the three-form to zero for our purposes, the M-theory effective action can be written at the eight derivatives level as

$$S_M = \frac{1}{2\kappa_{11}^2} \int d^{11}x \sqrt{-\hat{g}} \left(\hat{R} + \frac{\left(4\pi\kappa_{11}^2\right)^{2/3}}{(2\pi)^4 \cdot 3^3 \cdot 2^{13}} \left(\hat{t}_8 \hat{t}_8 \hat{R}^4 - \frac{1}{24} \epsilon_{11} \epsilon_{11} \hat{R}^4 \right) \right).$$
(4.31)

To simplify the computations, we will take $\kappa_{11} = 1.^{14}$ The \hat{t}_8 and ϵ_{11} tensors are defined in complete analogy with the Type IIA case. In fact, after writing these terms as contractions of four Riemann tensors, the 11d R^4 term look exactly the same as the 10d one. More details about these structures can be found in appendix A.

As it is well-known, the circle compactification ansatz from 11 to 10 dimensions is:

$$d\tilde{s}^{2} = e^{-\frac{1}{6}\phi} ds^{2} + e^{\frac{4}{3}\phi} \left(dz - C_{\mu} dx^{\mu} \right)^{2} .$$
(4.32)

Plugging this into the effective action recovers the 10d Einstein frame Type IIA effective action up to the eight-derivative level. Instead of doing this and then taking the extremal $AdS_2 \times S^8$ near-horizon ansatz, we perform the computation at the level of M-theory. That is, we first encode the $AdS_2 \times S^8$ ansatz into the 11d metric, and then we plug it into the 11d effective action. This allows us to read off the 10d Lagrangian evaluated at the horizon. For this last step, we need to take into account the relation between the 11d and 10d determinant of the metric appearing in the action. This gives the following relation:

$$\mathcal{L}_{IIA}|_h = g_s^{-1/6} \mathcal{L}_M|_{11d\,ansatz} \,. \tag{4.33}$$

To encode the extremal $AdS_2 \times S^8$ near-horizon geometry in the 11d metric ansatz, we just plug equation (4.1) into (4.32) to get

$$d\tilde{s}^{2} = -g_{s}^{-1/6}v_{1}\left(r^{2}dt^{2} + \frac{1}{r^{2}}dr^{2}\right) + g_{s}^{-1/6}v_{2}\,d\Omega_{8}^{2} + g_{s}^{4/3}\left(dz - e\,r\,dt\right)^{2}\,.$$
 (4.34)

Evaluated on this ansatz, the 11d Ricci scalar reads

$$\hat{R} = g_s^{1/6} \frac{\beta^2 e^2 g_s^{3/2} - 4v(\beta - 28)}{2v^2}, \qquad (4.35)$$

where we introduced new variables $v = v_2$ and $\beta = v_2/v_1$ as we did in section 4.1.1. When plugged into (4.33), this recovers the two-derivative piece of the Type IIA Lagrangian evaluated at the horizon (cf. equation (4.6)).

¹⁴Even though this might seem to imply that the results will be in 11d Planck units, when mapping to the type IIA Einstein frame, the 11d Planck scale gets traded for the 10d one. Thus, the final results are actually in 10d Planck units.

Similarly, we evaluate the M-theory R^4 terms in this ansatz, getting

$$\left(\hat{t}_{8} \hat{t}_{8} - \frac{1}{24} \epsilon_{11} \epsilon_{11} \right) \hat{R}^{4} = \frac{3g_{s}^{2/3}}{4v^{8}} \left(287\beta^{8} e^{8}g_{s}^{6} - 1392\beta^{7} e^{6}g_{s}^{9/2} v + 32\left(83\beta^{2} + 308 \right) \beta^{4} e^{4}g_{s}^{3}v^{2} - 768\left(3\beta^{3} + 28\beta - 280 \right) \beta^{2} e^{2}g_{s}^{3/2}v^{3} + 256\left(3\beta^{4} + 56\beta^{2} - 3360\beta + 3360 \right) v^{4} \right).$$

$$(4.36)$$

As a check, we can recover the Type IIA R^4 result from the previous section by setting e = 0 and plugging (4.36) into (4.33) (cf. equation (4.28) with the $g_s^{1/2}$ factor appearing in (4.12)).

Putting these two results together in the 11d Lagrangian, using equation (4.33) and plugging the result into the definition of the entropy function in (4.2), we finally obtain:

$$\mathcal{E}(N,g_s,v,\beta,e) = eN - \frac{8\pi^4 v^3 \left(\beta^2 e^2 g_s^{3/2} - 4(\beta - 28)v\right)}{105\beta} \\ - \left(\frac{\pi}{2}\right)^{2/3} \frac{g_s^{1/2}}{2580480 v^3\beta} \left(287\beta^8 e^8 g_s^6 - 1392\beta^7 e^6 g_s^{9/2}v + 32\left(83\beta^2 + 308\right)\beta^4 e^4 g_s^3 v^2 \\ - 768\left(3\beta^3 + 28\beta - 280\right)\beta^2 e^2 g_s^{3/2} v^3 + 256\left(3\beta^4 + 56\beta^2 - 3360\beta + 3360\right)v^4\right).$$

$$(4.37)$$

Here we have already performed the integral in equation (4.2), taking into account the determinant of the metric evaluated in the ansatz as given in (4.3).

The entropy function turns out to be rather involved, and we cannot extremize it analytically. Doing so numerically would require fixing N to a set of different values, extremizing the entropy function for each of these and then fitting the dependence on N of the various quantities $(g_s, v, ...)$. Instead of doing so, let us perform a change of variables inspired by the scalings found in section 4.1.2. We define

$$g_s = \tilde{g}_s N^{3/4}, \qquad v = \tilde{v} N^{1/16}, \qquad \beta = \tilde{\beta}, \qquad e = \tilde{e} N^{-1/2}.$$
 (4.38)

Introducing this into the entropy function, we get

$$\begin{aligned} \mathcal{E}(N,\tilde{g}_{s},\tilde{v},\tilde{\beta},\tilde{e}) &= \sqrt{N} \Bigg[\tilde{e} - \frac{8\pi^{4}\tilde{v}^{3} \left(\tilde{\beta}^{2}\tilde{e}^{2}\tilde{g}_{s}^{3/2} - 4(\tilde{\beta} - 28)\tilde{v} \right)}{105\tilde{\beta}} \\ &- \left(\frac{\pi}{2} \right)^{2/3} \frac{\tilde{g}_{s}^{1/2}}{2580480 \tilde{v}^{3}\beta} \left(287\tilde{\beta}^{8}\tilde{e}^{8}\tilde{g}_{s}^{6} - 1392\tilde{\beta}^{7}\tilde{e}^{6}\tilde{g}_{s}^{9/2}\tilde{v} + 32 \left(83\tilde{\beta}^{2} + 308 \right) \tilde{\beta}^{4}\tilde{e}^{4}\tilde{g}_{s}^{3}\tilde{v}^{2} \\ &- 768 \left(3\tilde{\beta}^{3} + 28\tilde{\beta} - 280 \right) \tilde{\beta}^{2}\tilde{e}^{2}\tilde{g}_{s}^{3/2}\tilde{v}^{3} + 256 \left(3\tilde{\beta}^{4} + 56\tilde{\beta}^{2} - 3360\tilde{\beta} + 3360 \right) \tilde{v}^{4} \right) \Bigg]. \end{aligned}$$

$$(4.39)$$

This change of variables does the important job of factoring out all the N-dependence in the entropy function, which is very non-trivial since all the terms had to conspire to give the

same overall \sqrt{N} factor. Thanks to this, we can now focus on the terms in parenthesis and look for its N-independent extrema numerically. If an extremum exist, then we see that the scalings we found in section 4.1.2 are automatically guaranteed. Furthermore, given that the overall factor in front of the entropy function is \sqrt{N} , the scaling of the entropy from the microscopic counting in section 3.1 is also automatically guaranteed.

Finally, we look for extrema of the terms in parenthesis in the previous equation or, equivalently, of $\mathcal{E}(1, \tilde{g}_s, \tilde{v}, \tilde{\beta}, \tilde{e})$. This cannot be done analytically, so we perform a numerical search. The result is the following extremum:

$$\tilde{g}_s \approx 2.90365$$
, $\tilde{v} \approx 0.159653$, $\beta \approx 14.0556$, $\tilde{e} \approx 0.0479134$. (4.40)

For these values, one can check that the derivatives of the entropy function are 14 orders of magnitude smaller than the value of function itself. This indicates that, to a very good approximation, this is an extremum of the entropy function. To make this extremum more apparent, we can fix one of the variables to its value at the extremum and display the three-dimensional gradient flow. This is shown in figure 1, where we can see the presence of a non-trivial extremum. As it can be checked also by direct computation, we also see that the extremum is in fact a saddle point. This fact is not relevant for the Wald entropy formalism, that only requires the presence of an extremum. It would be interesting to understand the thermodynamical stability of this stretched horizon, somewhat along the lines of [55]. However, this analysis presumably requires a systematic inclusion of even higher derivative terms to achieve control of order one factors.

In conclusion, we found that the Type IIA action including *all* terms up to the eight derivative level captures the presence of the species scale stretched horizon for the D0 solution. In addition, the entropy associated to the horizon nicely fits the microscopic counting and the near-horizon $AdS_2 \times S^8$ is not scale separated. This gives support to the idea that, even though a proper geometric treatment of a stretched horizon requires the inclusion of all the infinite amount of higher derivative operators, the classical level and the (complete) first non-trivial higher derivative level seem to suffice for capturing its presence and its scaling properties with the species scale.

As in the case of 10d \mathbb{R}^4 , the new set of higher derivative terms can be understood as arising from integrating out the SDC tower of D0-branes, in the spirit of Emergence. Indeed, since all these couplings are related by 11d Poincaré invariance, the computation of the 4-point amplitude is essentially unchanged. From the 10d perspective, the difference is that instead of computing 4-graviton scattering one should consider processes involving the RR 1-form field. Their worldline vertex operators are directly related [108], hence the structure of the 1-loop integral is essentially unchanged, and the changes only involve the kinematic tensor structure. This computation is directly not available in the literature, but it can be regarded as a 10d decomposition of the computation of the 11d \mathbb{R}^4 correction from the 4-point 1-loop superparticle amplitude in [109]. This suffices to show the emergence from the SDC tower of the required higher derivative terms beyond the purely gravitational sector.

5 Conclusions

In this work we have clarified the links among the different avatars of the species scale in the swampland program: (i) as cutoff of the effective theory specially in the presence of the



Figure 1. Gradient flows of the entropy function with one of the variables fixed. Lighter/yellow colors denotes bigger values for the gradient, while darker/blue colors denote smaller ones.

infinite SDC tower of light particles near infinite distance points in moduli space, (ii) the scale at which terms in the infinite series of higher derivative corrections in the gravitational sector start competing with the two-derivative level, (iii) as the size of the smallest black hole which admits a description within the EFT.

Our analysis uses small black holes, whose cores probe infinite distance limits in field space, at which we can characterize the SDC tower of states leading to the species scale in avatar (i). Upon integrating out the tower of states, the theory generates a series of higher-dimensional operators, naturally leading to avatar (ii) of the species scale. Finally, including these corrections in the black hole solution triggers the appearance of a stretched horizon of a size given by the species scale in avatar (iii).¹⁵

¹⁵It is worth noticing that we consider BPS small black holes, and that the quantum gravity effects we are taking into account come in the form of supersymmetry protected higher derivative corrections. It would be interesting to see how our results might extend to cases with less supersymmetry.

We have provided quantitative examples of this phenomenon, both in a class of 4d small black holes, and in the fascinating case of 10d type IIA D0-branes. The latter case turned out to be highly non-trivial, as the appearance of the species scale horizon required understanding of 10d 8-derivative terms involving not only the curvature but also the RR 2-form field strength. The nice packaging of those terms into an 11d R^4 correction is a tantalizing hint that the species scale can serve as a powerful probe of the UV physics of M-theory.

The 10d D0-brane example displays an interesting feature: the objects making up the small black hole are precisely of the same kind as those in the SDC tower for the corresponding infinite distance limit. This feature arises in the microscopic explanation of the entropy for other black holes in string theory; but the remarkable fact about the D0-brane case is that the higher derivative terms coming from integrating out the tower of D0-branes are crucial to generate a geometric horizon for the D0-brane black hole itself!

This motivates us to entertain the following picture. According to the most radical interpretation of the Emergence Proposal [35–38], not only these higher derivative terms but the very dynamics of gravity would emerge from integrating out the towers of states. In this spirit, the emergence of species scale stretched horizons studied in this work would potentially apply to any black hole horizon. This suggests a possible general picture for the emergence of horizons in quantum gravity: in the UV, the natural quantum system to consider would be that formed by the states in the tower, with no dynamical gravity. When going down to the IR, the species in the tower are integrated out. It is reasonable to expect that this effective description cannot describe the previous physical system in full detail, but it should be able to codify its coarse-grained properties, i.e., its macrostate. How is this made possible? In the IR, gravity emerges and the system would be replaced by a black hole that precisely reproduces this macrostate. In this sense, the emergence of a horizon would be somehow required for the effective description to be able to keep the macroscopic information about the formerly considered physical system.

It is also tantalizing to speculate that the entropy of the black hole is nothing but a reflection of the entanglement entropy of the species after their degrees of freedom are traced out. This goes in the spirit of ER=EPR [110], the N-portrait picture [111], or of the literature on thermal M(atrix) theory black holes, see e.g. [112–118] (also [119, 120] for more recent developments). On a related note, it would be interesting to understand if and how this system made out of species could radiate and reproduce the evaporation of the black hole, perhaps along the lines of [31].

Back to more earthly matters, our work suggest several interesting questions for further research, for instance:

- We have focused on the interpretations of the species scale in asymptotic regions of moduli space, which are those naturally explored by small black holes. It would be interesting to extend our concept of unification of the three notions of species scale and gain a similar understanding in the interior of moduli space, connecting with the proposal in [30, 47, 48, 121], perhaps using large black hole probes as in [28].
- Even in asymptotic regimes, there is a strong ongoing activity in understanding the precise treatment of the species scale in the process of integrating out the SDC tower of states, with different proposals [39–43]. We hope the appearance of such computations in our setup can serve to clarify the proper physical procedures.

- The non-trivial constraints on higher derivative corrections for the appearance of stretched horizons are reminiscent of similar conditions derived from positivity constraints in the swampland program or in the S-matrix bootstrap, see e.g. [62, 122–124]. It would be interesting to explore this possible connections further.
- The $AdS_2 \times S^8$ ansatz in the entropy function analysis suggests, via holography, the possible appearance of IR superconformal behaviour in the worldvolume theory on the D0-brane, once the equivalent to the higher derivative corrections are taken into account. It would be interesting to find further quantitative evidence for this proposal from the field theory side.
- It would be interesting to see whether some of our results extend to lower codimension cases. Higher-dimensional objects that probe infinite distance in field space and are singular in the EFT (such as the End-of-the-World branes of [20, 24, 26, 125–128]¹⁶) could, in the same way as the small black holes, be shown to develop a species scale sized horizon using a modified version of the entropy function formalism.
- In some instances, higher derivative corrections have been shown to create new singularities on the horizon of large black holes [139]. This is to be put in contrast with our work, where curvature corrections instead smooth out singularities. It would be interesting to better understand the mechanisms that make it such that curvature corrections can seemingly both create and smooth out singularities in the EFT.
- There are interesting recent development regarding the possibility of log corrections to the species scale [41, 48, 85]. In this respect, it is tantalizing that our microscopic considerations in section 3.1 lead to such corrections. It would be interesting to exploit our techniques to shed some light on this question.

We hope to come back to these questions in the near future.

Acknowledgments

We thank Roberta Angius, Alberto Castellano, Lucia Gómez Cordova, Jesús Huertas, Luis Ibáñez, Shota Komatsu, Wolfgang Lerche, Luca Melotti, Fernando Marchesano, Jake McNamara, Miguel Montero, Tomás Ortín and Irene Valenzuela for useful conversations. The work of M.D. and A.U. is supported through the grants CEX2020-001007-S and PID2021-123017NB-I00, funded by MCIN/AEI/10.13039/501100011033 and by ERDF A way of making Europe.

A Tensor structure of R^4 corrections

In this appendix we recall the form of the $t_8 t_8 R^4$ and $\epsilon_{10} \epsilon_{10} R^4$ terms arising in 10d type IIA theory (and closely related to those in 11d M-theory). We follow closely the notation in [36].

Let us start with the $t_8 t_8 R^4$ structure, which denotes

$$t_8 t_8 R^4 = t_8^{A_1 \cdots A_8} t_8^{B_1 \cdots B_8} R_{A_1 A_2 B_1 B_2} \cdots R_{A_7 A_8 B_7 B_8}, \qquad (A.1)$$

¹⁶For the related topic of solutions in theories with dynamical tadpoles, see [129–132] for early work and [133–138] for related recent developments.

where the t_8 tensor can be written in terms of the metric as¹⁷

$$t_8^{A_1\cdots A_8} = \frac{1}{5} \Big[-2 \left(g^{A_1A_3} g^{A_2A_4} g^{A_5A_7} g^{A_6A_8} + g^{A_1A_5} g^{A_2A_6} g^{A_3A_7} g^{A_4A_8} + g^{A_1A_7} g^{A_2A_8} g^{A_3A_5} g^{A_4A_6} \right) \\ + 8 \left(g^{A_2A_3} g^{A_4A_5} g^{A_6A_7} g^{A_8A_1} + g^{A_2A_5} g^{A_6A_3} g^{A_4A_7} g^{A_8A_1} + g^{A_2A_5} g^{A_6A_7} g^{A_8A_3} g^{A_4A_6} \right) \\ - (A_1 \leftrightarrow A_2) - (A_3 \leftrightarrow A_4) - (A_5 \leftrightarrow A_6) - (A_7 \leftrightarrow A_8) \Big].$$
(A.2)

After contracting all the metrics in t_8 with the Riemman tensors in (A.1), the $t_8 t_8 R^4$ structure can be written as:

$$t_{8}t_{8}R^{4} = 12 \left(R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{1}A_{2}A_{3}A_{4}} \right)^{2} + 192 R_{A_{1}}^{A_{5}} R^{A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{2}}^{A_{7}} R^{A_{8}}R_{A_{5}A_{7}A_{6}A_{8}} - 192 R_{A_{1}A_{2}A_{3}}^{A_{5}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{4}}^{A_{6}A_{7}A_{8}}R_{A_{5}A_{6}A_{7}A_{8}} + 24 R_{A_{1}A_{2}}^{A_{5}A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{3}A_{4}}^{A_{7}A_{8}}R_{A_{5}A_{6}A_{7}A_{8}} + 384 R_{A_{1}}^{A_{5}} R^{A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{2}}^{A_{7}} R^{A_{8}}R_{A_{4}A_{8}A_{5}A_{7}} - 96 R_{A_{1}A_{2}}^{A_{5}A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{3}A_{5}}^{A_{7}A_{8}}R_{A_{4}A_{6}A_{7}A_{8}} .$$
(A.3)

Similarly, $\epsilon_{10}\epsilon_{10}R^4$ structure denotes

$$\epsilon_{10}\epsilon_{10}R^{4} = \epsilon_{10}^{C_{1}C_{2}A_{1}\cdots A_{8}} \epsilon_{10 C_{1}C_{2}B_{1}\cdots B_{8}} R^{B_{1}B_{2}}{}_{A_{1}A_{2}} \cdots R^{B_{7}B_{8}}{}_{A_{7}A_{8}}$$

$$= -2 \cdot 8! \, \delta^{A_{1}}{}_{[B_{1}} \cdots \delta^{A_{8}}{}_{B_{8}]} R^{B_{1}B_{2}}{}_{A_{1}A_{2}} \cdots R^{B_{7}B_{8}}{}_{A_{7}A_{8}},$$
(A.4)

where, in the last line, we have used the usual identity for contracting Levi-Civita tensors (cf. equation (A.1) in [36]). Contracting all the Kronecker delta functions with the Riemann tensors in (A.4), the $\epsilon_{10}\epsilon_{10}R^4$ structure can be written as:

$$\begin{split} &\frac{1}{8}\epsilon_{10}\epsilon_{10}R^{4} = \\ &-12\left(R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{1}A_{2}A_{3}A_{4}}\right)^{2} \\ &-192\,R_{A_{1}}^{A_{5}}A_{a}^{A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{2}}^{A_{7}}A_{a}^{A_{8}}R_{A_{5}A_{7}A_{6}A_{8}} \\ &+192\,R_{A_{1}A_{2}A_{3}}^{A_{5}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{4}}^{A_{6}A_{7}A_{8}}R_{A_{5}A_{6}A_{7}A_{8}} \\ &-24\,R_{A_{1}A_{2}}^{A_{5}A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{3}}^{A_{7}}A_{5}^{A_{8}}R_{A_{5}A_{6}A_{7}A_{8}} \\ &-24\,R_{A_{1}A_{2}}^{A_{5}A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{3}}^{A_{7}}A_{5}^{A_{8}}R_{A_{4}A_{8}A_{6}A_{7}} \\ &+384\,R_{A_{1}A_{2}}^{A_{5}A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{3}}^{A_{7}}A_{5}^{A_{8}}R_{A_{4}A_{7}A_{6}A_{8}} - 768\,R^{A_{1}A_{2}}R_{A_{1}}^{A_{3}}A_{4}^{A_{7}}R_{A_{4}A_{5}A_{6}}R_{A_{1}A_{2}A_{3}A_{4}}R_{A_{4}}^{A_{5}A_{6}A_{7}} \\ &+384\,R^{A_{1}A_{2}}R_{A_{1}}^{A_{3}A_{4}A_{5}}R_{A_{2}}^{A_{6}}A_{4}^{A_{7}}R_{A_{4}A_{7}A_{6}A_{8}} - 768\,R^{A_{1}A_{2}}R_{A_{3}A_{4}A_{5}A_{6}}R^{A_{3}A_{4}A_{5}A_{6}} \\ &-1536\,R^{A_{1}A_{2}}R_{A_{1}}^{A_{3}A_{4}A_{5}}R_{A_{2}}^{A_{6}}A_{4}^{A_{7}}R_{A_{3}A_{7}A_{5}A_{6}} + 768\,R^{A_{1}A_{2}}R^{A_{3}A_{4}}R_{A_{1}}^{A_{2}}A_{6}^{A}R_{A_{3}A_{4}A_{5}A_{6}} \\ &-168\,R_{A_{1}}^{A_{3}}R^{A_{1}A_{2}}R_{A_{2}}^{A_{4}A_{5}A_{6}}R_{A_{3}A_{4}A_{5}A_{6}} - 32\,R\,R_{A_{1}A_{2}}^{A_{5}A_{6}}R^{A_{1}A_{2}A_{3}A_{4}}R_{A_{3}A_{4}A_{5}A_{6}} \\ &+128\,R\,R_{A_{1}}^{A_{5}}R_{A_{2}}^{A_{4}A_{5}A_{6}}R_{A_{2}A_{4}A_{5}A_{6}} - 768\,R^{A_{1}A_{2}}R^{A_{3}A_{4}}R_{A_{1}}^{A_{5}}A_{3}^{A_{6}}}R_{A_{2}A_{4}A_{5}A_{6}} \\ &-384\,R^{A_{1}A_{2}}R^{A_{3}A_{4}}R_{A_{1}A_{3}}^{A_{5}A_{6}}R_{A_{2}A_{4}A_{5}A_{6}} + 1536\,R_{A_{1}}^{A_{3}}R^{A_{1}A_{2}}R_{A_{4}A_{5}}R_{A_{2}A_{4}A_{5}} \\ &-384\,R^{A_{1}A_{2}}R^{A_{3}A_{4}}R_{A_{1}A_{3}}R^{A_{1}A_{2}}R_{A_{4}A_{5}}R_{A_{2}A_{4}A_{5}} \\ &-384\,R^{A_{1}A_{2}}R^{A_{3}A_{4}}R_{A_{1}A_{3}}R^{A_{1}A_{2}}R_{A_{4}}} \\ &-384\,R^{A_{1}A_{2}}R^{A_{3}A_{4}}R_{A_{1}A_{3}}R^{A_{1}A_{2}}R_{A_{4}}} \\ &-384\,R^{A_{1}A_{2}}R^{A_{3}A_{4}}R_{A_{1}A_{3}}R^{A_{1}A_{2}}R_{A_{4}}} \\ &-256\,R\,R_{A_{1}}^{A_{3}}R^{A_{1}A_{2}}R_{A_{2}A_{4}}} \\ &+96\,R^{2}\,R_{$$

¹⁷Notice the change of prefactor with respect to equation (B.3) of [36], so that, after expressing this structure as contractions of four Riemann tensors, the result matches that in [95].

Evaluating on Mathematica each of the contractions in (A.3) and (A.5) on the $AdS_2 \times S^8$ ansatz in (4.1), adding them up, and substituting $v = v_2$ and $\beta = v_2/v_1$, we get to the result reported in equation (4.28).

As we used in section 4.4, both the $\hat{t}_8 \hat{t}_8 \hat{R}^4$ and the $\frac{1}{24} \epsilon_{11} \epsilon_{11} \hat{R}^4$ structures appearing in the M-theory effective action in equation (4.31) look exactly the same as the terms we just discussed. This is, once written in terms of contractions of 11d Riemman tensors, they precisely reduce to the ones in (A.3) and (A.5). Evaluating these contractions on the 11d metric ansatz in (4.34) with Mathematica, we recover the result in (4.36).

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- E. Palti, The Swampland: Introduction and Review, Fortsch. Phys. 67 (2019) 1900037 [arXiv:1903.06239] [INSPIRE].
- M. van Beest, J. Calderón-Infante, D. Mirfendereski and I. Valenzuela, Lectures on the Swampland Program in String Compactifications, Phys. Rept. 989 (2022) 1
 [arXiv:2102.01111] [INSPIRE].
- [3] M. Graña and A. Herráez, The Swampland Conjectures: A Bridge from Quantum Gravity to Particle Physics, Universe 7 (2021) 273 [arXiv:2107.00087] [INSPIRE].
- [4] N.B. Agmon, A. Bedroya, M.J. Kang and C. Vafa, Lectures on the string landscape and the Swampland, arXiv:2212.06187 [INSPIRE].
- [5] G. Dvali, Black Holes and Large N Species Solution to the Hierarchy Problem, Fortsch. Phys. 58 (2010) 528 [arXiv:0706.2050] [INSPIRE].
- [6] G. Dvali and C. Gomez, Quantum Information and Gravity Cutoff in Theories with Species, Phys. Lett. B 674 (2009) 303 [arXiv:0812.1940] [INSPIRE].
- [7] G. Dvali and D. Lüst, Evaporation of Microscopic Black Holes in String Theory and the Bound on Species, Fortsch. Phys. 58 (2010) 505 [arXiv:0912.3167] [INSPIRE].
- [8] G. Dvali and C. Gomez, Species and Strings, arXiv:1004.3744 [INSPIRE].
- [9] G. Dvali, C. Gomez and D. Lüst, Black Hole Quantum Mechanics in the Presence of Species, Fortsch. Phys. 61 (2013) 768 [arXiv:1206.2365] [INSPIRE].
- [10] H. Ooguri and C. Vafa, On the Geometry of the String Landscape and the Swampland, Nucl. Phys. B 766 (2007) 21 [hep-th/0605264] [INSPIRE].
- [11] M. Etheredge et al., Sharpening the Distance Conjecture in diverse dimensions, JHEP 12 (2022) 114 [arXiv:2206.04063] [INSPIRE].
- [12] D. van de Heisteeg, C. Vafa, M. Wiesner and D.H. Wu, Bounds on Field Range for Slowly Varying Positive Potentials, arXiv:2305.07701 [INSPIRE].
- [13] M. Etheredge et al., Running Decompactification, Sliding Towers, and the Distance Conjecture, arXiv:2306.16440 [INSPIRE].
- [14] J. Calderón-Infante, A. Castellano, A. Herráez and L.E. Ibáñez, Entropy Bounds and the Species Scale Distance Conjecture, arXiv:2306.16450 [INSPIRE].

- [15] S. Ferrara, R. Kallosh and A. Strominger, N = 2 extremal black holes, Phys. Rev. D 52 (1995) R5412 [hep-th/9508072] [INSPIRE].
- S. Ferrara and R. Kallosh, Supersymmetry and attractors, Phys. Rev. D 54 (1996) 1514
 [hep-th/9602136] [INSPIRE].
- S. Ferrara and R. Kallosh, Universality of supersymmetric attractors, Phys. Rev. D 54 (1996) 1525 [hep-th/9603090] [INSPIRE].
- [18] S. Ferrara, G.W. Gibbons and R. Kallosh, Black holes and critical points in moduli space, Nucl. Phys. B 500 (1997) 75 [hep-th/9702103] [INSPIRE].
- [19] Y. Hamada, M. Montero, C. Vafa and I. Valenzuela, Finiteness and the swampland, J. Phys. A 55 (2022) 224005 [arXiv:2111.00015] [INSPIRE].
- [20] R. Angius et al., At the end of the world: Local Dynamical Cobordism, JHEP 06 (2022) 142 [arXiv:2203.11240] [INSPIRE].
- [21] R. Angius, J. Huertas and A.M. Uranga, Small black hole explosions, JHEP 06 (2023) 070 [arXiv:2303.15903] [INSPIRE].
- [22] G. Buratti, J. Calderón and A.M. Uranga, Transplanckian axion monodromy!?, JHEP 05 (2019) 176 [arXiv:1812.05016] [INSPIRE].
- [23] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, Swampland Conjectures for Strings and Membranes, JHEP 02 (2021) 006 [arXiv:2006.15154] [INSPIRE].
- [24] G. Buratti, M. Delgado and A.M. Uranga, Dynamical tadpoles, stringy cobordism, and the SM from spontaneous compactification, JHEP 06 (2021) 170 [arXiv:2104.02091] [INSPIRE].
- [25] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, The EFT stringy viewpoint on large distances, JHEP 09 (2021) 197 [arXiv:2104.05726] [INSPIRE].
- [26] G. Buratti, J. Calderón-Infante, M. Delgado and A.M. Uranga, Dynamical Cobordism and Swampland Distance Conjectures, JHEP 10 (2021) 037 [arXiv:2107.09098] [INSPIRE].
- [27] R. Angius, M. Delgado and A.M. Uranga, Dynamical Cobordism and the beginning of time: supercritical strings and tachyon condensation, JHEP 08 (2022) 285 [arXiv:2207.13108]
 [INSPIRE].
- [28] M. Delgado, M. Montero and C. Vafa, Black holes as probes of moduli space geometry, JHEP 04 (2023) 045 [arXiv:2212.08676] [INSPIRE].
- [29] J. Huertas and A.M. Uranga, Aspects of dynamical cobordism in AdS/CFT, JHEP 08 (2023) 140 [arXiv:2306.07335] [INSPIRE].
- [30] N. Cribiori, D. Lüst and G. Staudt, Black hole entropy and moduli-dependent species scale, Phys. Lett. B 844 (2023) 138113 [arXiv:2212.10286] [INSPIRE].
- [31] N. Cribiori, D. Lüst and C. Montella, Species entropy and thermodynamics, JHEP 10 (2023)
 059 [arXiv:2305.10489] [INSPIRE].
- [32] G. Dvali and M. Redi, Black Hole Bound on the Number of Species and Quantum Gravity at LHC, Phys. Rev. D 77 (2008) 045027 [arXiv:0710.4344] [INSPIRE].
- [33] P.A. Cano, P.F. Ramírez and A. Ruipérez, The small black hole illusion, JHEP 03 (2020) 115 [arXiv:1808.10449] [INSPIRE].
- [34] P.A. Cano, A. Murcia, P.F. Ramírez and A. Ruipérez, On small black holes, KK monopoles and solitonic 5-branes, JHEP 05 (2021) 272 [arXiv:2102.04476] [INSPIRE].

- [35] B. Heidenreich, M. Reece and T. Rudelius, The Weak Gravity Conjecture and Emergence from an Ultraviolet Cutoff, Eur. Phys. J. C 78 (2018) 337 [arXiv:1712.01868] [INSPIRE].
- [36] T.W. Grimm, E. Palti and I. Valenzuela, Infinite Distances in Field Space and Massless Towers of States, JHEP 08 (2018) 143 [arXiv:1802.08264] [INSPIRE].
- [37] B. Heidenreich, M. Reece and T. Rudelius, Emergence of Weak Coupling at Large Distance in Quantum Gravity, Phys. Rev. Lett. 121 (2018) 051601 [arXiv:1802.08698] [INSPIRE].
- [38] P. Corvilain, T.W. Grimm and I. Valenzuela, The Swampland Distance Conjecture for Kähler moduli, JHEP 08 (2019) 075 [arXiv:1812.07548] [INSPIRE].
- [39] F. Marchesano and L. Melotti, EFT strings and emergence, JHEP 02 (2023) 112 [arXiv:2211.01409] [INSPIRE].
- [40] A. Castellano, A. Herráez and L.E. Ibáñez, The emergence proposal in quantum gravity and the species scale, JHEP 06 (2023) 047 [arXiv:2212.03908] [INSPIRE].
- [41] R. Blumenhagen, A. Gligovic and A. Paraskevopoulou, The emergence proposal and the emergent string, JHEP 10 (2023) 145 [arXiv:2305.10490] [INSPIRE].
- [42] R. Blumenhagen, N. Cribiori, A. Gligovic and A. Paraskevopoulou, *Demystifying the Emergence Proposal*, arXiv:2309.11551 [INSPIRE].
- [43] R. Blumenhagen, N. Cribiori, A. Gligovic and A. Paraskevopoulou, *The Emergent M-theory Limit*, arXiv:2309.11554 [INSPIRE].
- [44] S.D. Mathur, The Fuzzball proposal for black holes: An Elementary review, Fortsch. Phys. 53 (2005) 793 [hep-th/0502050] [INSPIRE].
- [45] I. Bena and N.P. Warner, Black holes, black rings and their microstates, Lect. Notes Phys. 755 (2008) 1 [hep-th/0701216] [INSPIRE].
- [46] B.D. Chowdhury and A. Virmani, Modave Lectures on Fuzzballs and Emission from the D1-D5 System, in the proceedings of the 5th Modave Summer School in Mathematical Physics, Modave, Belgium, August 17–21 (2009) [arXiv:1001.1444] [INSPIRE].
- [47] D. van de Heisteeg, C. Vafa, M. Wiesner and D.H. Wu, Moduli-dependent Species Scale, arXiv:2212.06841 [INSPIRE].
- [48] D. van de Heisteeg, C. Vafa and M. Wiesner, Bounds on Species Scale and the Distance Conjecture, Fortsch. Phys. 71 (2023) 2300143 [arXiv:2303.13580] [INSPIRE].
- [49] B. de Wit, J. Hoppe and H. Nicolai, On the Quantum Mechanics of Supermembranes, Nucl. Phys. B 305 (1988) 545 [INSPIRE].
- [50] M.R. Douglas, D.N. Kabat, P. Pouliot and S.H. Shenker, *D-branes and short distances in string theory*, *Nucl. Phys. B* 485 (1997) 85 [hep-th/9608024] [INSPIRE].
- [51] T. Banks, W. Fischler, S.H. Shenker and L. Susskind, M theory as a matrix model: A Conjecture, Phys. Rev. D 55 (1997) 5112 [hep-th/9610043] [INSPIRE].
- [52] N. Itzhaki, J.M. Maldacena, J. Sonnenschein and S. Yankielowicz, Supergravity and the large N limit of theories with sixteen supercharges, Phys. Rev. D 58 (1998) 046004 [hep-th/9802042] [INSPIRE].
- [53] T. Ortiz, H. Samtleben and D. Tsimpis, Matrix model holography, JHEP 12 (2014) 096 [arXiv:1410.0487] [INSPIRE].

- [54] MONTE CARLO STRING/M-THEORY (MCSMC) collaboration, *Precision test of gauge/gravity duality in D0-brane matrix model at low temperature*, *JHEP* **03** (2023) 071 [arXiv:2210.04881] [INSPIRE].
- [55] A. Sinha and N.V. Suryanarayana, Extremal single-charge small black holes: Entropy function analysis, Class. Quant. Grav. 23 (2006) 3305 [hep-th/0601183] [INSPIRE].
- [56] A. Sen, Extremal black holes and elementary string states, Mod. Phys. Lett. A 10 (1995) 2081
 [hep-th/9504147] [INSPIRE].
- [57] A. Dabholkar and S. Nampuri, Quantum black holes, Lect. Notes Phys. 851 (2012) 165 [arXiv:1208.4814] [INSPIRE].
- [58] T. Mohaupt, Black hole entropy, special geometry and strings, Fortsch. Phys. 49 (2001) 3 [hep-th/0007195] [INSPIRE].
- [59] N. Cribiori et al., Large and small non-extremal black holes, thermodynamic dualities, and the Swampland, JHEP 10 (2022) 093 [arXiv:2202.04657] [INSPIRE].
- [60] N. Cribiori, A. Gnecchi, D. Lüst and M. Scalisi, On the correspondence between black holes, domain walls and fluxes, JHEP 05 (2023) 033 [arXiv:2302.03054] [INSPIRE].
- [61] N. Arkani-Hamed, L. Motl, A. Nicolis and C. Vafa, The String landscape, black holes and gravity as the weakest force, JHEP 06 (2007) 060 [hep-th/0601001] [INSPIRE].
- [62] C. Cheung and G.N. Remmen, Infrared Consistency and the Weak Gravity Conjecture, JHEP 12 (2014) 087 [arXiv:1407.7865] [INSPIRE].
- [63] C. Cheung, J. Liu and G.N. Remmen, Proof of the Weak Gravity Conjecture from Black Hole Entropy, JHEP 10 (2018) 004 [arXiv:1801.08546] [INSPIRE].
- [64] N. Arkani-Hamed, Y.-T. Huang, J.-Y. Liu and G.N. Remmen, Causality, unitarity, and the weak gravity conjecture, JHEP 03 (2022) 083 [arXiv:2109.13937] [INSPIRE].
- [65] E. Lauria and A. Van Proeyen, $\mathcal{N} = 2$ Supergravity in D = 4, 5, 6 Dimensions, arXiv:2004.11433 [D0I:10.1007/978-3-030-33757-5].
- [66] J.M. Maldacena, A. Strominger and E. Witten, Black hole entropy in M theory, JHEP 12 (1997) 002 [hep-th/9711053] [INSPIRE].
- [67] G. Lopes Cardoso, B. de Wit and T. Mohaupt, Corrections to macroscopic supersymmetric black hole entropy, Phys. Lett. B 451 (1999) 309 [hep-th/9812082] [INSPIRE].
- [68] G. Lopes Cardoso, B. de Wit and T. Mohaupt, Deviations from the area law for supersymmetric black holes, Fortsch. Phys. 48 (2000) 49 [hep-th/9904005] [INSPIRE].
- [69] G. Lopes Cardoso, B. de Wit and T. Mohaupt, Macroscopic entropy formulae and nonholomorphic corrections for supersymmetric black holes, Nucl. Phys. B 567 (2000) 87
 [hep-th/9906094] [INSPIRE].
- [70] B. Pioline, Lectures on black holes, topological strings and quantum attractors, Class. Quant. Grav. 23 (2006) S981 [hep-th/0607227] [INSPIRE].
- [71] A. Sen, Black hole entropy function and the attractor mechanism in higher derivative gravity, JHEP 09 (2005) 038 [hep-th/0506177] [INSPIRE].
- [72] S.-J. Lee, W. Lerche and T. Weigand, Emergent strings, duality and weak coupling limits for two-form fields, JHEP 02 (2022) 096 [arXiv:1904.06344] [INSPIRE].
- [73] S.-J. Lee, W. Lerche and T. Weigand, Emergent strings from infinite distance limits, JHEP 02 (2022) 190 [arXiv:1910.01135] [INSPIRE].

- [74] R. Gopakumar and C. Vafa, M theory and topological strings. 1, hep-th/9809187 [INSPIRE].
- [75] R. Gopakumar and C. Vafa, M theory and topological strings. 2, hep-th/9812127 [INSPIRE].
- [76] M.B. Green, M. Gutperle and P. Vanhove, One loop in eleven-dimensions, Phys. Lett. B 409 (1997) 177 [hep-th/9706175] [INSPIRE].
- [77] H. Ooguri and C. Vafa, Summing up D instantons, Phys. Rev. Lett. 77 (1996) 3296
 [hep-th/9608079] [INSPIRE].
- [78] M. Gutperle, A note on heterotic / type I-prime duality and D0-brane quantum mechanics, JHEP 05 (1999) 007 [hep-th/9903010] [INSPIRE].
- [79] A. Collinucci, P. Soler and A.M. Uranga, Non-perturbative effects and wall-crossing from topological strings, JHEP 11 (2009) 025 [arXiv:0904.1133] [INSPIRE].
- [80] C. Petersson, P. Soler and A.M. Uranga, *D-instanton and polyinstanton effects from type I' D0-brane loops*, JHEP 06 (2010) 089 [arXiv:1001.3390] [INSPIRE].
- [81] N. Piazzalunga and A.M. Uranga, *M-theory interpretation of the real topological string*, *JHEP* 08 (2014) 054 [arXiv:1405.6019] [INSPIRE].
- [82] E. Gonzalo, L.E. Ibáñez and Á.M. Uranga, Modular symmetries and the swampland conjectures, JHEP 05 (2019) 105 [arXiv:1812.06520] [INSPIRE].
- [83] I. Antoniadis, S. Ferrara, R. Minasian and K.S. Narain, R⁴ couplings in M and type II theories on Calabi-Yau spaces, Nucl. Phys. B 507 (1997) 571 [hep-th/9707013] [INSPIRE].
- [84] E. Witten, String theory dynamics in various dimensions, Nucl. Phys. B 443 (1995) 85 [hep-th/9503124] [INSPIRE].
- [85] N. Cribiori and D. Lüst, A Note on Modular Invariant Species Scale and Potentials, Fortsch. Phys. 71 (2023) 2300150 [arXiv:2306.08673] [INSPIRE].
- [86] D. Berenstein, A toy model for the AdS / CFT correspondence, JHEP 07 (2004) 018 [hep-th/0403110] [INSPIRE].
- [87] V. Balasubramanian, J. de Boer, V. Jejjala and J. Simon, The Library of Babel: On the origin of gravitational thermodynamics, JHEP 12 (2005) 006 [hep-th/0508023] [INSPIRE].
- [88] P. Erdös and J. Lehner, The distribution of the number of sumands in the partitions of a positive integer, Duke Math. J. 8(2) (1941) 335.
- [89] R.E. Crandall, Probability density for partitions of n with k parts, [https://www.reed.edu/physics/faculty/wheeler/documents/Quantum%20Mechanics/Miscell aneous%20Essays/Partitions%20&%20Separability/Crandall%20on%20Partitions.pdf].
- [90] G. Skezeres, Some asymptotic formulae in the theory of partitions (II), Quart. J. of Math. (Oxford) 4 (1953) 96.
- [91] T. Anous et al., Areas and entropies in BFSS/gravity duality, SciPost Phys. 8 (2020) 057 [arXiv:1911.11145] [INSPIRE].
- [92] D. Lüst, E. Palti and C. Vafa, AdS and the Swampland, Phys. Lett. B 797 (2019) 134867
 [arXiv:1906.05225] [INSPIRE].
- [93] M.B. Green and J.H. Schwarz, Supersymmetrical String Theories, Phys. Lett. B 109 (1982) 444 [INSPIRE].
- [94] D.J. Gross and E. Witten, Superstring Modifications of Einstein's Equations, Nucl. Phys. B 277 (1986) 1 [INSPIRE].

- [95] M.T. Grisaru and D. Zanon, σ Model Superstring Corrections to the Einstein-hilbert Action, Phys. Lett. B 177 (1986) 347 [INSPIRE].
- [96] M.T. Grisaru, A.E.M. van de Ven and D. Zanon, Two-Dimensional Supersymmetric Sigma Models on Ricci Flat Kähler Manifolds Are Not Finite, Nucl. Phys. B 277 (1986) 388 [INSPIRE].
- [97] M.T. Grisaru, A.E.M. van de Ven and D. Zanon, Four Loop Divergences for the N = 1 Supersymmetric Nonlinear Sigma Model in Two-Dimensions, Nucl. Phys. B 277 (1986) 409
 [INSPIRE].
- [98] T.W. Grimm, K. Mayer and M. Weissenbacher, *Higher derivatives in Type II and M-theory on Calabi-Yau threefolds*, *JHEP* **02** (2018) 127 [arXiv:1702.08404] [INSPIRE].
- [99] M.B. Green and P. Vanhove, D instantons, strings and M theory, Phys. Lett. B 408 (1997) 122 [hep-th/9704145] [INSPIRE].
- [100] J.G. Russo and A.A. Tseytlin, One loop four graviton amplitude in eleven-dimensional supergravity, Nucl. Phys. B 508 (1997) 245 [hep-th/9707134] [INSPIRE].
- [101] J. Polchinski, M theory and the light cone, Prog. Theor. Phys. Suppl. 134 (1999) 158 [hep-th/9903165] [INSPIRE].
- [102] L. Susskind, Holography in the flat space limit, AIP Conf. Proc. 493 (1999) 98
 [hep-th/9901079] [INSPIRE].
- [103] N.A. Nekrasov, On the size of a graviton, in the proceedings of the 14th International Workshop on High-Energy Physics and Quantum Field Theory (QFTHEP 99), Moscow, Russian Federation, May 27 – June 02 (1999), p. 401–411 [hep-th/9909213] [INSPIRE].
- [104] T. Crisford, G.T. Horowitz and J.E. Santos, Testing the Weak Gravity Cosmic Censorship Connection, Phys. Rev. D 97 (2018) 066005 [arXiv:1709.07880] [INSPIRE].
- [105] G.T. Horowitz and J.E. Santos, Further evidence for the weak gravity cosmic censorship connection, JHEP 06 (2019) 122 [arXiv:1901.11096] [INSPIRE].
- [106] I.R. Klebanov and M.J. Strassler, Supergravity and a confining gauge theory: Duality cascades and chi SB resolution of naked singularities, JHEP 08 (2000) 052 [hep-th/0007191] [INSPIRE].
- [107] J.T. Liu and R. Minasian, Higher-derivative couplings in string theory: dualities and the B-field, Nucl. Phys. B 874 (2013) 413 [arXiv:1304.3137] [INSPIRE].
- [108] M.B. Green, M. Gutperle and H.H. Kwon, Light cone quantum mechanics of the eleven-dimensional superparticle, JHEP 08 (1999) 012 [hep-th/9907155] [INSPIRE].
- [109] K. Peeters, J. Plefka and S. Stern, Higher-derivative gauge field terms in the M-theory action, JHEP 08 (2005) 095 [hep-th/0507178] [INSPIRE].
- [110] J. Maldacena and L. Susskind, Cool horizons for entangled black holes, Fortsch. Phys. 61 (2013) 781 [arXiv:1306.0533] [INSPIRE].
- [111] G. Dvali and C. Gomez, Black Hole's Quantum N-Portrait, Fortsch. Phys. 61 (2013) 742
 [arXiv:1112.3359] [INSPIRE].
- [112] T. Banks, W. Fischler, I.R. Klebanov and L. Susskind, Schwarzschild black holes from matrix theory, Phys. Rev. Lett. 80 (1998) 226 [hep-th/9709091] [INSPIRE].
- [113] I.R. Klebanov and L. Susskind, Schwarzschild black holes in various dimensions from matrix theory, Phys. Lett. B 416 (1998) 62 [hep-th/9709108] [INSPIRE].
- [114] G.T. Horowitz and E.J. Martinec, Comments on black holes in matrix theory, Phys. Rev. D 57 (1998) 4935 [hep-th/9710217] [INSPIRE].

- [115] M. Li, Matrix Schwarzschild black holes in large N limit, JHEP 01 (1998) 009
 [hep-th/9710226] [INSPIRE].
- [116] T. Banks, W. Fischler, I.R. Klebanov and L. Susskind, Schwarzschild black holes in matrix theory. 2, JHEP 01 (1998) 008 [hep-th/9711005] [INSPIRE].
- [117] L. Susskind, Matrix theory black holes and the Gross-Witten transition, hep-th/9805115
 [INSPIRE].
- [118] M. Li and E.J. Martinec, Probing matrix black holes, hep-th/9801070 [INSPIRE].
- [119] A.V. Smilga, Comments on thermodynamics of supersymmetric matrix models, Nucl. Phys. B 818 (2009) 101 [arXiv:0812.4753] [INSPIRE].
- T. Wiseman, On black hole thermodynamics from super Yang-Mills, JHEP 07 (2013) 101 [arXiv:1304.3938] [INSPIRE].
- [121] D. Andriot, Bumping into the Species Scale with the Scalar Potential, Fortsch. Phys. 71 (2023)
 2300139 [arXiv:2305.07480] [INSPIRE].
- [122] A. Adams et al., Causality, analyticity and an IR obstruction to UV completion, JHEP 10 (2006) 014 [hep-th/0602178] [INSPIRE].
- [123] B. Bellazzini, C. Cheung and G.N. Remmen, Quantum Gravity Constraints from Unitarity and Analyticity, Phys. Rev. D 93 (2016) 064076 [arXiv:1509.00851] [INSPIRE].
- [124] Y. Hamada, T. Noumi and G. Shiu, Weak Gravity Conjecture from Unitarity and Causality, Phys. Rev. Lett. 123 (2019) 051601 [arXiv:1810.03637] [INSPIRE].
- [125] R. Blumenhagen, N. Cribiori, C. Kneissl and A. Makridou, Dynamical cobordism of a domain wall and its companion defect 7-brane, JHEP 08 (2022) 204 [arXiv:2205.09782] [INSPIRE].
- [126] R. Blumenhagen, N. Cribiori, C. Kneissl and A. Makridou, Dimensional Reduction of Cobordism and K-theory, JHEP 03 (2023) 181 [arXiv:2208.01656] [INSPIRE].
- [127] I. Basile, S. Raucci and S. Thomée, Revisiting Dudas-Mourad Compactifications, Universe 8 (2022) 544 [arXiv:2209.10553] [INSPIRE].
- [128] R. Blumenhagen, C. Kneissl and C. Wang, Dynamical Cobordism Conjecture: solutions for end-of-the-world branes, JHEP 05 (2023) 123 [arXiv:2303.03423] [INSPIRE].
- [129] E. Dudas and J. Mourad, Brane solutions in strings with broken supersymmetry and dilaton tadpoles, Phys. Lett. B 486 (2000) 172 [hep-th/0004165] [INSPIRE].
- [130] R. Blumenhagen and A. Font, Dilaton tadpoles, warped geometries and large extra dimensions for nonsupersymmetric strings, Nucl. Phys. B 599 (2001) 241 [hep-th/0011269] [INSPIRE].
- [131] E. Dudas, J. Mourad and C. Timirgaziu, Time and space dependent backgrounds from nonsupersymmetric strings, Nucl. Phys. B 660 (2003) 3 [hep-th/0209176] [INSPIRE].
- [132] E. Dudas, G. Pradisi, M. Nicolosi and A. Sagnotti, On tadpoles and vacuum redefinitions in string theory, Nucl. Phys. B 708 (2005) 3 [hep-th/0410101] [INSPIRE].
- [133] S. Raucci, On codimension-one vacua and string theory, Nucl. Phys. B 985 (2022) 116002
 [arXiv:2206.06399] [INSPIRE].
- [134] R. Antonelli and I. Basile, Brane annihilation in non-supersymmetric strings, JHEP 11 (2019)
 021 [arXiv:1908.04352] [INSPIRE].
- [135] A. Mininno and A.M. Uranga, Dynamical tadpoles and Weak Gravity Constraints, JHEP 05 (2021) 177 [arXiv:2011.00051] [INSPIRE].

- [136] I. Basile, On String Vacua without Supersymmetry: brane dynamics, bubbles and holography, Ph.D. thesis, Scuola Normale Superiore, Pisa, Italy (2020) [arXiv:2010.00628] [INSPIRE].
- [137] I. Basile, Supersymmetry breaking, brane dynamics and Swampland conjectures, JHEP 10 (2021) 080 [arXiv:2106.04574] [INSPIRE].
- [138] J. Mourad and A. Sagnotti, A 4D IIB flux vacuum and supersymmetry breaking. Part I. Fermionic spectrum, JHEP 08 (2022) 301 [arXiv:2206.03340] [INSPIRE].
- [139] G.T. Horowitz, M. Kolanowski, G.N. Remmen and J.E. Santos, Extremal Kerr Black Holes as Amplifiers of New Physics, Phys. Rev. Lett. 131 (2023) 091402 [arXiv:2303.07358] [INSPIRE].
5

Global summary of results and discussion

In this thesis, we investigated the implications of implementing dynamical change of the topology of spacetime, as should be allowed by a background independent theory of QG. In a first part, we studied dynamical realizations of the processes predicted by the cobordism conjecture, in the context of the Swampland Program. This allowed us to characterize the singular extended objects that play the role of implementing these processes. Then, we studied how bordism groups can detect global gauge/diffeomorphism anomalies in theories of quantum gravity. Finally we explored the different ways in which (singular and smooth) extended objects in string theory can, or not, be used as probes of UV physics. Let us briefly summarize the key points and the main results of each of the seven articles.

In Chapter 2, we studied various aspects of dynamical realizations of cobordisms to nothing. These are solutions to the equations of motion that run along a spacetime dimension and where the compact space pinches of at finite distance in spacetime.

In the first article, we considered string theory vacua with tadpoles for dynamical fields in the scalar potential and uncovered universal features of the resulting spacetime dependent solutions. We argued that the solutions can extend only a finite spacetime distance Δ away in the spacetime dimensions over which the fields vary, scaling as $\Delta^n \sim \mathcal{T}$ with the strength of the tadpole \mathcal{T} . We showed that naive singularities arising at this distance scale are physically replaced by ends of spacetime, related to the cobordism defects of the swampland cobordism conjecture and involving stringy ingredients like orientifold planes and branes, or exotic variants thereof. We illustrated these phenomena in large classes of examples, including $AdS_5 \times T^{1,1}$ with 3-form fluxes, 10d massive IIA, M-theory on K3, the 10d non-supersymmetric USp(32) strings, and type IIB compactifications with 3-form fluxes and/or magnetized D-branes. We also described a 6d string model whose tadpole triggers spontaneous compactification to a semirealistic 3-family MSSM-like particle physics model.

In the second article, we continued studying spacetime-dependent solutions to string theory models with tadpoles for dynamical fields. We confirmed in many new explicit string theoretic examples, that such solutions have necessarily finite extent in spacetime, and are capped off by boundaries at a finite distance, in a dynamical realization of the Cobordism Conjecture. We showed that as the configuration approaches these cobordism walls of nothing, the scalar fields run off to infinite distance in moduli space, allowing to explore the implications of the Swampland Distance Conjecture. We uncovered new interesting scaling relations linking the moduli space distance and the SDC tower scale to spacetime geometric quantities, such as the distance to the wall and the scalar curvature. We show that walls at which scalars remain at finite distance in moduli space correspond to domain walls separating different (but bordant) theories/vacua; this still applies even if the scalars reach finite distance singularities in moduli space, such as conifold points. We illustrated our ideas with explicit examples in massive IIA theory, M-theory on CY threefolds, and 10d non-supersymmetric strings. In 4d $\mathcal{N} = 1$ theories, our framework reproduces a recent proposal to explore the SDC using 4d string-like solutions.

In the third article, we explain why one observes these universal patterns in the dynamical realization of cobordism, as spacetime dependent solutions of Einstein gravity coupled to scalars containing such end-of-the-world 'branes'. The latter appear in effective theory as a singularity at finite spacetime distance at which scalars go off to infinite field space distance. We provided a local description near the end-of-the-world branes, in which the solutions simplify dramatically and are characterized in terms of a critical exponent, which controls the asymptotic profiles of fields and the universal scaling relations among the spacetime distance to the singularity, the field space distance, and the spacetime curvature. The analysis did not rely on supersymmetry. We studied many explicit examples of such Local Dynamical Cobordisms in string theory, including 10d massive IIA, the 10d nonsupersymmetric USp(32) theory, Bubbles of Nothing, $4d \mathcal{N} = 1$ cosmic string solutions, the Klebanov-Strassler throat, Dp-brane solutions, brane configurations related to the D1/D5systems, and small black holes. Our framework encompasses diverse recent setups in which scalars diverge at the core of defects, by regarding them as suitable end-of-the-world branes. We also explored the interplay of Local Dynamical Cobordisms with the Distance Conjecture and other swampland constraints.

Finally, in the fourth article, we took a step in the direction of studying timedependent dynamical cobordisms. We described timelike linear dilaton backgrounds of supercritical string theories as time-dependent Dynamical Cobordisms in string theory, with their spacelike singularity as a boundary defining the beginning of time. We proposed and provided compelling evidence that its microscopic interpretation corresponds to a region of (a strong coupling version of) closed tachyon condensation. We argued that this beginning of time is closely related to (and shares the same scaling behaviour as) the bubbles of nothing obtained in a weakly coupled background with lightlike tachyon condensation. As an intermediate result, we also provided the description of the latter as lightlike Dynamical Cobordism.

In Chapter 3, we studied Dai-Freed anomalies in non-supersymmetric string theories. These anomalies arise in theories with dynamical gravity and where spacetime topology change should be allowed.

More specifically, in the article, we studied the three tachyon-free non-supersymmetric string theories in ten dimensions. These theories provide a handle on quantum gravity away from the supersymmetric lamppost. Despite having been around for decades, they have not been shown to be fully consistent; although local anomalies cancel due to versions of the Green-Schwarz mechanism, there could be global anomalies, not cancelled by the Green-Schwarz mechanism, that could become fatal pathologies. We computed the twisted string bordism groups that control these anomalies via the Adams spectral sequence, showing that they vanish completely in two out of three cases (Sugimoto and $SO(16)^2$) and showing a partial vanishing also in the third (Sagnotti 0'B model). We also computed lower-dimensional bordism groups of the non-supersymmetric string theories, which are of interest to the classification of branes in these theories via the Cobordism Conjecture. Using an anomaly inflow argument, we were able to propose a worldvolume content of the $SO(16)^2$ NS5-brane. As a byproduct of our techniques and analysis, we also reprove that the outer \mathbb{Z}_2 automorphism swapping the two E_8 factors in the supersymmetric heterotic string is also non-anomalous.

Lastly, in Chapter 6, we considered different types of black holes that could be used as probes of UV physics.

In the first article, we argued that supersymmetric BPS black holes can act as efficient finite energy probes of the moduli space geometry thanks to the attractor mechanism. We focused on 4d $\mathcal{N} = 2$ compactifications and captured aspects of the effective field theory near the attractor values in terms of physical quantities far away in moduli space. Furthermore, we illustrated how the standard distance in moduli space can be related asymptotically to the black hole mass. We also computed a measure of the resolution with which BPS black holes of a given mass can distinguish far away points in the moduli space. The black hole probes may lead to a deeper understanding of the Swampland constraints on the geometry of the moduli space.

In the second paper, we instead considered singular black holes, that take us to the limits of the domain of validity of the whole notion of an EFT of QG, the species scale Λ_s . We used black holes of vanishing horizon area (small black holes) in effective field theories coupled to quantum gravity to shed light on how the three different physical manifestations of the species scale Λ_s relate to each other. (i) Near the small black hole core, a scalar field runs to infinite distance in moduli space, a regime in which the Swampland Distance Conjecture predicts a tower of exponentially light states, which lower Λ_s . (ii) We integrate out modes in the tower and generate via Emergence a set of higher derivative corrections, showing that Λ_s is the scale at which such terms become relevant. (iii) Finally, higher derivative terms modify the black hole solution and grant it a non-zero, species scale sized stretched horizon of radius Λ_s^{-1} , showcasing the species scale as the size of the smallest possible black hole describable in the effective theory. We present explicit 4d examples of small black holes in 4d $\mathcal{N} = 2$ supergravity, and the 10d example of type IIA D0-branes. The emergence of the species scale horizon for D0-branes requires a non-trivial interplay of different 8-derivative terms in type IIA and M-theory, providing a highly non-trivial check of our unified description of the different phenomena associated to the species scale.

6

Resumen global de los resultados y discusión

En esta tesis, investigamos las implicaciones de implementar el cambio dinámico de la topología del espacio tiempo, como debería permitirlo una teoría independiente del fondo de gravedad cuántica (QG). En una primera parte, estudiamos realizaciones dinámicas de los procesos predichos por la conjetura de cobordismo, en el contexto del Programa Swampland. Esto nos permitió caracterizar los objetos extendidos singulares que desempeñan el papel de implementar estos procesos. Luego, estudiamos cómo los grupos de bordismo pueden detectar anomalías globales de calibre/difeomorfismo en teorías de gravedad cuántica. Finalmente, exploramos las diferentes formas en que los objetos extendidos (singulares y suaves) en la teoría de cuerdas pueden, o no, ser utilizados como sondas de la física ultravioleta (UV). Resumamos brevemente los puntos clave y los principales resultados de cada uno de los siete artículos.

En el Capítulo 2, estudiamos varios aspectos de las realizaciones dinámicas de cobordismos hacia la nada. Estas son soluciones a las ecuaciones de movimiento que recorren una dimensión del espacio tiempo y donde el espacio compacto se estrangula a una distancia finita en el espacio tiempo.

En el primer artículo, consideramos vacíos de la teoría de cuerdas con "tadpoles" para campos dinámicos en el potencial escalar y descubrimos características universales de las soluciones resultantes dependientes del espacio tiempo. Argumentamos que las soluciones solo pueden extenderse una distancia finita en el espacio tiempo Δ en las dimensiones del espacio tiempo sobre las cuales los campos varían, escalando como $\Delta^n \sim \mathcal{T}$ con la fuerza del "tadpole" \mathcal{T} . Mostramos que las singularidades ingenuas que surgen a esta escala de distancia son físicamente reemplazadas por extremos del espacio tiempo, relacionados con los defectos de cobordismo de la conjetura de cobordismo y que involucran ingredientes de la teoría de cuerdas como planos orientables y branas, o variantes exóticas de estos. Ilustramos estos fenómenos en grandes clases de ejemplos, incluyendo AdS₅ × $T^{1,1}$ con flujos de 3-formas, IIA masivo en 10d, M-teoría en K3, las cuerdas USp(32) no supersimétricas en 10d, y compactificaciones de tipo IIB con flujos de 3-formas y/o branas magnetizadas. También describimos un modelo de cuerda en 6d cuyo "tadpole" desencadena una compactificación espontánea a un modelo de física de partículas MSSM-like semi-realista de 3 familias.

En el segundo artículo, continuamos estudiando soluciones dependientes del espacio tiempo para modelos de teoría de cuerdas con "tadpoles" para campos dinámicos. Confirmamos en muchos nuevos ejemplos explícitos de teoría de cuerdas, que tales soluciones tienen necesariamente una extensión finita en el espacio tiempo, y están tapadas por límites a una distancia finita, en una realización dinámica de la Conjetura de Cobordismo. Mostramos que a medida que la configuración se acerca a estas paredes de cobordismo hacia la nada, los campos escalares se alejan a una distancia infinita en el espacio de móduli, permitiendo explorar las implicaciones de la Conjetura de Distancia. Descubrimos nuevas relaciones de escala interesantes que vinculan la distancia en el espacio de móduli y la escala de la torre SDC a cantidades geométricas del espacio tiempo, como la distancia a la pared y la curvatura escalar. Mostramos que las paredes en las que los escalares permanecen a una distancia finita en el espacio de móduli corresponden a paredes de dominio que separan diferentes teorías/vacíos (pero bordantes); esto sigue aplicándose incluso si los escalares alcanzan singularidades a una distancia finita en el espacio de móduli, como puntos de conifolds. Ilustramos nuestras ideas con ejemplos explícitos en la teoría IIA masiva, M-teoría en tresfolds CY, y cuerdas no supersimétricas en 10d. En teorías 4d $\mathcal{N} = 1$, nuestro marco reproduce una propuesta reciente para explorar la SDC usando soluciones tipo cuerda en 4d.

En el tercer artículo, explicamos por qué se observan estos patrones universales en la realización dinámica del cobordismo, como soluciones dependientes del espacio tiempo de la gravedad de Einstein acoplada a escalares que contienen tales 'branas' al final del mundo. Estas últimas aparecen en la teoría efectiva como una singularidad a una distancia finita en el espacio tiempo en la que los escalares se alejan a una distancia infinita en el espacio de campo. Proporcionamos una descripción local cerca de las branas al final del mundo, en la que las soluciones se simplifican drásticamente y se caracterizan en términos de un exponente crítico, que controla los perfiles asintóticos de los campos y las relaciones de escala universales entre la distancia en el espacio tiempo a la singularidad, la distancia en el espacio de campo y la curvatura del espacio tiempo. El análisis no se basó en la supersimetría. Estudiamos muchos ejemplos explícitos de tales Cobordismos Dinámicos Locales en la teoría de cuerdas, incluyendo IIA masivo en 10d, la teoría USp(32) no supersimétrica en 10d, Burbujas de la Nada, soluciones de cuerdas cósmicas 4d \mathcal{N} = 1, la "throat" de Klebanov-Strassler, soluciones de Dp-branas, configuraciones de branas relacionadas con los sistemas D1/D5, y pequeños agujeros negros. Nuestro marco abarca diversas configuraciones recientes en las que los escalares divergen en el núcleo de los defectos, al considerarlos como branas adecuadas al final del mundo. También exploramos la interacción de los Cobordismos Dinámicos Locales con la Conjetura de Distancia y otras restricciones del Swampland.

Finalmente, en el cuarto artículo, dimos un paso en la dirección de estudiar cobordismos dinámicos dependientes del tiempo. Describimos fondos de dilatón lineal timelike de teorías de cuerdas supercríticas como Cobordismos Dinámicos dependientes del tiempo en la teoría de cuerdas, con su singularidad espacial como un límite que define el comienzo del tiempo. Propusimos y proporcionamos evidencia convincente de que su interpretación microscópica corresponde a una región de (una versión de acoplamiento fuerte de) condensación cerrada de taquiones. Argumentamos que este comienzo del tiempo está estrechamente relacionado con (y comparte el mismo comportamiento de escala que) las burbujas de la nada obtenidas en un fondo débilmente acoplado con condensación de taquiones null. Como resultado intermedio, también proporcionamos la descripción de esta última como Cobordismo Dinámico lightlike. En el Capítulo 3, estudiamos las anomalías Dai-Freed en teorías de cuerdas no supersimétricas. Estas anomalías surgen en teorías con gravedad dinámica y donde se debería permitir el cambio en la topología del espacio tiempo.

Más específicamente, en el artículo, estudiamos las tres teorías de cuerdas no supersimétricas sin taquiones en diez dimensiones. Estas teorías proporcionan un manejo de la gravedad cuántica lejos del poste de luz supersimétrico. A pesar de haber estado presentes durante décadas, no se ha demostrado que sean completamente consistentes; aunque las anomalías locales se cancelan debido a versiones del mecanismo de Green-Schwarz, podría haber anomalías globales, no canceladas por el mecanismo de Green-Schwarz, que podrían convertirse en patologías fatales. Calculamos los grupos de bordismo de cuerdas torcidas que controlan estas anomalías a través de la secuencia espectral de Adams, mostrando que desaparecen por completo en dos de los tres casos (Sugimoto y $SO(16)^2$) y mostrando un desvanecimiento parcial también en el tercero (modelo 0'B de Sagnotti). También calculamos grupos de bordismo de dimensiones inferiores de las teorías de cuerdas no supersimétricas, que son de interés para la clasificación de branas en estas teorías mediante la Conjetura de Cobordismo. Utilizando un argumento de flujo de anomalías, pudimos proponer un contenido de volumen mundial de la NS5-brana de $SO(16)^2$. Como subproducto de nuestras técnicas y análisis, también demostramos que el automorfismo externo \mathbb{Z}_2 que intercambia los dos factores E_8 en la cuerda heterótica supersimétrica también es no anómalo.

Por último, en el Capítulo 6, consideramos diferentes tipos de agujeros negros que podrían utilizarse como sondas de la física UV.

En el primer artículo, argumentamos que los agujeros negros BPS supersimétricos pueden actuar como eficientes sondas de energía finita de la geometría del espacio de móduli gracias al mecanismo atrayente. Nos centramos en compactificaciones $\mathcal{N} = 2$ de 4d y capturamos aspectos de la teoría efectiva cerca de los valores atrayentes en términos de cantidades físicas lejanas en el espacio de móduli. Además, ilustramos cómo la distancia estándar en el espacio de móduli puede relacionarse asintóticamente con la masa del agujero negro. También calculamos una medida de la resolución con la que los agujeros negros BPS de una masa dada pueden distinguir puntos lejanos en el espacio de móduli. Las sondas de agujeros negros pueden llevar a una comprensión más profunda de las restricciones del Swampland en la geometría del espacio de móduli.

En el segundo artículo, en cambio, consideramos agujeros negros singulares, que nos llevan a los límites del dominio de validez de toda la noción de una EFT de QG, la escala de especies Λ_s . Utilizamos agujeros negros con área de horizonte que tiende a cero (agujeros negros pequeños) en teorías efectivas acopladas a la gravedad cuántica para arrojar luz sobre cómo se relacionan entre sí las tres diferentes manifestaciones físicas de la escala de especies Λ_s . (i) Cerca del núcleo del agujero negro pequeño, un campo escalar se aleja a una distancia infinita en el espacio de móduli, un régimen en el que la Conjetura de Distancia predice una torre de estados exponencialmente ligeros, que disminuyen Λ_s . (ii) Integramos los modos en la torre y generamos mediante la Emergencia un conjunto de correcciones de derivadas superiores, mostrando que Λ_s es la escala en la que dichos términos se vuelven relevantes. (iii) Finalmente, los términos de derivadas superiores modifican la solución del agujero negro y le otorgan un horizonte estirado de tamaño de escala de especies de radio Λ_s^{-1} , mostrando la escala de especies como el tamaño del agujero negro posible más pequeño que se puede describir en la teoría efectiva. Presentamos ejemplos explícitos de agujeros negros pequeños en supergravedad $\mathcal{N} = 2$ de 4d y el ejemplo de 10d de las D0-branas de tipo IIA. La emergencia del horizonte de escala de especies para las D0-branas requiere una interacción no trivial de diferentes términos de 8 derivadas en tipo IIA y M-teoría, proporcionando una verificación altamente no trivial de nuestra descripción unificada de los diferentes fenómenos asociados a la escala de especies.